FILE NAME

"Report.VCP.v00674.2004-09-23.RAWP_Site2A_2B.pdf"

NOTE

Contains folded images. Images must be copied in their entirety.

Images may contain color.

- Documents must be returned in the same arrangement in which they were received.

Please return electronic copies to Sally Dewes

The Port Authority of New York and New Jersey

Site Investigation and Conceptual Remedial Workplan Site 2A/2B (VCP 00674-2) HHMT - Port Ivory Facility September 2004

> 40 Western Avenue, Staten Island, New York HMM 232952

DOM STATE OF NEW YORK DEPARTMENT OF HEALTH

Flanigan Square, 547 River Street, Troy, New York 12180-2216

Antonia C. Novello, M.D., M.P.H., Dr.P.H. Commissioner Dennis P. Whalen Executive Deputy Commissioner

December 22, 2004

Mr. Tom Gibbons NYS Department of Environmental Conservation Division of Environmental Remediation 625 Broadway – 11th Floor Albany, NY 12233-70

Re:

HHMT – Port Ivory Facility Operable Unit 2 (OU-2) Site # V006742 Staten Island, Richmond County

Dear Mr. Gibbons:

I have reviewed the September 2004*Site Investigation and Conceptual Remedial Workplan* and the December 2004 *Substructure Soil Gas and Ambient and Indoor Air Sampling Plan* for the above-referenced site and offer the following comments:

Site Investigation and Conceptual Remedial Workplan

- 1) Section 10.2, page 247 states "...detected in sediment samples in excess of the Human Health Bioaccumulation criteria during the Remedial Investigation." Provide further explanation of this criteria.
- 2) Section 6.4, page 172 replace "SVG" with SGV to reflect Standard Guidance Values.
- 3) Section 11, page 251, states that during the removal of UST -7, petroleum impacted soil was encountered during the excavation, please indicate whether end point samples were collected and analyzed.

Substructure Soil Gas and Ambient and Indoor Air Sampling Plan

- 1) Collect **two** subsurface soil gas samples from Building 40.
- 2) Please note that flow rates for purging and collecting soil gas samples must not exceed 0.2 liters per minute.

If you have any comments, please contact me at (518) 402-7860.

Sincerely,

Julia M. Guastella Public Health Specialist I Bureau of Environmental Exposure Investigation

cc: G. Litwin / G. Laccetti / file B. Devine – MARO J. Prudhomme – NYCDOH R. Cozzy – NYSDEC D. Walsh – NYSDEC, Reg. 2 D. Greeley – NYCDEP

P:\Bureau\Sites\Region_2\RICHMOND\V006742\OU-2_SIRW.doc



Infrastructure and Environment 27 Bleeker Street Millburn, NJ 07041-1008 T 973.379.3400 www.hatchmott.com



September 22, 2004

Hatch Mott

MacDonald

Mr. Thomas Gibbons Project Manager New York State Department of Environmental Conservation Remedial Bureau B, Section D Division of Environmental Remediation 625 Broadway Albany, New York 12233-9773

RE: Report Submission HHMT – Port Ivory Facility 40 Western Avenue State Island, New York ID #'s V00615, V00674, V00675

Dear Mr. Gibbons:

By way of this correspondence, Hatch Mott MacDonald (HMM), on behalf of the Port Authority of New York and New Jersey (the Port Authority) respectfully submits the initial reports required pursuant to the Voluntary Cleanup Program (VCP) Agreements executed for the HHMT-Port Ivory Facility. As per the above referenced VCP Agreements, please find enclosed four copies (3 bound and 1 unbound) of the *Revised - Site Investigation and Conceptual Remedial Workplan* for Site 1(VCP Site 00615-2) dated September 2004 and the *Site Investigation and Conceptual Remedial Workplan* for Site 2A/2B (VCP Site 00674-2) dated September 2004. As per the agreement, copies are being forwarded concurrently to the New York State Department of Health as well as to the Project Attorney at the New York State Department of Environmental Conservation (NYSDEC). As per our letter dated September 15, 2004, the summary report for Site 3 will be submitted on September 30, 2004.

Please note, the NYSDEC Marine Resources Program is currently involved with a wetland restoration project situated south of the HHMT-Port Ivory Facility. The NYSDEC has contacted the Port Authority with regard to certain actions, which may be undertaken at Bridge Creek. The NYSDEC's project includes sampling at Bridge Creek and the Port Authority has requested information generated from the NYSDEC's project. HMM/Port Authority will be reviewing analytical data and other information, as made available by NYSDEC, to determine if the findings have any bearing on conditions present at the HHMT-Port Ivory Facility. Pertinent information related to the Port Authority's compliance with the VCP Program will be forwarded to your attention.

P:\232952wmd\Operable Unit Reports\NYSDEC Correspondence\Report Submission.doc

Hatch Mott MacDonald

If you have questions or need further information, please contact me at 973.912.2475.

Very truly yours,

Hatch Mott MacDonald

ety Kohlociat

Tennifer Nulty Kohlsaat Associate T 973.912.2475 F 973.912.2400 jennifer.kohlsaat@hatchmott.com

cc:

G. Laccetti, NYSDOH (w/enc. via US Mail) D. D'Ambrosio, NYSDEC (w/enc. via US Mail) E. Aldrich, PANYNJ (w/enc. via US Mail)

Mr. Gibbons, Page 2, September 22, 2004











í.

1.0	Exe	ecutive Summary	1
2.0	Intr	roduction	
	2.1	Objective	6
	2.2	Site Location and Description	6
		*	
3.0	Bac	kground	23
	3.1	Site 2A/2B History	23
	3.2	Hydrogeologic Setting	25
		3.2.1 Soils	26
		3.2.2 Groundwater	26
4.0	Env	vironmental Site Investigation Workplan	29
	4.1	Previous Environmental Investigation Efforts	
	4.2	Sampling Plan	54
		4.2.1 USTs	58
		4.2.2 Precipitate at Bridge Creek	61
		4.2.3 Previously Identified Soil and Groundwater Contamination (P&G AO	Cs)63
		4.2.4 Railroad Tracks and Siding	65
		4.2.5 Staining	66
		4.2.6 Pits and Drains	
		4.2.7 Former Structures	
		4.2.8 Historic Fill Material	
		4.2.9 Groundwater	
	4.3	QA / QC and Health and Safety	
			· · · · · · · · · · · · · · · · · · ·
5.0	SL -	- Field Investigation	
5.0	5.1	Pre-Investigation Field Activities	
	5.2	GPR/EM Survey – Potential UST Areas	
	5.3	Soil Boring Construction and Sampling	
		5.3.1 Potential UST Areas	149
	, i i i i i i i i i i i i i i i i i i i	5.3.2 Previously Identified AOCs	149
	-	(Areas B, E, F2, Railyard)	. (<u>)</u>
		5.3.3 Railroad Tracks and Sidings	
		5.3.4 Surface Staining	151
اھي. ا	ан Т	5.3.5 Pits and Drains	151
	an N	5.3.6 Former Structures.	151
		5.3.7 Historic Fill Material	152
	5.4	Monitoring Well Installation	153
	5.5	Groundwater Sampling	154
	5.6	Surface Water and Sediment Inspection and Sampling	155

 $\left\{ \cdot \right\}$

 $A_{2}^{(i)}$

i





SI R	esults	. 155
6.1	Site Hydrogeology	155
	6.1.1 Soil	156
	6.1.2 Groundwater	157
6.2	GPR/EM – Potential UST Areas	161
6.3	Soil Analytical Data	161
	6.3.1 Volatile Organic Compounds	162
	6.3.2 Semi-Volatile Organic Compounds	162
	6.3.3 Polychlorinated Biphenyls	164
	6.3.4 Pesticides	164
	6.3.5 Metals	164
	6.3.6 Cyanide and Total Phenolics	166
	6.3.7 Petroleum Hydrocarbons/Oil and Grease	168
	6.3.8 pH	168
6.4	Groundwater Analytical Data	172
	6.4.1 Volatile Organic Compounds	172
	6.4.2 Semi-Volatile Organic Compounds	172
	6.4.3 Polychlorinated Biphenyls	184
	6.4.4 Pesticides	184
	6.4.5 Metals	184
	6.4.6 Cyanide and Total Phenolics	184
	6.4.7 Petroleum Hydrocarbons/Oil and Grease	185
	6.4.8 pH	185
	Necución /Conclusions	105
		196
7.1	Soll	100
1.2	Groundwater	101
7.3	Bridge Creek – Surface water/Sediment	101
/.4		191
Ren	nedial Investigation Workplan	. 193
8.1	Proposed Actions - Potential UST Areas	193
8.2	Proposed Actions - Potential Petroleum Impacted Areas	195
8.3	Proposed Actions - Bridge Creek	196
	SI R 6.1 6.2 6.3 6.4 6.4 SI E 7.1 7.2 7.3 7.4 Ren 8.1 8.2 8.3	SI Results 6.1 Site Hydrogeology 6.1.2 Groundwater 6.2 GPR/EM – Potential UST Areas. 6.3 Soil Analytical Data 6.3.1 Volatile Organic Compounds 6.3.2 Semi-Volatile Organic Compounds 6.3.3 Polychlorinated Biphenyls 6.3.4 Pesticides. 6.3.5 Metals. 6.3.6 Cyanide and Total Phenolics. 6.3.7 Petroleum Hydrocarbons/Oil and Grease 6.3.8 pH. 6.4 Groundwater Analytical Data 6.4.1 Volatile Organic Compounds 6.4.2 Semi-Volatile Organic Compounds 6.4.3 Polychlorinated Biphenyls 6.4.4 Pesticides. 6.4.5 Metals. 6.4.6 Cyanide and Total Phenolics 6.4.7 Petroleum Hydrocarbons/Oil and Grease 6.4.8 pH. 6.4 Cyanide and Total Phenolics 6.4.7 Petroleum Hydrocarbons/Oil and Grease 6.4.8 pH. SI Discussion /Conclusions. 7.1 Soil 7.2 Groundwater 7.3 Bridge Creek – Surface Water/Sediment. 7.4 Historic Fill Remedial Investigation Workplan 8.1 Proposed Actions - Potential UST Areas. 8.2 Proposed Actions - Potential Detroleum Impacted Area



9.0	RI – Field Investigation	
	9.1 RI Sampling Procedures/Methodology	
	9.1.1 Potential UST Areas	
	9.1.2 Areas B-2 and B-3	
	9.1.3 Area GW-14	
	9.1.4 Area OP-1	
	9.1.5 Area A-5	
	9.1.6 Area FS-1B	
	9.1.7 Area GW-1	
	9.1.8 Area MW-10D	
	9.1.9 Bridge Creek - Surface Water/Sediment	
	9.2 RI-Analytical Results (Soil)	
	9.2.1 Volatile Organic Compounds	
	9.2.2 PAH Compounds	
	9.3 RI – Analytical Results (Sediment/Surface Water)	
	9.3.1 Sediment	
	9.3.1.1 Volatile Organic Compounds	
	9.3.1.2 Semi-Volatile Organic Compounds	234
	9.3.1.3 Pesticides and Polychlorinated Biphenyls	234
	9.3.1.4 Metals	234
	9.3.1.5 Petroleum Hydrocarbons/Oil and Grease	235
	9.3.1.6 Cyanide and Total Phenolics	235
	9.3.1.7 pH	235
	9.3.2 Surface Water	235
	9.3.2.1 Volatile Organic Compounds	235
	9.3.2.2 Semi-Volatile Organic Compounds	242
	9.3.2.3 Pesticides and Polychlorinated Biphenyls	242
	9.3.2.4 Metals	242
	9.3.2.5 Petroleum Hydrocarbons/Oil and Grease	242
	9.3.2.6 Cyanide and Total Phenolics	
	9.3.2.7 pH	242
	9.4 RI Summary	242
10.0	Exposure Assessment	
	10.1 Exposure Setting	
	10.2 Nature of On-Site Contaminants	245
	10.3 Potential Exposure Points and Routes	248
	10.4 Receptor Populations	249
	10.5 Exposure Assessment Summary	
11.0	UST Removal Efforts	250





12.0	Summary of Remedial Actions		
	12.1	Proposed Remedial Actions	251
	12.2	Completed Remedial Actions	252
		12.2.1 Area B-3/B-2	257
		12.2.2 Area GW-14	257
		12.2.3 Area OP-1	
		12.2.4 Area A-5	
		12.2.5 Area FS-1B	264
13.0	Sumn	nary of Proposed Site Development Actions	267
14.0	Conc	lusions	281

List of Appendices

Appendix A -Geophysical Report

Appendix B – Soil Boring and Well Logs, Field Screening Results, Site Investigation

Appendix C - Groundwater Sampling Logs, Site Investigation

Appendix D – Development Information



List of Figures

(all figures are provided in text by order of reference)

Figure 1 – Site Location Map 4
Figure 2 – Site 2A/2B Layout
Figure 3 – Existing Building Locations – Year 2000
Figure 4 – Cross Section Location Map 27
Figure 5 - Cross-Section
Figure 6 - Areas of Concern (AOC) Map 62
Figure 7 - Site Investigation Sample Locations Map, Site 2A/2B
Figure 8 – Potentiometric Surface Contour Map, Shallow Wells
Figure 9 - Water Table Contour Map160
Figure 10 - Distribution of VOCs and SVOCs in Soil,
Site 2A/2B, Site Investigation163
Figure 11 - Distribution of Pesticides and PCBs, in Soil,
Site 2A/2B, Site Investigation
Figure 12 - Distribution of Select Metals in Soil, Site 2A/2B, Site Investigation 167
Figure 13 - Distribution of TPHC in Soil, Site 2A/2B, Site Investigation
Figure 14 - Distribution of Oil and Grease in Soil, Site 2A/2B, Site Investigation 170
Figure 15 - Distribution of pH in Soil, Site 2A/2B, Site Investigation
Figure 16 - Distribution of Metals, VOCs and SVOCs in Groundwater, Site 2A/2B,
Site Investigation173
Figure 17 - Proposed Remedial Investigation AOCs, Petroleum/Non-Petroleum Areas
and Potential UST Areas194
Figure 18 - Remedial Investigation Sample Locations and Analytical Results Map,
Site 2A/2B
Figure 19 - Analytical Results, Sediment Site 2A/2B, Remedial Investigation
Figure 20 - Proposed Hot Spot Excavation Areas, Remedial Action, Site 2A/2B 253
Figure 21 - Remedial Action - Hot Spot Excavation Limits and Confirmation Sample
Locations254
Figure 22 – Surcharge Monitoring Well Locations Map and Analytical Data



List of Tables

(all tables are provided in text in order of first reference)

Table 1 - Summary of Site Buildings10
Table 2 - Summary of Historical Environmental Reports and Information
Table 3 - Summary of Environmental Database Listings 55
Table 4 - Summary of SI Investigative Actions and Sampling
Tables 5A-5E - Soil Analytical Results, Site 2A/2B, Site Investigation
Table 5A – Volatile Organic Compounds 84
Table 5B – Semi–Volatile Organic Compounds96
Table 5C - Pesticides and PCBs108
Table 5D – Metals 119
Table 5E – TPHC, Oil and Grease, pH, Cyanide & Total Phenolics
Tables 6A–6E – Groundwater Analytical Results – Site Investigation
Table 6A – Volatile Organic Compounds174
Table 6B - Semi-Volatile Organic Compounds176
Table 6C - Pesticides and PCBs178
Table 6D - Metals
Table 6E – TPHC, Oil and Grease, pH, Cyanide & Total Phenolics
Table 7 - Summary of Remedial Investigation Sampling, Site 2A/2B 197
Table 8 - Summary of RI Field Observations, Site 2A/2B 202
Tables 9A & 9B - Soil Analytical Results, Site 2A/2B, Remedial Investigation
Table 9A – Volatile Organic Compounds 220
Table 9B – PAH Compounds 222
Tables 10A-E - Sediment Analytical Results - RI, Site 2A/2B
Table 10A – Volatile Organic Compounds 228
Table 10B – Semi–Volatile Organic Compounds 229
Table 10C - Pesticides and PCBs, Sediment231
Table 10D – Metals 232
Table 10E - TPHC, Oil and Grease, pH, Cyanide & Total Phenolics



List of Tables (continued)

Tables 11A–E – Surface Water Analytical Results – RI, Site 2A/2B			
Table 11A - Volatile Organic Compounds 236			
Table 11B - Semi-Volatile Organic Compounds 237			
Table 11C - Pesticides and PCBs239			
Table 11D - Metals			
Table 11E - TPHC, Oil and Grease, pH, Cyanide & Total Phenolics			
Tables 12 – Summary of Remedial Action Sampling 255			
Tables 13A & 13B - Soil Analytical Results, Site 2A/2B, Area GW-14, Remedial Action			
Table 13A – Volatile Organic Compounds, Area GW–14			
Table 13B – PAH Compounds, Area GW–14			
Tables 14A & 14B – Soil Analytical Results, Site 2A/2B, Area OP-1, Remedial Action			
Table 14A - Volatile Organic Compounds, Area OP-1			
Table 14B - PAH Compounds, Area OP-1 262			
Tables 15A & 15B – Soil Analytical Results, Site 2A/2B, Area A-5, Remedial Action			
Table 15A – Volatile Organic Compounds, Area A–5			
Table 15B – PAH Compounds, Area A–5266			
Tables 16A & 16B – Soil Analytical Results, Site 2A/2B, Area FS-1, Remedial Action			
Table 16A – Volatile Organic Compounds, Area FS–1			
Table 16B - PAH Compounds, Area FS-1			
Tables 17 A–C – Surcharge Pilot Study – Groundwater Results			
Table 17A - Groundwater Results, BTEX 272			
Table 17B - Groundwater Results, Acid Extractables 273			
Table 17C- Groundwater and Surface Water Results, TAL Metals and pH 275			

P:\232952wmd\Operable Unit Reports\Operable Unit 2\Post VCP Revisions\Report\July 2004 Site 2 TOC_new 8-24-04.doc



1.0 EXECUTIVE SUMMARY

The Port Authority of New York and New Jersey (Port Authority) is currently redeveloping the former Procter & Gamble (P&G) Port Ivory Facility located at 40 Western Avenue in Staten Island, Richmond County New York. On behalf of the Port Authority, Hatch Mott MacDonald (HMM) has performed assessment and investigation activities to characterize site conditions and delineate historic fill material and contaminants in environmental media at the site. These efforts have been undertaken based upon a commercial/industrial end use for Port Ivory. Specifically, the Port Authority intends to utilize the 40 Western Avenue facility, now known as the Howland Hook Marine Terminal (HHMT) - Port Ivory Facility for a container terminal and intermodal facility in conjunction with the adjacent Howland Hook Marine Terminal.

As part of the overall site redevelopment, the Port Authority entered into the New York State Department of Environmental Conservation (NYSDEC) Voluntary Cleanup Program (VCP) in July 2004 (VCP Agreement Site V-00674-2, VCP Index Number W2-0986-02-04). The Port Authority's objective for entering into the VCP program with NYSDEC is to address the presence of contamination due to prior site activities unrelated to the Port Authority. To accommodate the redevelopment schedule for the northwestern portion of the HHMT-Port Ivory Facility, the NYSDEC has agreed to expedite the review of information pertaining to certain portions of the HHMT-Port Ivory Facility. Thus, the Port Authority agreed to address the HHMT-Port Ivory Facility as four individual "Sites" and present assessment, investigation and remedial action information/documentation for each individual Site. This report addresses Site 2A/2B, which consists of the eastern and southern portion of Block 1400, Lot 1 and the southern portion of Block 1338, Lot 1.

Overall, the assessment and investigation activities undertaken at Site 2A/2B have revealed the presence of historic fill material; and several contaminants at relatively low concentrations in samples collected from soil and groundwater at Site 2A/2B. The presence of the historic fill material and contaminants in environmental media is consistent with the highly urbanized and historically industrial nature of the site and surrounding area. Subsequent investigative efforts successfully delineated potential petroleum-impacted areas and accessible petroleum-impacted areas have been addressed through source area excavation and removal; the removal of petroleum-impact soil was performed in conjunction with ongoing site redevelopment activities, prior to entering the VCP Program. Based on the results of the assessment, site investigation and remedial investigation activities, the Port Authority's planned usage of

1

Site 2A/2B Report

 $\mathbf{2}$



the site as an intermodal facility and container terminal is not inconsistent with the levels of contamination noted to be present in site soil and groundwater. To address structural issues presented by the presence of fill material, the Port Authority's development plan includes a process of surcharging portions of Site 1 and Site 2A/2B, with geotechnically suitable clean fill, to achieve a stable base for future construction. Thus, the proposed development plan will result in the use of engineering controls (an environmental cap), which will minimize potential impacts to human health and the environment. As part of the geotechnical site preparation work, the Port Authority performed a surcharge pilot study at an area of Site 1 in 2002/2203. The study included the systematic placement of soil/fill over an area measuring approximately 75 feet by 75 feet and the measurement of settlement. The pilot study included a review of potential environmental impacts to groundwater and Bridge Creek. The environmental evaluation performed as part of the pilot study did not reveal any adverse impacts as a result of the compaction process. Although the pilot study did not reveal the presence of adverse impacts to groundwater or Bridge Creek, additional monitoring efforts are proposed to confirm the findings of the pilot study.



2.0 INTRODUCTION

At the request of the Port Authority of New York and New Jersey (Port Authority), Hatch Mott MacDonald (HMM) performed assessment, investigation and remedial actions at the now former Procter & Gamble (P&G) Port Ivory Facility located at 40 Western Avenue in/Staten Island, Richmond County New York. The location of the subject site is presented on Figure 1. The initial phase of the project consisted of the performance of a Phase I Environmental Site Assessment (ESA) and a supplemental file review of the entire facility. The Phase I ESA was performed in accordance with the Standard Practice for Environmental Site Assessment E1527: Phase I ESA Process, as set forth by the American Society for Testing and Materials (ASTM). Performance of the ESA and the supplemental file review identified numerous recognized and/or potential environmental conditions, as defined by ASTM E1527, at the above referenced site. Upon completion of the Phase I ESA and the supplemental file review, the Port Authority requested that HMM prepare an Environmental Site Investigation Workplan (ESIW) to evaluate the Areas of Concern (AOCs) and subsequently, to implement the proposed Site Investigation (SI) activities for the entire facility. The purpose of the SI was to assess current (year 2000) environmental conditions at this facility.

Based on the findings of the SI and subsequent to the Port Authority's purchase of the P&G Port Ivory Facility at 40 Western Avenue Site), HMM prepared a remedial investigation workplan (RIW) designed to evaluate potential issues related to petroleum, which were identified through prior assessment and investigation. The RIW also included review of nine potential UST areas; four of the potential UST Areas were identified at Site 2A and no potential UST Areas were identified at Site 2B. The objective of the delineation was to resolve these issues in preparation for redevelopment of the entire P&G Port Ivory Facility; upon transfer of ownership the property/facility was designated as the Howland Hook Marine Terminal (HHMT) – Port Ivory Facility. The remedial investigation (RI) of petroleum/non-petroleum areas was performed during the spring/summer of 2002. Although building demolition and other construction related activities impeded the installation of test pits as part of proposed RI efforts for potential UST Areas (UST1, UST3, UST4 and UST7) at Site 2A, further assessment of some of the areas was accomplished through the performance of certain demolition actions including removal of concrete pads and building footings. The potential UST Areas were evaluated in the fall/winter 2002 through the Spring 2003. As described later in this report, the activities at the UST1,

3

site tr





UST3, and UST4 Area did not reveal the presence of any USTs and the activities at the UST7 Area identified the presence of two USTs. The RI successfully delineated the horizontal and vertical extent of petroleum/non-petroleum oils in soil accessible areas of Site 2A/2B. Based on the field screening and analytical results from the RI, hot-spot excavation has been identified as the appropriate remedial action (RA) for the identified petroleum/non-petroleum-impacted areas. The remainder of the areas will be addressed through site redevelopment. To accommodate site redevelopment efforts, hot-spot excavation at certain potential petroleum-impacted areas has been implemented and information pertaining to excavation and post-excavation (confirmation) sampling is presented herein (See Section 12). The remaining residual contaminants will be addressed through site redevelopment, which will include engineering controls such as the placement of environmental caps (soil, gravel, asphalt, concrete, etc.). To the extent feasible, the Port Authority has performed assessment, investigation and remediation activities in accordance with NYSDEC requirements and is committed to redeveloping this site in a manner which ensures human health and the environment given the proposed site usage

As part of the overall site redevelopment, the Port Authority entered into the New York State Department Environment Conservation (NYSDEC) Voluntary Cleanup Program (VCP) in July 2004. The Port Authority's objective for entering into the VCP program with NYSDEC was to address the identified contamination due to prior site activities unrelated to the Port Authority. Prior to entering the VCP program, the Port Authority performed assessment, investigation and remedial activities to address the subject site (known after December 2000 as the HHMT – Port Ivory Facility) in its entirety. During discussions with the NYSDEC, the Port Authority identified that it had established different redevelopment schedules for the individual site parcels. As a result, the NYSDEC has agreed to expedite the review of information pertaining to certain portions of this site and the Port Authority agreed to address the HHMT-Port Ivory Facility as four individual sites, thereby, presenting assessment, investigation and remedial action information/documentation for each individual site. The four units or sites have been defined as follows: Site 1 consists of the northwestern portion of Block 1400, Lot 1; Site 2A/2B consists of the eastern and southern portions, known as Site 2A and a small area of the southern portion of Block 1338, Lot 1 referred to as Site 2B; Site 3 consists of the northern portion of Block 1338, Lot 1; and Future Site 4 consists of Block 1309, Lot 10. Block 1309, Lot 10 has been designated as "Site 4" on mapping provided in the VCP Agreements for Sites 1, 2A/2B and 3. However, the Port Authority has not executed a VCP Agreement for Block 1309, Lot 10. As such, the Block 1309, Lot 10 parcel will be referenced as "Future Site 4" for this report.

This report addresses Site 2A/2B pursuant to the July VCP Agreement (VCP Agreement Site V-00674-2). The limits of Site 2A/2B in relation to the remainder of the HHMT-Port Ivory Facility, including the designations of the three other Sites are presented in Figure 2. The map also depicts the easements located on Site 2A/2B as well as the remainder of the site.

2.1 Objective

The objective of this report is to describe the actions undertaken at this site to characterize, delineate and address contamination present in environmental media at Site 2A/2B. This report also includes a summary of analytical data as well as field observations generated through the performance of sampling and other evaluation efforts. Analytical data from the site and remedial investigations, as well as from remedial actions performed to date, is presented in tabular form and pertinent information is provided on maps and described in applicable text sections. This report also includes a summary of remedial actions that were undertaken at certain petroleum-impacted areas. These efforts were performed prior to entering the VCP Program and were done to proactively address areas as part of active site demolition activities. Please note, to facilitate review of the assessment, investigation and remedial actions described herein, an overview of site history focusing on Site 2A/2B has been included in Section 3.1 of this report. The specific sampling and remedial actions described in this report were developed based on a predetermined end-use for the HHMT-Port Ivory Facility including Site 2A/2B. The Port Authority is redeveloping this former industrial site for use as an intermodal/container storage facility with Site 2A functioning primarily as the intermodal component of the facility.

2.2 Site Location and Description

As previously stated, the overall subject site is located at 40 Western Avenue, Staten Island, Richmond County, New York and is comprised of the three following tax blocks/lots: Block 1309, Lot 10, Block 1338, Lot 1 and Block 1400, Lot 1. The latitude/longitude of the site, as determined from the HHMT-Port Ivory Facility center, is 40 degrees 38 minutes 15 seconds (N)/74 degrees 10 minutes 50 seconds (W). This report addresses Site 2A and 2B, which consists of the eastern portion of Block 1400, Lot 1 and the southern portion of Block 1338, Lot 1, south of building 70 and 74. At the time of the Phase I and SI activities, the entire facility was owned by P&G; the Port Authority purchased the site from P&G in December 2000 and the site is now known as the HHMT - Port Ivory Facility. Subsequent to the



-



purchase of the site, the Port Authority performed RI activities. The Port Authority has also addressed some of the petroleum-impacted areas and certain of the tank areas (tanks formerly used by P&G). Generally, the excavation activities were undertaken in conjunction with site demolition and redevelopment efforts and were performed prior to entering the VCP Program.

Site 2A/2B encompasses 28.6 acres of the 124.31 acre former manufacturing facility and is comprised of Site 2A (23.94 acres) and Site 2B (4.66 acres). At the time of the purchase by the Port Authority, the HHMT- Port Ivory Facility site was improved by 68 buildings. Site 2A included the following buildings: 30-Series Buildings; 40-Series Buildings; Building 1; Building 20; Building S-35 and portions of Buildings 12 and 13. The locations of the site buildings (Year 2000) are presented on Figure 3. The site was formerly utilized for the manufacturing of consumer products including soap, detergent and foodstuffs. Reportedly P&G initiated manufacturing operations in the early 1900s and ceased operations in approximately 1991. Generally, Site 2A was utilized for offices, boiler buildings, furnace structures, manufacturing and development buildings, locomotive maintenance, security stations, cafeteria, sewage treatment plant and the reservoir for fire protection. Site 2A was home to product delivery, several underground storage tanks (USTs), former aboveground storage tanks (ASTs), and an underground network of piping for drainage control/Site 2B encompasses the areas south of the 70-Series Buildings and includes the southern portion of part of the 70's building complex: southern portion of Buildings 70, 70 A/B/C, 70F, 70G and 72. The Site 2B/Site 3 boundary line extends along the southern building wall of Buildings 74 and 75 such that Site 2B does not include the interior of Buildings 74 and 75 but includes exterior areas to the south of the buildings. Site 2B was utilized for storage and warehousing of finished products, production and packaging of orange juice and is characterized by the presence of railroad tracks. A summary of the site buildings present during the Phase I ESA and Year 2000 SI is provided in Table 1.

Site entrance/exit ways are located along Western Avenue and Richmond Terrace. Western Avenue extends in a north-south direction between Block 1400, Lot 1 (Sites 1 and 2A) and Block 1338, Lot 1 (Sites 2B and 3) and terminates at Richmond Terrace. One of the three parcels, Block 1309, Lot 10 (Future Site 4) is situated north of Richmond Terrace and the two remaining parcels, Block 1400, Lot 1 (Sites 1 and 2A) and Block 1338, Lot 1 (Sites 2B and 3), are situated south of Richmond Terrace. The overall layout of Site 2 as well as the remainder of the HHMT-Port Ivory Facility is presented on Figure 2.

8





Building	Reported Information	Observations/Comments
Identification		
Building 1/ Woodburning Boiler	This seven-story building, encompassing 50,337 square feet, was built in 1983 in response to rising oil prices. Building 1 houses a wood burning boiler and associated wood storage hopper. The boiler (south end) and storage hopper (north end) comprise the majority of the building area. The boiler is suspended from the ceiling structure to allow for the expansion and contraction of the interior steam tubes. Former storage areas for parts, materials and equipment utilized in the operation of the boiler are located on the first and second floor of the structure. Wood is reported to have been processed in a building (Building 1A) located to the northwest of Building 1 and stored in an area to the south of Building 1B. Wood is reported to have been supplied to Building 1 via a conveyor belt system. The conveyor belt initiated at the wood storage area located to the southwest of Building 1B and entered Building 1 at the northwestern corner of the seventh floor.	The building is constructed with a concrete floor and sheet metal walls and ceiling. Parts and equipment utilized in the operation of the boiler unit were stored in this building at the time of the assessment and investigation activities.
Building 12/ Machine Shop (Partially located on Site 2A)	Building 12 is a two-story building which occupies 15,128 square feet and was built in 1918. According to P&G, this building was utilized as the "central" machine shop for the facility, contained typical equipment for a machine shop (i.e., grinders, lathers, saws, presses, etc.) and was used (2 nd floor) for the storage of parts, equipment, and machinery.	The first floor and the eastern portion of the second floor are constructed with a concrete floor, brick walls and a concrete ceiling. The western portion of the second floor (i.e., the Locker Room) is constructed with a ceramic tile floor, a combination of sheet rock and ceramic tile floors and a drop panel (2' x 2' tile) ceiling. Overhead loading dock doors providing access to the exterior are located along the northern and western walls of Building 13.
Building 13/ Engineering (Partially located on Site 2A)	This two-story 6,040 square foot building was built in 1916 to have been used solely for office/administrative purposes including, in particular, housing the Engineering Department.	The building is constructed with a combination of ceramic tile/linoleum or concrete flooring, sheet rock walls and a drop (2' x 2' tile) panel ceiling. An Electric Room is located on the second floor of this building. Inspection of this room noted the presence of several wall-mounted transformer units and electrical panels. This room was constructed with a $9"x 9"$ vinyl tile floor. Two office trailers, formerly utilized for additional office space, were noted to be situated in the area located immediately north of Building 13.



Hatch Mott MacDonald

I.M

Building	Reported Information	Observations/Comments
Identification		
Identification Building 19/ Fire Pump House - Building 20/ Locomotive House	This one-story 252 square foot building was built in 1962, and was utilized to house pumps associated with the facility's fire suppression system. Fire suppression water is maintained in a 2.5 million gallon reservoir located immediately adjacent to (east of) this building. This building is reported to be connected, via underground piping, to a second water reservoir and pump house (i.e., Building 30). According to P&G, no chemicals were added to the fire suppression system. The 2.5 million gallon water storage reservoir is reported to be filled on an as-needed basis through pipe connections to the City of New York water supply system.	Building 19 is constructed with concrete floor and walls and a metal deck ceiling. Inspection of this building noted the presence of an oil-fired (diesel) generator. {Staining, was noted along the underside_of-the-generator_and_the_underlying concrete-floor. Inspection of the underlying concrete floor noted same to be intact and free of breaches in its integrity. Based on the construction of the building and the fact that the floor is set at a lower elevation than the doorway, it is unlikely that discharges, if any, from the generator could migrate to the exterior area surrounding this building. Diesel fuel utilized in the operation of the generator unit is stored in a \$500-gallon. AST-located along the southern exterior of the building. The AST is situated inside:of-a-containment dike:constructed with a concrete floor and masonry walls. Visual inspection of the containment dike did not reveal the presence of any indications that a discharge has occurred within same. Building 20 is constructed with a combination concrete and bare soil flooring, brick walls and a concrete ceiling. The building is divided into three bays with overhead doors along the southern wall of the building providing access to the railroad sidings at the exterior. Inspection of the soil-floor.in the ceastern and western have noted the presence of staining.in
		several-areas. A subsurface maintenance pit constructed of concrete is located in the central bay of this building. At the time of the assessment and investigation activities, snow removal equipment, an air compressor, hoses and parts as well as several small containers (i.e., five-gallons or less) of
		various lubricants, oils and other petroleum products were noted to be present in this building. Concrete pads are located along the western and northern sides of this building. A non- paved, gravel covered area measuring approximately 50 feet by 100 feet was noted on the eastern side of Building 20. No staining was noted on the gravel surface.
Thirty Series Buildin	gs	
1		1



Table 1			
Summary of Site Buildings			
Site 2A/2B: HHMT - Port Ivory Facility			

Hatch MottSummary of Site BuildingsMacDonaldSite 2A/2B: HHMT - Port Ivory Facility			
Building Identification	Reported Information	Observations/Comments	
Building 30/Fresh Water Pump House	This one-story, 740 square foot structure was built in 1915 and formerly functioned as one of the two, on-site pump houses associated with the facility fire suppression system. The building is located immediately to the north of a concrete lined water reservoir with a reported capacity of approximately 50,000-gallons. This pump house is reported to be connected, via underground piping, to the main pump house, Building 19.	Building 30 is constructed with a concrete floor, a combination of concrete and brick walls and a concrete ceiling. The building is divided into two levels, separated by a metal grate floor/ceiling. Metal steps provide access to the lower level of Building 30. Visual inspection of the building noted the presence of a diesel oil-fired generator and two large pump units. Diesel oil associated with the generator unit is stored in a cl_000:gallon_(approximate) AST>suspended underneath the metal-grate floor. The fill and vent pipe system associated with the AST extends through the eastern wall of the building. Staining:was noted on the generator-unit as-well-as_on_the_underlying=concrete_pad. However, staining was noted to be limited to the concrete pad.	
Building 31/Comet Warehouse	This one-story, 9,000 square foot building was constructed in 1909 and was utilized for storage.	The building was constructed with a concrete floor, brick walls and wood ceilings. Due to potential safety concerns, no access to this building was provided to HMM. As a result of the building's deteriorated condition, this building, as well as Building 33 and 33A did not have a roof at the time of inspection.	
Building 31A/ Glycerine House	This two-story building occupies 8,000 square feet and was constructed in 1909. According to a representative of P&G, this building was utilized as a storage area for equipment and parts associated with the boiler house. The first floor of the building was used for miscellaneous storage and the second floor of the building was utilized for office space.	Building 31A is constructed with a concrete floor, brick walls and a wooden roof. A fenced-in storage area is located along the exterior of the western wall of Building 31A. The storage area is underlain with concrete and separated into storage areas via masonry walls. Based on the signage on the fencing, this area was formerly utilized to store (empty and full) pressurized gas cylinders (i.e., acetylene, oxygen, etc.). A pad-mounted compactor is located along the exterior of the southwestern corner of this building. Inspection of same proved unremarkable.	

1	

Hatch MottSummary of Site BuildingsMacDonaldSite 2A/2B: HHMT - Port Ivory Facility		
Building Identification	Reported Information	Observations/Comments
Building 32/ Boiler House	This two level (one floor and a basement), 29,860 square foot building was built in 1909 and was utilized to house two boiler units (i.e., natural gas and diesel oil) as well as the main and two secondary generator units. The generators are reported to be powered by steam manufactured during the operation of the boiler units. According to P&G, the two boiler units were removed from the northern portion of this building (the floor cut-out was noted at the time of the inspection). Natural gas is reported to be provided by Con- Edison. Diesel fuel is reported to have been stored in an UST formerly located along the eastern exterior of the building.	Building 32 is constructed with a concrete floor, brick walls and a metal deck ceiling. Reportedly, the cut-outs present in the flooring/ceiling were to allow for equipment clearance. The main electric room for the facility is situated in Building 32. This room is constructed with a concrete floor, a combination of concrete and brick walls and a metal deck ceiling. Inspection of this room noted the presence of several pad-mounted switch and breaker box units. Four 55-gallon drums of Sunoco TH Fluid were noted to be stored along the eastern wall of the electric room. The main generator (7500 kva) and two secondary generators (1500 kva) are located in Building 32. These generators are seated on the concrete floor in the basement and protrude through to the first floor via the floor cut-outs.
Building 32A/ Boiler House	This two level (one floor and a basement), 3,200 square foot building was built in 1909 and was utilized to house a diesel oil-fired generator unit used on weekends and during maintenance of the boilers located in Building 32. Diesel fuel is reported to have been stored in an UST formerly located along the eastern exterior of the building.	Inspection of the building noted similar construction to Building 32. Some <u>staining</u> was noted on the concrete flooring: However, the flooring was noted to be intact. A 3;000-gallon AST-(approximate) is located in the northwestern- corner of the basement of this building. The AST is reported to be associated with a cooling tower located on the roof of this building.
Building 33/ Kettle House 1	This three-story, 54,000 square foot building was built in 1909 and was utilized in soap manufacturing. Specifically, a large kettle utilized in the melting of soap was located in this building. The kettle was heated via steam generated in the Boiler House (i.e., Buildings 32 and 32A). The kettle was described as a large tank that extended to the second floor of the building. The upper floors were utilized for storage purposes as well as to monitor the kettle operations.	Due to potential safety concerns, HMM was not provided access to this building. As a result of the building's deteriorated condition, this building, as well as Building 31 and 33A did not have a roof at the time of inspection. The building was noted to be constructed with a concrete floor, brick walls and wood ceilings.



Hatch Mott MacDonald

hav

Building	Reported Information	Observations/Comments
Identification		
Building 33A/ Comet Making (Kettle House #3)	This three-story, 20,100 square foot building was built in 1909 and was utilized for activities associated with soap and Comet cleanser manufacturing. A large kettle utilized in the melting of soap and the manufacture of Comet cleanser was formerly located in this building. The kettle was heated via steam generated in the Boiler House (i.e., Buildings 32 and 32A). The kettle was described as a large tank that extended to the second floor of the building. The upper floors were utilized for storage purposes as well as to monitor the kettle operations.	HMM was not provided access to this building. As a result of the building's deteriorated condition, this building, as well as Building 31 and 33 did not have a roof at the time of inspection. The building was noted to be constructed with a concrete floor, brick walls and wood ceilings.
Building 34/ Comet Warehouse	Building 34, a one-story 8,568 square foot structure, was built in 1929 and was utilized for the storage of cardboard packaging for Comet cleanser.	Inspection of this building noted same to be constructed with concrete floors, brick walls and a wood ceiling.
Building 35/ Comet Packing	This three-story (two floors and a basement) 25,336 square foot structure was built in 1909 and was utilized for packaging activities for Comet cleanser. A bridge, located in the southeastern corner of the second floor, connects this building to Building 42C. This bridge was utilized to transfer manufactured liquids stored in this building (manufactured in Building 38) to the packaging facility located in Building 42C. An elevated area located in the southwestern corner of the second floor is reported to have been utilized for the storage of packaging materials as well as spare parts for the packaging machinery.	Visual inspection of this building noted same to be constructed with a combination of concrete and wooden floors, brick and masonry walls and wooden ceilings.
Building S-35/ North Gate House	This one-story 336 square foot building was constructed in 1930 and was utilized as a Guard House (security station).	Visual reconnaissance of the building noted the northern portion to be utilized as a Guard House and the southern portion to be utilized to house electrical equipment (i.e., transformers, switch boxes, breaker boxes, etc.)
Building 36/ Comet Warehouse	This five-story (plus partial basement) 26,390 square foot building was built in 1923 and was utilized to warehouse packaging materials associated with the Comet cleanser line. According to P&G, a conveyor system in a tunnel in the the basement was utilized to transport finished Comet product to Building 43A.	Visual inspection of this building noted that all five floors were constructed with wooden floors, brick walls and wooden ceilings. No access was provided to the tunnel at the time of inspection.
Building 38/Liquids Making (HSC)	This two-story 2,992 square foot building was built in 1927 and was utilized for the manufacturing of Solo liquid detergent. The manufacturing process is reported to have consisted of blending operations. Mixing vats are reported to have been situated on the first floor of the building and to have protruded, through "cut-outs", to the second floor of the building.	This building is constructed with wooden floors, brick walls and a wooden ceiling. Inspection of the building noted two circular "cut-outs" in the second floor. These "cut-outs" appear to denote the locations of the former blending vats.

1

14



Hatch Mott MacDonald

Building	Reported Information	Observations/Comments
Identification		
Forty Series Buildi	ngs	
Building 40/ Cafeteria	Building 40 is a three-story (i.e., two floor plus a basement) 25,608 square foot structure constructed in 1916 and utilized for cafeteria activities, office/administrative tasks and for employee training activities (i.e., classrooms/ demonstration rooms).	The building is constructed with a ceramic tile floor (some portions), concrete walls and wooden ceilings. A sump pump system and a gas-fired heating unit were noted in the basement. The sump system is reported to discharge, via aboveground piping and hoses, to the southeastern exterior of the building and eventually, Western Avenue.
Building 41/Offices	This two-story 21,500 square foot building was constructed in 1907 and was utilized for office/administrative purposes as well as locker rooms (northeastern corner of the first floor).	The building is constructed with carpeted and linoleum floors, sheet rock walls, and drop panel ceilings. A HVAC room with a single HVAC unit and transformer is located on the second floor of Building 41.
Building S-41/ Natural Gas Meter House	This one-story 588 square foot structure was built in 1955 and formerly housed components of the natural gas entry system. The system was shut down and replaced by new connections from the street into the individual buildings.	Inspection of this structure noted same to be constructed of concrete (i.e., floor, walls and ceiling). The structure is divided into two rooms (north and south) which contain the natural gas piping, valves and meters.
Building S-42/ Transformer Shelter	This one-story 315 square foot structure was built in 1927 and houses the main electrical switches and breaker boxes.	This building is constructed with a concrete floor, brick walls and a concrete ceiling. At the time of the May 2000 site reconnaissance, this structure contained six breaker boxes, several switch boxes and several back-up batteries. A concrete dike system is located along the southern exterior of Building 42. This dike contains five exterior transformer units. No staining was noted on the units or the concrete base of the containment dike.
Building 42A/ Bar Soap Storage	This four-story 63,200 square foot structure was built in 1918 and was utilized for the manufacture and packaging, including the storage of packaging materials, of Coast Soap.	Building 42 was noted to be constructed with a combination of tile, wood and concrete floors, masonry and brick walls and wood and/or concrete ceilings. Two circular "cut-outs" in the ceiling of the first floor of Building 42A denote the locations of two:former: <u>ASTs:or-circular.storage:vessels</u> . The tanks are reported to have been utilized to store(scrap:soap_which was collected for re-use. A floor drain was noted in the southwestern corner of the first floor of Building 42A. According to P&G, all interior drains were/are connected into the sanitary sewer system or, in the case of drains that collected materials from process operations, connected to process systems.

-	-

Table 1	
Summary of Site Buildings	
Site 2A/2B: HHMT - Port Ivory Facility	

Hatch MottSummary of Site BuildingsMacDonaldSite 2A/2B: HHMT - Port Ivory Facility		y
Building	Reported Information	Observations/Comments
Building 42B/Bar Soap Processing/ Packaging	This three-story 81,300 square foot structure was built in 1908 and was utilized for the processing and packaging of finished bars of Ivory Soap and a machine shop for packaging equipment. A "dry house" was located on the eastern side of this floor. The "dry house" consisted of a conveyor belt system that traversed through a heat enclosure. After exiting the "dry house", the bars were cut, stamped and packaged.	Inspection of this building noted same to be constructed with concrete floors, masonry walls and wooden ceilings. A machine-shop, utilized to maintain the packaging machines located in Buildings 42A, 42B and 42C was formerly located along the eastern side of the first floor of this building. The northern side of this building is open to the second story and formerly housed a large storage tank which contained liquid Ivory soap. The liquid soap was transferred to this location from the Hydrolizer Building which was formerly located to the north of Building 16; the Hydrolizer was demolished prior to the May 2000 site reconnaissance. The liquid soap was then pumped from this AST to the third floor of the building for drying, shaping, cutting and packaging. Several floor drains were noted on the third floor of this building. These drains were utilized to collect water from the plotter stamps which transferred the "Ivory" stamp onto the soap bars. This representative added that water collected via these drains was re-blended and recycled.
Building 42C/ Liquids/Solo Packing and Storage	This three-story 81,300 square foot structure was built in 1908 and was utilized for the storage of cardboard packaging for the finished soap products and for packaging activities for Mr. Clean products.	The building is constructed with concrete floors, a combination of masonry and brick walls and wooden ceilings. An electrical room containing a transformer unit and other wall-mounted electrical panels is located in this building. Inspection of the third floor of this building noted the presence of several square pads surrounded by trench drain systems. According to P&G, machinery associated with the packaging of Mr. Clean cleaning solution was formerly located on these pads. The trench drain systems were utilized to collect any spilled cleaning solution and transfer same back into the process.

Hatch Mott MacDonald

h

• Building	Reported Information	Observations/Comments
Identification		
Building 43/ LSP Granules	This three-story 37,380 square foot structure was built in 1927 and was utilized for the packaging of synthetic granule type soaps and detergents (i.e., Ivory Snow and Ivory Flakes). Soap granules were separated based on their size and then placed into the buggies which dumped same, through floor openings in the northern portion of this building, into hoppers located on the second floor. The various sized granules were blended together to prepare Ivory Snow and Ivory Flakes.	Inspection of this building noted same to be constructed with a concrete floor, masonry walls and combination of steel and wood ceilings. Inspection of the first floor of Building 43 noted the presence of a concrete pad along the northern wall of the building and a trench drain situated on the southern side of the pad. Further, the entrance/exit of a subsurface tunnel was noted in the northern portion of the building. According to a representative of P&G, the tunnel formerly housed a conveyor belt system, which was utilized to transfer raw product from the "forty series" buildings to Building 43 for packaging. No access to the tunnel was provided at the time of the inspection.
Building 43A/ LSP Granules	This three-story 55,296 square foot structure was built in 1927 and was utilized for the packaging of synthetic granule type soaps and detergents (i.e., Ivory Snow and Ivory Flakes).	Inspection of this building noted same to be constructed with a concrete floor, masonry walls and combination of steel and wood ceilings.
Building 44/ LSP Granules	This four-story 24,176 square foot structure was built in 1912 and was utilized for the packaging of synthetic granule type soaps and detergents.	Inspection of this building noted same to be constructed with a combination concrete and ceramic tile floor, masonry walls and concrete ceilings.
Building 45/Main Gate Guard House	Building 45, a one-story 344 square foot structure, is utilized as a security house for the main gate of the facility. The structure was built in 1982.	Inspection of this building noted same to be constructed with non-slip floor tiles over concrete, a combination of masonry and glass walls and a drop panel ceiling $(2' \times 2' \text{ tiles})$.
Building 46/Sewage Pump Station	This structure was built in 1970. Although given a building designation, the Sewerage Pump Station is a covered concrete pit. The structure was never occupied by employees. The pump station is utilized to house the piping, pumps and valves associated with the transfer of sanitary sewage from the facility and to the New York City sanitary sewer system.	The facility's sanitary sewage holding and pre-treatment system is located west of Building 46. Sanitary flow from the facility, via gravity flow, is directed to two concrete pits with wastes pumped into two 71,000-gallon holding tanks located within a concrete dike. P&G's pretreatment of the wastes consisted of pH adjustment/correction via the addition of either sulfuric acid or sodium hydroxide. These materials were noted to be stored in three 250-gallon plastic tote-type storage tanks The tanks are located over a sloped concrete pad which leads into the diked area. After pre-treatment, the wastes are pumped to the NYC sanitary sewer system via the piping located within Building 46. Building 46 reportedly contains a mechanically actuated valve system which monitors pH and closes the system in the event of a problem.

17



webstand and the second second

Table 1 Hatch Mott Summary of Site Buildings MacDonald Site 2A/2B: HHMT - Port Ivory Facility		
Building Identification	Reported Information	Observations/Comments
Building 300/ Water Treatment	This two-story 3,268 square foot structure (one floor and a basement) was constructed in 1929. Operations in this building 'are reported to have consisted of the treatment of water utilized in process operations.	This building is constructed with a concrete floor, brick walls and a metal deck ceiling. Equipment utilized in the water treatment operations was noted to be present in this building. (Two-ASTs, located in a concrete containment area along the southern exterior wall of this building, and a third-AST, situated on a concrete pad along the southwestern corner of the building, were also noted to be present. The two tanks, located within the containment area are reported to contain acid:and caustic utilized in the water treatment process. The third AST is reported to be utilized to store treated "neutralized"-water.—A sump is located within the concrete containment berm associated with the two exterior tanks. Several floor and trench drains were noted around the water treatment equipment. These drains are reported to transfer collected liquids to the sanitary sewer system.
Seventy Series Bui	ldings	
Building 70/ Citrus Office (Boundary line of Sites 2B and 3 runs along the southern building wall of this Building)	Building 70 is a one- story 8,000 square foot structure built in 1959 and utilized for office/administration activities for Citrus Hill and Duncan Hines Bake Mix (DHBM) Division.	Inspection of this building was impeded by lack of lighting. The building appeared to be constructed with concrete floors, masonry walls and a concrete ceiling. Floor drains in the bathroom, locker room and shower areas are reported to be connected to the sanitary sewer system.
Building 70A/ DHBM Making (Boundary line of Sites 2B and 3 runs along the southern building wall of this Building)	This one-story 1,220 square foot structure was built in approximately 1959 and was utilized for the blending and mixing the various dry ingredients used in the formulation of Duncan Hines Bake Mixes. The dry ingredients (i.e., sugar, flour, etc.) are reported to have been transferred to the building via a rail siding located along the southern side of this building. The ingredients were "blown" into the building and placed into large mixing vats.	This building is constructed with concrete floors, masonry walls and concrete ceilings. The building contains a mezzanine level which is reported to have been utilized to fill and monitor the blending vats. Several sealed overhead doorways were noted along the northern wall of this building. These doorways were formerly utilized to transfer the blended cake mixes into the adjacent building, Building 70B, for packaging. A large pad-mounted transformer unit as well as other pad-mounted electrical equipment was noted along the southwestern corner of this building. A concrete truck ramp is located along the southeastern corner of Building 70A.—Tank= saddles_were_noted_along the western side of the ramp. No indications of staining were noted on or adjacent to the tank saddles.

Hatch Mott Summary of Site Buildings MacDonald Site 2A/2B: HHMT - Port Ivory Facility		
Building	Reported Information	Observations/Comments
Identification Building 70B/ DHBM Packing (Boundary line of Sites 2B and 3 runs along the southern building wall of this Building)	This two-story 14,140 square foot building was built in 1959 and was used for the packing of the cake mixes and the storage of cardboard packing materials.	Building 70B is constructed with a combination of concrete and vinyl tile flooring, masonry walls and a concrete ceiling. The eastern portion of this building was noted to consist of a single floor which included a hoist system. The system is reported to have been used to transfer packaging materials from the second floor storage area to the first floor packing area. Inspection of the second floor of the facility noted several floor drains and a mop well. These drains are reported to be connected to the sanitary sewer system servicing the subject site. Minor-staining was noted on the concrete- flooring in these areas. However, the flooring was noted to be intact.
Buildings 70C and 70D/Warehouse (Boundary line of Sites 2B and 3 runs along the southern building wall of Building 70C)	These two building designations are combined into a single one-story warehouse structure reported to have been built in 1959. Building 70C occupies 14,271 square feet and 70D occupies 34,829 feet. These building are reported to have been utilized for warehousing purposes as well as the packaging of Citrus Hill orange juice (eastern portion of the warehouse area).	This building is constructed with concrete floors, masonry walls and a metal deck ceiling. Electrical equipment, a hot water heater and a transformer are located along the western wall of this area. Several trench drains are located in the eastern portion of this building. According to P&G, these trench drain systems formerly surrounded the packaging equipment associated with the packaging of Citrus Hill orange juice. These floor drains, as well as the remainder of the floor drains located in the facility, are either connected to the sanitary sewer system, or in the case of drains that collect liquids from process operations, are connected back into the process.
Building 70F/ DHBM Kitchen (Partially located in Site 2B)	This one-story structure was constructed in 1978 and occupies 644 square feet. According to a representative P&G, this building was utilized for the testing of the DHBM.	This area was constructed with a ceramic tile floor, masonry walls and a metal deck ceiling. Inspection of this building noted counter top areas along the eastern side of this area.
Building 70G/ DHBM Cotactor Building (Partially located in Site 2B)	This two-story 1,804 square foot structure was constructed in 1978 and was utilized for the storage of compressors associated with the operations formerly conducted in Building 70A.	This building designation appears to correspond with an alcove located along the southern wall of Building 70A. Construction of this area proved similar to that noted in Building 70A.

	Hatch	Mot
him	MacD	nal
	IVIACIO	ומו ונ

Building	Reported Information	Observations/Comments
Identification		
Building 72/ DHBM Loading Dock (Partially located in Site 2B)	This one-story 3,100 square foot loading dock structure is reported to have been built in 1962. Building 72 is not actually a building as it does not maintain any exterior walls. The north side of the structure is bounded by the southern wall of Building 70A. The southern side is open to allow for the loading and unloading of materials associated with the making of the Duncan Hines Bake Mixes.	The structure was constructed of concrete. No indications of staining were noted with regard to the concrete loading dock.
Building 74/ Warehouse Building (Boundary line of Sites 2B and 3 runs along the southern building wall of this Building)	This one-story 103,400 square foot building was constructed in 1980 and is reported to have been utilized for the warehousing of various P&G manufactured (finished) products.	This building was noted to be constructed with a concrete floor, masonry walls and a metal deck ceiling. Three overhead doors, providing access to the adjacent warehouse building to the east (i.e., Building 75), are located along the eastern wall of this building. An elevated mezzanine area is located in the western portion of this building. In addition, a raised concrete pad with a masonry room is located immediately to the east of this mezzanine area. These areas are reported to have been associated with the packaging operation of the Citrus Hill orange juice conducted in the adjacent Buildings 70C and 70D. Inspection of these areas noted the presence of air chilling/handling equipment to be located on the second floor mezzanine area. This equipment appeared to have been utilized to supply cold air to the aforementioned masonry room. A large switch box/breaker box is located west of the masonry/cold room. Floor drains were noted to be present in a concrete diked area in the northeastern corner of the raised concrete area. The floor drains are reportedly tied to the sanitary system.
Building 75/ Warehouse Building (Boundary line of Sites 2B and 3 runs along the southern building wall of this Building)	This one-story 103,400 square foot building was constructed in 1980 and is reported to have been utilized for the warehousing and distribution of the various P&G manufactured products.	This building is constructed with a concrete floor, masonry walls and a metal deck ceiling. Approximately twenty overhead loading dock doors are located along the eastern wall of this building. These doors provide access to the loading/unloading dock area located to the east of Building 75. In addition, three overhead doors are located along the western wall of this building and provide access to the adjacent warehouse building (i.e., Building 74). A transformer unit is mounted on a metal support bracket in the southwestern corner of this building. A masonry room is



Hatch Mott MacDonald

Table 1 Summary of Site Buildings Site 2A/2B: HHMT - Port Ivory Facility

	Site 24/2D. IIIIvii - Fort Rony Facility	y
Building	Reported Information	Observations/Comments
Identification		
		located in the northeastern corner of this building. According
		to P&G, this room was utilized as a climate control room in
		the Crisco food oils manufacturing process. A concrete sub-
		level is located in the southwestern corner of Building 75.
		Inspection of this area noted the presence of an electric hot
		water heater, an approximately=3;000-gallon=AST_and a sump
		and pump system servicing an adjacent bathroom facility. The
		AST-is-utilized-to-store-hot-water heated by a solar panel on
L		the roof of the building.

NOTES:

Ï

- (1) Buildings are located on Site 2A unless otherwise noted.
- (2) All facility buildings are reported to have been heated by steam fired heating units. Steam was provided to the individual buildings by the facility's boiler houses.
- (3) Several of the facility buildings contain freight elevators. All of the facility elevators are reported to be cable operated and do not contain any hydraulic pistons. The cable operation system is reported to be located on the roofs of the respective buildings.
- (4) Several floor drains and trench drain system were noted in several of the on-site buildings. According to P&G, all floor/trench drain systems are either connected into the sanitary sewer system servicing the subject site or direct collected materials back (recycled) into the process operations.
- (5) All bathrooms are reported to be connected into the sanitary sewer system servicing the subject site. According to P&G, no septic systems or dry wells are currently or have ever been located on the subject site.
- (6) The subject site buildings are to be serviced via sprinkler systems for fire protection. According to a representative of P&G, the fire suppression system is a "water-only" system. Water utilized in this system is stored in two reservoirs located adjacent to Building 19 and Building 30. The reservoirs are supplied with water via the New York City water supply system.
- (7) The P&G representative who accompanied HMM on the site inspection was unable to provide any information with regard to the storage and/or usage of petroleum products and/or hazardous materials in subject site buildings.

P:\232952wmd\Operable Unit Reports\Operable Unit 2\Post VCP Revisions\Data Tables\Site 2 - Table 01 building inspection summary.doc

21
22



The entire HHMT-Port Ivory Facility is and has been serviced by connections to the potable water and sanitary sewer system of New York City. No septic systems, potable water wells or dry wells are reported to be or ever to have been located on the subject site. Stormwater generated on the site is directed via sheet flow to on-site catch basins. These catch basins discharge, through the facility's underground stormwater sewer system including permitted outfalls, to the adjacent waterways, roadways and marshland areas. Electrical service is supplied to the subject site via connection to the Consolidated Edison system servicing this section of Staten Island.

In addition, several utility easements and pipelines traverse the HHMT-Port Ivory Facility; these easements predated the Port Authority's ownership of the facility. With regard to Site 2A/2B, Colonial Pipeline and Exxon (now known as ExxonMobil) maintain easements. Colonial Pipeline maintains a 10foot pipeline easement that initiates south of Site 2B, extends through the western portion of Site 2B and continues in a north/south direction along the western property boundary of Site 2A. The easement, traverses through Site 2A entering the southwestern corner of Site 1, continuing across Richmond Terrace and through the western portion of Future Site 4 (Block 1309, Lot 10). The easement terminates at the northern end of Future Site 4 (Block 1309, Lot 10). ExxonMobil maintains an 18-foot easement that is located primarily to the east of the Colonial Pipeline easement. The easement initiates to the west of Site 2A and the Colonial Easement, and continues to the east for approximately 90 feet where it turns to the north through Site 2A. This easement parallels the Colonial Pipeline easement throughout Site 1, however, this easement extends in an easterly direction along the southern boundary of Block 1309, Lot 10 beyond Richmond Terrace. Texas Pipeline is noted as having an easement of unknown width traversing east/west through Site 2B. The easement originates to the south extending into Site 2B and is maintained by SOHIO. Texas Eastern Pipeline is also noted as having an easement of unknown width that parallels the Texas Pipeline. The presence of an abandoned 10-foot easement formerly maintained by Tidewater Pipe Co. LTD. traverses Site 2B as well. The pipeline originates to the west of Site 2B and traverses the length of the site before turning to the north extending through Site 3 and the eastern portion of Future Site 4. The termination point of the Tidewater easement is unknown. The locations of the easements are presented on Figure 2. \checkmark



3.0 BACKGROUND

In the early 1900s, P&G developed portions of the HHMT-Port Ivory Facility for use as a consumer goods manufacturing facility. The initial development included portions of Sites 1, 2A and 4. Over the years, P&G acquired additional acreage (Site 2B and 3) and emplaced fill materials at low-lying areas of Sites 1 and 2 (2A) and Future Site 4 expanding the original facility (i.e., the original P&G/Port I/ory Facility) to include the current HHMT-Port Ivory Facility limits, as shown on Figures 1 and 2. The site was utilized for consumer goods manufacturing from development until 1991. The specific consumer goods produced at the facility and the operations/activities performed at specific site areas changed based upon corporate requirements. A discussion of the current and historical physical setting of Site 2A/2B, including a summary of historical operations specific to Site 2A/2B are included herein.

3.1 Site 2A/2B History

According to representatives of P&G and information provided in reports supplied by same, P&G constructed the initial Port Ivory manufacturing facility at this site in 1906-1907. The original 77-acre facility included Sites 1 and 2A (Block 1400, Lot 1) and Future Site 4 (Block 1309, Lot 10) and was developed on an open, vegetated, marshy area. Additional acreage is reported to have been gained through the acquisition of Site 2B and 3 (Block 1338, Lot 1) as well as the filling of additional marshlands at all four Sites. The fill used by P&G in conjunction with site development is reported to have included the following: sand, silt, gravel mixed with debris, cinders generated from on-site coal-fired boilers, calcium carbonate and other carbonate salts generated as a by-product from soap manufacturing processes, spent diatomaceous filter earth from vegetable oil refining operations, and carbonaceous filter material from glycerin recovery operations. Visual review of subsurface conditions during SI and RI activities indicates that all of the above listed materials may have been emplaced at Site 2A; limited fill materials are reported to have been placed at Site 2B. It should be noted that the presence of fill material is considered an existing condition with regard to the HHMT-Port Ivory Facility.

Historical information sources indicate some variability in the operations performed at specific site locations throughout the operation of the facility. However, in general, Sites 1 and 2A (Block 1400, Lot 1) were utilized as a single facility for soap, comet, and glycerin manufacturing and utility functions (i.e., boiler houses, wood processing for the boilers, locomotive maintenance, etc.). Site 2A was the location

Site 2A/2B Report

X



of the manufacturing buildings, the boilers, the rail yard, locomotive maintenance buildings, raw product manufacturing/storage, and other activities associated with the processing, packaging, and storage of products from the early 1900's to the closure of activities. Other buildings included the offices, cafeterias, sewage treatment facility, fire suppression systems, and the chimneystack were reported to have operated and existed on Site 2A. Portions of the inactive railway system remain at Site 2A.

Historical information sources also identified structures and ASTs that were not present at the site during initial assessment activities. With regard to Site 2A, structures were noted to be present north of Buildings 12 and 13 and west of the 30-Series Buildings. A structure referenced as the Kettle House was located northeast of the former processing building and south of Building 17. This building appeared to be situated on both Site 1 and Site 2A. Historical mapping shows buildings identified as Building 10, Building 11, Building 10A, Building 22, Building 23 and Building 14 (Lye) in the open area north of Building 12 and 13, south of Building 17 and west of the 30-Series Buildings. In addition, mapping identified the presence of ASTs located adjacent to Buildings 11, 23, and 14. Further, historical mapping shows the positioning of a network of railroad tracks running adjacent to the former 20-Series Buildings as well as additional tracking alongside existing buildings. Historical mapping did not identify the presence of former structures or tanks a Site 2B.

Based on historical mapping and information provided in reports prepared by P&G, the following materials were stored in ASTs present and/or maintained at storage areas located at Site 2A: caustics, various vegetable and fish oils, fuel oil, waste oil, soap, spent acids, spent nickel catalyst, grease, coke and resin. The storage methods are not identified on the maps. Several ASTs on the Block 1400, Lot 1 parcel (Sites 1 and 2A) were labeled on historical Sanborn Maps as being "hydrogen holders". Historical maps also identify the use of underground storage tanks (USTs) at the site including four areas (referenced as UST1, UST3, UST4 and UST7 in the SI and RI) on Site 2A. Historical information indicates the following tank contents: variations in grades of oil in one or more tanks at Areas UST1, UST3, UST4 and UST7. Tanks containing ethanol and enzymes are also reported to be present at Site 2A generally at locations north/northeast of Building 13 and west of Building 31.

Historical aerial photos and mapping before the 1900's, indicate that Site 2B was characterized by marshland prior to development by P&G. Previous site activities performed at Site 2B consisted primarily of railway. Buildings 70-75, primarily situated on Site 3, were used in process operations performed by the Citrus Hill and Duncan Hines divisions of P&G. The Citrus Hill Division

formulated/packaged orange juice and operations centered in Buildings 70, 70C, 70D, and 70J (citrus shop). The Duncan Hines Baking Mix (DHBM) operations were located in Buildings 70A (productions), 70B (packaging), 70F (kitchen), 70G (contractor), 70I (personnel), 72 (loading dock), 74 (warehouse), and 75 (warehousing). The 70-Series Buildings were constructed in the late 1950's, with modifications and additional construction from the late 1970's to the mid-1980's.

3.2 Hydrogeologic Setting

Hydrogeologic provinces within Staten Island include both the Atlantic Coastal plain and the Triassic lowlands section of the Piedmont physiographic province. The Precambrian-Cretaceous unconformity defines the boundary between these two physiographic provinces extending northeastward from Fresh Kills to north of Stapleton, continuing eastward across Long Island. The low-lying plain in extreme northwest Staten Island consists of glacial outwash deposits and tidal marsh. Outwash deposits consist chiefly of stratified fine to coarse sand and gravel, while shore and marsh deposits consist of sand, organic clays and silts. These deposits are generally thin and probably no thicker than 15 feet.

The subsurface unconsolidated deposits at Site 2A/2B, as well as the remainder of the site, include a complex of stratified drift, glacial till, and tidal marsh deposits consisting of glacial outwash, marsh deposits, and artificial (non-indigenous) fill. In general, the following six soil and rock strata have been identified at the subject site area (listed from ground surface to top of bedrock) (1) non-indigenous fill consisting of sand, silt, clay, and gravel in a generally loose condition covering most of the subject site with a maximum thickness of about 19.5 feet (2) organic clays and peats, consisting of soft and highly compressible tidal marsh deposits, to a maximum thickness of approximately 27 feet. (3) and deposits consisting of loose to medium dense sand from marine or glacio-fluvial deposits extending eastward across the site and ranging in thickness from 5 to 16 feet; (4) glacial clay, silt, sand and gravel, deposits (primarily of clay and silt) ranging in thickness from less than 10 to 60 feet; (5) weathered shale, partially decomposed or weathered shale; and (6) generally unweathered, competent shale, located at depths of 45 to 72 feet below sea level. A deep bedrock-aquifer monitoring well (LF-DW-1) was installed on the Future Site 4 by P&G prior to May 1993 in conjunction with landfill closure procedures. Bedrock of the Passaic Formation was encountered at approximately 47 feet below ground surface (bgs). Soil borings installed as part of the SI (November and December of 2000) and the RI (2002/2003) confirmed the five upper soil and rock strata; the SI and RI did not include an evaluation of competent bedrock. However, as part of the SI for groundwater at Site 2 three wells, MW-7D, MW-10D, and MW-15D were installed to evaluate the deeper aquifer (Section 3.2.2 and 6.1.2). At locations MW-7D and MW-15D, bedrock was

- 25

26



present at a depth of approximately 70 feet bgs in Site 2A, while bedrock was encountered at 40 feet bgs in Site 2B at the PG-MW-10D location.

The Passaic Formation underlies Site 2A/2B, as well as the remainder of the subject site, and consists of reddish-brown to greyish-red siltstone and shale, with a maximum thickness of 3,600 meters. Dense by its nature, the Passaic Formation exhibits very little primary porosity. However, characteristic vertical or near vertical joints and fractures provide for limited transmission and storage of water. These openings decrease with depth, resulting in decreased permeability and specific yield with distance from the surface. Separations between bedding planes also allows for limited permeability as well as limited transmissivity and storage of water. According to available technical literature, the Passaic Formation exhibits a regional bedding strike of north 50 degrees east and a dip of 9 to 15 degrees to the northwest.

3.2.1 Soils

The three shallowest units described in the above paragraph constitute the soils of the subject site area (i.e., non-indigenous fill on top of organic clays and peat or sand deposits). Essentially, the SI and the RI confirmed that fill material was placed upon tidal salt-marsh or sand deposits at Site 2A/2B to raise the elevation of the land to allow for development. Activities associated with the SI and RI indicated that the soil strata of the site was consistent with that documented in the site area. The presence of fill material at this site is further described in Sections 6.1.1 and 7.4. To provide a visual presentation of Site 1 soil conditions, HMM prepared a cross section diagram based on upon the points identified on Figure 4. Soil conditions are presented geo-spatially in Figure 5, Cross-Section.

3.2.2 Groundwater

. · · · · ·

Groundwater was encountered at depths ranging from approximately 2 to 12 feet bgs across Site 2A and 2B; groundwater depth was estimated based upon information gained through recording water levels in existing and newly installed monitoring wells. The depth to groundwater at Site 2A/2B is consistent with conditions noted at the remainder of the site with the exception of PAMW-11D located on the northeast corner of Site 3 (Block 1338). Groundwater was encountered at a depth of approximately 22 feet bgs, coinciding with the higher topographic location, as compared to the rest of the site. In the shallow



.









sections of bedrock in the area (+/- 150 feet bgs), groundwater is stored within bedding-plane separation and secondary porosity developed by fractures (e.g., joints and faults). Water in the Newark Supergroup of Staten Island occurs under unconfined or confined conditions, depending on the degree of confinement in the overlying deposits and the hydraulic interconnections within the shale and sandstone. Generally, groundwater occurrence in unconsolidated deposits in the site area depends on the sand, silt, and clay compositions of the glacial outwash and non-indigenous fill. Information from the groundwater investigation component of the SI indicates groundwater conditions are generally consistent with that of the area. According to previous environmental investigations (performed by P&G) as well as limited information from the SI (performed by the Port Authority), tidally influenced potentiometric fluctuations were not observed in on-site monitoring stations with the exception of monitoring points directly adjacent to the Kill Van Kull. However, the SI included only a limited review of this issue. Observations during excavation activities associated with building demolition and utility repair/removal indicates the potential for tidal influences at Site 2A.

Movement of fresh groundwater on Staten Island is seaward. Although the unconsolidated deposits and bedrock are hydraulically connected, most of the flow occurs horizontally within the glacial deposits due to their greater hydraulic conductivity. The horizontal flow is estimated to range from less than 0.1 to approximately 1.5 feet/day in glacial deposits comprised of sand and gravel. Estimates of recharge rates on Staten Island are comparable to Kings and Queens Counties, approximately 0.25 to 0.5 million gallons per day per square mile.

Groundwater is not currently used for public water supply on Staten Island. Before 1970, however, the surface water supply from upstate New York was supplemented by pumping a maximum of 5 million gallons per day of groundwater from aquifers beneath Staten Island. Higher pumping rates induced saline groundwater infiltration. Due to saline intrusion of aquifers in the area caused by increased withdrawal, future development of aquifers for potable purposes in the general area is unlikely.

4.0 ENVIRONMENTAL SITE INVESTIGATION WORKPLAN

As previously stated, HMM performed a Phase I ESA of the entire HHMT-Port Ivory Facility. This effort identified AOCs based upon several site inspections, interviews of available representatives of P&G, review of historical information sources (site plans, aerial photographs, Sanborn Fire Insurance Maps) and review of an electronic database search. The AOCs included both site-wide AOCs and area specific AOCs. Thus, an ESIW was developed to address the entire HHMT-Port Ivory Facility including



both area-specific AOCs and site-wide AOC as well as to provide information on current environmental conditions at the site for the purpose of acquisition. The information provided in the following section focuses on efforts undertaken at Site 2A/2B. However, given the site-wide perspective of the ESIW, the information presented in this section also includes or references efforts undertaken at other Sites, as appropriate. Such information is presented for completeness and is provided to convey the comprehensive nature of the SI effort.

4.1 Previous Environmental Investigation Efforts

HMM reviewed documents pertaining to site history and previous environmental investigations in conjunction with the performance of the Phase I ESA and a supplemental file review. The documents included in the review were limited to those made available by P&G. Overall, the documents identified a number of AOCs that were evaluated, to varying degrees, by the prior site owner, P&G. The AOCs involved soil and groundwater, USTs (underground storage tanks) and the presence of a white precipitate material along the eastern bank of Bridge Creek, which runs along the western border of Sites 1 and 2A (Block 1400, Lot 1). A list of the documents included in the review and a brief summary of the contents of same are provided in Table 2. The information provided in Table 2 reflects all documents and reports and, therefore, provides information pertaining to the entire site. To facilitate review, information pertaining to Site 2A/2B has been presented in bold type. In addition, an environmental database report was obtained as part of the Phase I ESA. The electronic database search performed by Environmental Data Resources, Inc. identified that the subject site was included in several American Society of Testing and Materials (ASTM) standard and non-standard environmental record sources. These sources include the following:

- The United States Protection Agency (USEPA), Resource Conservation Recovery Information System (RCRIS) Facilities Large Quantity Generators (LQG) List, December 12, 1999;
- The New York State Department of Environmental Conservation (NYSDEC) Inventory of State Hazardous Disposal Sites (SHWS) List, February 4, 2000;
- NYSDEC, Leaking Underground Storage Tank Incident Reports (LTANKS) List, January 2000;
- NYSDEC, Petroleum Bulk Storage Database (UST) List, January 2000;
- NYSDEC, Chemical Bulk Storage Database (CBS UST) List, January 2000;
- NYSDEC, Chemical Bulk Storage Database (CBS AST) List, January 2000;
- NYSDEC, Major Oil Storage Facilities Database (MOSF UST) List, January 2000;
- NYSDEC, Major Oil Storage Facilities Database (MOSF AST) List, January 2000;
- USEPA Facility Index System (FINDS) List, dated October 1999; and,
- NYSDEC Spills Information Database (Spills) List dated January 2000.

 \varkappa



Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report	Report Topic	Description of Activities and Analytical	Report Conclusions
Identification	Area(s) Of Concern	Results ⁽²⁾	
Phase II Environmental Assessment - Wood Yard, McLaren Hart/Hart Environmental Engineering Corp., prepared for Owl Energy Resources, Inc., November 19, 1991	A 1 to 2 acre wood yard is reported to have been present at the site prior to the 1950s. Further, a water gas holder, four gas purifiers and a coke storage area are reported to have been located at the wood yard. The area is reported to contain coal tars and residues. This report describes an investigation of soil and groundwater at the former wood yard and an attempt to identify the presence of an underlying clay "liner/layer" at this portion of the site.	The investigation included the installation and sampling of four soil borings and the completion of three of the four borings as monitoring wells. Also, four borings were installed for geotechnical purposes. The soil borings did not identify the presence of a clay "liner" beneath the Wood Yard area. (TPHC and BN, compounds, mostly TICs, are reported to have been detected in one or more soil-samples one from soil boring. Also, VO compounds and/or VO TICs, below regulatory criteria were detected in samples from this boring. The report references that the TPHC, detected in soil may be from a leaking hydraulic lift. Di- n-butyl phthalate is reported to have been detected in all soil samples. According to the report, this compound is often detected in soils high in organics and therefore does not pose a threat. The investigation revealed the presence of wood as well as cinder fill. Some elevated readings were recorded on field instrumentation. Analytical results from groundwater samples identify TPHC and BN-TICs in the sample from one well; the same location as the elevated soil results. Atsheen was noted on water in this well. Also, all groundwater samples are reported to have revealed elevated- concentrations of phenols.	The levels of contaminants detected in soil and groundwater were not regarded as an area of concern. Elevated field readings were attributed to the presence of marshlands and underlying peat. The report noted a potential reporting requirement with regard to TPHC. No additional actions are proposed with regard to soil and additional, sampling is recommended to further evaluate phenols in groundwater.



Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
<i>Final Report, Tax Block 1400,</i> Dames & Moore, January 24, 1992	This report presents a summary of investigative activities performed to address nine AOCs identified on this parcel: Area A West Tank Field (southwest of Building 16), Area B S&S Tank Field, Area C Oleum Tank Field, Area E S&S Fat Trap, Area F1 Spent Nickel Catalyst, Area F2 Waste Oil Drum Storage, Area H Former Rosin Area, Area R Northwest Corner of Soap Manufacturing Area (suspected calcium carbonate fill area), and Area P Former Product Unloading Pit. This report also provides information pertaining to the placement of fill materials at Block 1400. The by-products, identified at this parcel include the following: spent zinc and nickel catalyst recovered from fat processing operations (hydrolyzer); spent carbonaceous filter material from glycerine purification; turpentine from recovery of resin from tree soap; coke ash from hydrogen making operations; waste oils from servicing vehicles, locomotives and equipment, and, kettle bottoms? The report also identifies that a site plan notes a "rosin storage area" at the northwest corner of the soap manufacturing area. The area identified as the "rosin storage area" is noted to be unpaved at the time of the investigation. Waste oil is reported to have been used to lubricate rail switches on this parcel. There is some reference but no resolution to UST issues.	Installed and sampled soil borings and wells to investigate the listed areas. The investigation is reported to have revealed the presence of fill material from 2 to 17 feet at areas on the western/northwestern portion of Block 1400. A geophysical survey is reported to have been unsuccessful due to metal interference. A groundwater mound is noted along the northwest portion of this parcel in the area of GW-8, GW-14, CS1 and CS3. Groundwater flows radially off the mound. The mounding is attributed to the presence of a thick layer of low permeability calcium carbonate at the western portion of Block 1400, Lot 1.	No specific conclusions are provided in report.
	Area A: ASTs containing caustics and vegetable oil were formerly located southwest of Building #16.	Installed and sampled soil borings and one well to evaluate this area. During drilling, indications of fat, oil, grease (FOG) and FPHC are noted to extend to the groundwater table. Analytical results confirm the presence of varying concentrations of FOG and TPHC in soil. cpH-was recorded at levels ranging from above 9 to almost-12., pH of the calcium carbonate material was recorded at 9.99 for all sampled intervals.	No specific conclusions are provided in report.

h.,



Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
<i>Continued - Final Report, Tax Block 1400,</i> Dames & Moore, January 24, 1992	Area B: ASTs containing vegetable oils, tallow and tailings/soap bottoms from hydrolyzer were located south of hydrolyzer and east of west tank field. The tank field area was not equipped with a containment berm and surface runoff from this area flowed to unpaved areas including overflowing of a zipper drain located along the western boundary. An AST containing phenol alkane was formerly located southwest of the S&S tank field.	Installed and sampled 6 soil borings and one well to evaluate this area. Elevated levels of FOG and TPHC are reported to have been detected in all borings, extending to groundwater. A floating hydrocarbon layer is was noted at GW-14 and a sheen was noted with regard to GW-7. Zinc is reported to have been detected in soil samples. No calcium carbonate materials is reported to have been identified in borings from this area.	Report identifies a railroad siding and former oil tanks as potential sources of petroleum in soil. Catalyst material is identified as the likely source of the zinc.
	Area C: An AST used for oleum, waste sulfuric acid and acid wastewater was located northwest of Building #17. A former toluene tank (closed in place in December 1989) is reported to be located in the vicinity of Area C.	Installed and sampled 2 soil borings and 1 well to evaluate this area. Calcium carbonate detected at this area. pH levels are reported to increase with depth, over 8 to over 12.	Conclude washwater did not impact area. pH levels (elevated) are attributed to migration from upgradient sources.
	Area E: A steel UST designed t collect and trap oil and grease present in wastewater stream is located southwest of the S&S Tank Field, near the <u>phenol storage area</u> . Historical information indicates elevated zinc concentrations in wastewater flowing to this trap.	Installed and sampled 3 borings and a well. Investigation indicates that vegetable oil is visibly present in the saturated zone and that FOG and TPHC were detected at varying concentrations in soil samples. Nickel and zinc were detected above background concentrations in soil samples. pH is reported to have been recorded at slightly acidic levels in soil samples.	Conclude that FOG, TPHC and metals are likely to be associated with trap usage. No conclusion is provided for slightly acidic pH.
	Area F1: Open drums containing spent nickel-catalyst and an unknown liquid were noted northwest of Building #16. The asphalt surface in this area was noted to be cracked, stained and deteriorated. A paint shed is reported to have been located west of the drum storage pad.	Miscellaneous fill including calcium carbonate fill is reported to have been identified at this area. pH is recorded between 9 and slightly over 12. FOG and <u>TPHC</u> are reported to have been detected in samples from unsaturated zone PCBs are reported to have been detected in at least one soil sample.	FOG, TPHC, pH attributed to former activities including caustics/alkaline zones found in the calcium carbonate. Recommend, excavation to address PCBs.





Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
<i>Continued - Final Report, Tax Block 1400,</i> Dames & Moore, January 24, 1992	Area F2: Open drums were noted to be present on an asphalt storage pad located east of product unloading terminal and south of fatty acid storage tanks. The asphalt surface in this area was noted to be cracked, stained and deteriorated.	Investigation revealed <u>black staining of soil</u> and elevated readings were recorded during field screening. FOG and TPHC are reported to have been detected in soil samples from the unsaturated zone.	The report concludes that waste oil storage may have impacted this area.
	Area H and Area R (Area H/R): Site plans reportedly identified an area at the northeast corner of the main soap manufacturing area as a rosin storage area. Rosin was produced through the separation of resin from turpentine. A surface water body was originally located at this area and filled with calcium carbonate.	Calcium carbonate material was identified ranging in thickness from 15.5 to 17 feet. Elevated pH levels were recorded in samples and were noted to increase with depth. No turpentine related compounds are reported to have been detected and nickel concentrations are reported to be consistent with background.	Conclude that the highly alkaline zones were the cause of the elevated pH.
	Area P: Pits, used for unloading raw materials from tankers and rail cars, are reported to have been located in alleyways next to the main soap building. The pits are reported to have been closed.	FOG is reported to have been detected and slightly elevated pH levels (approximately 9) recorded in soil samples.	Conclude that the levels of FOG and pH may be from former transfer operations conducted at this area.
	Groundwater: Groundwater was identified as an issue with regard to the southern portion of Block 1400.	Installed and sampled monitoring wells at various locations on Block 1400. FOG and TPHC reported to have been detected in samples from Areas A, B, C, E, F1, F2 & H/R. Free product is reported to have been noted at GW-14 and a sheen was noted on the water surface of GW-10, GW-13, GW-14, GW-17 and CS-1. An elevated pH level was recorded in the sample from CS-1. Lead, nickel and zinc were reported to have been detected in samples from certain wells.	Recommend a groundwater treatment system including pH adjustment, oil/water separation to remove free product, clarification and settling to remove solids and precipitates, and liquid phase carbon adsorption to reduce PHC levels.

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Final Report Soil Environmental Investigation, Tax Block 1309, Dames & Moore, April 20, 1992	This report presents a summary of investigative activities performed to address two AOCs identified on this parcel: Area D Oil Pump House (Bldg S-29) and Area I Fly Ash Storage Area. This report also identifies a 1988 Memorandum of Understanding (MOU) which was executed between Procter & Gamble and the NYSDEC regarding the discharge from the pipes rupture and the referenced "oil lens". This report also provides historical information including information pertaining to the placement of fill materials at Block 1309, Lot 1.	Installed and sampled soil borings installed at Area D and test pits at Area I. Analytical results are compared to "background levels". Groundwater encountered from 2.2 to 9 ft bsg. Generally the groundwater noted to exist in fill material and silt layers.	No specific conclusions provided in report.
	Area D is located south of two fuel oil ASTs in dock area. The ASTs are located in a diked area described as being lined with a synthetic geotextile material. Area D is in the vicinity of previously performed investigation associated with a deak in fuel a oil-transfer piping at the eastern portion of dock. This report references a BB&L Report describing the efforts undertaken to address the fuel oil rupture. The pipe is reported to have been repaired and the contamination associated with the pipe rupture to have been addressed.	Area D: Samples were analyzed for TPHC, FOG, nickel and pH. Nickel and pH were included in the analyses due to information indicating that the pump house area was filled with diatomaceious earth from vegetable oil operations at the site. Results indicated varying concentrations of FOG and <u>TPHC in both unsaturated</u> and saturated zone. Nickel detected in samples. pH recorded at the 8 to 9 range.	Report noted higher concentrations of TPHC and FOG present in upper two feet. Nickel referenced as being at concentrations below levels of concern.
	Area I is located at the northern portion of this parcel and is the location of a temporary fly ash stockpile area. Investigation initiated in response to elevated concentrations of lead (exceeded extraction procedure toxicity) in samples from fly ash. Assert that the elevated lead is from demolition debris containing lead based paint.	Test pits were installed from surface to 3 ft bsg. Fill material (silt, sand mixed with ash, gravel, bricks overlying calcium carbonate) was noted in test pits from this area. Samples from the test pits were analyzed for pH, zinc and lead. pH was recorded at levels of 9 to 10 in fill samples. Zinc and lead, also were detected in soil samples.	Zinc and lead referenced as being at concentrations below levels of concern. Elevated pH attributed to fill, including calcium carbonate.



×

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Final Report Soil and Groundwater Environmental Investigation, Tax Block 1338S, Dames & Moore, April 20, 1992	This report presents a summary of investigative activities performed to address 6 AOCs identified on the southern portion of the Block 1338 parcel: Area G Former Vegetable Oil Tank Farm, Area K Fill Area and Coal Storage, Area M Area East of Edible Oils Buildings #52-56, Area N Former Vegetable Oil Fat Trap, Area P1 Former Product Unloading Pit and Area Q1 Existing Scale Pit. The report also provides historical information including information pertaining to the placement of fill materials at the southern portion of Block 1338 and identifies that spent diatomaceous earth from edible oil refining and spent nickel catalyst from edible oils are the by-products of the "food area". The report references a geophysical survey performed by Blackhawk Geosciences which identified USTs at Area M, specifically east of Buildings #53/54 and east of Building #56.	Soil and groundwater investigation consisting of the installation and sampling of soil borings and wells is reported to have been performed at each of these AOCs. Based on the groundwater investigation performed at the southern portion of Block 1338, groundwater at this portion of the site is reported to exist at depths ranging from 2.2 to 9 feet bsg and to flow toward Bridge Creek	No specific conclusions provided in report.
	Area G: ASTs containing vegetable oil and caustics were formerly located at this area. <u>Nickel-catalyst-was-stored in this</u> area after tanks were dismantled. An investigation is reported to have been undertaken due to cracking and expansion joints in the concrete pad at this area.	Investigative efforts did not reveal any free phase vegetable oil but did identify black staining of soil in this area. Nickel, lead and zinc are reported to have been detected below background levels. pH was recorded at levels of 9 to 10 in surface and subsurface samples.	No specific conclusions provided in report.
	Area K: Fill is reported to have been placed in the southeastern portion of this parcel in the area of Buildings #74 and #75. In addition, this area is reported to have been used for coal storage. Also, an unknown black material was found during the foundation investigation for Buildings #74 and #75.	Installed and sampled soil borings and wells.	No specific conclusions provided in report.
	Area M: ASTs containing vegetable oil and caustics were present at the area east and southeast of Buildings #52 and #56. Also, unloading pits and railroad sidings are reported to have been present at this area. Fill is reported to have been placed at this area. UST(s) may also have been present in this area.	Installed and sampled 5 soil borings and 1 well at this area. Analytical results revealed the presence of low levels of TPHC and FOG in soil samples. Nickel is not reported to have been detected at an elevated concentration and pH was recorded at levels ranging from 8 to above 10. The report does not identify the location(s) of any UST(s) at this area.	No specific conclusions provided in report.



Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - Final Report Soil and Groundwater Environmental Investigation, Tax Block 1338S, Dames & Moore, April 20, 1992	Area N: A vegetable oil fat trap, "super fat trap", is located south of Building #56. An oil/water separator system including a UST, now filled with coarse aggregate, also is located in this area.	Installed and sampled soil borings which revealed the presence of <u>black staining of soil</u> . FOG was detected in soil samples and pH was recorded at relatively neutral levels. Nickel was detected below background.	No specific conclusions provided in report.
	Area P1 - Concrete pits were formerly located at the bottom of the rail siding unloading area, east of the Edible Oils Building. The pits were filled in and capped with asphalt/concrete.	Area P1: Low concentrations of TPHC and FOG were detected in soil samples. pH was recorded at levels ranging from almost 7 to slightly over 9.	No specific conclusions provided in report.
	Area O: This area is an existing scale pit and includes equipment for weighing trailers and rail cars at the site. Construction records indicate that the pit is constructed of concrete and is 10 feet deep.	Area O: <u>FPHC</u> and FOG were detected in soil samples and pH was generally recorded in the 7 to slightly above 8 range.	No specific conclusions provided in report.
	Groundwater was considered of concern with regard to the southern portion of Block 1338.	Groundwater: Installed and sampled 5 wells at the southern portion of Block 1338. Samples were analyzed for TPHC, FOG, zinc, lead, nickel, and pH. Report identifies isolated incidences of elevated TPHC concentrations and notes that higher concentrations are away from the production areas of this portion of Block 1338. Elevated concentration of lead and zincOne monitoring well, PG-PA-MW- 10D was included within The Operable Unit 2 report.	States that the presence of TPHC in wells upgradient of production areas suggests that contaminants may be from off -site sources. State that TPHC has had a limited impact on groundwater. Overall Remedial Approach included in report states that the tar-like material (with elevated levels of TPHC may be impacting groundwater.

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Final Report Soil and Groundwater Environmental Investigation, Tax Block 1338N, Dames & Moore, April 20, 1992	This report presents investigative actions performed at two AOCs: Area L Filled Area (southeast of Building #64) and Area Q2 Former Scale Pit located at the northern portion of Block 1338. The report indicates that <u>paints and solvents were</u> :likely used in refurbishing operations at an old copper shop. Recent operations are identified as warehousing in Buildings #80, #60, #67N and #67S.	Investigation included the installation and sampling of soil borings and wells. Also performed a geophysical survey to identify USTs. The survey is not successful due to metallic interference from railroad tracks, metal piping, etc. Groundwater at the portion of the site occurs at 5.5-8.5 feet bsg and primarily in miscellaneous fill. Groundwater flow is reported to be to the southwest.	No remedial action is proposed to address either AOC or the northern portion of Block 1338.
	Area L: A sludge pond is reported to have been located south of Building #67 and southeast of Building #64. The report indicated that investigation was necessary to evaluate the type of materials utilized to fill the sludge pond. Also, investigation efforts were undertaken to evaluate impacts from a copper shop.	Installed and sampled two soil borings and a monitoring well. Some petroleum staining of soil-is-noted in one boring. The report references the recording of elevated pH levels in soil samples.	The report concludes that the investigation did not identify impacts to the area from former uses and did not support that the areas had been used as a sludge pond. Also concludes that the elevated pH may be associated with caustics.
	Area Q2: A truck scale was previously operated at the area west of Building #60. The scale is reported to be constructed of concrete.	Results do not identify the presence of TPHC or FOG and pH was recorded in the 6 to 8 range.	No remedial action is proposed based on analytical results.
	Groundwater was considered an area of concern with regard to the northern portion of Block 1338.	Wells were installed and sampled. TPHC and FOG were not detected at elevated concentrations in groundwater. Nickel,-lead-and-zinc-were detected in one site monitoring.well.(GW-5) from this area.	No remedial action proposed for groundwater.

39

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
<i>Results of Sampling</i> <i>for Toluene and</i> <i>Metals,</i> Recon Systems, Inc., December 11, 1992	The report presents and summarizes sampling performed to delineate toluene and TPHC contamination in groundwater and to supplement a previously completed feasibility study.	In December 1992 samples were collected from 10 wells: GW-7, GW-10, GW-11R, GW-12, GW-14, GW- 17, RS-1, CS-3, Code Well and MW-5 (across Richmond Terrace). Samples from 5 wells (GW-10, GW-11R, RS-1, Code Well and MW-5) were analyzed for VO. Field measurements (pH, temperature and conductivity) were recorded for all 10 wells and dissolved oxygen was recorded for five wells.	This report concludes that this round of sampling confirms the results of previous sampling rounds and states that the presence of ₁ toluene will be addressed as part of the groundwater treatability study. No further action is proposed for metals as concentrations are below NYC sewer discharge levels.
	VO analysis of groundwater samples.	<u>Toluene was detected in samples from 3 of the well</u> samples tested for VO compounds.	The report states that the December 1992 sampling round indicates that toluene contamination is centered at GW=11R.
	Metals analysis of groundwater samples.	Samples from all 10 wells were analyzed for cadmium, chromium, copper, cyanide, lead, mercury, nickel and zinc. Low concentrations of copper and zinc are reported to have been detected in all wells. Chromium and nickel are reported to have been detected in some of the wells.	All concentrations of metals are reported to have been below NYC sewer discharge levels.
	pH assessment of groundwater samples.	The level of pH is reported to have been outside the acceptable federal drinking water range of 6.5-8.5 in four wells: Code Well, RS-1, CS-3 and GW-14.	The results are reported to confirm previous sampling rounds with regard to pH.



Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - Results of Sampling for Toluene and Metals, Recon Systems, Inc., December 11, 1992	TPHC analysis of groundwater samples.	Samples from two wells, GW-12 and GW-17 were analyzed for TPHC. TPHC was detected in the magnetic sample from GW-12 and was not detected in the sample from GW-17. (NOTE): Insufficient information was made available to identify the locations on former locations of all above listed wells. Generally, wells are/were located on the northern portion of Operable Unit 1, central portion of Operable Unit 2 and southwestern portion of Operable Unit 4.	The level and extent of the TPHC is reported to be consistent with results of previous investigations.
UST Storage Tank Removal and Site Assessment Report, Recon Systems, Inc., February 19, 1993 (Draft Version)	This report provides a summary of removal efforts for nine USTs. The report also includes an Appendix, which consists of information associated with five of the nine tank removals performed by CODE Environmental. The CODE report is listed as a separate report in this table. The Recon report also includes a letter from Recon to the NYSDEC informing them of the intended removal of three tanks (one 8,000 gallon tank at Building #20 and two 10,000 gallon tanks at Building #56), which had never been included on the tank registration for the facility. These tanks are reported to have been identified through a review of historical site plans. It appears likely that these tanks identified in the letter were removed as part of the closure effort described in this report. It should be noted that the two 10,000 gallons described in this report. According to the report, a representative of the NYSDEC Water Program witnessed the closure efforts for all tanks.	The following USTs are reported to have been closed: one 8,000 gallon No. 6 oil-UST at Building #20; two 8,000 gallon No. 6 oil-UST and one 8,000 gallon No. 2, oil.UST at Building #56; one 1,000 gallon diesel fuel UST at Building #18 (Excavation A); one 2,000 gallon unleaded gas UST at Building #12 (Excavation B); one 3,000 gallon diesel UST at Building #32 (Excavation C); and, one 12,500 gallon No. 6 oil UST and one 12,500 gallon No. 2 oil UST at Building #32A (Excavation D). The closure included removal of tanks, removal of soil (based on field screening), the collection and analysis of post-excavation samples and the restoration of each tank area via the placement of clean fill. Some dewatering is reported to have been performed and resultant materials collected and transported from the site for disposal at an appropriate facility.	The report states that all accessible contaminated soil was removed from tank areas. No- <u>exceedences</u> are reported with regard to <u>VO</u> compounds and only a <u>few-exceedences</u> are reported with regard to <u>CPAH</u> compounds.

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - UST Storage Tank Removal and Site Assessment Report, Recon Systems, Inc., February 19, 1993 (Draft Version) –	Removal of one 8,000 gallon UST-containing No6 oil from the Building #20 Area.	Building #20 Excavation: A'8,000 gallon UST	No additional actions were recommended for this area.
	Removal of two 8,000 gallon UST containing No. 6 oil and one. 8,000 gallon tank containing No. 2 oil from the Building #56 Area.	Building #56 Excavation: Two-8,000 gallon USTs containing No6 oil and one 8,000 gallon UST containing No2 oil were removed. Based on the presence of stained soil and oil sheen on the groundwater, 325 tons of soil were removed from the tank area. Due to the presence of electric lines, some contaminated soil was left in place. The matter was assigned NYSDEC Number 920-3754. Six post excavation samples were collected from the interval immediately above groundwater and analyzed for BN+15. Analytical results from the sample collected below the electric line revealed the presence of CPAH- compounds in excess of NYSDEC standards. BN compounds were either not detected or were detected below cleanup standards in the other samples.	No additional actions were recommended for this area.

where it years

- 41

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - UST Storage Tank Removal and Site Assessment Report, Recon Systems, Inc., February 19, 1993 (Draft Version)	Removal of one 1,000 gallon UST-containing diesel-fuel from the Building #1B Area.	Building #1B Excavation: A-1-000 gallon-UST_ containing diesel-fuel-was removed. Contaminated soil was encountered during the removal effort and approximately:50 tons of soil is reported to have been removed from the tank area. The matter was assigned NYSDEC Number 920-3697. Four post-excavation samples were collected from the interval immediately above groundwater and analyzed for BN+15 and VO+15. Analytical results revealed the presence of CPAH: compounds in excess of NYSDEC standards in two of the four samples.	No additional actions were recommended for this area.
	(Removal of one 2,000 gallon-UST-containing unleaded gas from the Building #12 Area.	Building #12 Excavation::A:2;000 gallon-UST containing unleaded gasoline was removed. No: contaminated soil or holes were observed during the removal. Four post-excavation samples were collected (three from the excavation and one from along the supply line) and analyzed for VO. The concentrations are reported to have been below cleanup standards.	No additional actions were recommended for this area.
	Removal of one-3,000 gallon UST containing diesel fuel from the Building #32 Area.	Building #32 Excavation: A-3;000 gallon: UST. enclosed in a vault was removed and approximately '50 tons of soil were removed from the tank area. The matter was assigned NYSDEC Number 920-3697 (same number as Building 1 Excavation). The excavation was extended to groundwater and is reported to have been limited by the presence of an electric line along the eastern portion of the tank area. Two post-excavation samples were collected from the interval immediately above groundwater and analyzed for BN+15 and VO+15. No targeted BN or VO compounds were detected. Low concentrations of VO TICs were detected.	No additional actions were recommended for this area.

- - / ?



Report	Report Topic	Description of Activities and Analytical	Report Conclusions
Identification	Area(s) Of Concern	Results ⁽²⁾	
Continued - UST Storage Tank Removal and Site Assessment Report, Recon Systems, Inc., February 19, 1993 (Draft Version)	Removal of one 12;500 gallon UST containing No. 6 oil and one 12,500 gallon UST containing No. 2 oil from the Building #32A Area.	Building #32A Excavation: Two 12,500 gallon USTs were removed and approximately 75 tons of soil were removed from the area surrounding the tank. The matter was assigned NYSDEC Number 920-4269. The excavation was extended to groundwater and is reported to have been limited by the presence of buildings on three sides and an electric line. All	No additional actions were recommended for this area.

-43

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Site Assessment Summary Report Closure of Underground Storage Tank Systems, CODE Environmental Services, September 1992 (included in Appendix 1 of Recon UST Report, dated February 19, 1993)	Area(s) Of Contern This report provides a summary of the removal efforts, undertaken for 5-USTs: one.1,000 gallon UST formerly used to store diesel oil; one.2,000 gallon UST storing gasoline; one. 3,000 gallon UST containing diesel oil; one.12,500 gallon UST containing fuel oil; and one.12,500 gallon UST containing fuel oil. This report references a different sampling regime than described in the February 1993 Recon report. The report identifies a closure approval dated June 22, 1992. This report is provided as an appendix to the February 1993 Recon report.	Results Tanks and impacted soil, if any, were removed from five site locations in June/July 1992. One 1:000-gallon-steel tank formerly used to store diesel fuel was removed. Approximately 160-170-gallons of diesel-fuel and sludge present in the vault encasing the UST were removed and drummed for disposal. Samples are reported to have been collected from the sides and bottom of the excavation and analyzed for TPHC. One 2:000-gallon-steel tank located at Building #12 and used to store gasoline was removed. The tank was encased in concrete with concrete and soil overlying same. Samples are reported to have been collected from the sides and bottom of the excavation and analyzed for TPHC and BTEX. The NYSDEC ordered the excavation backfilled in July 1992. One 3:000-gallon-steel tank located at Building #32 and used to store diesel fuel was removed. During excavation activities, it was determined that a leak from the feed lines had impacted surrounding soil The NYSDEC was notified (920-3697) of the discharge and the excavation was backfilled at the direction of the NYSDEC No reference to sampling is included in the discussion. Two 12.500 gallon steel tanks, one used to store No.2	No conclusions were provided in the report.
		(fuel oil and one used to store No. 6 oil, were removed. The tanks were encased as well as being horizontally cross-braced with large steel I beams. The No. 6 oil tank was grouted and embedded in the building abutment.	

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Area F Soil Remediation Report, Recon Systems, Inc., March 16, 1993	This report describes soil excavation and sampling performed to- address previously delineated PCB contamination in soil at Area- tE The report states that Area F was first identified as an area of concern during a SI performed by Dames & Moore and subsequently the extent of the PCB contamination was delineated through a soil boring investigation performed by Recon in 1992. A report documenting the delineation activities is reported to have been prepared and submitted to P&G in June 1992.	Excavation activities were performed in February 1993. The excavation boundaries are reported to have been based upon the results of a soil boring investigation performed in 1992 and to have been centered about sample:FB-3 which reported the highest PCB-, concentration of 150 ppm. The excavation was extended to a depth of approximately 3 feet bsg. Approximately 150 cubic yards (221-tons) of soil was excavated and nine post-excavation samples, were collected from the resultant excavation area. (PCBs were either not detected tor were detected below the minimum detection limit in 5 samples. Detectable levels of Aroclor-1254 were identified in the remaining four samples with the highest concentration recorded at 0.49 ppm, below the NYSDEC standard for PCBs of 1 ppm.	No further action was proposed for Area F.
Site Assessment, Soils Delineation and Impact to Groundwater in Area K at the Port Ivory Facility, Recon Systems, Inc., October 15, 1993	Report describes a groundwater investigation undertaken to determine if groundwater in monitoring wells near Area K (GW- 16 and 1) had been impacted. The report states that soil investigations performed by Dames & Moore and Recon identified TPHC, VOCs and BNs in soil samples from Area K. This report references a November 1992 report by Recon Results of Soil Investigation in Areas F and K. This report was not provided for HMM's review. However, the October 1993 report states that the 1992 report provides a summary of delineation efforts at Area K. The delineation efforts at Area K are reported to have included 54 test pits, field screening and collection/analysis of 17 soil samples. The effort revealed the presence of "elevated" levels of TPHC in soil samples collected from areas exhibiting a black tar-like substance. The October 1993 report reiterated the conclusion of the 1992 report and stated that the noted hydrocarbons were likely to be immobile due to their high viscosity but indicated that a groundwater investigation was necessary to confirm this conclusion.	In December 1992, Recon obtained samples from wells GW-16 and GW-1. Samples were analyzed for PP+40 including cyanides and phenols. Analytical results are reported to have been below NYSDEC action levels except for cyanides, 2(-1,-1-dimethyl)phenol; arsenic, chromium, copper, lead and zinc. The:levels of the above listed contaminants are reported to have been within one order of magnitude of corresponding NYSDEC action levels. To confirm results, the wells were re-sampled-in-March-1993 for cyanide, arsenic, chromium, copper, lead and zinc. Analytical results revealed similar levels of the noted contaminants.	The report asserted that residential <u>exposure</u> from the subsurface contamination would be <u>minimal so long as the soil</u> , was not disturbed. Also, stated that soil bound hydrocarbons have not impacted groundwater at this portion of the site. Further, states that the metals in groundwater may be from fill rather than industrial activities. No further action is proposed for groundwater since it is not used for potable purposes.

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Environmental Site Assessment Summary Report of Tax Block 1400, Recon Environmental Group, October 18, 1994	According to this report, environmental due diligence studies were performed to characterize environmental conditions of this parcel and that all issues have been addressed at this parcel. The report states that P&G has completed several projects to eliminate site contamination and that the one remaining active project is a groundwater remediation project which is described in this report. The report indicates that the proposed groundwater recovery system would induce a constant flow across the site thereby mobilizing compounds that are adsorbed to soil. These mobilized compounds can be recovered and treated thereby remediating soil.	The previously identified concerns and response actions, as presented and described in this report, are as follows: Bridge Creek Calcium Deposits; Former Raw Product and By-product AST Areas; Wastewater Treatment; Drum Storage; Former Rosin Storage Area; Representative Railroad Switch and Equipment Areas; Product Unloading Areas; Closure of UST Systems; Wood Yard; Building 20; and Groundwater Sampling and Analysis.	Groundwater
	Bridge Creek Calcium Deposits	Two investigations were performed to determine the sources and extent of the white precipitate in Bridge Creek. Studies involved sediment and groundwater sampling and analysis. Results of both studies revealed high pH levels and the conclusion was that the material was calcium carbonate.	This report states that the high pH will be addressed through the proposed groundwater remediation program.
	Former Raw Product and By-product AST Areas	Three AST Areas (Areas A, B & C) were investigated by Dames & Moore in 1992. Each area is reported to have been investigated with soil borings and at least one monitoring well. Analytical results from soil samples are reported to have indicated levels of FOG, TPHC, pH and zincGroundwater results are reported to have indicated elevated levels of FOG, TPHC, pH, zinc and lead. All ASTs are reported to have been removed. This report also comments that a UST used to hold toluene near Area C was closed in- place and filled with concrete in 1989.	The report states that elevated concentrations of contaminants in groundwater will be addressed through the proposed groundwater remediation program.





Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - Environmental Site Assessment Summary Report of Tax Block 1400, Recon Environmental Group, October 18, 1994	Wastewater Treatment Drum Storage	The S&S Fat Trap (Area E) handled wastewater from the hydrolyzer building. Soil borings and a well were installed at this area. Analytical results revealed the presence of elevated-concentrations of FOG, TPHC, nickel and zinc:	The report states that elevated concentrations of contaminants in , groundwater will be addressed through the proposed groundwater remediation program.
	Drum Storage	Area F1 (Spent Nickel Catalyze Drum Storage Area) and Area F2 (Waste Oil Drum Storage Area) were evaluated through the installation and sampling of soil borings and wells. Analytical results from Area F1 revealed the presence of elevated levels of pH, TPHC, FOG, zinc and PCBs. Analytical results from Area F2 revealed elevated levels of FOG and TPHC. Additional sampling was performed to delineate the extent of the PCBs detected in soil at Area F1. Subsequently, soil excavation was performed to address the PCBs.	-PCB contaminated soil was removed and no. further action is necessary based on post-excavation sampling.
	Former Rosin Storage Area	This area, Area H, was investigated through the installation and sampling of soil borings and a well. Elevated pH is reported to have been recorded in soil and groundwater.	This report states that the <u>high pH</u> will be addressed through the proposed groundwater remediation program.
	Representative Railroad Switch and Equipment Areas	Representative railroad switch, tie and equipment (Area O) is reported to have been sampled by Dames & Moore. Reportedly, the investigation did not identify any negative impact associated with the railroad equipment. The specific sample location was not identified.	No actions are proposed for railroad equipment on this parcel.



Table 2Summary of Historical Environmental Reports and Information
Site 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - Environmental Site Assessment Summary Report of Tax Block 1400, Recon Environmental Group, October 18, 1994	Product Unloading Areas	Concrete lined pits which have been filled in and capped with asphalt or concrete were formerly used for <u>unloading raw</u> product from tankers and rail cars. These pits were evaluated through the collection of soil samples. Analytical results indicated elevated levels of FOG and pH.	Conclude that induced groundwater flow from the groundwater treatment system will remediate these soils.
•	Closure of UST Systems	The report states that Recon and CODE supervised and documented the decommissioning of the following USTs: 1;000 gallon diesel (B1B), 2;000 : gallon gasoline (B12), 3,000 gallon diesel (B32), 12,500 gallon #2 (353) 12,500 gallon #6 (354) and a 8,000 gallon #6 (Building #20). Impacted soil is reported to have been removed from the former B1B, B32, 353, and 354 and some impacted soil is reported to have been left in place adjacent to Buildings #20, #32, #32A and #56 due to the presence of buildings and/or utilities.	Conclude that no further action is necessary given that the source(s) and the majority of the contaminated soil was removed.
	Wood Yard	Historical maps are reported to identify a 1 to 2 acre wood yard which had contained a coal gasification raw- material storage area prior to the 1950s. This is reported to be discussed in a 1991 McLaren Hart report which was not provided to HMM during the document review. The area is reported to have been investigated to, determine if any coal tar residue had impacted soil or groundwater. The investigation revealed elevated levels, of TPHC, VO and BN in soil and TPHC, BN and phenols in groundwater.	Groundwater quality will be addressed in the proposed groundwater remediation program.

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Continued - Environmental Site Assessment Summary Report of Tax Block 1400, Recon Environmental Group, October 18, 1994	Building #20	Building 20 is reported to have been utilized as a locomotive repair shop. Analysis of samples from the stained soil floor indicated elevated levels of TPHC, VO; BN; metals and low levels of PCBs. A McLaren Hart report (1992) is reported to have concluded that the sampling results did not contain any contaminants above cleanup guidance values or that would pose a threat to human health. The 1992 McLaren Hart report was not provided to Killam during the document review.	No actions were proposed for this area.
	Groundwater Sampling and Analysis	Floating product is reported to have been observed on the water surface of-wells on Block 1400 and elevated levels of pH are reported to have been recorded with regard to groundwater samples. Reportedly, Dames & Moore and McLaren Hart recommended a groundwater investigation and remediation program (free-phase product removal and pH neutralization) and, Recon performed an investigation which included testing to delineate the high pH, toluene and product plumes on this parcela and a pump test to evaluate hydraulic parameters for use in a preliminary design.	Conclude that groundwater remediation (coalescing oil/water separator, air stripper and acid addition to address TPHC, toluene and high, .pH) is warranted.

Table 2Summary of Historical Environmental Reports and Information
Site 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

-120

Sector - Conception -

Report	Report Topic	Description of Activities and Analytical	Report Conclusions
Identification	Area(s) Of Concern	Results ⁽²⁾	
	Groundwater Contamination	Groundwater remediation: This report states that <u>Recon-was.going to develop:a preliminary treatment</u> design to be utilized in permit negotiations with New York City. The proposed design scheme was to include 10 recovery wells pumping water to 3 input wells in the treatment system. Water from-three wells.contaminated.with-TPHC-was to be pumped to an oil/water separator and water from the two wells exhibiting elevated-levels-of-toluene, was to be pump to an equalization tank. The effluent from the oil/water separator and the air stripper was to be mixed, in an equalization tank, with water from the wells from the <u>carea</u> with_high_pH. From the equalization tank, the water was to be pumped to an existing pH control system. An inline static mixer was to be added along with an acid addition system as the primary pH control and the existing pH control system was to be used as a backup. It was proposed to discharge the treated effluent to the sewer.	Report concludes that groundwater remediation is_needed to address TPHC, toluene and pH.

manuta mite

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Landfill Closure Construction Certification Report, Levine-Fricke- Recon (LFR), July 18, 1997	Documents the field procedures implemented to achieve physical closure of the P&G landfill in accordance with 6 NYCRR Part 360 and the landfill closure plan dated January 1997. This report also includes permits, correspondence, disposal documentation and cover material certification associated with the landfill closure. The report states that P&G previously demonstrated the non-hazardous condition of the landfill and, as allowed on a case-by-case basis, P&G had demonstrated that specific landfill closure requirements in Section 360-2.15 Landfill Closure and Post Closure Criteria were not applicable. Therefore, NYSDEC is reported to have addressed the closure according to Section 360-2.14 Industrial/Commercial Waste Monofills which allows for closure requirements to be modified based on pollution potential of waste.	The approved closure activities included site clearing to remove surface debris, brush clearing, placement of one foot of cover and the establishment of vegetation. Materials removed from the landfill area included the following: scrap metal, tires, telephone poles, railroad ties, vegetative debris and one box of sharps.	No additional actions are proposed for the landfill with the exception of the post-closure groundwater monitoring and maintenance.
Landfill Cover Maintenance Manual and Groundwater Monitoring Plan, LFR, April 14, 1998	Describes maintenance and groundwater monitoring for closure of the C&D Landfill located on Block 1309. This report provides maps which depict the landfill area, the locations of 7 wells and groundwater contours.	No investigative actions are included in this report. The report sets forth a five year sampling and maintenance program including all 7 monitoring wells (MW-1,2,3,4,5,6 and DW-1) located within the landfill. The proposed maintenance plan includes a semi-annual inspection to ensure the integrity of soil cover and vegetation.	No conclusions are provided in this report.
Landfill Closure Plan, LFR, April 14, 1998	This report documents the closure of the landfill at the Port Ivory facility in accordance with NYCRR Part 360. The report states previous investigation(s) revealed that soil and groundwater are free of significant contamination and therefore do not pose a threat to human health or the environment.	No activities performed in conjunction with this report.	Closure will include a deed restriction

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Report Identification	Report Topic Area(s) Of Concern	Description of Activities and Analytical Results ⁽²⁾	Report Conclusions
Update on the Report on the Recommended Treatment System for Groundwater Contaminated with NAPL, Toluene and High pH, Recon Systems, Inc. March 28, 1995 Amendment to Remove Economic Information, May 13, 1999	The report presents updated information pertaining to the proposed treatment system for groundwater contaminated with NAPL, toluene and high pH:	The report does not include any additional testing activities. Rather, the report provides an updated design based on data generated since issuance of previous design report in 1993. The changes to the design system include fewer recovery wells due to a reported NAPL dissipation (one area of concern remaining) and diminished extent of the high pH area as well as increased water hardness.	The report concludes that recent sampling results necessitate revision to the previously described treatment system. The revised design calls for fewer recovery wells, elimination of the oil/water separator, addition of a sludge thickening system (if needed due to recent high hardness measurements) and a scaled down stripper system. Also, economic information is referenced as having been removed from this report.
Investigation of Calcium Deposits, Blasland, Bouck & Lee, September 1999	According to this report an area on the western side of the site, along Bridge Creek, was formerly occupied by calcium carbonate drying beds. In addition, several ASTs containing caustic materials were located approximately 250 feet east of Outfall G. White precipitate is reported to have been noticed several times along the banks. In response to the noted precipitate, P&G is reported to have initiated a pH level monitoring program. The purpose of this investigation was to identify and map the extent of the precipitate occurrences in Bridge Creek and attempt to determine the source area of the precipitate.	The investigation/study included the following: collection and analysis of sediment samples from the bed of Bridge Creek; collection of water samples from selected outfalls that intermittently discharge to the creek; installation and sampling of 7 wells; water table measurements hydraulic conductivity testing; hydrochemical sampling (pH, conductivity and temperature); and review of previously recorded pH values. Samples collected as part of this investigation were analyzed for indicator inorganic constituents (chloride, sulfate, nitrate, fluoride, chromium, arsenic, barium, cadmium, calcium, cyanide, iron,	The report concludes that groundwater with an elevated pH exists over much of the study area and that the flow of the high pH groundwater through the subsurface lime deposits has resulted in the dissolution of the deposits and the release of calcium products.

ananan --- a naraadagadag

Table 2Summary of Historical Environmental Reports and InformationSite 2A/2B: HHMT - Port Ivory Facility⁽¹⁾

Ŧ

53

Identification Area(s) Of Concern	Results"	
lead zind cald gro elev Out G. wit rec sim Cro hac res sys in I tha	ead, manganese, mercury, copper, silver, sodium, inc and selenium). The report also includes alculations estimating potential rate of discharge to roundwater into Bridge Creek. Two areas of levated pH were identified through this study, Outfall G Area and an area 500 feet north of Outfall G. The second area is presumed to be associated with a groundwater seepage point. The levels of pH ecorded between 1986 and 1989 were generally imilar. Investigation revealed that pH of Bridge Creek was historically elevated and that the levels ad been declining since 1985/6 due to a delayed esponse to the installation of an underground piping ystem at the AST area in 1984. Given the similarity npH levels between 1986 and 1989, it was concluded hat the precipitate either stabilized or is forming at	The discharging of this calcium enriched groundwater into surface water exhibiting a lower pH may cause the precipitation and deposition of calcium salts. Furthermore, the soils and groundwater reflect many of the chemical parameters indicative of the saline to brackish waters natural to Bridge Creek.

NOTES:

(1) Information provided in this table is as presented in the listed reports.

(2) Activities and results are as described in the reports. All activities were performed by or on behalf of P&G.

Site 2A/2B Report



A summary of the listings as well as commentary regarding the basis for the listings, as feasible and appropriate, is provided in Table 3. It should be noted that HMM contacted the NYSDEC with regard to the site's inclusion on the NYSDEC Inventory of SHWS. Based on the discussion, it was determined that the site had been included on the SHWS Inventory based on the presence of a "potential" C&D landfill situated on Future Site 4. As P&G characterized and closed the C&D landfill in accordance with NYSDEC regulations, it did not appear appropriate for the site to be included in the SHWS inventory based on the criteria for that database. As such, HMM requested that NYSDEC provide written confirmation of the delisting of the subject site from the NYS database. A copy of the correspondence issued by the NYSDEC was included in Appendix A of the Site 1 Report.

The NYSDEC LTANKS List includes listings for five UST removals within Site 2A. No USTs are reported to have been present at Site 2B. The tank removal efforts at Site 2A were performed by P&G,

several years before the transfer of ownership to the Port Authority. The locations of the former USTs are as follows: east of Building 20 (case number 920-3451); south of Building 12 (no case number); east of Building 32 (case number 920-3697), and two USTs located east of Building 32A (case number 920-4269). The UST in the vicinity of Building 20 had a capacity of 8,000 gallons, contained #6 heating oil and was situated within a concrete vault. Approximately 200 tons of soil is reported to have been removed from the tank area to address impacted soil. Soil is reported to have been excavated to the groundwater table; however, due to the proximity of Building 20 a portion of the vault and some contaminated soil was left in place. A 2,000 gallon unleaded gasoline UST was removed from the Building 12 Area. No discharge or impacted soil is reported to have been observed at this area. A 3,000 gallon UST containing diesel fuel was removed from the Building 32 Area and two USTs, (one 12,500 gallon UST containing #6 oil and one 12,500 gallon UST containing #2 oil), were removed from the Building 32A area. As the UST removals were reported to have been performed in accordance with NYSDEC procedures and with NYSDEC oversight, as appropriate, no additional investigative efforts were included in the SI. Please note, a discussion of former UST issues is provided in Section 4.2.1. The inclusion of the subject site on the remainder of the above listed databases will be addressed as part of overall HHMT-Port Ivory Facility redevelopment.

4.2 Sampling Program

The AOCs identified at the site through performance of the Phase I ESA are as follows: USTs, Fill Material, Previously Identified Soil and Groundwater Contamination (i.e., AOCs identified and investigated by P&G and described in environmental reports prepared for P&G), the Closed C&D

81

Table 3 Summary of Environmental Database Listings Site 2A/2B: HHMT - Port Ivory Facility

Database	Database Date	Additional Information
USEPA, Resource Conservation Recovery Information System (RCRIS) Facilities - Large Quantity Generators (LQG) List	December 12, 1999	The subject site is listed on the USEPA, RCRIS Facilities - LQG List dated December 12, 1999. Review of this site listing notes that P&G is permitted as a LQG (Record Date August 13, 1980) and assigned USEPA ID Number NYD000249961. One violation appears to be associated with this site listing and is associated with the requirements Compliance Evaluation Inspection. P&G is reported to have complied with these requirements on September 25, 1986. Based on review of the site listing, it appears that no outstanding violations are associated with the site's listing as a LQG.
The NYSDEC Inventory of Hazardous Disposal Sites (SHWS) List	April 1999	The subject site's inclusion on NYSDEC, HSWDS List dated April 1999 is associated with the presence of the C&D Landfill on Block 1309. This listing also identifies that P&G maintains an USEPA Identification Number NYD980507537 and operates a wastewater treatment system to control pH in the sanitary waste stream. After some acidulation occurs, the sludge from the treatment system is reported to be removed from the subject site. No other off-site disposal activities are identified in this listing. The listing comments that the abandoned landfill reported to be on-site does not have a liner or a leachate collection system and that P&G disposed of wastes, generated from their manufacturing processes, on-site. A consent order, executed in March 1992, is identified in this listing. Further, the consent order is reported to have required site investigation and closure (in accordance with Part 360) of the landfill. This investigation is reported to be currently under review. Although information provided by representatives of DEC have confirmed that the landfill was closed in accordance with prevailing regulations and that the case is considered closed by the Department. Post-monitoring requirements were performed by P&G and are currently being performed by the Port Authority. HMM has contacted the NYSDEC regarding the site's inclusion on this list and has been informed that the site should no longer be included in the SHWS Inventory. At the request of HMM, the NYSDEC has issued a letter stating that the site should be de-listed.
NYSDEC, Petroleum Bulk Storage Database (UST) List	April 2000	The listing identifies three USTs (PBS Number 2-600767) formerly located on the subject site. One 8,000 gallon and two 10,000 gallon USTs containing 1,2 or 4 fuel oil are reported to have been closed/removed in August 1992. Tanks are reported to have been constructed of steel/carbon and associated piping is reported to have been constructed of steel/iron.
NYSDEC, Chemical Bulk Storage Database (CBS UST) List	January 2000	This listing notes that P&G formerly utilized oner10,000-gallon UST,-was registered under CBS Registration Number 2-000128, for the storage of toluene. The tank is reported to have been installed in January 1950 and its current status is noted as "temporarily out of service/closed-in place". No date for the closing of the tank was

Table 3 Summary of Environmental Database Listings Site 2A/2B: HHMT - Port Ivory Facility

Database	Database Date	Additional Information
		provided in the EDR Listing. The tank and piping are reported to be constructed of
		steel/carbon steel and situated within a secondary containment vault. According to
		<i>P&G</i> , contamination was identified in conjunction with the former toluene tank area.
		Please note, the toluene tank was not specifically evaluated as part of the site
		investigation since P&G indicated it was a closed issue with the NYSDEC. However,
		investigation actions were performed in the vicinity of the former toluene tank.
NYSDEC, Chemical Bulk Storage Database	January 2000	This listing notes the subject site formerly maintained nine ASTs under CBS
(CBS AST) List	· · · · · · · · · · · · · · · · · · ·	Registration Number 2-000128. All tanks are reported to have been closed.
NYSDEC, Major Oil Storage Facilities	January 2000	This listing notes the subject site formerly maintained eight USTs under MOSF
Database (MOSF UST) List	-	Facility Identification Number 2-2160. The facility status is listed as inactive.
		The tanks ranged in size from 550 gallons to 12,000 gallons and all are reported
		to have contained petroleum products (fuel oil, diesel or unleaded gasoline). The
		listing indicates that all of the:USTs were removed with NYSDEC oversight and
		does not identify any outstanding required actions.
YSDEC, Major Oil Storage Facilities	January 2000	This listing notes the subject site formerly maintained five ASTs under MOSF Facility
Database (MOSF AST) List		Identification Number 2-2160. The facility status is listed as inactive. Three tanks
		with capacities of 550, 275 and 250 gallons are reported to have contained diesel fuel-
		and two tanks, each with a capacity of 420,000 gallons are reported to have contained
· · · · · · · · · · · · · · · · · · ·		No1,-2 and 4 fuel oil.
USEPA Facility Index System (FINDS) List	October 1999	The FINDS List typically contains "pointers" and information indicating that the site
		is listed on other database sources within RCRIS. Review of this site listing notes
		other pertinent environmental site listings to include listings on the Aerometric
		Information Retrieval System, Facility System (AIRS/FS), Enforcement Docket System
		(DOCKET), National Compliance Database (NCDB) and Section Seven Tracking
		System (SSTS).
NYSDEC Spills Information Database	January 2000	The site is listed on the NYSDEC-SPIELS three times The first case, Spill Number
(Spills) List		8907474, is associated with a discharge that occurred on October 26, 1989. The spill
		is reported to be associated with the detection of toluene, contamination discovered
		during the analysis of soil samples obtained from the toluene tank area during closure
		of the UST. The listing identifies that the NYSDEC was informed of the discharge and
		that this agency_closed-the-spill case citing that same all not pose an immediate
		aanger to neatin and the environment; the spill case was closed on August 14, 1990.
		The issuing comments that P&G asserted that the contamination was confined to an
		upper aquijer situated on top of a timestone layer. The second spill, Spill Number
		outronorted amount of an unraported material from a varial into the Kill Van Kull A
		anreported amount of an unreported material-from a vessel into the Kill Van Kull. A
		creanup_contractor_is_reported to have been catted to the site and handled-the



Database	Database Date	Additional Information
		remediation of same. The spill case was closed by the NYSDEC on November 28, 1986. The third_spill, Spill Number 8906834, was noted to be associated with a <u>simulated_exercise</u> involving P&G, the New York City Police Department and the NYSDEC conducted on October 12, 1989. No actual materials are reported to have been discharged to environmental media. The spill case was closed the same day. As all three of the above spill cases were reported to the NYSDEC, investigated by same and eventually closed by this agency, no site investigation activities appear to be warranted with regard to the spills. Please note, this workplan includes the performance of investigative activities in the area of the former toluene tank.

Notes: Database information is provided in an electronic database search, performed by EDR in May 2000.

57
Landfill, Railroad Tracks and Sidings, Surface Staining, Pits and Drains, Former Structures, Listing of the Site (P&G Port Ivory Facility) in Environmental Databases, Area Sites of Concern (i.e., sites of known environmental concern in the vicinity of the subject site), Wetlands, Asbestos-Containing Materials, and Lead-Based Paint. The objective of the investigative/sampling effort was to develop a better understanding of year 2000 site conditions, including levels of contaminants present in various environmental media (soil, groundwater, sediment and surface water). A description of the individual AOCs present within the limits of Site 2A/2B and the investigative actions proposed to evaluate each AOC are provided in the following sections. In addition, descriptions are provided for site-wide AOCs to the extent that such are relevant to Site 2A/2B. Please note, no investigative efforts were included for three AOCs identified in the Phase I ESA: (1) Area Sites (i.e., sites of known environmental concern in the vicinity of the subject site); (2) Wetlands; and, (3) Asbestos-Containing Materials/Lead-Based Paint. The Port Authority has addressed issues associated with the site's inclusion in environmental databases as part of the overall acquisition of the property. Further, no efforts were undertaken for the Closed C&D Landfill since this AOC is not related to Site 2A/2B.

The SI for soil included the collection of discrete 6-inch samples and, to the extent possible, the SI soil boring and well locations were biased toward areas exhibiting indications of contamination and sample selection was based upon the results of field screening with a bias toward the interval(s) exhibiting indications of contamination. The SI also included the collection and analysis of soil samples from beneath the water table due to unique strata identified only below saturated depths at certain locations.

4.2.1 USTs

Hatch Mott

MacDonald

According to P&G, no active USTs or oil/water separators were present at the HHMT-Port Ivory Facility site in 2000. However, USTs were formerly utilized at the subject site to store toluene and various petroleum products including ethanol, diesel fuel, No. 2 fuel oil, No. 6 fuel oil and unleaded gasoline. P&G also utilized grease traps and oil/water separators in process operations. The environmental database report indicates that P&G closed or removed eight USTs containing various fuel products; five of these tanks were located at Site 2A. Based on the information in reports provided by P&G, five USTs were removed (12,500 gallon #2 oil; 12,500 gallon #6 oil; 8,000 gallon #6 Oil; 2,000 gallon unleaded gasoline: and a 3,000 gallon diesel) from Site 2A (by P&G). According to P&G, the tank closures for Site 2A were performed in accordance with NYSDEC regulations and with NYSDEC oversight, as appropriate. A summary of the tank information included in the database report for the HHMT-Port Ivory

58

✻

×

Site 2A/2B Report



Facility is provided in Table 3. In addition, UST removal/closure efforts undertaken for tanks located at Site 2A are described below. Information provided in P&G reports is presented in Table 2. Given that the removal/abandonment actions were performed with NYSDEC oversight, no SI actions were proposed for soil at the former UST areas. However, the site-wide groundwater investigation included a review of groundwater quality at locations throughout Site 2A.

Former Tank Areas

- Building 20: One concrete vaulted 8,000-gallon UST containing #6 oil was removed from the area east of Building 20. The presence of stained soil was observed during the tank removal and was addressed through the removal of approximately 200 tons of impacted soil from the tank area. NYSDEC assigned case number 920-3451 to the closure/removal effort. Due to the proximity of the tank to the foundation of Building 20, some impacted soil and a portion of the containment vault were left in place. Four post-excavation samples were collected from the 0-6 inch interval above the groundwater table and analyzed for BN+15. Analytical results revealed the presence of CPAH compounds in excess of NYSDEC standards (NYSDEC Standards/guidance in place at the time of removal) in three out of four samples. The excavation is reported to have been backfilled with clean soil.
- Building 12: One 2,000-gallon steel UST containing unleaded gasoline was removed from the area south of Building 12. No contaminated soil was observed during the tank removal and no holes were observed in the tank upon it removal. Post-excavation samples confirmed concentrations below applicable cleanup standards. The P&G report did not recommend any additional efforts with regard to the tank removal. The excavation was backfilled with clean soil.
- Building 32: One 3,000-gallon concrete vaulted UST containing diesel fuel was removed from the area east of Building 32. Approximately 50 tons of soil was removed from the area surrounding the UST based upon visual signs of staining. The closure was assigned case number #920-3697. The excavation was extended to the groundwater table to address impacted soil. However, remedial efforts were limited due to the proximity of underground utilities along the eastern sidewall and building foundations. Two post-excavation samples were collected from the 0-6 inch interval above the groundwater table and analyzed for volatile organics compounds (VOCs) and semi-volatile organic compounds (SVOCs). No contaminants are reported to have been detected at concentrations in excess of NYSDEC standards. The excavation was backfilled with clean soil.

59



- Hatch Mott MacDonald
- Building 32A: Two 12,500-gallon USTs were removed east of Building 32A. One UST was utilized for the storage of #6 oil and the other was utilized to store #2 oil. Approximately 75 tons of soil were removed from the area surrounding the USTs to address visually impacted soil. The closure was assigned case number #920-4269. The excavation was extended to the groundwater table to address visually impacted soil. However, remedial efforts were limited due to the presence of building foundations and underground utilities. All accessible impacted soil was reported to have been removed. Four post-excavation samples were collected from the 0-6 inch interval above the groundwater table. The samples were analyzed for BN+15 and VO+15. No targeted BN compounds were detected. Low concentrations of target VO compounds, below regulatory levels, were detected in one sample. The excavation was backfilled with clean soil.

As previously presented, P&G stated that the five UST closure were performed in accordance with NYSDEC procedures and with NYSDEC oversight, as applicable. Tanks containing ethanol and enzymes were also reported to be present at Site 2A, generally at locations north/northeast of Building 13 and west of Building 31. No closure documentation was provided by P&G, however, information obtained from P&G representatives indicated that these tanks had been removed and that no discharges had been reported with respect to the tanks. Based on historical mapping, it appeared that the locations of the tanks might correspond with one of the potential UST Areas, UST3. Thus, the area was to be reviewed in conjunction with the UST3 Area investigation described below.

In addition to "known" former tank areas, HMM's review of reports and Sanborn Maps revealed the potential for additional USTs to be present at nine (9) locations across the entire site (UST1-UST9). Four of the potential tank areas, UST1, UST3, UST4 and UST7 were identified at Site-2A. Therefore, based on available information, it was proposed to perform a ground penetrating radar (GPR)/electromagnetic (EM) survey at each of the nine potential UST areas (multiple tanks were identified at five of the nine potential tank areas). The proposed SI also included the installation and sampling of soil borings at areas where the GPR/EM survey identified potential tanks. The need to perform laboratory analyses for soil samples was to be based upon the results of field screening and the type of analysis was to be based upon former tank contents, if known. In those instances where the contents of potential tanks could not be established, it was proposed to analyze samples for total petroleum hydrocarbons (TPHC) and the target compounds list (TCL) including volatiles and semi-volatiles, target analyte list (TAL) metals, and polychlorinated biphenyls (PCBs). Although a site-wide groundwater investigation was proposed as part

- 60



of the SI for this site (Section 4.2.9), it was proposed to perform groundwater investigation activities, as necessary, at potential UST Areas. Specifically, it was proposed to convert one soil boring per potential tank area to a temporary well, as necessary and feasible, to assess groundwater conditions in the vicinity of any identified USTs. Analysis of groundwater samples from temporary wells was to be based on former contents of the tanks. However, in the absence of such information, it was proposed to analyze groundwater samples for TPHC and TCL. The nine potential tanks areas, UST1 through UST9, are identified on Figure 6. The potential UST Areas located within Site 2A are: UST1, UST3, UST4, and UST7. The specific SI activities implemented for soil at the four potential tank areas, located within Site 2 are described in Sections 5.2 and 5.3.1.

4.2.2 Precipitate at Bridge Creek

Reports provided by P&G identified the presence of one or more types of precipitates along the banks of Bridge Creek and described various efforts (inspection of the creek bed, performance of chemical and physical testing of the noted precipitates) undertaken to evaluate the noted precipitates. Bridge Creek extends along the western border of Site 1 and the western border of the southern portion of Site 2A. The reports provided by P&G indicated that the noted material had the potential to be associated with prior filling activities at the site. The reports did not identify a significant environmental issue with regard to the presence of the precipitate. However, precipitate at Bridge Creek was included in the proposed SI to evaluate current (year 2000) conditions relative to this issue. Specifically, it was proposed to evaluate current conditions with regard to the noted precipitate through visual review and the collection and laboratory analysis of sediment/precipitate samples and surface water. The initial phase of the proposed investigation was to include a visual reconnaissance of the creek bed at both low and high tides on two separate occasions (i.e., two low tide and two high tide inspections). In addition, it was proposed to obtain representative samples of precipitate, if any, noted to be present and to obtain surface water samples from Bridge Creek to identify current (year 2000) water quality. The number and location of precipitate and surface water samples were to be dependent upon the conditions observed during the proposed visual reconnaissance. All samples, precipitate and surface water, were to be submitted for TAL Metals and pH analysis. The initial SI activities performed to evaluate this AOC are presented in Section 5.6. As precipitate was only observed at locations adjacent to Site 1, no samples were from Site 2A as part of the SI. Thus, a discussion of analytical results from the Year 2000 SI is provided in the Site I Report. However, additional efforts were undertaken to evaluate Bridge Creek as part of the RI for Site 2A/2B. These additional efforts were performed proximate to Site 2A and, thus, results are presented



herein. Please refer to Section 8.3 for proposed remedial investigation sampling and to Section 9.3 for analytical results associated with the proposed surface water and sediment sampling at Bridge Creek.

4.2.3 Previously Identified Soil and Groundwater Conțamination (P&G AOCs)

Reports provided by P&G identified numerous AOCs. Table 2, Summary of Historical Environmental Reports and Information, provides pertinent information associated with the AOCs identified by P&G. The reports identify and describe remedial efforts undertaken by P&G with regard to the following areas/issues: the C&D Landfill, the presence of PCBs in soil at Area F1, USTs and a groundwater investigation including PG-GW-1. The C&D Landfill, situated on Future Site 4 (Block 1309, Lot 10), is not included as part of the VCP Program as regulatory oversight is provided by the NYSDEC Division of Solid Waste pursuant to the landfill closure. Area F1 is situated on Site 1 and a discussion of actions undertaken by P&G at Area F1 is provided in the Site 1 Report. Overall, the reports provided by P&G identified that contaminants and/or elevated pH were detected/recorded in one or more soil and/or groundwater samples collected from the majority of these AOCs. In addition, some of the available reports commented upon the presence of black staining in the soil and free-phase floating product on the water surface in monitoring wells.

Previous reports referenced the closure and removal of underground storage tanks located site wide and specifically at Site 2A. As previously presented in this report, the closure efforts for the USTs at Site 2A include: a 10,000 gallon #6 tank (Building 20); a 2,000 gallon unleaded gasoline UST (Building 12); one 3,000 gallon diesel UST (Building 32); one 12,500 gallon #6 oil UST (Building 32A); and one 12,500 gallon #2 oil (Building 32A). Actions taken in regards to these UST areas are described in Section 4.2.1.

Except as detailed for USTs, the P&G reports do not identify or describe any remedial actions undertaken, by P&G, to address contaminants identified in soil at Site 2A; P&G did not specifically identify the presence of contamination in soil or groundwater at Site 2B. Rather, P&G asserted in reports, that the contaminants detected in soil at Site 2A, as well as the rest of the site, are relatively immobile and that residential (human) exposure would be minimal so long as the soil was undisturbed (i.e., contaminants in soil do not present a risk with regard to contact). The elevated pH levels in groundwater were attributed to certain fill material and free-phase product (free product) was attributed to prior usage of vegetable oils and other petroleum products. Overall, P&G indicated that no actions were necessary with regard to site groundwater given that groundwater was not utilized for potable purposes at the site or in the immediate surrounding area. However, a few of the reports from the early 1990s included recommendations to

63

2

N



address product and elevated pH in groundwater at Block 1400 (Sites 1 and 2A) including the central and southern portions of Site 2A.

Given the identification of contaminants in soil and groundwater at the site and the length of time that had elapsed since P&G's investigative efforts (the majority of sampling was performed in the early 1990s), it was proposed to perform SI sampling activities for both soil and groundwater at the areas identified as AOCs by P&G. The AOC designations that are located in Site 2A are as follows: Area B, Area E, Area FZ, Area P, the eastern portion of Area A, and the Rail Yard. The AOC designation that is located in Site 2B is Area Q1. The locations of the P&G AOCs are presented on Figure 6. The number of samples proposed for each of P&G's AOCs was based upon the contaminants detected during P&G's investigations, the level of completeness of reports relating to individual AOCs, historical information provided through review of Sanborn Maps and historic aerial photographs and site conditions at the time of the Phase I ESA. Please note, the identification letters/names assigned to the AOCs by P&G have been utilized in this report to provide easy reference to investigative efforts described in P&G reports. Table 2 \checkmark provides a summary of information contained in previous environmental reports.

For the purposes of the SI, fill material was regarded as a separate site issue. A discussion of site-wide fill material and investigative efforts proposed to address same, as related to Site 2A/2B, are presented in Section 4.2.8. Given the presence of fill throughout the site, the SI was designed to integrate the evaluation of the fill material with P&G AOCs as well as other AOCs identified as part of the Phase I ESA.

Typically, the depth of an investigative soil boring would be based upon the type of issue(s) identified at each AOC. However, given the presence of fill material, the SI utilized all soil borings to evaluate and characterize fill material and individual AOCs. As such, the SI included the installation of soil borings to a depth of approximately 15 feet below surface grade, regardless of AOC, to evaluate fill material. This approximate depth was deemed sufficient given that no information had been obtained to indicate that contaminants at the P&G AOCs exist at depths greater than 15 feet. Although it was proposed to base the analytical suite for each AOC upon the results of field screening, it was assumed that the samples would be analyzed for TCL volatiles organic compounds (VOCs), TCL semi-volatile organic compounds (SVOCs), TAL metals, pesticides, PCBs, TPHC, oil & grease (O/G), pH, total cyanide and phenolics.

With regard to previously identified contaminants in groundwater, it was proposed to obtain and analyze groundwater samples to establish current (year 2000) groundwater quality. The groundwater component of the SI is presented in Section 4.2.9.

4.2.4 Railroad Tracks and Sidings

Visual inspection of the site identified the presence of railroad tracks and sidings. In addition, review of historical records revealed that additional tracks and sidings were formerly present at the HHMT-Port Ivory Facility. A 1994 summary report stated that some limited testing was performed to evaluate representative railroad switches, ties and equipment. The reports concluded that testing of the representative railroad equipment did not reveal any "negative impact". However, insufficient information was provided in available reports to determine if prior evaluations adequately assessed railroad tracks and sidings. As such, it was proposed to obtain samples from locations adjacent to representative portions of the on-site railroad system to confirm that the railroad system had not impacted soil at the HHMT-Port Ivory Facility. The NYSDEC does not have an established program for the evaluation of current or former railroad systems, therefore, it was proposed to select sample locations based on current conditions as well as information presented on Sanborn Maps and aerial photographs sets.

To maximize the time and cost efficiency of the proposed sampling effort, it was proposed to integrate the sampling for this AOC with that designed for other AOCs and the site-wide fill evaluation. The SI included the installation and sampling of approximately 27 soil borings to evaluate this AOC. The sampling program established that 17 of the 26 borings proposed to evaluate this AOC would also be utilized to evaluate other AOCs. All 27 soil borings would be utilized as part of the site-wide fill evaluation.

Based on the current and former locations of railroad tracks and sidings, the SI proposed to install 12 of the 27 soil borings at Site 2A and one of the 26 soil borings at Site 2B. As previously stated, it was proposed to install all soil borings to a depth of approximately 15 feet below surface grade. The sampling______ proposed for this AOC included the collection of samples from a discrete 6-inch interval within the upper four feet of soil. The sampling program included an analytical suite comprised of TPHC, VO+10, base neutral (BN) compounds, PCBs and TAL metals.

Q.



4.2.5 Staining

Visual inspection of the site revealed the presence of staining adjacent to and on the soil flooring within Building 20 (Site 2A) and adjacent to Building 60B (Site 3). Based on visual observations, the proposed SI included the collection of representative samples from each visually stained area including two locations within Building 20. It was proposed to collect a minimum of one sample per stained area. It was proposed to install 5 of the 6 soil borings proposed for this AOC on Site 2A/2B. Specifically, all 5 soil borings are proposed to be installed on Site 2A. The sampling program included an analytical suite comprised of TCL volatiles organic compounds (VOCs), TCL semi-volatile organic compounds (SVOCs), TAL metals, pesticides, PCBs, TPHC, oil & grease (O/G), pH, total cyanide and phenolics.

4.2.6 Pits and Drains

Pits and drains, most sealed or filled with gravel, were noted at both interior and exterior site locations. In addition, P&G reports identified the presence of oil/water separator systems and described limited investigative efforts performed to evaluate conditions at and near oil/water separator systems. These reports identified the presence of contaminants in environmental media in samples from the oil/water separator areas but concluded that the concentrations of contaminants detected did not warrant remedial actions. Given the aforementioned, the SI included a review of pits and drains through visual inspection, as possible, as well as through the installation and sampling of soil borings.

Sampling was proposed at 28 locations at or adjacent to pits and drains identified in the field and/or through review of reports and historical information sources. Eleven soils borings were to be installed at Site 2A. It was acknowledged that it would not be possible to accomplish the proposed soil sampling at a portion of the 28 locations due to the presence of structures and utilities. As described in previous sections, it was proposed to integrate the sampling program for pits and drains with the sampling programs designed to address other AOCs and the fill evaluation. Specifically, it was proposed to utilize all soil borings for the fill evaluation and 11 of the 28 soil borings for other AOC investigations. As proposed for other AOC investigations, all soil borings were to be installed to a depth of 15 feet below surface grade as part of the fill evaluation. With regard to soil sampling for pits and drains, it was proposed to obtain representative samples from a discrete 6-inch interval within the upper six feet of the soil. Soil samples would be analyzed for TPHC, TCL, pH and oil and grease. Specific sample selection within the designated interval was to be based upon the results of field screening.



X

67

4.2.7 Former Structures

Review of Sanborn Maps and aerial photographs revealed the presence of former structures, ASTs and railroad tracks and sidings at various locations throughout the subject site. With regard to Site 2, review of historical information sources revealed the following: the presence of additional structures (buildings and tanks) at the Rail Yard; the presence of buildings and structures east and south of Buildings 12 and 13; existing buildings and structures (Building 30s and 40s); Building 100 (Furnace); and the former structure, Building S-35. In addition, review of historical information sources also revealed the presence of discolored areas, debris piles and possible fill material at various site locations. The discolored areas, debris piles and sidings are discussed in Section 4.2.8, Fill Material. Concerns associated with former railroad tracks and sidings are discussed in Section 4.2.4. The stained areas noted in Building 20 are discussed in Section 4.2.5.

Given the above, the SI included the installation and sampling of soil borings at former building and AST areas. The purpose of the sampling proposed for this AOC was to evaluate areas formerly utilized as part of process operations as identified through the presence of structures, storage areas, etc. It should also be noted that some of the P&G AOCs include areas of former structures, in particular, ASTs. Sampling efforts for P&G AOCs are described in Section 4.2.3 of this report. As with other AOCs, the sampling proposed to evaluate former structures was integrated with the proposed sampling for other AOCs and fill material. Please note, the vast majority of the sampling proposed for other AOCs represented investigation of prior activities including some type of structure (ASTs, structures, etc.). However, based on the locations of former structures and debris piles, it was proposed to install and sample nine additional soil borings to address this AOC. Three of the nine additional soil borings (FS-1, FS-2 and FS-3) were to be installed at Site 2A; the location of proposed soil boring FS-1 was located in close proximity to the boundary line between Site 1 and Site 2A. With regard to sample selection for former structures, it was proposed to obtain representative samples from a discrete 6-inch interval within the upper four feet of the soil and to analyze the samples for TPHC, TCL, pH and oil and grease. As previously stated, it was proposed to utilize all soil borings for the evaluation of site-wide fill material. Therefore, it was proposed to advance all soil borings installed to evaluate this AOC to a depth of 15 feet below surface grade. Specific sample selection within the designated interval was to be based upon the results of field screening.



4.2.8 Historic Fill Material

According to representatives of P&G and information provided in reports, P&G placed a variety of fill material at the HHMT-Port Ivory Facility to raise the grade for site development. The fill materials present at the site include non-native soil/sand, construction debris (wood, bricks, glass, concrete); ash, from the furnace operations, slag, vegetative debris; and by-products from on-site production activities (calcium carbonate, spent diatomaceous filter earth, and spent carbonaceous filter material). The specific composition of the fill is reported to vary with location. Information from P&G's investigations indicate that elevated pH as well as some contaminants detected in samples from the site, both soil and groundwater, may be attributable to the fill material. Reports provided by P&G also describe the presence of black staining at a few locations throughout the HHMT-Port Ivory Facility.

No comprehensive report has been provided which summarized the locations and concentrations of fill material, contaminants both related and unrelated to fill material, and/or the reasoning behind the occurrences of "black staining". Thus, the SI included a site-wide sampling program to assess current site soil conditions and to identify the limit(s) of fill material. As the NYSDEC guidance documents do not provide sampling frequency and/or analytical requirements for the investigation of fill, the sampling program referenced the New Jersey Department of Environmental Protection (NJDEP) program for general guidance.

The NJDEP has stipulated a minimum frequency of four samples per acre to establish the presence of fill material. However, the NJDEP guidance documents recognize that on larger sites a lower frequency of samples provides sufficient site coverage with regard to the evaluation of fill. In most cases, the NJDEP has accepted a sampling frequency of one sample per acre at larger sites. Given the number of soil borings being installed to evaluate other AOCs and the intent to utilize all of these borings for information pertaining to fill material, it was proposed to install and sample soil borings at locations not otherwise evaluated through the overall sampling program. Specifically, it was proposed to install and sample 25 additional soil borings to provide adequate site-wide coverage with regard to fill. Five of the additional 25 soil borings (Fill-1 through Fill-5) were to be installed in Site 2A while two (Fill-9 and Fill-10) of the additional 25 soil borings were to be installed in Site 2B. In total, the evaluation of other AOCs included the installation and sampling of 97 soil borings. Therefore, the site-wide fill evaluation include a total of 120 soil borings are to be based upon GPR/EM survey results (See Section 4.2.1). Additionally, the five



Hatch Mott MacDonald

borings slated for visual review of the former sludge pond at Area L at Site 3 were not included in the evaluation of fill material.

Forty soil borings were proposed to evaluate fill at Site 2A/2B. Based on information provided in P&G reports, it was proposed to advance soil borings to a depth of approximately 15 feet below surface grade. To determine the types and extent of fill material at the site, it was proposed to perform a visual assessment of soil conditions at each soil boring location. Further, it was proposed to obtain samples from each type of fill material and submit for laboratory analysis to determine contaminants, if any, that may be present within the fill material. The goal of the fill evaluation program was to determine the extent and nature of the various fill material reported to be present at the site. As such, the proposed program included the analysis of a representative number of samples from each type of fill material noted to be present at the site, regardless if the fill was situated within the saturated zone.

4.2.9 Groundwater

Previous investigative efforts performed at the site identified the presence of contaminants and elevated pH in site groundwater. In addition, the presence of free product and/or a sheen on groundwater was identified at a few locations on Site 2A. As the majority of the groundwater sampling presented in the P&G reports was collected in the early 1990s, it was proposed to perform a groundwater investigation for the purpose of identifying current groundwater quality.

The initial phase of the groundwater investigation program proposed for the overall site included the sampling of a representative number of existing wells and the installation and sampling of additional groundwater wells. To establish the number of useable wells at the site, it was proposed to perform a physical inspection of existing wells as well as to identify the presence of free product and to record, to the extent possible, water levels for all existing wells. The groundwater sampling program assumed that a minimum of 12 monitoring wells would be determined to be in adequate condition (i.e., suitable for. sampling).

Based on information regarding groundwater quality and the presence of fill material provided in P&G reports, it was proposed to install and sample 15 shallow monitoring wells at locations at the interior and around the perimeter of the site. Five of the 15 wells were to be located on Site 2A. In addition, given that information provided by P&G indicates that a confining layer exists below the noted fill material at locations throughout the site, it was also proposed to install eight deeper monitoring wells to evaluate

 \star



groundwater quality below the confining layer. The deeper wells were to be situated, to the extent possible, adjacent to eight of the proposed shallow wells to establish well couplets at eight site locations. Two of the well pairs were to be installed on Site 2A and one of the well pairs was to be installed at Site 2B. Upon completion of well installation activities, it was proposed to collect samples from a portion of the existing wells (assumed to be 12 site wells including 4 wells on Site 2A and one well on Site 2B) and all newly installed wells (assumed to be 23 wells with 4 wells installed on Site 2A and one well to be installed on Site 2B). Prior to the performance of sampling, it was proposed to redevelop existing monitoring wells included in the proposed sampling program. The groundwater samples would be analyzed for TPHC, TCL, oil and grease and pH.

As stated in Section 4.2.1, temporary wells proposed for UST areas were based upon the results of GPR/EM and soil investigation activities. The groundwater investigation described above does not include temporary wells installed to evaluate potential UST areas.

4.3 QA/QC and Health and Safety

The Port Authority has developed protocols for field sampling, which are designed to protect the health and safety of on-site personnel and minimize public exposure. These protocols ensure that data generated from field efforts meet required QA/QC standards and result in data that is reproducible, accurate, representative, comparable and complete. The protocols are presented in the Port Authority's *Field Standard Operating Procedures Manual* dated January 1995. Thus, the ESIW proposed to perform all field-sampling activities in accordance with the Port Authority's QA/QC and Health and Safety protocol's as presented in the Port Authority *Field Standard Operating Procedures Manual* dated January 1995. If desired by the NYSDEC pursuant to the VCP Program, the Port Authority will provide a copy of the *Field Standard Operating Procedures Manual*.

In accordance with Port Authority protocols, it was proposed to utilize Hampton-Clarke, Inc./Veritech Laboratories (NY certification number 11408) of Fairfield, New Jersey for laboratory services associated with the SI. As appropriate, field protocols for the SI are described and/or referenced in Section 5.0. In accordance with NYSDEC requirements, data will be evaluated with Division of Environmental Remediation Data Usability Summary Report (DUSR) guidelines. The DUSR associated with the SI will be provided under separate cover.



5.0 SI – FIELD INVESTIGATION

This section describes the SI activities undertaken to evaluate the AOCs identified at Site 2A/2B. Due to the site-wide nature of many of the AOCs, numerous sample locations were utilized to evaluate multiple AOCs at Site 2A/2B. Based on information from historical sources and previous environmental reports, a variety of fill material was placed at the site. As such, all soil borings installed at Site 2A/2B were utilized as part of the site-wide fill evaluation. The investigation also included the laboratory analysis of a number of samples collected from intervals below the water table. As stated in Section 4.2, the purpose of sampling below the water table was to better characterize historic fill material present at the site sampling unique strata situated below the saturated zone.

This SI included investigation of soil and groundwater at Site 2A/2B. Samples were collected from sediment and surface water of Bridge Creek as part of the site-wide SI, however, as previously stated the samples were collected adjacent to Site 1. Thus, the results of the SI surface water and sediment sampling were presented in the Site 1 Report. The soil component of the Site 2A/2B SI consisted of the installation of 40 soil borings and the collection of 74 soil samples for laboratory analysis, excluding UST area samples. Due to the presence of reinforced concrete and/or utilities, it was not possible to fully install and sample seven of the soil borings proposed for Site 2A/2B. The soil borings that could not be fully installed and sampled were as follows: Fill-6, FS-2, RR-12, RR-13 and RR-14 from Site 2A and Q1-2 and PA-MW-10 from Site 2B. Given the comprehensive nature of the SI and the overall sampling frequency at Site 2A/2B, the Port Authority proposed to review field information and analytical results and determine if additional efforts would be necessary at these seven locations.

The potential UST investigation included the performance of a GPR/EM survey, the installation of seven soil borings resulted in the collection of 12 soil samples from the soil borings installed at the four potential UST areas on Site 2A/2B. One temporary monitoring well was installed and sampled as part of – the UST investigation. In total, the SI for soil at Site 2A/2B included the installation of 47 soil borings and the collection of 86 soil samples.

A minimum of one sample was collected and submitted for laboratory analysis from all soil borings with the exception of the temporary well and the two deeper wells installed at locations PAMW-7D and PAMW-15D. The purpose of the temporary well was to obtain information regarding groundwater quality at the potential UST area. The installation of the deeper wells was to establish shallow/deep well

 \star

Site 2A/2B Report



Hatch Mott

pairs at certain site locations for use in the groundwater evaluation effort. Given the close proximity of other soil borings to the three well locations (i.e., PA-TMW-01, PAMW-7D, PAMW-15D), no additional soil sampling was deemed warranted. Please note, soils were reviewed during boring/well installation activities and no unusual soil conditions were noted with regard to these locations.

The groundwater portion of this investigation included converting 5 soil borings into groundwater monitoring wells, installing one temporary monitoring well, recording water levels from all newly installed wells and four existing wells, reviewing wells for the presence of free product and the collection and laboratory analysis of 10 groundwater samples from five newly installed wells, four existing wells, and one temporary well. Two samples were obtained and analyzed from PG-PAMW-15D due to the presence of sediment. Traditional groundwater samples were not obtained from monitoring wells GW-14 and OP-1 due to the presence of free product, however, samples of the free product were submitted for identification.

A summary of the investigative actions and sampling activities performed as part of this SI is presented in Table 4. Please note, the table is organized by AOC and includes a brief summary of the types of issues identified through the performance of the Phase I and the supplemental file review, identification of the actions and sampling efforts undertaken to evaluate each AOC, soil boring and sample reference/identification numbers and, as appropriate, analytical parameters. Soil boring and well locations for Site 2A/2B, as feasible, are presented on Figure 7.

All sampling and field investigation activities were performed in accordance with the Port Authority's *Field Standard Operating Procedures Manual* dated January 1995 and NYSDEC protocols. All sample analyses were performed by a New York State certified analytical laboratory (Hampton-Clarke, Inc./Veritech Laboratories (NY certification number 11408). Field screening for VO vapors utilizing a photo-ionization detector (PID) was performed during the sampling activities and was utilized in sample selection as well as in overall site characterization.

It should be reiterated that the site was not in operation at the time of the inspection; therefore the sampling program was based, to a large extent, on information from documents provided by P&G. HMM did not observe operations and therefore could not assess issues associated with daily operating practices including housekeeping, hazardous material and petroleum storage, etc.



,



Hatch Mott MacDonald

AOC	Description of Issues	Description of Actions and Sampling	Sampling Methodology
Potential USTs	Sanborn Maps identified nine areas which may include USTs.	GPR/EM Survey performed at each area to attempt to identify tanks. 16 soil borings at the site with 7 soil borings from site 2A/2B were installed: UST1-2, UST2-1, UST2-1A, UST2-1B, UST2-2, UST2-3, UST4-1, UST4-2, UST5-2, UST6-2, UST6-3, UST7-1, UST7-1A, UST7-1B, UST7-2 and UST9-1.	Soil E418.1, SW6010, SW7471, SW8081, SW8082, SW8260, SW8270, SW9014, SW9045, SW9065, SW9071
		30 soil samples from the site with 12 samples from site 2A/2B were submitted for laboratory analysis: UST1-2(12- 14), UST1-2(2-4), UST2-1(6-7), UST2-1(8-10), UST2- 1A(0-2), UST2-1B(2-4), UST2-1B(4-6), UST2-2(4-6), UST2-2(10-12), UST2-3(2-4), UST2-3(12-14), UST2-3(8- 9), UST4-1(14-15), UST4-1(2-4), UST4-1(10-11), UST4- 2(12-14), UST4-2(4-6), UST5-2(4-6), UST6-2(4-6), UST6- 2(8-10), UST6-2(16-18), UST6-3(14-16), UST6-3(1.5-2), UST7-1(8-10), UST7-1A(0-2), UST7-1B(2-3.5), UST7- 2(8-10), UST7-2(10-12), UST9-1(8-10) and UST9-1(2-4). 2 temporary wells from the site with 1 temporary well from site 2A/2B were installed and sampled. TMW-01 and TMW-02	Groundwater E624, E625, E200.7, E245.2 E150.1, E418.1, E1664, E335.2, E420.1, E608
Precipitate at Bridge Creek	Investigative efforts by P&G identified the presence of a precipitate material along the banks of Bridge Creek.	The portion of Bridge Creek located along the western side of the site was visually reviewed during two low tide and two high tide periods. Sediment/precipitate samples and surface water samples were collected and analyzed. 5 sediment samples were submitted for laboratory analysis: SED-1, SED-2, SED-3, SED-4 and SED-5. 3 surface water samples were submitted for laboratory analysis: SW-1, SW-2 and SW-3.	Sediment SW6010, SW7471 Surface Water 200.7, E245.2, 335.2



AOC	Description of Issues	Description of Actions and Sampling	Sampling Methodology
P&G AOCs	Historical reports identified AOCs at the subject site which had been evaluated, to some degree, by P&G. Information pertaining to AOCs (Areas A through I, Areas K through R and the Wood Yard) is described in Table 2. Soil borings were installed and sampled at these areas. The soil boring and sample references for each AOC are listed below. Groundwater actions are described under the groundwater AOC.	Soil borings were installed and sampled. 58 soil borings from the site with 13 soil borings from site 2A/2B were installed: A-1, A-2, A-3, A-4, A-5, A-6, B-02, B-2 (B-02A), B-1, B-3, B-4, PAMW-1, D-1, D-2, D-3, D- 4, D-4A, E-1, F1-3, F2-2, G-2, G-3, G-4, G-5, G-5A, G-6, G-7(N), G-8, G-9, G-10, H/R-1, H/R-2, H-R-3, I-1, K-1, K- 2, L-1, L-2, L3(FILL), L-4, L-5, L-6, M-01, M-2, M-3, M- 4, M-5, MW-04, PAMW-4, P-1, P-2, P-3, Q1-1, WOOD- 01C, WOOD-3, WOOD-5. 107 soil samples from the site with 25 soil samples from site 2A/2B were submitted for laboratory analysis: A-1(2- 4), A-2(6-8), A-2(2-4), A-2(0-2), A-3(10-12), A-3(6-8), A- 3(2-4), A-4(12-14), A-4(6-8), A-5(2-4), A-6(01-3), B-1(2- 4), B-1(6-8), B-1(9-10), B-02(2-4), B-02(6-8), B-02A(8- 10), B-3(2-4), B-3(6-8), B-4(2-4), PAMW-1(2-4), PAMW- 1(4-6), PAMW-1(10-12), D-1(0-2), D-1(6-8), D-1(18-20), D-2(0-2), D-2(6-8), D-3(0-2), D-4(0-2), D-4A(6-8), E- 1(0.2-2), E-1(4-6), E-1(10-12), F1-3(1-3), F1-3(3-5), F2- 2(2-4), F2-2(8-10), G-2(0-2), G-2(4-6), G-2(6-8), G-3(0-2), G-4(6-8), G-5(4-6), G-5A(8-10),G-6(4-6),G-6(6-8),G- 7(N)(8-10), G-7(N)(10-12), G-8(1-2), G-8(6-7), G-9(4-6), G-10(2-4), H/R-1(1-3), H/R-1(3-4), H/R-2(0-1.5), H/R- 2(1.5-3.5), H/R-3(0.3-1), H/R-3(1-3), I-1(0-2), I-1(2-4), K- 1(2-4), K-1(5-6), K-2(0-2), K-2(2-4), L-1(2-4), L-1(6-8), L-2(8-10), L-2(10-12), L3FILL(2-4), L3FILL(8-10), L3FILL(12-14), L-4(0-2), L-4(6-8), L-5(2-4), L-5(8-10), L- 6(6-7.5), L-6(7.5-8), M-01(0-2), M-01(2-4), M-2(2-4), M- 2(4-6), M-3(2-4), M-4(2-4), M-4(4-6), P-1(2-4), P-1(8- 10), P-2(2-4), P-2(4-6), P-3(2-4), P-3(6-8), Q1-1(2-4), Q1- 1(4-6), WOOD-3(4-6), WOOD-3(6-8), WOOD-5(0-2), WOOD-5(2-4), WOOD-5(4-6), WOOD-5(6-8), WOOD-5(6-8), WOOD- 5(8-10) and WOOD-5(14-16).	Soil E418.1, SW6010, SW7471, SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071
Area A West Tank Field (Southwest of Building 16)/Block 1400		6 soil borings from the site with 2 soil borings from site $2A/2B$ were installed: A-1, A-2, A-3, A-4, A-5, A-6, 10 samples from the site with 3 soil samples from site $2A/2B$ were submitted for laboratory analysis: A-1(2-4), A-2(0-2), A-2(2-4), A-3(2-4), A-3(6-8), A-3(10-12), A-4(6-8), A-4(12-14), A-5(2-4) and A-6(1-3).	Soil E418.1, SW6010, SW7471 SW8081, SW8260 SW8260 SW8270, SW9014, SW9045 SW9065,
Area B Former Raw Product and By-product AST Areas/Block 1400		Area B investigation was performed in conjunction with Area A investigation. 4 soil borings were installed on site 2A/2B: B-1, B-02(B- 02A), B-3, B-4	See sampling methodology for Area A.





Table 4 Summary of Investigative Actions and Sampling Site 2A/2B: HHMT Port Ivory Facility

AOC	Description of Issues	Description of Actions and Sampling	Sampling Methodology		
		11 soil samples from site 2A/2B were submitted for laboratory analysis: B-1(2-4), B-1(6-8), B-1(9-10), B-02(2- 4), B-02(6-8), B-02A(8-10), B-3(2-4), B-3(6-8), B-4(2-4), B-4(5-6), and B-4(6-7).			
	_	Note: Samples B-4(5-6) and B-4(6-7) were analyzed for Volatile Organic Compounds only.			
Area C Former Oleum AST and Acid Wastewater Area/Block 1400		2 soil borings were installed: PA-MW-1 and PA-MW-1D 3 samples were submitted for laboratory analysis: PA-MW- 1(2-4), PA-MW-1(3-4.5) and PA-MW-1(8-10). All samples submitted for analysis were from PAMW-1.	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045		
Area D Fuel Oil AST Area/Block 1309		 5 soil borings were installed: D-1, D-2, D-3, D-4 and D-4A. 8 samples were submitted for laboratory analysis: D-1(0-2), D-1(6-8), D-1(18-20), D-2(0-2), D-2(6-8), D-3(1-3), D-4(0-2) and D-4A(6-8). 	SW9065, SW9071 Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045		
Area E S&S Tank Field, Super Fat Trap/Block 1400		 1 soil boring was installed on site 2A/2B: E-1. 3 samples from site 2A/2B were submitted for laboratory analysis: E-1(0.2-2), E-1(4-6) and E-1(10-12). 	SW9065, SW9071 Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071		
Area F1 Spent Nickel Catalyst Drum Storage Area/Block 1400		 1 soil boring was installed: F1-3 2 samples were submitted for laboratory analysis: F1-3(1-3), F1-3(3-5). 	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071		
Area F2 Waste Oil Drum Storage Area/Block 1400		 soil boring was installed on site 2A/2B: F2-2. samples from site 2A/2B were submitted for laboratory analysis: F2-2(2-4) and F2-2(8-10). 	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071		
Area G Former Vegetable AST Area/Block 1338		10 soil borings were installed: G-2, G-3, G-4, G-5, G-5A, G-6, G-7(N), G-8, G-9 and G-10.	Soil E418.1, SW6010, SW7471		



Table 4Summary of Investigative Actions and SamplingSite 2A/2B: HHMT Port Ivory Facility

	Description of Issues	Description of Actions and Sampling	Sampling Methodology		
AOC					
		15 samples were submitted for laboratory analysis: G- 2(0-2), G-2(4-6), G-2(6-8), G-3(0-2), G-4(6-8), G-5(4-6), G-5A(8-10), G-6(4-6), G-6(6-8), G-7(N)(8-10), G- 7(N)(10-12), G-8(1-2), G-8(6-7), G-9(4-6) and G-10(2-	SW8081, SW8082, SW8260 SW8270, SW9014, SW9045		
		4).	SW9065, SW9071		
Area H and Area R (Area H/R)		3 soil borings were installed: H/R-1, H/R-2 and H/R-3.	E418.1, SW6010, SW7471		
Storage Area/Block 1400		o samples were submitted for laboratory analysis: H/R - 1(1-3), H/R -1(3-4.5), H/R -2(0-1.5), H/R -2(1.5-3.5), H/R - 3(0.3-1) and H/R -3(1-3).	SW8081, SW8082, SW8260		
			SW8270, SW9014, SW9045 SW9065 SW9071		
4		1 soil boring was installed: I-1.	Soil		
Area I Temporary Fly		2 samples were submitted for laboratory analysis: 1-1/0-	E418.1, SW6010, SW7471		
Area/Block 1309		2) and I-1(2-4).	SW8081, SW8082, SW8260		
			SW8270, SW9014, SW9045 SW9065 SW9071		
		· · · · · · · · · · · · · · · · · · ·	Soil		
Area K /Block 1338		2 soil borings were installed: K-1 and K-2.	E418.1, SW6010, SW7471		
		4 samples were submitted for laboratory analysis: K-1(2- 4), K-1(5-6), K-2(0-2) and K-2(2-4).	SW8081, SW8082, SW8260		
			SW8270, SW9014, SW9045 SW0065 SW0071		
			Soil		
Area L Filled Area (southeast		6 soil borings were installed: L-1, L-2, L3(FILL), L-4, L- 5 and L-6.	E418.1, SW6010, SW7471		
64)/Block 1338		13 samples were submitted for laboratory analysis: L-	SW8081, SW8082, SW8260		
		1(2-4), L-1(6-8), L-2(8-10), L-2(10-12), L3(F1LL)(2-4), L3(F1LL)(8-10), L3(F1LL)(12-14), L-4(0-2), L-4(6-8), L-5(2-4), L-5(2	SW8270, SW9014, SW9045		
		5(2-4), $L-5(6-10)$, $L-0(6-7.5)$ and $L-0(7.5-6)$.	5 8 9065, 5 8 9071		
Area M Area		7 soil borings were installed: M-01, M-2, M-3, M-4, M- 5, MW-04 and PAMW-4.	Soil E418.1, SW6010,		
East of Edible Oils Buildings			SW7471 SW8081, SW8082,		
1338		(2-4), M-01 (0-2,) M-2 (2-4), M-2 (4-6), M-3 (2-4), M-4 (1-2), M-4 (2-4), M-4 (6-8), M-5 (6-6.5), PAMW-4 (0-2)	SW8260 SW8270, SW9014, SW9045		
		and PA-MW-04 (4-6).	SW9065, SW9071		
Area N Super Fat Trap Area/Block 1338		Evaluation of this area has been included with evaluation of Area G.	See sampling methodology for Area G.		

37

Ŷ



*

Area P Former Product Unloading		3 soil borings were installed on site 2A/2B: P-1, P-2 and P-3.	Soil E418.1, SW6010, SW7471 SW8081, SW8082
Pit/Block 1400		6 soil samples from site 2A/2B were submitted for laboratory analysis: P-1(2-4), P-1(8-10), P-2(2-4), P-2(4- 6), P-3(2-4) and P-3(6-8).	SW8260 SW8270, SW9014, SW9045 SW9065 SW9071
Area Q1		1 soil boring was installed on site 2A/2B: Q1-1.	Soil E418.1, SW6010,
Existing Scale Pit/Block 1338		2 samples from site 2A/2B were submitted for laboratory analysis: Q1-1(2-4) and Q1-1(4-6).	SW7471 SW8081, SW8082, SW8260
			SW8270, SW9014, SW9045 SW9065, SW9071
Area R Northwest Corner of Soap Manufacturing		Evaluation of this area has been included with evaluation of Area H	See sampling methodology for Area H.
Area (suspected calcium carbonate fill area)/Block 1400			
Wood Yard		5 soil borings were installed: WOOD-1B(not sampled), WOOD-01C, WOOD-03, WOOD-3 and WOOD-05.	Soil E418.1, SW6010, SW7471
		11 samples were submitted for laboratory analysis: WOOD-01C(10-12), WOOD-3(0.5-2), WOOD-3(2-4), WOOD-3(2-4), WOOD-3(6-8), WOOD-5(0-2), WOOD- 5(2-4), WOOD-5(4-6), WOOD-5(6-8), WOOD-5(8-10) and WOOD-5(14-16).	SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071
Closed C&D Landfill	P&G operated a construction and demolition (C&D) waste landfill at Block 1309. The landfill has been closed in accordance with applicable regulations. Post-closure requirements include both groundwater monitoring and landfill cap maintenance.	No actions were undertaken as part of the site investigation.	Not Applicable
Railroad Tracks and Sidings	Visual inspection of the site identified the presence of railroad tracks, sidings and equipment throughout the subject site. Investigative efforts were undertaken to document environmental quality.	Soil borings were installed and sampled. 27 soil borings at the site with 13 soil borings from site 2A/2B were installed: RR-01, RR-02, RR-03, RR-04, RR-05, RR-06, RR-07 , RR-8, RR-10, RR-15, PAMW-5, PAMW-6, A-4 , A-5, B-4 , G-8, H/R-3, L-1, PAMW-4, MW-04, M-3, P-1 , P-3 , P-2 , Q1-1 , WOOD-1B(not sampled) and WOOD-01C.	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071



		46 samples from the site with 24 samples from site 2A/2B were submitted for laboratory analysis: RR-01(0- 1.2) , RR-01(1.2-2) , RR-02(0-2) , RR-03(1.5-2) , RR- 04(0-2) , RR-04(2-4) , RR-05(0-2) , RR-05(2-4) , RR- 06(0-2) , RR-06(2-4) , RR-07(0-2) , RR-07(2-4) , RR- 8(2- 4), RR-8(6-8) , RR-10(2-4) , RR-07(0-2) , RR-07(2-4) , RR-8(2- 4), RR-8(6-8) , RR-10(2-4) , RR-10(8-10) , RR-15(4-6) , RR-15(0-2) , PAMW-5(0-2) , PAMW-6(0-2) , PAMW- 6(2-4) , PAMW-6(4-6) , PAMW-6(6-8) , PAMW-6(8-10) , A-4(6-8) , A-4(12-14) , A-5(2-4) , B-4(2-4) , B-4(5-6) , B- 4(6-7) , G-8(1-2) , G-8(6-7) , H/R-3(0-2) , L-1(2-4) , L-1(6- 8), M-4(2-4) , PA-MW-04(6-8) , M-3(2-4) , P-1(2-4) , P- 1(8-10) , P-3(2-4) , P-3(6-8) , P-2(2-4) , Q1-1(2-4) , Q1- 1(4-6) and WOOD-01C(10-12).	
Staining	Staining was noted on the soil flooring in two bays of Building #20 as well as south of Building 60B.	Soil borings were installed and sampled. 6 soil borings at the site with 5 soil borings from site 2A/2B were installed: STAIN-1, STAIN-02, STAIN-03, STAIN-3B, RR-06 and RR-07.	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071
		2A/2B were submitted for laboratory analysis: STAIN- 1(0-2), STAIN-1(4-6), STAIN-02(1-2), STAIN-02(2-3), STAIN-03(1-1.5), STAIN-03(1.5-2.5), STAIN-3B(0-2), STAIN-3B(2-4), RR-06(0-2) RR-06(2-4), RR-07(0-2) and RR-07(2-4).	
Pits and Drains	Pits and drains, some sealed with gravel, were noted at both interior and exterior site locations. In addition, reports identify the presence of oil/water separator systems.	A visual inspection was performed, as feasible, to assess conditions at pits and drains. Soil borings were installed and sampled at and adjacent to current and former pits and drains. 21 soil borings at the site with 11 soil borings from site 2A/2B were installed: PD-1 , PD-3 , PD-4 , PD-4A (No sample collected), PD-5 , PD-6, PD-8, PD-9, PD-10, PD- 11, PD-13(not sampled), PD-14, A-4, A-5, P-1, P-3, P-2, RR-03 , RR-15, PAMW-5 and STAIN-02 .	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071
		37 samples from the site with 22 samples from site 2A/2B were submitted for laboratory analysis: PD-1(2- 4) , PD-1(10-12) , PD-3(4-6) , PD-4(8-10) , PD-5(0-2) , PD-5(2-4) , PD-6(6-8) , PD-6(12-14) , PD-8(2-4) , PD-8(8- 10), PD-8(16-17) , PD-9(4-6) , PD-9(8-10) , PD-10(2-4) , PD-10(6-8) , PD-11(4-6) , PD-14(2-4) , PD-14(6-8) , A- 4(6-8) , A-4(12-14) , A-5(2-4) , P-1(2-4) , PD-14(6-8) , A- 4(6-8) , A-4(12-14) , A-5(2-4) , P-1(2-4) , PD-14(6-8) , A- 4(6-8) , A-4(12-14) , A-5(2-4) , P-1(2-4) , P-1(8-10) , P- 2(2-4) , P-2(4-6) , P-3(2-4) , P-3(6-8) , RR-03(0-2) , RR- 15(0-2), RR-15(4-6) , STAIN-02(1-2) , STAIN-02(2-3) , PAMW-7(2-4) , PAMW-7(4-6) , PAMW-7(6-8) , PAMW-7(8-10) , and PAMW-5 (0-2).	
Former Structures	Review of Sanborn Maps and aerial photographs reveal the presence of former structures, ASTs, railroad tracks and sidings,	Soil borings were installed and sampled at areas formerly occupied by structures, debris piles and discolored areas. 26 soil borings at the site with 10 soil borings from site 2A/2B were installed: FS-1 (No sample collected), FS-	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014,

79



\$

Hatch Mott MacDonald

	at various locations throughout the subject site. Review of some of the historical sources also revealed the presence of discolored areas and/or debris piles.	1A (No sample collected), FS-1B, FS-2, FS-3, FS-4, FS-6, FS-7, FS-8, PAMW-4, PAMW-7 , PAMW-8, A-3, M-3, WOOD-1B(not sampled), WOOD-01C, WOOD-3, WOOD-5, RR-01, RR-04, RR-05, PD-1, PD-3, PD-8, PD-9 and PD-11.	SW9045 SW9065, SW9071
		submitted for laboratory analysis: FS-1B(0-2), FS-1B(6-7), FS-1B(12-13.5), FS-2(2-4), FS-2(8-10), FS-2(17-18), FS-3(2-4), FS-3(6-8), FS-4(0-2), FS-4(2-4), FS-6(0-2), FS-6(4-6), FS-7(2-4), FS-7(8-10), FS-8(0-2), PAMW-4(0-2), PAMW-4(4-6), PAMW-7(2-4), PAMW-7(4-6), PAMW-7(6-8), PAMW-8(0-2), PAMW-8(4-6), A-3(2-4), A-3(6-8), A-3(10-12), M-3(2-4), WOOD-01C(10-12), WOOD-3(0-2), WOOD-3(2-4), WOOD-5(0-2), WOOD-5(2-4), WOOD-5(4-6), WOOD-5(6-8), WOOD-5(8-10), WOOD-5(8-10), WOOD-5(8-10), WOOD-5(14-16), WOOD-3(2-4), WOOD-3(6-8), RR-01(0-1.2), RR-01(1.2-2), RR-04(0-2), RR-04(2-4), RR-05(0-2), PD-1(2-4), PD-8(16-17), PD-9(4-6), PD-9(8-10) and PD-11(4-6).	
Fill Material	P&G placed a variety of fill material at the subject site. The fill materials present at the site include soil/sand, construction debris (wood, bricks, glass, concrete), ash from boiler operations, slag, vegetative debris and by-products from production activities (calcium carbonate, spent diatomaceous filter earth, and spent carbonaceous filter material). The presence of black staining of site soil was noted in P&G reports.	 Soil borings were installed throughout the site to characterize the type and extent of fill material. Representative samples were submitted for laboratory analysis to determine the presence/absence of contaminants in fill materials. 23 soil borings at the site with 7 soil borings from site 2A/2B were installed: Fill-1 (No sample collected), Fill-2, Fill-3, Fill-4, Fill-5, Fill-7, Fill-8, Fill-10, Fill-11, Fill-12, Fill-3, Fill-4, Fill-5, Fill-7, Fill-8, Fill-10, Fill-11, Fill-12, Fill-25, PAMW-10D(Fill-9), PAMW-11D(Fill18), PAMW-12(Fill-19), PAMW-13(Fill-23), and PAMW-14D(Fill-24). 60 samples at the site with 20 from site 2A/2B were submitted for laboratory analysis: Fill-2(0.7-3.), Fill-3(0-2), Fill-3(2-4), Fill-5(2-4), Fill-4(4-6), Fill-4(2-4), Fill-4(4-6), Fill-4(4-6), Fill-10(5-8), Fill-5(2-4), Fill-5(6-8), Fill-7(1.5-2.5), Fill-7(2.5-4), Fill-7(10-12), Fill-8(0-2), Fill-8(6-8), Fill-10(3-4), Fill-10(6-8), Fill-11(0-2), Fill-11(2-4), Fill-12(0-2), Fill-13(1-3), Fill-13(3-5), Fill-16(2-4), Fill-14(6-8), Fill-15(4-6), Fill-15(12-13), Fill-16(2-4), Fill-20(4-6), Fill-12(0(5-8), Fill-120(0-2), Fill-20(0-2), Fill-20(10-12), Fill-20(12-14), Fill-20(6-8), Fill-20(0-2-2), Fill-20(10-12), Fill-20(12-14), Fill-20(6-8), Fill-20(0-2), FAMW-7(6-8), PAMW-7(8-10), FAMW-7(6-6), PAMW-7(2-4), PAMW-7(2-4), PAMW-7(4-6), PAMW-7(4-6), PAMW-10D(7-8), PAMW-10D(6-2), PAMW-10D(4-6), PAMW-12(0-2), PAMW-12(0-2), PAMW-13(0-2), PAMW-13(0-2), PAMW-14D(0-2) and PAMW-14D(4-6). 	Soil E418.1, SW6010, SW7471 SW8081, SW8082, SW8260 SW8270, SW9014, SW9045 SW9065, SW9071



敻

Hatch Mott MacDonald

Table 4Summary of Investigative Actions and SamplingSite 2A/2B: HHMT Port Ivory Facility

Groundwater	P&G reports identified the presence of contaminants, elevated pH and free phase product in site monitoring wells.	 Samples were obtained from a representative number of existing wells and additional wells were installed and sampled to evaluate current groundwater quality. All wells were examined for the presence of free product and samples of identified free product were submitted for fingerprinting. 17 wells at the site with 5 from site 2A/2B were installed and sampled: PAMW-14D, PAMW-15, PAMW-15D (two rounds of samples submitted for laboratory analysis), PAMW-1, PAMW-1D, PAMW-4, PAMW-04D, PAMW-5, PAMW-6, PAMW-6D, PAMW-7, PAMW-7D, PAMW-8, PAMW-10D, PAMW-11D, PAMW-12 and PAMW-13. 	Groundwater E624, E625, E200.7 E245.2, E150.1, E418.1 E1664, E335.2, E420.1 E608 Free Product/Fingerprint GCFID
		The following 14 existing wells at the site with 4 from site 2A/2B were included in the sampling effort: BW-13, CS-7, EW-3, EW-6, GW-10, GW-3, GW-5, GW-7, GW- 9, MW-3, MW-04 (duplicate samples submitted), PZ-1, RS-1, and RS-2. Finger printing was performed on 4 wells at the site with 2 from site 2A/2B: GW-14, OP-1, GW-16 and EW-18.	:

(1): This table identifies samples collected to identify individual AOCs. Given that samples were utilized to address multiple AOCs, samples may be listed under more than one AOC. Thus, this table should **not** be utilized to calculate the total number of samples collected through the SI.

(2). Soil borings, wells and sample designations for Site 2A/2B are presented in bold type.

(3): The prefix "PG" has not been included for well designations.



5.1 Pre-Investigation Field Activities

Prior to the initiating sampling efforts, HMM performed a series of pre-investigative field tasks consisting of the following:

- Site walk
- Review of available Sanborn maps and information from the Phase I Environmental Site Assessment
- Field screening, cataloging and inspection of the existing monitoring wells on site. (Depth to water, total depth of well, presence of free phase product, physical condition of well and protective casing, etc.)
- Mark out of all soil boring and groundwater monitoring well locations in accordance with predetermined AOCs.
- Met with site operations personnel as well as former P&G employees to discuss boring and monitoring well locations and possible underground utilities.
- Contact and met with representatives of the pipeline companies concerning the presence of various pipelines that transect the site.
- Contact and met with representatives of the local utility companies and authorities regarding the location of public utilities.
- Supervised the geophysical team from Hager-Richter Geoscience, Inc (Hager-Richter) who field screened all proposed soil boring and monitoring well locations for internal underground utilities as well as possible UST locations using geophysical techniques.

5.2 GPR/EM Survey - Potential UST Areas

The June 2000 Phase I ESA identified the potential presence of one or more USTs at four locations at Site 2A (UST1, UST3, UST4 and UST7); no potential USTs were identified at Site 2B. This conclusion was based upon a review of Sanborn Fire Insurance Maps, information in portions of reports provided by P&G and limited information provided by representatives of P&G. Based on the information obtained through the performance of the Phase I ESA, a geophysical and electromagnetic survey was performed at the following site areas:

- UST1 Area: North of Building 20;
- UST3 Area: North of the east of Building 13;
- UST4 Area: West of Buildings 34 and 38; and
- UST7 Area: West of Building 43A and South of Building S-35.



HMM retained Hager-Richter to perform a survey to evaluate the presence of USTs at the above listed locations. Hager-Richter utilized ground penetrating radar (GPR) and electromagnetic (EM) methods to assess the potential UST locations. The findings of the GPR/EM survey are presented in Section 6.2 and a copy of the Hager-Richter Geophysical Report is provided in Appendix A of this report.

5.3 Soil Boring Construction and Sampling

In November and December 2000, soil borings were installed to evaluate subsurface soil conditions and to determine the extent of fill material present on the subject site in accordance with ASTM D 1586-84 sampling protocol. Samples were field screened and visually reviewed to establish site lithology. Representative samples were submitted for chemical analysis to evaluate AOCs.

The first six feet of the boreholes were advanced using stainless steel hand augers. Any samples obtained from this interval and slated for chemical analysis were collected via a decontaminated hand auger. Upon reaching six feet bgs, the boreholes were advanced using a truck mounted drill rig with six-inch hollow stem augers (HSA). The boreholes were advanced continuously using HSAs with three-inch diameter split spoons from which the samples were obtained. Split spoons were taken from approximately six feet bgs to 16 feet bgs or until native material was encountered.

Soil samples were collected from the borings in the following manner. Samples collected for VOC analysis were immediately removed from the two-foot interval of the split spoon and placed in laboratory containers. Samples obtained for analysis other than VOC were homogenized in a stainless steel mixing bowl and transferred utilizing a stainless steel trowel to the appropriate laboratory containers. Upon completion of the soil boring, the abandoned borehole was pressure-grouted with cement-bentonite grout to ground surface.

The specifics of the SI for soil are presented by AOC in the following sections. A summary of the soil borings installed and samples collected as part of the SI of Site 2A/2B are presented in Table 4. Soil boring locations are presented on Figure 7 Analytical results for SI soil sampling are discussed in Section 6.3 and analytical summary tables for soil (Table 5A-5E) are provided subsequent to first reference, organized by specific classes of contaminants. Soil boring logs, including field screening information such as PID readings and visual observations, associated with the SI are provided in Appendix B. As previously stated, soil borings and samples were utilized to evaluate multiple AOCs at



Location	Recommended	PG-A-4	PG-A-4	PG-A-5	PG-B-1	PG-B-1	PG-B-1	PG-B-02	PG-B-02
Sample Date	Soil	11/16/2000	11/16/2000	11/14/2000	12/4/2000	12/4/2000	12/4/2000	11/16/2000	11/16/2000
Sample ID	Cleanup	PG-A-04	PG-A-04	PG-A-05	PG-B-01	PG-B-01	PG-B-01	PG-B-02	PG-A-02
Sample Depth	Objective	6-8'	12-14'	2-4'	2-4'	6-8'	9-10'	2-4'	6-8'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.006 8 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
1,1,2-TRICHLOROETHANE	6	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
1,1-DICHLOROETHANE	0.2	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0 059 U	0.0069 U
1,1-DICHLOROETHYLENE	0.4	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
1,2-DICHLOROETHANE	0.1	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
1,2-DICHLOROPROPANE	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
2-CHLOROETHYL VINYL ETHER	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
ACROLEIN	NS	0.021 U	0.16 U	0.019 U	0.017 U	0.018 U	0.019 U	0.018 U	0.021 U
ACRYLONITRILE	NS	0.0095 U	0.074 U	0.0086 U	0.0079 U	0.0083 U	0.0088 U	0.0082 U	0.0096 U
BENZENE	0.06	0.0014 U	0.011 U	0.0012 U	0.0011 U	0.0012 U	0.0013 U	0.0012 U	0.0014 U
BROMODICHLOROMETHANE	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
BROMOFORM	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
BROMOMETHANE	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
CARBON TETRACHLORIDE	0.6	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
CHLOROBENZENE	1.7	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
CHLOROETHANE	1.9	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
CHLOROFORM	0.3	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
CHLOROMETHANE	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
CIS-1,3-DICHLOROPROPENE	NS	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
DIBROMOCHLOROMETHANE	N/A	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
DICHLOROMETHANE	0.1	0.0029 JB	0.051 JB	0.0027 JB	0.0022 JB	0.0020 JB	0.0033 JB	0.0047 JB	0.013
ETHYLBENZENE	5.5	0.0014 U	0.011 U	0.0012 U	0.0011 U	0.0012 U	0.0013 U	0.0012 U	0.0075
M&P-XYLENES	1.2*	0.0027 U	0.021 U	0.0025 U	0.0023 U	0.0024 U	0.0025 U	0.0024 U	0.021
METHYLBENZENE	1.5	0.0014 U	0.011 U	0.0012 U	0.0011 U	0.0012 U	0.0013 U	0.0012 U	0.0058
O-XYLENE	1.2*	0.0014 U	0.011 U	0.0012 U	0.0011 U	0.0012 U	0.0013 U	0.0012 U	0.0072
TETRACHLOROETHYLENE	1.4	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
TRICHLOROETHYLENE	0.7	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
VINYL CHLORIDE	0.2	0.0068 U	0.053 U	0.0062 U	0.0057 U	0.0060 U	0.0063 U	0.0059 U	0.0069 U
TOTAL VOCs	10	0.0029	0.051	0.0027	0.0022	0.002	0.0033	0.0047	0.0545

U Undetectable Levels

~******

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard

and the second secon

Location	Recommended	PG-B-02A	PG-B-3	PG-B-3	PG-B-4	PG-B-4	PG-B-4	PG-E-1	PG-E-1
Sample Date	Soil	11/16/2000	12/4/2000	12/4/2000	12/4/2000	12/4/2000	12/4/2000	11/13/2000	11/13/2000
Sample ID	Cleanup	PG-B-02A	PG-B-03	PG-B-03	PG-B-04	PG-B-04	PG-B-04	PG-E-01	PG-E-01
Sample Depth	Objective	8-10'	2-4'	6-8'	2-4'	5-6'	6-7'	0.2-2'	4-6'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
1,1,2-TRICHLOROETHANE	6	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
1,1-DICHLOROETHANE	0.2	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0 057 U	0.0058 U
1,1-DICHLOROETHYLENE	0.4	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
1,2-DICHLOROETHANE	0.1	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
1,2-DICHLOROPROPANE	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
2-CHLOROETHYL VINYL ETHER	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
ACROLEIN	NS	2.5 U	0.021 U	0.025 U	0.018 U	0.075 U	0.018 U	0.017 U	0.017 U
ACRYLONITRILE	NS	0.30 U	0.0096 U	0.011 U	0.0082 U	0.035 U	0.0082 U	0.0079 U	0.0081 U
BENZENE	0.06	0.16 U	0.0014 U	0.0016 U	0.0012 U	0.005 U	0.0012 U	0.0011 U	0.0012 U
BROMODICHLOROMETHANE	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
BROMOFORM	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
BROMOMETHANE	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
CARBON TETRACHLORIDE	0.6	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
CHLOROBENZENE	1.7	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
CHLOROETHANE	1.9	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
CHLOROFORM	0.3	0.82 U	0.0069 U	0.0082 U	0.0016 J	0.025 U	0.0060 U	0.0057 U	0.0058 U
CHLOROMETHANE	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
CIS-1,3-DICHLOROPROPENE	NS	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
DIBROMOCHLOROMETHANE	N/A	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025·U	0.0060 U	0.0057 U	0.0058 U
DICHLOROMETHANE	0.1	0.82 U	0.0042 JB	0.0031 JB	0.0044 JB	0.025 U	0.0035 JB	0.0057 U	0.0034 JB
ETHYLBENZENE	5.5	0.16 U	0.0014	0.0016 U	0.0012 U	0.005 U	0.0012 U	0.0011 U	0.0012 U
M&P-XYLENES	1.2*	0.33 U	0.0030	0.0033 U	0.0055	0.01 U	0.0024 U	0.0023 U	0.0023 U
METHYLBENZENE	1.5	0.16 U	0.0014 U	0.0016 U	0.0012 U	0.005 U	0.0012 U	0.0011 U	0.0012 U
O-XYLENE	1.2*	0.16 U	0.0014 U	0.0016 U	0.0014	0.005 U	0.0012 U	0.0011 U	0.0012 U
TETRACHLOROETHYLENE	1.4	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.00 5 7 U	0.0058 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.43	0.0060 U	0.0057 U	0.0058 U
TRICHLOROETHYLENE	0.7	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
VINYL CHLORIDE	0.2	0.82 U	0.0069 U	0.0082 U	0.0059 U	0.025 U	0.0060 U	0.0057 U	0.0058 U
TOTAL VOCs	10	ND	0.0086	0.0031	0.0129	0.43	0.0035	ND	0.0034

X

and the second secon

8285⁻¹000-100-0002820

U Undetectable Levels

-9,99

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard

÷ .

Location	Recommended	PG-E-1	PG-F2-2	PG-F2-2	PG-P-1	PG-P-1	PG-P-2	PG-P-2	PG-P-3
Sample Date	Soil	11/13/2000	11/20/2000	11/20/2000	11/22/2000	11/22/2000	11/30/2000	11/30/2000	11/22/2000
Sample ID	Cleanup	PG-E-01	PG-F2-2	PG-F2-2	PG-P-01	PG-P-01	PG-P-02	PG-P-02	PG-P-03
Sample Depth	Objective	10-12'	2-4'	8-10'	2-4'	8-10'	2-4'	4-6'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
1,1,2-TRICHLOROETHANE	6	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.00 57 U
1,1-DICHLOROETHANE	0.2	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
1,1-DICHLOROETHYLENE	0.4	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
1,2-DICHLOROETHANE	0.1	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
1,2-DICHLOROPROPANE	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
2-CHLOROETHYL VINYL ETHER	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
ACROLEIN	NS	0.028 U	0.016 U	0.017 U	0.016 U	0.019 U	0.017 U	0.018 U	0.017 U
ACRYLONITRILE	NS	0.013 U	0.0075 U	0.0081 U	0.0075 U	0.0088 U	0.0081 U	0.0082 U	0.0080 U
BENZENE	0.06	0.0019 U	0.0011 U	0.0012 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U
BROMODICHLOROMETHANE	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
BROMOFORM	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
BROMOMETHANE	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
CARBON TETRACHLORIDE	0.6	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
CHLOROBENZENE	1.7	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.00 5 7 U
CHLOROETHANE	1.9	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
CHLOROFORM	0.3	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
CHLOROMETHANE	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
CIS-1,3-DICHLOROPROPENE	NS	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
DIBROMOCHLOROMETHANE	N/A	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
DICHLOROMETHANE	0.1	0.0094 U	0.0048 J	0.0054 J	0.0035 JB	0.0030 JB	0.0087 B	0.0061 B	0.0043 JB
ETHYLBENZENE	5.5	0.0019 U	0.0011 U	0.0012 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U
M&P-XYLENES	1.2*	0.0038 U	0.0022 U	0.0023 U	0.0022 U	0.0025 U	0.0023 U	0.0024 U	0.0023 U
METHYLBENZENE	1.5	0.0019 U	0.0011 U	0.0012 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U
O-XYLENE	1.2*	0.0019 U	0.0011 U	0.0012 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U
TETRACHLOROETHYLENE	1.4	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
TRICHLOROETHYLENE	0.7	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
VINYL CHLORIDE	0.2	0.0094 U	0.0054 U	0.0058 U	0.0054 U	0.0063 U	0.0058 U	0.0060 U	0.0057 U
TOTAL VOCs	10	ND	0.0048	0.0054	0.0035	0.0030	0.0087	0.0061	0.0043

* V#21003502880. 277

U Undetectable Levels

1019

ND Not Detectived

NS No Standard

COLLASS SALES

1982 5 1983

* Total Xylene Recommended Cleanup Standard

00°2X*2



Location	Recommended	PG-P-3	PG-PD-1	PG-PD-1	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5	PG-Q1-1
Sample Date	Soil	11/22/2000	11/21/2000	11/21/2000	11/29/2000	12/2/2000	12/2/2000	12/2/2000	11/30/2000
Sample ID	Cleanup	PG-P-03	PG-PD-01	PG-PD-01	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5	PG-Q1-1
Sample Depth	Objective	4-6'	2-4'	10-12'	4-6'	8-10'	0-2'	2-4'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
	1								
1,1,1-TRICHLOROETHANE	0.8	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
1,1,2-TRICHLOROETHANE	6	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
1,1-DICHLOROETHANE	0.2	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
1,1-DICHLOROETHYLENE	0.4	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
1,2-DICHLOROETHANE	0.1	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
1,2-DICHLOROPROPANE	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
2-CHLOROETHYL VINYL ETHER	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
ACROLEIN	NS	0.017 U	0.017 U	0.019 U	0.019 U	0.018 U	0.017 U	0.017 U	0.017 U
ACRYLONITRILE	NS	0.0080 U	0.0078 U	0.0088 U	0.0086 U	0.0085 U	0.0079 U	0.0078 U	0.0079 U
BENZENE	0.06	0.0011 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U	0.0011 U	0.0011 U
BROMODICHLOROMETHANE	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
BROMOFORM	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
BROMOMETHANE	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
CARBON TETRACHLORIDE	0.6	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
CHLOROBENZENE	1.7	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
CHLOROETHANE	1.9	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
CHLOROFORM	0.3	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
CHLOROMETHANE	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
CIS-1,3-DICHLOROPROPENE	NS	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
DIBROMOCHLOROMETHANE	N/A	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
DICHLOROMETHANE	0.1	0.0042 JB	0.0073	0.0026 JB	0.0045 JB	0.0063 B	0.0060 B	0.0050 JB	0.0064 B
ETHYLBENZENE	5.5	0.0011 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U	0.0011 U	0.0011 U
M&P-XYLENES	1.2*	0.0023 U	0.0022 U	0.0025 U	0.0025 U	0.0024 U	0.0023 U	0.0022 U	0.0023 U
METHYLBENZENE	1.5	0.0011 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U	0.0011 U	0.0011 U
O-XYLENE	1.2*	0.0011 U	0.0011 U	0.0013 U	0.0012 U	0.0012 U	0.0011 U	0.0011 U	0.0011 U
TETRACHLOROETHYLENE	1.4	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
TRICHLOROETHYLENE	0.7	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
VINYL CHLORIDE	0.2	0.0057 U	0.0056 U	0.0063 U	0.0062 U	0.0061 U	0.0057 U	0.0056 U	0.0057 U
TOTAL VOCs	10	0.0042	0.0073	0.0026	0.0045	0.0063	0.0060	0.0050	0.0064

U Undetectable Levels

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard



Location	Recommended	PG-Q1-1	PG-RR-1	PG-RR-1	PG-RR-2	PG-RR-3	PG-RR-4	PG-RR-4	PG-RR-5
Sample Date	Soil	11/30/2000	11/3/2000	11/3/2000	11/4/2000	11/3/2000	11/4/2000	11/4/2000	11/7/2000
Sample ID	Cleanup	PG-Q1-1	PG-RR-01	PG-RR-01	PG-RR-02	PG-RR-03	PG-RR-04	PG-RR-04	PG-RR-05
Sample Depth	Objective	4-6'	0-1.2'	1.2-2'	0-2'	1.5-2'	0-2'	2-4'	0-2'
Concentration	МG/КС	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
								-	
1,1,1-TRICHLOROETHANE	0.8	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
1,1,2-TRICHLOROETHANE	6	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
I,I-DICHLOROETHANE	0.2	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
1,1-DICHLOROETHYLENE	0.4	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
1,2-DICHLOROETHANE	0.1	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
1,2-DICHLOROPROPANE	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
2-CHLOROETHYL VINYL ETHER	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
ACROLEIN	NS	0.023 U	0.017 U	0.017 U	0.018 U	0.017 U	0.017 U	2.3 U	0.016 U
ACRYLONITRILE	NS	0.011 U	0.0081 U	0.0080 U	0.0082 U	0.0081 U	0.0078 U	1.1 U	0.0075 U
BENZENE	0.06	0.0015 U	0.0012 U	0.0011 U	0.0012 U	0.0012 U	0.0011 U	0.16 U	0.0011 U
BROMODICHLOROMETHANE	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
BROMOFORM	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
BROMOMETHANE	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
CARBON TETRACHLORIDE	0.6	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
CHLOROBENZENE	1.7	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
CHLOROETHANE	1.9	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
CHLOROFORM	0.3	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
CHLOROMETHANE	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
CIS-1,3-DICHLOROPROPENE	NS	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
DIBROMOCHLOROMETHANE	N/A	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
DICHLOROMETHANE	0.1	0.0096 B	0.011 B	0.0087 B	0.0051 JB	0.0082 B	0.017 B	0.78 U	0.0050 J
ETHYLBENZENE	5.5	0.0015 U	0.0012 U	0.0011 U	0.0012 U	0.0012 U	0.0011 U	0.16 U	0.0011 U
M&P-XYLENES	1.2*	0.0031 U	0.0023 U	0.0023 U	0.0024 U	0.0023 U	0.0022 U	0.31 U	0.0 022 U
METHYLBENZENE	1.5	0.0015 U	0.0012 U	0.0011 U	0.0012 U	0.0012 U	0.0011 U	0.16 U	0.0011 U
O-XYLENE	1.2*	0.0015 U	0.0012 U	0.0011 U	0.0012 U	0.0012 U	0.0011 U	0.16 U	0.0011 U
TETRACHLOROETHYLENE	1.4	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
TRICHLOROETHYLENE	0.7	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
VINYL CHLORIDE	0.2	0.0077 U	0.0058 U	0.0057 U	0.0059 U	0.0058 U	0.0056 U	0.78 U	0.0054 U
TOTAL VOCs	10	0.0096	0.0110	0.0087	0.0051	0.0082	0.017	ND	0.005

U Undetectable Levels

ND Not Detectived

NS No Standard

Total Xylene Recommended Cleanup Standard

and the second second

Location	Recommended	PG-RR-5	PG-RR-6	PG-RR-6	PG-RR-7	PG-RR-7	PG-STAIN-02	PG-STAIN-02	PG-STAIN-03
Sample Date	Soil	11/7/2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000	11/11/2000	11/11/2000	11/10/2000
Sample ID	Cleanup	PG-RR-05	PG-RR-06	PG-RR-06	PG-RR-07	PG-RR-07	PG-ST-02	PG-ST-02	PG-ST-03
Sample Depth	Objective	2-4'	0-2'	2-4'	0-2'	2-4'	1-2'	2-3'	1-1.5'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.0054 U	0.0059 U	0.0060 U	0.73 U				
1,1,2,2-TETRACHLOROETHANE	0.6	0.0054 U	0.0059 U	0.0060 U	0.73 U				
1,1,2-TRICHLOROETHANE	6	0.0054 U	0.0059 U	0.0060 U	0.73 U				
1,1-DICHLOROETHANE	0.2	0.0054 U	0.0059 U	0.0060 U	0.73 U				
1,1-DICHLOROETHYLENE	0.4	0.0054 U	0.0059 U	0.0060 U	0.73 U				
1,2-DICHLOROETHANE	0.1	0.0054 U	0.0059 U	0.0060 U	0.73 U				
1,2-DICHLOROPROPANE	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
2-CHLOROETHYL VINYL ETHER	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
ACROLEIN	NS	0.016 U	0.018 U	0.018 U	2.2 U				
ACRYLONITRILE	NS	0.0075 U	0.0082 U	0.0083 U	1.0 U				
BENZENE	0.06	0.0011 U	0.0012 U	0.0012 U	0.15 U				
BROMODICHLOROMETHANE	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
BROMOFORM	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
BROMOMETHANE	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
CARBON TETRACHLORIDE	0.6	0.0054 U	0.0059 U	0.0060 U	0.73 U				
CHLOROBENZENE	1.7	0.0054 U	0.0059 U	0.0060 U	0.73 U				
CHLOROETHANE	1.9	0.0054 U	0.0059 U	0.0060 U	0.73 U				
CHLOROFORM	0.3	0.0054 U	0.0059 U	0.0060 U	0.73 U				
CHLOROMETHANE	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
CIS-1,3-DICHLOROPROPENE	NS	0.0054 U	0.0059 U	0.0060 U	0.73 U				
DIBROMOCHLOROMETHANE	N/A	0.0054 U	0.0059 U	0.0060 U	0.73 U				
DICHLOROMETHANE	0.1	0.0044 J	0.0057 B	0.0071 B	0.0054 JB	0.0061 B	0.0034 JB	0.0026 JB	0.17 J
ETHYLBENZENE	5.5	0.0011 U	0.0012 U	0.0012 U	0.15 U				
M&P-XYLENES	1.2*	0.0022 U	0.0024 U	0.0024 U	0.37				
METHYLBENZENE	1.5	0.0011 U	0.0016	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0012 U	0.15 U
O-XYLENE	1.2*	0.0011 U	0.0012 U	0.0012 U	0.24				
TETRACHLOROETHYLENE	1.4	0.0054 U	0.0059 U	0.0060 U	0.73 U				
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0054 U	0.0059 U	0.0060 U	0.73 U				
TRANS-1,3-DICHLOROPROPENE	0.3	0.0054 U	0.0059 U	0.0060 U	0.73 U				
TRICHLOROETHYLENE	0.7	0.0054 U	0.0059 U	0.0060 U	0.73 U				
VINYL CHLORIDE	0.2	0.0054 U	0.0059 U	0.0060 U	0.73 U				
TOTAL VOCs	10	0.0044	0.0073	0.0071	0.0054	0.0061	0.0034	0.0026	0.78

U Undetectable Levels

19

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard

ry taiter in marchine in the

Location	Recommended	PG-STAIN-03	PG-STAIN-3B	PG-STAIN-3B	PG-FS-2	PG-FS-2	PG-FS-2	PG-FS-3	PG-FS-3
Sample Date	Soil	11/10/2000	11/11/2000	11/11/2000	11/17/2000	11/17/2000	11/17/2000	11/15/2000	11/15/2000
Sample ID	Cleanup	PG-ST-03	PG-ST-3B	PG-ST-3B	PG-FS-02	PG-FS-02	PG-FS-02	PG-FS03	PG-FS03
Sample Depth	Objective	1.5-2.5	0-2'	2-4'	2-4'	8-10'	17-18'	2-4'	6-8'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
1,1,2-TRICHLOROETHANE	6	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
I,1-DICHLOROETHANE	0.2	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
1,1-DICHLOROETHYLENE	0.4	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
1,2-DICHLOROETHANE	0.1	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
I,2-DICHLOROPROPANE	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
2-CHLOROETHYL VINYL ETHER	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
ACROLEIN	NS	2.4 U	0.016 U	0.016 U	0.017 U	0.018 U	0.017 U	0.017 U	0.018 U
ACRYLONITRILE	NS	1.I U	0.0075 U	0.0075 U	0.0079 U	0.0085 U	0.0078 U	0.0080 U	0.0082 U
BENZENE	0.06	0.16 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U	0.0012 U
BROMODICHLOROMETHANE	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
BROMOFORM	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
BROMOMETHANE	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
CARBON TETRACHLORIDE	0.6	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
CHLOROBENZENE	1.7	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
CHLOROETHANE	1.9	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
CHLOROFORM	0.3	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
CHLOROMETHANE	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
CIS-1,3-DICHLOROPROPENE	NS	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
DIBROMOCHLOROMETHANE	N/A	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
DICHLOROMETHANE	0.1	0.79 U	0.0054 U	0.0026 JB	0.0024 JB	0.0049 JB	0.0029 JB	0.0021 JB	0.0059 U
ETHYLBENZENE	5.5	0.16 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U	0.0012 U
M&P-XYLENES	1.2*	0.19 J	0.0022 U	0.0022 U	0.0023 U	0.0024 U	0.0022 U	0.0023 U	0.0024 U
METHYLBENZENE	1.5	0.16 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U	0.0012 U
O-XYLENE	1.2*	0.16 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U	0.0012 U
TETRACHLOROETHYLENE	1.4	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
TRICHLOROETHYLENE	0.7	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
VINYL CHLORIDE	0.2	0.79 U	0.0054 U	0.0054 U	0.0057 U	0.0061 U	0.0056 U	0.0057 U	0.0059 U
TOTAL VOCs	10	0.19	ND	0.0026	0.0024	0.0049	0.0029	0.0021	ND

U Undetectable Levels

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard

1.110000793



Location	Recommended	PG-FILL-02	PG-FILL-03	PG-FILL-03	PG-FILL-03	PG-FILL-04	PG-FILL-04	PG-FILL-04
Sample Date	Soil	11/3/2000	11/4/2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PGFILL02	PGFILL03	PGFILL03	PGFILL03	PGFILL04	PGFILL04	PGFILL04
Sample Depth	Objective	0.7-3'	0-2'	2-4'	4-6'	0-2'	2-4'	4-6'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
				•				
1,1,1-TRICHLOROETHANE	0.8	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
1,1,2-TRICHLOROETHANE	6	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
1,1-DICHLOROETHANE	0.2	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
1,1-DICHLOROETHYLENE	0.4	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
I,2-DICHLOROETHANE	0.1	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
1,2-DICHLOROPROPANE	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
2-CHLOROETHYL VINYL ETHER	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
ACROLEIN	NS	0.019 U	0.016 U	0.017 U				
ACRYLONITRILE	NS	0.0090 U	0.0075 U	0.0077 U	0.0077 U	0.0081 U	0.0078 U	0.0080 U
BENZENE	0.06	0.0013 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U
BROMODICHLOROMETHANE	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
BROMOFORM	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
BROMOMETHANE	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
CARBON TETRACHLORIDE	0.6	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
CHLOROBENZENE	1.7	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
CHLOROETHANE	1.9	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
CHLOROFORM	0.3	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
CHLOROMETHANE	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
CIS-1,3-DICHLOROPROPENE	NS	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
DIBROMOCHLOROMETHANE	N/A	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
DICHLOROMETHANE	0.1	0.012 B	0.0054 U	0.0027 J	0.0033 J	0.0054 J	0.0091 B	0.0075 B
ETHYLBENZENE	5.5	0.0013 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U
M&P-XYLENES	1.2*	0.0026 U	0.0022 U	0.0022 U	0.0022 U	0.0023 U	0.0022 U	0.0023 U
METHYLBENZENE	1.5	0.0013 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U
O-XYLENE	1.2*	0.0013 U	0.0011 U	0.0011 U	0.0011 U	0.0012 U	0.0011 U	0.0011 U
TETRACHLOROETHYLENE	1.4	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
TRICHLOROETHYLENE	0.7	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
VINYL CHLORIDE	0.2	0.0065 U	0.0054 U	0.0056 U	0.0056 U	0.0058 U	0.0056 U	0.0057 U
TOTAL VOCs	10	0.0120	ND	0.0027	0.0033	0.0054	0.0091	0.0075

U Undetectable Levels

1982

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard



Location	Recommended	PG-FILL-04	PG-FILL-5	PG-FILL-5	PG-FILL-10	PG-FILL-10	PG-UST1-2	PG-UST1-2
Sample Date	Soil	11/6/2000	11/18/2000	11/18/2000	12/1/2000	12/1/2000	11/20/2000	11/20/2000
Sample ID	Cleanup	PGFILL04	PG-FILL-5	PG-FILL-5	PG-FILL10	PG-FILL10	PG-UST1-2	PG-UST1-2
Sample Depth	Objective	6-8'	2-4'	6-8'	3-4'	6-8'	2-4'	12-14'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
1,1,2-TRICHLOROETHANE	6	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
1,1-DICHLOROETHANE	0.2	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.01 2 U	0.0054 U	0.0060 U
1,1-DICHLOROETHYLENE	0.4	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
1,2-DICHLOROETHANE	0.1	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
1,2-DICHLOROPROPANE	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
2-CHLOROETHYL VINYL ETHER	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
ACROLEIN	NS	0.019 U	0.016 U	0.017 U	0.016 U	0.036 U	0.016 U	0.018 U
ACRYLONITRILE	NS	0.0087 U	0.0075 U	0.0081 U	0.0076 U	0.016 U	0.0075 U	0.0082 U
BENZENE	0.06	0.0012 U	0.0011 U	0.0012 U	0.0011 U	0.0024 U	0.0011 U	0.0012 U
BROMODICHLOROMETHANE	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
BROMOFORM	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
BROMOMETHANE	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
CARBON TETRACHLORIDE	0.6	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
CHLOROBENZENE	1.7	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
CHLOROETHANE	1.9	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
CHLOROFORM	0.3	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
CHLOROMETHANE	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
CIS-1,3-DICHLOROPROPENE	NS	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.0 12 U	0.0054 U	0.0060 U
DIBROMOCHLOROMETHANE	N/A	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
DICHLOROMETHANE	0.1	0.0066 B	0.0033 JB	0.0051 J	0.0096 B	0.023 B	0.0055	0.0049 J
ETHYLBENZENE	5.5	0.0012 U	0.0011 U	0.0012 U	0.0011 U	0.0024 U	0.0011 U	0.0012 U
M&P-XYLENES	1.2*	0.0025 U	0.0022 U	0.0023 U	0.0022 U	0.0048 U	0.0022 U	0.0024 U
METHYLBENZENE	1.5	0.0012 U	0.0011 U	0.0012 U	0.0011 U	0.0024 U	0.0011 U	0.0012 U
O-XYLENE	1.2*	0.0012 U	0.0011 U	0.0012 U	0.0011 U	0.0024 U	0.0011 U	0.0012 U
TETRACHLOROETHYLENE	1.4	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
TRICHLOROETHYLENE	0.7	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
VINYL CHLORIDE	0.2	0.0062 U	0.0054 U	0.0058 U	0.0055 U	0.012 U	0.0054 U	0.0060 U
TOTAL VOCs	10	0.0066	0.0033	0.0051	0.0096	0.023	0.0055	0.0049

U Undetectable Levels

44

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard

102.320.0

Location	Recommended	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1	PG-UST7-1A
Sample Date	Soil	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/28/2000	11/28/2000
Sample ID	Сleanup	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1	PG-UST7-1A
Sample Depth	Objective	2-4'	10-11'	14-15'	4-6'	12-14'	8-10'	0-2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
1,1,2-TRICHLOROETHANE	6	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
1,1-DICHLOROETHANE	0.2	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
1,1-DICHLOROETHYLENE	0.4	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
1,2-DICHLOROETHANE	0.1	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
1,2-DICHLOROPROPANE	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
2-CHLOROETHYL VINYL ETHER	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
ACROLEIN	NS	0.019 U	0.021 U	0.097 U	0.020 U	0.022 U	0.018 U	0.016 U
ACRYLONITRILE	NS	0.0088 U	0.0095 U	0.045 U	0.0092 U	0.010 U	0.0082 U	0.0075 U
BENZENE	0.06	0.0013 U	0.0014 U	0.0065 U	0.0013 U	0.0015 U	0.0012 U	0.0011 U
BROMODICHLOROMETHANE	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
BROMOFORM	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
BROMOMETHANE	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
CARBON TETRACHLORIDE	0.6	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
CHLOROBENZENE	1.7	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
CHLOROETHANE	1.9	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
CHLOROFORM	0.3	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
CHLOROMETHANE	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U ·
CIS-1,3-DICHLOROPROPENE	NS	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
DIBROMOCHLOROMETHANE	N/A	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
DICHLOROMETHANE	0.1	0.0076 B	0.0078	0.030 J	0.0037 J	0.0038 JB	0.0075 B	0.0060 B
ETHYLBENZENE	5.5	0.0013 U	0.0014 U	0.0065 U	0.0013 U	0.0015 U	0.0012 U	0.0011 U
M&P-XYLENES	1.2*	0.0025 U	0.0027 U	0.013 U	0.0027 U	0.0030 U	0.0024 U	0.0022 U
METHYLBENZENE	1.5	0.0013 U	0.0026	0.0065 U	0.0013 U	0.0044	0.0012 U	0.0011 U
O-XYLENE	1.2*	0.0013 U	0.0014 U	0.0065 U	0.0013 U	0.0015 U	0.0012 U	0.0011 U
TETRACHLOROETHYLENE	1.4	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.007 5 U	0.0059 U	0.0054 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
TRICHLOROETHYLENE	0.7	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0013 J
VINYL CHLORIDE	0.2	0.0063 U	0.0068 U	0.032 U	0.0067 U	0.0075 U	0.0059 U	0.0054 U
TOTAL VOCs	10	0.0076	0.0104	0.0300	0.0037	0.0082	0.0075	0.0073

U Undetectable Levels

40

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard



Location	Recommended	PG-UST7-1B	PG-UST7-2	PG-UST7-2	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7
Sample Date	Soit	11/28/2000	11/21/2000	11/21/2000	11/13/2000	11/13/2000	11/13/2000
Sample ID	Cleanup	PG-UST7-1B	PG-UST7-2	PG-UST7-2	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07
Sample Depth	Objective	2-3.5'	8-10'	10-12'	2.5-4'	4-6'	6-8'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,1,1-TRICHLOROETHANE	0.8	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
1,1,2,2-TETRACHLOROETHANE	0.6	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
1,1,2-TRICHLOROETHANE	6	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
1,1-DICHLOROETHANE	0.2	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
1,1-DICHLOROETHYLENE	0.4	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
1,2-DICHLOROETHANE	0.1	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
1,2-DICHLOROPROPANE	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
2-CHLOROETHYL VINYL ETHER	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
ACROLEIN	NS	0.017 U	0.086 U	0.090 U	0.017 U	0.017 U	0.018 U
ACRYLONITRILE	NS	0.0079 U	0.040 U	0.042 U	0.0077 U	0.0080 U	0.0083 U
BENZENE	0.06	0.0011 U	0.0057 U	0.0060 U	0.0011 U	0.0011 U	0.0012 U
BROMODICHLOROMETHANE	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
BROMOFORM	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
BROMOMETHANE	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
CARBON TETRACHLORIDE	0.6	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
CHLOROBENZENE	1.7	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
CHLOROETHANE	1.9	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
CHLOROFORM	0.3	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
CHLOROMETHANE	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
CIS-1,3-DICHLOROPROPENE	NS	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
DIBROMOCHLOROMETHANE	N/A	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
DICHLOROMETHANE	0.1	0.0075 B	0.016 JB	0.015 JB	0.0018 JB	0.0026 JB	0.0020 JB
ETHYLBENZENE	5.5	0.0011 U	0.0057 U	0.0060 U	0.0011 U	0.0011 U	0.0012 U
M&P-XYLENES	1.2*	0.0014 J	0.011 U	0.012 U	0.0022 U	0.0023 U	0.0024 U
METHYLBENZENE	1.5	0.0011 U	0.0057 U	0.0060 U	0.0011 U	0.0011 U	0.0012 U
O-XYLENE	1.2*	0.0024	0.0057 U	0.0060 U	0.0011 U	0.0011 U	0.0012 U
TETRACHLOROETHYLENE	1.4	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
TRANS-1,2-DICHLOROETHYLENE	0.3	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
TRANS-1,3-DICHLOROPROPENE	0.3	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
TRICHLOROETHYLENE	0.7	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
VINYL CHLORIDE	0.2	0.0057 U	0.029 U	0.030 U	0.0056 U	0.0057 U	0.0060 U
TOTAL VOCs	10	0.0113	0.0160	0.0150	0.0018	0.0026	0.0020

U Undetectable Levels

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard
| Location | Recommended | PG-PA-MW-7 | PG-PA-MW-10D | PG-PA-MW-10D | PG-MW10D | PG-MW10D |
|----------------------------|-------------|------------|--------------|--------------|------------|------------|
| Sample Date | Soil | 11/13/2000 | 11/27/2000 | 11/27/2000 | 11/27/2000 | 11/27/2000 |
| Sample ID | Cleanup | PG-PAMW-07 | PG-PAMW10D | PG-PAMW10D | PG-MW10D | PG-MW10D |
| Sample Depth | Objective | 8-10' | 0.5-2' | 4-6' | 7-8' | 8-10' |
| Concentration | MG/KG | MG/KG | MG/KG | MG/KG | MG/KG | MG/KG |
| - | | | | | | |
| 1,1,1-TRICHLOROETHANE | 0.8 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 1,1,2,2-TETRACHLOROETHANE | 0.6 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 1,1,2-TRICHLOROETHANE | 6 | 0.0057 Ų | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 1,1-DICHLOROETHANE | 0.2 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 1,1-DICHLOROETHYLENE | 0.4 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 1,2-DICHLOROETHANE | 0.1 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 1,2-DICHLOROPROPANE | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| 2-CHLOROETHYL VINYL ETHER | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| ACROLEIN | NS | 0.017 U | 0.017 U | 0.017 U | 0.021 U | 2.9 U |
| ACRYLONITRILE | NS | 0.0079 U | 0.0077 U | 0.0079 U | 0.0095 U | 1.3 U |
| BENZENE | 0.06 | 0.0011 U | 0.0011 U | 0.0011 U | 0.0014 U | 0.32 |
| BROMODICHLOROMETHANE | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| BROMOFORM | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| BROMOMETHANE | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| CARBON TETRACHLORIDE | 0.6 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| CHLOROBENZENE | 1.7 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| CHLOROETHANE | 1.9 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| CHLOROFORM | 0.3 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| CHLOROMETHANE | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| CIS-1,3-DICHLOROPROPENE | NS | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| DIBROMOCHLOROMETHANE | N/A | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| DICHLOROMETHANE | 0.1 | 0.0021 JB | 0.0021 JB | 0.0024 JB | 0.010 B | 0.96 U |
| ETHYLBENZENE | 5.5 | 0.0011 U | 0.0011 U | 0.0011 U | 0.0014 U | 0.28 |
| M&P-XYLENES | 1.2* | 0.0023 U | 0.0022 U | 0.0023 U | 0.0027 U | 0.72 |
| METHYLBENZENE | 1.5 | 0.0011 U | 0.0011 U | 0.0011 U | 0.0027 | 0.43 |
| O-XYLENE | 1.2* | 0.0011 U | 0.0011 U | 0.0011 U | 0.0014 U | 0.24 |
| TETRACHLOROETHYLENE | 1.4 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| TRANS-1,2-DICHLOROETHYLENE | 0.3 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| TRANS-1,3-DICHLOROPROPENE | 0.3 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| TRICHLOROETHYLENE | 0.7 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| VINYL CHLORIDE | 0.2 | 0.0057 U | 0.0056 U | 0.0057 U | 0.0068 U | 0.96 U |
| TOTAL VOCs | 10 | 0.0021 | 0.0021 | 0.0024 | 0.0127 | 1.99 |

.

U Undetectable Levels

e qui diffe

ND Not Detectived

NS No Standard

* Total Xylene Recommended Cleanup Standard

Leastion	Pacammandad	PC-A-4	PC-A-4	PC-A-5	PC_R.1	PC-B-1	PC-B-1	PC_B_02	PC_B_02	PC_B-02A
Second Date	Sall	11/16/2000	11/16/2000	11/14/2000	12/4/2000	12/4/2000	12/4/2000	11/16/2000	1467000	11/16/2000
Sample Date	501	DC 1 01	11/10/2000	DC A 05	DC D 01	12/4/2000	127472000 DC D 01	DC D 03	DC A 02	DC D 034
Sample ID	Cleanup	PG-A-04	PG-A-04	PG-A-05	PG-B-01	PG-B-01	PG-B-01	PG-B-02	PG-A-02	PG-B-02A
Sample Depth	Objective	6-8'	12-14	2-4'	2-4	6-8.	9-10	2-4	6-8.	8-10
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,2,4-TRICHLOROBENZENE	3.4	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
1,2-BENZPHENANTHRACENE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
1,2-DICHLOROBENZENE	7.9	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
1,2-DIPHENYLHYDRAZINE	NS	0.046 U	0.071 U	1.6 U	0.038 U	0.040 U	0.042 U	0.039 U	1.9 U	0.044 U
1,4-DICHLOROBENZENE	8.5	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2,4,6-TRICHLORORPHENOL	0.1	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2,4-DICHLOROPHENOL	0.4	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2.4-DIMETHYLPHENOL	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2,4-DINITROPHENOL	0.200 or MDL	0.46 U	0.71 U	16 U	0.38 U	0.40 U	0.42 U	0.39 U	19 U	0.44 U
2.4-DINITROTOLUENE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2.6-DINITROTOLUENE	1	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2-CHLORORNAPHTHALENE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
2-CHLOROPHENOL	0.8	0.2311	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 11
2-NITROPHENOI	0.330 or MDL	0 23 11	0 35 11	8211	0 19 11	0 20 11	0 21 11	0 20 11	9311	0.22.12
2 INTROPRETATIONE	N/A	0 23 11	0 35 11	8 7 11	0 19 11	0.2011	0.211	0 20 11	9311	0.2211
1,5-DICHEOROBENZIDINE	0.100 or MDI	0.22 0	0.35 U	8.2.0	0.190	0.20 U	0.21 U	0.20 U	0.3.0	0.22 0
4,6-DINITRO-O-CRESOL	0.100 01 1102	0.23 0	0.33 0	8.2 0	0.190	0.20 U	0.21 0	0.20 0	9.3 0	0.22 0
4-BROMOPHENYLPHENYL ETHER	NS	0.23 0	0.35 ()	8.20	0.190	0.20 0	0.21 0	0.20 0	9.30	0.22 0
4-CHLORO-3-METHYLPHENOL	0.240 OF MDL	0.23 U	0.35 U	8.20	0.19 0	0.20 U	0.21 0	0.20 ()	9.3 0	10.22 0
4-CHLOROPHENYLPHENYL ETHER	NS	0.23 U	0.35 U	8.2 U	0.19 0	0.20 U	0.21 U	0.20 U	9.3 0	0.22 U
4-NITROPHENOL	0.100 or MDL	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
ACENAPHTHENE	50	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
ACENAPHTHYLENE	41	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
ANTHRACENE	50	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BENZIDINE	NS	0.46 U	0.71 U	16 U	0.38 U	0.40 U	0.42 U	0.39 U	19 U	0.44 U
BENZO[A]ANTHRACENE	0.224 or MDL	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BENZO[A]PYRENE	0.061 or MDL	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.099 J	0.20 U	9.3 U	0.22 U
BENZO[B]FLOURANTHENE	1.1	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BENZO[G,H,I]PERYLENE	50	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BENZOIKIFLOURANTHENE	1.1	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BENZYL BUTYL PHTHALATE	50	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BIS(2-CHLOROFTHOXY)METHANE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BIS(2-CHLOROFTHYL)FTHER	NS	0.231	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	0 23 11	0 35 11	8211	0 19 11	0 20 11	0.21.11	0 20 11	9311	0 22 11
BIS(2-CHEOROISOT ROT TEJETTER	50	0.19 10	0.33 0	8211	0.074 JB	0.079 /B	0.21.0	0.20 0	0.3.U	0.003 IB
BIS(2-EINTHEXTL)FHIMALATE	81	0.1835	0.15 31	18.2 U	0.1011	0.07330	0.21 0	0.07930	0.2.11	0.033.70
DI-N-BOTTL PHTHALATE	50	0.25 0	0.071 18	8.2.0	0.170	0.1578	0.038 JB	0.200	0.2.11	0.45
DI-N-OCTYL PHTHALATE	0.014 or MDI	0.10 JB	0.073 JB	8.2 0	0.004 18	0.054 JB	0.045 JB	0.041 JB	9.3 0	0.037 JB
DIBENZ(A,HJAN THRACENE	0.014 01 MDL	0.23 U	0.35 0	8.20	0.190	0.20 0	0.21 0	0.20 0	9.3 0	0.22 0
DIETHYL PHTHALATE	7.1	0.23 ()	0.35 0	8.20	0.190	0.20 0	0.21 U	0.20 U	9.3 0	0.22 0
DIMETHYL PHTHALATE	2	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
FLUORANTHENE	50	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
FLUORENE	50	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
HEXACHLORO-1,3-BUTADIENE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
HEXACHLOROBENZENE	0.41	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
HEXACHLOROCYCLOPENTADIENE	NS	0.68 U	1.1 U	25 U	0.57 U	0.60 U	0.63 U	0.59 U	28 U	0.66 U
HEXACHLOROETHANE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
INDENO[1,2,3-CD]PYRENE	3.2	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
ISOPHORORNE	4.4	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
M-DICHLOROBENZENE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
N-NITROSO-DI-N-PROPYLAMINE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
N-NITROSODIMETHYLAMINE	NS	0.23 U	0.35 U	8.2 U	0.19 U	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
N-NITROSODIPHENVI AMINE	NS	0.23 U	0.35 U	8.2 U	0.1911	0.20 U	0.21 U	0.20 U	9.3 U	0.22 U
NAPHTHAI ENE	13	0.231/	0.351	8.2 11	0.1911	0.201	0.21 U	0.20 U	9.3 U	0.22 []
NITROBENZENE	0.200 or MDI	0 23 11	0351	8711	0 1911	0 20 11	0 21 11	0 20 11	9311	0 22 11
DENTACHLODODUENOI	1.0 or MDI	0.23 1/	0 35 11	8213	0 10 11	0.2017	0.21 U	0.201/	0.11	0 22 11
PUENANTUDENE	50 50	0.23 U	0.35 0	0.20	0.19.0	0.20 0	0.21 U	0.20 U	0.212	0.22 1/
PHENANIHKENE		0.25 0	0.33 0	0.20	0.190	0.200	0.21 0	0.20 0	0.213	0.22 0
PHENOL		0.23 0	0.261	16.2 0	0.190	0.200	0.21 U	0.20 0	19.3 U	0.22 0
PYRENE		0.23 U	0.35 U	8.2.0	0.190	0.20 0	0.21 0	0.20 0	19.3 U	0.22 0
TOTAL SVOCs	1 500	0.28	0.573	מאן	0.138	0.263	10.202	10.12	IND	10.58

U Undetectable Levels NS No Standard

\$

MDL Method Detection Limit

96

and the second second

**

No. of the state o

s¥nak .

A SALE AND AND A SALE A

at the second

Instrume Sole Duckbool Pure No Pure No <th< th=""><th>Location</th><th>Recommended</th><th>PC-B-3</th><th>PC-R-1</th><th>PC_B.4</th><th>PC-F-1</th><th>PC_F_1</th><th>PC-E-1</th><th>PC-F2-2</th><th>PC-52-2</th></th<>	Location	Recommended	PC-B-3	PC-R-1	PC_B.4	PC-F-1	PC_F_1	PC-E-1	PC-F2-2	PC-52-2
marging marging <t< td=""><td>Sample Data</td><td>Sail</td><td>13/4/3000</td><td>12/4/2000</td><td>12/4/2000</td><td>11/12/2000</td><td>11/12/2000</td><td>11/12/2000</td><td>1100000</td><td>11/20/2000</td></t<>	Sample Data	Sail	13/4/3000	12/4/2000	12/4/2000	11/12/2000	11/12/2000	11/12/2000	1100000	11/20/2000
Description Description PARAGE <	Sample Date	Cleanup	PC P 03	DC D 02	DC D 04	DC E 01	PC E 01	PC E 01	PC 52 2	PC 52.2
Description Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	Sample Donth	Objective	10-0-03 IN 1	ru-b-03	2 4	0.2.2	1.6	10.12	2 4	R 10'
DALE DECK DECK <thdeck< th=""> DECK DECK <thd< td=""><td>Concentration</td><td>MC/KC</td><td>2-4 MG/KG</td><td>MCMC</td><td>12-4 MC/KG</td><td>0.2-2 MG/KG</td><td>4-0 MG/KG</td><td>MGNG</td><td>MGRG</td><td>MG/KG</td></thd<></thdeck<>	Concentration	MC/KC	2-4 MG/KG	MCMC	12-4 MC/KG	0.2-2 MG/KG	4-0 MG/KG	MGNG	MGRG	MG/KG
Description Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	1 2 4-TRICHLOROBENZENE	3.4	1211	0 27 11	0 20 11	0 19 13	0 19 11	03111	0 18 11	0 19 11
Discrition BoomestZenie 1.9 1.0 0.37U 0.80U 0.80U <td>1.2.PENZPHENANTHRACENE</td> <td>J.4 NS</td> <td>1.2.0</td> <td>0.27 0</td> <td>0.20 0</td> <td>0.190</td> <td>0.1917</td> <td>0.310</td> <td>0.18 U</td> <td>0.19.0</td>	1.2.PENZPHENANTHRACENE	J.4 NS	1.2.0	0.27 0	0.20 0	0.190	0.1917	0.310	0.18 U	0.19.0
Dependence NS Description Description <thdesc< td=""><td>1.2-DICHLOPORENZENE</td><td>79</td><td>1.20</td><td>0.27 U</td><td>0.2011</td><td>0.30</td><td>0.191</td><td>0.31 U</td><td>0.18 0</td><td>0.190</td></thdesc<>	1.2-DICHLOPORENZENE	79	1.20	0.27 U	0.2011	0.30	0.191	0.31 U	0.18 0	0.190
Description Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	1.2-DIPHENVI HVDP AZINE	NS	0.23.11	0.27 0	0.200	0.190	0.190	0.510	0.18 0	0.190
Description Dist		85	0.25 0	0.033 0	0.039 0	0.038 0	0.0390	0.005 0	0.030 0	0.059 0
Description 0.4 12U 0.20U 0.10U 0.10U 0.11U 0.18U 0.0893 2-DIMITION PRINDL 0.200 PMDL 2.11U 0.51U 0.51U<	2 4 6-TRICHLOROBENEENOL	0.1	1.2.0	0.27 1	0.20 0	0.1911	0.19.0	0.31 U	0.18 1/	0.092 1
DADMITRYLIPHENOL NS 12U 627U	2 4-DICHLOROPHENOL	0.4	1.2.0	0.27 U	0.20 0	0.1911	0.1911	0.310	0.18 U	0.059 1
Density Density Density <thdensity< th=""> Density <thdensity< th=""></thdensity<></thdensity<>	2 4-DIMETHYL PHENOL	NS	1.2.0	0.27 []	0 20 11	0.1911	0.1911	0.31 U	0 18 11	0 19 11
2-ADMINSTORULENE DEN 2/21 <th2 21<="" th=""> 2/21 2/21</th2>	2.4-DINITROPHENOL	0.200 or MDI	2.311	0.55 U	0 39 11	0 38 11	0 39 11	0.63 U	0.16 0	0 39 11
2x-DBNTROTOLIENE 1 12U 021U 0.91U	2.4-DINITROTOLUENE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
2CHLORONNAPHTHALEVE NS 12U 0.2U 0.2U <th0.2u< th=""> 0.2U 0.2U</th0.2u<>	2.6-DINITROTOLUENE	1	1.2 U	0.27 U	0.20 U	0.19 U	0.191	0.31 U	0.18 U	0.19 U
2CHLOROPHENOL 0.9 1 ± U 0.20 U 0.9 U	2-CHLORORNAPHTHALENE	NS	1.2 U	0.27 1	0.2013	0.1911	0.19 U	0.31 U	0.18 U	0.040 J
SANTROPHENOL 0.330 or MOL 121 0.27U 0.19 U 0.19 U 0.18 U 0.18 U 3-DOCHLOROBENZIDINE N/A 12 U 0.27 U 0.20 U 0.19 U 0.19 U 0.11 U 0.18 U	2-CHLOROPHENOL	0.8	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
J.Y.DCHLOROBENZIDNE N/A 12 U 927 U 0.90 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U 4-SDINTGO-CRESQL 0.100 or MDL 12 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U 4-CHLOROHENCLPHENVLEHENDL 0.240 or MDL 12 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U 4-CHLOROHENCL 0.240 or MDL 12 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U 4-CHLOROHENCL 0.100 or MDL 12 U 0.27 U 1.2 0.19 U 0.19 U 0.18 U 0.08 U 0.18 U 0.06 L 4-CENAPHTRINE 50 1.2 U 0.27 U 1.1 0.040 U 0.31 U 0.18 U 0.06 L 0.18 U 0.06 L 0.18 U 0.06 L 0.20 U 0.21 U	2-NITROPHENOL	0.330 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
A-DINTRO-O-CRESOL 0.100 or MDL 1:2U 0.27U 0.29U 0.19U 0.19U 0.11U 0.18U 0.18U 0.19U 4BROMOPHENYLPHENYL ETHER NS 1:2U 0.27U 0.20U 0.19U 0.31U 0.18U 0.19U 4CHLORO-J-MENYLPHENYL ETHER NS 1:2U 0.27U 0.20U 0.19U 0.31U 0.18U 0.19U ACRAPHTHENE 50 1:2U 0.27U 0.20U 0.19U 0.31U 0.18U 0.18U 0.19U ACENAPHTHENE 50 1:2U 0.27U 1:2 0.19U 0.31U 0.18U 0.0621 ACENAPHTHENE 50 1:2U 0.27U 1:1 0.19U 0.31U 0.18U 0.0421 BENZOJANTHEACENE 0.224 or MOL 1:2U 0.27U 1:2 0.31 0.19U 0.31U 0.18U 0.0431 BENZOJANTHEACENE 50 1:2U 0.27U 1:2 0.31 0.19U 0.31U 0.18U 0.19U BENZOJANTHEACENE	3,3'-DICHLOROBENZIDINE	N/A	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
ABBOWDHENVLPHENVLETHER NS 12.U 0.27U 0.20U 0.19U 0.19U 0.11U 0.18.U 0.19U ACHLORD-JMETTUYPEEVOL 0.200 of MDL 1.2U 0.22U 0.20U 0.19U 0.19U 0.11U 0.18.U 0.19U AVITRACENE NS 1.2U 0.27U 0.20U 0.19U 0.19U 0.19U 0.19U 0.18U 0.18U 0.19U AVITRACENE 50 1.2U 0.27U 1.2 0.19U 0.19U 0.11U 0.18U 0.18U 0.0621 ACENAPHTHINE 41 0.411 0.27U 1.06661 0.19U 0.31U 0.18U 0.042 ACENAPHTHYLENE 50 1.2U 0.27U 1.6 0.631 0.19U 0.31U 0.18U 0.043 BENZQIAJAVITKACENE 0.224 or MDL 1.2U 0.27U 1.2 0.31 0.19U 0.31U 0.18U 0.19U BENZQIAJAURANTHENE 1.2U 0.27U 0.70 0.111 0.19U 0.31U <td< td=""><td>4,6-DINITRO-O-CRESOL</td><td>0.100 or MDL</td><td>1.2 U</td><td>0.27 U</td><td>0.20 U</td><td>0.19 U</td><td>0.19 U</td><td>0.31 U</td><td>0.18 U</td><td>0.19 U</td></td<>	4,6-DINITRO-O-CRESOL	0.100 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
ACHLORD-JMETHYLPHENOL 0.24 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.11 U 0.18 U 0.01 U ACHLOROPHENYLPHENYL ETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.11 U 0.18 U 0.18 U 0.19 U ALTROPHENOL 0.100 or MDL 1.2 U 0.27 U 1.2 U 0.19 U 0.19 U 0.19 U 0.18 U 0.06 J ACENAPHTHYLENE 41 0.41 U 0.27 U 1.1 U 0.19 U 0.19 U 0.18 U 0.06 J 0.19 U 0.18 U 0.06 J 0.19 U 0.19 U 0.18 U 0.06 J 0.19 U 0.19 U 0.18 U 0.04 J BENZORAPHENE 0.06 J 0.05 U 0.27 U 0.75 O 0.23 C 0.19 U 0.11 U 0.18 U 0.18 U 0.04 J BENZORAPHENENE 50 1.2 U 0.27 U 0.70 U 0.11 J 0.19 U 0.11 U 0.18 U 0.18 U 0.40 J BENZORAPTICHENY ENENE 50 1.2 U 0.27 U <	4-BROMOPHENYLPHENYL ETHER	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
ACHLOROPHENYLPHENYL ETHER NS 12U 027U 020U 0.19U 0.19U 0.18U 0.18U 0.19U ANTROPHENGL 0.100 or MDL 1.2U 0.27U 1.2 0.19U 0.19U 0.11U 0.18U 0.18U 0.621 ACENAMTHANENE 41 0.41 0.27U 1.1 0.664 0.19U 0.31U 0.18U 0.6621 ACENAMTHANCENE 50 1.2U 0.27U 1.1 0.664 0.19U 0.31U 0.18U 0.040 BENZOJANTHRACENE 0.224 or MDL 0.2U 0.31U 0.18U 0.040 0.31U 0.18U 0.041 BENZOJANTHRACENE 0.224 or MDL 0.2U 0.27U 0.76 0.425 0.19U 0.31U 0.18U 0.18U 0.19U BENZOJANTHRACENE 1.1 1.2U 0.27U 0.20 0.19U 0.31U 0.18U 0.18U 0.19U BENZOJANJVLENENE 1.1 1.2U 0.27U 0.20U 0.19U 0.11U 0.18U	4-CHLORO-3-METHYLPHENOL	0.240 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.079 J
ENTROPHENCL 0.100 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U ACENAPHTHENE 50 1.2 U 0.27 U 1.2 0.19 U 0.19 U 0.31 U 0.18 U 0.062 J ACENAPHTHYLENE 41 0.41 J 0.27 U 1.1 0.046 J 0.31 U 0.18 U 0.062 J ANTRACENE 50 1.2 U 0.27 U 1.1 0.046 J 0.31 U 0.18 U 0.045 J 0.99 U 0.31 U 0.18 U 0.045 J 0.19 U 0.31 U 0.18 U 0.049 J 0.91 U 0.31 U 0.18 U 0.49 J 0.91 U 0.92 U 0.92	4-CHLOROPHENYLPHENYL ETHER	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
ACENAPITHENE 50 12 U 0.27 U 12 0.19 U 0.31 U 0.18 U 0.0621 ACENAPITHYLENE 41 0.41 J 0.27 U 0.11 J 0.19 U 0.31 U 0.18 U 0.046 J ANTHACENE 50 1.2 U 0.27 U 1.1 0.046 J 0.19 U 0.31 U 0.18 U 0.046 J BENZOJANTHRACENE NS 2.2 U 0.52 U 0.32 U 0.31 U 0.18 U 0.43 U 0.43 J BENZOJANTHRACENE 0.024 Or MDL I.2 U 0.27 U 0.75 0.23 U 0.31 U 0.18 U 0.18 U 0.19 U BENZOJGJUPRENLENE 50 1.2 U 0.27 U 0.70 0.11 J 0.19 U 0.31 U 0.18 U 0.19 U BENZOJGULARNTHENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BENZOJGULARNTHENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.40 J BENZOJGULARNTHENE 51 1	4-NITROPHENOL	0.100 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
ACEBAPHTHYLENE 41 0.41 0.27 0 0.11 0.19 0.31 0 0.18 0 0.0461 ANTHRACENE 50 1.2 0.27 1 0.0661 0.31 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.31 0.18 0.31 0.18 0.31 0.18 0.31 0.31 0.31 <td< td=""><td>ACENAPHTHENE</td><td>50</td><td>1.2 U</td><td>0.27 U</td><td>1.2</td><td>0.19 U</td><td>0.19 U</td><td>0.31 U</td><td>0.18 U</td><td>0.062 J</td></td<>	ACENAPHTHENE	50	1.2 U	0.27 U	1.2	0.19 U	0.19 U	0.31 U	0.18 U	0.062 J
ANTRACENE 50 1.2 U 0.27 U 1.1 0.64 J 0.9 U 0.18 U 0.19 U BENZDIANE NS 2.3 U 0.55 U 0.39 U 0.30 U 0.31 U 0.18 U 0.04 U 0.04 U 0.04 U 0.04 U 0.01 U 0.11 U 0.18 U 0.04 U 0.04 U 0.04 U 0.01 U 0.18 U 0.04 U 0.04 U 0.01 U 0.18 U 0.01 U 0.18 U 0.19 U 0.18 U 0.01 U 0.18 U 0.01 U 0.18 U 0.04 U 0.18 U 0.04 U 0.18 U 0.04 U	ACENAPHTHYLENE	41	0.41 J	0.27 U	0.11 J	0.19 U	0.19 U	0.31 U	0.18 U	0.046 J
BENZDIANE NS 2.3.U 0.55 U 0.39 U 0.39 U 0.30 U 0.36 U 0.36 U 0.39 U BENZQA(ANTHRACENE 0.224 or MDL 1.2 U 0.27 U 6.3 U 0.19 U 0.31 U 0.18 U 0.043 J BENZQA(PYRENE 0.061 or MOL 1.2 U 0.27 U 1.2 0.31 U 0.18 U 0.19 U BENZQA(PYRENE 1.1 1.2 U 0.27 U 0.70 U 0.14 J 0.19 U 0.31 U 0.18 U 0.19 U BENZQA(K)FLORANTHENE 1.1 1.2 U 0.27 U 0.70 U 0.14 J 0.19 U 0.31 U 0.18 U 0.19 U BENZQA(K)FLORANTHENE 1.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BENZQA(K)FLORANTHENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROSTROPLICITIER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.040 J DIN-BUTZ/FITHALATE	ANTHRACENE	50	1.2 U	0.27 U	1.1	0.046 J	0.19 U	0.31 U	0.18 U	0.19 U
BENZQIAJANTHRACENE 0.224 or MOL 1.2 0.27 6.3 0.31 0.19 0.18 0.0431 BENZQIAJPYRENE 0.061 or MDL 1.2 0.27 0.75 0.23 0.19 0.31 0.18 0.040 1.1 0.19 0.31 0.18 0.040 </td <td>BENZIDINE</td> <td>NS</td> <td>2.3 U</td> <td>0.55 U</td> <td>0.39 U</td> <td>0.38 U</td> <td>0.39 U</td> <td>0.63 U</td> <td>0.36 U</td> <td>0.39 U</td>	BENZIDINE	NS	2.3 U	0.55 U	0.39 U	0.38 U	0.39 U	0.63 U	0.36 U	0.39 U
BENZQAJPYRENE 0.061 or MOL 1.2U 0.27 U 0.75 0.25 0.19 U 0.11 U 0.18 U 0.19 U BENZQGAJPYRENE 50 1.2U 0.27 U 1.2 0.31 U 0.19 U 0.31 U 0.18 U 0.19 U BENZQGAJPYRENE 50 1.2U 0.27 U 0.70 0.11 J 0.19 U 0.31 U 0.18 U 0.40 J BENZQGAJPYRENE 1.1 1.2U 0.27 U 0.70 0.11 U 0.19 U 0.31 U 0.18 U 0.40 J BENZQALORETHOXYMETHANE NS 1.2U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROETHOXYMETHANE NS 1.2U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROETHXYLETHE NS 1.2U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROETHXYLETHER NS 1.2U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.045 J	BENZO[A]ANTHRACENE	0.224 or MDL	1.2 U	0.27 U	1	0.31	0.19 U	0.31 U	0.18 U	0.043 J
BERZQ[B]FLOURANTHENE 1.1 1.2 U 0.27 U 1.2 0.14 J 0.19 U 0.11 U 0.18 U 0.19 U BENZQ[FLOURANTHENE 1.1 1.2 U 0.27 U 0.70 0.11 J 0.19 U 0.11 U 0.18 U 0.040 J BENZQ[FLOURANTHENE 1.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.040 J BENZQ[FLOURANTHENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROETHYL)ENTHALATE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROETHYL)ENTHALATE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROISOROPOYL)ENTHALATE S.1 1.2 U 0.27 U 0.20 B 0.40 J 0.10 J 0.07 J 0.18 U 0.046 J DIN-NUTYL PHTHALATE S.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.060 J <td>BENZO[A]PYRENE</td> <td>0.061 or MDL</td> <td>1.2 U</td> <td>0.27 U</td> <td>0.75</td> <td>0.25</td> <td>0.19 U</td> <td>0.31 U</td> <td>0.18 U</td> <td>0.19 U</td>	BENZO[A]PYRENE	0.061 or MDL	1.2 U	0.27 U	0.75	0.25	0.19 U	0.31 U	0.18 U	0.19 U
BENZQICH.IJPERVLENE 50 1.2 U 0.27 U 0.17 J 0.14 J 0.19 U 0.31 U 0.18 U 0.19 U BENZQI(KJFLOURANTHENE 1.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.040 J BENZYL BUTYL PHTHALATE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U BIS2-CHLOROETHOXYMETHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS2-CHLOROETHYL)ETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS2-CHLOROETHYL)ETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.043 J DIN-BUTYL PHTHALATE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J DIETHYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J DIETHYL PHT	BENZO[B]FLOURANTHENE	1.1	1.2 U	0.27 U	1.2	0.31	0.19 U	0.31 U	0.18 U	0.19 U
BERZQIKJELOURANTHENE 1.1 1.2 U 0.70 0.11 J 0.19 U 0.31 U 0.18 U 0.040 J BENZYL BUTYL PITHALATE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.13 U 0.18 U 0.19 U BIS(2-CHLOROETHOXY)METHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROETHOXY)METHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROETHOXY)METHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-ETHYREXYL)PHTHALATE 50 0.32 JB 2.4B 0.37 B 0.27 0.20 L 0.07 J 0.18 U 0.043 J DIN-OCTYL PHTHALATE 6.1 1.2 U 0.27 U 0.26 B 0.19 U 0.31 U 0.18 U 0.046 J DIBENZAJNJANTRACENE 0.014 or MOL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J DIMETHYL PHT	BENZO[G,H,I]PERYLENE	50	1.2 U	0.27 U	0.17 J	0.14 J	0.19 U	0.31 U	0.18 U	0.19 U
BENZYL BUTYL PHTHALATE 50 1.2 U 0.2 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROETHOXY)METHARE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROETHV)ETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS(2-CHLOROISOPROPYL)ETHER NS 1.2 U 0.27 U 0.20 B 0.40 J 0.07 J 0.80 U 0.48 U 0.48 U 0.49 U DIN-BUTYL PHTHALATE 50 1.2 U 0.27 U 0.20 B 0.40 J 0.07 J 0.18 U 0.49 J DIN-BUTYL PHTHALATE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.46 J DIBETHYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.46 J DIMETHYL PHTHALATE 2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.46 J FLUORENE	BENZOKIFLOURANTHENE	1.1	1.2 U	0.27 U	0.70	0.11 J	0.19 U	0.31 U	0.18 U	0.040 J
BIS/2-CHLOROETHOXY/METHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROETHYLJETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROISOPROPYLJETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS/2-CHLOROISOPROPYLJETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.01 U 0.18 U 0.19 U BIS/2-CHLOROISOPROPYLJETHER S.1 1.2 U 0.27 U 0.20 B 0.0401 U 0.01 U 0.18 U 0.403 U DIN-OCTYL PHTHALATE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.404 J DIBENZ[A,HJANTHRACENE 0.014 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.404 J DIBENTYL, PHTHALATE 2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.404 J FL	BENZYL BUTYL PHTHALATE	50	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
BIS2-CHLOROETHYLJETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U BIS2-CHLOROISOPROPYLETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BIS2-ETHYLEXYLD/PITHALATE 50 0.32 JB 2.24 B 0.37 B 0.27 O 0.22 O 0.077 J 0.18 U 0.043 J DI-N-BUTYL PITHALATE 50 1.2 U 0.27 U 0.28 B 0.19 U 0.053 J 0.15 U 0.046 J 0.060 J DIENZ(AHJANTHACENE 0.014 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J DIENTYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J FLUORANTHERE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.065 J FLUORENE 80 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.066 J HEXACHL	BIS(2-CHLOROETHOXY)METHANE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
BISC-CHLOROISOPROPYLJETHER NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U BISC-ETHYHEXYLJPHTHALATE 50 0.32 JB 2.4 B 0.37 B 0.27 U 0.22 U 0.20 J 0.075 JB 0.098 JB DI-N-BUTYL PHTHALATE 5.1 1.2 U 0.27 U 0.20 B 0.040 J 0.10 J 0.077 J 0.18 U 0.040 J DI-N-OCTYL PHTHALATE 50 1.2 U 0.27 U 0.260 G////////////////////////////////////	BIS(2-CHLOROETHYL)ETHER	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
BIS(2-ETHYHEXYL)PHTHALATE 50 0.32 JB 2.4 B 0.37 B 0.27 0.22 0.20 0.075 JB 0.098 JB DI-N-BUTYL PHTHALATE 8.1 1.2 U 0.27 U 0.20 B 0.40 J 0.10 J 0.077 J 0.18 U 0.443 J DI-N-OCTYL PHTHALATE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.404 J DIBENZ(A.HJANTHRACENE 0.014 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.447 J DIMETHYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.447 J DIMETHYL PHTHALATE 2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.404 J FLUORANTHENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.404 J FLUORANTHENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.61 J HEXACHLOROCH_J-BUTAN	BIS(2-CHLOROISOPROPYL)ETHER	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
DI-N.BUTYL PHTHALATE 8.1 1.2 U 0.27 U 0.20 B 0.40 J 0.10 J 0.07 J 0.18 U 0.043 J DI-N-OCTYL PHTHALATE 50 1.2 U 0.27 U 0.28 B 0.19 U 0.053 J 0.15 J 0.046 J 0.060 J DIBENZ[A,H]ANTHRACENE 0.14 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.047 J DIMENT[A,H]ANTHRACENE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.047 J DIMENT[A,H]ANTHRACENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.040 J FLUORENE 50 1.2 U 0.27 U 0.24 U 0.19 U 0.31 U 0.18 U 0.060 J HEXACHLORO-1,3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U IND	BIS(2-ETHYHEXYL)PHTHALATE	50	0.32 JB	2.4 B	0.37 B	0.27	0.22	0.20 J	0.075 JB	0.098 JB
DI-N-OCTYL PHTHALATE 50 1.2 U 0.27 U 0.28 B 0.19 U 0.053 J 0.15 J 0.046 J 0.060 J DIBEXZ(A.HJANTRACENE 0.014 or MOL 1.2 U 0.27 U 0.060 J 0.049 J 0.31 U 0.18 U 0.19 U DIETHYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J DIMETHYL PHTHALATE 2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J FLUORANTHENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J FLUORENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J HEXACHLOROCHJS-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.66 I J HEXACHLOROETJANEN NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U INDERHORORNE 4.4 <td< td=""><td>DI-N-BUTYL PHTHALATE</td><td>8.1</td><td>1.2 U</td><td>0.27 U</td><td>0.20 B</td><td>0.040 J</td><td>0.10 J</td><td>0.077 J</td><td>0.18 U</td><td>0.043 J</td></td<>	DI-N-BUTYL PHTHALATE	8.1	1.2 U	0.27 U	0.20 B	0.040 J	0.10 J	0.077 J	0.18 U	0.043 J
DIBENZ[A,H]ANTHRACENE 0.014 or MDL 1.2 U 0.27 U 0.060.1 0.049 U 0.19 U 0.18 U 0.19 U DIETHYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.047 J DIMETHYL PHTHALATE 2 1.2 U 0.27 U 4.2 0.40 0.19 U 0.31 U 0.18 U 0.040 J FLUORANTHENE 50 1.2 U 0.27 U 4.2 0.40 0.19 U 0.31 U 0.18 U 0.040 J FLUORANTHENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.065 J HEXACHLORO-1.3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.82 U 0.57 U 0.58 U 0.54 U 0.58 U 0.54 U 0.58 U HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.11 U 0.18 U	DI-N-OCTYL PHTHALATE	50	1.2 U	0.27 U	0.28 B	0.19 U	0.053 J	0.15 J	0.046 J	0.060 J
DIETHYL PHTHALATE 7.1 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.047 J DIMETHYL PHTHALATE 2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J FLUORANTHENE 50 1.2 U 0.27 U 4.2 0.40 0.19 U 0.31 U 0.18 U 0.040 J FLUORENE 50 1.2 U 0.27 U 0.84 0.19 U 0.31 U 0.18 U 0.060 J HEXACHLORO-1,3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.22 U 0.59 U 0.57 U 0.58 U 0.94 U 0.54 U 0.58 U HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.22 U 0.19 U 0.31 U 0.18 U 0.19 U IDOENO(1,2,3-CDJPYRENE 3.2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U ISOPHORONE 4.4 1.2 U <td>DIBENZ[A,H]ANTHRACENE</td> <td>0.014 or MDL</td> <td>1.2 U</td> <td>0.27 U</td> <td>0.060 J</td> <td>0.049 J 🔬 🔬</td> <td>0.19 U</td> <td>0.31 U</td> <td>0.18 U</td> <td>0.19 U</td>	DIBENZ[A,H]ANTHRACENE	0.014 or MDL	1.2 U	0.27 U	0.060 J	0.049 J 🔬 🔬	0.19 U	0.31 U	0.18 U	0.19 U
DIMETHYL PHTHALATE 2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.046 J FLUORANTHENE 50 1.2 U 0.27 U 4.2 0.40 0.19 U 0.31 U 0.18 U 0.040 J FLUORENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.065 J HEXACHLORO-1.3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLORO-1.3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.82 U 0.57 U 0.58 U 0.94 U 0.54 U 0.58 U INDENO[1,2,-CD]PYRENE 3.2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS <t< td=""><td>DIETHYL PHTHALATE</td><td>7.1</td><td>1.2 U</td><td>0.27 U</td><td>0.20 U</td><td>0.19 U</td><td>0.19 U</td><td>0.31 U</td><td>0.18 U</td><td>0.047 J</td></t<>	DIETHYL PHTHALATE	7.1	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.047 J
FLUORANTHENE 50 1.2 U 0.27 U 4.2 0.40 0.19 U 0.087 J 0.18 U 0.040 J FLUORENE 50 1.2 U 0.27 U 0.84 0.19 U 0.19 U 0.31 U 0.18 U 0.065 J HEXACHLORO-1,3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.82 U 0.57 U 0.58 U 0.94 U 0.54 U 0.54 U 0.58 U HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-N	DIMETHYL PHTHALATE	2	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.046 J
FLUORENE 50 1.2 U 0.27 U 0.84 0.19 U 0.19 U 0.11 U 0.18 U 0.065 J HEXACHLORO-1,3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U HEXACHLOROBENZENE 0.41 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE </td <td>FLUORANTHENE</td> <td>50</td> <td>1.2 U</td> <td>0.27 U</td> <td>4.2</td> <td>0.40</td> <td>0.19 U</td> <td>0.087 J</td> <td>0.18 U</td> <td>0.040 J</td>	FLUORANTHENE	50	1.2 U	0.27 U	4.2	0.40	0.19 U	0.087 J	0.18 U	0.040 J
HEXACHLORO-1,3-BUTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U HEXACHLOROBENZENE 0.41 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.82 U 0.59 U 0.57 U 0.58 U 0.94 U 0.54 U 0.58 U HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.661 J HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.667 J INDENG[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.67 J M-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U	FLUORENE	50	1.2 U	0.27 U	0.84	0.19 U	0.19 U	0.31 U	0.18 U	0.065 J
HEXACHLOROBENZENE 0.41 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.061 J HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.82 U 0.59 U 0.57 U 0.58 U 0.94 U 0.54 U 0.58 U HEXACHLOROCYCLOPENTADIENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.061 J INDENO[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.17 J 0.12 J 0.19 U 0.31 U 0.18 U 0.067 J ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J N-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19	HEXACHLORO-1,3-BUTADIENE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
HEXACHLOROCYCLOPENTADIENE NS 3.5 U 0.82 U 0.59 U 0.57 U 0.58 U 0.94 U 0.54 U 0.58 U HEXACHLOROETHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.17 J 0.12 J 0.19 U 0.31 U 0.18 U 0.19 U ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J M-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE <td>HEXACHLOROBENZENE</td> <td>0.41</td> <td>1.2 U</td> <td>0.27 U</td> <td>0.20 U</td> <td>0.19 U</td> <td>0.19 U</td> <td>0.31 U</td> <td>0.18 U</td> <td>0.061 J</td>	HEXACHLOROBENZENE	0.41	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.061 J
HEXACHLOROETHANE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.17 J 0.12 J 0.19 U 0.31 U 0.18 U 0.19 U ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J M-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.31 U 0.18 U 0.19 U NITROSODIPHENYLAMINE 1.2 U <t< td=""><td>HEXACHLOROCYCLOPENTADIENE</td><td>NS</td><td>3.5 U</td><td>0.82 U</td><td>0.59 U</td><td>0.57 U</td><td>0.58 U</td><td>0.94 U</td><td>0.54 U</td><td>0.58 U</td></t<>	HEXACHLOROCYCLOPENTADIENE	NS	3. 5 U	0.82 U	0.59 U	0.57 U	0.58 U	0.94 U	0.54 U	0.58 U
INDENO[1,2,3-CD]PYRENE 3.2 1.2 U 0.27 U 0.17 J 0.12 J 0.19 U 0.31 U 0.18 U 0.19 U ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J M-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL <td>HEXACHLOROETHANE</td> <td>NS</td> <td>1.2 U</td> <td>0.27 U</td> <td>0.20 U</td> <td>0.19 U</td> <td>0.19 U</td> <td>0.31 U</td> <td>0.18 U</td> <td>0.19 U</td>	HEXACHLOROETHANE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
ISOPHORORNE 4.4 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.067 J M-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.59 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U	INDENO[1,2,3-CD]PYRENE	3.2	1.2 U	0.27 U	0.17 J	0.12 J	0.19 U	0.31 U	0.18 U	0.19 U
M-DICHLOROBENZENE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.59 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.42 0.18 U 0.19 U NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.31 U 0.18 U	ISOPHORORNE	4.4	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.067 J
N-NITROSO-DI-N-PROPYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.59 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.59 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.19 U 0.42 0.18 U 0.19 U NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.31 U 0.18 U 0.045 J	M-DICHLOROBENZENE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIMETHYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.19 U 0.42 0.18 U 0.19 U NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PHENANTHRENE 50 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PYRENE 50	N-NITROSO-DI-N-PROPYLAMINE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.59	0.19 U
N-NITROSODIPHENYLAMINE NS 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.19 U 0.42 0.18 U 0.19 U NAPHTHALENE 0.200 or MDL 1.2 U 0.27 U 1.1 0.19 U 0.42 0.18 U 0.19 U NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PHENANTHRENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PYRENE 50 1.2 U 0.27	N-NITROSODIMETHYLAMINE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
NAPHTHALENE 13 1.2 U 0.27 U 1.1 0.19 U 0.42 0.18 U 0.19 U NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PHENANTHRENE 50 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.045 J PYRENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PYRENE 50 1.2 U 0.27 U 2.8 0.48 0.19 U 0.31 U 0.18 U 0.19 U TOTAL SVOCs	N-NITROSODIPHENYLAMINE	NS	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
NITROBENZENE 0.200 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PHENANTHRENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U PHENANTHRENE 50 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.045 J PYRENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PYRENE 50 1.2 U 0.27 U 2.8 0.48 0.19 U 0.31 U 0.18 U 0.19 U TOTAL SVOCs 500 0.73 2.4 20.35 3.005 0.373 1.154 0.711 1.023	NAPHTHALENE	13	1.2 U	0.27 U	1.1	0.19 U	0.19 U	0.42	0.18 U	0.19 U
PENTACHLOROPHENOL 1.0 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PHENANTHRENE 50 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.045 J PYRENE 50 1.2 U 0.27 U 2.8 0.48 0.19 U 0.31 U 0.18 U 0.19 U TOTAL SVOCs 500 0.73 2.4 20.35 3.005 0.373 1.154 0.711 1.023	NITROBENZENE	0.200 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
PHENANTHRENE 50 1.2 U 0.27 U 2.9 0.17 J 0.19 U 0.22 J 0.18 U 0.045 J PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.045 J PYRENE 50 1.2 U 0.27 U 0.20 U 0.19 U 0.19 U 0.31 U 0.18 U 0.19 U TOTAL SVOCs 500 0.73 2.4 20.35 3.005 0.373 1.154 0.711 1.023	PENTACHLOROPHENOL	1.0 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
PHENOL 0.03 or MDL 1.2 U 0.27 U 0.20 U 0.19 U 0.31 U 0.18 U 0.19 U PYRENE 50 1.2 U 0.27 U 2.8 0.48 0.19 U 0.31 U 0.18 U 0.19 U TOTAL SVOCs 500 0.73 2.4 20.35 3.005 0.373 1.154 0.711 1.023	PHENANTHRENE	50	1.2 U	0.27 U	2.9	0.17 J	0.19 U	0.22 J	0.18 U	0.045 J
PYRENE 50 1.2 U 0.27 U 2.8 0.48 0.19 U 0.31 U 0.18 U 0.19 U TOTAL SVOCs 500 0.73 2.4 20.35 3.005 0.373 1.154 0.711 1.023	PHENOL	0.03 or MDL	1.2 U	0.27 U	0.20 U	0.19 U	0.19 U	0.31 U	0.18 U	0.19 U
TOTAL SVOCs 500 0.73 2.4 20.35 3.005 0.373 1.154 0.711 1.023	PYRENE	50	1.2 U	0.27 U	2.8	0.48	0.19 U	0.31 U	0.18 U	0.19 U
	TOTAL SVOCs	500	0.73	2.4	20.35	3.005	0.373	1.154	0.711	1.023

U Undetectable Levels NS No Standard

MDL Method Detection Limit

1.78° 8.1

dia dia 1

ķ

Damp Date Sale Diat Number of the set of the se	Location	Recommanded	PC-P-1	PC-P-1	PC_P_1	PC-P-2	PC-P-3	PC-P-3	PC-PD-1	PC-PD-1
Dampin Di Che-Di C-P-Di CP-Di		Seil	11/22/2000	11/22/2000	11/20/2000	11/30/2000	11/22/2000	11/22/2000	11/01/2000	11/21/2000
Stappin Depth Objective 24* Dir V Performation P		Cleanum	BC D 01	PC P 01	PC P 02	PG P 02	PC P 01	PC P 03	PC PD 01	PC PD A1
Sample might Description Park of Park	Sample ID	Cleanup	PG-P-01	PG-P-01	PG-P-02	r0-r-02	PG-P-03	rG-r-03	PG-PD-01	10-10-01
Concentration Proc. No. 1_2_2 No. <	Sample Depth	Objective	2-4	8-10	2-4	4-0	2-4	0-8 NGWC	2-4	10-12
Light Relation Construction Light Relation Construction Light Relation Construction Light Relation Construction Light Relation Light Relation Construction S 0.11	Concentration	MG/KG	MU/KU	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MU/KU	MG/KG
Label Process Construction 2 = Constres <td< td=""><td>1,2,4-TRICHLUROBENZENE</td><td>3.4</td><td>0.18 U</td><td>0.21 0</td><td>0.19 U</td><td>0.20 0</td><td>0.190</td><td>0.190</td><td>0.19 0</td><td>0.21 0</td></td<>	1,2,4-TRICHLUROBENZENE	3.4	0.18 U	0.21 0	0.19 U	0.20 0	0.190	0.190	0.19 0	0.21 0
Label Charlow Construction Discrete Discrete <thdiscre< th=""> Discrete <thdiscret< td=""><td>1,2-BENZPHENANTHRACENE</td><td>NS 7.0</td><td>0.18 U</td><td>0.063 3</td><td>0.190</td><td>0.20 0</td><td>0.19 0</td><td>0.19 0</td><td>0.19 0</td><td>0.21 0</td></thdiscret<></thdiscre<>	1,2-BENZPHENANTHRACENE	NS 7.0	0.18 U	0.063 3	0.190	0.20 0	0.19 0	0.19 0	0.19 0	0.21 0
L3.DHTMULTURZZINE NS 008/L	1,2-DICHLOROBENZENE	7.9	0.18 0	0.21 0	0.19 0	0.20 0	0.190	0.190	0.19 0	0.210
LADIC MODERNATIVE C.3. DIEU CUIU CUIU DUEU DUEU <thdueu< th=""> DUEU DUEU<td>1,2-DIPHENYLHYDRAZINE</td><td>NS D E</td><td>0.036 0</td><td>0.042 0</td><td>0.039 0</td><td>0.040 0</td><td>0.038 0</td><td>0.038 U</td><td>0.0370</td><td>0.042 0</td></thdueu<>	1,2-DIPHENYLHYDRAZINE	NS D E	0.036 0	0.042 0	0.039 0	0.040 0	0.038 0	0.038 U	0.0370	0.042 0
LALFRAINGONPRENUL 0.1 <th0.1< th=""> 0.1 <th0.1< th=""></th0.1<></th0.1<>	1,4-DICHLOROBENZENE	8.5	0.18 U	0.21 U	0.19 0	0.20 U	0.19 U	0.19 0	0.19 0	0.21 0
LADIA RUNOWIEWOL D.N D.18 U D.21 U D.19 U D.11 U <thd.11 th="" u<=""> <thd.11 th="" u<=""> <thd.1< td=""><td>2,4,6-TRICHLORORPHENOL</td><td>0.1</td><td>0.18 U</td><td>0.21 0</td><td>0.19 U</td><td>0.20 U</td><td>0.190</td><td>0.19 0</td><td>0.190</td><td>0.21 U</td></thd.1<></thd.11></thd.11>	2,4,6-TRICHLORORPHENOL	0.1	0.18 U	0.21 0	0.19 U	0.20 U	0.190	0.19 0	0.190	0.21 U
24-DBM 0.200 m130 0.210 0.200 0.190 0.190 0.190 0.190 0.210 24-DBM NORMENOL 0.200 m140 0.210 0.190 0.210 0.190 0.210 0.190 0.210 0.190 0.210 0.190 0.210 0.190 0.210 0.190 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100 0.100	2.4-DICHLOROPHENOL	0.4	0.18 0	0.21 0	0.19 0	0.20 0	0.19 0	0.190	0.19 0	0.21 0
LADIMINOPLENUL D.200 PMC D340 D420 D340 D3410 D3410 <thd3410< th=""> <th< td=""><td>2,4-DIMETRYLPHENOL</td><td></td><td>0.18 0</td><td>0.21 0</td><td>0.190</td><td>0.20 U</td><td>0.19 0</td><td>0.19 0</td><td>0.19 0</td><td>0.21 U</td></th<></thd3410<>	2,4-DIMETRYLPHENOL		0.18 0	0.21 0	0.190	0.20 U	0.19 0	0.19 0	0.19 0	0.21 U
LADIMIROULLERE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.11 U <th0.11 th="" u<=""> 0.11 U <th0.11< td=""><td>2,4-DINITROPHENOL</td><td>0.200 BF MDL</td><td>0.36 0</td><td>0.42 0</td><td>0.39 U</td><td>0.40 0</td><td>0.38 0</td><td>0.38 0</td><td>0.37 0</td><td>0.42 0</td></th0.11<></th0.11>	2,4-DINITROPHENOL	0.200 BF MDL	0.36 0	0.42 0	0.39 U	0.40 0	0.38 0	0.38 0	0.37 0	0.42 0
LADIMINOLOLISME 1 0.18 U 0.19 U 0.11 U 2-CHLORONAPHINOL 0.38 U 0.11 U 0.19 U 0.11 U 2-CHLORONAPHINOL 0.38 O 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.11 U 3-DICHLOROBENZIDINE NAA 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.11 U 4-BROMOPHENYLPHENYL ETHER NS 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U 4-ETIROPHIENTUL 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U 4-ETIROPHIENTUL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U 4-ETIROPHIENTUL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U	2,4-DINITROTOLUENE	1	0.18 U	0.21 0	0.19 U	0.20 U	0.190	0.190	0.19 0	0.21 0
24.11.0X08NAMH INALESE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.21 U 2ATIROMENOL 0.30 O MOL 0.18 U 0.21 U 0.19 U 0.11 U 4.20 U 0.19 U 0.19 U 0.19 U 0.11 U 4.20 U 0.19 U 0.19 U 0.11 U 4.21 U 0.19 U 0.19 U 0.11 U 4.21 U 4.11 U 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.21 U 4.11 U 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.21 U 4.11 U 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U 4.21 U 0.19 U 0.19 U 0.19 U 0.11 U	2,6-DINITROTOLUENE	1	0.18 U	0.21 0	0.19 0	0.20 U	0.19 0	0.190	0.190	0.21 U
DATE OF THE CONSTRUCT DATE DATE OF THE CONSTRUCT DATE OF THE CONSTRUCT <thdate construct<="" of="" th="" the=""> <thdate co<="" of="" td="" the=""><td>2-CHLOROKNAPHTHALENE</td><td><u>N5</u></td><td>0.18 U</td><td>0.21 0</td><td>0.19 0</td><td>0.20 0</td><td>0.19 0</td><td>0.190</td><td>0.19 0</td><td>0.21 0</td></thdate></thdate>	2-CHLOROKNAPHTHALENE	<u>N5</u>	0.18 U	0.21 0	0.19 0	0.20 0	0.19 0	0.190	0.19 0	0.21 0
Description Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	2-CHLOROPHENOL	0.0	0.18 0	0.21 U	0.19 0	0.20 0	0.19 0	0.190	0.19 0	0.21 U
J.S.D.CLOROBENZIONE N/A DIS U DIS U <thdis th="" u<=""> DIS U <thdis th="" u<=""></thdis></thdis>	2-NITROPHENOL		0.18 U	0.21 0	0.19 0	0.20 0	0.190	0.190	0.19 U	0.21 U
Generative-or-CRESOL. 0.100 0.100 0.100 0.100 0.100 0.100 0.1010 0.1010 0.021 ABCMOORHENVLPHENVL 0.180 0.21 0.19 0.20 0.19 0.19 0.20 0.19 0.19 0.21 0.19 0.19 0.21 0.19 0.19 0.21 0.19 0.21 0.19 0.19 0.21 0.19 0.21 0.19 0.21 0.19 0.19 0.21 0.19 0.19 0.19 0.21 0.19 0.1	13.3-DICHLOROBENZIDINE		0.18 U	0.21 U	0.19 U	0.20 U	0.19 0	0.190	0.19.0	0.21 0
*#ARXMONTARY LETHER NS Dis 0 0.41 U 0.42 U <th< td=""><td>4,0-DINTIRU-U-CRESUL</td><td></td><td>0.18 U</td><td>0.21 U</td><td>0.190</td><td>0.20 0</td><td>0.19 0</td><td>0.19 0</td><td>0.19 0</td><td>0.21 U</td></th<>	4,0-DINTIRU-U-CRESUL		0.18 U	0.21 U	0.190	0.20 0	0.19 0	0.19 0	0.19 0	0.21 U
CHLOROPHENT/LIFENCY Disc Disc <thdisc< th=""> Disc <thdisc< th=""> Disc <thdisc< th=""></thdisc<></thdisc<></thdisc<>	4-BRUMOPHENYLPHENYLEIHER		0.18 U	0.21 U	0.19.0	0.20 0	0.19 0	0.19 U	0.19.0	0.21 U
CALONATION LETTERY CS 0.21 U 0.19 U 0.11 U 0.12 U 0.19 U 0.19 U 0.11 U <th< td=""><td>4-CHLORO-3-METHYLPHENOL</td><td></td><td>0.18 U</td><td>0.21 U</td><td>0.190</td><td>0.20 0</td><td>0.19 0</td><td>0.19.0</td><td>0.190</td><td>0.21 U</td></th<>	4-CHLORO-3-METHYLPHENOL		0.18 U	0.21 U	0.190	0.20 0	0.19 0	0.19.0	0.190	0.21 U
NUMENTIAL 0.10 U 0.17	4-UNUTROBUENOU		0.10 U	0.21 0	0.190	0.20 0	0.19 0	0.19.0	0.19 0	0.21 U
ALE-INVEDICE OC OLD OLD <th< td=""><td></td><td>5.100 01 1101</td><td>0.10 U</td><td>0.21 0</td><td>0.19.0</td><td>0.004 1</td><td>0.19 0</td><td>0.19 0</td><td>0 19 11</td><td>0.21 U</td></th<>		5.100 01 1101	0.10 U	0.21 0	0.19.0	0.004 1	0.19 0	0.19 0	0 19 11	0.21 U
ALE NATH AFLENE Y1 018 U 021 U 019 U 019 U 019 U 021 U BATTRACENE 50 0.16 U 0.21 U 0.19 U 0.13 J 0.19 U 0.21 U BENZOJAJ PYRENE 0.024 J 0.043 J 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U BENZOJAJ PYRENE 0.024 J 0.095 J 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BENZOJAJ PYRENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BENZOJAJPERTNENE 1.1 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BENZOJAJPERTNENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BENZOJAJPERTNENE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.21 U BENZOJAJPERTNENE	ACENAPHTHENE	41	0.18 U	0.21 0	0.190	0.094 J	0.19 U	0.190	0.19 0	0.21 U
ANTINACENE 33 018 U 021 U 019 U 019 U 019 U 019 U 021 U BENZIDINE NS 0.36 U 0.42 U 0.39 U 0.40 U 0.38 U 0.38 U 0.31 U 0.42 U BENZOJAJANTIRACENE 0.224 or MDL 0.18 U 0.064 J 0.19 U 0.21 U 0.19 U 0.19 U 0.19 U 0.10 U 0.10 U 0.10 U 0.19 U	ALENAPHINILENE	50	0.18 U	0.21 U	0.190	0.20 0	0.190	0.190	0.190	0.21 U
BERZIGLANTE NS 0.38 U 0.19 U 0.21 U BENZOGIANJPERTHER NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21	ANTHRACENE	NC	0.18 U	0.21 0	0.190	0.13 J	0.190	0.19 0	0.190	0.210
BER.COL(A)RTREACENCE 0.124 OF MOL 0.18 U 0.083 J 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U BER.COL(A)PTENE 1.1 0.044 J 0.095 J 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U BER.COL(A)PTERVLENE 50 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U BER.COL(A)PTERVLENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U BER.COL(A)RORTHOXYMETHARE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U BIS2-CHLOROISOPROPYL)ETHER NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BIS2-CHLOROISOPROPYL)ETHER NS 0.18 U 0.21 U 0.19 U 0.22 U 0.19 U 0.19 U 0.21 U BIS2-CHLOROISOPROPYL)ETHER NS 0.18 U 0.21 U 0.19 U 0.22 U 0.19	BENZIDINE		0.30 0	0.42 0	0.390	0.40 0	0.38 0	0.38 0	0.37 0	0.42 0
BENCOURANTHENE 0.001 0.13 0.001 0.13 0.001 0.13 0.001 0.13 0.001 0.13 0.001 0.13 0.001 0.13 0.001 0.13 0.001 0.13 0.01 0.13 0.01 0.13 0.	BENZOLAJANTHRACENE	0.224 01 MDL	0.18.1	0.043 J	0.19 0	0.044 J	0.190	0.190	0.190	0.21 U
BENZOGI, IPEDURANTIENE 1.1.2 0.049.7 0.059.7 0.19 0.20 0.19 0.19 0.21 0 BENZOGI, IPERVLENE 50 0.18 0.21 0.19 0.20 0.19 0.19 0.21 0.19 0.21 0.19 0.19 0.19 0.19 0.21 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.21 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.19 0.21 0.21 0.19 0.19 0.19 0.19 0.19 0.21 0.21 0.19 0.20 0.19 0.19 0.21 0.21 0.21 0.19 0.19 0.19 0.21 0.21 0.19 0.21 0.21 0.19 0.21 0.19 0.19 0.21 0.19 0.19 0.19 0.19	BENZOLAJPYRENE	1 1	0.18 0	0.000 J	0.19 0	0.20 U	0.191	0.190	0.190	0.21 U
BENCOUCH, IPERLENC Dist Dist <thdist< th=""> Dist Dist</thdist<>	DENZO(B)FLOURANTHENE	50	0.044 J	0.093 1	0.190	0.20 0	0.19.0	0.190	0.190	0.21 U
BENZURJENCIANTRENE 2.12 0.18 U 0.21 U 0.19 U 0.11 U <	DENZOUGH, IJPERT LENE	1 1	0.18 U	0.21 0	0.1917	0.20 U	0.19.0	0.190	0.19 U	0.210
BEACTLESTICTUSTION OLIGU OLIGU <td>DENZYL DUTYL BUTHALATE</td> <td>50</td> <td>0.18 U</td> <td>0.21 U</td> <td>0.191</td> <td>0.20 0</td> <td>0.191</td> <td>0.19.0</td> <td>0.19.0</td> <td>0.21 U</td>	DENZYL DUTYL BUTHALATE	50	0.18 U	0.21 U	0.191	0.20 0	0.191	0.19.0	0.19.0	0.21 U
BIS(2-CHLOROETHYLETHER NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BIS(2-CHLOROETHYLETHER NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BIS(2-CHLOROETHYLETHER NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U BIS(2-CHLOROETHYLETHER NS 0.18 U 0.21 U 0.064 J 0.080 J 0.41 B 0.38 B 0.086 J B 0.22 U DIN-BUTYL PHTHALATE 50 0.18 U 0.060 J 0.064 J 0.089 J 0.058 J 0.19 U 0.21 U DIBENZ(A,HJANTHRACENE 0.014 or MDL 0.18 U 0.21 U 0.19 U 0.21 U 0.19 U 0.19 U 0.21 U DIMETYL PHTHALATE 7.1 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PLUORANTHENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U HEXACHLOROETAJENE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U <td< td=""><td>BENZTE BUTTE PHTHALATE</td><td>NC</td><td>0.18 0</td><td>0.21 0</td><td>0.190</td><td>0.20 0</td><td>0.19.0</td><td>0.190</td><td>0.190</td><td>0.210</td></td<>	BENZTE BUTTE PHTHALATE	NC	0.18 0	0.21 0	0.190	0.20 0	0.19.0	0.190	0.190	0.210
BIS(2-CHLOROEINTL)ETHER NS 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U BIS(2-CHLOROEINTL)ETHER NS 0.18 U 0.21 U 0.080 J 0.41 B 0.38 B 0.086 JB 0.22 U 0.19 U 0.19 U 0.19 U 0.21 U BIS(2-CHLOROEINTL)ETHER NS 0.18 U 0.030 J 0.064 J 0.080 J 0.058 J 0.19 U 0.19 U 0.21 U DI-N-OCTYL PHTHALATE 50 0.18 U 0.020 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U DIBENZ(A,HJANTHRACENE 0.014 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U DIMETHYL PHTHALATE 7.1 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U DIMETHYL PHTHALATE 7.1 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U 0.19 U 0.19 U 0.21 U FLUORANTHENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.	BIS(2-CHLOROETHOXT)METHANE	NS	0.18 U	0.21 U	0.191	0.20 U	0.190	0.190	0.19 U	0.21 U
BIS(2-ETI/HALATE 50 0.34 B 0.32 B 0.064 J 0.030 J 0.41 B 0.38 B 0.086 JB 0.22 B DI-N-BUTYL PHTHALATE 50 0.34 B 0.032 B 0.064 J 0.080 J 0.41 B 0.38 B 0.086 JB 0.22 B DI-N-BUTYL PHTHALATE 50 0.18 U 0.086 J 0.040 J 0.064 J 0.080 J 0.41 B 0.38 B 0.086 JB 0.22 U DI-N-CTYL PHTHALATE 50 0.18 U 0.02 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U DIENYL, PHTHALATE 7.1 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U DIENTYL, PHTHALATE 7.1 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PLUORANTHENE 50 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U HEXACHLOROELZENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U HEXACHLOROETHANE	BIS(2-CHLOROETHTE)ETHER	NS	0.181	0.21 0	0.190	0.20 0	0.19.0	0.191/	0.190	0.21 0
Disperimentation Disperimentation <thdisperimentation< th=""> <thdisperimentation< t<="" td=""><td>DIS(2 ETHVIEVVI) DUTUALATE</td><td>50</td><td>0.18 U</td><td>0.210</td><td>0.13 0</td><td>0.200</td><td>0.190</td><td>0.190</td><td>0.096 /P</td><td>0.21 C</td></thdisperimentation<></thdisperimentation<>	DIS(2 ETHVIEVVI) DUTUALATE	50	0.18 U	0.210	0.13 0	0.200	0.190	0.190	0.096 /P	0.21 C
Dirk-OCTYL Dirk-OCTYL <thdirk-octyl< th=""> Dirk-OCTYL Dirk-OCT</thdirk-octyl<>	DIN DUTYL BUTHALATE	81	0.34 B	0.52 6	0.1011	0.030 J	0.052 IP	0.10.1	0.080 15	0.22 B
DIRENCIAL Dire	DLN-OCTVI BUTHALATE	50	0.1811	0.086 1	0.130	0.042 3	0.089 1	0.058 1	0.19.0	0.210
DIDELIGATION District	DIBENZIA HIANTHRACENE	0.014 or MDI	0.1811	0.21.11	0.1911	0.2011	0 19 11	0.0901	0.19.0	0.21 U
DILTITE TIMITE TIMITE ATTE Production Producin	DISENVI PHTHALATE	7.1	0.1811	0.21 12	0.1911	0.2011	0.1911	0.19.0	0.1911	0.21 U
DIALTIC INTECTINE ATE C OTO OTO <thoto< th=""> OTO <thoto< th=""></thoto<></thoto<>		2	0.1811	0.21 U	0.1911	0.20 0	0.19.11	0.1911	0.1911	0.21 U
1100001111102 0100 0110 </td <td>FLUORANTHENE</td> <td>50</td> <td>0 18 11</td> <td>0.21.11</td> <td>0 1911</td> <td>0.23</td> <td>0 19 11</td> <td>0 19 11</td> <td>0 19 11</td> <td>0.2111</td>	FLUORANTHENE	50	0 18 11	0.21.11	0 1911	0.23	0 19 11	0 19 11	0 19 11	0.2111
International and the set of the	FLUORENE	50	0 18 11	0.2111	0 19 11	0.10 1	0 19 11	0 19 1	0 19 U	0.21 U
Instructions Instruction	HEXACHLORO-L 3-BUTADIENE	NS	0.18 U	0.21 U	0.19 U	0.20 U	0.191/	0.19 U	0.19 U	0.21 U
Instruction	HEXACHLOROBENZENE	0,41	0.18 U	0.21 U	0.19 U	0.20 U	0.1911	0.191/	0.19 U	0.2111
International Constraint Interna	HEXACHLOROCYCLOPENTADIENE	NS	0.54 U	0.63 U	0.58 U	0.60 U	0.57 U	0.57 U	0.56 U	0.63 U
INDENDITION IND IND <th< td=""><td>HEXACHLOROFTHANE</td><td>NS</td><td>0.18.U</td><td>0.21 U</td><td>0.19 U</td><td>0.20 U</td><td>0.19 U</td><td>0.19 U</td><td>0.1911</td><td>0.21 U</td></th<>	HEXACHLOROFTHANE	NS	0.18.U	0.21 U	0.19 U	0.20 U	0.19 U	0.19 U	0.1911	0.21 U
INCLUE OF FUNCTION INCOLOR	INDENO[1 2.3-CD]PYRENE	3.2	0.18 U	0.21 U	0.19 U	0.20 U	0.19 U	0.19 U	0.19 U	0.21 U
INDICATION IND	ISOPHORORNE	4.4	0 18 11	0.2111	0 19 11	0 20 11	0 1911	0 19 11	0 19 11	0.21.0
N-NITROSO-DI-N-PROPYLAMINE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U N-NITROSO-DI-N-PROPYLAMINE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U N-NITROSODIMETHYLAMINE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U N-NITROSODIPHENYLAMINE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NAPHTHALENE 13 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NTROBENZENE 0.200 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19	M-DICHLOROBENZENE	NS	0.18 U	0.21 U	0.19 U	0.20 U	0.19 U	0.19.U	0.191	0.21 U
NNITROSODIMETHYLAMINE NS 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U N-NITROSODIMETHYLAMINE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U N-NITROSODIPHENYLAMINE NS 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NAPHTHALENE 13 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NTROBENZENE 0.200 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U <td< td=""><td>N-NITROSO-DI-N-PROPYLAMINE</td><td>NS</td><td>0.18 U</td><td>0.21 U</td><td>0.1911</td><td>0.20 U</td><td>0.19 U</td><td>0.19 U</td><td>0.191/</td><td>0.2111</td></td<>	N-NITROSO-DI-N-PROPYLAMINE	NS	0.18 U	0.21 U	0.1911	0.20 U	0.19 U	0.19 U	0.191/	0.2111
NNTIROSODIPHENYLAMINE NS 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U NAPHTHALENE 13 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NAPHTHALENE 13 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NITROBENZENE 0.200 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U PYRENE <	N-NITROSODIMETHYLAMINE	NS	0.181	0.21 U	0.191/	0.20 U	0.1911	0.191/	0.1911	0.21 U
NAPHTHALENE 13 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U NITROBENZENE 0.200 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PYRENE 50 0.18 U 0.21 U 0.19 U 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U PYRENE 50 0.18 U 0.058 J 0.19 U 0.16 J 0.19 U 0.19 U 0.21 U	N-NITROSODIPHENYI AMINE	NS	0.18 U	0.21 U	0.1911	0.20 U	0.19 U	0.19 U	0.19 U	0.21 U
NTROBENZENE 0.200 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PYRENE 50 0.18 U 0.058 J 0.19 U 0.16 J 0.19 U 0.19 U 0.19 U 0.21 U TOTAL SVOCs 500 0.422 0.775 0.104 1.354 0.551 0.438 0.086 0.22	NAPHTHALENE	13	0.18 U	0.21 U	0.19 U	0.20 U	0.19 U	0.19 U	0.19 U	0.21 U
PENTACHLOROPHENOL 1.0 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PYRENE 50 0.18 U 0.058 J 0.19 U 0.16 J 0.19 U 0.19 U 0.21 U TOTAL SVOCs 500 0.422 0.775 0.104 1.354 0.551 0.438 0.086 0.22	NITROBENZENE	0.200 or MDL	0.18 U	0.21 U	0.19 U	0.20 U	0.19 U	0.19 U	0.19 U	0.21 U
PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.19 U 0.21 U PHENANTHRENE 50 0.18 U 0.21 U 0.19 U 0.41 0.19 U 0.19 U 0.19 U 0.19 U 0.21 U PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.21 U PYRENE 50 0.18 U 0.058 J 0.19 U 0.16 J 0.19 U 0.19 U 0.21 U TOTAL SVOCs 500 0.422 0.775 0.104 1.354 0.551 0.438 0.086 0.22	PENTACHLOROPHENOL	1.0 or MDL	0.18 U	0.21 U	0.19 U	0.20 U	0.19 U	0.19 U	0.19 U	0.21 U
PHENOL 0.03 or MDL 0.18 U 0.21 U 0.19 U 0.20 U 0.19 U 0.19 U 0.19 U 0.21 U PYRENE 50 0.18 U 0.058 J 0.19 U 0.16 J 0.19 U 0.19 U 0.19 U 0.21 U TOTAL SVOCs 500 0.422 0.775 0.104 1.354 0.551 0.438 0.086 0.22	PHENANTHRENE	50	0.18 U	0.21 U	0.19 U	0.41	0.19 U	0.19 U	0.19 U	0.21 U
PYRENE 50 0.18 U 0.058 J 0.19 U 0.16 J 0.19 U 0.19 U 0.19 U 0.21 U TOTAL SVOCs 500 0.422 0.775 0.104 1.354 0.551 0.438 0.086 0.22 U Undetectable Levels 0.755 0.104 1.354 0.551 0.438 0.086 0.22	PHENOL	0.03 or MDL	0.18 U	0.21 U	0.19 U	0.20 U	0.19 U	0.19 U	0.19 U	0.21 U
TOTAL SVOCs 500 0.422 0.775 0.104 1.354 0.551 0.438 0.086 0.22 U Undetectable Levels 0 0.775 0.104 1.354 0.551 0.438 0.086 0.22	PYRENE	50	0.18 U	0.058 J	0.19 U	0.16 J	0.19 U	0.19 U	0.19 U	0.21 U
U Undetectable Levels	TOTAL SVOCs	500	0.422	0.775	0.104	1.354	0.551	0.438	0.086	0.22
	U Undetectable Levels			1	<u></u>	<u>L</u>	<u>.</u>	•		

U Undetectable Lev NS No Standard

MDL Method Detection Limit

ţ.

200/×-4-

ŝ

F	Deserves				DC DD C			
Location	Recommended	FG-FD-3	rG-rD-4	rG-rD-3	rG-rD-5	rG-rA-MW-/	PG-PA-MW-/	rG-PA-MW-7
Sample Date	501	11/29/2000	12/2/2000	12/2/2000	12/2/2000	11/13/2000	11/13/2000	11/13/2000
Sample ID	Cleanup	PG-PD-3	PG-PD-4	PG-PD-S	PG-PD-5	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07
Sample Depth	Objective	4-6	8-10	0.7-1.5	2-4	2.5-4	4-6'	6-8
	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,2,4-TRICHLOROBENZENE	3.4	0.21 0	0.20 U	0.19 0	0.19 0	0.190	0.190	0.20 0
1,2-BENZPHENAN IHRACENE	NS	0.21 U	0.20 0	0.15 J	0.190	0.19 0	0.190	0.20 0
1,2-DICHLOROBENZENE	7.9	0.21 0	0.20 0	0.19 0	0.19 0	0.19 0	0.19 0	0.20 0
1,2-DIFHENT LHT DRAZINE	85	0.041 0	0.041 0	0.038 0	0.0370	0.0370	0.038 0	0.040 0
2.4.6 TRICHLOROBENZENE	0.5	0.21 0	0.20 0	0.19 U	0.190	0.19 0	0.19 0	0.20 0
2.4.0-TRICHEORORFHENOL	0.1	0.210	0.20 0	0.19 U	0.190	0.19.0	0.19 0	0.20 0
2.4-DIMETHVI PHENOI	NS	0.2112	0.2011	0.19.0	0.19 0	0.19.0	0.19.0	0.20 0
2.4-DIMETHTERNOL	0 200 or MDI	0.21 0	0.20 0	0.19 0	0.17 []	0.130	0.19 0	0.200
	NS	0.2111	0.2011	0.1011	0.1911	0.970	0.58 0	0.000
2.6-DINITROTOLUENE	1	0.21 0	0.20 0	0.191	0.1911	0.191	0.190	0.20 U
2-CHLORORNAPHTHALENE	NS	0 21 11	0.20 0	0.191	0.1911	0.191	0.19.0	0.200
2-CHLOROPHENOI	0.8	0.21.0	0 20 11	0 19 11	0 19 11	0 19 11	0.19.0	0.2011
2-NITROPHENOL	0.330 or MDL	0.21 U	0.20 U	0.1911	0.1911	0.1911	0 19 11	0.2011
3 3'-DICHLOROBENZIDINE	N/A	0.21 U	0 20 11	0 19 11	0 19 11	0 19 U	0 19 U	0 20 U
4.6-DINITRO-O-CRESOL	0.100 or MDL	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
4-BROMOPHENYLPHENYL ETHER	NS	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
4-CHLORO-3-METHYLPHENOL	0.240 or MDL	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
4-CHLOROPHENYLPHENYL ETHER	NS	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
4-NITROPHENOL	0.100 or MDL	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
ACENAPHTHENE	50	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
ACENAPHTHYLENE	41	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
ANTHRACENE	50	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
BENZIDINE	NS	0.41 U	0.41 U	0.38 U	0.37 U	0.37 U	0.38 U	0.40 U
BENZO[A]ANTHRACENE	0.224 or MDL	0.21 U	0.20 U	0.12 J	0.19 U	0.19 U	0.19 U	0.20 U
BENZO[A]PYRENE	0.061 or MDL	0.21 U	0.20 U	0.14 J	0.19 U	0.19 U	0.19 U	0.20 U
BENZO[B]FLOURANTHENE	1.1	0.21 U	0.20 U	0.28	0.19 U	0.19 U	0.19 U	0.20 U
BENZO[G,H,I]PERYLENE	50	0.21 U	0.20 U	0.050 J	0.19 U	0.19 U	0.19 U	0.20 U
BENZO[K]FLOURANTHENE	1.1	0.21 U	0.20 U	0.077 J	0.19 U	0.19 U	0.19 U	0.20 U
BENZYL BUTYL PHTHALATE	50	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
BIS(2-CHLOROETHOXY)METHANE	NS	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
BIS(2-CHLOROETHYL)ETHER	NS	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
BIS(2-ETHYHEXYL)PHTHALATE	50	0.23 B	0.11 JB	0.19 B	0.12 JB	0.045 J	0.063 J	0.082 J
DI-N-BUTYL PHTHALATE	8.1	0.21 U	0.093 J	0.075 J	0.19 U	L 0600	0.19 U	0.068 J
DI-N-OCTYL PHTHALATE	50	0.21 U	0.075 JB	0.20 B	0.085 JB	0.039 J	0.046 J	0.057 J
DIBENZ[A,H]ANTHRACENE	0.014 or MDL	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
DIETHYL PHTHALATE	/.1	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
DIMETHYL PHTHALATE	2	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
FLUORANTHENE	50	0.21 U	0.20 U	0.17 J	0.19 U	0.19 U	0.19 U	0.20 U
FLUORENE	50	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
HEXACHLORO-1,3-BUTADIENE	NS	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
HEXACHLOROBENZENE	0.41	0.21 U	0.20 0	0.19 U	0.19 U	0.19 U	0.19 U	0.20 U
HEXACHLOROCYCLOPENTADIENE	NS	0.62 U	0.61 U	0.57 U	0.56 U	0.56 U	0.57 U	0.60 U
HEXACHLOROETHANE	NS 2.2	0.21 U	0.20 0	0.19 U	0.19 U	0.19 0	0.19 U	0.20 0
INDENO[1,2,3-CD]PYRENE	3.2	0.21 U	0.20 U	0.049 J	0.19 U	0.19 U	0.19 U	0.20 U
ISOPHORORNE	4.4	0.21 U	0.20 U	0.19 U	0.19 U	0.19 U	0.19 U	0.20 0
M-DICHLOROBENZENE	NS NS	0.21 U	0.20 0	0.19 U	0.19 0	0.19 0	0.19 U	0.20 U
N-NITKOSO-DI-N-PROPYLAMINE	NS NS	0.21 U	0.20 0	0.19 0	0.19 0	0.19 0	0.19 U	0.20 U
N NITROSODIBLENVI ANDE	NS	0.21 0	0.20 0	0.19 0	0.19 0	0.19 0	0.19 0	0.20 0
IN-INITRUSODIPHENYLAMINE	13	0.21 0	0.20 0	0.190	0.19 0	0.19 U	0.190	0.20 0
NATHIHALENE		0.21 U	0.20 0	0.046 J	0.19 U	0.190	0.19 0	0.20 U
NITROBENZENE		0.21 0	0.20 0	0.190	0.190	0.190	0.19 0	0.20 0
PENTACHLOROPHENOL	1.0 01 PIDE	0.21 0	0.20 0	0.190	0.190	0.19 0	0.19 0	0.20 0
DUENOI		0.21 0	0.20 0	0.090 1	0.190	0.19 0	0.19 0	0.20 U
DVDENE	50	0.21.0	0.20 0	0.15 0	0.19.0	0.19.0	0.19.0	0.2011
	500	0.21 0	0.200	1 905	0.19 0	0.174	0.190	0.207
LIOIAL SYDES		0.25	10.410	11.003	10.203	10.1.74	10.103	10.201

U Undetectable Levels

ş

🕐 hasila 🖛 innatitiratikkan

NS No Standard

MDL Method Detection Limit

2

64000 400

三次教

Location	Recommended	PG-PA-MW-7	PG-PA-MW-10D	PG-PA-MW-10D	PG-PA-MW10D	PG-PA-MW10D	PG-01-1
Sample Date	Sail	11/13/2000	11/27/2000	11/27/2000	11/07/2000	11/77/2000	11/30/2000
	Channen	DC DAMW 07	BC BAMWIOD	PC PAMWIOD	PC MW10D	PC MW10D	PG 01 1
	Cleanup	PU-FAMW-07	PG-PAMWIOD	A CI	7 0 ¹	PG-MW10D	2.4
Sample Depth	MC/KC	MCMC	MC/VC	MORG	MCMC	MCAC	MG/KG
L 2 4 TRICHLORODENZENE	14	0 10 11	0 10 11	0.1011	0.22.11	12.11	0.101
1,2,4-TRICHLOROBENZENE	5.4 NS	0.191	0.190	0.19 0	6.25 0	13 U	0.19 0
1,2-BENZPHENANTHRACENE	7.0	0.190	0.12.3	0.55	0.2	13.0	0.49
1,2-DICHLOROBENZENE	1.5	0.190	0.190	0.190	0.23 0		0.190
1,2-DIPHENYLHYDRAZINE	85	0.038 0	0.037 0	0.038 0	0.046 0	2.6 U	0.038 0
1,4-DICHLOROBENZENE	0.5	0.190	0.190	0.190	0.23 U	13 U	0.190
2,4,6-TRICHLORORPHENOL	0.1	0.190	0.190	0.190	0.23 U	13 U	0.190
2,4-DICHLOROPHENOL	0.4	0.19 U	0.190	0.19 U	0.23 0	13 U	0.19 U
2,4-DIMETHYLPHENOL	NS 0.200 at MDI	0.190	0.190	0.19 U	0.23 0	13 0	0.19 0
2,4-DINITROPHENOL	0.200 OF MDL	0.38 U	0.37 U	0.38 U	0.46 0	26 0	0.38 U
2,4-DINITROTOLUENE	NS	0.190	0.19 0	0.19 U	0.23 0	13 0	0.19.0
2,6-DINITROTOLUENE	1	0.19 0	0.190	0.19 0	0.23 0	13 0	0.190
2-CHLORORNAPHTHALENE	NS 0.9	0.19 U	0.19 U	0.190	0.23 0	13 0	0.19 0
2-CHLOROPHENOL	0.0	0.190	0.19 U	0.19 0	0.23 0	13.0	0.190
2-NITROPHENOL		0.19.0	0.19 0	0.19 0	0.23 U	13 U	0.19 U
3.5-DICHLOROBENZIDINE		0.19 0	0.190	0.19 0	0.23 U	130	0.190
4,0-DINITRO-O-CRESOL		0.190	0.19 U	0.19 U	0.23 U	12 U	0.19 U
4-BROMOPHENYLPHENYL ETHER		0.190	0.19 U	0.19 0	0.23 U	U 51	0.19 0
4-CHLORO-3-METHYLPHENOL	U.240 OF MUL	0.191	0.19 0	0.19 0	0.23 U	130	0.19 0
4-CHLOROPHENYLPHENYL EIHER	NS 0.100 at MDI	0.190	0.19 U	0.190	0.23 U	13 0	0.190
4-NITROPHENOL	0.100 01 MDL	0.19 0	0.19 U	0.190	0.23 0	13 0	0.190
ACENAPHTHENE	50	0.190	0.19 U	0.190	0.23	13.0	0.0693
ACENAPHTHYLENE	41	0.190	0.19.0	0.190	0.64	2.6 J	0.095 J
ANTHRACENE	50	0,19 U	0.19 0	0.096 J	1.2	13.0	0.20
BENZIDINE	NS 0.224 or MDL	0.38 (0.37 0	0.38 0	0.46 U	26 0	0.38 0
BENZO[A]ANTHRACENE	0.224 OF MOL	0.190	0.10 J	0,36	4.4. Barren 6. S	13.0	0.49
BENZO[A]PYRENE	0.061 OF MDL	0.19 0	0, H, J. 2 40. 37 1. 19	0.32	4.3	13.0	0.46
BENZO(B)FLOURANTHENE	1.1	0.190	0.13 J	0.50	7.9	13 0	0.55
BENZOLG,H,IJPERYLENE	50	0.19 0	0.081 J	0.13 3	2.5	13.0	0.17 J
BENZO[K]FLOURANTHENE	1.1	0.190	0.038 J	0.14 J	2.92	13.0	0.30
BENZYL BUTYL PHTHALATE	50	0.19 0	0.19 0	0.190	0.23 U	13.0	0.19 0
BIS(2-CHLOROETHOXY)METHANE	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.190
BIS(2-CHLOROETHYL)ETHER	NS	0.19 0	0.19 0	0.190	0.23 0	130	0.190
BIS(2-CHLOROISOPROPYL)ETHER	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
BIS(2-ETHYHEXYL)PHTHALATE	50	0.11 1	0.14 JB	0.26 B	0.42 B	13.0	0.60
DI-N-BUTYL PHTHALATE	6.1	0.19 0	0.22 B	0.068 18	0.096 JB	13.0	0.068 J
DI-N-OCTYL PHTHALATE	50 0.014 or MDL	0.091 J	0.074 J	0.15 J	0.23 U	130	0.039 J
DIBENZ[A,H]ANTHRACENE	0.014 OF MDL	0.19 ()	0.19 U	0.050 J	0.92	13 0	0.19 U
DIETHYL PHTHALATE	7.1	0.19 U	0.19 U	0.19 U	0.23 U	13.0	0.19 U
DIMETHYL PHTHALATE	2	U.19 U	0.19 U	0.19 U	0.23 U	130	0.19 U
FLUORANTHENE	50	0.19 U	0.21	0.79	5.0	13.0	0.88
FLUORENE	50	U.19 U	0.051 J	0.19 U	0.27	13.0	0.13 J
HEXACHLORO-1,3-BUTADIENE	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
HEXACHLOROBENZENE	0.41	0.19 U	0.19 U	10.19 U	0.23 U	13 U	0.19 U
HEXACHLOROCYCLOPENTADIENE	NS	0.57 U	0.56 U	0.57 U	0.68 U	38 U	0.57 U
HEXACHLOROETHANE	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
INDENO[1,2,3-CD]PYRENE	3.2	0.19 U	0.068 J	0.14 J	2.6	13.0	0.15 J
ISOPHORORNE	4.4	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
M-DICHLOROBENZENE	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
N-NITROSO-DI-N-PROPYLAMINE	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
N-NITROSODIMETHYLAMINE	NS	0.19 U	0.19 U	10.19 U	0.23 U	13.0	U.19 U
N-NITROSODIPHENYLAMINE	NS	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
NAPHTHALENE	13	0.19 U	0.052 J	10.19 U	0.18 J	130	10.15 J
NITROBENZENE	0.200 or MDL	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
PENTACHLOROPHENOL	1.0 or MDL	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
PHENANTHRENE	50	0.19 U	0.22	0.37	1.1	13 U	0.95
PHENOL	0.03 or MDL	0.19 U	0.19 U	0.19 U	0.23 U	13 U	0.19 U
PYRENE	50	0.19 U	0.27	0.77	4.8	13 U	0.95
TOTAL SVOCs	500	0.201	1.884	4.474	45.656	2.6	6.741

U Undetectable Levels

NS No Standard MDL Method Detection Limit

1

\$

A CONTRACTOR OF A CONTRACT

\$. T.Y.

2000

Acres 100.

×.

. N.

`~~ \$

-40

Location	Recommended	PG-01-1	PG-RR-1	PG-RR-1	PG-RR-2	PG-RR-3	PG-RR-4	PG-RR-4
Sample Date	Soil	11/30/2000	11/3/2000	11/3/2000	11/4/2000	11/3/2000	11/4/2000	11/4/2000
Sample ID	Cleanun	PG-01-1	PG-RR-01	PG-RR-01	PG-RR-07	PG-RR-03	PG-RR-04	PG-RR-04
Sample Denth	Objective	4-6'	0.1.2	1 2 2	0.2	1 5.2'	0.6.2	3_4'
Concentration	мс/кс	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MGRG
1 2 4-TRICHLOROBENZENE	34	0.2611	0 19 11	0 1911	0 20 11	0.97.11	0 1911	0 21 11
1 2-BENZPHENANTHRACENE	NS	0.26 U	1.5	0.19.0	0.57	11	0.73	0.21 0
1.2-DICHLOROBENZENE	7.9	0.2611	0 19 11	0 1911	0.32	0.9711	0.1911	0.21 0
1 2-DIPHENVI HYDRAZINE	NS	0.051 11	0.03911	0.03811	0.03911	0.970	0.037.11	0.04711
14 DICHLOROBENZENE	8.5	0.2611	0 1911	0.1911	0.0000	0.9711	0.1911	0.21.13
246-TRICHLORORPHENOL	0.1	0.26 U	0 19 11	0 1911	0.2011	0 97 11	0 19 11	0.21.11
2 4-DICHLOROPHENOL	0.4	0.26 U	0 19 11	0 1911	0 20 11	0.97.11	0 19 11	0 21 11
2.4-DIMETHYLPHENOL	NS	0.26 U	0.19 U	0.1911	0.2011	0.97 U	0.19 U	0.21 U
2.4-DINITROPHENOL	0.200 or MDL	0.51 U	0.39 U	0.38 U	0.39 U	1.9 U	0.37 U	0.42 U
2.4-DINITROTOLUENE	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
2.6-DINITROTOLUENE	1	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
2-CHLORORNAPHTHALENE	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
2-CHLOROPHENOL	0.8	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
2-NITROPHENOL	0.330 or MDL	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
3,3'-DICHLOROBENZIDINE	N/A	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
4,6-DINITRO-O-CRESOL	0.100 or MDL	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
4-BROMOPHENYLPHENYL ETHER	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
4-CHLORO-3-METHYLPHENOL	0.240 or MDL	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
4-CHLOROPHENYLPHENYL ETHER	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
4-NITROPHENOL	0.100 or MDL	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
ACENAPHTHENE	50	1.7	0.053 J	0.19 U	0.20 U	0.50 J	0.19 U	0.21 U
ACENAPHTHYLENE	41	0.26 U	0.41	0.19 U	0.11 J	3.3	0.19 U	0.21 U
ANTHRACENE	50	0.26 U	0.32	0.044 J	0.16 J	2.0	0.049 J	0.21 U
BENZIDINE	NS	0.51 U	0.39 U	0.38 U	0.39 U	1.9 U	0.37 U	0.42 U
BENZO[A]ANTHRACENE	0.224 or MDL	0.26 U	0.96	0.12 J	0.29	8.9	0.20	0.21 U
BENZO[A]PYRENE	0.061 or MDL	0.26 U	1:2	0.15 J	0.35	15	0.19	0.21 U
BENZO[B]FLOURANTHENE	1.1	0.054 J	3.2	0.32	0.96	26	0.18 J	0.21 U
BENZO[G,H,I]PERYLENE	50	0.26 U	0.51	0.074 J	0.16 J	3.5	0.12 J	0.21 U
BENZO[K]FLOURANTHENE	1.1	0.26 U	0.19 U	0.19 U	0.20 U	1r 🦾	0.17 J	0.21 U
BENZYL BUTYL PHTHALATE	50	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
BIS(2-CHLOROETHOXY)METHANE	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
BIS(2-CHLOROETHYL)ETHER	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
BIS(2-ETHYHEXYL)PHTHALATE	50	0.11 J	0.075 J	0.19 U	0.052 J	0.23 J	0.052 J	0.071 J
DI-N-BUTYL PHTHALATE	8.1	0.26 U	0.19 U	0.19 U	0.060 J	0.97 U	0.046 J	0.062 J
DI-N-OCTYL PHTHALATE	50	0.095 J	0.071 J	0.19 U	0.064 J	0.35 J	0.067 J	0.057 J
DIBENZ[A,H]ANTHRACENE	0.014 or MDL	0.26 U	0.29	0.046 J	0.089 J	0.97 U	0.19 U	0.21 U
DIETHYL PHTHALATE	7.1	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
DIMETHYL PHTHALATE	2	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.057 J
FLUORANTHENE	50	0.49	1.5	0.18 J	0.39	6.6	0.33	0.21 U
FLUORENE	50	0.35	0.081 J	0.19 U	0.042 J	0.47 J	0.19 U	0.21 U
HEXACHLORO-1,3-BUTADIENE	NS 0.41	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
HEXACHLOROBENZENE	0.41	0.26 U	0.19 0	0.190	0.20 U	0.97 U	0.19 0	0.21 0
HEXACHLOROCYCLOPENTADIENE	NS	0.77 U	0.58 U	0.57 U	0.59 U	2.9 U	0.56 U	0.62 U
HEXACHLOROETHANE	NS 2.2	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
INDENO[1,2,3-CD]PYRENE	3.2	0.26 U	0.55	0.085 J	0.17 J	4.3 3 2 2	0.10 J	0.21 U
ISOPHORORNE	4.4	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 U	0.21 U
M-DICHLOKOBENZENE	NS NS	0.26 U	0.190	0.19 0	0.20 0	0.97 0	0.19 0	0.21 0
IN-NITROSO-DI-N-PROPYLAMINE	NS	0.26 U	0.19 0	0.19.0	0.20 0	0.97 U	0.19 0	0.21 U
N-NITROSODIMETHYLAMINE	NS NS	0.26 U	0.19 0	0.19 0	0.20 0	0.97 0	0.19 0	0.21 U
N-NITROSODIPHENYLAMINE	NS 17	0.26 0	0.1910	0.19 0	0.20 U	0.97 U	0.190	0.21 0
NAPHIHALENE	13 0 200 or MD!	0.12 J	0.34	0.039 J	0.50	2.2	0.12.1	0.21 0
NITKOBENZENE	0.200 OF MDL	0.26 U	0.19 U	0.19 U	0.20 U	0.97 U	0.19 0	0.21 0
PEN FACHLOROPHENOL	I.U UF MUL	0.26 U	0.19 0	0.19 0	0.20 U	0.97 U	0.19 0	0.21 0
PHENANTHRENE		0.11.0	0.60	0.085 J	0.47	2.0	0.21	0.21 0
PHENOL		0.26 U	0.190	0.19 0	0.20 U	0.97 0	0.190	0.21 U
PYKENE	50	0.29	1.5	U.16 J	0.42	9.3	0.34	0.21 U
TOTAL SVOCs	500	3.319	113.16	11.483	4.807	106.65	12.404	0.247

U Undetectable Levels NS No Standard

\$

MDL Method Detection Limit

Sec.

ł.

÷.

	n	DC DD C	DC DD C	nc nn c	DC DD (DC DD 7	DC DD 7	DC STAIN 02	DC STAIN 02
Location	Recommended	rG-RR-5	PG-RR-5	rG-RR-0	PG-RR-0	FG-RR-/	PG-RR-7	PG-STAIN-02	PG-STAIN-02
Sample Date	Soil	11///2000	11///2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000	11/11/2000	11/11/2000
Sample ID	Cleanup	PG-KR-05	PG-RR-05	PG-RK-06	PG-RR-06	PG-RR-07	PG-RR-07	PG-S1-02	PG-S1-02
Sample Depth	Objective	1-2"	2-4	1-2	2-4	0-2	2-4	1-2	2-3
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,2,4-TRICHLOROBENZENE	3.4	0.18 U	0.18 U	0.18 0	0.18 U	0.18 U	0.18 0	0.20 U	0.20 U
1,2-BENZPHENANTHRACENE	NS	0.055 J	0.18 U	0.079 J	0.18 U	0.18 U	0.11 J	0.20 U	0.20 U
1,2-DICHLOROBENZENE	7.9	0.18 U	0.20 U	0.20 U					
1,2-DIPHENYLHYDRAZINE	NS	0.036 U	0.039 U	0.040 U					
1.4-DICHLOROBENZENE	8.5	0.18 U	0.20 U	0.20 U					
2,4,6-TRICHLORORPHENOL	0.1	0.18 U	0.20 U	0.20 U					
2.4-DICHLOROPHENOL	0.4	0.18 U	0.20 U	0.20 U					
2,4-DIMETHYLPHENOL	NS	0.18 U	0.18 U	0.18 0	0.18 0	0.18 0	0.18 0	0.20 0	0.20 ()
2,4-DINITROPHENOL	0.200 OF MDL	0.36 U	0.39 0	0.40 U					
2,4-DINITROTOLUENE	NS	0.18 U	0.18 0	0.20 0	0.20 U				
2,6-DINITROTOLUENE	1	0.18 U	0.18 U	0.18 0	0.18 0	0.18 0	0.18 0	0.20 U	0.20 U
2-CHLORORNAPHTHALENE	NS 0.9	0.18 U	0.18 U	0.18 0	0.18 U	0.18 0	0.18 U	0.20 0	0.20 U
2-CHLOROPHENOL	0.0	0.18 0	0.18 U	0.18 0	0.18 U	0.18 U	0.18 U	0.20 0	0.20 U
2-NITROPHENOL		0.18 U	0.18 U	0.18 0	0.18 U	0.18 U	0.18 0	0.20 U	0.20 0
AC DRITTED O CERCOL	0.100 or MDI	0.18 U	0.18 U	0.18 U	0.10 0	0.10 U	0.18 U	0.20 0	0.20 0
4,0-DINITKO-O-CRESOL		0.18 U	0.18 U	0.18 U	0.18 U	0.10 U	0.18 U	0.20 0	0.20 0
4-BROMOPHENYLPHENYL ETHER	NS 0.240 or MDI	0.18 U	0.20 0	0.20 0					
4-CHLORO-3-METHYLPHENOL	0.240 OF MDL	0.18 U	0.18 U	0.18 U	0.18 U	0.18 0	0.18 U	0.20 0	0.20 0
4-CHLOKOPHENYLPHENYL ETHER		0.18.0	0.18 U	0.20 0	0.200				
4-NITROPHENOL	0.100 OF MDL	0.18 0	0.18 U	0.20 U	0.20 U				
ACENAPHTHENE	<u> </u>	0.18 U	0.18 0	0.18 U	0.18 U	0.18 U	0.18 U	0.20 U	0.20 0
ACENAPHIHYLENE	41 50	0.18 U	0.18 U	0.18.0	0.18 U	0.18 U	0.18 U	0.20 0	0.20 U
ANTHRACENE	30	0.18 U	0.18 0	0.18 0	0.18 U	0.16 U	0.16 U	0.20 0	0.20 0
BENZIDINE	0.224 or MDL	0.36 U	0.36 U	0.36 U	0.36 U	0.36 0	0.36 U	0.39 0	0.40 0
BENZOLAJAN I HRACENE	0.224 OF MDL	0.18 U	0.18 U	0.054 J	0.18 U	0.18 U	0.074 J	0.20 U	0.20 U
BENZO[A]PYKENE	1 1	0.18 U	0.18 U	0.034 J	0.18 U	0.18 0	0.123	0.20 U	0.20 0
BENZO[B]FLOUKAN THENE	50	0.18 U	0.18 U	0.123	0.18 U	0.18 U	0.22	0.20 U	0.20 0
BENZOIG,H,IJPERYLENE	11	0.18 U	0.18 0	0.043 J	0.18 13	0.18 U	0.17 1	0.20 U	0.20 0
BENZOK FLOUKANTHENE	50	0.18 U	0.18 1	0.18.0	0.181	0.18 0	0.1811	0.20 0	0.20 0
BENZIL BUTTL FRITALATE	NS	0.18 U	0.181	0.18.0	0.1813	0.18 U	0.18 []	0.20 U	0.20 U
BIS(2-CHLOROETHUX) ETHER	NS	0.18 U	0.18 0	0.18 U	0.18 U	0.18 U	0.1811	0.20 U	0.20 0
BIS(2-CHLOROETHTL)ETHER	NS	0.18 U	0.18 U	0.18.0	0.18 11	0.18 U	0.18 U	0.20 U	0.20 U
DIS(2 ETUVIEVVI) BUTUALATE	50	0.18 U	0.18 0	0.18 U	0.18 U	0.180	0.15 0	0.12 19	0.12 19
DIN BUTY BUTHALATE	8.1	0.18 0	0.18.11	0.18 U	0.1811	0.045 5	0.068 1	0.067.1	0.2011
DI-N-DOTTE PHTHALATE	50	0.054 [0.18 U	0.086 IB	0.078 IB	0.077 IB	0.11 /B	0.2011	0.20 0
DIBENZIA HIANTHRACENE	0.014 or MDI	0.1811	0.1811	0.1811	0.18 U	0.07735	0.039 1	0.2011	0.2013
DISTRYL PHTHALATE	7.1	0 18 17	0 18 11	0 18 11	0 18 11	0.18.11	0 18 11	0.2011	0.20 0
DIMETHYL PHTHALATE	2	0 18 11	0 18 11	0 18 11	0 18 11	0.18 11	0.181	0 20 11	0 20 11
FLUORANTHENE	50	0 18 1/	0.18 U	0.081 1	0.18 U	0.18 []	0.071 1	0 20 11	0.20 U
FLUORENE	50	0.181/	0.18 U	0.1811	0.181	0.18 U	0.1811	0.2011	0 20 11
HEXACHLORO-1.3-BUTADIENE	NS	0.18 U	0.20 U	0.20 U					
HEXACHLOROBENZENE	0.41	0 18 11	0.18.0	0.18 U	0.18.11	0.18 U	0.18.U	0.2011	0 20 11
HEXACHLOROCYCLOPENTADIENE	NS	0.54 U	0.5411	0.59 U	0.60 U				
HEXACHLOROETHANE	NS	0.18 U	0.20 U	0.20 U					
INDENO[1 2 3-CDIPYRENE	3.2	0.18 U	0.1811	0.18 U	0.18 U	0.18 U	0.096 J	0.20 U	0.2011
ISOPHORORNE	4.4	0.18 U	0.20 U	0.2011					
M-DICHLOROBENZENE	NS	0 18 11	0.1811	0.181/	0.18 U	0.18 U	0.18 U	0.20 U	0 20 11
N-NITROSO-DI-N-PROPYLAMINE	NS	0.1811	0.18 U	0.2011	0.2011				
N-NITROSODIMETHYLAMINE	NS	0 18 11	0 18 11	0 18 11	0.18 U	0.18 U	0 18 U	0 20 11	0 20 11
N-NITROSODIPHENVI AMINE	NS	0 18 11	0.18 U	0.18 U	0.181/	0.18 U	0.18 U	0.2011	0 20 11
NAPUTHALENE	13	0.1811	0.18.11	0.090 1	0 18 1/	0.1811	0.181/	0 20 11	0 20 11
NITROBENZENE	0.200 or MDI	0.18 U	0.181	0.18 U	0.18 U	0.18 U	0.18 U	0.20 U	0.20 U
PENTACHLOROPHENOL	1.0 or MDL	0.18 U	0.20 U	0.20 U					
PHENANTHRENE	50	0.064 J	0.18 U	0.11 J	0.18 U	0.18 U	0.046 1	0.20 U	0.20 U
PHENOL	0.03 or MDL	0.18 U	0.20 U	0.20 U					
PYRENE	50	0.18 U	0.18 U	0.079 J	0.18 U	0.18 U	0.14 J	0.20 U	0.20 1/
TOTAL SVOCs	500	0.232	ND	0.796	0.078	0.25	1.414	0.187	0.13
	1	1	L	1	1	1	1	1	1

U Undetectable Levels NS No Standard

\$

MDL Method Detection Limit

		· · · · · · · · · · · · · · · · · · ·						
Location	Recommended	PG-STAIN-03	PG-STAIN-03	PG-STAIN-3B	PG-STAIN-3B	PG-FS-2	PG-FS-2	PG-FS-2
Sample Date	Soil	11/10/2000	11/10/2000	11/11/2000	11/11/2000	11/17/2000	11/17/2000	11/17/2000
Sample ID	Cleanup	PG-ST-03	PG-ST-03	PG-ST-3B	PG-ST-3B	PG-FS-02	PG-FS-02	PG-FS-02
Sample Depth	Objective	1.7-2.5'	2.5-3.5'	0-2'	2-4'	2-4'	8-10	17-18'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,2,4-TRICHLOROBENZENE	3.4	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
1,2-BENZPHENANTHRACENE	NS	77	2.4	0.55	0.18 U	0.19 U	0.20 U	0.19 U
1,2-DICHLOROBENZENE	7.9	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
1,2-DIPHENYLHYDRAZINE	NS	3.9 U	0.21 U	0.036 U	0.036 U	0.038 U	0.041 U	0.037 U
1,4-DICHLOROBENZENE	8.5	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2,4,6-TRICHLORORPHENOL	0.1	19 U	1.I U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2,4-DICHLOROPHENOL	0.4	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2,4-DIMETHYLPHENOL	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2,4-DINITROPHENOL	0.200 or MDL	39 U	2.1 U	0.36 U	0.36 U	0.38 U	0.41 U	0.37 U
2,4-DINITROTOLUENE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2,6-DINITROTOLUENE	1	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2-CHLORORNAPHTHALENE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2-CHLOROPHENOL	0.8	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
2-NITROPHENOL	0.330 or MDL	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
3,3'-DICHLOROBENZIDINE	N/A	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
4,6-DINITRO-O-CRESOL	0.100 or MDL	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
4-BROMOPHENYLPHENYL ETHER	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
4-CHLORO-3-METHYLPHENOL	0.240 or MDL	19 U	1.I U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
4-CHLOROPHENYLPHENYL ETHER	NS	19 U	1. 1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
4-NITROPHENOL	0.100 or MDL	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
ACENAPHTHENE	50	200	5.6	0.059 J	0.18 U	0.19 U	0.20 U	0.19 U
ACENAPHTHYLENE	41	32	0.80 J	0.044 J	0.18 U	0.19 U	0.20 U	0.19 U
ANTHRACENE	50	140	4.1	0.13 J	0.18 U	0.19 U	0.20 U	0.19 U
BENZIDINE	NS	39 U	2.1 U	0.36 U	0.36 U	0.38 U	0.41 U	0.37 U
BENZO[A]ANTHRACENE	0.224 or MDL	86	2.6	0.44	0.18 U	0.19 U	0.20 U	0.19 U
BENZO[A]PYRENE	0.061 or MDL	44 🐀 👘 🦾	13.	0.47	0.18 U	0.19 U	0.20 U	0.19 U
BENZO[B]FLOURANTHENE	1.1	63	2	1.1	0.18 U	0.19 U	0.20 U	0.19 U
BENZO[G,H,I]PERYLENE	50	18 J	0.52 J	0.24	0.18 U	0.19 U	0.20 U	0.19 U
BENZO[K]FLOURANTHENE	1.1	22	0.66 J	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
BENZYL BUTYL PHTHALATE	50	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
BIS(2-CHLOROETHOXY)METHANE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
BIS(2-CHLOROETHYL)ETHER	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	19 U	1.1 U	0.18-U	0.18 U	0.19 U	0.20 U	0.19 U
BIS(2-ETHYHEXYL)PHTHALATE	50	19 U	0.26 JB	0.30 B	0.087 JB	0.19 U	0.050 J	0.19 U
DI-N-BUTYL PHTHALATE	8.1	19 U	1.1 U	0.086 J	0.056 J	0.19 U	0.20 U	0.19 U
DI-N-OCTYL PHTHALATE	50	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
DIBENZ[A,H]ANTHRACENE	0.014 or MDL	6.3 J	1.1 U	0.13 J	0.18 U	0.19 U	0.20 U	0.19 U
DIETHYL PHTHALATE	7.1	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
DIMETHYL PHTHALATE	2	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
FLUORANTHENE	50	350	9.9	0.66	0.055 J	0.19 U	0.20 U	0.19 U
FLUORENE	50	190	5.3	0.054 J	0.18 U	0.19 U	0.20 U	0.19 U
HEXACHLORO-1,3-BUTADIENE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
HEXACHLOROBENZENE	0.41	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
HEXACHLOROCYCLOPENTADIENE	NS	58 U	3.2 U	0.54 U	0.54 U	0.57 U	0.61 U	0.56 U
HEXACHLOROETHANE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
INDENO[1,2,3-CD]PYRENE	3.2	19 J	0.58 J	0.23	0.18 U	0.19 U	0.20 U	0.19 U
ISOPHORORNE	4.4	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
M-DICHLOROBENZENE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
N-NITROSO-DI-N-PROPYLAMINE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
N-NITROSODIMETHYLAMINE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
N-NITROSODIPHENYLAMINE	NS	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
NAPHTHALENE	13	420	11	0.18 J	0.18 U	0.19 U	0.20 U	0.19 U
NITROBENZENE	0.200 or MDL	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
PENTACHLOROPHENOL	1.0 or MDL	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
PHENANTHRENE	50	540	16	0.49	0.18 U	0.19 U	0.20 U	0.19 U
PHENOL	0.03 or MDL	19 U	1.1 U	0.18 U	0.18 U	0.19 U	0.20 U	0.19 U
PYRENE	50	230	7.1	0.77	0.043 J	0.19 U	0.20 U	0.19 U
TOTAL SVOCs	500	2437.3	70.12	5.933	0.241	ND	0.05	ND
Least the second s	A		· · · · · · · · · · · · · · · · · · ·		• • • •	·	·	•

U Undetectable Levels NS No Standard

\$.

MDL Method Detection Limit

_

1.9.2 0.000

A 100

1.685

Sample Date Sample Dist Classo 11/1/S000 <	Location	Recommended	PG-FS-3	PG-FS-3	PG-FILL-02	PG-FILL-03	PG-FILL-03	PG-FILL-03	PG-FILL-04
Sample Dg1 Sample Dg1 Sample Dg1Cenample Dg1 ConcentrationCenample Dg1 Concentration	Sample Date	Soil	11/15/2000	11/15/2000	11/3/2000	11/4/2000	11/6/2000	11/6/2000	11/6/2000
Sample Depth Objective 2-4 6-5-7 0.5-7	Sample ID	Cleanup	PG-FS03	PG-FS03	PGFILL02	PGFILL03	PGFILL03	PGFILL03	PGFILL04
Casesentiane MGRG	Sample Depth	Objective	2-4'	6.5-8'	0.7-3'	0.5-2	2-4'	4-6'	0-2'
1.4.+TRCH0.000EN2ENE 3.4 0.19 U 0.20 U 0.12 U 0.19 U 0.19 U 0.19 U 1.2.BORZHIGNANGACENE NS 0.68 U 0.20 U 0.21 U 0.18 U 0.19 U	Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
D_BERNAMIRBACENE NS 0.481 0.81 0.58 0.631 0.633 0.633 0.633 0.633 0.634 0.634 0.634 0.634 0.634 0.634 0.634 0.634 0.637	1,2,4-TRICHLOROBENZENE	3.4	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
L3-DECRAORGENZENE 7.9 0.19 U 0.20 U 0.21 U 0.19 U 0.19 U 0.19 U L-DORIGNUTVARCATE NS 0.38 U 0.39 U 0.20 U 0.11 U 0.19 U <	1,2-BENZPHENANTHRACENE	NS	0.048 J	0.20 U	0.18 J	1.6	0.58	0.053 J	0.33
I_D_DIPLESYLITYORAZENE NS 0.08U 0.04U 0.08U 0.02U 0.18U 0.19U 0.19U <td>1,2-DICHLOROBENZENE</td> <td>7.9</td> <td>0.19 U</td> <td>0.20 U</td> <td>0.22 U</td> <td>0.18 U</td> <td>0.19 U</td> <td>0.19 U</td> <td>0.19 U</td>	1,2-DICHLOROBENZENE	7.9	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
I-DICILICROBENZENE 8.5 0.19 U 0.20 U 0.21 U 0.18 U 0.19 U <t< td=""><td>1,2-DIPHENYLHYDRAZINE</td><td>NS</td><td>0.038 U</td><td>0.039 U</td><td>0.043 U</td><td>0.036 U</td><td>0.037 U</td><td>0.037 U</td><td>0.039 U</td></t<>	1,2-DIPHENYLHYDRAZINE	NS	0.038 U	0.039 U	0.043 U	0.036 U	0.037 U	0.037 U	0.039 U
2.4.5-TRCH.0000PPIENOL 0.19 0.20 0.21 0.18 0.19 0.19 0.19 2.4-DICILLORAPHINOL NS 0.19 0.20 0.18 0.19 0.19 0.19 2.4-DICILLORAPHINOL NS 0.19 0.20 0.18 0.19 0.20 0.18 0.19 0.20 0.18 0.19 0.20 0.18 0.19 0.20 0.18 0.19<	1,4-DICHLOROBENZENE	8.5	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
Za-DBCRINCHINOL 0.9 0.9 0.92 0.81 0.91 0.91 0.91 Za-DMRTRYHENOL NS 0.19 0.31 0.35 0.31 <td>2,4,6-TRICHLORORPHENOL</td> <td>0.1</td> <td>0.19 U</td> <td>0.20 U</td> <td>0.22 U</td> <td>0.18 U</td> <td>0.19 U</td> <td>0.19 U</td> <td>0.19 U</td>	2,4,6-TRICHLORORPHENOL	0.1	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
2.4-DINTROPHENOL NS 0.19 U 0.20 or MDL 0.20 or MDL 0.30 U 0.20 U 0.10 U 0.30 U	2,4-DICHLOROPHENOL	0.4	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
2.4-DINTROPHENOL 0.200 mMDL 0.39 U 0.36 U 0.37 U 0.39 U 0.30 U 0.22 U 0.18 U 0.39 U	2,4-DIMETHYLPHENOL	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
2.4_DNITROTOLUENE NS 0.19 U 0.20 U 0.18 U 0.19 U <th0< td=""><td>2,4-DINITROPHENOL</td><td>0.200 or MDL</td><td>0.38 U</td><td>0.39 U</td><td>0.43 U</td><td>0.36 U</td><td>0.37 U</td><td>0.37 U</td><td>0.39 U</td></th0<>	2,4-DINITROPHENOL	0.200 or MDL	0.38 U	0.39 U	0.43 U	0.36 U	0.37 U	0.37 U	0.39 U
2.5.DINTROTOLUENE 1 0.9 U 0.20 U 0.18 U 0.19 U 0.19 U 0.19 U 2.CHLORORNAMITTALENE NS 0.19 U 0.20 U 0.18 U 0.19 U 0.19 U 2.CHLORORNAMITTALENE NA 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 3.JOCHLOROBENZIDINE NA 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U	2,4-DINITROTOLUENE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
2-CHLORORNAPHTHALENE NS 0 19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 2-CHLOROPHENOL 0.33 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 3-NDICHLORORENZIDINE NA 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 3-SUCICLORORENZIDINE NA 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 456 DMITHOLPHENVL EIHEN 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U CHLICROPHENDIL 0.100 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U ACERNAMITINE 50 0.19 U 0.20 U 0.22 U 0.06 J 0.19 U 0.19 U ACERNAMITINE 50 0.19 U 0.20 U 0.22 U 0.06 J 0.19 U 0.19 U ACERNAMITINE 50 0.19 U 0.22 U 0.06 J 0.19 U 0.19 U ACERNAMITINENE 50 0.19 U	2,6-DINITROTOLUENE	1	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
2-CHLOROPHENOL 0.8 0.19 0.20 0.22 0.18 0.19	2-CHLORORNAPHTHALENE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
2-NITROPIENOL 0.330 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 3-DICHLOROBENZIDINE N/A 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U	2-CHLOROPHENOL	0.8	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
3-DOLLOROBENZIDNE N/A 0.19 U 0.20 U 0.21 U 0.18 U 0.19 U 0.19 U 0.19 U 6-DONTRO-CRESOL 0.100 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.05 U 0.02 U 0.05 U 0.05 J 0.19 U 0.19 U 0.04 U 0.02 U 0.03 U 0.19 U 0.22 U 0.06 J 0.19 U 0.21 U 0.04 J 0.21 U 0.04 U 0.22 U 0.03 U 0.37 U	2-NITROPHENOL	0.330 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
44-DINTRO-O-CRESOL 0.10 or MDL 0.19 U 0.20 U 0.18 U 0.19 U 0.19 U 0.19 U 45R0MOPHENVLPHENVL LETHER NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U CHILGROPHENOL 0.100 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U CHILGROPHENOL 0.100 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U ACENANHTINLENE 50 0.19 U 0.20 U 0.22 U 0.064 J 0.19 U 0.04 J ACENANHTINLENE 41 0.19 U 0.20 U 0.22 U 0.36 U 0.31 U 0.19 U 0.20 U ACENANHTINLENE 50 0.19 U 0.20 U 0.22 U 0.36 U 0.31 U 0.37 U 0.37 U 0.39 U BENZOJAJANTHRACENE 0.56 I 0.30 U 0.20 U 0.22 U 0.66 U 0.31 U 0.21 U<	3,3'-DICHLOROBENZIDINE	N/A	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
4-BRCMOOPHENYLPHENYL ETHER NS 0.19 U 0.20 U 0.21 U 0.18 U 0.19 U 0.19 U 6-CHLORO-SMETHYLPHENDL 0.20 U 0.22 U 0.18 U 0.19 U 0.05 U 0.02 U 0.03 U 0.21 U 0.64 J 0.19 U 0.22 U 0.64 J 0.37 U 0.39 U 0.37 U 0.39 U 0.31 U 0.30 U 0.31	4,6-DINITRO-O-CRESOL	0.100 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
4-CHLORO-J-METHYLPHENOL 0.240 Or MOL [0.19 U 0.20 U 0.21 U 0.18 U 0.19 U 0.20 U 0.21 U 0.36 U 0.36 U 0.37 U 0.32 U 0.11 U 0.20 U 0.21 U 0.21 U 0.21 U 0.23 U 0.36 U 0.37 U 0.32 U 0.37 U 0.32 U 0.32 U 0.36 U 0.32 U 0.32 U 0.36 U 0.32 U 0.32 U 0.32 U 0.32 U 0.32 U	4-BROMOPHENYLPHENYL ETHER	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
4-CHLOROPHENYLETHER NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U ANTROPHENDL 0.100 Or MDL 0.19 U 0.20 U 0.22 U 0.09 U 0.05 U 0.19 U 0.20 U 0.22 U 0.064 I 0.19 U 0.20 U 0.22 U 0.64 I 0.19 U 0.27 U 0.37 U 0.35 U 0.19 U 0.20 U 0.21 U 0.56 U 0.19 U 0.10 U 0.20 U 0.22 U 0.66 U 0.34 U 0.19 U 0.10 U <t< td=""><td>4-CHLORO-3-METHYLPHENOL</td><td>0.240 or MDL</td><td>0.19 U</td><td>0.20 U</td><td>0.22 U</td><td>0.18 U</td><td>0.19 U</td><td>0.19 U</td><td>0.19 U</td></t<>	4-CHLORO-3-METHYLPHENOL	0.240 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
ANTROPHENCL 0.100 or MCL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U ACENAPHTHENE 41 0.19 U 0.20 U 0.22 U 0.669 J 0.058 J 0.19 U 0.19 U ACENAPHTHENE 41 0.19 U 0.20 U 0.22 U 0.669 J 0.068 J 0.19 U 0.04 J ANTHRACENE 50 0.19 U 0.20 U 0.12 J 0.33 U 0.37 U 0.37 U 0.32 U BENZOJAPATRACENE 0.224 or MCL 0.642 J 0.20 U 0.12 J 1.3 0.47 0.19 U 0.22 U BENZOJAPATRANENE 0.051 Or MCL 0.043 J 0.20 U 0.48 J 1.4 0.59 0.07 J 0.45 BENZOJAPATRNENE 1.1 0.19 U 0.20 U 0.22 U 0.66 O 0.3 U 0.19 U	4-CHLOROPHENYLPHENYL ETHER	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
ACENAPITHENE 50 0.19 U 0.20 U 0.22 U 0.09 J 0.058 J 0.19 U 0.19 U ACENAPITHYLENE 41 0.19 U 0.20 U 0.22 U 0.33 0.12 J 0.19 U 0.040 J BENZDIA 0.39 U 0.20 U 0.22 U 0.33 0.12 J 0.19 U 0.040 J BENZDIAJONE 0.224 or MDL 0.042 J 0.20 U 0.011 J 1.3 0.37 U 0.39 U 0.22 U 0.36 U 0.37 U 0.22 U 0.21 (************************************	4-NITROPHENOL	0.100 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
ACENAPITHYLENE 41 0.19 U 0.20 U 0.22 U 0.064 J 0.064 J 0.19 U 0.040 J BENZDAR 50 0.19 U 0.20 U 0.33 U 0.33 U 0.37 U </td <td>ACENAPHTHENE</td> <td>50</td> <td>0.19 U</td> <td>0.20 U</td> <td>0.22 U</td> <td>0.099 J</td> <td>0.058 J</td> <td>0.19 U</td> <td>0.19 U</td>	ACENAPHTHENE	50	0.19 U	0.20 U	0.22 U	0.099 J	0.058 J	0.19 U	0.19 U
ANTHRACENE 50 0.19 U 0.20 U 0.22 U 0.33 0.12 J 0.19 U 0.040 J BENZIDNE NS 0.38 U 0.39 U 0.34 U 0.37 U 0.32 U 0.32 U 0.34 U 0.19 U	ACENAPHTHYLENE	41	0.19 U	0.20 U	0.22 U	0.064 J	0.046 J	0.19 U	0.19 U
BENZDIANTHAACENE NS 0.38 U 0.39 U 0.43 U 0.36 U 0.37 U 0.32 U 0.14 U 0.19 U 0.14 U 0.19 U 0	ANTHRACENE	50	0.19 U	0.20 U	0.22 U	0.33	0.12 J	0.19 U	0.040 J
BENZQ(A)NTHRACENE 0.224 or MUL 0.041 0.20 0.121 13.8 0.19 U 0.22 BENZQ(A)PRENE 0.051 //D 0.001 //D 0.011 1.4 0.59 0.073 //D 0.45 BENZQ(A)PRENE 1.1 0.051 //D 0.20 U 0.041 //D 1.4 0.59 0.073 //D 0.45 BENZQ(B)FLOURANTHENE 1.1 0.19 U 0.20 U 0.22 U 0.66 0.34 0.19 U 0.19 U BENZQ(K)FLOURANTHENE 1.1 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U <td>BENZIDINE</td> <td>NS</td> <td>0.38 U</td> <td>0.39 U</td> <td>0.43 U</td> <td>0.36 U</td> <td>0.37 U</td> <td>0.37 U</td> <td>0.39 U</td>	BENZIDINE	NS	0.38 U	0.39 U	0.43 U	0.36 U	0.37 U	0.37 U	0.39 U
BENZQIGLORANTHENE 0.0051 of PMU 0.031 0.20 U 0.081 J 1.1.4 0.053 0.070 U 0.043 BENZQIGLORANTHENE 1.1 0.051 U 0.20 U 0.22 U 0.66 0.34 0.19 U 0.19 U BENZQIGLURANTHENE 1.1 0.19 U 0.20 U 0.22 U 0.66 0.34 0.19 U 0.19 U BENZVL GUTVL PHTHALATE 50 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U <td>BENZO[A]ANTHRACENE</td> <td>0.224 or MDL</td> <td>0.042 J</td> <td>0.20 U</td> <td>0.12 J</td> <td>13.8%</td> <td>0.47</td> <td>0.19 U</td> <td>0.22</td>	BENZO[A]ANTHRACENE	0.224 or MDL	0.042 J	0.20 U	0.12 J	13.8%	0.47	0.19 U	0.22
BENZOGUBJELOURANTHENE 1.1 0.051 J 0.20 U 0.21 U 0.66 0.34 0.19 U 0.10 J BENZOGUAJPERVLENE 50 0.19 U 0.20 U 0.22 U 0.66 0.34 0.19 U 0.19 U BENZOGUAJPERVLENE 1.1 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U BENZOGUAJPERVLENE 50 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U BESC2-CHLOROETHYL)ETHER NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U	BENZO[A]PYRENE	0.061 OF MDL	0.043 J	0.20 U	0.081 J	1.4	0.54	0.19 U	0.21
BENZQUKLDURANTHENE 50 0.19 U 0.20 U 0.22 U 0.66 0.34 0.19 U 0.19 U BENZQUKLDURANTHENE 1.1 0.19 U 0.20 U 0.064 J 1.3 0.53 0.19 U 0.19 U BENZQUKLDURANTHENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U BIS2C-CHLOROETHVLY.PHTHALATE S0 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U BIS2C-CHLOROETHVLPHTHER NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U BIS2C-CHLOROETHVLPHTHALATE 50 0.091 JB 1.3 B 0.12 J 0.078 J 0.043 J 0.19 U DI-N-BUTYL PHTHALATE 50 0.044 JB 0.11 JB 0.23 U 0.073 J 0.073 J 0.073 J 0.017 J 0.099 U 0.19 U DIBENZ[A,H]ANTHRACENE 0.19 U 0.20 U 0.22 U 0.35 0.19 U	BENZO[B]FLOURANTHENE	1.1	0.051 J	0.20 U	0.14 J	1.4	0.59	0.073 J	0.45
BENZYL BUTYL PHTHALATE 50 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U BIS/2-CHLOROETHOXY)METHANE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U </td <td>BENZOLGH,IJPERYLENE</td> <td>50</td> <td>0.190</td> <td>0.20 0</td> <td>0.22 0</td> <td>0.66</td> <td>0.34</td> <td>0.190</td> <td>0.10 J</td>	BENZOLGH,IJPERYLENE	50	0.190	0.20 0	0.22 0	0.66	0.34	0.190	0.10 J
BENZYL BUTYL PHTHALATE 30 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U BIS(2-CHLOROETHAYL)ETHAR NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U <td>BENZOKIFLOURANTHENE</td> <td>50</td> <td>0.190</td> <td>0.20 U</td> <td>0.064 J</td> <td>1.5</td> <td>0.55</td> <td>0.19 C</td> <td>0.190</td>	BENZOKIFLOURANTHENE	50	0.190	0.20 U	0.064 J	1.5	0.55	0.19 C	0.190
BIS(2-CHLOROETHACTMETHER NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U BIS(2-CHLOROETHYL)ETHER NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U BIS(2-CHLOROISOPROPYL)ETHER NS 0.19 U 0.091 JB 1.3 B 0.12 J 0.038 J 0.21 U 0.19 U 0.19 U 0.19 U DIN-BUTYL PHTHALATE S0 0.091 JB 1.3 B 0.12 J 0.038 J 0.043 J 0.19 U 0.19 U </td <td>BENZYL BUTYL PHTHALATE</td> <td>50</td> <td>0.190</td> <td>0.20 U</td> <td>0.22 0</td> <td>0.18 U</td> <td>0.190</td> <td>0.190</td> <td>0.190</td>	BENZYL BUTYL PHTHALATE	50	0.190	0.20 U	0.22 0	0.18 U	0.190	0.190	0.190
BIS(2-CHLOROE)HTER NS 0.19 U 0.20 U 0.18 U 0.19 U 0.19 U 0.19 U BIS(2-CHLOROE)SOPROPYLETHER NS 0.19 U 0.20 U 0.22 U 0.08 J 0.21 U 0.19 U 0.19 U 0.19 U BIS(2-CHLOROSOPROPYLETHER NS 0.19 U 0.22 U 0.078 J 0.043 J 0.19 U 0.19 U DI-N-DCTYL PHTHALATE 8.1 0.19 U 0.20 U 0.22 U 0.078 J 0.043 J 0.19 U 0.19 U DIENZ(A,HJANTHRACENE 0.014 or MDL 0.19 U 0.20 U 0.22 U 0.35 0.16 J 0.19 U 0.19 U DIETHYL PHTHALATE 7.1 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U DIMETHYL PHTHALATE 7.1 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U DIMETHYL PHTHALATE 50 0.079 J 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PLUORANTHENE 50 0.19 U 0.20 U 0.22 U 0.084 J 0.19 U 0.19 U 0.19 U	BIS(2-CHLOROETHOXY)METHANE	NS NC	0.190	0.20 0	0.22 0	0.18 U	0.190	0.190	0.190
BIS(2-ETLUMENTAGE TELETINE NS 0.19 U 0.09 U 0.12 U 0.18 U 0.19 U 0.19 U DI-N-BUTYL PHTHALATE 50 0.091 JB 1.3 B 0.12 J 0.038 J 0.21 U 0.043 J 0.19 U 0.19 U DI-N-BUTYL PHTHALATE 50 0.044 JB 0.11 JB 0.23 U 0.057 JB 0.17 JB 0.059 JB 0.11 JB DIBENZ(A.HJANTHRACENE 0.014 or MDL 0.19 U 0.20 U 0.22 U 0.35 L 0.16 J 0.19 U 0.19 U DIETHYL PHTHALATE 7.1 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U DIMETHYL PHTHALATE 2 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U FLUORENE 50 0.079 J 0.20 U 0.22 U 0.84 J 0.19 U 0.19 U 0.19 U HEXACHLORO-1,3-BUTADIENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U HEXACHLOROCYCLOPENTADIENE NS<	BIS(2-CHEOROETHTE)ETHER	NO	0.19.0	0.20 U	0.22 0	0.18 0	0.19.0	0.19.0	0.19 U
INSCREPTION TRACKIE 30 0.071/B 0.012 0.035/J 0.21 0.035/J 0.21 0.015/J 0.021/J 0.022/J 0.35/J 0.016/J 0.019/J 0.19/J 0.20/J 0.22/J 0.35/J 0.016/J 0.19/J 0.19/J 0.20/J 0.22/J 0.18/J 0.19/J 0.19/J 0.19/J 0.19/J 0.20/J 0.22/J 0.08/J 0.19/J 0.19/J 0.20/J 0.22/J 0.08/J 0.19/J 0.19/J 0.19/J 0.19/J 0.19/J 0.19/J 0.19/J <th0.19 j<="" th=""> 0.19/J 0.19/J</th0.19>	BIS(2-CHLOROISOPROFIL)ETHER	50	0.001 1P	1.1.8	0.12 0	0.180	0.190	0.19.0	0.19.0
Diskont Primatzine Disk Diskont Primatzine Diskont Primatzine <td>DIN BUTYL BUTHALATE</td> <td>81</td> <td>0.091 JB</td> <td>0.039 1</td> <td>0.12 J</td> <td>0.078 3</td> <td>0.043 1</td> <td>0.190</td> <td>0.19.0</td>	DIN BUTYL BUTHALATE	81	0.091 JB	0.039 1	0.12 J	0.078 3	0.043 1	0.190	0.19.0
DINKOCH LEHRALAR Question District District <thdistrict< th=""> District District</thdistrict<>	DIN OCTVI BUTUALATE	50	0.044 IB	0.11 /B	0.22 0	0.057 18	0.17 IB	0.059 IB	0.11 18
DIREMENTIATE 0.17 U 0.19 U 0	DIRENZIA HIANTHRACENE	0.014 or MDI	0.19.11	0.2011	0.2211	0.35	0161	0 1911	0.076 1
DIMETRIA Discrete	DIETHVI PHTHAI ATE	7.1	0.1911	0.20 U	0 22 11	0.1811	0.191/	0.1917	0.1911
Display Display <thdisplay< th=""> <th< td=""><td>DIMETHYL PHTHALATE</td><td>2</td><td>0.191</td><td>0.20 U</td><td>0.22 U</td><td>0.18 U</td><td>0.191/</td><td>0.19 U</td><td>0.19 U</td></th<></thdisplay<>	DIMETHYL PHTHALATE	2	0.191	0.20 U	0.22 U	0.18 U	0.191/	0.19 U	0.19 U
FLUCRENE 50 0.19 U 0.20 U 0.22 U 0.084 J 0.19 U 0.19 U 0.19 U HEXACHLORO-1,3-BUTADIENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U HEXACHLOROE-1,3-BUTADIENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U HEXACHLOROEVENE 0.41 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U HEXACHLOROCYCLOPENTADIENE NS 0.57 U 0.59 U 0.65 U 0.54 U 0.56 U 0.56 U 0.58 U HEXACHLOROCTHANE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U INDENO[1,2,3-CDJPYRENE 3.2 0.19 U 0.20 U 0.22 U 0.66 0.31 0.19 U 0.19 U ISOPHORORNE 4.4 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSODI-N-PROPYLAMINE NS 0.19 U 0.20 U	FLUORANTHENE	50	0.079 J	0.20 U	0.14 J	1.9	0.70	0.19 U	0.35
HEXACHLORO-1,3-BUTADIENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U HEXACHLOROBENZENE 0.41 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U HEXACHLOROCYCLOPENTADIENE NS 0.57 U 0.59 U 0.65 U 0.54 U 0.56 U 0.56 U 0.58 U HEXACHLOROETHANE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U M-DICHLOROBENZENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U	FLUORENE	50	0.19 U	0.20 U	0.22 U	0.084 J	0.19 U	0.19 U	0.19 U
HEXACHLOROBENZENE 0.41 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U HEXACHLOROCYCLOPENTADIENE NS 0.57 U 0.59 U 0.65 U 0.54 U 0.56 U 0.56 U 0.58 U HEXACHLOROCYCLOPENTADIENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 0.19 U 0.20 U 0.22 U 0.66 0.31 0.19 U 0.19 U ISOPHORORNE 4.4 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U M-DICHLOROBENZENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U <t< td=""><td>HEXACHLORO-1,3-BUTADIENE</td><td>NS</td><td>0.19 U</td><td>0.20 U</td><td>0.22 U</td><td>0.18 U</td><td>0.19 U</td><td>0.19 U</td><td>0.19 U</td></t<>	HEXACHLORO-1,3-BUTADIENE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
HEXACHLOROCYCLOPENTADIENE NS 0.57 U 0.59 U 0.65 U 0.54 U 0.56 U 0.56 U 0.58 U HEXACHLOROETHANE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U	HEXACHLOROBENZENE	0.41	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
HEXACHLOROETHANE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U INDENO[1,2,3-CD]PYRENE 3.2 0.19 U 0.20 U 0.22 U 0.66 0.31 0.19 U 0.10 J ISOPHORORNE 4.4 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U M-DICHLOROBENZENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U	HEXACHLOROCYCLOPENTADIENE	NS	0.57 U	0.59 U	0.65 U	0.54 U	0.56 U	0.56 U	0.58 U
INDENO[1,2,3-CD]PYRENE 3.2 0.19 U 0.20 U 0.22 U 0.66 0.31 0.19 U 0.10 J ISOPHORORNE 4.4 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U M-DICHLOROBENZENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U N-NITROSODIPHENYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U	HEXACHLOROETHANE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
ISOPHORORNE 4.4 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U M-DICHLOROBENZENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U NNITROSODIPHENYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.42 U 0.18 U 0.19 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.42 U 0.18 U<	INDENO[1,2,3-CD]PYRENE	3.2	0.19 U	0.20 U	0.22 U	0.66	0.31	0.19 U	0.10 J
M-DICHLOROBENZENE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIPHENYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.56 0.18 U 0.19 U 0.19 U 0.19 U NTROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PHENATHRENE 50 0.19 U 0.20 U <td>ISOPHORORNE</td> <td>4.4</td> <td>0.19 U</td> <td>0.20 U</td> <td>0.22 U</td> <td>0.18 U</td> <td>0.19 U</td> <td>0.19 U</td> <td>0.19 U</td>	ISOPHORORNE	4.4	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
N-NITROSO-DI-N-PROPYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIPHENYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.56 0.18 U 0.19 U 0.19 U 0.19 U NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PHENANTHRENE 50 0.19 U 0.20 U 0.45 S 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U	M-DICHLOROBENZENE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
N-NITROSODIMETHYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U N-NITROSODIPHENYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.56 0.18 U 0.27 0.19 U 0.19 U NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PHENANTHRENE 50 0.19 U 0.20 U 0.45 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U PHENOL 0.03 or MDL 0.19 U 0.20 U 0.45 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U 0.20 U 0.23	N-NITROSO-DI-N-PROPYLAMINE	NS	0.19 U	0.20 U	0.22 Ŭ	0.18 U	0.19 U	0.19 U	0.19 U
N-NITROSODIPHENYLAMINE NS 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U NAPHTHALENE 13 0.19 U 0.20 U 0.56 0.18 U 0.27 0.19 U 0.18 J NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PHENANTHRENE 50 0.19 U 0.20 U 0.45 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U PYRENE 50 0.077 J 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U PYRENE 50 0.077 J 0.20 U 0.23 2.2 0.84 0.19 U 0.33 TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227<	N-NITROSODIMETHYLAMINE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
NAPHTHALENE 13 0.19 U 0.20 U 0.56 0.18 U 0.27 0.19 U 0.18 J NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.33 C PYRENE 500 0.475	N-NITROSODIPHENYLAMINE	NS	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
NITROBENZENE 0.200 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PHENANTHRENE 50 0.19 U 0.20 U 0.45 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PYRENE 50 0.077 J 0.20 U 0.23 2.2 0.84 0.19 U 0.33 TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227 2.806	NAPHTHALENE	13	0.19 U	0.20 U	0.56	0.18 U	0.27	0.19 U	0.18 J
PENTACHLOROPHENOL 1.0 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PHENANTHRENE 50 0.19 U 0.20 U 0.45 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PYRENE 50 0.077 J 0.20 U 0.23 2.2 0.84 0.19 U 0.33 TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227 2.806	NITROBENZENE	0.200 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
PHENANTHRENE 50 0.19 U 0.20 U 0.45 1.2 0.38 0.042 J 0.31 PHENOL 0.03 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PYRENE 50 0.077 J 0.20 U 0.23 2.2 0.84 0.19 U 0.33 TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227 2.806	PENTACHLOROPHENOL	1.0 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
PHENOL 0.03 or MDL 0.19 U 0.20 U 0.22 U 0.18 U 0.19 U 0.19 U 0.19 U PYRENE 50 0.077 J 0.20 U 0.23 2.2 0.84 0.19 U 0.33 TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227 2.806	PHENANTHRENE	50	0.19 U	0.20 U	0.45	1.2	0.38	0.042 J	0.31
PYRENE 50 0.077 J 0.20 U 0.23 2.2 0.84 0.19 U 0.33 TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227 2.806	PHENOL	0.03 or MDL	0.19 U	0.20 U	0.22 U	0.18 U	0.19 U	0.19 U	0.19 U
TOTAL SVOCs 500 0.475 1.449 2.315 14.72 6.357 0.227 2.806	PYRENE	50	0.077 J	0.20 U	0.23	2.2	0.84	0.19 U	0.33
	TOTAL SVOCs	500	0.475	1.449	2.315	14.72	6.357	0.227	2.806

U Undetectable Levels

NS No Standard

\$

الالآبيل مانده وروداؤن روبو المحارر فالمرتفة بالماج وها

MDL Method Detection Limit

waith mini and the contraction

~

T	Bassan dad			DO DU L OL	DO FRI C		DO 1741 40	
Location	Recommended	PG-FILL-04	PG-FILL-04	rG-FILL-04	PG-FILL-5	PG-FILL-5	PG-FILL-10	PG-FILL-10
Sample Date	Soil	11/6/2000	11/6/2000	11/6/2000	11/18/2000	11/18/2000	12/1/2000	12/1/2000
Sample ID	Cleanup	PGFILL04	PGFILL04	PGFILL04	PG-FILL-5	PG-FILL-5	PG-FILL10	PG-FILL10
Sample Depth	Objective	2-4'	4-6'	7-8'	2-4'	6-8'	3-4'	6.4-8.2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,2,4-TRICHLOROBENZENE	3.4	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
1,2-BENZPHENANTHRACENE	NS	0.19 U	0.12 J	0.21 U	0.18 U	0.19 U	0.23	0.43
1,2-DICHLOROBENZENE	7.9	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
1,2-DIPHENYLHYDRAZINE	NS	0.037 U	0.038 U	0.042 U	0.036 U	0.039 U	0.037 U	0.079 U
1,4-DICHLOROBENZENE	8.5	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2,4,6-TRICHLORORPHENOL	0.1	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2,4-DICHLOROPHENOL	0.4	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2,4-DIMETHYLPHENOL	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2,4-DINITROPHENOL	0.200 or MDL	0.37 U	0.38 U	0.42 U	0.36 U	0.39 U	0.37 U	0.79 U
2,4-DINITROTOLUENE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2,6-DINITROTOLUENE	1	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2-CHLORORNAPHTHALENE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2-CHLOROPHENOL	0.8	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
2-NITROPHENOL	0.330 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
3,3'-DICHLOROBENZIDINE	N/A	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
4,6-DINITRO-O-CRESOL	0.100 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
4-BROMOPHENYLPHENYL ETHER	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
4-CHLORO-3-METHYLPHENOL	0.240 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
4-CHLOROPHENYLPHENYL ETHER	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
4-NITROPHENOL	0.100 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
ACENAPHTHENE	50	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.14 J
ACENAPHTHYLENE	41	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.069 J	0.40 U
ANTHRACENE	50	0.19 U	0.041 J	0.21 U	0.18 U	0.19 U	0.054 J	0.17 J
BENZIDINE	NS	0.37 U	0.38 U	0.42 U	0.36 U	0.39 U	0.37 U	0.79 U
BENZO[A]ANTHRACENE	0.224 or MDL	0.19 U	0.082 J	0.21 U	0.18 U	0.19 U	0.24	0,42
BENZO[A]PYRENE	0.061 or MDL	0.19 U	0.076 J	0.081.J	0.18 U	0.19 U	0.17 J	0,34 J
BENZO[B]FLOURANTHENE	1.1	0.19 U	0.11 J	0.21 U	0.18 U	0.19 U	0.18 J	0.62
BENZO[G,H,I]PERYLENE	50	0.19 U	0.045 J	0.21 U	0.18 U	0.19 U	0.11 J	0.11 J
BENZO[K]FLOURANTHENE	1.1	0.19 U	0.054 J	0.21 U	0.18 U	0.19 U	0.063 J	0.30 J
BENZYL BUTYL PHTHALATE	50	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
BIS(2-CHLOROETHOXY)METHANE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
BIS(2-CHLOROETHYL)ETHER	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
BIS(2-ETHYHEXYL)PHTHALATE	50	0.19 U	0.19 U	0.055 J	0.041 J	0.19 U	0.070 JB	0.27 JB
DI-N-BUTYL PHTHALATE	8.1	0.19 U	0.10 J	0.14 J	0.18 U	0.19 U	0.050 J	0.18 J
DI-N-OCTYL PHTHALATE	50	0.19 B	0.14 JB	0.10 JB	0.18 U	0.19 U	0.037 J	0.15 J
DIBENZ[A,H]ANTHRACENE	0.014 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
DIETHYL PHTHALATE	7.1	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
DIMETHYL PHTHALATE	2	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
FLUORANTHENE	50	0.19 U	0.15 J	0.21 U	0.18 U	0.19 U	0.35	0.67
FLUORENE	50	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.15 J
HEXACHLORO-1,3-BUTADIENE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
HEXACHLOROBENZENE	0.41	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
HEXACHLOROCYCLOPENTADIENE	NS	0.56 U	0.57 U	0.62 U	0.54 U	0.58 U	0.55 U	1.2 U
HEXACHLOROETHANE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
INDENO[1,2,3-CD]PYRENE	3.2	0.19 U	0.039 J	0.21 U	0.18 U	0.19 U	0.078 J	0.11 J
ISOPHORORNE	4.4	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
M-DICHLOROBENZENE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
N-NITROSO-DI-N-PROPYLAMINE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
N-NITROSODIMETHYLAMINE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
N-NITROSODIPHENYLAMINE	NS	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
NAPHTHALENE	13	0.19 U	0.050 J	0.21 U	0.18 U	0.19 U	0.18 U	1.9
NITROBENZENE	0.200 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
PENTACHLOROPHENOL	1.0 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
PHENANTHRENE	50	0.19 U	0.21	0.21 U	0.18 U	0.19 U	0.18 J	0.58
PHENOL	0.03 or MDL	0.19 U	0.19 U	0.21 U	0.18 U	0.19 U	0.18 U	0.40 U
PYRENE	50	0.19 U	0.13 J	0.21 U	0.037 J	0.19 U	0.62	0.58
TOTAL SVOCs	500	0.19	1.347	0.376	0.078	ND	2.501	7.12

U Undetectable Levels

*

NS No Standard

MDL Method Detection Limit

1

Pro deserve

ž,

digrania

Location	Recommended	PG-UST1-2	PG-UST1-2	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1
Sample Date	Soil	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/28/2000
Sample ID	Cleanup	PG-UST1-2	PG-UST1-2	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1
Sample Denth	Objective	2-4'	12-14	2-4'	10-11.5	14-15.5	4-6'	12-14	8-10
Concentration	MG/KG	MG/KG	мажа	MG/KG	MG/KG	MG/KG	мажа	MG/KG	MG/KG
1.2 4-TRICHLOROBENZENE	34	0.18.U	0 20 11	0 21 11	0.2311	0.22 []	0.22 []	0.2511	0.2011
1 2-BENZPHENANTHRACENE	NS	0.74	0.161	0.29	0.23 U	0.22 U	0.18 J	0.34	0.10 J
1.2-DICHLOROBENZENE	7.9	0.18 U	0.2011	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
1.2-DIPHENVLHYDRAZINE	NS	0.03611	0.040 11	0.042 U	0.0461	0.043 U	0.044 U	0.050 U	0.039 U
1 4-DICHLOROBENZENE	8.5	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
2.4.6-TRICHLORORPHENOL	0.1	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
2.4-DICHLOROPHENOL	0.4	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.097 J	0.25 U	0.20 U
2.4-DIMETHYLPHENOL	NS	0.18 U	0.059 J	0.21 U	0.23 U	0.22 U	0.19 J	0.25 U	0.20 U
2.4-DINITROPHENOL	0.200 or MDL	0.36 U	0.40 U	0.42 U	0.46 U	0.43 U	0.44 U	0.50 U	0.39 U
2.4-DINITROTOLUENE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
2.6-DINITROTOLUENE	1	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
2-CHLORORNAPHTHALENE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
2-CHLOROPHENOL	0.8	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.062 J	0.25 U	0.20 U
2-NITROPHENOL	0.330 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
3,3'-DICHLOROBENZIDINE	N/A	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
4,6-DINITRO-O-CRESOL	0.100 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
4-BROMOPHENYLPHENYL ETHER	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
4-CHLORO-3-METHYLPHENOL	0.240 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
4-CHLOROPHENYLPHENYL ETHER	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
4-NITROPHENOL	0.100 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
ACENAPHTHENE	50	0.19	0.098 J	0.44	0.16 J	0.22 U	0.15 J	0.15 J	0.063 J
ACENAPHTHYLENE	41	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.050 J	0.25 U	0.20 U
ANTHRACENE	50	0.56	0.048 J	0.42	0.23 U	0.22 U	0.066 J	0.14 J	0.20 U
BENZIDINE	NS	0.36 U	0.40 U	0.42 U	0.46 U	0.43 U	0.44 U	0.50 U	0.39 U
BENZOJAJANTHRACENE	0.224 or MDL	0.87	0.077 J	0.25	0.063 J	0.22 U	0.099 J	0.39	0.065 J
BENZO[A]PYRENE	0.061 or MDL	0.58	0.056 J	0.18 J	0.23 U	0.22 U	0.063 J	0.29	0.20 U
BENZO[B]FLOURANTHENE	1.1	0.80	0.11 J	0.28	0.054 J	0.22 U	0.12 J	0.40	0.20 U
BENZO[G,H,I]PERYLENE	50	0.33	0.043 J	0.11 J	0.23 U	0.22 U	0.070 J	0.10 J	0.20 U
BENZOKIFLOURANTHENE	1.1	0.33	0.20 U	0.084 J	0.23 U	0.22 U	0.22 U	0.16 J	0.20 U
BENZYL BUTYL PHTHALATE	50	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
BIS(2-CHLOROETHOXY)METHANE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
BIS(2-CHLOROETHYL)ETHER	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
BIS(2-ETHYHEXYL)PHTHALATE	50	0.067 JB	0.24 B	0.073 JB	0.17 JB	0.15 JB	0.28 B	0.23 JB	0.11 J
DI-N-BUTYL PHTHALATE	8.1	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.065 JB	0.20 U
DI-N-OCTYL PHTHALATE	50	0.18 U	0.045 J	0.21 U	0.073 J	0.054 J	0.055 J	0.14 J	0.047 J
DIBENZ[A,H]ANTHRACENE	0.014 or MDL	0.041 J	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
DIETHYL PHTHALATE	7.1	0.18 U	0.041 J	0.21 U	0.23 U	0.22 U	0.064 J	0.25 U	0.20 U
DIMETHYL PHTHALATE	2	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
FLUORANTHENE	50	2.2	0.21	1.0	0.32	0.22 U	0.23	0.71	0.064 J
FLUORENE	50	0.26	0.083 J	0.32	0.087 J	0.22 U	0.12 J	0.13 J	0.20 U
HEXACHLORO-1,3-BUTADIENE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
HEXACHLOROBENZENE	0.41	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
HEXACHLOROCYCLOPENTADIENE	NS	0.54 U	0.60 U	0.63 U	0.68 U	0.65 U	0.67 U	0.75 U	0.59 U
HEXACHLOROETHANE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
INDENO[1,2,3-CD]PYRENE	3.2	0.33	0.041 J	0.097 J	0.23 U	0.22 U	0.061 J	0.086 J	0.20 U
ISOPHORORNE	4.4	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.052 J	0.25 U	0.20 U
M-DICHLOROBENZENE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
N-NITROSO-DI-N-PROPYLAMINE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
N-NITROSODIMETHYLAMINE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
N-NITROSODIPHENYLAMINE	NS	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
NAPHTHALENE	13	0.11 J	0.51	0.28	0.21 J	0.22 U	0.61	0.29	0.20 U
NITROBENZENE	0.200 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.22 U	0.22 U	0.25 U	0.20 U
PENTACHLOROPHENOL	1.0 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.081 J	0.22 U	0.25 U	0.040 J
PHENANTHRENE	50	2.4	0.24	0.96	0.22 J	0.22 U	0.28	0.60	0.16 J
PHENOL	0.03 or MDL	0.18 U	0.20 U	0.21 U	0.23 U	0.047 J	0.14 J	0.25 U	0.20 U
PYRENE	50	1.6	0.15 J	0.72	0.22 J	0.22 U	0.20 J	0.63	0.28
TOTAL SVOCs	500	11.408	2.211	5.504	1.577	0.332	3.239	4.851	0.929
U Undetectable Levels									

NS No Standard

-

MDL Method Detection Limit

_ .

South States

*

.

ų,

	n	DC LISTE LA	DC VOTE ID	DO NOTE A	
Location	Recommended	PG-UST/-IA	PG-051/-1B	PG-0517-2	PG-US17-2
Sample Date	Soll	11/28/2000	11/28/2000	11/21/2000	11/21/2000
Sample ID	Cleanup	PG-UST7-IA	PG-UST7-1B	PG-UST7-2	PG-UST7-2
Sample Depth	Objective	0-2'	2-3.5	8-10'	10-12'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
1,2,4-TRICHLOROBENZENE	3.4	1.8 U	1.9 U	0.19 U	0.20 U
1,2-BENZPHENANTHRACENE	NS	0.82 J	2.7	0.31	0.61
1,2-DICHLOROBENZENE	7.9	1.8 U	1.9 U	0.19 U	0.20 U
1,2-DIPHENYLHYDRAZINE	NS	0.36 U	0.38 U	0.038 U	0.040 U
1,4-DICHLOROBENZENE	8.5	1.8 U	1.9 U	0.19 U	0.20 U
2,4,6-TRICHLORORPHENOL	0.1	1.8 U	1.9 U	0.19 U	0.20 U
2,4-DICHLOROPHENOL	0.4	1.8 U	1.9 U	0.19 U	0.20 U
2,4-DIMETHYLPHENOL	NS	1.8 U	1.9 U	0.19 U	0.20 U
2,4-DINITROPHENOL	0.200 or MDL	3.6 U	3.8 U	0.38 U	0.40 U
2,4-DINITROTOLUENE	NS	1.8 U	1.9 U	0.19 U	0.20 U
2,6-DINITROTOLUENE	1	1.8 0	1.9 0	0.19 U	0.20 U
2-CHLORORNAPHTHALENE	NS	1.8 U	1.9 U	0.19 U	0.20 U
2-CHLOROPHENOL	0.8	1.8 U	1.9 U	0.19 U	0.20 U
2-NITROPHENOL	0.330 OF MDL	1.8 U	1.9 U	0.19 U	0.20 U
3,3'-DICHLOROBENZIDINE	N/A	1.8 U	1.9 U	0.19 U	0.20 U
4,6-DINITRO-O-CRESOL	0.100 or MDL	1.8 U	1.9 U	0.19 U	0.20 U
4-BROMOPHENYLPHENYL ETHER	NS	1.8 U	1.9 U	0.19 U	0.20 U
4-CHLORO-3-METHYLPHENOL	0.240 OF MDL	1.8 U	1.9 U	0.19 U	0.20 U
4-CHLOROPHENYLPHENYL ETHER	NS	1.8 U	1.9 U	0.19 U	0.20 U
4-NITROPHENOL	0.100 OF MDL	1.8 U	1.9 U	0.19 U	0.20 U
ACENAPHTHENE	50	1.8 0	0.45 J	0.26	0.27
ACENAPHTHYLENE	41	1.8 U	1.9 U	0.19 U	0.20 U
ANTHRACENE	50	1.8 U	1.0 J	0.60	0.49
BENZIDINE	NS 0.774 av MDI	3.6 U	3.8 U	0.38 U	0.40 U
BENZOLAJANTHRACENE	0.224 OF MDL	0.43.3	2.5	0.32	0.4
BENZO(AJPYKENE	1 1	1.8 U	1,1 J	0.19	0.28
BENZO(BIFLOUKANTHENE	50	0.05 J	1.2.3	0.12 J	0.10 J
BENZOICHELOURANTHENE	1 1	0.70 1	0.80 1	0.12 J	0.18 J
DENZUL BUTVI BUTUALATE	50	1.911	1 0 1 1	0.10.13	0.0751
BENZTE BOTTE THIMALATE	NS	1 9 11	1.50	0.19 U	0.20 U
BIS(2-CHLOROFTHYL)ETHER	NS	1.811	1911	0.191	0.20 U
BIS(2-CHLOROISOPROPYL)ETHER	NS	1.811	1911	0.191	0.20 0
BIS(2-ETHYHEXYI)PHTHALATE	50	1.811	1911	0.191	0.20 C
DI-N-BUTYL PHTHALATE	8.1	1.8 0	191	0.191	0.11 /B
DI-N-OCTYL PHTHALATE	50	1.80	1911	0 19 11	0.057.1
DIBENZIA HIANTHRACENE	0.014 or MDL	1811	1911	0 19 11	0 20 11
DIETHYL PHTHALATE	7.1	181	1.911	0 19 11	0 20 11
DIMETHYL PHTHALATE	2	1.8 U	1.91	0.19 U	0.20 U
FLUORANTHENE	50	1.8 U	0.69 J	0.25	0.27
FLUORENE	50	1.8 U	0.86 J	0.38	0.20 U
HEXACHLORO-1.3-BUTADIENE	NS	1.8 U	1.9 U	0.19 U	0.20 U
HEXACHLOROBENZENE	0.41	1.8 U	1.9 U	0.19 U	0.20 U
HEXACHLOROCYCLOPENTADIENE	NS	5.4 U	5.7 U	0.57 U	0.60 U
HEXACHLOROETHANE	NS	1.8 U	1.9 U	0.19 U	0.20 U
INDENO[1,2,3-CD]PYRENE	3.2	1.8 U	1.9 U	0.044 J	0.050 J
ISOPHORORNE	4.4	1.8 U	1.9 U	0.19 U	0.20 U
M-DICHLOROBENZENE	NS	1.8 U	1.9 U	0.19 U	0.20 U
N-NITROSO-DI-N-PROPYLAMINE	NS	1.8 U	1.9 U	0.19 U	0.20 U
N-NITROSODIMETHYLAMINE	NS	1.8 U	1.9 U	0.19 U	0.20 U
N-NITROSODIPHENYLAMINE	NS	1.8 U	1.9 U	0.19 U	0.20 U
NAPHTHALENE	13	1.8 U	1.9 U	0.19 U	0.056 J
NITROBENZENE	0.200 or MDL	1.8 U	1.9 U	0.19 U	0.20 U
PENTACHLOROPHENOL	1.0 or MDL	1.8 U	1.9 U	0.19 U	0.20 U
PHENANTHRENE	50	1.8 U	1.4 J	1.4	0.20 U
PHENOL	0.03 or MDL	1.8 U	1.9 U	0.19 U	0.20 U
PYRENE	50	0.59 J	6.1	1.8	2.9
TOTAL SVOCs	500	3.78	18.69	5.86	6.268

U Undetectable Levels

NS No Standard

8

MDL Method Detection Limit



Location	Recommended	PG-A-4	PG-A-4	PG-A-5	PG-B-1	PG-B-1	PG-B-1	PG-B-02	PG-B-02
Sample Date	Soil	11/16/2000	11/16/2000	11/14/2000	12/4/2000	12/4/2000	12/4/2000	11/16/2000	11/16/2000
Sample ID	Cleanup	PG-A-04	PG-A-04	PG-A-05	PG-B-01	PG-B-01	PG-B-01	PG-B-02	PG-A-02
Sample Depth	Objective	6-8'	12-14'	2-4'	2-4'	6-8'	9-10'	2-4'	6-8'
Concentration	мд/кд	MG/KG	MG/KĠ	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
4,4'-DDE	2.1	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.35	0.038	0.0039 U	0.0046 U
4,4'-DDT	2.1	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.042	0.0039 U	0.0046 U
ALDRIN	0.041	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
ALPHA-BHC	0.11	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
BETA-BHC	0.2	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
CHLORDANE	0.54	0.0091 U	0.014 U	0.0082 U	0.0076 U	0.04 U	0.0084 U	0.0078 U	0.0093 U
DELTA-BHC	0.3	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
DIELDRIN	0.044	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.3	0.0042 U	0.0039 U	0.0046 U
ENDOSULFAN I	0.9	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.00 4 2 U	0.0039 U	0.0046 U
ENDOSULFAN II	0.9	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
ENDOSULFAN SULFATE	1	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
ENDRIN	0.1	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.39	0.00 42 U	0.0039 U	0.0046 U
ENDRIN ALDEHYDE	NS	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.087	0.0042 U	0.0039 U	0.0046 U
ENDRIN KETONE	N/A	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
GAMMA-BHC (LINDANE)	0.06	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
HEPTACHLOR	0.1	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.091	0.0042 U	0.0039 U	0.0046 U
HEPTACHLOR EPOXIDE	0.02	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.069	0.0042 U	0.0039 U	0.0046 U
METHOXYCHLOR	NS	0.0046 U	0.0071 U	0.0041 U	0.0038 U	0.02 U	0.0042 U	0.0039 U	0.0046 U
TOXAPHENE	NS	0.046 U	0.071 U	0.041 U	0.038 U	0.2 U	0.042 U	0.039 U	0.046 U
AROCLOR 1016	NS	0.023 U	0.035 U	0.021 U	0.019 U	0.4 U	0.021 U	0.02 U	0.023 U
AROCLOR 1221	NS	0.023 U	0.035 U	0.021 U	0.019 U	0.4 U	0.021 U	0.02 U	0.023 U
AROCLOR 1232	NS	0.023 U	0.035 U	0.021 U	0.019 U	0.4 U	0.021 U	0.02 U	0.023 U
AROCLOR 1242	NS	0.023 U	0.035 U	0.021 U	0.019 U	0.4 U	0.021 U	0.02 U	0.023 U
AROCLOR 1248	NS	0.023 U	0.035 U	0.021 U	0.019 U	0.4 U	0.021 U	0.02 U	0.023 U
AROCLOR 1254	NS	0.023 U	0.035 U	0.021 U	0.019 U	6.6	0.021 U	0.02 U	0.023 U
AROCLOR 1260	NS	0.023 U	0.035 U	0.021 U	0.019 U	0.4 U	0.29	0.02 U	0.023 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	ND	6.6	0.29	ND	ND

U Undetectable Levels

••••••••

~**

ND Not Detectived

NS No Standard



Location	Recommended	PG-B-02A	PG-B-3	PG-B-3	PG-B-4	PG-E-1	PG-E-1	PG-E-1
Sample Date	Soil	11/16/2000	12/4/2000	12/4/2000	12/4/2000	11/13/2000	11/13/2000	11/13/2000
Sample ID	Cleanup	PG-B-02A	PG-B-03	PG-B-03 [,]	PG-B-04	PG-E-01	PG-E-01	PG-E-01
Sample Depth	Objective	8-10'	2-4'	6-8'	2-4'	0.2-2'	4-6'	10-12'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
4,4'-DDE	2.1	0.0044 U	0.11	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
4.4'-DDT	2.1	0.0044 U	0.023 U	0.0055 U	0.025	0.0038 U	0.0039 U	0.0063 U
ALDRIN	0.041	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
ALPHA-BHC	0.11	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
BETA-BHC	0.2	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
CHLORDANE	0.54	0.0088 U	0.046 U	0.011 U	0.039	0.0076 U	0.0078 U	0.013 U
DELTA-BHC	0.3	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
DIELDRIN	0.044	0.0044 U	0.13	0.0055 U	0.0067	0.0038 U	0.0039 U	0.0063 U
ENDOSULFAN I	0.9	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
ENDOSULFAN II	0.9	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
ENDOSULFAN SULFATE	1	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
ENDRIN	0.1	0.0044 U	0.14	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
ENDRIN ALDEHYDE	NS	0.0044 U	0.23	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
ENDRIN KETONE	N/A	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
GAMMA-BHC (LINDANE)	0.06	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
HEPTACHLOR	0.1	0.0044 U	0.023 U	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
HEPTACHLOR EPOXIDE	0.02	0.0044 U	0.029	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
METHOXYCHLOR	NS	0.0044 U	0.06	0.0055 U	0.0039 U	0.0038 U	0.0039 U	0.0063 U
TOXAPHENE	NS	0.044 U	0.23 U	0.055 U	0.039 U	0.038 U	0.039 U	0.063 U
AROCLOR 1016	NS	0.022 U	0.23 U	0.027 U	0.02 U	0.019 U	0.019 U	0.031 U
AROCLOR 1221	NS	0.022 U	0.23 U	0.027 U	0.02 U	0.019 U	0.019 U	0.031 U
AROCLOR 1232	NS	0.022 U	0.23 U	0.027 U	0.02 U	0.019 U	0.019 U	0.031 U
AROCLOR 1242	NS	0.022 U	0.23 U	0.027 U	0.02 U	0.019 U	0.019 U	0.031 U
AROCLOR 1248	NS	0.022 U	0.23 U	0.027 U	0.02 U	0.019 U	0.019 U	0.031 U
AROCLOR 1254	NS	0.022 U	2.4	0.027 U	0.02 U	0.019 U	0.019 U	0.031 U
AROCLOR 1260	NS	0.022 U	0.23 U	0.027 U	0.24	0.019 U	0.019 U	0.031 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	2.4	ND	0.24	ND	ND	ND

U Undetectable Levels

ND Not Detectived

NS No Standard

Location	Recommended	PG-F2-2	PG-F2-2	PG-P-1	PG-P-1	PG-P-2	PG-P-2	PG-P-3
Sample Date	Soil	11/20/2000	11/20/2000	11/22/2000	11/22/2000	11/30/2000	11/30/2000	11/22/2000
Sample ID	Cleanup	PG-F2-2	PG-F2-2	PG-P-01	PG-P-01	PG-P-02	PG-P-02	PG-P-03
Sample Depth	Objective	2-4'	8-10'	2-4'	8-10'	2-4'	4-6'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
4,4'-DDE	2.1	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
4,4'-DDT	2.1	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ALDRIN	0.041	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ALPHA-BHC	0.11	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
BETA-BHC	0.2	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
CHLORDANE	0.54	0.0072 U	0.0078 U	0.0072 U	0.0084 U	0.0078 U	0.0079 U	0.041 U
DELTA-BHC	0.3	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
DIELDRIN	0.044	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ENDOSULFAN I	0.9	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ENDOSULFAN II	0.9	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ENDOSULFAN SULFATE	1	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ENDRIN	0.1	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ENDRIN ALDEHYDE	NS	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
ENDRIN KETONE	N/A	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
GAMMA-BHC (LINDANE)	0.06	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
HEPTACHLOR	0.1	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
HEPTACHLOR EPOXIDE	0.02	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
METHOXYCHLOR	NS	0.0036 U	0.0039 U	0.0036 U	0.0042 U	0.0039 U	0.004 U	0.021 U
TOXAPHENE	NS	0.036 U	0.039 U	0.036 U	0.042 U	0.039 U	0.04 U	0.21 U
AROCLOR 1016	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
AROCLOR 1221	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
AROCLOR 1232	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
AROCLOR 1242	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
AROCLOR 1248	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
AROCLOR 1254	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
AROCLOR 1260	NS	0.018 U	0.019 U	0.018 U	0.021 U	0.019 U	0.02 U	0.021 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND						

U Undetectable Levels

ware the second of the second

000001500

ND Not Detectived

NS No Standard

110

- and a man the second second

Location	Recommended	PG-P-3	PG-PD-1	PG-PD-1	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5
Sample Date	Soil	11/22/2000	11/21/2000	11/21/2000	11/29/2000	12/2/2000	12/2/2000	12/2/2000
Sample ID	Cleanup	PG-P-03	PG-PD-01	PG-PD-01	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5
Sample Depth	Objective	6-8'	2-4'	10-12'	4-6'	8-10'	0.7-1.5'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
4,4'-DDE	2.1	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
4,4'-DDT	2.1	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0054	0.0037 U
ALDRIN	0.041	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
АLРНА-ВНС	0.11	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
BETA-BHC	0.2	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
CHLORDANE	0.54	0.0081 U	0.0075 U	0.0084 U	0.041 U	0.0081 U	0.0076 U	0.0075 U
DELTA-BHC	0.3	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
DIELDRIN	0.044	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
ENDOSULFAN I	0.9	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
ENDOSULFAN II	0.9	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
ENDOSULFAN SULFATE	1	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
ENDRIN	0.1	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
ENDRIN ALDEHYDE	NS	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
ENDRIN KETONE	N/A	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
GAMMA-BHC (LINDANE)	0.06	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
HEPTACHLOR	0.1	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
HEPTACHLOR EPOXIDE	0.02	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
METHOXYCHLOR	NS	0.0041 U	0.0037 U	0.0042 U	0.021 U	0.0041 U	0.0038 U	0.0037 U
TOXAPHENE	NS	0.041 U	0.037 U	0.042 U	0.21 U	0.041 U	0.038 U	0.037 U
AROCLOR 1016	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.019 U	0.019 U
AROCLOR 1221	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.019 U	0.019 U
AROCLOR 1232	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.019 U	0.019 U
AROCLOR 1242	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.019 U	0.019 U
AROCLOR 1248	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.019 U	0.019 U
AROCLOR 1254	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.019 U	0.019 U
AROCLOR 1260	NS	0.02 U	0.019 U	0.021 U	0.021 U	0.02 U	0.036	0.019 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	ND	ND	0.036	ND

U Undetectable Levels

*1969

ND Not Detectived

NS No Standard

1

Location	Recommended	PG-Q1-1	PG-Q1-1	PG-RR-1	PG-RR-1	PG-RR-2	PG-RR-3	PG-RR-4
Sample Date	Soil	11/30/2000	11/30/2000	11/3/2000	11/3/2000	11/4/2000	11/3/2000	11/4/2000
Sample ID	Cleanup	PG-Q1-1	PG-Q1-1	PG-RR-01	PG-RR-01	PG-RR-02	PG-RR-03	PG-RR-04
Sample Depth	Objective	2-4'	4-6'	0-1.2'	1.2-2'	0-2'	1.5-2'	0.6-2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
· .								
4,4'-DDD	2.9	0.0038 U	0.026 U	0.078	0.0038 U	0.0039 U	0.019 U	0.0037 U
4,4'-DDE	2.1	0.0052	0.026 U	0.078	0.012	0.017	0.019 U	0.0037 U
4,4'-DDT	2.1	0.0038 U	0.026 U	0.21	0.026	0.0039 U	0.039	0.0037 U
ALDRIN	0.041	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ALPHA-BHC	0.11	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
BETA-BHC	0.2	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
CHLORDANE	0.54	0.0076 U	0.051 U	0.039 U	0.018	0.0078 U	0.039 U	0.0075 U
DELTA-BHC	0.3	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
DIELDRIN	0.044	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ENDOSULFAN I	0.9	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ENDOSULFAN II	0.9	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ENDOSULFAN SULFATE	1	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ENDRIN	0.1	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ENDRIN ALDEHYDE	NS	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
ENDRIN KETONE	N/A	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.05	0.0037 U
GAMMA-BHC (LINDANE)	0.06	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0045	0.029	0.0037 U
HEPTACHLOR	0.1	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
HEPTACHLOR EPOXIDE	0.02	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
METHOXYCHLOR	NS	0.0038 U	0.026 U	0.019 U	0.0038 U	0.0039 U	0.019 U	0.0037 U
TOXAPHENE	NS	0.038 U	0.26 U	0.19 U	0.038 U	0.039 U	0.19 U	0.037 U
AROCLOR 1016	NS	0.019 U	0.026 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U
AROCLOR 1221	NS	0.019 U	0.026 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U
AROCLOR 1232	NS	0.019 U	0.026 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U
AROCLOR 1242	NS	0.019 U	0.026 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U
AROCLOR 1248	NS	0.019 U	0.026 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U
AROCLOR 1254	NS	0.019 U	0.026 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U
AROCLOR 1260	NS	0.019 U	0.026 U	0.13	0.019 U	0.02 U	0.019 U	0.019 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	0.13	ND	ND	ND	ND

U Undetectable Levels

~%®

ND Not Detectived

NS No Standard

200 A 1996 (2)



Location	Recommended	PG-RR-4	PG-RR-5	PG-RR-5	PG-RR-6	PG-RR-6	PG-RR-7	PG-RR-7
Sample Date	Soil	11/4/2000	11/7/2000	11/7/2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PG-RR-04	PG-RR-05	PG-RR-05	PG-RR-06	PG-RR-06	PG-RR-07	PG-RR-07
Sample Depth	Objective	3-4'	1-2'	2-4'	1-2'	2-4'	0-2'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0042 U	0.0036 U					
4,4'-DDE	2.1	0.0042 U	0.0036 U					
4,4'-DDT	2.1	0.0042 U	0.0036 U					
ALDRIN	0.041	0.0042 U	0.0036 U					
ALPHA-BHC	0.11	0.0042 U	0.0036 U					
BETA-BHC	0.2	0.0042 U	0.0036 U					
CHLORDANE	0.54	0.0083 U	0.0072 U					
DELTA-BHC	0.3	0.0042 U	0.0036 U					
DIELDRIN	0.044	0.0042 U	0.0036 U					
ENDOSULFAN I	0.9	0.0042 U	0.0036 U					
ENDOSULFAN II	0.9	0.0042 U	0.0036 U					
ENDOSULFAN SULFATE	1	0.0042 U	0.0036 U					
ENDRIN	0.1	0.0042 U	0.0036 U					
ENDRIN ALDEHYDE	NS	0.0042 U	0.0036 U					
ENDRIN KETONE	N/A	0.0042 U	0.0036 U					
GAMMA-BHC (LINDANE)	0.06	0.0042 U	0.0036 U					
HEPTACHLOR	0.1	0.0042 U	0.0036 U					
HEPTACHLOR EPOXIDE	0.02	0.0042 U	0.0036 U					
METHOXYCHLOR	NS	0.0042 U	0.0036 U					
TOXAPHENE	NS	0.042 U	0.036 U					
AROCLOR 1016	NS	0.021 U	0.018 U					
AROCLOR 1221	NS	0.021 U	0.018 U					
AROCLOR 1232	NS	0.021 U	0.018 U					
AROCLOR 1242	NS	0.021 U	0.018 U					
AROCLOR 1248	NS	0.021 U	0.018 U					
AROCLOR 1254	NS	0.021 U	0.018 U					
AROCLOR 1260	NS	0.021 U	0.018 U					
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND						

.

U Undetectable Levels

10000 10000 CONSTRUCTION

and the second state of th

19**19***

ND Not Detectived

NS No Standard



Location	Recommended	PG-STAIN-02	PG-STAIN-02	PG-STAIN-03	PG-STAIN-03	PG-STAIN-3B	PG-STAIN-3B	PG-FS-2
Sample Date	Soil	11/11/2000	11/11/2000	11/10/2000	11/10/2000	11/11/2000	11/11/2000	11/17/2000
Sample ID	Cleanup	PG-ST-02	PG-ST-02	PG-ST-03	PG-ST-03	PG-ST-3B	PG-ST-3B	PG-FS-02
Sample Depth	Objective	1-2'	2-3'	1.7-2.5'	2.5-3.5'	0-2'	2-4'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
4,4'-DDE	2.1	0.0039 U	0.004 U	0.078 U	0.01	0.0089	0.0036 U	0.0038 U
4,4'-DDT	2.1	0.0039 U	0.004 U	0.078 U	0.0042 U	0.006	0.0036 U	0.0038 U
ALDRIN	0.041	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
ALPHA-BHC	0.11	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
BETA-BHC	0.2	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
CHLORDANE	0.54	0.0078 U	0.008 U	4.5	0.082	0.0072 U	0.0072 U	0.0076 U
DELTA-BHC	0.3	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
DIELDRIN	0.044	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0092	0.0036 U	0.0038 U
ENDOSULFAN I	0.9	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
ENDOSULFAN II	0.9	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
ENDOSULFAN SULFATE	. 1	0.0039 U	0.004 U	0.62	0.0042 U	0.0036 U	0.0036 U	0.0038 U
ENDRIN	0.1	0.0039 U	0.004 U	0.078 U	0.03	0.023	0.0036 U	0.0038 U
ENDRIN ALDEHYDE	NS	0.0039 U	0.004 U	0.078 U	0.0064	0.0055	0.0036 U	0.0038 U
ENDRIN KETONE	N/A	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
GAMMA-BHC (LINDANE)	0.06	0.0039 U	0.004 U	0.078 U	0.0053	0.0042	0.0036 U	0.0038 U
HEPTACHLOR	0.1	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
HEPTACHLOR EPOXIDE	0.02	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
METHOXYCHLOR	NS	0.0039 U	0.004 U	0.078 U	0.0042 U	0.0036 U	0.0036 U	0.0038 U
TOXAPHENE	NS	0.039 U	0.04 U	0.78 U	0.042 U	0.036 U	0.036 U	0.038 U
AROCLOR 1016	NS	0.02 U	0.02 U	0.097 U	0.021 U	0.018 U	0.018 U	0.019 U
AROCLOR 1221	NS	0.02 U	0.02 U	0.097 U	0.021 U	0.018 U	0.018 U	0.019 U
AROCLOR 1232	NS	0. 02 U	0.02 U	0.097 U	0.021 U	0.018 U	0.018 U	0.019 U
AROCLOR 1242	NS	0.0 2 U	0.02 U	0.097 U	0.021 U	0.018 U	0.018 U	0.019 U
AROCLOR 1248	NS	0.02 U	0.02 U	0.097 U	0.021 U	0.018 U	0.018 U	0.019 U
AROCLOR 1254	NS	0.02 U	0.02 U	0.097 U	0.021 U	0.018 U	0.018 U	0.019 U
AROCLOR 1260	NS	0.02 U	0.02 U	0.097 U	0.021 U	0.17	0.018 U	0.019 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	ND	0.17	ND	ND

U Undetectable Levels

-968

ND Not Detectived

NS No Standard

~###



Table 5C Soil Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-FS-2	PG-FS-2	PG-FS-3	PG-FS-3	PG-FILL-02	PG-FILL-03	PG-FILL-03	PG-FILL-03	PG-FILL-04
Sample Date	Soit	11/17/2000	11/17/2000	11/15/2000	11/15/2000	11/3/2000	11/4/2000	11/6/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PG-FS-02	PG-FS-02	PG-FS03	PG-FS03	PGFILL02	PGFILL03	PGFILL03	PGFILL03	PGFILL04
Sample Depth	Objective	8-10'	17-18'	2-4'	6.5-8'	0.7-3'	0.5-2'	2-4'	4-6'	0-2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
4,4'-DDE	2.1	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.011	0.0036 U	0.0037 U	0.0037 U	0.0039 U
4,4'-DDT	2.1	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0084	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ALDRIN	0.041	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ALPHA-BHC	0.11	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
BETA-BHC	0.2	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
CHLORDANE	0.54	0.0081 U	0.0075 U	0.0077 U	0.0078 U	0.0087 U	0.0072 U	0.0074 U	0.0074 U	0.0078 U
DELTA-BHC	0.3	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
DIELDRIN	0.044	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ENDOSULFAN I	0.9	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ENDOSULFAN II	0.9	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ENDOSULFAN SULFATE	1	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ENDRIN	0.1	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ENDRIN ALDEHYDE	NS	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
ENDRIN KETONE	N/A	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
GAMMA-BHC (LINDANE)	0.06	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
HEPTACHLOR	0.1	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
HEPTACHLOR EPOXIDE	0.02	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0037 U	0.0039 U
METHOXYCHLOR	NS	0.0041 U	0.0037 U	0.0038 U	0.0039 U	0.0043 U	0.0036 U	0.0037 U	0.0054	0.0039 U
TOXAPHENE	NS	0.041 U	0.037 U	0.038 U	0.039 U	0.043 U	0.036 U	0.037 U	0.037 U	0.039 U
AROCLOR 1016	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.018 U	0.019 U	0.019 U	0.019 U
AROCLOR 1221	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.018 U	0.019 U	0.019 U	0.019 U
AROCLOR 1232	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.018 U	0.019 U	0.019 U	0.019 U
AROCLOR 1242	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.018 U	0.019 U	0.019 U	0.019 U
AROCLOR 1248	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.018 U	0.019 U	0.019 U	0.048
AROCLOR 1254	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.018 U	0.019 U	0.019 U	0.019 U
AROCLOR 1260	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.022 U	0.11	0.054	0.019 U	0.05
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	ND	ND	0.11	0.054	ND	0.098

U Undetectable Levels

and a second second

NTA THE STATE

ND Not Detectived

NS No Standard

and a second second



Table 5C Soil Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-FILL-04	PG-FILL-04	PG-FILL-04	PG-FILL-5	PG-FILL-5	PG-FILL-10	PG-FILL-10	PG-UST1-2	PG-UST1-2
Sample Date	Soil	11/6/2000	11/6/2000	11/6/2000	11/18/2000	11/18/2000	12/1/2000	12/1/2000	11/20/2000	11/20/2000
Sample ID	Cleanup	PGFILL04	PGFILL04	PGFILL04	PG-FILL-5	PG-FILL-5	PG-FILL10	PG-FILL10	PG-UST1-2	PG-UST1-2
Sample Depth	Objective	2-4'	4-6'	7-8 '	2-4'	6-8'	3-4'	6.4-8.2'	2-4'	12-14'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
4,4'-DDD	2.9	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0065	0.92	0.0036 U	0.004 U
4,4'-DDE	2.1	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0046	0.24	0.0036 U	0.004 U
4,4'-DDT	2.1	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ALDRIN	0.041	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ALPHA-BHC	0.11	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
BETA-BHC	0.2	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
CHLORDANE	0.54	0.0075 U	0.0077 U	0.0083 U	0.0072 U	0.0078 U	0.0073 U	0.16 U	0.0072 U	0.0079 U
DELTA-BHC	0.3	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
DIELDRIN	0.044	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ENDOSULFAN I	0.9	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ENDOSULFAN II	0.9	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ENDOSULFAN SULFATE	1	0.0037 U	0.0038 U	0.0042 U	0.00 36 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ENDRIN	0.1	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
ENDRIN ALDEHYDE	NS	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 Ù	0.079 U	0.0036 U	0.004 U
ENDRIN KETONE	N/A	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
GAMMA-BHC (LINDANE)	0.06	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
HEPTACHLOR	0.1	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
HEPTACHLOR EPOXIDE	0.02	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
METHOXYCHLOR	NS	0.0037 U	0.0038 U	0.0042 U	0.0036 U	0.0039 U	0.0037 U	0.079 U	0.0036 U	0.004 U
TOXAPHENE	NS	0.037 U	0.038 U	0.042 U	0.036 U	0.039 U	0.037 U	0.79 U	0.036 U	0.04 U
AROCLOR 1016	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
AROCLOR 1221	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
AROCLOR 1232	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
AROCLOR 1242	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
AROCLOR 1248	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
AROCLOR 1254	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
AROCLOR 1260	NS	0.019 U	0.019 U	0.021 U	0.018 U	0.019 U	0.018 U	0.04 U	0.018 U	0.02 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	ND	ND	ND	ND	ND	ND

U Undetectable Levels

ND Not Detectived

NS No Standard

Location	Recommended	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1	PG-UST7-1A	PG-UST7-1B	PG-UST7-2
Sample Date	Soil	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/28/2000	11/28/2000	11/28/2000	11/21/2000
Sample ID	Cleanup	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1	PG-UST7-1A	PG-UST7-1B	PG-UST7-2
Sample Depth	Objective	2-4'	10-11.5'	14-15.5'	4-6'	12-14'	8-10'	0-2'	2-3.5'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
·									<u></u>	
4,4'-DDD	2.9	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
4,4'-DDE	2.1	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
4,4'-DDT	2.1	0.0042 U	0.0046 U	0.0043 U	0.081	0.005 U	0.0039 U	0.36	0.15	0.0038 U
ALDRIN	0.041	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
ALPHA-BHC	0.11	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
BETA-BHC	0.2	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
CHLORDANE	0.54	0.0084 U	0.0091 U	0.0087 U	0.0089 U	0.01 U	0.0078 U	0.036 U	0.038 U	0.0077 U
DELTA-BHC	0.3	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
DIELDRIN	0.044	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.13	0.019 U	0.0038 U
ENDOSULFAN I	0.9	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
ENDOSULFAN II	0.9	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
ENDOSULFAN SULFATE	1	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
ENDRIN	0.1	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.07	0.019 U	0.0038 U
ENDRIN ALDEHYDE	NS	0.0042 U	0.0046 U	0.0043 U	0.016	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
ENDRIN KETONE	N/A	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
GAMMA-BHC (LINDANE)	0.06	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
HEPTACHLOR	0.1	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
HEPTACHLOR EPOXIDE	0.02	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
METHOXYCHLOR	NS	0.0042 U	0.0046 U	0.0043 U	0.0044 U	0.005 U	0.0039 U	0.018 U	0.019 U	0.0038 U
TOXAPHENE	NS	0.042 U	0.046 U	0.043 U	0.044 U	0.05 U	0.039 U	0.18 U	0.19 U	0.038 U
AROCLOR 1016	NS	0.021 U	0.023 U	0.022 U	0.022 U	0.025 U	0.02 U	0.018 U	0.019 U	0.019 U
AROCLOR 1221	NS	0.021 U	0.023 U	0.022 U	0.022 U	0.025 U	0.02 U	0.018 U	0.019 U	0.019 U
AROCLOR 1232	NS	0.021 U	0.023 U	0.022 U	0.022 U	0.025 U	0.02 U	0.018 U	0.019 U	0.019 U
AROCLOR 1242	NS	0.021 U	0.023 U	0.022 U	0.022 U	0.025 U	0.02 U	0.018 U	0.019 U	0.019 U
AROCLOR 1248	NS	0.021 U	0.023 U	0.022 U	0.022 U	0.025 U	0.02 U	0.018 U	0.019 U	0.019 U
AROCLOR 1254	NS	0.021 U	0.023 U	0.022 U	0.022 U	0.025 U	0.02 U	0.82	0.34	0.019 U
AROCLOR 1260	NS	0.021 U	0.023 U	0.022 U	0.2	0.025 U	0.02 U	0.018 U	0.019 U	0.019 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	0.2	ND	ND	0.82	0.34	ND

U Undetectable Levels

ND Not Detectived

NS No Standard

**



Table 5C Soil Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-UST7-2	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-10D	PG-PA-MW-10D	PG-MW10D	PG-MW10D
Sample Date	Soil	11/21/2000	11/13/2000	11/13/2000	11/13/2000	11/13/2000	11/27/2000	11/27/2000	11/27/2000	11/27/2000
Sample ID	Cleanup	PG-UST7-2	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07	PG-PAMW10D	PG-PAMW10D	PG-MW10D	PG-MW10D
Sample Depth	Objective	10-12'	2.5-4'	4-6'	6-8'	8-10'	0.5-2'	4-6'	7-8'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
· · · · · · · · · · · · · · · · · · ·										
4,4'-DDD	2.9	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.011	0.014	0.67	0.0051 U
4,4'-DDE	2.1	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0064	0.0077	0.082	0.0051 U
4,4'-DDT	2.1	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0063	0.023 U	0.0051 U
ALDRIN	0.041	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ALPHA-BHC	0.11	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ВЕТА-ВНС	0.2	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
CHLORDANE	0.54	0.008 U	0.0074 U	0.0077 U	0.008 U	0.0076 U	0.0074 U	0.029	0.22	0.01 U
DELTA-BHC	0.3	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
DIELDRIN	0.044	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ENDOSULFAN I	0.9	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ENDOSULFAN II	0.9	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ENDOSULFAN SULFATE	1	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ENDRIN	0.1	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ENDRIN ALDEHYDE	NS	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
ENDRIN KETONE	N/A	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
GAMMA-BHC (LINDANE)	0.06	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
HEPTACHLOR	0.1	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
HEPTACHLOR EPOXIDE	0.02	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
METHOXYCHLOR	NS	0.004 U	0.0037 U	0.0038 U	0.004 U	0.0038 U	0.0037 U	0.0038 U	0.023 U	0.0051 U
TOXAPHENE	NS	0.04 U	0.037 U	0.038 U	0.04 U	0.038 U	0.037 U	0.038 U	0.23 U	0.051 U
AROCLOR 1016	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.019 U	0.023 U	0.026 U
AROCLOR 1221	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.019 U	0.023 U	0.026 U
AROCLOR 1232	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.019 U	0.023 U	0.026 U
AROCLOR 1242	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.019 U	0.023 U	0.026 U
AROCLOR 1248	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.019 U	0.023 U	0.026 U
AROCLOR 1254	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.019 U	0.023 U	0.026 U
AROCLOR 1260	NS	0.02 U	0.019 U	0.019 U	0.02 U	0.019 U	0.019 U	0.031	0.25	0.026 U
TOTAL PCBs	1.0(Surface)/ 10(Sun-Surface)	ND	ND	ND	ND	ND	ND	0.031	0.25	ND

U Undetectable Levels

ND Not Detectived

NS No Standard

Location	Recommended	PG-A-4	PG-A-4	PG-A-5	PG-B-1	PG-B-1	PG-B-1
Sample Date	Soil ·	11/16/2000	11/16/2000	11/14/2000	12/4/2000	12/4/2000	12/4/2000
Sample ID	Cleanup	PG-A-04	PG-A-04	PG-A-05	PG-B-01	PG-B-01	PG-B-01
Sample Depth	Objective	6-8'	12-14'	2-4'	2-4'	6-8'	9-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	1200 U	22000	1500	3300	8500	3700
ANTIMONY	SB	2 U	3.1 U	1.8 U	1.6 U	1.7 U	1.8 U
ARSENIC	7.5 or SB	19	8.9	21	8.5	2.4 U	4.7
BARIUM	300 or SB	21	360	30	60	120	47
BERYLLIUM	0.16 (HEAST) or SB	0.55 U	2.2	0.49 U	0.45 U	0.56	0.51 U
CADMIUM	1 or SB	0.41 U	0.64 U	0.37 U	0.34 U	0.36 U	0.38 U
CALCIUM METAL	35,000*	1400 U	20000	1200 U	18000	1300	1300 U
CHROMIUM	10 or SB	7.9	13	4.9 U	7.5	18	9.1
COBALT	30 or SB	3	13	3.5	4.7	10	3.8
COPPER	25 or SB	22	69	20	37	13	15
IRON	2,000 or SB	3200 U	26000	7600	16000	22000	13000
LEAD	500*	10	22	16	31	14	7.5
MAGNESIUM	5,000*	810 U	7600	730 U	1700	4100	890
MANGANESE	5,000*	22 U	150	30	120	340	56
NICKEL	13 or SB	9.4	36	8.7	13	19	5.7
POTASSIUM	43,000*	550 U	19000	490 U	1100	1500	510 U
SELENIUM	2 or SB	3.4 U	5.3 U	3.1 U	3	3 U	3.2 U
SILVER	SB	1.7	6.3	0.62 U	1	0.63	0.63 U
SODIUM	8,000*	860	26000	490 U	1200	1700	510 U
THALLIUM	300*	1.6 U	2.6 U	1.5 U	1.4 U	1.4 U	1.5 U
VANADIUM	150 or SB	15	40	12 U	14	28	20
ZINC	20 or SB	14 U	200	23	31	51	26
MERCURY	0.1	0.19 U	0.3 U	0.18 U	0.16 U	0.17 U	0.18 U

U Undetectable Levels

SB Site Background

1

Location	Recommended	PG-B-02	PG-B-02	PG-B-02A	PG-B-3	PG-B-3	PG-B-4
Sample Date	Soil	11/16/2000	11/16/2000	11/16/2000	12/4/2000	12/4/2000	12/4/2000
Sample ID	Cleanup	PG-B-02	PG-A-02	PG-B-02A	PG-B-03	PG-B-03	PG-B-04
Sample Depth	Objective	2-4'	6-8'	8-10'	2-4'	6-8'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	2600	2000	1500	35000	15000	4400
ANTIMONY	SB	1.7 U	2 U	1.9 U	2.8	2.4 U	3.9
ARSENIC	7.5 or SB	4.6	3.6	3.6	4.6	4	37
BARIUM	300 or SB	43	57	31	230	350	170
BERYLLIUM	0.16 (HEAST) or SB	0.47 U	0.56 U	0.53 U	0.79	2.2	1.1
CADMIUM	1 or SB	0.35 U	0.42 U	14	0.42 U	0.49 U	0.35 U
CALCIUM METAL	35,000*	9600	3800	6000	15000	3700	19000
CHROMIUM	10 or SB	7.1	5.6 U	5.3 U	40	18	19
COBALT	30 or SB	2.5	2.3 U	2.2 U	13	11	10
COPPER	25 or SB	17	17	17	73	57	210
IRON	2,000 or SB	7800	4800	9400	28000	9000	20000
LEAD	500*	20	6.7	27	200	10	230
MAGNESIUM	5,000*	690 U	820 U	780 U	2500	3400	8800
MANGANESE	5,000*	69	22 U	110	150	59	210
NICKEL	13 or SB	11	7.4	6.7	53	19	600
POTASSIUM	43,000*	560	560 U	530 U	5700	6200	580
SELENIUM	2 or SB	2.9 U	3.5 U	3.3 U	3.5 U	4.1 U	2.9 U
SILVER	SB	0.68	0.69 U	0.66 U	6.6	6.7	0.59 U
SODIUM	8,000*	470 U	560 U	530 U	15000	5400	470 U
THALLIUM	300*	1.4 U	1.7 U	1.6 U	1.7 U	2 U	1.4 U
VANADIUM	150 or SB	12 U	14 U	13 U	14 U	23	36
ZINC	20 or SB	12 U	35	1100	170	30	580
MERCURY	0.1	0.17 U	0.2 U	0.19 U	0.48	0.23 U	0.34

U Undetectable Levels

 γ

SB Site Background

Location	Recommended	PG-E-1	PG-E-1	PG-E-1	PG-F2-2	PG-F2-2	PG-P-1
Sample Date	Soil	11/13/2000	11/13/2000	11/13/2000	11/20/2000	11/20/2000	11/22/2000
Sample ID	Cleanup	PG-E-01	PG-E-01	PG-E-01	PG-F2-2	PG-F2-2	PG-P-01
Sample Depth	Objective	0.2-2'	4-6'	10-12'	2-4'	8-10'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	4500	1900	7300	2300	6400	2200
ANTIMONY	SB	1.6 U	1.7 U	2.7 U	1.6 U	1.7 U	1.6 U
ARSENIC	7.5 or SB	6.4	5.3	12	2.2 U	2.3 U	2.5
BARIUM	300 or SB	93	44	390	15	17	18
BERYLLIUM	0.16 (HEAST) or SB	0.63	0.47 U	1.9	0.43 U	0.47 U	0.43 U
CADMIUM	1 or SB	0.34 U	0.35 U	0.57 U	0.33 U	0.35 U	0.33 U
CALCIUM METAL	35,000*	8300	3800	7400	1100 U	2300	1100 U
CHROMIUM	10 or SB	8.5	4.7 U	7.5 U	4.3 U	5.9	5.2
COBALT	30 or SB	6.8	3.1	3.1 U	1.8 U	1.9 U	2.2
COPPER	25 or SB	100	60	28	20	4.4 U	13
IRON	2,000 or SB	16000	8300	23000	6100	12000	7200
LEAD	500*	170	70	70	9.1	4.7 U	21
MAGNESIUM	5,000*	3500	850	2300	640 U	710	640 U
MANGANESE	5,000*	190	54	84	24	110	66
NICKEL	13 or SB	32	36	7.2	2.7 U	4.6	4.6
POTASSIUM	43,000*	810	470 U	2200	430 U	910	430 U
SELENIUM	2 or SB	2.8 U	2.9 U	4.7 U	2.7 U	2.9 U	2.7 U
SILVER	SB	0.57 U	0.66	0.94 U	0.54 U	0.58 U	0.54 U
SODIUM	8,000*	450 U	470 U	17000	430 U	470 U	430 U
THALLIUM	300*	1.4 U	1.4 U	2.3 U	1.3 U	1.4 U	1.3 U
VANADIUM	150 or SB	19	12 U	19 U	11 U	12 U	11 U
ZINC	20 or SB	540	990	120	22	25	40
MERCURY	0.1	0.24	1.2	0.27 U	0.15 U	0.17 U	0.15 U

U Undetectable Levels

SB Site Background

Location	Recommended	PG-P-1	PG-P-2	PG-P-2	PG-P-3	PG-P-3	PG-PD-1
Sample Date	Soil	11/22/2000	11/30/2000	11/30/2000	11/22/2000	11/22/2000	11/21/2000
Sample ID	Cleanup	PG-P-01	PG-P-02	PG-P-02	PG-P-03	PG-P-03	PG-PD-01
Sample Depth	Objective	8-10'	2-4'	4-6'	2-4'	6-8'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	5200	2600	3100	8700	7400	4700 U
ANTIMONY	SB	1.8 U	1.7 U	1.7 U	1.7 U	1.7 U	8.1 U
ARSENIC	7.5 or SB	2.5 U	4.2	3.1	2.3 U	2.3 U	13
BARIUM	300 or SB	20	19	21	73	58	170
BERYLLIUM	0.16 (HEAST) or SB	0.51 U	0.47 U	0.48 U	0.78	0.73	2.2 U
CADMIUM	1 or SB	0.38 U	0.35 U	0.36 U	0.34 U	0.34 U	1.7 U
CALCIUM METAL	35,000*	1300 U	580 U	600 U	3400	45000	16000
CHROMIUM	10 or SB	10	4.7 U	5.6	18	16	22 U
COBALT	30 or SB	3.1	1.9 U	2 U	10	9.2	9.3 U
COPPER	25 or SB	5.6	82	4.5 U	16	6.1	92
IRON	2,000 or SB	12000	5100	9000	20000	18000	13000 U
LEAD	500*	7.9	17	4.8 U	12	14	80
MAGNESIUM	5,000*	1100	420 U	450	4900	5000	3300 U
MANGANESE	5,000*	47	38	27	860	1300	91 U
NICKEL	13 or SB	5.4	14	3.3	23	21	23
POTASSIUM	43,000*	520	140	250	1900	2500	450 U
SELENIUM	2 or SB	3.2 U	2.9 U	3 U	2.9 U	2.9 U	14 U
SILVER	SB	0.63 U	0.58 U	0.6 U	0.57 U	0.57 U	2.8 U
SODIUM	8,000*	510 U	460 U	480 U	460 U	460 U	450 U
THALLIUM	300*	1.5 U	1.4 U	1.4 U	1.4 U	1.4 U	6.7 U
VANADIUM	150 or SB	19	12 U	12 U	25	26	56 U
ZINC	20 or SB	23	64	14	53	42	240
MERCURY	0.1	0.18 U	0.17 U	0.17 U	0.16 U	0.16 U	0.16 U

U Undetectable Levels

SB Site Background

-20

Location	Recommended	PG-PD-1	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5	PG-Q1-1
Sample Date	Soil	11/21/2000	11/29/2000	12/2/2000	12/2/2000	12/2/2000	11/30/2000
Sample ID	Cleanup	PG-PD-01	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5	PG-Q1-1
Sample Depth	Objective	10-12'	4-6'	8-10'	0.7-1.5'	2-4'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	1100	8600	7200	4200	4400	3700
ANTIMONY	SB	1.8 U	1.8 U	1.8 U	1.6 U	1.6 U	1.6 U
ARSENIC	7.5 or SB	2.5 U	2.5 U	2.4 U	9.3	2.2 U	10
BARIUM	300 or SB	13	38	47	50	66	67
BERYLLIUM	0.16 (HEAST) or SB	0.51 U	0.72	0.89	0.57	0.8	0.45 U
CADMIUM	1 or SB	0.38 U	0.37 U	0.37 U	0.34 U	0.34 U	0.34 U
CALCIUM METAL	35,000*	1300 U	15000	1100	47000	27000	26000
CHROMIUM	10 or SB	5.1 U	19	13	17	11	11
COBALT	30 or SB	2.1 U ·	8.8	9.1	4.1	5.7	4.7
COPPER	25 or SB	5.8	12	29	72	9.5	44
IRON	2,000 or SB	4500	22000	17000	14000	14000	14000
LEAD	500*	5.1 U	16	13	47	10	75
MAGNESIUM	5,000*	750 U	11000	2900	14000	3300	12000
MANGANESE	5,000*	21 U	270	280	130	630	190
NICKEL	13 or SB	3.1 U	19	15	19	12	15
POTASSIUM	43,000*	510 U	1900	810	470	1200	450
SELENIUM	2 or SB	3.2 U	3.1 U	3 U	2.8 U	2.8 U	2.8 U
SILVER	SB	0.63 U	0.62 U	0.61 U	0.57	0.56 U	0.57 U
SODIUM	8,000*	510 U	1900	490 U	550	1100	450 U
THALLIUM	300*	1.5 U	1.5 U	1.5 U	1.4 U	1.3 U	1.4 U
VANADIUM	150 or SB	13 U	26	24	52	20	26
ZINC	20 or SB	17	61	40	87	28	110
MERCURY	0.1	0.18 U	0.18 U	0.17 U	0.39	0.16 U	0.16 U

U Undetectable Levels

SB Site Background

Location	Recommended	PG-Q1-1	PG-RR-1	PG-RR-1	PG-RR-2	PG-RR-3	PG-RR-4
Sample Date	Soil	11/30/2000	11/3/2000	11/3/2000	11/4/2000	11/3/2000	11/4/2000
Sample ID	Cleanup	PG-Q1-1	PG-RR-01	PG-RR-01	PG-RR-02	PG-RR-03	PG-RR-04
Sample Depth	Objective	4-6'	0-1.2'	1.2-2'	0-2'	1.5-2'	0.6-2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	3000	1400	2500	840	1900	3400
ANTIMONY	SB	2.2 U	7.3	2	3	1.9	1.6 U
ARSENIC	7.5 or SB	9.2	980	290	150	250	11
BARIUM	300 or SB	42	170	22	100	120	73
BERYLLIUM	0.16 (HEAST) or SB	0.62 U	0.47 U	0.46 U	0.47 U	0.47 U	0.45 U
CADMIUM	1 or SB	0.46 U	1.4	0.34 U	0.35 U	0.39	0.34 U
CALCIUM METAL	35,000*	770 U	37000	570 U	3100	10000	61000
CHROMIUM	10 or SB	10	23	4.6	9.6	8.4	7.9
COBALT	30 or SB	3.3	5.1	1.9	4.2	4.6	2.2
COPPER	25 or SB	42	150	12	51	69	22
IRON	2,000 or SB	14000	24000	9200	23000	13000	8700
LEAD	500*	39	210	11	120	120	52
MAGNESIUM	5,000*	620	21000	580	1300	5900	27000
MANGANESE	5,000*	43	210	41	140	130	120
NICKEL	13 or SB	8.2	65	15	41	130	22
POTASSIUM	43,000*	250	370	170	310	230	600
SELENIUM	2 or SB	3.8 U	2.9 U	2.9 U	2.9 U	2.9 U	2.8 U
SILVER	SB	0.77 U	0.58 U	0.57 U	0.59 U	0.58 U	0.56 U
SODIUM	8,000*	610 U	460 U	460 U	470 U	460 U	450 U
THALLIUM	300*	1.8 U	1.4 U	1.4 U	1.4 U	1.4 U	1.3 U
VANADIUM	150 or SB	18	45	13	23	21	11 U
ZINC	20 or SB	47	270	30	61	110	40
MERCURY	0.1	0.22 U	0.21	0.16 U	0.17 U	0.17 U	0.16 U

U Undetectable Levels

SB Site Background

* Eastern USA Background

an de contracter l'insulation de la contract de la contraction de la contraction de la contraction de la contra

Location	Recommended	PG-RR-4	PG-RR-5	PG-RR-5	PG-RR-6	PG-RR-6	PG-RR-7
Sample Date	Soil	11/4/2000	11/7/2000	11/7/2000	11/6/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PG-RR-04	PG-RR-05	PG-RR-05	PG-RR-06	PG-RR-06	PG-RR-07
Sample Depth	Objective	3-4'	1-2'	2-4'	1-2'	2-4'	0-2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	8500	3000	4700	2800	1500	770
ANTIMONY	SB	1.8 U	1.6 U	1.6 U	4.2	1.6 U	1.6 U
ARSENIC	7.5 or SB	7.5	89	180	140	12	2.8
BARIUM	300 or SB	96	73	23	96	19	29
BERYLLIUM	0.16 (HEAST) or SB	0.5 U	0.43 U	0.43 U	0.43 U	0.43 U	0.43 U
CADMIUM	1 or SB	0.37 U	0.33 U	0.32 U	0.33 U	0.33 U	0.32 U
CALCIUM METAL	35,000*	3700	1100 U	1100 U	3800	540 U	540 U
CHROMIUM	10 or SB	13	7.5	7.1	6.8	4.3 U	5.9
COBALT	30 or SB	8	2.7	2.2	2.4	1.8 U	2.7
COPPER	25 or SB	19	42	44	65	12	11
IRON	2,000 or SB	19000	14000	11000	19000	11000	3600
LEAD	500*	22	29	22	200	24	17
MAGNESIUM	5,000*	3800	640 U	630 U	1200	390 U	390 U
MANGANESE	5,000*	410	43	33	79	28	30
NICKEL	13 or SB	14	130	16	9	2.7 U	2.6 U
POTASSIUM	43,000*	1100	210	180	270	140	120
SELENIUM	2 or SB	3.1 U	2.7 U	2.7 U	2.8	2.7 U	2.7 U
SILVER	SB	0.62 U	0.54 U				
SODIUM	8,000*	1900	97 U	96 U	430 U	430 U	430 U
THALLIUM	300*	1.5 U	1.3 U				
VANADIUM	150 or SB	20	14	16	17	11 U	15
ZINC	20 or SB	54	30	21	46	11 U	15
MERCURY	0.1	0.18 U	0.15 U	0.15 U	0.15 Ŭ	0.15 U	0.15 U

U Undetectable Levels

SB Site Background

Location	Recommended	PG-RR-7	PG-STAIN-02	PG-STAIN-02	PG-STAIN-03	PG-STAIN-03
Sample Date	Soil	11/6/2000	11/11/2000	11/11/2000	11/10/2000	11/10/2000
Sample ID	Cleanup	PG-RR-07	PG-ST-02	PG-ST-02	PG-ST-03	PG-ST-03
Sample Depth	Objective	2-4'	1-2'	2-3'	1.7-2.5'	2.5-3.5'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
			-			
ALUMINUM (FUME OR DUST)	33,000*	710	3100	4900	1200	1100 U
ANTIMONY	SB	1.6 U	1.7 U	1.7 U	1.7 U	1.8 U
ARSENIC	7.5 or SB	8.5	31	50	110	43
BARIUM	300 or SB	43	12 U	18	71	66
BERYLLIUM	0.16 (HEAST) or SB	0.43 U	0.58	0.65	0.56	0.51 U
CADMIUM	1 or SB	0.33 U	0.35 U	0.36 U	0.35 U	0.38 U
CALCIUM METAL	35,000*	70000	1900	1200	8900	1300 U
CHROMIUM	10 or SB	6	7.6	7.6	15	5.1 U
COBALT	30 or SB	1.8 U	3.7	2 U	2.6	2.1 U
COPPER	25 or SB	14	16	70	52	18
IRON	2,000 or SB	4200	11000	8500	38000	38000
LEAD	500*	23	5.1	8.3	110	39
MAGNESIUM	5,000*	41000	930	710 U	1900	750 U
MANGANESE	5,000*	88	43	41	34	21 U
NICKEL	13 or SB	4	4.5	3.6	6.9	13
POTASSIUM	43,000*	190	470 U	480 U	980	1400
SELENIUM	2 or SB	2.7 U	2.9 U	3 U	5.7	6
SILVER	SB	0.54 U	0.59 U	0.6 U	0.58 U	0.63 U
SODIUM	8,000*	430 U	470 U	480 U	2100	2100
THALLIUM	300*	1.3 U	1.4 U	1.4 U	2.3	2.9
VANADIUM	150 or SB	12	15	13	47	23
ZINC	20 or SB	60	20	16	18	13 U
MERCURY	0.1	0.15 U	0.17 U	0.17 U	0.17 U	0.52

U Undetectable Levels

SB Site Background

Location	Recommended	PG-STAIN-3B	PG-STAIN-3B	PG-FS-2	PG-FS-2	PG-FS-2	PG-FS-3
Sample Date	Soil	11/11/2000	11/11/2000	11/17/2000	11/17/2000	11/17/2000	11/15/2000
Sample ID	Cleanup	PG-ST-3B	PG-ST-3B	PG-FS-02	PG-FS-02	PG-FS-02	PG-FS03
Sample Depth	Objective	0-2'	2-4'	2-4'	8-10'	17-18'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	2100	2000	4300	4500	1000	3000
ANTIMONY	SB	13	1.6 U	1.6 U	1.8 U	1.6 U	1.7 U
ARSENIC	7.5 or SB	350	31	5.5	6.8	2.2 U	30
BARIUM	300 or SB	160	11 U	32	18	11 U	54
BERYLLIUM	0.16 (HEAST) or SB	0.67	0.5	0.54	1.9	0.45 U	0.46 U
CADMIUM	1 or SB	1.8	0.32 U	0.34 U	0.37 U	0.34 U	0.34 U
CALCIUM METAL	35,000*	44000	1100 U	1100 U	1200 U	1500	1300
CHROMIUM	10 or SB	83	7.7	11	14	5.3	8.6
COBALT	30 or SB	7.4	1.9	1.9 U	7.5	1.9 U	1.9 U
COPPER	25 or SB	260	5.7	26	9.4	4.3 U	40
IRON	2,000 or SB	45000	10000	14000	17000	2700 U	12000
LEAD	500*	950	4.5	6.8	4.9 U	4.5 U	96
MAGNESIUM	5,000*	22000	630 U	1200	720 U	660 U	680 U
MANGANESE	5,000*	270	45	46	110	37	40
NICKEL	13 or SB	34	4.3	5.6	5	2.7 U	5.2
POTASSIUM	43,000*	430 U	430 U	590	490 U	450 U	470
SELENIUM	2 or SB	4.1	2.7 U	2.8 U	3 U	2.8 U	2.9 U
SILVER	SB	0.54 U	0.54 U	0.57 U	0.61 U	0.56 U	0.57 U
SODIUM	8,000*	1200	430 U	450 U	490 U	450 U	460 U
THALLIUM	300*	1.3 U	1.3 U	1.4 U	1.5 U	1.3 U	1.4 U
VANADIUM	150 or SB	26	11	21	30	11 U	14
ZINC	20 or SB	330	15	36	42	11 U	81
MERCURY	0.1	0.24	0.15 U	0.16 U	0.17 U	0.16 U	0.16 U

U Undetectable Levels

1996

SB Site Background

* Eastern USA Background

127

Location	Recommended	PG-FS-3	PG-FILL-02	PG-FILL-03	PG-FILL-03	PG-FILL-03
Sample Date	Soil	11/15/2000	11/3/2000	11/4/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PG-FS03	PGFILL02	PGFILL03	PGFILL03	PGFILL03
Sample Depth	Objective	6.5-8	0.7-3'	0.5-2'	2-4'	4-6'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	5800	4300	3900	2500	3300
ANTIMONY	SB	1.7 U	1.9 U	1.6 U	19	1.6 U
ARSENIC	7.5 or SB	2.5	45	13	9.3	3
BARIUM	300 or SB	130	210	87	100	29
BERYLLIUM	0.16 (HEAST) or SB	0.68	0.52 U	0.86	0.45	0.44 U
CADMIUM	1 or SB	0.35 U	1	0.44	0.4	0.33 U
CALCIUM METAL	35,000*	3000	21000	27000	23000	2200
CHROMIUM	10 or SB	11	22	11	6.1	5.4
COBALT	30 or SB	7.2	7.6	4.8	3.7	2.2
COPPER	25 or SB	14	93	66	2600	25
IRON	2,000 or SB	15000	39000	19000	13000	9600
LEAD	500*	22	680	63	420	28
MAGNESIUM	5,000*	3400	10000	5000	9600	1000
MANGANESE	5,000*	290	150	240	96	40
NICKEL	13 or SB	16	17	24	180	7.6
POTASSIUM	43,000*	1300	480	290	350	300
SELENIUM	2 or SB	2.9 U	3.3	2.7 U	2.8 U	2.8 U
SILVER	SB	0.59 U	0.65 U	0.54 U	11	0.56 U
SODIUM	8,000*	470 U	520 U	430 U	440 U	470
THALLIUM	300*	1.4 U	1.6 U	1.3 U	1.3 U	1.3 U
VANADIUM	150 or SB	20	30	32	14	11 U
ZINC	20 or SB	93	120	280	320	40
MERCURY	0.1	0.17 U	0.18 U	0.22	0.19	0.16 U

U Undetectable Levels

1946

SB Site Background

Location	Recommended	PG-FILL-04	PG-FILL-04	PG-FILL-04	PG-FILL-04	PG-FILL-5	PG-FILL-5
Sample Date	Soil	11/6/2000	11/6/2000	11/6/2000	11/6/2000	11/18/2000	11/18/2000
Sample ID	Cleanup	PGFILL04	PGFILL04	PGFILL04	PGFILL04	PG-FILL-5	PG-FILL-5
Sample Depth	Objective	0-2'	2-4'	4-6'	7-8'	2-4'	6-8'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	2300	2000	3600	2400	3200	2600
ANTIMONY	SB	1.7 U	1.6 U	1.7 U	1.8 U	1.6 U	1.7 U
ARSENIC	7.5 or SB	10	8.8	4.7	3	4.6	3.3
BARIUM	300 or SB	140	70	50	13 U	26	17
BERYLLIUM	0.16 (HEAST) or SB	0.47 U	0.45 U	0.46 U	0.5 U	0.58	0.47
CADMIUM	1 or SB	0.48	0.34 U	0.34 U	0.37 U	0.32 U	0.35 U
CALCIUM METAL	35,000*	3700	1300	1100	910	1100 U	1200 U
CHROMIUM	10 or SB	4.7 U	4.9	6.8	5 U	7.8	5.9
COBALT	30 or SB	4.5	5.1	5	4.6	4 .	2.8
COPPER	25 or SB	48	49	22	20	11	8.7
IRON	2,000 or SB	16000	14000	13000	6800	11000	10000
LEAD	500*	76	43	16	12	8.2	4.7 U
MAGNESIUM	5,000*	680	450	950	450 U	800	690 U
MANGANESE	5,000*	63	66	120	38	86	53
NICKEL	13 or SB	15	10	10	5.1	4.6	3.6
POTASSIUM	43,000*	270	260	470	260	430 U	470 U
SELENIUM	2 or SB	2.9 U	2.8 U	2.9 U	3.1 U	2.7 U	2.9 U
SILVER	SB	0.58 U	0.56 U	0.57 U	0.62 U	0.54 U	0.58 U
SODIUM	8,000*	460 U	450 U	460 U	500 U	430 U	470 U
THALLIUM	300*	1.4 U	1.3 U	1.4 U	1.5 U	1.3 U	1.4 U
VANADIUM	150 or SB	12 U	13	15	13 U	16	14
ZINC	20 or SB	130	110	53	24	26	22
MERCURY	0.1	0.26	0.16 U	0.16 U	0.24	0.15 U	0.17 U

U Undetectable Levels

1989

SB Site Background

* Eastern USA Background

129

Location	Recommended	PG-FILL-10	PG-FILL-10	PG-UST1-2	PG-UST1-2	PG-UST4-1	PG-UST4-1
Sample Date	Soil	12/1/2000	12/1/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000
Sample ID	Cleanup	PG-FILL10	PG-FILL10	PG-UST1-2	PG-UST1-2	PG-UST4-1	PG-UST4-1
Sample Depth	Objective	3-4'	6.4-8.2'	2-4'	12-14'	2-4'	10-11.5'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	4000	6500	2800	1000 U	3600	3600
ANTIMONY	SB	1.6 U	3.6	1.8	1.7 U	1.8 U	2 U
ARSENIC	7.5 or SB	4.8	180	11	2.4 U	50	33
BARIUM	300 or SB	67	360	70	12 U	210	90
BERYLLIUM	0.16 (HEAST) or SB	0.55	1.1	0.43 U	0.48 U	0.6	0.55
CADMIUM	1 or SB	0.33 U	2.6	0.32 U	0.3 <mark>6</mark> U	0.62	0.41 U
CALCIUM METAL	35,000*	5100	6300	4200	1200 U	21000	5200
CHROMIUM	10 or SB	16	66	6.9	4.8 U	8.6	5.5 U
COBALT	30 or SB	4.9	38	3.3	2 U	8	9
COPPER	25 or SB	22	220	38	4.5 U	370	41
IRON	2,000 or SB	13000	35000	12000	4300	32000	17000
LEAD	500*	26	260	300	4.8 U	310	16
MAGNESIUM	5,000*	1500	1500	1600	700 U	13000	1500
MANGANESE	5,000*	430	67 .	130	19 U	150	60
NICKEL	13 or SB	8.1	230	9.7	2.9 U	57	110
POTASSIUM	43,000*	390	440	430 U	480 U	590	810
SELENIUM	2 or SB	2.7 U	6.7	2.7 U	3 U	4.9	4.3
SILVER	SB	0.55 U	1.2 U	0.54 U	0.6 U	0.63 U	0.68 U
SODIUM	8,000*	440 U	950 U	430 U	480 U	510 U	550 U
THALLIUM	300*	1.3 U	2.9 U	1.3 U	1.4 U	1.5 U	1.6 U
VANADIUM	150 or SB	34	130	14	12 U	25	18
ZINC	20 or SB	41	640	100	39	1200	3500
MERCURY	0.1	0.1 <mark>6</mark> U	0.47	1	0.17 U	0.45	0.19 U

U Undetectable Levels

SB Site Background

* Eastern USA Background

130
Table 5DSoil Analytical ResultsMetalsSite 2A/2B HHMT- Port Ivory Facility

Location	Recommended	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1	PG-UST7-1A	PG-UST7-1B
Sample Date	Soil	11/20/2000	11/20/2000	11/20/2000	11/28/2000	11/28/2000	11/28/2000
Sample ID	Cleanup	PG-UST4-1	PG-UST4-2	PG-UST4-2	PG-UST7-1	PG-UST7-1A	PG-UST7-1B
Sample Depth	Objective	14-15.5'	4-6'	12-14'	8-10'	0-2'	2-3.5'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	6300	1400	1500	7600	2500	990
ANTIMONY	SB	1.9 U	1.9 U	2.2 U	1.7 U	1.6 U	1.6 U
ARSENIC	7.5 or SB	5.1	8.4	13	2.4 U	2.3	2.3 U
BARIUM	300 or SB	76	61	43	31	49	16
BERYLLIUM	0.16 (HEAST) or SB	0.62	0.53 U	0.6 U	0.47 U	0.43 U	0.45 U
CADMIUM	1 or SB	0.39 U	0.4 U	0.45 U	0.35 U	0.32 U	0.34 U
CALCIUM METAL	35,000*	3700	3700	6700	1500	69000	20000
CHROMIUM	10 or SB	13	5.3 U	6 U	11	9.2	4.5 U
COBALT	30 or SB	9.2	2.2 U	2.5 U	6.7	3.3	1.9 U
COPPER	25 or SB	25	16	38 • 🧳	7.3	45	19
IRON	2,000 or SB	20000	7100	18000	14000	10000	4300
LEAD	500*	65	13	49	8.5	120	23
MAGNESIUM	5,000*	3600	790 U	1400	3400	31000	11000
MANGANESE	5,000*	300	22	100	180 .	130	68
NICKEL	13 or SB	22	9.3	11	14	17	8.4
POTASSIUM	43,000*	4100	530 U	1900	1400	430 U	450 U
SELENIUM	2 or SB	3.2 U	3.3 U	3.7 U	2.9 U	2.7 U	2.8 U
SILVER	SB	0.65 U	0.74	0.79	0.59 U	0.54 U	0.57 U
SODIUM	8,000*	4000	530 U	4400	720	470	450 U
THALLIUM	300*	1.6 U	1.6 U	1.8 U	1.4 U	1.3 U	1.4 U
VANADIUM	150 or SB	23	13 U	15 U	18	17	13
ZINC	20 or SB	160	79	480	44	160	83
MERCURY	0.1	0.24	0.19 U	0.21 U	0.17 U	0.83	0.16 U

U Undetectable Levels

SB Site Background

* Eastern USA Background

Table 5DSoil Analytical ResultsMetalsSite 2A/2B HHMT- Port Ivory Facility

Location	Recommended	PG-UST7-2	PG-UST7-2	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7
Sample Date	Soil	11/21/2000	11/21/2000	11/13/2000	11/13/2000	11/13/2000	11/13/2000
Sample ID	Cleanup	PG-UST7-2	PG-UST7-2	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07
Sample Depth	Objective	8-10'	10-12'	2.5-4'	4-6'	6-8'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	8200	5200	4200	2600	2600	6200
ANTIMONY	SB	1.7 U	1.7 U	1.6 U	1.7 U	1.7 U	1.6 U
ARSENIC	7.5 or SB	4.7	6.6	4.2	2.9	2.4 U	2.3 U
BARIUM	300 or SB	110	92	16	12	12 U	57
BERYLLIUM	0.16 (HEAST) or SB	0.94	0.53	0.44 U	0.46 U	0.48 U	0.45 U
CADMIUM	1 or SB	0.34 U	0.36 U	0.33 U	0.34 U	0.36 U	0.34 U
CALCIUM METAL	35,000*	2300	3600	1100 U	1100 U	1200 U	1100 U
CHROMIUM	10 or SB	19	14	7.2	5.6	6.3	7.8
COBALT	30 or SB	11	6.3	5.4	1.9 U	2 U	5.3
COPPER	25 or SB	15	22	4.2 U	4.4 U	4.6 U	8.2
IRON	2,000 or SB	23000	19000	12000	9200	9000	11000
LEAD	500*	12	40	4.4 U	4. 6 U	4.8 U	5.1
MAGNESIUM	5,000*	4100	2600	1100	680 U	710 U	1800
MANGANESE	5,000*	430	310	150	49	51	92
NICKEL	13 or SB	41	14	4.7	2.8 U	2.9 U	8.1
POTASSIUM	43,000*	1600	850	570	490	490	690
SELENIUM	2 or SB	2.9 U	3 U	2.8 U	2.9 U	3 U	2.8 U
SILVER	SB	0.94	0.83	0.56 U	0.57 U	0.6 U	0.6
SODIUM	8,000*	460 U	480 U	440 U	700	890	3900
THALLIUM	300*	1.4 U	1.4 U	1.3 U	1.4 U	1.4 U	1.4 U
VANADIUM	150 or SB	28	22	15	12	.13	14
ZINC	20 or SB	54	59	22	15	16	29
MERCURY	0.1	0.16 U	0.17 U	0.16 U	0.16 U	0.17 U	0.16 U

U Undetectable Levels

SB Site Background

* Eastern USA Background

Table 5DSoil Analytical ResultsMetalsSite 2A/2B HHMT- Port Ivory Facility

Location	Recommended	PG-PA-MW-10D	PG-PA-MW-10D	PG-MW10D	PG-MW10D
Sample Date	Soil	11/27/2000	11/27/2000	11/27/2000	11/27/2000
Sample ID	Cleanup	PG-PAMW10D	PG-PAMW10D	PG-MW10D	PG-MW10D
Sample Depth	Objective	0.5-2'	4-6'	7-8'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
ALUMINUM (FUME OR DUST)	33,000*	3000	2900	2800	3600
ANTIMONY	SB	1.6 U	1.6 U	2.4	2.2 U
ARSENIC	7.5 or SB	3.9	7.6	140	12
BARIUM	300 or SB	42	35	51	120
BERYLLIUM	0.16 (HEAST) or SB	0.58	0.49	0.93	0.62 U
CADMIUM	1 or SB	0.33 U	0.34 U	0.44	0.46 U
CALCIUM METAL	35,000*	3800	3100	120000	5800
CHROMIUM	10 or SB	6.7	5	13	6.2 U
COBALT	30 or SB	3.2	2.8	3.9	3.1
COPPER	25 or SB	17	19	95	6.4
IRON	2,000 or SB	10000	8100	19000	6600
LEAD	500*	15	19	84	24
MAGNESIUM	5,000*	2000	2100	52000	1700
MANGANESE	5,000*	190	160	210	120
NICKEL	13 or SB	5.1	7.5	29	4.2
POTASSIUM	43,000*	440 U	450 U	550 U	620 U
SELENIUM	2 or SB	2.8 U	2.8 U	3.4 U	3.8 U
SILVER	SB	0.56 U	0.57 U	0.68 U	0.78
SODIUM	8,000*	440 U	450 U	550 U	1900
THALLIUM	300*	1.3 U	1.4 U	1.6 U	1.8 ⁻ U
VANADIUM	150 or SB	17	14	40	15 U
ZINC	20 or SB	25	51	170	17
MERCURY	0.1	0.16 U	0.38	0.19 U	0.22 U

U Undetectable Levels

SB Site Background

* Eastern USA Background

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-A-4	PG-A-4	PG-A-5	PG-B-1	PG-B-1	PG-B-1
Sample Date	Soil	11/16/2000	11/16/2000	11/14/2000	12/4/2000	12/4/2000	12/4/2000
Sample ID	Cleanup	PG-A-04	PG-A-04	PG-A-05	PG-B-01	PG-B-01	PG-B-01
Sample Depth	Objective	6-8'	12-14'	2-4'	2-4'	6-8'	9-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	47 U	74	12000	39 U	41 U	43 U
OIL & GREASE	NS	91 U	2100	9700	76 U	80 U	84 U
CYANIDE	***	0.34 U	0.53 U	0.74	0.28 U	0.3 U	0.32 U
pH	NS	10	11	4.9	10	8.8	8.0
TOTAL PHENOLICS	NS	1.7 U	2.7 U	1.5 U	1.4 U	1.5 U	1.6 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-B-02	PG-B-02	PG-B-02A	PG-B-3	PG-B-3	PG-B-4
Sample Date	Soil	11/16/2000	11/16/2000	11/16/2000	12/4/2000	12/4/2000	12/4/2000
Sample ID	Cleanup	PG-B-02	PG-A-02	PG-B-02A	PG-B-03	PG-B-03	PG-B-04
Sample Depth	Objective	2-4'	6-8'	8-10'	2-4'	6-8'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	1200	10000	890	15000	13000	1400
OIL & GREASE	NS	1800	20000	460	22000	4200	2000
CYANIDE	***	0.29 U	0.35 U	0.33 U	0.35 U	0.41 U	0.29 U
рН	NS	8.8	7.9	8.0	9.1	9.2	7.8
TOTAL PHENOLICS	NS	1.5 U	1.7 U	1.6 U	1.7 U	2.0 U	1.5 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-E-1	PG-E-1	PG-E-1	PG-F2-2	PG-F2-2	PG-P-1	PG-P-1
Sample Date	Soil	11/13/2000	11/13/2000	11/13/2000	11/20/2000	11/20/2000	11/22/2000	11/22/2000
Sample ID	Cleanup	PG-E-01	PG-E-01	PG-E-01	PG-F2-2	PG-F2-2	PG-P-01	PG-P-01
Sample Depth	Objective	0.2-2'	4-6'	10-12'	2-4'	8-10'	2-4'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	150	120	96	37 U	40 U	37 U	43 U
OIL & GREASE	NS	1200	2600	430	140	78 U	72 U	84 U
CYANIDE	***	0.58	0.84	0.47 U	0.27 U	0.29 U	0.27 U	0.32 U
pH	NS	7.4	7.7	8.7	9.9	6.4	6.4	6.9
TOTAL PHENOLICS	NS	1.4 U	1.4 U	2.4 U	1.4 U	1.4 U	1.4 U	1. 6 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Table 5E Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-P-2	PG-P-2	PG-P-3	PG-P-3	PG-PD-1	PG-PD-1
Sample Date	Soil	11/30/2000	11/30/2000	11/22/2000	11/22/2000	11/21/2000	11/21/2000
Sample ID	Cleanup	PG-P-02	PG-P-02	PG-P-03	PG-P-03	PG-PD-01	PG-PD-01
Sample Depth	Objective	2-4'	4-6'	2-4'	6-8'	2-4'	10-12'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	40 U	40 U	74	39 U	38 U	43 U
OIL & GREASE	NS	220	79 U	110	77 U	170	370
CYANIDE	***	0.29 U	0.30 U	0.29 U	0.29 U	0.28 U	0.32 U
pH	NS	6.7	7.3	7.3	7.8	8.4	7.0
TOTAL PHENOLICS	NS	1.4 U	1.8	1.4 U	1.4 U	1.4 U	1.6 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Table 5ESoil Analytical Results

TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5	PG-Q1-1	PG-Q1-1
Sample Date	Soil	11/29/2000	12/2/2000	12/2/2000	12/2/2000	11/30/2000	11/30/2000
Sample ID	Cleanup	PG-PD-3	PG-PD-4	PG-PD-5	PG-PD-5	PG-Q1-1	PG-Q1-1
Sample Depth	Objective	4-6'	8-10'	0.7-1.5'	2-4'	2-4'	4-6'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	42 U	41 U	62	38 U	39 U	260
OIL & GREASE	NS	82 U	81 U	190	75 U	280	810
CYANIDE	***	0.31 U	0.3 U	0.28 U	0.28 U	0.29	0.45
pH	NS	8.8	7.7	12	10	8.9	7.4
TOTAL PHENOLICS	NS	1.5 U	1.5 U	1.4 U	1.4 U	1.4 U	1.9 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Oil & Grease over 10,000 mg/kg have been bolded and highlight

138

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-RR-1	PG-RR-1	PG-RR-2	PG-RR-3	PG-RR-4	PG-RR-4
Sample Date	Soil	11/3/2000	11/3/2000	11/4/2000	11/3/2000	11/4/2000	11/4/2000
Sample ID	Cleanup	PG-RR-01	PG-RR-01	PG-RR-02	PG-RR-03	PG-RR-04	PG-RR-04
Sample Depth	Objective	0-1.2'	1.2-2'	0-2'	1.5-2'	0.6-2'	3-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	130	42	40 U	110	160	71
OIL & GREASE	NS	NA	NA	NA	NA	NA	NA
CYANIDE	***	0.68	0.47	0.29 U	0.29 U	0.28 U	0.31 U
pH	NS	NA	NA	NA	NA	NA	NA
TOTAL PHENOLICS	NS	1.4 U	1.4 U	1.5 U	1.4 U	1.4 U	1.6 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-RR-5	PG-RR-5	PG-RR-6	PG-RR-6	PG-RR-7	PG-RR-7
Sample Date	Soil	11/7/2000	11/7/2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PG-RR-05	PG-RR-05	PG-RR-06	PG-RR-06	PG-RR-07	PG-RR-07
Sample Depth	Objective	1-2'	2-4'	1-2'	2-4'	0-2'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	37 U ·	37 U	37 U	37 U	86	250
OIL & GREASE	NS	NA	NA	72 U	72 U	290	7500
CYANIDE	***	0.27 U					
pH	NS	NA	NA	6.5	5.0	3.8	4.7
TOTAL PHENOLICS	NS	1.4 U	1.3 U	1.4 U	1.4 U	1.3 U	1.4 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-STAIN-02	PG-STAIN-02	PG-STAIN-03	PG-STAIN-03	PG-STAIN-3B
Sample Date	Soil	11/11/2000	11/11/2000	11/10/2000	11/10/2000	11/11/2000
Sample ID	Cleanup	PG-ST-02	PG-ST-02	PG-ST-03	PG-ST-03	PG-ST-3B
Sample Depth	Objective	1-2'	2-3'	1.7-2.5'	2.5-3.5'	0-2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	40 U	49	3600	170	680
OIL & GREASE	NS	78 U	80 U	7400	84 U	1500
CYANIDE	***	0.29 U	0.30 U	0.29 U	0.32 U	0.27 U
pH	NS	7.2	7.0	5.3	4.3	6.9
TOTAL PHENOLICS	NS	1.5 U	1.5 U	5.5	1.6 U	1.3 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results

TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-STAIN-3B	PG-FS-2	PG-FS-2	PG-FS-2	PG-FS-3	PG-FS-3
Sample Date	Soil	11/11/2000	11/17/2000	11/17/2000	11/17/2000	11/15/2000	11/15/2000
Sample ID	Cleanup	PG-ST-3B	PG-FS-02	PG-FS-02	PG-FS-02	PG-FS03	PG-FS03
Sample Depth	Objective	2-4'	2-4'	8-10'	17-18'	2-4'	6.5-8'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	37 U	70	41 U	38 U	49	40 U
OIL & GREASE	NS	72 U	76	130	75 U	77 U	78 U
CYANIDE	***	0.27 U	0.28 U	0.30 U	0.28 U	0.29 U	0.29 U
pH	NS	6.9	7.2	6.6	7.7	6.6	7.2
TOTAL PHENOLICS	NS	1.3 U	1.4 U	1.5 U	1.4 U	1.4 U	1.5 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Table 5E Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-FILL-02	PG-FILL-03	PG-FILL-03	PG-FILL-03	PG-FILL-04	PG-FILL-04
Sample Date	Soil	11/3/2000	11/4/2000	11/6/2000	11/6/2000	11/6/2000	11/6/2000
Sample ID	Cleanup	PGFILL02	PGFILL03	PGFILL03	PGFILL03	PGFILL04	PGFILL04
Sample Depth	Objective	0.7-3'	0.5-2'	2-4'	4-6'	0-2'	2-4'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	330	100	69	38 U	120	45
OIL & GREASE	NS	1400	12000	11000	190	640	310
CYANIDE	***	0.32 U	0.27 U	0.28 U	0.28 U	0.29 U	0.28 U
рН	NS	7.2	7.7	7.5	7.9	6.8	5.3
TOTAL PHENOLICS	NS	1.4 U					

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-FILL-04	PG-FILL-04	PG-FILL-5	PG-FILL-5	PG-FILL-10	PG-FILL-10
Sample Date	Soil	11/6/2000	11/6/2000	11/18/2000	11/18/2000	12/1/2000	12/1/2000
Sample ID	Cleanup	PGFILL04	PGFILL04	PG-FILL-5	PG-FILL-5	PG-FILL10	PG-FILL10
Sample Depth	Objective	4-6'	7-8'	2-4'	6-8'	3-4'	6.4-8.2'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	39 U	43 U ,	37 U	40 U	37 U	160
OIL & GREASE	NS	150	83 U	72 U	78 U	95	2000
CYANIDE	***	0.29 U	0.31 U	0.27 U	0.29 U	0.27 U	0.6 U
pH	NS	6.3	6.6	7.2	6.2	8.2	7.5
TOTAL PHENOLICS	NS	1.4 U	1.6 U	1.3 U	1.4 U	1.4 U	3.0 U

U Undetectable Levels

1999

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Table 5E Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-UST1-2	PG-UST1-2	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2
Sample Date	Soil	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000	11/20/2000
Sample ID	Cleanup	PG-UST1-2	PG-UST1-2	PG-UST4-1	PG-UST4-1	PG-UST4-1	PG-UST4-2
Sample Depth	Objective	2-4'	12-14'	2-4'	10-11.5'	14-15.5'	4-6'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
						-	
PETROLEUM HYDROCARBONS	NS	60	41 U	200	45 U	92	51 U
OIL & GREASE	NS	270	7 9 U	3300	690	220	470
CYANIDE	***	0.27 U	0.30 U	0.32 U	0.77	0.32 U	0.33 U
рН	NS	7.1	6.7	6.5	7.8	10	7.4
TOTAL PHENOLICS	NS	1.3 U	1.5 U	1.6 U	1.7 U	1.6 U	1.7 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-UST4-2	PG-UST7-1	PG-UST7-1A	PG-UST7-1B	PG-UST7-2
Sample Date	Soil	11/20/2000	11/28/2000	11/28/2000	11/28/2000	11/21/2000
Sample ID	Cleanup	PG-UST4-2	PG-UST7-1	PG-UST7-1A	PG-UST7-1B	PG-UST7-2
Sample Depth	Objective	12-14'	8-10'	0-2'	2-3.5'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	150	290	5500	12000	1100
OIL & GREASE	NS	830	580	14000	24000	2200
CYANIDE	***	0.37 U	0. 29 U	0.27 U	0.28 U	0.29 U
pH	NS	11	9.1	8.4	8.2	7.3
TOTAL PHENOLICS	NS	1.9 U	1.5 U	1.3 U	1.4 U	1.4 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-UST7-2	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7	PG-PA-MW-7
Sample Date	Soil	11/21/2000	11/13/2000	11/13/2000	11/13/2000	11/13/2000
Sample ID	Cleanup	PG-UST7-2	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07	PG-PAMW-07
Sample Depth	Objective	10-12'	2.5-4'	4-6'	6-8'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	460	38 U	51	41 U	39 U
OIL & GREASE	NS	5300	74 U	840	80	76 U
CYANIDE	***	0.30 U	0.65	0.59	0.30 U	0.28 U
pH	NS	7.4	8.0	8.4	7.5	9.3
TOTAL PHENOLICS	NS	1.5 U	1.4 U	1.4 U	1.5 U	1.4 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and

Table 5E Soil Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	PG-PA-MW-10D	PG-PA-MW-10D	PG-MW10D	PG-MW10D
Sample Date	Soil	11/27/2000	11/27/2000	11/27/2000	11/27/2000
Sample ID	Cleanup	PG-PAMW10D	PG-PAMW10D	PG-MW10D	PG-MW10D
Sample Depth	Objective	0.5-2'	4-6'	7-8'	8-10'
Concentration	MG/KG	MG/KG	MG/KG	MG/KG	MG/KG
PETROLEUM HYDROCARBONS	NS	38 U	39	2400	660
OIL & GREASE	NS	140	83	11000	55000
CYANIDE	***	0.28 U	0.28 U	0.34 U	0.38 U
pH	NS	7.6	6.9	6.8	5.2
TOTAL PHENOLICS	NS	1.4 U	1.4 U	1.7 U	1.9 U

U Undetectable Levels

NS No Standard

NA Not Analyzed

*** Site Specfifc Standard

Although there is no standard, Petroleum Hydrocarbons and



Site 2A/2B. Generally, SI soil samples were collected from the intervals revealing indications of contamination based on field screening and/or the presence of fill material.

5.3.1 Potential UST Areas

Four potential UST areas (UST1, UST3, UST4 and UST7) were identified at Site 2A/2B. As described in Section 5.2, a GPR/EM survey was performed at each area. In accordance with the ESIW developed for potential UST areas, soil borings were installed and sampled from each potential UST area. Specifically seven soil borings were installed to evaluate potential UST Areas. One soil boring was installed and sampled at UST1 (UST1-2). Two soil samples were collected from the soil boring installed at UST1 and submitted for laboratory analyses. No soil boring was installed at proposed location UST1-1 due to the presence of concrete and underground utilities. Several other soil borings were installed to the south of the UST1 area. In addition, no the subsurface investigation for UST3 was combined with the investigation of "former structures" in the Buildings 12/13 area. Please see discussion under Section 5.3.6. Two soil borings, UST4-1 and UST4-2, were installed at UST4. Five soil samples were collected from UST4 Area and submitted for laboratory analyses. Four soil borings, UST7-1, UST7-1A, UST7-1B, and UST7-2 were installed at UST7; soil borings UST7-1A and UST7-1B are not depicted on Figure 7 due to their close proximity to UST7-1. Five soil samples were collected from the soil borings installed at UST7 and submitted for laboratory analysis. One temporary monitoring well (TMW-01) was installed in the UST7 Area in order to further evaluate subsurface conditions. Sample designations and depths are summarized in Table 4 under the Potential USTs heading. All samples were analyzed for the parameters specified in the ESIW and listed on Table 4 included in Section 5.1.

5.3.2 Previously Identified AOCs (Areas A, B, E, F2, P, and Q1)

Thirteen soil borings were installed in Site 2A to evaluate those AOCs (Areas A, B, E, F2, P and Q1) previously identified by P&G on Site 2A/2B. With respect to the above listed areas, it was not possible to install two of the proposed soil borings: F2-1 at the F2 Area (Site 2A) and Q1-2 at the Q1 Area (Site 2B). Based on the presence of impediments, the need for installation of these borings and collection of samples was deferred until review of field screening information and analytical results from other samples collected at Areas F2 and Q1.



Hatch Mott MacDonald

Two soil borings (A-4 and A-5) were installed at Site 2A to evaluate Area A. Three soil samples were collected from the soil borings and submitted for laboratory analysis. It should be noted that the soil borings installed to evaluate the western limit of Area A, soil borings A-1, A-2, A-3, and A-6, are situated in Site 1. As such, the majority of the information generated through the installation and sampling of these borings was presented in the previously submitted Site 1 Report.

Five soil borings were installed to evaluate Area B (B-1, B-2, B-02A, B-3 and B-4); due to the scale of the map soil boring location B-2 and B-2A are both represented by the B-2 designation. Nine soil samples were collected from Area B and submitted for laboratory analysis. One soil boring (F2-2) was installed at Area F2 with two soil samples submitted for laboratory analysis. One soil boring was installed at Area E (E-1) with three samples submitted for laboratory analysis. Three soil borings (P-1, P-2, and P-3) were installed at Area P with six samples submitted for laboratory analysis. One soil boring (O1-1) was installed at Site 2B with two samples submitted for laboratory analysis.

Designations for samples collected from the above listed borings are summarized in Table 4 under the individual AOC headings. Analytical parameters for samples for each AOC were generally consistent with those proposed in the ESIW and also are presented in Table 4 $^{\triangleright}$

Railroad Tracks and Sidings 5.3.3

Thirteen soil borings were installed and sampled within Site 2A/2B to evaluate this AOC. Three additional soil borings (RR-12, RR-13 and RR-14) were proposed for railroad tracks and sidings at Site 2A. Given the number of other soil borings which were installed and sampled to evaluate this AOC, it was determined adequate coverage was accomplished without these locations with respect to railroad tracks and sidings at Site 2A. However, the need for additional sampling was to be based upon field screening information and analytical results from sampling at other railroad track and siding samples locations.

The thirteen soil borings installed to evaluate railroad tracks and sidings are as follows: RR-01, RR-02, RR-03, RR-04, RR-05, RR-06, RR-07, A-4, B-4, P-1, P-2, P-3 and Q1-1. Twenty-four soil samples were collected from the soil borings to evaluate this AOC. The sample designations and depths are presented in Table 4^vunder the Railroad Tracks and Sidings heading. Analytical parameters for samples for this AOC were generally consistent with those proposed in the ESIW and also are presented in Table 4. \checkmark



5.3.4 Surface Staining

Five soil borings were installed at locations within Site 2A to evaluate this AOC and in particular staining noted at Building 20. The soil borings are as follows: STAIN-02, STAIN-03, STAIN-3B, RR-06 and RR-07. Soil boring STAIN-3 was installed in the center of a stained area situated within Building 20 and STAIN-3B was installed at the fringe of the same stained area. Due to close proximity of these two soil borings to one another, only soil boring STAIN-3 is presented on Figure 7. Ten samples were collected from these borings and submitted for laboratory analysis. The sample designations and depths for samples collected from soil borings installed to evaluate this AOC are presented in Table 4. Analytical parameters for samples for this AOC were generally consistent with those proposed in the ESIW and also are presented in Table 4.

5.3.5 Pits and Drains

Eleven soil borings were installed at locations within Site 2A/2B to evaluate this AOC. In addition, due to the presence of reinforced concrete and/or utilities, it was not possible to install soil borings at several additional locations (RR-12, RR-13, RR-14, Q1-2, PD-13), which were originally proposed to evaluate pits and drains at Site 2A/2B. Based on information from other area samples, these additional samples were not deemed warranted as part of the SI for pits and drains.

The eleven soil borings installed to evaluate pits and drains at Site 2A/2B are as follows: PD-1, PD-3, PD-4, PD-4A, PD-5, A-4, P-1, P-2, P-3, RR-03, STAIN-02. Twenty-two samples were collected from these borings and submitted for laboratory analysis. The sample designations and depths for samples collected from soil borings installed to evaluate this AOC are presented in Table 4 under the Pits and Drains heading. It should be noted that a soil boring was also installed at location adjacent to PD-4 (designated as PD-4A) to provide additional visual review given the presence of subsurface concrete; no sample (for analysis) was collected from this location. Analytical parameters for samples for this AOC were generally consistent with those proposed in the ESIW and also are presented in Table 4.

5.3.6 Former Structures

Ten soil borings were installed at locations within Site 2A/2B to evaluate this AOC. The soil borings are as follows: FS-1, FS-1A, FS-2, FS-3, PAMW-7, RR-01, RR-04, RR-05, PD-1 and PD-3. Soil borings FS-1, FS-1A, FS-2 and FS-3 were also utilized to review potential UST area, UST3. Sixteen samples



were collected from these borings and submitted for laboratory analysis. It should be noted that no soil samples were collected from soil borings FS-1 and FS-1A. A discussion of analytical results from samples collected from soil boring FS-1B situated in Site 1 was provided in the Site 1 Report. These soil borings were field screened and information was utilized to evaluate issues associated with former structures in the vicinity of FS-1/1A/1B. Based on visual review at these locations, no further sampling was warranted at the FS-1/1A locations. Again, additional information associated with the FS-1B location is provided in the Site 1 Report. The sample designations and depths for samples collected from soil borings installed to evaluate this AOC are presented in Table 4 under the Former Structures heading. Analytical parameters for samples for this AOC were generally consistent with those proposed in the ESIW and also are presented in Table 4.

5.3.7 Historic Fill Material

As previously stated, all soil borings installed during the site investigation were utilized as part of the sitewide historic fill evaluation. Similarly, all soil borings installed at Site 2A/2B were utilized to characterize historic fill material at this portion of the HHMT-Port Ivory facility. Seven additional soil borings Fill-1, Fill-2, Fill-3, Fill-4, Fill-5, Fill-9/PA-MW-10D and Fill-10 were installed at locations within Site 2A/2B to evaluate historic fill material. Twenty soil samples were collected from the seven soil borings and submitted for laboratory analysis. No sample was collected from location Fill-1. In addition, Fill-6 was not able to be installed due to the presence of utilities. Given the comprehensive nature of the fill evaluation at Site 2A/2B and the entire HHMT-Port Ivory Facility, it was determined that sufficient sampling had been performed to evaluate fill materials. However, the need for additional sampling at the Fill-6 location was to be based upon field screening information and analytical results from the overall fill evaluation effort.

A total of 47 soil borings were installed and sampled (including locations at potential UST areas) to evaluate historic fill material at Site 2A/2B. Please note, the information provided in Table 4 under the Fill Material heading presented information related to the four additional soil borings installed to evaluate historic fill material. Analytical parameters for samples for this AOC were as proposed in the ESIW and are presented in Table 4.



5.4 Monitoring Well Installation

The site-wide groundwater investigation included the installation and sampling of 17 wells and the sampling of 14 existing wells. In conjunction with this investigation, five monitoring wells (PA-MW-7, PA-MW-7D, PA-MW-10D, PA-MW-15, PA-MW-15D) and one temporary well (TMW-01) were installed at Site 2A and one monitoring well (MW-10D) was installed at Site 2B. One additional well, MW-10 was proposed to be installed at Site 2B. However, it was determined that existing well shallow well, MW-3 provided adequate coverage at that area. Prior to installation of the monitoring well, borings were constructed to identify soils and contamination, if any, at the proposed monitoring well locations. All wells were installed in the overburden layer. However, to determine conditions both above and below the naturally occurring peat layer, three shallow wells were screened in fill or native material above the peat layer (where present) and two deep wells were screened in native material below the peat layer. The temporary well was screened in fill material above the peat layer. At two locations on Site 2A, monitoring wells were placed to form a shallow/deep well pair in combination with existing well, MW-3.

The monitoring wells were constructed of two-inch outside diameter (O.D.) Schedule 40 polyvinyl chloride (PVC) pipe in a borehole that was eight inches in diameter. The boreholes were drilled with a truck mounted drill rig utilizing HSAs and <u>mud rotary drilling techniques</u>. As described in the soil boring methodology section, hand augering was performed for the first six feet of each location to avoid contacting utilities. The screens of the shallow wells were set across the uppermost water table using tenfoot sections of 0.02-inch (20 slot) slotted screen. The screens of the deep wells were set in the most permeable layer above bedrock and below the peat layer (as present) using ten-foot sections of 0.01 (10 slot) slotted screen. The screened sections of the wells were packed with well-graded sand pack, 95 percent coarser than the slot size, which extended one foot above the top of the screen. The elevation of the top of the sand was checked by lowering a weighted measuring tape in the annular space of the wells. To avoid bridging, both the sand and granular bentonite seal were installed by carefully placing small quantities of sand and granular bentonite. The remaining annular space was backfilled with a cement/bentonite grout mix.

Site 2A/2B Report



The wells were completed at the ground surface by either extending a PVC riser to approximately three feet above grade, with a locking compression cap and a fitted steel protective casing over the PVC and embedded into a concrete pad constructed at the ground surface; or the PVC was cut to grade and equipped with a locking compression cap and a steel protective flush mount to fit over the PVC and embedded into the ground surface in a pad constructed of concrete. A keyed-alike lock was installed on the steel casing and the compression cap to hinder tampering with the wells. The concrete pads were sloped away from the wells to prevent water from collecting around same.

Following completion, the newly installed wells and all existing wells to be sampled were developed. Existing wells were redeveloped due to the prolonged time period from previous sampling efforts. A permanent water level measurement mark was etched on top of the PVC casing to allow for accurate, and consistent water level measurements over time. In accordance with Port Authority protocol, the monitoring wells were allowed to equilibrate for a period of two weeks prior to sampling, as feasible given project time constraints.

5.5 Groundwater Sampling

Groundwater sampling was performed in November and December 2000. Prior to sampling activities, the following was performed: condition of each monitoring well was visually inspected for signs of damage or tampering, the lock and well cap was removed so a PID reading could be obtained, depth of water, depth of free product (if present), and depth to bottom of each monitoring well was obtained and recorded.

All monitoring wells were purged prior to sampling. Purging was accomplished by removing a predetermined volume of standing water utilizing a peristaltic or submersible pump. The purge rate depended on the yield of the well, and did not exceed the well development discharge rate. At the start and end of the purging process, the discharge water was monitored and recorded for pH, temperature, dissolved oxygen, turbidity, and specific conductivity.

Subsequent to the completion of purging, groundwater samples were collected after the well had recovered to a volume sufficient for sampling, or no later than two hours from the end time of purging. Samples were collected using poly-Teflon bailers. Bailers were lowered into the well at the screened interval. Once the bailer was filled, it was retrieved and the groundwater was poured into the proper laboratory containers while minimizing aeration. The containers were then labeled, placed on ice, and



delivered to the laboratory for analysis. As previously stated, 10 groundwater samples were collected from wells located on Site 2 as part of the site-wide groundwater investigation. Specifically, the following wells were sampled: PG-PA-MW-7, PG-PA-MW-7D, PG-PA-MW-10D, PG-PA-MW-15, and PG-PA-MW-15D (five newly installed wells); PG-TMW-01 (one temporary well); and, PG-MW-3, PG-GW-10, PG-GW-7, and PG-PZ-1 (four existing wells). Please note, due to the presence of sediment/indications of high turbidity, samples were collected from PG-PA-MW-15D on November 20, and November 30, 2000. Free product samples were collected from PG-OP-1 and PG-GW-14 for GC fingerprint analysis.

5.6 Surface Water and Sediment Inspection and Sampling

HMM performed several visual inspections of Bridge Creek during both low and high tide events during the weeks of October 29, 2000, November 5, 2000 and November 13, 2000. The purpose of the inspections was to determine if the precipitate material identified in environmental reports provided by P&G were present. The visual inspections revealed the presence of a "white-ish precipitate" material at numerous locations in near shore sediments during low tide. All locations were located at the northern portion of Bridge Creek, proximate to Site 1. The material was not observed to be present at any locations during high tide periods. As such, surface and sediment sampling was performed during low tide on November 21, 2000. In each case, sediment samples included the "white-ish precipitate" material and surface water samples were taken from locations adjacent to Site 1. As such, no further information pertaining to SI sampling of Bridge Creek is provided in this report. Sampling of Bridge Creek (sediment and surface water) was performed at locations adjacent to Site 2A in conjunction with RI activities. Information pertaining to RI sampling is provided in Sections 8.3 and 9.3 of this report.

6.0 SI RESULTS

The SI for Site 2A/2B consisted of a variety of tasks designed to establish current (year 2000) environmental conditions for the purposes of acquisition and potential site redevelopment. The results of the SI efforts are presented in this section.



Soil and groundwater sampling efforts have provided information to better characterize site hydrogeology including the types and general extent of fill material present at this site. Given that fill material was present at the site prior to the Port Authority's purchase, fill material has been regarded as an existing condition and is referenced as historic fill to reflect same. Soil and groundwater are described below. As stated in Section 5.6, the SI included the collection of sediment and surface water samples from locations adjacent to Site 1. As such, sampling information including analytical results is included in the Site 1 Report. Sampling of Bridge Creek (sediment and surface water) was performed at locations adjacent to Site 2A in conjunction with RI activities. Information pertaining to RI sampling is provided in Sections

8.3 and 9.3 of this report

6.1.1 Soil

Three general types of historic fill material were identified through the SI program with regard to Site 2A/2B: (1) urban fill including non-native soil, vegetative debris, construction debris (wood, bricks, glass, concrete); (2) cinder fill consisting primarily of ash and ash-type materials with some slag; and, (3) by-products from on-site production and manufacturing activities (calcium carbonate, spent diatomaceous earth, and spent carbonaceous filter material). The specific composition of the historic fill was noted to vary with location and frequently all three types were noted in varying quantities at the same location. Urban fill was observed at varying thicknesses and depths at locations throughout Site 2A and 2B. In particular, non-native soil (reddish brown silty clay with coarse gravel or coarse-to-fine brown sand) was noted to be present at the northern, eastern, and southern portions of Site 2A as observed at soil boring locations Fill-5, P-1, P-3, PD-4B, FS-3, F2-2, UST 7-1 and PA-MW-7.

The second type of historic fill material, cinders, was noted to be present in notable quantities in certain locations within Site 2A and 2B. Cinder fill was observed in the central portion of Site 2A in soil borings A-5, B-2, B-4, B-3, P-2, FS-1A, E-1, F1-3, UST1-2, UST4-1 and PA-MW-15 as well as at Site B in soil boring Fill-9/PA-MW-10D.

The third type of historic fill material, by-product fill, was encountered just underneath the surface as well as at varying depths primarily in the western to central/northern portion of Site 2A. The by-product fill ranged in color from white to green to blue to gray and black in various depths and widths in boring locations on Site 2A. The by-product fill material was readily distinguishable from the other types of



historic fill encountered. The by-product fill material has very small grain sizes and takes on paste/powder-like characteristics, and in some locations is pure white.

Native material has been defined as peat or very well sorted light brown to orange brown to red brown sands and silts. The majority of the borings installed during the SI were advanced to the depth of the brown peat layer. The brown peat horizon was located at depths ranging from 10 to 20 feet bgs. The peat layer was noted to be present at most, but not all, boring locations. In some instances, the peat layer may exists below the completion depth of the soil borings. SI soil boring logs, including those logs for borings, which were finished as monitoring wells are provided in Appendix B. Figure 5, Cross-Section, presents soil conditions at Site 2A/2B

6.1.2 Groundwater

Based on the findings of HMM's SI groundwater program and considering the data generated by prior site investigation activities, the general hydrogeologic regime in overburden saturated zones consists of two components; an upper aquifer, comprised of unconsolidated materials described previously in Section 3.2.2 (indigenous granular soils, operational by-product fill and/or non-indigenous fill materials), and; a deeper water bearing zone comprised of native glacio-fluvial sediments (i.e., gravel, sand, silt and clay). A discontinuous peat layer that occurs at approximately 10 to 24 feet bgs separates these zones of saturation. This peat layer was encountered in the majority of the soil borings installed during HMM's SI program. Please note, the groundwater description provided herein generally reflects site-wide conditions. As appropriate, specific commentary is provided for Site 2A/2B.

Figure 8 depicts the plot of the water table contours for the shallow overburden aquifer during the November 2000 groundwater sampling event. As depicted on Figure 8, the shallow overburden saturated zone exhibits a hydraulic gradient of variable orientation and magnitude. The flow regime for the shallow, saturated overburden zone reflects the non-homogenous nature of the upper surface aquifer. The groundwater contour plot depicts directional flow components to the north, west, and southwest, with a groundwater flow divide oriented from northwest to southeast, across the northern portion of the site. Groundwater flow direction in the upper aquifer across the central and southern portion of Site 2A is generally from the northeast to the southwest flowing towards Bridge Creek. In the northern portion of the site the flow direction changes from the southwest to the northeast, towards the Arthur Kill. Gradual hydraulic gradients are mapped for the southern and central portion of Site 2A, adjacent to Bridge Creek. Less permeable fill materials (by-product fill) deposited in this portion of the site may contribute to the

157



Site 2A/2B Report

mounding/hydraulic divide effects observed along the western boundary of Site 2A. Flow dynamics in the upper aquifer, particularly the Site 2A portion of the Site 2A/2B Site, must take into consideration the low permeability and low porosity characteristics of the by-product fill material. The thicknesses and depths of the by-product fill, may cause changes in flow patterns as groundwater is diverted either horizontally or vertically to find the path of least resistance. The data set indicates shallow hydraulic gradients in the northeast and southwest portions of the site (Site 2A and 3) on the flanks of the groundwater divide. These are areas where SI borings indicated either granular, reworked native soils along with soil fill, or in-situ, undisturbed soil deposits. Indigenous, unconsolidated granular soils consisting of gravel, sand and silt, with little to trace quantities of clay, were encountered in the deep monitoring well borings installed as part of HMM's SI program (MW-7D, PG-PA-MW-10D, and MW-15D). As described previously, deep wells were screened in the first encountered permeable formation beneath the observed (or inferred) depth of the peat layer.

Hatch Mott

MacDonald

Figure 9, Water Table Contour Map, depicts the plot of the groundwater surface for the November 2000 groundwater sampling event. As depicted on Figure 9, the deep overburden saturated zone exhibits a consistent magnitude of hydraulic gradient across the entire site, and indicates flow direction components toward adjacent surface water bodies (Bridge Creek to the west, the Arthur Kill to the west/southwest and the Kill Van Kull to the North). Across Site 2A/2B, the flow direction is in the northwesterly direction with relatively constant gradient, as it moves towards Bridge Creek. This contrasts with Figure 8 for the shallow overburden aquifer underlying Site 2A, which reflected the effects on hydraulic magnitude and gradient due to aquifer heterogeneity. This disparity in the magnitude and orientation of hydraulic gradient between the shallow and deep overburden aquifers suggests that the peat layer and/or less permeable fill materials limit vertical groundwater movement from the shallow, overburden aquifer to the deep overburden aquifer, establishing predominantly horizontal flow conditions in each of these saturated overburden zones.

HMM employed the use of data loggers to assess whether groundwater movement beneath the subject site was influenced by the tidal fluctuations of the surface water body adjacent to the northern end of Future Site 4. Data loggers were placed in four shallow monitoring wells and one deep monitoring well to monitor groundwater fluctuations in the shallow and deep overburden aquifers. An additional data logger station was established on the Kill Van Kull to monitor fluctuations in the surface water elevation of that water body. Review of the collected elevation data indicated no correlation between tidal fluctuations of the Kill Van Kull and groundwater levels in the shallow and deep aquifers beneath the site. However,

159

 \mathcal{X}



recent site building demolition and redevelopment activities appear to contradict this information. Specifically, variable water levels have been observed in shallow excavations installed as part of the removal of concrete pads and foundations and the subsurface modifications to utilities.

6.2 GPR/EM - Potential UST Areas

Due to a limited GPR signal, presumably the result of surface and subsurface features such as concrete slabs, metal piping and rail spurs. The GPR/EM survey proved inconclusive with regard to identifying USTs at the four potential locations at Site 2A (UST1, UST3, UST4 and UST7). However, based on information obtained through the GPR/EM survey, soil borings were installed to further evaluate conditions at each of these potential UST areas; soil borings for UST3 were combined with proposed sampling for former structures at Buildings 12/13. Soil boring locations were selected through the findings of the Hager-Richter survey as well as available information from soil borings installed at adjacent areas to evaluate other AOCs. In one location (UST7 Area), a temporary well was installed, in addition to soil borings, to further evaluate the subsurface conditions and attempt to identify impacts to groundwater, if any. A discussion of the sampling frequency for each potential UST area is provided in Section 5.3.1. No potential UST areas were identified at Site 2B.

6.3 Soil Analytical Data

As described in Section 5, 86 soil samples were collected from 47 soil borings at Site 2A/2B. The locations of the SI soil borings are presented on Figure 7. The samples from the soil borings were submitted for specific laboratory analysis based upon the types of contaminants likely to be present at each AOC. Table 4 provides the analyses for specific samples. The analytical results for HMM's sampling efforts of soil are presented in Tables 5A-5F; presented on the following pages. Figures 10 through 16 provide soil boring locations as well as pertinent analytical data; the figures are presented after first reference in the following subsections. For discussion purposes, the results have been compared, as appropriate, to current NYSDEC regulatory criteria. The criteria utilized are Recommended Soil Cleanup Objectives (RSCOs) as set forth in the December 2000 NYSDEC Division of Technical and Administrative Guidance Memorandum (TAGM) 4046, which incorporates the guidance criteria included in the NYSDEC Spill Technology and Remediation Series (STARS). Please note, the reference of these standards in this report does not represent any agreement or concurrence that same are appropriate for usage at this site. In addition, reports of previous investigations described the presence of fat, oil, grease or "FOG" in soil at the site. As such, one of the goals of the SI phase of this project was to identify issues associated with petroleum and non-petroleum substances in soil at the site. In the absence of a regulatory

SUIL DAT



а

standard for TPHC, O/G or FOG, a threshold value of 10,000 mg/kg was employed for TPHC and O/G in soil samples for this SI. This threshold value was utilized for comparison purposes only and to identify any areas, which might warrant additional subsurface review prior to site development. This threshold value is not intended as a site specific standard for petroleum or non-petroleum materials in soil at this site. A discussion of the analytical results from the soil component of the SI is provided below.

6.3.1 Volatile Organic Compounds

VOCs were either not detected or were detected below NYSDEC RSCOs in all but three soil samples collected from Site 2A/2B. Specifically, trans-1,3-dichloropropene was detected at a concentration of 0.43 mg/kg from sample B-4 (5-6 feet); dichloromethane was detected at a concentration of 0.17 mg/kg from sample PG-STAIN-03; and benzene was detected at a concentration of 0.32 mg/kg from sample PG-MW10D. Two of the three samples, B-4 (5-6 feet) and PG-STAIN-03 were collected from locations on Site 2A and the other sample, PG-MW-10D (8-10 feet), was collected from a location at Site 2B. The total VOC concentration was below the NYSDEC guidance criteria of 10 mg/kg for all samples from Site 2A/2B including the results for the above three listed samples. In fact, the highest concentration of total VOCs is 1.99 mg/kg, detected in sample PG-MW-10D (8-10 feet). Total VOCs have been calculated for each sample and are presented in Table 5A. Please refer to Table 5A and Figure 10 for VOC results.

6.3.2 Semi-Volatile Organic Compounds

A number of SVOCs were detected in soil samples collected from Site 2A/2B. However, the vast majority of these compounds were detected below NYSDEC TAGM RSCOs. The following SVOCs were detected at concentrations above corresponding RSCOs in one or more soil samples from Site 2A/2B: pyrene, phenanthrene, acenaphthene, anthracene, naphthalene, benzo(b)fluoranthene, benzo(a)pyrene, benzo(a)anthracene, dibenz[a,h]anthracene, benzo(k)flouranthene, indeno(1,2,3-cd)pyrene, flouranthene, flourene, and phenol. Analytical results revealed the presence of SVOCs in excess of 50 mg/kg in only one sample, Sample PG-STAIN-03, which was collected from within a stained area at Building 20. The SVOCs detected in excess of 50 mg/kg are as follows: acenaphthene, anthracene, benzo(a)anthracene, benzo(b)flouranthene, flourenthene, flourenthene, flourenthene, and pyrene. One total SVOC concentration, PG-STAIN-03 was detected in excess of the NYSDEC guidance criteria of 500 mg/kg for total SVOCs. Again, this sample was collected from a "stained" area from the surfical interval. The sample collected from the deeper interval at location PG-STAIN-3 and samples collected form the surficial and subsurface intervals of an adjacent location (PG-STAIN-3B) at the fringe of the stained area revealed significantly lower concentrations of SVOCs. Please refer to Table







		of				
7 (0) Re	Ŋ				
G P	ROGRA	AM MA	NAGEF	2		
					-	
F PIN	Revisio	n DEDA	DTM	ENIT	Appro	ved
RII	VG L	VEPA	RIM	ENI		
an da 1 h ang an 1 h ang a						
Service and a service of				n a a general and the second secon		
Ck	(14	.00	10	T 1	ΔN	D
00 20		133 IVC	8, L RY	OT FA	1) CIL	.ITY
	SITI	E 2/	\/2 E	3		
RI D	BU1 SV(FIOI DCs	N O	F V SO	OC: IL,	5
EI	INV	ES7	ſIG	ATI(ON	
) su ns, res us	ibject ideas servec sed wi	to co s, des d to 'thout	onditio signs Port its w	ns in and Aúthc rritten	contr meth ority cons	act. ods and ent.
у	Dro	awn t	ру	Che	cked	by
	FIC	JUR	E 1	10		

5B and Figure 10 for a summary of SVOC results for soil at Sites 2A/2B. Total SVOCs have been calculated and are presented in Table 5B.

6.3.3 Polychlorinated Biphenyls

Hatch Mott

MacDonald

No specific PCBs were detected in soil samples collected from Site 2A/2B. No samples collected from the surficical interval exhibited a concentration in excess of the NYSDEC TAGM RSCO for surface soil of 1 mg/kg. None of the samples collected from the subsurface exceeded the NYSDEC guidance criteria for PCBs in subsurface soil of 10 mg/kg. Please refer to Table 5C and Figure 11 for a summary of PCB results.

6.3.4 Pesticides

Five pesticide compounds were detected in soil samples collected from various locations at Site 2A/2B. Dieldrin, endrin, heptachlor, heptachlor epoxide, and chlordane were the only contaminants detected in excess of corresponding NYSDEC TAGM RSCOs in samples collected from Site 2A/2B. Specifically, dieldrin was detected at concentrations in excess of its corresponding RSCO in three samples; endrin was detected at concentrations in excess of its corresponding RSCO in two samples; heptachlor was detected in concentrations in excess of its corresponding RSCO in two samples; heptachlor was detected in concentrations in excess of its corresponding RSCO in one sample; chlordane was detected in excess of its RSCO in two samples; and, heptachlor epoxide was detected in excess of its RSCO in only a single sample. Please refer to Table 5C and Figure 11 for a summary of pesticides results.

6.3.5 Metals

All TAL metals were detected in at least one soil sample collected as part of the SI of Sites 2A and 2B. The NYSDEC TAGM generally regards site background as an appropriate concentration for the 24 TAL metals and only provides RSCOs for a portion of the metals included in the TAL. RSCOs are provided for the following metals: arsenic, barium, beryllium, cadmium, chromium, cobalt, copper, iron, mercury, nickel, selenium, vanadium and zinc. Given the presence of fill material and the urban nature of the site, it is difficult to establish a site background concentration for metals. As such, in the absence of a specified RCSO, the upper limit of the Eastern USA Background Range, as provided in the TAGM was utilized for comparison purposes. It is important to recognize that the presence of a metal above an established background concentration does not constitute an exceedance of a regulatory standard. As the NYSDEC TAGM does not include a regional background concentration nor RSCOs for antimony or silver, no discussion of exceedances is provided for these metals. Analytical results revealed

164



,

*

THE FORT AUTHORITY OFNY&NJ ENGINEERING PROGRAM MANAGER No. Date Revision Approved ENGINEERING DEPARTMENT (BLOCK 1400, LOT 1 AND BLOCK 1338, LOT 1) HHMT-PORT IVORY FACILITY SITE 2A/2B **DISTRIBUTION OF PESTICIDES AND PCBs IN** SOIL, SITE INVESTIGATION This drawing subject to conditions in contract. All inventions, ideas, designs and methods herein are reserved to Port Authority and may not be used without its written consent. Designed by Drawn by Checked by

of

FIGURE 11

м.

1

.

Site 2A/2B Report



Hatch Mott MacDonald

exceedances of RSCOs in one or more soil samples for 12 of the 13 of the metals with established guidance criteria; antimony was not detected in excess of its RSCO in any of the soil samples from Site 2A/2B. It should be noted that the NYSDEC has not established a guidance criteria for lead but does review concentrations related to this metal on a case by case basis. The Eastern US Background guidance for lead is 500 mg/kg. Analytical results from this sampling performed at Site 2A/2B revealed concentrations of lead ranging from not detected to 950 mg/kg.

Arsenic was detected in the majority of the samples collected from Site 2A/2B; the highest concentration of arsenic (980 mg/kg) was detected in sample PG-RR-1 (0-1.2 feet). Cadmium was detected above the RSCO ranging in four samples; concentrations above the RSCO ranged from 1.4 mg/kg in sample PG-RR-1 (0-1.2 feet) to 14 mg/kg in sample PG-B-02A (8-10 feet). Chromium was detected in excess of the RSCO in many of the samples; concentrations above the RSCO of 10 mg/kg ranged from 11 mg/kg in several samples to 83 mg/kg in sample PG-STAIN-3B (0-2 feet). Mercury was detected in excess of the RSCO of 0.1 mg/kg in 18 samples collected from Site 2A/2B. Exceedances ranged from 0.19 mg/kg in sample PG-FILL-03 (2-4 feet) to 1.2 mg/kg/in sample PG-E-1 (4-6 feet).

Please refer to Table 5D^V and Figure 12 for a summary of metals results.

6.3.6 Cyanide and Total Phenolics

Cyanide was detected in several soil samples collected from Site 2A/2B. In all instances, cyanide was detected at a concentration of less than 1 mg/kg. The NYSDEC has not established guidance criteria for cyanide in soil. Rather, the NYSDEC establishes guidance criteria for cyanide on a case-by-case basis. Please refer to Table 5E for a summary of cyanide results.

Total phenolics was detected in two samples collected from Site 2A/2B, both samples were collected from Site 2A. The concentrations were 1.8 mg/kg from sample PG-P-2 (4-6 feet) and 5.5 mg/kg from sample PG-STAIN-03 (1.7-2.5 feet). The NYSDEC has not established guidance criteria for total phenolics in soil. Rather, the NYSDEC establishes guidance criteria for total phenolics on a case-by-case basis. Please refer to Table 5E for a summary of total phenolics results.

-m


	ц	EGEND	
	UTILITY EASEMENT		
and the second sec	RAILROAD TRACKS	ANALYTE	INDICATES COMPOUND FOR WHICH LABORATORY ANALYSIS WAS CONDUCTED
	SITE BOUNDARY	RES	CONTAMINANT CONCENTRATION IN MG/KG
PG-PA-MW-5	YEAR 2000 SITE INVESTIGATION MONITORING WELL LOCATION	DEPTH	DEPTH IN FEET BELOW GROUND SURFACE
PG-WOOD-3	YEAR 2000 SITE INVESTIGATION SOIL BORING LOCATION		
PG-RS-1	PRE-EXISTING P&G MONITORING WELL LOCATION		
PG-58-2	PRE-EXISTING P&G SOIL BORING LOCATION		

HMT-PORT IVORY FACILIT									
SITE	SITE 2A/2 UTION OI TALS IN S INVESTIG	B SELECT SOIL, ATION							
nis drawing su II inventions, erein are res nay not be us	Ibject to conditi ideas, designs served to Port sed without its	ons in contract. s and methods Authority and written consent.							
esigned by ate	Drawn by	Checked by							
ontract lumber									
rawing umber	FIGURE	12							

(BLOCK 1400, LOT 1 AND DI OOK 4000 I OT 4)

Title

No. Date Revision Approved ENGINEERING DEPARTMENT

THE POPER AUTHORITY OFNYGNJ

ENGINEERING PROGRAM MANAGER

Sheet



6.3.7 Petroleum Hydrocarbons/Oil and Grease

TPHC and O/G were detected in the majority of SI soil samples collected from Site 2A/2B as well as throughout the site. The NYSDEC does not currently maintain a standard for TPHC or O/G in soil. For general guidance purposes related to total petroleum hydrocarbon concentrations, TPHC and O/G results were compared to a threshold value of 10,000 mg/kg. As described in Sections 6.3.1 and 6.3.2, samples were also analyzed for VOC and SVOC compounds and concentrations were compared to corresponding NYSDEC RSCOs including NYSDEC guidance values of 10 mg/kg for total VOCs, 50 mg/kg for individual SVOCs and 500 mg/kg for total SVOCs.

Overall, four samples from Site 2A/2B exhibited a concentration of TPHC above 10,000 mg/kg. All of the samples exhibiting TPHC at or above the guidance threshold were collected from locations at Site 2A. The samples were collected from locations at Area A, Area B and UST7. The concentrations above the guidance threshold are as follows: 12,000 mg/kg at sample PG-A-5 (2-4 feet); 15,000 mg/kg at sample PG-B-3 (2-4 feet); 13,000 at sample PG-B-3 (6-8 feet) and 12,000 mg/kg at sample UST7-1B (2-3.5 feet). Two of the samples also exhibited concentrations of O/G in excess of 10,000 mg/kg. TPHC was not detected above 10,000 mg/kg in any of the samples collected from Site 2B. O/G was detected at a concentration in excess of 10,000 mg/kg in eight samples. The detections above the guidance threshold were identified at locations from both Site 2A/2B and ranged from 11,000 mg/kg in sample PG-Fill-03 (2-4 feet) to 55,000 mg/kg in sample PG-MW-10D. Two of the eight samples from Site 2A/2B also exhibited concentrations of TPHC in excess of 10,000 mg/kg. Please refer to Table 5E and Figures 13 and 14 for a summary of TPHC and O/G/results; TPHC results are summarized on Figure 13 and O/G results are summarized on Figure 14.

6.3.8 pH

The pH values recorded for soil samples collected from Site 2A/2B ranged from 3.8 to 12 with a little less then the majority, approximately 46%, of the values falling between 7.0 and 8.5. It should be noted that all of the samples exhibiting pH concentrations at or above 10 were collected from soil borings installed into the by-product fill material present at the Site A. Three samples exhibiting low pH, below 5, were also were noted at Site A. The three samples were all obtained from locations near Building 20 and were noted in samples from surface to approximately 4 feet bgs. Please refer to Table 5E and Figure 15 for a summary of the pH results.

 \mathcal{K}

 \times



and the second second



of				
AM MA	NAGEF	$\overline{\mathbf{x}}$		
)ri			Appro	ved.
)EPA	RTM	ENT		
133 133	LO 8, L		AN 1)	D
E 2/	× 1 4/2E	гА 3	UIL	
ON SOIL	OF _,	TP	HC	IN
ESI	IG	ATI	ON	
to co s, des d to vithout	onditic signs Port its v	ns in and Autho vritten	contr meth ority cons	act. ods and ent.
awn I	ру	Che	cked	by
GUF	RE	13		Ministrator
Martin Martin				



		GEND	
			······································
	UTILITY EASEMENT	PG-RS-1	PRE-EXISTING P&G
	RAILROAD TRACKS	t ₽	MONITORING WELL LOCATION
	SITE BOUNDARY	PG-SB-2	PRE-EXISTING P&G
			SOIL BORING LOCATION
i	YEAR 2000 SITE INVESTIGATION	ANALYTE	INDICATES COMPOUND FOR WHICH
	MONITORING WELL LOCATION		LABORATORY ANALYSIS WAS CONDUCTED
3	YEAR 2000 SITE INVESTIGATION	RES	CONTAMINANT CONCENTRATION IN MG/KG
	SOIL BORING LOCATION		
		DEPTH	DEPTH IN FEET BELOW GROUND SURFACE

Sheet	of	
		BRETY
OFNY	S NJ	
ENGINEERING	PROGRAM MANAGE	R
No. Data	Davisias	
ENGINEER	RING DEPARTM	ENT
(BLUC	CK 1400, LC	OT 1 AND
ННМТ-Р	ORT IVORY	FACILITY
	SITE 2A/2	В

DISTRIBUTION OF OIL AND **GREASE IN SOIL,** SITE INVESTIGATION

This drawing subject to conditions in contract. All inventions, ideas, designs and methods herein are reserved to Port Authority and may not be used without its written consent.

Designed by Drawn by Checked by Date

Contract Number

Drawing Number

FIGURE 14





The pH values recorded for soil samples collected from Site 2B ranged from 5.2 to 8.9 with 50%, of the values falling between 7.0 and 8.5. Please refer to Table 5E for a summary of the pH results.

6.4 Groundwater Analytical Data

Groundwater samples were collected from all five of the newly installed groundwater monitoring wells, one temporary well and four of the existing monitoring wells in Site 2A/2B. Table 4 presents the specific analyses for groundwater samples. In addition, field pH was recorded for all groundwater samples. The analytical results for HMM's sampling efforts are presented in Tables 6A-6E, included in the following pages. Figure 7 presents monitoring well locations and Figure 16 presents pertinent groundwater analytical data for Site 2A/2B.

For discussion purposes, the results have been compared, as appropriate, to current NYSDEC Ambient Water Quality Standards and Guidance Values (SGVs). The NYSDEC classifies all groundwater in New York State as GA, potential drinking water source, unless otherwise reclassified. At this time, this represent the only guidance available for ambient groundwater. Please note, the reference of these standards in this report does not represent any agreement or concurrence that same are appropriate for usage at this site. A discussion of the analytical results from the groundwater component of the investigation is provided below.

6.4.1 Volatile Organic Compounds

What about VC mw-

No VOCs were detected above NYSDEC groundwater SVGs in groundwater samples from Site 2A/2B. Please refer to Table 6A and Figure 16, for VOC results.

6.4.2 Semi-Volatile Organic Compounds

One SVOC was detected at concentrations in excess of its corresponding NYSDEC groundwater SVG. Specifically, bis(2-ethylhexyl)phthalate was detected above the SVG of 5 ug/l in two samples. This compound was detected at 8.1 ppb in the sample from PG-PZ-1 and 9.2 ug/l in the sample from PG-TMW-01. Bis(2-ethylhexyl)phthalate is frequently identified as a laboratory contaminant, and in fact, this compound was identified as being a laboratory contaminant in other groundwater samples collected with regard to this project. Please refer to Table 6B and Figure 16, for a summary of SVOC results.



_ _ _ _ _ _ _ _ _ _ _



Shee	***			of				
		7 (3)						
ENGI	NEERI	NG P	ROGRA	M MA	NAGER	2		
No.	Dat	e f	Revisic NG L	n DEP4	RTM	EN7	Appro	ved
Titie (E	B B MT	DCł LO(-PC	< 14 CK DRT	133 IVC	LO 8, L)RY	0T 1 .01 7 F/	AN 1) ACIL	D
DI	STF VC GR	RIB DCs ROU IN	SIT UTI 5, A JND VES	E 2 ON ND WA	A/2 OF SV TE SAT	B Me DC: R, S ION	ETAI s IN SITE N	LS,
This All ii here may	draw nvent in ai not	ing s tions, re re be u	ubject , idea serve sed w	to c s, de d to ithou	onditic signs Port t its v	ons i and Auth writte	n cont meth nority n cons	ract. ods ond sent.
Desi Date Cont	gned tract	by	Dr	awn	by	Ch	ecked	by
Num	ber							



Table 6A Groundwater Analytical Results Volatile Organic Compounds Site 2A/2B HHMT Port Ivory Facility

Location	Recommended	Recommended	PG-GW-10	PG-GW-7	PG-MW-3	PG-PA-MW-7	PG-PA-MW-7D	PG-PA-MW-10D
Sample Date	Groundwater	Groundwater	11/24/2000	11/24/2000	11/28/2000	11/27/2000	11/30/2000	11/30/2000
Concentration	Cleanup	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard	Criteria						
	UG/L	UG/L						
1,1,1-TRICHLOROETHANE	5	NG	0.44 U	0.44 U	0.44 U	0.44 U	0.44 U	0.44 U
1,1,2,2-TETRACHLOROETHANE	5	NG	0.42 U	0.42 U	0.42 U	0.42 U	0.42 U	0.42 U
1,1,2-TRICHLOROETHANE	1	NG	0.50 U	0.50 U	0.50 U	0.50 U	0.50 U	0.50 U
1,1-DICHLOROETHANE	5	NG	0.35 U	0.35 U	0.35 U	0.35 U	0.35 U	0.35 U
1,1-DICHLOROETHYLENE	5	NG	0.41 U	0.41 U	0.41 U	0.41 U	0.4 <u>1</u> U	0.41 U
1,2-DICHLOROETHANE	0.6	NG	0.44 U	0.44 U	0.44 U	0.44 U	0.44 U	0.44 U
1,2-DICHLORORPROPANE	1	NG	0.44 U	0.44 U	0.44 U	0.44 U	0.44 U	0.44 U
2-CHLOROETHYL VINYL ETHER	NS	NG	1. <u>1 U</u>	1.1 U	1.1 U	1.1 U	1.1 U	1.1 U
ACROLEIN	5	NG	3.0 U	3.0 U	3.0 U	3.0 U	3.0 U	3.0 U
ACRYLONITRILE	5	NG	6.6 U	6.6 U	6.6 U	6.6 U	6.6 U	6.6 U
BENZENE	1	NG	0. <u>32</u> U	0.32 U	0.32 U	0.32 U	0.32 U	0.32 U
BROMODICHLOROMETHANE	NS	50	0.30 U	0.30 U	0.30 U	0.30 U	0.30 U	0.30 U
BROMOFORM	NS	50	0.32 U	0.32 U	0.32 U	0.32 U	0.32 U	0.32 U
BROMOMETHANE	5	NG	0.55 U	0.55 U	0.55 U	0.55 U	0.55 U	0.55 U
CARBON TETRACHLORIDE	5	NG	0.23 U	0.23 U	0.23 U	0.23 U	0.23 U	0.23 U
CHLOROBENZENE	5	NG	0.25 U	0.25 U	0.25 U	0.25 U	0.25 U	0.25 U
CHLOROETHANE	5	NG	0.52 U	0.52 U	0.52 U	0.52 U	0.52 U	0.52 U
CHLOROFORM	7	NG	0.45 U	0.45 U	0.45 U	0.45 U	0.45 U	0.45 U
CHLOROMETHANE	. 5	NG	0.32 U	0.32 U	0.32 U	0.32 U	0.32 U	0.32 U
CIS-1,3-DICHLOROPROPENE	5	NG	0.35 U	0.35 U	0.35 U	0.35 U	0.35 U	0.35 U
DIBROMOCHLOROMETHANE	NS	50	0.41 U	0.41 U	0.41 U	0.41 U	0.41 U	0.41 U
DICHLOROMETHANE	5	NG	0.85 U	0.85 U	0.85 U	0.85 U	0.85 U	0.85 U
ETHYLBENZENE	5	NG	0.15 U	0.15 U	0.15 U	0.15 U	0.15 U	0.15 U
M&P-XYLENES	5&5	NG	0.81 U	0.81 U	0.81 U	0.81 U	0.81 U	0.81 U
METHYLBENZENE	5	NG	3.2	0.24 U	0.24 U	0.24 U	3.3	0.24 U
O-XYLENE	5	NG	0.36 U	0.36 U	0.36 U	0.36 U	0.36 U	0.36 U
TETRACHLOROETHYLENE	5	NG	0.34 U	0.34 U	0.34 U	0.34 U	0.34 U	0.34 U
TRAMS-1,2-DICHLOROETHYLENE	5	NG	0.46 U	0.46 U	0.46 U	0.46 U	0.46 U	0.46 U
TRANS-1,3-DICHLOROPROPENE	NS	NG	0.24 U	0.24 U	0.24 U	0.24 U	0.24 U	0.24 U
TRICHLOROETHYLENE	5	NG	0.37 U	0.37 U	0.37 U	0.37 U	0.37 U	0.37 U
VINYL CHLORIDE	2	NG	0.67 U	0.67 U	0.67 U	2.0	0.67 U	0.67 U

U Undetectable Levels

NS No Standard

NG No Guidance

Shows Figure

174

Table 6A Groundwater Analytical Results Volatile Organic Compounds Site 2A/2B HHMT Port Ivory Facility

Location	Recommended	PG-PA-MW-15	PG-PA-MW-15D	PG-PA-MW-15D	PG-PZ-1	PG-TMW-01
Sample Date	Groundwater	11/20/2000	11/20/2000	11/30/2000	11/24/2000	11/29/2000
Concentration	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard					
	UĠ/L					
1,1,1-TRICHLOROETHANE	5	0.22 U	0.22 U	0.44 U	0.44 U	0.44 U
1,1,2,2-TETRACHLOROETHANE	- 5	0.42 U	0.42 U	0.42 U	0.42 U	0.42 U
1,1,2-TRICHLOROETHANE	1	0.51 U	0.51 U	0.50 U	0.50 U	0.50 U
1,1-DICHLOROETHANE	5	0.54 U	0.54 U	0.35 U	0.35 U	0.35 U
1,1-DICHLOROETHYLENE	5	0.99 U	0.99 U	0.41 U	0.41 U	0.41 U
1,2-DICHLOROETHANE	0.6	0.22 U	0.22 U	0.44 U	0.44 U	0.44 U
1,2-DICHLORORPROPANE	1	0.73 U	0.73 U	0.44 U	0.44 U	0.44 U
2-CHLOROETHYL VINYL ETHER	NS	0.77 U	0.77 U	1.1 U	1.1 U	1.1 U
ACROLEIN	5	5.1 U	5.1 U	3.0 U	3.0 U	3.0 U
ACRYLONITRILE	. 5	2.0 U	2.0 U	6.6 U	6.6 U	6.6 U
BENZENE	1	0.15 U	0.15 U	0.32 U	0.32 U	0.32 U
BROMODICHLOROMETHANE	NS	0.27 U	0.27 U	0.30 U	0.30 U	0.30 U
BROMOFORM	NS	0.34 U	0.34 U	0.32 U	0.32 U	0.32 U
BROMOMETHANE	5	0.72 U	0.72 U	0.55 U	0.55 U	0.55 U
CARBON TETRACHLORIDE	5	0.36 U	0.36 U	0.23 U	0.23 U	0.23 U
CHLOROBENZENE	5	0.34 U	0.34 U	0.25 U	0.25 U	0.25 U
CHLOROETHANE	5	0.88 U	0.88 U	0.52 U	0.52 U	0.52 U
CHLOROFORM	7	0.27 U	0.27 U	0.45 U	0.45 U	0.45 U
CHLOROMETHANE	5	0.66 U	0.66 U	0.32 U	0.32 U	0.32 U
CIS-1,3-DICHLOROPROPENE	5	0.43 U	0.43 U	0.35 U	0.35 U	0.35 U
DIBROMOCHLOROMETHANE	NS	0.42 U	0.42 U	0.41 U	0.41 U	0.41 U
DICHLOROMETHANE	5	1.0 U	1.0 U	0.85 U	0.85 U	0.85 U
ETHYLBENZENE	5	0.57 U	0.57 U	0.15 U	0.15 U	0.15 U
M&P-XYLENES	5&5	0.90 U	0.90 U	0.81 U	0.81 U	0.81 U
METHYLBENZENE	5	1.5	1.6	0.24 U	0.24 U	0.24 U
O-XYLENE	5	0.55 U	0.55 U	0.36 U	0.36 U	0.36 U
TETRACHLOROETHYLENE	5	0.59 U	0.59 U	0.34 U	0. 34 U	0.34 U
TRAMS-1,2-DICHLOROETHYLENE	5	0.94 U	0.94 U	0.46 U	0.46 U	0.46 U
TRANS-1,3-DICHLOROPROPENE	NS	0.21 U	0.21 U	0.24 U	0.24 U	0.24 U
TRICHLOROETHYLENE	5	0.64 U	0.64 U	0.37 U	0.37 U	0.37 U
VINYL CHLORIDE	2	0.69 U	0.69 U	0.67 U	0.67 U	0.67 U

U Undetectable Levels

NS No Standard

NG No Guidance

•

Table 6B Groundwater Analytical Results Semi-Volatile Organic Compounds Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	PG-GW-10	PG-GW-7	PG-MW-3	PG-PA-MW-7	PG-PA-MW-7D	PG-PA-MW-10D
Sample Date	Groundwater	Groundwater	11/24/2000	11/24/2000	11/28/2000	11/27/2000	11/30/2000	11/30/2000
Concentration	Cleanup	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard	Criteria	l I		ļ			
	UG/L	UG/L				\$	1	
1,2,4-TRICHLOROBENZENE	5	NG	0.27 U	0.27 U				
1,2-BENZPHENANTHRACENE	NS	0.002	0.30 U	0.30 U				
1,2-DICHLOROBENZENE	3	NG	0.26 U	0.26 U				
1,2-DIPHENYLHYDRAZINE	NS	NG	0.24 U	0.24 U				
1,4-DICHLOROBENZENE	3	NG	0.20 U	0.20 U				
2,4,6-TRICHLOROPHENOL	NS	NG	2.1 U	2.1 U				
2,4-DICHLOROPHENOL	5	NG	2.0 U	2.0 U				
2,4-DIMETHYLPHENOL	NS	50	1.4 U	1.4 U				
2,4-DINITRPHENOL	NS	10	0.47 U	0.47 U				
2,4-DINITROTOLUENE	5	NG	0.16 U	0.16 U				
2,6-DINITROTOLUENE	5	NG	0.27 U	0.27 U				
2-CHLORONAPHTHALENE	NS	10	0.22 U	0.22 U				
2-CHLOROPHENOL	NS	NG	1.4 U	1.4 U				
2-NITROPHENOL	NS	NG	2.1 U	2.1 U				
3,3'-DICHLOROBENZIDINE	5	NG	2.7 U	2.7 U				
4,6-DINITRO-O-CRESOL	NS	NG	1.2 U	1.2 U				
4-BROMOPHENYLPHENYL ETHER	NS	NG	0.23 U	0.23 U				
4-CHLORO-3-METHYLPHENOL	NS	NG	1.9 U	1.9 U				
4-CHLORORPHENLYPHENYL ETHER	NS	NG	0.32 U	0.32 U				
4-NITROPHENOL	NS	NG	1.6 U	1.6 U				
ACENAPHTHENE	NS	20	0.31 U	0.31 U				
ACENAPHTHYLENE	NS	NG	0.26 U	0.26 U				
ANTHRACENE	NS	50	0.25 U	0.25 U				
BENZIDINE	5	NG	3.4 U	3.4 U				
BENZO{A}ANTHRACENE	NS	0.002	0.20 U	0.20 U				
BENZO{A}PYRENE	ND	NG	0.24 U	0.24 U				
BENZO { B } FLOURANTHENE	NS	0.002	0.49 U	0.49 U				
BENZO {G,H,I}PERYLENE	NS	NG	0.36 U	0.36 U				
BENZO {K}FLOURANTHENE	NS	0.002	0.50 U	0.50 U				
BENZYL BUTYL PHTHALATE	NS	50	0.29 U	0.29 U				
BIS(2-CHLOROETHOXY)METHANE		NG	0.21 U	0.21 U	0.21 U	0.21 U	0.21 0	0.21 U
BIS(2-CHLOROETHYL)ETHER		NG	0.15 U	0.15 U	0.15 U	0.15 U	0.15 0	0.15 U
BIS(2-CHLOROISOPROPYL)ETHER		NG	1.5	0.14 0	0.14 U	0.14 U	0.14 0	0.14 0
BIS(2-ETHYLREATLICALATE		NG	0.26 11	0.26 11	4.2 B	2.1 B	2.5 B	3.0 B
DI-N-BUTTL PHTHALATE	NS	50	0.200	0.200	0.20 U	0.20 0	1.2	0.26 0
DIPENZIA LIANTHRACENE	NS	NG	0.80 0	0.34 U	0.34 11	0.80 0	0.2411	0.24 11
DIBENZIA, IJAN I HRACENE	NS	50	0.34 U	0.3111	0.34 0	0.34 0	0.34 U	0.34 U
	NS	50	0.2411	0.24 U	0.2411	0.24 U	0.2411	0.310
FLUORANTHENE	NS	50	0.29 11	0.2911	0.2911	0.2911	0.24 0	0.29 []
FLUORENE	NS	50	0.28 U	0.28 U	0.28 []	0.28 U	0.2811	0.28 11
HEXACHLORO-1.3-BUTADIENE	0.5	NG	0.25 U	0.25 U				
HEXACHLOROBENZENE	0.04	NG	0.28 U	0.28 U				
HEXACHLOROCYCLOPENTADIENE	5	NG	2.5 U	2.5 U				
HEXACHLOROETHANE	5	NG	0.26 U	0.26 U				
INDENO[1,2,3-CD]PYRENE	NS	0.002	0.34 U	0.34 U				
ISOPHORONE	NS	50	0.21 U	0.21 U				
M-DICHLOROBENZENE	3	NG	0.27 U	0.27 U				
N-NITROSO-DI-N-PROPYLAMINE	NS	NG	0.22 U	0.22 U				
N-NITROSODIMETHYLAMINE	NS	NG	0.28 U	0.28 U				
N-NITROSODIPHENYLAMINE	NS	50	0.32 U	0.32 U	0.32 U	0.32 U	0.32 Ū	0.32 U
NAPHTHALENE	NS	10	0.36 U	0.36 U				
NITROBENZENE	0.4	NG	0.23 U	0.23 U				
PENTACHLOROPHENOL	1(Total Phenols)	NG	2.0 U	2.0 U				
PHENANTHRENE	NS	50	0.27 U	0.27 U				
PHENOL	1(Total Phenois)	NG	1.2 U	1.2 U				
PYRENE	NS	50	0.27 U	0.27 U				

U Undetectable Levels NS No Standard

NG No Guidance * Principal organic contaminant standard for groundwater of 5 ug/L applies

Table 6B Groundwater Analytical Results Semi-Volatile Organic Compounds Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	PG-PA-MW-15	PG-PA-MW-15D	PC-PA-MW-15D	PC-P7-1	PC.TMW-01
Sample Date	Groundwater	Groundwater	11/20/2000	11/20/2000	11/30/2000	11/24/2000	11/20/2000
Concentration	Cleanun	Cleanup	ПСЛ	10/2000		11/24/2000	11/29/2000
Concentration	Standard	Creatup	100/12	00/L	100/L	UG/L	JUGIL
	UG/L	UG/L					
1.2.4-TRICHLOROBENZENE	5	NG NG	0.27 U	0.23 U	0.27 U	0.27 11	0.27 11
1 2-BENZPHENANTHRACENE	NS	0.002	0.30 U	0.37 11	0 30 11	0.30 11	0.30 11
1 2-DICHLOROBENZENE	3	NG	0.261	0361	0.261	0.26 U	0.36 U
1 2-DIPHENVLHYDRAZINE	NS	NG	0 24 11	0.2611	0.24 U	0.24 U	0.200
1 4-DICHLOROBENZENE	3	NG	0.20 U	0 38 11	0.20 U	0.24 0	0.24 0
2.4.6-TRICHLOROPHENOL	NS	NG	2111	161	2111	2111	2111
	5	NG	2011	261	2011	2011	2011
2.4-DIMETHVI PHENOI	NS	50	141	151	141	141	1411
2 4-DINITEPHENOI	NS	10	0.4711	2711	0 47 11	0.47.11	04711
2.4-DINITROTOLUENE	5	NG	0.161	0.84 11	0.16 U	0.1611	0.470
2.6 DINITROTOLUENE	5	NG	0.2711	0.44 U	0.100	0.10 0	0.100
2-CHI ORONAPHTHAL FNE	NS	10	0.22 11	0.2911	0.22 11	0.27 U	0.27 U
2-CHLOROPHENOI	NS	NG	141	241	141	1411	141
2-NITROPHENOL	NS	NG	2.1 U	1.91	2111	2111	2111
3 3'-DICHLOROBENZIDINE	5	NG	2.7 U	341	2711	2711	2711
4 6-DINITRO-O-CRESOL	NS	NG	1.2 []	1811	1211	1211	1211
4-BROMOPHENYI PHENYI ETHER	NS	NG	0.23 U	0.75 11	0.23.11	0.2311	0.23 11
4-CHI ORO-3-METHYI PHENOL	NS	NG	1911	2511	1911	191	1911
4-CHLOROPHENI VPHENYI ETHER	NS	NG	0.32 []	0.5511	03211	0.32 11	0.32 11
4 NITROPHENOI	NS	NG	161	2211	1611	161	161
	NS	20	03111	0.40 U	0.31.11	0.21 11	1.0 0
ACENALITIENE	NS	NG	0.310	0.40 0	0.310	0.310	0.26 1
ANTHRACENE	NS	50	0.25 U	0.34 0	0.20 0	0.26 U	0.26 U
RENZIONE	- 115	NG	241	2011	2411	2 4 11	2 4 11
DENZIDINE DENZO (A) ANTURACENE	NIS	0.002	0.20 11	0.41 U	0.2011	0.20.11	3.4 U
DENZO (A) PVPENE	ND	0.002 NG	0.20 0	0.47 U	0.20 0	0.20 0	0.20 U
BENZO (B) ELOURANTHENE	NS	0.002	0.24 0	0.47 U	0.24 0	0.24 0	0.24 0
BENZO(GHUPERVIENE	NS	0.002 NG	0.45 U	0.45 U	0.49.0	0.490	0.49 0
BENZO(K)FLOURANTHENE	NS	0.002	0.50 U	0.67 U	0.50 U	0.50 U	0.50 U
BENZVI BUTVI PHTHALATE	NS	50	0.29 11	0.07.0	0.30 0	0.2011	0.30 U
BIS(2-CHLOROFTHOXY)METHANE	5	NG	0.21 U	0.31 U	0.27 U	0.290	0.23 0
BIS(2-CHLOROFTHYL)FTHER	1	NG	0.15 U	0.58 U	0.1511	0.15 U	0.15 U
BIS(2-CHLOROLSOPROPYL)FTHER	5	NG	0.14 U	0.31 U	0.15 0	0.14 U	0.13 U
BIS(2-FTHVI HFXVI)PHTHAI ATF	5	NG	38B	20 B 30 51 - 128 44 51 53	23B	81	0.14 0
DLN_BUTYI PHTHAIATE	50	NG	0.2611	03011	0.2611	0.2611	0.26.11
DINOCTVI PHTHALATE	NS	50	0.20 U	0.50 0	12	0.20 U	0.20 0
DIBENZIA HIANTHRACENE	NS	NG	0.34 U	0.46 U	03411	0.30 U	0.34 U
DIETHYL PHTHALATE	NS	- 50	031 U	0.32 11	0.31 U	03111	0.34 0
DIMETHYL PHTHALATE	NS	50	0.24 []	0.52 C	0.24 U	0.24 11	0.24 11
FLUORANTHENE	NS	50	0.29 U	0.42 U	0.2911	0.29 []	0.24 0
FLUORFNE	NS	- 50	0.28 U	033 U	0.2811	0.29 U	0.29 U
HEXACHLORO-1.3-BUTADIENE	0.5	NG	0.25 U	0.61 U	0.25 U	0.25 U	0.25 U
HEXACHLOROBENZENE	0.04	NG	0.28 U	0.64 U	0.28 U	0.28 U	0.28 11
HEXACHLOROCYCLOPENTADIENE	5	NG	2511	3.011	2511	2511	2511
HEXACHLOROETHANE	5	NG	0.26 U	0.74 U	0.26 U	0.2611	0.2611
INDENO[1 2.3-CD]PYRENE	NS	0.002	0.34 U	0.38 U	0.34 U	0 34 11	03411
ISOPHORONE	NS	50	0.21 U	0.41 U	0.21 U	0.21 U	0.21 U
M-DICHLOBOBENZENE	3	NG	0.27 U	0.48 U	0.27 U	0 27 11	0 27 U
N-NITROSO-DI-N-PROPYLAMINE	NS	NG	0.22 U	0.47 U	0.22 U	0.22 U	0.22 U
N-NITROSODIMETHYLAMINE	NS	NG	0.28 U	0.89 U	0.28 U	0.28 U	0.28 U
N-NITROSODIPHENYI AMINE	NS	50	0.32 U	0.44 U	0.32 U	0.32 11	0 32 U
NAPHTHAI ENE	NS	10	0.36 U	0.26 U	0.36 U	0.3611	0.36 U
NITROBENZENE	0.4	NG	0.23 U	0.63 U	0.23 11	0.23 U	0.23 []
PENTACHI OROPHENOI	1(Total Phenole)		2011	1.4 U	2.011	2011	2011
PHENANTHRENE	NS	50	0.27 U	0.32 U	0.27 []	0.27 11	0.27 11
PHENOL	1(Total Phenole)	NG	1.2 U	1.4 U	1.2 U	1.2 U	1211
PYRENE	NS	50	0.27 U	0.17 U	0.27 U	0.27 U	0.27 U

U Undetectable Levels NS No Standard NG No Guidance

* Principal organic contaminant standard for groundwater of 5 ug/L applies



Table 6C Groundwater Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	PG-GW-10	PG-GW-7	PG-PA-MW-7	PG-MW-3	PG-PA-MW-7D	PG-PA-MW-10D
Sample Date	Groundwater	Groundwater	11/24/2000	11/24/2000	11/27/2000	11/28/2000	11/30/2000	11/30/2000
Concentration	Cleanup	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard	Criteria						
	UG/L	UGAL	1					
4,4'-DDD	0.3	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
4,4'-DDE	0.2	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
4,4'-DDT	0.2	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ALDRIN	NS	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ALPHA-BHC	NS	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
AROCLOR 1016	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1221	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1232	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1242	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1248	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1254	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1260	0.09**	NG	0.5 U	0.5 U				
BETA-BHC	0.04	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
CHLORDANE	0.05	NG	0.2 U	0.2 U				
DELTA-BHC	0.04	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
DIELDRIN	0.004	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ENDOSULFAN I	NS	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ENDOSULFAN II	NS	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ENDOSULFAN SULFATE	NS	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ENDRIN	NS	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ENDRIN ALDEHYDE	5	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
ENDRIN KETONE	5	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
GAMMA-BHC (LINDANE)	0.05	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
HEPTACHLOR	0.04	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
HEPTACHLOR EPOXIDE	0.03	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
METHOXYCHLOR	35	NG	0.02 U	0.02 U	0.02 U	0.1 U	0.02 U	0.02 U
TOXAPHENE	0.06	NG	1 U	1 U	10	10	10	1 U
AROCLOR 1016	0.09**	NG	0.5 U	0.5 U	0.5 U .	0.5 U	0.5 U	0.5 U
AROCLOR 1221	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1232	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1242	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1248	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1254	0.09**	NG	0.5 U	0.5 U				
AROCLOR 1260	0.09**	NG	0.5 U	0.5 U				
Total PCBs	0.09	NG	ND	ND	ND	ND		

U Undetectable Levels

NS No Standard

NG No Guidance

ND Not Detected

** Total PCBs

١

Table 6C Groundwater Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	PG-PA-MW-15	PG-PA-MW-15D	PG-PA-MW-15D	PG-PZ-1	PG-TMW-01
Sample Date	Groundwater	Groundwater	11/20/2000	11/20/2000	11/30/2000	11/24/2000	11/29/2000
Concentration	Cleanup	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard	Criteria				1	
	UG/L	UG/L					
4,4'-DDD	0.3	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
4,4'-DDE	0.2	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
4,4'-DDT	0.2	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ALDRIN	NS	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ALPHA-BHC	NS	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
AROCLOR 1016	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1221	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1232	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1242	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1248	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1254	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1260	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
BETA-BHC	0.04	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
CHLORDANE	0.05	NG	0.2 U	0.2 U	0.2 U	0.2 U	0.2 U
DELTA-BHC	0.04	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
DIELDRIN	0.004	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ENDOSULFAN I	NS	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ENDOSULFAN II	NS	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ENDOSULFAN SULFATE	NS	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ENDRIN	NS	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ENDRIN ALDEHYDE	5	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
ENDRIN KETONE	5	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
GAMMA-BHC (LINDANE)	0.05	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
HEPTACHLOR	0.04	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
HEPTACHLOR EPOXIDE	0.03	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
METHOXYCHLOR	35	NG	0.02 U	0.02 U	0.02 U	0.02 U	0.02 U
TOXAPHENE	0.06	NG	1 U	1 U	1 U	10	10
AROCLOR 1016	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1221	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1232	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1242	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1248	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1254	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
AROCLOR 1260	0.09**	NG	0.5 U	0.5 U	0.5 U	0.5 U	0.5 U
Total PCBs	0.09	NG ·	ND	ND	ND	ND	ND

U Undetectable Levels

NS No Standard

NG No Guidance

ND Not Detected

** Total PCBs

Table 6DGroundwater Analytical ResultsMetalsSite 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	PG-GW-10	PG-GW-7	PG-MW-3	PG-PA-MW-7	PG-PA-MW-7D	PG-PA-MW-10D
Sample Date	Groundwater	Groundwater	11/24/2000	11/24/2000	11/28/2000	11/27/2000	11/30/2000	11/30/2000
Concentration	Cleanup	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard	Criteria						
	UG/L	UG/L					· ·	
ALUMINUM								
(FUME OR DUST)	NS	NG	510	140	120	3100	58 U	440
ANTIMONY	3	NG	3.3 U	3.3 U				
ARSENIC	25	NG	3.6 U	110	6.7	13	5.7	3.6 U
BARIUM	1000	NG	37	110	280	39	1800	120
BERYLLIUM	NS	3	2.5 U	2.5 U				
CADMIUM	5	NG	1.4 U	1.4 U				
CALCIUM METAL	NS	NG	48000	76000	130000	5700	230000	69000
CHROMIUM	50	NG	16 U	16 U				
COBALT	NS	NG	4.6 U	4.6 U				
COPPER	200	NG	20 U	20 U				
IRON	300***	NG	610***	14000***	3700***	5300***	3500***	12000***
LEAD	- 25	NG	3.4 U	3.4 U				
MAGNESIUM	NS	35000	8400	7600	180000	1900	36000	21000
MANGANESE	300***	NG	16***	420***	170***	220***	1300***	570***
NICKEL	100	NG	15 U	15 U	57	15 U	15 U	15 U
POTASSIUM	NS	NG	2600	10000	7500	4100	2900	4600
SELENIUM	10	NG	20 U	20 U				
SILVER	50	NG	5.2 U	5.2 U				
SODIUM	20000	NG	440000	38000	1700000	220000	240000	450000
THALLIUM	NS	0.5	3.1 U	3.1 U				
VANADIUM	NS	NG	4.5	4.3 U	8.8	9.2	7.4	8.9
ZINC	NS	2000	20 U	20 U				
MERCURY	0.7	NG	0.21 U	0.21 U				

U Undetectable Levels

NS No Standard

NG No Guidance

*** Total for Iron and Maganese is > 500

x.

-180

Table 6D Groundwater Analytical Results Metals Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	PG-PA-MW-15	PG-PA-MW-15D	PG-PA-MW-15D	PG-PŻ-1	PG-TMW-01
Sample Date	Groundwater	Groundwater	11/20/2000	11/20/2000	11/30/2000	11/24/2000	11/29/2000
Concentration	Cleanup	Cleanup	UG/L	UG/L	UG/L	UG/L	UG/L
	Standard	Criteria					
	UG/L	_UG/L					
ALUMINUM							
(FUME OR DUST)	NS	NG	NA	58 U	58 U	1300	1600
ANTIMONY	3	NG	NA	3.3 U	3.3 U	3.3 U	3.3 U
ARSENIC	25	NG	NA	3.6 U	3.6 U	100	3.6 U
BARIUM	1000	NG	NA	63	240	30	39
BERYLLIUM	NS	3	NA	2.5 U	2.5 U	2.5 U	2.5 U
CADMIUM	5	NG	NA	1.4 U	1.4 U	1.4 U	1.4 U
CALCIUM METAL	NS	NG	NA	360000	330000	6200	36000
CHROMIUM	50	NG	NA	16 U	16 U	16 U	22
COBALT	NS	NG	NA	4.6 U	4.6 U	4.6 U	6.3
COPPER	200	NG	NA	20 U	20 U	20 U	20 U
IRON	300***	NG	NA	2200***	330***	21000***	3100***
LEAD	25	NG	NA	7.3	3.4 U	11	3.4 U
MAGNESIUM	NS	35000	NA	56000	49000	1800	8600
MANGANESE	300***	NG	NA	1500****	1200***	420***	2400***
NICKEL	100	NG	NA	15 U	25	15 U	28
POTASSIUM	NS	NG	NA	2600	2700	2400	6300
SELENIUM	10	NG	NA	20 U	20 U	20 U	20 U
SILVER	50	NG	NA	5.2 U	5.2 U	5.2 U	5.2 U
SODIUM	20000	NG	24000	78000	65000	240000	22000
THALLIUM	NS	0.5	NA	3.1 U	3.1 U	3.1 U	3.1 U
VANADIUM	NS	NG	NA	7	7.2	37	47
ZINC	NS	2000	NA	20 U	20 U	33	40
MERCURY	0.7	NG	0.21 U	0.21 U	0.21 U	0.21 U	0.21 U

U Undetectable Levels

NS No Standard

~

NG No Guidance

*** Total for Iron and Maganese is > 500

•,



Groundwater Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location		Recommended	Recommended	PG-GW-10	PG-GW-7	PG-MW-3	PG-PA-MW-7	PG-PA-MW-7D	PG-PA-MW-10D
Sample Date	}	Groundwater	Groundwater	11/24/2000	11/24/2000	11/28/2000	11/27/2000	11/30/2000	11/30/2000
Concentration		Cleanup	Cleanup						
		Standard	Criteria						•
PETROLEUM HYDROCARBONS	MG/L	NS	NG	1.0 U	1.1 U	1.0 U	1.1 U	1.0 U	1.0 U
OIL & GREASE	MG/L	NS	NG	7.9	12	25	9.8	18	22
CYANIDE	MG/L	0.2	NG	0.01 U	0.01 U	0.027	0.033	0.01 U	0.074
*pH	pH units	s NS	NG	7.72	6.6	6.91	7.42	7.31	6.76
TOTAL PHENOLICS	MG/L	0.001	NG	0.05 U	0.05 U				

182

U Undetectable Levels

NS No Standard

NG No Guidance

Note: pH listed is the pH recorded in the field



Groundwater Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location		Recommended	Recommended	PG-PA-MW-15	PG-PA-MW-15D	PG-PA-MW-15D	PG-PZ-1	PG-TMW-01
Sample Date		Groundwater	Groundwater	11/20/2000	11/20/2000	11/30/2000	11/24/2000	11/29/2000
Concentration		Cleanup	Cleanup					
·		Standard	Criteria					
PETROLEUM HYDROCARBONS	MG/L	NS	NG	1.0 U	1.1 U	1.1 U	1.1 U	1.0 U
OIL & GREASE	MG/L	NS	NG	1.1 U	1.1 U	25	19	13
CYANIDE	MG/L	0.2	NG	0.01 U	0.01 U	0.01 U	0.01 U	0.01
*pH	pH units	s NS	NG	6.59	7	7.4	7.02	7.12
TOTAL PHENOLICS	MG/L	0.001	NG	0.05 U	0.05 U	0.05 U	0.05 U	0.05 U

183

U Undetectable Levels

NS No Standard

NG No Guidance

Note: pH listed is the pH recorded in the field



6.4.3 Polychlorinated Biphenyls

PCBs were not detected in the groundwater samples from Site 2A/2B. It should be noted that all of the method detection limits for PCBs were above the NYSDEC groundwater SVGs. Please refer to Table 6C $^{\circ}$ for a summary of PCB results.

6.4.4 Pesticides

No pesticides were detected in groundwater samples collected from Site 2A and 2B. Please refer to Table 6C for a summary of pesticides results.

6.4.5 Metals

A number of TAL metals were detected in one or more groundwater samples collected as part of the groundwater investigation of Site 2A/2B. However, only six TAL metals (arsenic, barium, magnesiun, iron, manganese, and sodium) were detected in one or more groundwater samples at concentrations in excess of corresponding NYSDEC groundwater SVGs. Iron and sodium were detected in excess of NYSDEC groundwater SVGs in several groundwater samples collected from wells located throughout Site 1. Magnesium and manganese and sodium were detected in excess of NYSDEC groundwater SVGs in several groundwater samples collected from wells located throughout Site 1. Magnesium and manganese and sodium were detected in excess of NYSDEC groundwater SVGs in several groundwater samples collected from wells located throughout Site 2A/2B. Arsenic was detected in two samples, PG-PZ-1 and PG-GW-7 at concentrations of 100 ug/l and 110 ug/l respectively; both wells are located on Site 2A. Comparatively, barium was detected in excess of NYSDEC groundwater SVGs in only a single sample (PG-PA-MW-7D) at a concentration of 1,800 ug/L. Please refer to Table 6D and Figure 16, for a summary of metals results.

6.4.6 Cyanide and Total Phenolics

Cyanide was detected in four of the groundwater samples collected from wells at Site 2A/2B. <u>All</u> concentrations were below the NYSDEC SVG for cyanide. Please refer to Table 6E, for a summary of cyanide results.

Total phenolics were not detected in any of the groundwater samples collected from Site 2A/2B. Please refer to Table 6E, for a summary of total phenolic results.

6.4.7 Petroleum Hydrocarbons/Oil and Grease

<u>TPHC</u> was not detected in the groundwater samples from Site 2A/2B. In contrast, O/G was detected in six of eleven groundwater samples. The detectable ranges of O/G ranged from 7.9 mg/L to 25 mg/L in samples PG-GW-10 and PG-MW-3, respectively. Please refer to Table 6E, for a summary of TPHC and O/G results.

Free phase product was observed in two wells, GW-14 and OP-1, from Site 2A. No free phase product was observed in wells situated at Site 2B. Product thickness was noted as a sheen in both wells. Regardless, samples of product were submitted for GC fingerprinting analysis, however, sufficient product could not be collected to perform the required analyses.

6.4.8 pH

The pH was recorded for groundwater at all locations as part of field sampling. The field pH values have been included in Table 6E for reference purposes. Please note, the value included on Table 6E reflects the pH recorded just prior to sampling. The pH values recorded in conjunction with the groundwater sampling from Site 2A/2B ranged from 6.59 to 7.72. The field recorded pH values are included on groundwater sampling logs provided in Appendix C.

7.0 SI – DISCUSSIONS/CONCLUSIONS

The investigative efforts described herein were undertaken to confirm current site conditions as well as to evaluate previously identified AOCs at this site. Overall, the field component of the SI confirmed the presence of a variety of fill materials and identified several potential "oil" impacted areas including potential UST Areas. Analytical data has revealed the presence of contaminants at concentrations in excess of current NYSDEC regulatory guidance criteria in samples from soil and groundwater. However, the data generally indicates that site issues are related to petroleum and non-petroleum oils, pH and to some degree, metals. To a far lesser extent, VOCs and SVOCs were noted to be present at \vee , \vee , \vee is concentrations above NYSDEC guidance criteria in soil and groundwater. Generally analytical results have shown that former site usage did not substantially impact groundwater and that groundwater quality is typical to that of urban areas. It should be noted that the investigation described herein did not include



a geo-technical evaluation. As such, it does not identify or address any issues associated with the physical elements of the fill material including issues associated with future construction activities.

7.1 Soil

Volatile Organic Compounds

Analytical results identified the presence of only three VOCs at concentrations in excess of NYSDEC guidance criteria for soil in the 86 soil samples collected from Site 2A/2B including soil samples collected from the four potential UST areas. Specifically, trans-1,3-dichloropropene was detected at a concentration of 0.43 mg/kg from sample PG-B-4 (5-6 feet); dichloromethane was detected at a concentration of 0.17 mg/kg from sample PG-STAIN-03; and benzene was detected at a concentration of 0.32 mg/kg from sample PG-MW-10D.

Given the low levels or non-detectable levels of VOC compounds in soil and the proposed future site usage, no further action was deemed warranted with regard to this class of contaminants. However, additional actions were proposed to further evaluate the presence of benzene at the PG-MW-10D location at Site 2B. Please refer to Section 8.2 for a discussion of proposed RI actions at the PG-MW-10D location.

Semi-Volatile Organic Compounds

Analytical results indicated the presence of several SVOCs at concentrations in excess of NYSDEC guidance criteria in a number soil samples collected from Site 2A/2B. Generally, these soil results reveal that the SVOCs present in soil at the site consist predominantly of PAH compounds at concentrations only slightly above NYSDEC guidance criteria. In one instance, sample PG-STAIN-03 (1.7-2.5 feet), individual PAH compounds were detected at concentrations above in excess of the NYSDEC guidance of 50 mg/kg for individual SVOCs; the total SVOC concentration for this sample was above 500 mg/kg. However, this sample was collected from a stained area and analytical results from a sample collected at a slightly deeper interval (2.5-3.5 feet) revealed significantly lower concentrations of PAH compounds.

Except as outlined above, the relatively low concentrations of PAH compounds detected in soil samples is not unexpected given that fill material was emplaced at Site 2A/2B in conjunction with site development and that Site 2A/2B, as well as the remainder of the site, has been utilized in an industrial capacity for approximately 100 years. As discussed in Section 6.4.2, no PAH compounds were detected above

✻



NYSDEC guidance criteria in groundwater samples. Given the low levels of PAH compounds in soil, lack of adverse impacts to groundwater and the proposed future site usage, no further action is proposed with regard to this class of contaminants in soil. However, the PG-STAIN-03 location will be addressed in conjunction with demolition of Building 20 and overall site development. No delineation sampling was deemed warranted at the PG-STAIN-03 location based on other analytical data from this area, including deeper samples collected from the PG-STAIN-03 location and samples collected from an adjacent location (PG-STAIN-3B) and visual signs of staining.

Metals

Analytical data revealed the presence of a variety of metals at a wide range of concentrations including exceedances of NYSDEC guidance criteria in soil samples collected from Site 2A/2B. The presence of metals in soil at this site was not unexpected given that indigenous soils may contain concentrations of metals species at levels near or above regulatory criteria. The number and wide range of the concentrations of detected metals similarly was anticipated as a variety of fill materials were placed at Site 2A/2B as well as other areas of the site in conjunction with site development.

A review of the spatial distribution of the analytical results revealed two notable conditions with regard to the metals at Site 2A/2B. First, analytical data from the soil component of this SI has revealed the presence of elevated concentrations of arsenic at locations throughout Site 2A/2B as well as the remainder of the site. However, this contaminant seems to occur at higher than "site average" concentrations in many of the soil samples collected from locations adjacent to current and former railroad tracks. The presence of this contaminant was also noted, at a reduced frequency and at lower concentrations, in samples from locations not proximate to railroad tracks and sidings. Given the large portion of Site 2A/ 2B, which is currently occupied or which was historically occupied by railroad tracks and sidings, it is likely that the presence of arsenic at many locations may be attributable, in part, to railroad fill, bedding materials (cinders and clunkers) and railroad tie chemical preservatives. Therefore, the presence of this metal is considered ubiquitous to Site 2A/2B site based upon the connection of arsenic and railroad materials. Further, the anticipated usage of Site 2A/2B consists of an intermodal facility, which will include a rail system. Although arsenic was detected in Site 2A/2B groundwater, only 2 of 11 samples (from 10 well locations) exhibited a concentration in excess of NYSDEC guidance criteria. The presence of this contaminant in groundwater may reflect the urban nature of the site area rather than an impact from site activities. Accordingly, no further action was proposed with regard to arsenic in soil.

Site 2A/2B Report



Second, fewer metals appear to be present in the by-product (diatomaceous earth) fill material present at the site as compared to other fill/soil. Generally, the by-product fill material includes aluminum, barium, calcium, iron, magnesium and sodium and to a lesser extent, manganese and potassium. This assertion does not appear to be sustained at locations where the by-product fill is intermixed or located in close proximity to soil fill or cinder fill or in samples of the by-product fill collected from the surficial interval. Analytical results revealed concentrations of aluminum, iron, sodium, and manganese above NYSDEC guidance criteria in samples from numerous site wells, including wells located at other areas of the site. Further discussion of fill related issues are provided later in this section. With regard to Site 2A/2B, the by-product fill appears to be limited to certain portions of Site 2A. The soil borings installed as Site 2B did not reveal the presence of by-product fill material. In addition, the presence of metals in soil does not appear to have adversely impacted groundwater at Site 2A/2B.

Therefore, based on future site development, no further actions were proposed with regard to metals in soil. The Port Authority is considering various development options and strategies relative to the presence of fill material. Thus, the need for additional review of environmental quality of metals in soil related to fill materials will be reviewed as part of development planning and evaluation.

pН

Investigative efforts at the site have revealed that pH in soil at Site 2A/2B ranges from 3.8 to 12, with the majority of values falling between 7.0 and 8.5. Figure 15 presents pH values at the high and low end of the recorded values: readings presented on Figure 15 are those noted to equal or exceed 10 or those noted to equal or fall below 5. The geospatial presentation of the high and low recorded pH values reveals that the higher pH values, defined as values greater than or equal to 10, appear to be most frequently recorded in samples collected from locations situated at the northern/northwestern portion of Site 2A. In addition, a localized area of lower pH, defined as values less than or equal to 5, was identified in the area of Building 20. Based on visual review of soil borings from the SI, the area noted to exhibit higher pH at Site 2A/2B appear to be associated with fill material. Fill material will be addressed in conjunction with overall site redevelopment. Issues associated with lower pH may appear to be localized to the Building 20 area and to be associated with surficial soil. The surficial soil at Building 20 will be addressed in conjunction and site redevelopment.



Site 2A/2B Report

Hatch Mott MacDonald

Potential Oil Impacted Areas (TPHC/Oil & Grease)

Visual observations and the results of laboratory analyses identified several potential "oil" impacted areas at Site 2A/2B. These areas included: several areas observed to include black staining and a distinct petroleum odor and four areas exhibiting concentrations of TPHC in excess of 10,000 mg/kg. It should be noted that the analytical results for O/G and TPHC suggest that these areas may, in some instances, be impacted by non-petroleum materials. Free product was noted on the groundwater surface in two wells at Site 2A, OP-1 and GW-14. Black staining was noted at several site locations including a few at Site 2A/2B. Taken in concert with analytical results, it appeared that "oil" impacts might be present at the following locations: Eastern Portion of the Area B and Area A including OP-1, UST7 Area, and GW-14 Area. Based on field observations and analytical results, additional actions to evaluate potential "oil" issues for soil were proposed for the following areas: Eastern Portion of the Area B and Area A including OP-1, UST7 Area, and GW-14 Area. Please refer to Section 8.2 for proposed actions.

In many instances, the presence of black staining was noted at locations, which also were characterized by cinder-type fill material. To the extent possible, the list of potential oil-impacted areas provided above reflects "oil" issues which are not attributable to the presence of trace cinders in fill material. The presence of the cinder fill material at the site is described, along with other fill material, as a separate issue later in this section.

SI efforts did not identify "oil" impacted areas at potential UST areas UST1, UST3 or UST4. However, soil borings installed at the UST7 revealed the presence of TPHC in excess of 10,000 mg/kg as well as visual signs of potential petroleum impacts. In addition, the SI revealed the presence of petroleum-related contaminants at PG-STAIN-03 which is situated in the vicinity of UST1. Regardless of the detection of contaminants through the SI, additional actions were proposed for each potential UST Area to verify that no tanks or impacted soil remain at these areas given inconclusive results from the GPR/EM survey and difficulties encountered during soil boring installation activities. Please refer to Section 8 for a discussion of proposed investigative efforts for oil-impacted areas as well as the potential UST Areas.

7.2 Groundwater

Volatile Organic Compounds

Analytical results did not identify the presence of any VOCs above the NYSDEC SVGs in samples collected from Site 2A/2B. Therefore, no further action was proposed with regard to VOCs in

 \varkappa



groundwater. However, the Port Authority proposed to re-evaluate this no further action proposal upon completion of the actions proposed to evaluate the presence of "oil" areas in soil. Please refer to Section 8 for a description of proposed RI actions.

Semi-Volatile Organic Compounds

Analytical results indicate the presence of only one SVOC, bis(2-ethylhexyl)phthalate, at a concentration in excess of NYSDEC guidance criteria in samples from two monitoring wells, PG-PZ-1, and PG-TMW-01. Bis(2-ethylhexyl)phthalate is a common laboratory contaminant and is unlikely to be an issue with regard to this site. As such, no further action was proposed with regard to SVOCs in groundwater at Site 2A/2B.

Metals

Analytical data revealed the presence of only six TAL metals (arsenic, barium, manganese, iron and sodium) at concentrations in excess of NYSDEC guidance criteria in groundwater samples. Four of these metals (barium, manganese, iron and sodium) are generally regarded as secondary contaminants with regard to water quality and are more likely to be related to regional groundwater conditions. Arsenic was only detected at an elevated concentration in two samples and the presence of this contaminant is not unexpected given the urban nature of Site 2A/2B, the HHMT-Port Ivory Facility as well as the site area. Therefore, no further action was proposed with regard to metals in groundwater at Site 2A/2B.

рΗ

Investigative efforts at the site have revealed that pH in groundwater ranges from 6.59 to 7.72. Therefore, it does not appear that fringe pH levels recorded in soil samples at Site 2A/2B have had a deleterious impact on site groundwater. Further, as presented in the Site 1 Report, the pH recorded for surface water samples collected from Bridge Creek, situated downgradient of fill-containing areas, revealed levels of pH within the normal range for saline waters, 7.5 to 8.2. Given that groundwater is not utilized for potable purposes by the site or surrounding area and that the investigation did not identify any downgradient impacted receptors with regard to pH, no additional actions were proposed with regard to pH in soil or groundwater. However, it was proposed to address historic fill material in conjunction with overall site redevelopment. The Port Authority is considering various development options and strategies relative to the presence of fill material. Thus, it was proposed to consider the need for additional review of environmental quality of pH in groundwater as part of development planning and evaluation.

Ά



Potential Oil Impacted Areas (TPHC/Oil & Grease)

Visual observations and the results of laboratory analyses identified no potential "oil" issues with regard to groundwater at Site 2A/2B. TPHC was not detected in the groundwater samples from Site 2A/2B. In contrast, O/G was detected in nine of eleven groundwater samples. In addition, free phase product was observed in wells OP-1 and GW-14. In both instances, samples were collected but insufficient sample was available to perform fingerprint analyses. Additional RI actions were proposed with regard to both OP-1 and GW-14. Please refer to Section 8 for a discussion of proposed RI efforts.

7.3 Bridge Creek – Surface Water/Sediment

Samples were obtained from both the surface water and sediment of Bridge Creek during the SI of the HHMT-Port Ivory Facility. As presented in the Site 1 Report, analytical results revealed the presence of several metals at concentrations in excess of NYSDEC guidance and screening criteria. In addition, information provided in P&G reports states that the frequency and extent of precipitates noted to be present in the stream corridor has decreased significantly over the past decade (late 1980s to late 1990s). P&G attributed the decrease, in part, to a delayed response to the installation of underground piping and containment system at an AST Area in 1984. Given the cessation of manufacturing activities at the site, the occurrence of precipitate in Bridge Creek is expected to remain stable or decrease in frequency. Further, it is anticipated that the Port Authority's development of the site will continue to enhance the quality of Bridge Creek. As such, no further investigation or delineation was proposed for Bridge Creek relative to Site 1. However, given the lack of information pertaining to surface water and sediment quality at the portion of Bridge Creek situated downgradient of Site 2A, additional sampling was proposed with respect to that portion of Bridge Creek proximate to Site 2A. Please refer to Section 8.3 for proposed sampling for Bridge Creek.

7.4 Historic Fill

Initial assessment/investigative efforts revealed that P&C placed a variety of fill material at the subject site to raise the topographic grade to facilitate site development. The investigation noted the presence of three general types of fill: (1) urban fill (including soil fill, vegetative debris, construction debris, wood, bricks, glass, concrete); (2) cinder fill (consisting primarily of ash, coal pieces, spoils from the furnace, ash-type materials with some slag); and (3) by-products from onsite production and manufacturing activities (calcium carbonate, spent diatomaceous earth, and spent carbonaceous filter material). The



Hatch Mott MacDonald

specific composition of the fill was noted to vary with location and frequently all three types of fill were noted to be present in varying concentrations at the same location.

As described previously in this report, urban fill is present throughout the site. Further, this type of fill material is considered ubiquitous with regard to waterfront sites throughout Staten Island as well as the larger region. Although trace cinders are likely to be present in urban fill, more significant cinder fill layers were noted in fill material present at some portions of Site 2A/2B. However, cinder fill was notably absent at the eastern portions of Site 2A. The third type of fill present at this site consists of a combination of process by-products such as calcium carbonate, spent diatomaceous earth, and spent carbonaceous filter material. Although this material was noted to be variable with regard to moisture content and coloration, it was readily distinguishable from other fill materials as well as underlying native materials at the site. Based on the site-wide fill investigation, the by-product fill material was predominantly located along the western border of Site 2A. No by-product fill was observed in soil borings installed at Site 2B.

Analytical results revealed the presence of a variety of contaminants including TPHC, PAH compounds and metals at a wide range of concentrations in samples collected from or including the urban fill and the cinder fill. A review of contaminant profiles of samples from each of these fill materials did not identify contaminants which were more prevalent in either type of material. The contaminants detected in these media are generally regarded as "typical" urban fill (contaminants such as arsenic, beryllium, cadmium, lead and various petroleum hydrocarbons related to fossil fuel by-products including PAH compounds). In general, the potential exists for a wide variety of contaminants to be present in historic fill material and the contaminants present at a specific site are typically linked to the source or sources of the fill materials and the composition of same. For example, arsenic and petroleum related compounds are typically present in historic fill materials taken from old railyard sites and emplaced at sites throughout the New York Metro Region. The types of contaminants detected in the samples from urban and cinder fill present at the site support this assertion. In contrast, the contaminant profile of samples collected from the byproduct fill does distinguish this material from other site fill and native material. As previously stated in this report, the by-product fill appears to be characterized by an elevated pH value and the presence of metals such as aluminum, barium, calcium, iron, magnesium and sodium and to a lesser extent, manganese and potassium rather than typical fill metals (lead, arsenic, nickel, etc.). The by-product fill material is not characterized by the presence of VOCs, SVOCs, pesticides, PCBs, TPHC or O/G, although these types of contaminants were detected at locations where the by-product fill is intermixed or located



in close proximity to soil fill or cinder fill or in samples of the by-product fill collected from the surficial interval. Based on the investigation, no additional actions are proposed with regard to the presence of fill material at Site 2A/2B. However, fill material will be identified, as appropriate, during the remedial investigation proposed to evaluate potential petroleum/oil impacted areas. As previously stated, the Port Authority will address fill material, as necessary, in conjunction with the redevelopment of this site. The Port Authority is considering various development options and strategies relative to the presence of fill material. Thus, the need for additional review of environmental quality of fill material will be reviewed as part of development planning and evaluation.

8.0 REMEDIAL INVESTIGATION WORKPLAN

Overall, the SI of Site 2A/2B described herein has revealed the presence of relatively few issues that require additional investigation/delineation and/or remediation. Further, the proposed redevelopment of the property will address many of the site contaminant issues in conjunction with construction activities. However, the SI revealed the presence of several potential petroleum-impacted areas, which require further evaluation/delineation prior redevelopment of Site 2A/2B. As such the Port Authority developed a remedial investigation workplan (RIW) to further evaluate four potential UST areas (UST1, UST3, UST4 and UST7) as well as six other areas within Site 2A/2B, which exhibited indications of potential petroleum impacts. The specific actions proposed to further evaluate the potential UST areas and the potential petroleum-impacted areas are described in the following sections. Please note, the QA/QC and Health and Safety protocols for the RI were to be consistent with those set forth for the SI as identified in Section 4.3.

8.1 Proposed Actions - Potential UST Areas

As described in Section 6.2, the geophysical survey was inconclusive due to interference with utilities and other site features. As such, the Port Authority proposed to install test pits at the potential UST Areas within Site 2 (UST1, UST3, UST4 and UST7) for the purpose of locating USTs and/or impacted soil, if present. Additional actions at these areas, such as sample collection and analyses, were to be based upon results of the proposed test pit effort. The locations of UST1, UST3, UST4 and UST7 are presented on Figure 17. In addition, soil borings installed at UST7 revealed the presence of potential petroleum impacts. Thus, RI efforts were proposed at the general UST7 Area for the purpose of confirming the presence/absence of USTs as well as at specific locations (UST7-1A and UST7-1B) to delineate previously observed petroleum impacts.





8.2 Proposed Actions - Potential Petroleum-Impacted Areas

Hatch Mott

MacDonald

As previously stated, visual observations and the results of laboratory analyses identified several areas on Site 2A/2B, which were impacted by petroleum or non-petroleum oil materials. These identified areas include the following: the location of two wells exhibiting a sheen on the groundwater surface (PG-GW-14 and PG-GW-10D); several areas observed to include black staining and a distinct petroleum odor; two locations with potential petroleum related VOC exceedances; and a few areas exhibiting concentrations of TPHC in excess of 10,000 mg/kg. Therefore, based on field screening and analytical results, it was proposed to delineate the extent of potential petroleum impacts at the following areas: Area OP-1, Area B-2, Area B-3, Area MW-10D and Area GW-14. As stated in Section 8.1, RI efforts also were proposed at soil boring locations at the UST7 Area. No delineation actions were proposed for PG-STAIN-3 given the proximity of other soil borings. Rather, it was proposed to address the "stained area" during building demolition activities. It should be noted that delineation actions were also proposed at the A-5 and FS-1B location. Although soil boring A-5 is located within Site 2A, the majority of the RI actions undertaken to address the A-5 location were situated within Site 1. Likewise with SI boring FS-1B, a majority of the RI actions undertaken to address the FS-1B were situated in Site 1. Thus, a discussion of RI activities for the A-5 and FS-1B Areas location are also provided in this report. The remaining delineation information for the A-5 and FS-1B location was presented in the previously submitted Site 1 Report. It is important to note that some or all of the potential petroleum or "oil" impacts which were observed are likely to be from non-petroleum materials such as vegetable or fish oils which were used or produced at the facility. However, for the purposes of the RI, the encountered materials will be referred to as petroleum-impacted materials or petroleum-impacted soil, as appropriate. It should also be noted that activities were performed at adjacent to location GW-1, located immediately south of Building 74. These activities were performed to further review conditions at Site 3. However, for completeness, a brief overview of information related to conditions encountered at those locations is provided herein. A more detailed discussion is provided in the Site 3 Report.

The RI for the above listed areas was to be accomplished through the installation and sampling of soil borings. Specifically, it was proposed to install and sample soil borings approximately 15 feet to the north, south, east and west of the SI soil borings, which exhibited evidence (through field screening or analytical results) of petroleum impacts. All samples were to be field screened for indications of petroleum-related contamination. If evidence of petroleum-related contamination was observed, another boring was to be installed at a distance of approximately 15 feet from the previous boring. The RIW

195

 \times

X

196



8.3 Proposed Actions – Bridge Creek

Hatch Mott

MacDonald

P&G had identified the presence of a white precipitate materials along the banks of Bridge Creek. Although P&G reports did not identify a significant environmental issue with regard to the presence of the precipitate, sampling was performed as part of the HHMT-Port Ivory Facility SI to evaluate current (year 2000) conditions relative to this issue. The results of the SI undertaken at locations adjacent to Site 1 were presented in the Site 1 Report. As no samples were collected downgradient of Site 2A, it was proposed to collect surface water and sediment samples from Bridge Creek as part of the RI activities for Site 2A/2B. Overall, it was proposed to collect four sediment and four surface water samples from representative locations along Bridge Creek. Samples were to be analyzed for TCL VOCs, TCL SVOCs, TAL Metals, TPHC, O/G, cyanide, total phenolics and pH.

9.0 RI – FIELD INVESTIGATION

The objective of the RI was to determine the extent of potential petroleum impacts in soil at Site 2A/B. No additional groundwater investigation was proposed as part of the RI for Site 2A/2B. The RI was developed and implemented to coordinate with proposed redevelopment of the site for use as an intermodal facility. RI activities were performed from May through July 2002 and additional UST removal efforts were performed at UST7 in the Fall of 2003. The UST removal effort is described in Section 10, the UST removal is not considered a remedial action since it was previously noted as an area requiring further evaluation. In addition, subsurface conditions throughout Site 2A were reviewed as part of ongoing building demolition (including removal of building footings and concrete pads) during 2002 and 2003. These efforts served to address the former potential UST Areas (UST1, UST3, UST4 and UST7) at Site 2A; no potential UST Areas were identified at Site 2B. A summary of the soil borings and samples are presented in Table 7. The soil boring locations are presented on Figure 16. The RI (Site 2A/2B) included the following areas: Area A-5, Area B-2, Area B-3, Area OP-1, Area GW-14, Area PG-MW-10D, the eastern delineation of location A-5, the eastern delineation of FS-1B and the southerm





Table 7Summary of Remedial Investigation SamplingSite 2A/2B: HHMT- Port Ivory Facility

Initial AOC	Location	Description of Issues	Description of Actions and Sampling	Sampling Methodology (Soil)
Area A West Tank Field (Southwest of Building 16)/Block 1400	A-5	Boring A-5 is located in Site 2A. The RI borings in the eastern direction are part of Site 2A. The north, south, and western RI borings are in Site 1 and were discussed in a previous report.	Three soil borings were installed around A-5 in Site 2A: A-5-E1, A-5-E2, and A-5-E3. Two soil samples were collected and submitted for laboratory analysis: A5E3-S2(2-4'), A5E5-S3(4-6').	VOC 8260; BN 8270
Potential UST Areas (UST 1, UST3, UST4, UST7)	UST1, UST3, UST4 & UST7	Sanborn Maps identified the potential presence of USTs at four areas at Site 2A. The SI identified potential petroleum impacts in a soil boring installed at the UST7 Area.	The four potential UST areas were reviewed as part of ongoing building demolition activities. No USTs or presence of impacted material was encountered during demolition efforts at UST1, UST3 and UST4. During the demolition/investigation of the area designated UST4, underground footings, trenches, piping, catch basins, and concrete manholes were encountered. Soil in this area appeared to be fill material and an odor resembling the household soap, "comet" was noted. No UST was discovered during excavation within the UST4 Area. During the breaking up of concrete pads and a concrete foundation west of Building #43, two USTs were identified. The two USTs measured approximately 26 feet in length and approximately 8 to 10 feet in diameter. The USTs were situated side-by-side in individual concrete vaults. Impacted soil directly surrounding the USTs was removed and stockpiled on site. The USTs were removed and the excavation was backfilled, with on-site soil.	VOC 8260; BN 8270





Table 7Summary of Remedial Investigation SamplingSite 2A/2B: HHMT- Port Ivory Facility

Initial AOC	Location	Description of Issues	Description of Actions and Sampling	Sampling Methodology (Soil)
Area B Former Raw Product and By-Product AST areas/Block 1400	B-2, B-3	During SI activities B-2 and B-3 exhibited indications of potential petroleum/non-petroleum substances. The majority of the soil borings installed around B- 3 are located in Site 2A. However, the farthest western boring B-3-W2 lies in Site 1 and was previously discussed in a report addressing Site 1	Twenty-two soil borings were installed surrounding B-2: B-2-S8,B-2-S7, B-2-S6, B-2-S5, B-2-S4, B-2-S3, B-2-S2, B-2-S1, B-2-E1, B-2-E2, B-2-E3, B-2-E4, B-2- W1, B-2-W2, B-2-W3, B-2-W4, B-2-N1, B-2-N2, B-2-N3, B-2-N4, B-2-N5, and B-2-N6. Eight soil borings were installed surrounding B-3. Seven of the eight soil borings were located in Site 2A: B-3-S1, GW-14-S1/B-3-N1, B-3-W1, B-3-E1, B-3-E2, and B-3-E3. Five soil samples collected from the B-2 and B-3 areas in Site 2A were submitted for laboratory analysis: B2S8-S1 (0-2'), B2S8-S2(2-4'), B2S8- S3(4-6'), B2E4-S3(4-6'), B3E3-S3(5-5.5').	VOC 8260; BN 8270
Monitoring Wells	GW-14 OP-1	Existing monitoring wells OP-1 and GW-14 were further investigated due to the presence of sheen on groundwater. A majority of the soil borings installed around GW-14 are located in Site 2A. However, the farthest western borings GW-14- W3 and GW-14-W4 lie in Site 1 and were discussed in the Site 1 Report.	Ten soil borings were installed around the center-boring GW-14. Two of the 10 are in Site 1. Eight borings were located in Site 2A: GW-14-E1, GW-14-E2, GW- 14-E3, GW-14-W1, GW-14-W2, GW-14-N1, GW-14-N2, and GW-14-N3. Two soil samples were submitted for laboratory analysis: GW-14-E3-S3(5-5.5') and GW-14-N3(3.5-4'). Seven soil borings were installed around the center-point OP-1: OP-1-E1, OP-1- N1, OP-1-W1, OP-1-S1, OP-1-S2, OP-1-S3, and OP-1-S4. Full installation of soil borings OP-1-S3 and OP-1-S4 were impeded by the presence of reinforced concrete. Five soil samples were submitted for laboratory analyses: OP1E1-S3(4-6'), OP1E1-S2(2-4'), OP1N1-S1(4.5-5'), OP1S1-S3(4-6'), and OP1W1-S3(4-6').	VOC 8260; BN 8270



Table 7Summary of Remedial Investigation SamplingSite 2A/2B: HHMT- Port Ivory Facility

Initial AOC	Location	Description of Issues	Description of Actions and Sampling	Sampling Methodology (Soil)
MW-10D	MW-10D	Monitoring well MW-10D was further investigated based on the detectable concentrations of benzene in one soil sample collected from this location.	 Utilizing MW-10D as the center-point, soil boring soil borings were installed as previously stated in the above AOC. Note, the MW-10D location was referenced as "GW-10D" during the RI and therefore, the GW-10D prefix refers to soil boring/well locations surrounding MW-10D. Five soil borings were installed around the center-point MW-10D to evaluate soil conditions: GW-10D-N1, GW-10D-N2, GW-10D-N3, GW-10D-W1, AND GW-10D-W2. Three soil samples were submitted for laboratory analysis: GW10E1-S3 (4.5'), GW-10N3-S3(4-5'), and GW10W2-S3(4-6'). 	VOC 8260; BN 8270

Note: the "PG" prefix has been removed from the MW-10D designation.

P:\232952wmd\Operable Unit Reports\Operable Unit 2\Operable Unit 2 Table 9 Summary of Remedial investigation Sampling.doc



delineation of issues at Site 3 (i.e., soil borings installed adjacent to the GW-1 location). The specific actions undertaken at each AOC are presented below.

9.1 RI Sampling Procedures / Methodology

All soil boring installation and sampling activities were performed in accordance with the guidelines set forth in the Port Authority's *Field Standard Operating Procedures Manual* dated January 1995 and appropriate NYSDEC protocols. A detailed description of soil boring installation and sampling performed during the SI is provided in Section 5.3. As same procedures for soil boring installation and sampling were utilized during the RI, the information provided in this section is limited to those aspects particular to the RI. For general information pertaining to soil boring installation and sampling, please refer to Section 5.3. All field sampling activities were performed in accordance with the Port Authority's QA/QC and Health and Safety protocol's which are presented in the Port Authority *Field Standard Operating Procedures Manual* dated January 1995.

As proposed, initial delineation at each location consisted of the installation of soil borings approximately 15 feet to the north, south, east and west of the previous SI or target soil borings. Samples were collected from the soil borings and were screened continuously for indications of petroleum contamination utilizing visual, olfactory, and instrument methods. Field screening included documenting and recording the following, as appropriate and feasible: soil boring depth, date and time of installation and sampling, photo ionization readings (if applicable), presence of water, and soil strata description (color, grain size, etc.). In those instances when groundwater was encountered, field screening also included an assessment of the presence of a sheen or free product on the water table. If the sample was noted to exhibit indications of petroleum, another boring was constructed approximately 15 feet from the previous boring in the same direction. At some locations, it was necessary to utilize intervals greater then 15 feet. If field screening by visual, olfactory, or instrument methods did not reveal any indications of petroleum above background concentrations, the location was considered an endpoint for that target boring and soil samples were collected and submitted for laboratory analysis per the RIW. As stated above, soil samples were collected from the outermost borings (i.e., assumed delineation endpoints) to confirm the limits of the potential petroleum-impacted area. As feasible, three soil samples were collected from each endpoint boring in the following manner: one sample was collected from the 0 to 2 foot interval; one sample was collected from the soil/water table interface; and, one sample was collected from the interval corresponding to the midpoint depth, as measured from ground surface to soil boring depth. As described above, field screening was performed during the soil boring installation and sampling. The field screening did not

200



reveal indications of contamination at non-specified intervals of the sampled boring locations. Based on analytical results from the SI and the objective of the RI, soil samples were submitted to a New York State certified laboratory (Hampton-Clarke/Veritech Laboratories) for NYSDEC VOCs including MTBE and TBA (8260) and PAH Compounds (8270). Soil boring and sample designations, sample depths and analytical parameters are presented in Table 7 and field observations are presented in Table 8.

9.1.1 Potential UST Areas

The RIW included the installation of test pits at the four potential UST Areas (UST1, 3, 4 and 7) at Site 2A. The purpose of the proposed test pits was to confirm that USTs did not exist at these locations. No potential UST areas had been identified at Site 2B. During 2002 and 2003, contractors retained by the Port Authority initiated building and site demolition activities at Site 2A. As part of those activities, the contractors removed concrete building footings and slabs, which allowed for visual review of the potential UST areas. No indications of USTs, other subsurface structures, or impacted soil was identified at either the UST1 or UST3 areas. Further, upon review of additional site mapping, it appears that the tank depicted on historical mapping at the UST1 Area may actually be located to the east of Building 20. As described earlier in this report, P&G removed an 8,000 gallon UST from the area east of Building 20. The tank removal was performed in accordance with NYSDEC procedures and with NYSDEC oversight, as appropriate.

Investigative efforts at the UST4 Area revealed the presence of subsurface structures including concrete building footings/foundation elements, trenches, piping, catch basins, and concrete manholes. Soil in this area appeared to include limited quantities of by-product fill material. Field screening did not reveal any elevated readings on the PID. However, an odor similar to the household soap "Comet" was noted during the subsurface review. No USTs were discovered during excavation within the UST4 Area.

During the breaking up of concrete pads and a concrete foundation west of Building 43 and north of Buildings 34/37, two USTs were identified at the UST7 Area. Each UST measured approximately 26 feet in length and approximately 8 to 10 feet in diameter. The tanks were situated side-by-side in individual concrete vaults. Based on the presence of the USTs, the Port Authority implemented actions to remove the "discovered" tanks and limited visually impacted soil. Tank removal information is provided in Section 11.



Table 8Summary of RI Field ObservationsSite 2A/2B: HHMT - Port Ivory Facility

	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	Concern	D	Direction			(Yes/No)	Analysis
		DO FO 1D	Keierence	(12/02			(Yes/INO)
1.	FS-1 Area	PG-FS-IB-	15'E 01 FS-	6/3/02	0-2 brn. Sdy gravel, t. slit;	Yes	INO
	Block 1400	EI			2-4 some wa pale green sand, dk orn. m		
			Tiocated in		Sd., I Grv. Lit. om-pale green grease		
			Site 2A		4-4.5 diatomaceous earth white gray		
		DO FO 1D	1015	(12/02	GW(a) 4 OSg	37	NT.
2.	FS-1 Area	PG-FS-IB-	JUE GEG 1D	6/3/02	0-1' brn Grvly Sd, t. sit.	Yes	NO
	Block 1400	E2	01FS-1B		1 1-2° dK brn, Grviy Sd, t. sit		
					2-4° dk brn Grvly Sd, s. sit. wd cnips, siag		
					$\begin{bmatrix} PID 5.4 \end{bmatrix}$		
					3.0 tan yellow-pale green Sd/wd		
			•		4-5 moist mottled rust/orn/bik sit,		
					diatomaceous earth		[
				1	GW (2,4.5)		
	DO 1 A	DO ES 1D	45217	(12/02	5 diatomaceous earth	X7	N7
3.	FS-I Area	PG-FS-1B-	45 E	0/3/02	1 22 how hile Site Corrections	Yes	Yes
	BIOCK 1400	ES	01 FS-16		2.2.5' have helde Solar Carry along		AB28/99-00
					2.5. 4' comparete hule Side Gran		
					2.5-4 concrete, ork, Sdy Grv		
Ì					4-5 Concrete, Drk, Sdy Grv		
	· ·				Grue 5 5? has		
	OP 1 Area	DC OD 1 E1	1575	6/17/02	O 1' concento robon	X ₂	V
4.	De-1 Area	PG-OP-I-EI		0/1//02	1. 2' concrete, rebar	res	
}	DIOCK 1400		01 01-1		2 4' andra Sd. Creu na adam		AB9/95-0
	l				2-4 churs, Su, Giv, no odors		
	ļ		ļ		4-4.5 0111 SU 4.5.5.5' has all Scholt		
			ł		4.5-5.5 OIII OI SU,SI		
					5.5-6 uk offi Su, no odors		
					Gw (m 0.0 bsg		
1		\		1		1	



Table 8Summary of RI Field ObservationsSite 2A/2B: HHMT - Port Ivory Facility

	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
	Concern	D	Direction			(Yes/No)	Analysis
		DO OD 1 MI	Reference	6/12/02	0.0.252 ()		(Yes/No)
5.	OP-1 Area	PG-OP-I-NI	$15^{\prime} N$	6/13/02	0-0.25° Grv	Yes	Yes
	BIOCK 1400		01 OF-1		3.0.5.0° controllete		AD39384
					$G_{\rm W} @ 5.0^{\circ}$ hsg		
					Gw @ 5.0 03g		
6.	OP-1 Area	PG-OP-1-S1	16' S	6/13/02	(NOT COMPLETED REFUSAL)	Yes	Yes
	Block 1400		of OP-1	6/25/02	0-4" concrete rebar		AB60465
					4"-3'concrete rebar		
-		-			3-4.5' concrete		
· · ·					4.5-5' concrete rebar		
					5-6' tan m Sd		
					6-8' tan m Sd, blk f slt, c grv		4
					GW @ 0.5 DSg		
7.	OP-1 Area	PG-OP-1-S2	37' S	6/17/02	(NOT COMPLETED REFUSAL)	Yes	No
	Block 1400		of OP-1	6/24/02	(NOT COMPLETED REFUSAL)		
Q	OP 1 Area	PG OP-1 S3	77, 8	6/17/02	0.12 Grue Sd	Vag	No
0.	Block 1400	10-01-1-55	of OP-1	0/1//02	1-3 5' hrn slt Sd Gry	105	190
	DIOCK 1400				3.5-4.5'brn cl Sd: no odors		
					4.5-5.5' brn f Sd; no odors		
			1		5.5-5.75' brn f Sd, endrs; slty stn, very faint		
					odors		
					Gw @ 5.5'		
	OP 1 Area	PG OP 1 SA	45' 8	6/17/02	0 12 concrete	Vas	No
ש.	Block 1400	10-01-1-54	of OP-1	6/19/02	OBSTRUCTION	105	ONI
	DIUCK 1400			0119102			

-203




	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
	Concern	D	Direction			(Yes/No)	Analysis
ې <u>د د مېښې د کې</u>	· · · · · · · · · · · · · · · · · · ·		Reference			· · · · · · · · · · · · · · · · · · ·	(Yes/No)
10.	OP-1 Area	PG-OP-1-W1	15' W	6/17/02	OBSTRUCTION	Yes	Yes
	Block 1400		of OP-1	6/25/02	0-8" concrete		AB60466
-					8"-2' Grv. 1 ½"		
					2-4' blk f Sd.; 1 ¹ / ₂ " Grv		
					4-5' brn blk f Sd; 1 ¹ / ₂ " Grv; no sheen, no		
					odors		
					Gw @ 5' bsg		
11.	A-5 Area	PG-A-5-E1	15' E	5/29/02	0-1' concrete	Yes	No
	Block 1400		of A-5		1-2' blk slt Sd, Grv		
			· ·	1 .	2-5' blk slt Sd, Grv; odor, sheen		
					5-6' brn Sd		
		· ·			6-7' blk Sd		
					Gw @ 5' bsg		
12.	A-5 Area	PG-A-5-E2	30' E	5/29/02	0-1' concrete pad	Yes	No
	Block 1400		of A-5		1-2' dk brn-blk Sd slt, endrs, Grv		
					2-4' blk endrs, Grv; sli odor		
					4-4.5' blk endrs, Grv; odor		
					4.5-5.5' Brn-gry Sd , t. slt, blk viscous		
					material; odor		
			<u> </u>		Gw @ 5.3' bsg		
13.	A-5 Area	PG-A-5-E3	45' E	5/29/02	0-0.5' Grv	Yes	Yes
	Block 1400		of A-5		0.5-2° concrete pad		AB 58485-6
					2-3.5° concrete pad		5/28/02
					3.5-4' blk Grv, endrs, slag		
					4-5' blk Grv, endrs, slag		
					5-6' brn-blk Sd; no odor no sheen		
					Gw @ 5./ bsg		





	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
, i i i	Concern	D	Direction			(Yes/No)	Analysis
: 			Reference			: 	(Yes/No)
14.	B-2 Area	PG-B-2-E1	15' E	5/31/02	0-0.3' Grv, orange Sd	Yes	No
	Block 1400		of B-2	i	0.3-2.5' concrete pad		
				•	2.5-4' blk slt Sd, endrs, slag; odor		
					4-5' blk silt Sd, cndrs, slag; odor		
				- 	Gw @ 5' bsg		-
15.	B-2 Area	PG-B-2-E2	30' E	5/31/02	0-0.4' Grv, slt Sd	Yes	No
	Block 1400		of B-2		0.4-2' concrete pad	· · · · ·	
					2-3.6' blk endrs, slag; odors		
					3.6-4' brn Sd; odor		
		· · ·			4-5' brn Sd; odor		
		а Т			Gw @ 5' bsg		
16.	B-2 Area	PG-B-2-E3	45' E	5/31/02	0-0.1' asphalt	Yes	No
	Block 1400		of B-2		0.1-0.3' Grv		
					0.3-1.3' RR tie		
					1.3-2' blk Grv, endrs		
					2-2.4' blk Grv, endrs		
				ļ	2.4-2.7' blk Sd slt		
					2.7-4' brn Sd slt		
					4-4.3' brn Sd slt		
					4.3-5' f-m Sd; odor		
					Gw @ 5' bsg		
17.	B-2 Area	PG-B-2-E4	60' E	5/31/02	0-3.5' concrete bldg foundation	Yes	Yes
	Block 1400		of B-2		3.5-4' slt, s Grv		AB58797
					4-6' blk Sd, t Grv; no odor, no sheen		
					Gw @ 5' bsg		
	1	<u> </u>					

205





Hatch Mott MacDonald



1	Area of Concern	Soil Boring ID	Distance and Direction Reference	Date	Field Observations and PID Readings	Located on Map (Yes/No)	Laboratory Analysis (Yes/No)
18.	B-2 Area Block 1400	PG-B-2-N1	15' N of B-2	5/30/02	0-0.1 Grv 0.1-0.5' concrete pad 0.5-2' brn slt Sd, s blk Sd 2-4' blk slt Sd, cndrs, slag 4-5' blk slt Sd, s blk Sd Gw @ 5' bsg	Yes	No
19.	B-2 Area Block 1400	PG-B-2-N2	30' N of B-2	5/30/02	0-0.3' Grv 0.3-1' concrete pad 1-2' blk endrs, s Grv 2-3' blk endr, s Grv 3-3.2' concrete pad 3.2-4' blk endrs, s Grv; odor, sheen on Gw Gw @ 4.2' bsg	Yes	No
20.	B-2 Area Block 1400	PG-B-2-N3	45' N of B-2	5/30/02	0-0.2' Grv 0.2-1 concrete pad 1-2' blk endrs Grv 2-3' blk endrs Grv 3-3.3' concrete pad 3.3-4' blk endrs, s Grv; odor, sheen Gw @ 4' bsg	Yes	No
21.	B-2 Area Block 1400	PG-B-2-N4	60' N of B-2	5/30/02	0-0.2' Grv 0.2-2' concrete pad 2-3' concrete pad 3-4' brn blk cndrs, slag 4-5' brn Sd Gw @ 4' bsg	Yes	No





	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
	Concern	LD .	Direction Reference			(Yes/No)	Analysis (Yes/No)
22.	B-2 Area Block 1400	PG-B-2-N5	75' N of B-2	5/30/02	0-0.1' Grv 0.1-3' concrete pad 3-4' blk cndrs, slag; sheen on Gw Gw @ 4' bsg	Yes	No
23.	B-2 Area Block 1400	PG-B-2-N6	90' N of B-2	5/30/02	0.0-0.1'Grv 0.1-2' concrete pad 2-4' blk cndrs, slag; sheen Gw @ 4' bsg	Yes	No
24.	B-2 Area Block 1400	PG-B-2-N7	105' N of B-2	5/31/02	0.0-0.1' Grv 0.1'- REFUSAL CONCRETE	Yes	No
25.	B-2 Area Block 1400	PG-B-2-N8	120' N of B-2	6/24/02	0-4" asphalts 4"-1' blk f Sd; Grv 1-2' brn blk f Sd 2-4' blk f Sd; slt, 1" Grv, brn cls; strng odors, sheen on soil Gw @ 4' bsg	Yes	No
26.	B-2 Area Block 1400	PG-B-2-S1	15' S of B-2	5/31/02	0-0.2' Grv 0.2-2' concrete pad 2-4' cndrs slag 4-6' cndrs, slag; odor Gw @ 5.7'	Yes	No
27.	B-2 Area Block 1400	PG-B-2-S2	30' S of B-2	5/31/02	0-0.2' Grv, slt Sd 0.2-2' concrete pad 2-3.7' blk slt 3.7-4' blk cndrs, slag 4-5' blk cndrs, slag; odor Gw @ 4.8' bsg	Yes	No



-	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
e e e e e e	Concern	D	Direction			(Yes/No)	Analysis
			Reference		1 1 1 2 2		(Yes/No)
28.	B-2 Area	PG-B-2-S3	45' S	5/31/02	0-0.2' Grv	Yes	No
	Block 1400		of B-2		0.2-2' concrete		
					2-3' concrete		
					3-4' cndrs, Grv		
					4-6' endrs, Grv; odor		
					Gw @ 4.8' bsg		
							· · · · ·
29.	B-2 Area	PG-B-2-S4	60' S	6/4/02	0-0.2' c Grv	Yes	No
	Block 1400		of B-2		0.2-0.7' concrete pad		
					0.7-2' blk Sd slt , endrs, Grv		
					2-4' blk Sd slt, Grv, endrs		
					4-5' blk Sd slt, Grv, endrs; odor		
					Gw @ 5' bsg		
				<u></u>			
30.	B-2 Area	PG-B-2-85	75' 8	6/4/02	0-0.2° c Grv	Yes	No
	Block 1400		of B-2		0.2-0.7 reinforced concrete pad		
					0.7-2' concrete pad		
					2-3° concrete pad		
					3-4' blk sd Slt, Grv, endrs		
					4-5' blk sd Slt, Grv, endrs; odor		
					Gw @ 5' bsg		
21	B 2 Area	PG B 2 S6	00' \$	6/4/02	0.0.2' o Gry	Var	Ne
J1.	Block 1400	I U-D-2-50	fB_2	0/4/02	0.2 1' concrete nod	res	INO
	DIUCK 1400		01 D-2		1.2' how ad Site Corre		
					2 4' Slt distances out		
					4.5' Condra alors aday (0.2 DID)		
					4-3 Churs, slag; odor (0.3 PID)		
			ļ		Gw (J bsg		
L			· · · · · · · · · · · · · · · · · · ·				

-208



	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
	Concern	D	Direction			(Yes/No)	Analysis
	·		Reference			······································	(Yes/No)
32.	B-2 Area	PG-B-2-S7	105' S	6/5/02	0-0.5' c Grv	Yes	No
	Block 1400		of B-2		0.5-2' concrete pad		
					2-2.5' Slt, diatomaceous earth		
					2.5-4' blk Cndrs slag		
					4-6' blk Cndrs, slag; Odor		
				· · · · · · · · · · · · · · · · · · ·	Gw @ 5.5' bsg		
33.	B-2 Area	PG-B-2-S8	120' S	6/5/02	0-0.5' c Grv	Yes	Yes
	Block 1400		of B-2		0.5-1' concrete pad		AB58965-7
					1-2' gry-blk slt Sd		
					2-4' Cndrs, slag, cl slt		
					4-6' Cndrs, slag cl Slt, no odor		
					Gw @ 5.5' bsg		
34.	B-2 Area	PG-B-2-W1	15' W	5/29/02	0-0.5' Grv, s Sd	Yes	No
	Block 1400		of B-2		0.5-1' concrete pad		
				}	1-1.5' Sd		
				ļ	1.5-2' concrete pad	. · · · · · · · · · · · · · · · · · · ·	
					2-3' endrs; odors, sheen		
			-		3-4' cndrs		i
					4-5' cndrs		
 		·			Gw @ 3' bsg		· · · · · · · · · · · · · · · · · · ·
35.	B-2 Area	PG-B-2-W2	30' W	5/29/02	0-0.5' Grv	Yes	No
	Block 1400		of B-2		0.5-2' brn-blk Sd, Grv endrs		
					2-4' Grv, endrs		
		l			Gw @ 4' bsg	<u></u>	
36.	B-2 Area	PG-B-2-W3	45' W	5/30/02	0-0.3' coarse Grv	Yes	No
	Block 1400		of B-2		0.3-1' concrete pad		
					1-2' slt Sd, endrs, Grv		
					2-3.5' slt Sd, endrs, Grv		
L					Gw @ 3.5' bsg		

- 209





	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
to the second	Concern	D	Direction			(Yes/No)	Analysis
			Reference	an <u>Station</u>			(Yes/No)
37.	B-2 Area	PG-B-2-W4	60' W	5/30/02	0-0.7' Grv, slt Sd	Yes	Yes
	Block 1400		of B-2		0.7-2' blk slt Sd, endrs, coal pes	•	AB 58574-5
					2-2.8' blk slt Sd, endrs, coal pes		5/30/02
					2.8-3.5' cndrs, coal pcs, brn slt Sd		
					Gw @ 3.5' bsg		
38.	B-3 Area	PG-B-3-E1	15' E	6/24/02	0-4" concrete	Yes	No
	Block 1400		of B-3		4"-1' concrete		
					1-1.5' c Grv		
					1.5-3' blk. F Sd; Grv		
					3-4' blk, m. Sd, t. slt; strng odors/sheen		
					from soil/water agitation		
					Gw @ 3.5'bsg		
39.	B-3 Area	PG-B-3-E2	30' E	6/24/02	0-6" asphalts/ concrete	Yes	No
	Bloc k 1400		of B-3		6"-1' brn blk f Sd; 1" Grv		
					1-2' brn blk f Sd; 1" Grv, endrs, brk		
					2-3' blk f Sd; 1" Grv		
					3-4' blk m Sd, t. slt; sw agitation free		
					product, strng odors		
					Gw @ 4' bsg		
40.	B-3 Area	PG-B-3-E3	45' E	6/24/02	0-6" concrete	Yes	Yes
	Block 1400		of B-3		6"-3' concrete		AB60466
					3-3.5' gry blk f Sd, Grv		
					3.5-5' gry blk f Sd, Grv		
					5-6' blk f Sd, cl, Grv: there is an odor and		
					some type of product on the soil and		
					g. sw agitation produced soap		
	l				bubbles		



	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
	Concern	D	Direction			(Yes/No)	Analysis
1			Reference				(Yes/No)
41.	B-3 Area	PG-B-3-S1	15'S	6/20/02	0-4" concrete	Yes	No
	Block 1400		of B-3		4"-2' concrete		
					2-3.5' blk f Sd, Grv; strng odor, sheen on		
					soil		
					Gw @ 3.5' bsg		
42.	B-3 Area	PG-B-3-W1	30'W	6/21/02	0-6" asphalt	Yes	No
	Block 1400		of B-3		6"-1' blk f Sd, Grv and endrs		
					1-2' blk f Sd, Grv and endrs, sli odors		
			i		2-3' blk f Sd, Grv and endrs, m odor		
					3-4' blk f Sd, Grv/cndrs, strng odors, sheen		1
					on soil/Gw		
					Gw @ 3' bsg		
43.	GW-10D	PG-GW-	20' E	7/3/02	0-4" Grv	Yes	Yes
	Area	10D-E1	of GW-10D		4"-1' red brn coarse Sd, Grv		AB61895
	Block 1338				1-2' brn f slty Sd, t. Grv		
1					2-3' dk brn f slty Sd		
					3-4' dk brn f slty Sd		
					4-5' dk brn f slty Sd		
 					Gw @ 5' bsg		
44.	GW-10D	PG-GW-	Proximity to	7/2/02	0-2" Grv	Yes	No
	Area	10D-N1	Texas Eastern		2"-1' reddish slty Sd, Grv		
	Block 1338		pipeline, 21'		1-2' brn red f Sd, t. slt; sli odor		
			N of GW-10D		2-3' dk brn f Sd, t. slt; sli odor, sheen on		
					water		
					Gw @ 3'		
45.	GW-10D	PG-GW-	Proximity to	7/2/02	0-2" Grv	Yes	No
	Area	10D-N2	Texas Eastern	l	2"-1' reddish sity Sd, t Grv; sli odor		
	Block 1338		Pipeline,		1-2' brn red f Sd, t. slt; apparent odor		
L		<u> </u>	35.5' N of		2-3 brn t Sd, t. slt; apparent odor		1

- 211

1



an a	Area of Concern	Soil Boring ID	Distance and Direction Reference	Date	Field Observations and PID Readings	Located on Map (Yes/No)	Laboratory Analysis (Yes/No)
			GW-10D		3-4' brn f Sd, t. slt; apparent odor4-5' brn f Sd, t. slt; apparent odor		
46.	GW-10D Area Block 1338	PG-GW- 10D-N3	Proximity to Texas Eastern Pipeline; 47.5' N of GW-10D	7/2/02	0-2" Grv 2"-1' dk brn slty Sd, Grv 1-2' dk brn slty Sd, Grv 2-3' brn slty Sd, t. Grv 3-4'brn slty m-f Sd 4-5' brn slty m-f Sd 5-6' brn slty m-f Sd Gw @ 5' bsg	Yes	Yes AB60923
47.	GW-14 Area Block 1400	PG-GW-14- E1	15' E of GW-14	6/20/02	0-4" asphalt 4"-1' blk f Sd, 1 ½" Grv 1-2' blk f Sd, c. grv2-4' blk f Sd slt, os (wd) 4-5' blk slt f Sd; sheen on Gw, strng odor Gw @ 5' bsg	Yes	No
48.	MW-10D Area Block 1338	PG-GW- 10D-W1	Proximity to Texas Eastern Pipeline. 18' W of GW- 10D	7/2/02	0-2" Grv 2"-1' dk brn slty fine-coarse Sd, Grv 1-2' brn slty f-coarse Sd; sli odor 2-3' brn slty f-coarse Sd; sli odor 3-4' reddish brn f-coarse Sd; sli odor, sheen 4-5' reddish brn f-coarse Sd; sli odor, sheen Gw @ 5' bsg	Yes	No
49.	MW-10D Area Block 1338	PG-GW- 10D-W2	35' W of GW-10D	7/3/02	0-2" Grv 2"-1' red brn slty Sd, Grv 1-2' red brn slty Sd, Grv 2-3' dk brn f slty Sd, Grv 3-5' brn f slty Sd, t. Grv 5-6' yellowish cl Slt; strng odor	Yes	Yes AB61896

212



	Area of	Soil Boring	Distance and	Date	Field Observations and PID Readings	Located on Map	Laboratory
	Concern	D	Direction			(Yes/No)	Analysis
	······································	· · · ·	Reference		Crue Q 5' has		(Yes/No)
	CW 14 Arres	DC CW 14	2025 - 6	(120/02		X7	
50.	GW-14 Area	PG-GW-14-	30 ± 01	0/20/02	12 - 4 concrete	Yes	NO
	BIOCK 1400	EZ	GW14		4 -1 Grv, blk I Sa		
					1-2° brn sit i Sd		
				-	$2 - 3^{\circ}$ bm sit i Sd		
					3-4 Drn sit i Sd, os (mulch)		
					4.5-5 gry-om ci sil, os (mulch); sw		
					agitation confirm presence of pet on		
					$\int SO(1)$		
51.	GW-14 Area	PG-GW-14-	45'E	6/20/02	0-8" concrete	Yes	Yes
	Block 1400	E3	of GW-14		8"-4' blk f Sd. 1" Grv		AB60098
					4-6' blk f Sd, 1" Grv		
l					Gw @ 6' bsg	•	
52.	GW-14 Area	PG-GW-14-	16' N	6/19/02	0-1" thin asphalt	Yes	No
	Block 1400	N1	of GW-14		1"-2' blk m Sd, Grv		
					2-3' cndrs, Grv, Sd, blk f Sd; increasing		
					odors and stn, pety sheen, odors		
					Gw @ 3' bsg		
53.	GW-14 Area	PG-GW-14-	34'N	6/19/02	0-3" asphalt	Yes	No
	Block 1400	N2	of GW-14		3"-2.5' Grv, gry m Sd		
l					2.5-3.0' blk Sd, Grv, slag, moist, impacted		
					soil, odors, sheen on Gw		
	CTTV 1.4.4	DO OW 14	501 25	6/10/00	Gw @ 3.0'		
54.	GW-14 Area	PG-GW-14-	52' N	6/19/02	0-3" Grv	Yes	Yes
	BIOCK 1400	CNI	01 GW-14		5 -1 Grv, gry Sd		AB59996
					1-3.5 II. Brn W-S m Sd.; no odor		
		l	<u> </u>		5.5-4.0 Drn-blk m Sa, very sity stn very		





	Area of Concern	Soil Boring ID	Distance and Direction	Date	Field Observations and PID Readings	Located on Map (Yes/No)	Laboratory Analysis
			Reference				(Yes/No)
					faint odors; no sheen Gw @ 4.0' bsg		
55.	GW-14 Area Block 1400	PG-GW-14- S1	15' S of GW- 14 and 15' N of B-3	6/20/02	0-4" concrete 4"-2' concrete 2-4' gry-blk f Sd slt; sheen on Gw, strng odor Gw @ 4' bsg	Yes	No
56.	GW-14 Area Block 1400	PG-GW-14- W1	18'W of GW-14	6/19/02	0-3" asphalt 3"-1' Grv, gry Sd 1-2' Grv, blk Sd 2-3' blk slt Sd, Grv; stn soil, sli odors, no product, no sheen Gw @ 3.0' bsg	Yes	No
57.	GW-14 Area Block 1400	PG-GW-14- W2	33'W of GW14	6/20/02	0-4" asphalt 4"-2' blk f slt Sd, Grv 2-2.5' blk f Sd slt; free product noted, strng odor Gw @ 2.5'	Yes	No
58.	UST – 7 Area (not initiated because of site conditions)					·	

Note:

(1) The "PG" has been removed from well designations for this table:

PG-MW-10D was referenced as GW-10D during the RI. As such, soil borings installed surrounding location MW-10D utilized the prefix "GW-10D".

9.1.2 Areas B-2 and B-3

Hatch Mott

MacDonald

As proposed soil borings were installed and sampled to the north, south, east and west of SI soil borings B-2 and B-3. The specific number of soil borings and sample locations at each area are provided below.

B-2

Twenty-four RI soil borings were installed in the north, south, east, and west of soil boring B-2. Soil samples were field screened as described in Section 9.1 resulting in the establishment of endpoints at variable distances extending from this soil boring. Specifically, soil borings were installed as follows: four soil borings were installed to the west, eight soil borings were installed to the south, eight soil borings were installed to the north, and four soil borings were installed to the east. The distances in each direction were as follows: 60 feet to the east, 120 feet to the north, 120 feet to the south, and 60 feet to the west. Six soil samples were collected from various depths of endpoint soil borings and were submitted for laboratory analyses. The RI soil borings and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are provided in Table 8.

B-3

Seven RI soil borings were installed at locations north, south, east and west of the SI soil boring B-3. All of the borings are located on Site 2A; except for, the farthest westerly boring which is located in Site 1. This westerly boring has already been discussed in a previously submitted report to the NYSDEC addressing Site 1. Soil samples were field screened as described in Section 9.1 resulting in the establishment of endpoints at variable distances extending from B-3. Specifically, soil borings were installed as follows: three to the east of the soil boring, one to the north, one to the south, and two to the west. The distances in each direction were as follows: 45 feet to the east, 15 feet to the south, 45 feet to the west, 15 feet to the north. Due to the close proximity of B-3 to monitoring well GW-14, the RI boring installed to the north of B-3 was also utilized to delineate the southwestern limit of potential impacts associated with GW-14. Two soil samples were collected from various depths of endpoint soil borings and were submitted for laboratory analyses The soil boring and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are provided in Table 8.

Site 2A/2B Report



9.1.3 Area GW-14

As proposed, soil borings were installed surrounding the monitoring well GW-14. Eleven RI soil borings were installed at locations north, south, east, and west of the SI boring GW-14. Soil samples were field screened as described in Section 9.1 resulting in the establishment of endpoints 15 to 52 feet from this soil boring. Specifically, RI soil borings were installed as follows: three to the east, three to the north, fifteen feet to the south, and four to the west. The furthest westerly borings installed as part of the RI for GW-14 are located in Site 1; therefore these borings were discussed in the Site 1 Report. The distances in each direction were as follows: 52 feet to the north, 45 feet to the east, 15 feet to the south, and 60 feet to the west (into Site 1). As described in Section 9.1.2, soil borings to the southwest of GW-14 are also utilized in the delineation of B-3. This RI boring is located equidistant from both borings. Two soil samples were collected from various depths of endpoint soil borings and were submitted for laboratory analyses. The soil boring and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are provided in Table 8.

9.1.4 Area OP-1

As proposed, soil borings were installed surrounding the SI soil boring OP-1. Seven RI soil borings were installed at locations north, south, east, and west of SI soil boring OP-1. Soil samples were field screened as described in Section 9.1 resulting in the establishment of endpoints 15 to 20 feet from this soil boring. One RI soil boring was installed in the north, west, and eastern direction at a distance of approximately 15 feet from OP-1. Four RI soil borings were installed in the southern direction, however, concrete was encountered at approximately 2 feet bgs in two of the soil borings (OP1-S2 and OP1-S4). Based on the above, the distances in each direction were as follows: 15 feet to the north, west, and east; and 16 feet to the south. Five soil samples were collected from various depths endpoint soil borings and were submitted for laboratory analyses. The soil borings and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are presented in Table 8.

9.1.5 Area A-5

As proposed, soil borings were installed surrounding SI soil boring A-5, which is located in Site 1. As previously stated, the delineation of A-5 extended in an easterly direction to locations within Site 2A. The RI borings installed in the north, south and western directions were installed in Site 1 and information pertaining to those locations are provided in the Site 1 Report. Soil samples were field screened as described in Section 9.1 resulting in the establishment of endpoints ranging from 45 to 75 feet from this



soil boring. The distances in each direction were as follows: 75 feet to the north (OP1), 75 feet to the west (OP1), 75 feet to the south (OP1), and 45 feet to the east (OP2). Eight soil samples were collected from various depths on endpoint soil borings and were submitted for laboratory analyses. The soil boring and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are presented in Table 8.

9.1.6 Area FS-1B

As proposed, soil borings were installed surrounding SI soil boring FS-1B, which is located in Site 1. As previously stated, the delineation of FS-1B in an easterly direction to include locations within Site 2A. The RI borings installed in the north, south and western directions were installed in Site 1 and information pertaining to those locations are provided in the Site 1 Report. Three soil borings were installed in the eastern direction onto Site 2A. The RI soil borings installed to the north, south, and western directions were installed in Site 1, and discussed in a previously submitted report. Soil samples were field screened as described in Section 9.1 resulting in the establishment of an endpoint 45 feet (to the east) from this soil boring. Specifically, soil borings were installed as follows: two soil borings were installed to the north (Site 1), two soil borings were installed to the south (Site 2A). The distances in each direction were as follows: 30 feet to the north (Site 1), 30 feet to the south (Site 1), 45feet to the west (Site 1), and 30 feet to the east (Site 2A). Ten soil samples were collected from various depths of endpoint soil borings sand were submitted for laboratory analyses. With regard to Site 2A, 2 soil samples were submitted from RI boring FS1B-E3. The RI soil boring and sample designations, sample depths/and analytical parameters are provided in Table 7 and field observations are presented in Table 8.

9.1.7 Area GW-1

The GW-1 Area was investigated in order to assess the soil lying on the southern side of Building 74/75 (Site 2B). The RI implemented for the K-2 area (located on Site 3) included a review of conditions between the K-2 and the northern wall of Building 74/75. Based on field observations, the RI for Site 3 was expanded to include the area south of Building 74, which is located within Site 2B. Overall, three soil borings were installed at locations west of monitoring well GW-1. Soil samples were field screened according to the procedures outlined in Section 9.1. Distances in the western direction from GW-1 extended 265 feet. Three soil samples were collected from various depths; one from each of the soil borings, and submitted for laboratory analyses. No temporary monitoring wells were installed in this



area. The soil boring and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are presented in Table 8.

9.1.8 MW-10D Area

As proposed, soil borings were installed surrounding SI monitoring well MW-10D. Six soil borings were installed at locations north, west and east of MW-10D. No soil borings were installed to the south, due to the presence of an active Texas Eastern Pipeline: Soil samples were field screened as described in Section 9.1 resulting in the establishment of endpoints 20 to 48 feet from this boring. Specifically, soil borings were installed as follows: one to the east, three to the north and two to the west. The distances in each direction from MW-10D were as follows: 35 feet to the west, 20 feet to the east, and 48 feet to the north. Three soil samples were collected from various depths of the endpoint soil borings and were submitted for laboratory analyses. No temporary monitoring wells were installed in this area. The soil boring and sample designations, sample depths and analytical parameters are provided in Table 7 and field observations are presented in Table 8.

9.2 RI - Analytical Results (Soil)

Twenty-one soil samples were collected from 58 delineation soil borings installed at Site 2A/2B; the totals include soil borings and samples used to evaluate certain limits of AOCs located on Site 1 (A-5, GW-14 and B-3). The locations of the RI soil borings are presented on Figure 18. The RI was performed to delineate the extent of potential petroleum-impacted soil and, as such, samples were submitted for VOCs (8270) and PAH compounds (8260). Table 7 identifies soil boring and sample designations and Table 8 presents the findings of field screening including soil characterization. The analytical results for HMM's sampling efforts of soil are presented in Tables 9A and 9B. Figure 18 presents the RI samples locations and a summary of analytical results. For discussion purposes, the results have been compared, as appropriate, to current NYSDEC regulatory criteria. For the RI phase of this project, the criteria utilized are NYSDEC RSCOs. Please note, the reference of these standards in this report does not represent any agreement or concurrence that same are appropriate for usage at this site.

9.2.1 Volatile Organic Compounds

VOCs were either not detected or were detected below correspond RSCOs. Further, none of the samples exhibited a total concentration of VOCs in excess of the 10 mg/kg threshold established for this contaminant class. Table 9A, presents analytical results from VOC analysis.



1

•

.

na Na Kar		U		殿		Ø		Ý
N	7	G	IJ					

ENGINEERING PROGRAM MANAGER

lat					n			Annra	
		DIN				DTM		- Appro	
V L	L /	\ <i>\\\</i>	v 0	L			_ / \ /		
					• • • • • • • • • • • • • • • • • • •				
	WW								ter (1999) e dinang e (1994) e dinang e
								/	2010-000,
							11 mar 1 - 10 mar 1 - 10 mar 1 - 10 mar 1		(10 Januar 14 July 10
	r alfanc ar sa sa an		4 1.47 4 ,499,491 (4 , 4)			n	99. da 11	1997-1991-1991-1991-1991-1991-1991-1991	ers, stradt sin er stat sjanske s
***		100 vajag (100 vajag			er falst utt för hands under open utt för		ran kurshira va ta sakaran kursiya	arta antista attasta tanta attasta t	ant methodal de meneters
							areas - Contractory		

(BLOCK 1400, LOT 1 AND **BLOCK 1338, LOT 1)** HHMT-PORT IVORY FACILITY

SITE 2A/2B SAMPLE LOCATIONS AND ANALYTICAL RESULTS MAP, **REMEDIAL INVESTIGATION**

This drawing subject to conditions in contract. All inventions, ideas, designs and methods herein are reserved to Port Authority and may not be used without its written consent.

Designed by Drawn by Checked by

FIGURE 18

Table 9A Soil Analytical Results Volatile Organic Compounds Site 2A/2B HHMT Port Ivory Facility

Location	Recommended	A5E3-S2	A5E5-S3	B2S8-S1	B2S8-S2	B2S8-S3	B2E4-S3	B2W4-S1	B2W4-S2	B3E3-S3	GW14E3-S3
Sample Date	Soil	5/29/2002	5/29/2002	6/5/2002	6/5/2002	6/5/2002	5/31/2002	5/30/2002	5/30/2002	6/26/2002	6/21/2002
Area ID	Cleanup	PG-A-5	PG-A-5	PG-B-2	PG-B-2	PG-B-2	PG-B-2	PG-B-2	PG-B-2	PG-B-3	PG-GW-14
Sample Depth (feet)	Objective	2-4'	4-6'	0-2'	2-4'	4-6'	4-6'	0-2'	2-4'	5-5.5'	5.5-6'
Concentration	mg/kg	mg/kg	mg/kg	mg/L	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
1,3,5-trimethylbenzene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
4-isopropyltoluene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0018	0.0088
Benzene	0.06	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
Ethylbenzene	5.5	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
Isopropylbenzene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
M&p Xylenes	1.2*	0.0024U	0.0025U	0.0021U	0.0025U	0.0026U	2.0U	0.0027U	0.0026U	0.0015J	0.0025U
Methyl-t-butyl ether	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
Naphthalene	13	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
N-Butylbenzene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
N-Propylbenzene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1. 0 U	0.0014U	0.0013U	0.0013U	0.0012U
O-Xylene	1.2*	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
Sec-Butylbenzene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
t-Butyl Alcohol	NS	0.012U	0.013U	0.011U	0.012U	0.013U	10U	0.014U	0.013U	0.013U	0.012U
T-Butylbenzene	NS	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
Methylbenzene	1.5	0.0012U	0.0013U	0.0011U	0.0012U	0.0013U	1.0U	0.0014U	0.0013U	0.0013U	0.0012U
Total VOCs	50	ND	ND	ND	ND	ND	ND	ND	ND	0.0018	0.0088

NS No Standard

ND Not Detected

U Undetectable Levels

* Total Xylene Recommended Cleanup Standard

...

Table 9A Soil Analytical Results Volatile Organic Compounds Site 2A/2B HHMT Port Ivory Facility

Location	Recommended	GW-14-N3	OP1E1-S2	OP1E1-S3	OP1N1-S1	OP1S1-S3	OP1W1-S3	GW10E1-S3	GW10N3-S3	GW10W2-S3
Sample Date	a Soil	6/19/2002	6/17/2002	6/17/2002	6/13/2002	6/26/2002	6/26/2002	7/8/2002	7/2/2002	7/8/2002
Area ID	Cleanup	PG-GW-14	PG-OP-1	PG-OP-1	PG-OP-1	PG-OP-1	PG-OP-1	PG-MW-10D	PG-MW-10D	PG-MW-10D
Sample Depth (feet)	Objective	3.5-4'	2-4'	4-6'	4.5-5'	4-6'	4-6'	4-5'	4-5'	4-6'
Concentration	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
1,3,5-trimethylbenzene	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
4-isopropyltoluene	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Benzene	0.06	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Ethylbenzene	5.5	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Isopropylbenzene	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
M&p Xylenes	1.2*	0.0024U	0.0024U	0.0023U	0.0026U	0.0024U	0.0024U	ND	ND	ND
Methyl-t-butyl ether	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Naphthalene	13	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
N-Butylbenzene	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
N-Propylbenzene	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
O-Xylene	1.2*	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Sec-Butylbenzene	NS	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
t-Butyl Alcohol	NS	0.012U	0.012U	0.011U	0.013U	0.012U	0.012U	ND	ND	ND
T-Butylbenzene	NS	0.0012U	0.0012U	0. <u>00</u> 11U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Methylbenzene	1.5	0.0012U	0.0012U	0.0011U	0.0013U	0.0012U	0.0012U	ND	ND	ND
Total VOCs	50	ND	ND	ND	ND	ND	ND	ND	ND	ND

NS No Standard

ND Not Detected

U Undetectable Levels

* Total Xylene Recommended Cleanup Standa

Table 9B Soil Analytical Results PAH Compounds Site 2A/2B HHMT Port Ivory Facility

									the second s		
Location	Recommended	A5E3-S2	A5E5-S3	B2S8-S1	B2S8-S2	B2S8-S3	B2E4-S3	B2W4-S1	B2W4-S2	B3E3-S3	GW14E3-S3
Sample Date	Soil	5/29/2002	5/29/2002	6/5/2002	6/5/2002	6/5/2002	5/31/2002	5/30/2002	5/30/2002	6/26/2002	6/21/2002
Area ID	Cleanup	PG-A-5	PG-A-5	PG-B-2	PG-B-2	PG-B-2	PG-B-2	PG-B-2	PG-B-2	PG-B-3	PG-GW-14
Sample Depth (feet)	Objective	2-4'	4-6'	0-2'	2-4'	4-6'	4-6'	0-2'	2-4'	5-5.5'	5.5-6'
Concentration	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Acenaphthene	41	0.41U	0.42U	0.35U	0. <u>42U</u>	0.43U	0.39U	0.45U	0.44U	0.43U	0.41U
Anthracene	50	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.099J	0.44U	0.053 J	0.088 J
Benzo(a)anthracene	0.224 or MDL	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.073J	0.44U	0.14 J	0.14 J
Benzo(a)pyrene	0.061 or MDL	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.048J	0.44U	0.11 J	0.13 J
Benzo(b)fluoranthene	1.1	0.41U	0.42U	0.35U	0.42U	0.43U	0.04J	0.24J	0.44U	0.21 J	0.12 J
Benzo(g,h,I)perylene	50	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.073J	0.44U	0.088 J	0.042 J
Benzo(k)fluoranthene	1.1	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U .	0.05J	0.44U	0.057 J	0.11 J
1,2- Benzphenanthracene	0.4	0.41U	0.42U	0.35U	0.42U	0.43U	0.05J	0.17J	0.44U	0.16 J	0.12 J
Dibenzo(a,h)Anthracene	0.014 or MDL	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.45U	0.44U	0.43U	0.41U
Fluoranthene	50	0.41U	0.42U	0.35U	0.42U	0.43U	0.066J	0.19J	0.44U	0.33 J	0.51
Fluorene	50	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.45U	0.44U	0.059 J	0.41U
Indeno(1,2,3-cd)pyrene	3.2	0.41U	0.42U	0.35U	0. 42 U	0.43U	0.39U	0.073J	0.44U	0.077 J	0.044 J
Naphthalene	13	0.41U	0.42U	0.35U	0.42U	0.43U	0.39U	0.095J	0.65J	0.11 J	0.051 J
Phenanthrene	50	0.41U	0.42U	0.35U	0.42U	0.43U	0.11J	0.16J	0.44U	0.32 J	0.28 J
Pyrene	50	0.41U	0.42U	0.35U	0.42U	0.43U	0.048J	0.077J	0.44U	0.27 J	0.22 J
Total PAH Compounds	500	ND	ND	ND	ND	ND	0.274	1.622	0.65	1.984	1.855

MDL Method Detection Limit

NS No Standard

ND Not Detected

U Undetectable Levels

-

Table 9B Soil Analytical Results PAH Compounds Site 2A/2B HHMT Port Ivory Facility

		Τ	· · · · ·	1	T		1	T		<u> </u>
Location	Recommended	GW-14-N3	OP1E1-S2	OP1E1-S3	OP1N1-S1	OP1S1-S3	OP1W1-S3	GW10E1-S3	GW10N3-S3	GW10W2-S3
Sample Date	Soil	6/19/2002	6/17/2002	6/17/2002	6/13/2002	6/26/2002	6/26/2002	7/8/2002	7/2/2002	7/8/2002
Area ID	Cleanup	PG-GW-14	PG-OP-1	PG-OP-1	PG-OP-1	PG-OP-1	PG-OP-1	PG-MW-10D	PG-MW-10D	PG-MW-10D
Sample Depth (feet)	Objective	3.5-4'	2-4'	4-6'	4.5-5'	4-6'	4-6'	4-5'	4-5'	4-6'
Concentration	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Acenaphthene	41	2.0U	0.39U	0.38U	0.43U	0.097 J	0.40U	ND	0.33J	ND
Anthracene	50	2.0U	0.39U	0.38U	0.43U	0.28 J	0.40U	ND	1.1	ND
Benzo(a)anthracene	0.224 or MDL	2.0U	0.39U	0.38U	0.43U	0.68	0.066 J	0.071J	2.8	0.049J
Benzo(a)pyrene	0.061 or MDL	2.0U	0.39U	0.38U	0.43U	0.60	0.053 J	0.056J	1.5	0.049J
Benzo(b)fluoranthene	1.1	2.0U	0.069J	0.38U	0.43U	0.61	0.088 J	0.069J	2.3	0.062J
Benzo(g,h,I)perylene	50	2.0U	0.39U	0.38U	0.43U	0.33 J	0.40U	ND	0.3J	ND
Benzo(k)fluoranthene	1.1	2.0U	0.39U	0.38U	0.43U	0.22 J	0.40U	ND	1.1	ND
1,2- Benzphenanthracene	0.4	2.0U	0.055J	0.38U	0.43U	0.69	0.071 J	0.061J	2.0	0.067J
Dibenzo(a,h)Anthracene	0.014 or MDL	2.0U	0.39U	0.38U	0.43U	0.040 J	.0.40U	ND	0.05J	ND
Fluoranthene	50	2.0U	0.057J	0.38U	0.43U	1.2	0.12 J	0.13J	3.9	0.097J
Fluorene	50	2.0U	0.39U	0.38U	0.43U	0.088 J	0.40U	ND	0.56	ND
Indeno(1,2,3-cd)pyrene	3.2	2.0U	0.39U	0.38U	0.43U	0.27 J	0.40U	ND	0.31J	ND
Naphthalene	13	2.0U	0.39U	0.38U	0.43U	0.10 J	0.40U	ND	0.28J	ND
Phenanthrene	50	2.0U	0.045J	0.38U	0.045J	0.97	0.093 J	0.046J	2.8	0.043J
Pyrene	50	2.0U	0.044J	0.38U	0.43U	1.5	0.11 J	0.1J	4.0	0.079J
Total PAH Compounds	500	ND	0.27	ND	0.045	7.675	0.601	0.533	23.33	0.446

MDL Method Detection Limit

NS No Standard

ND Not Detected

U Undetectable Levels

ţ





9.2.2 PAH Compounds

The majority of PAH compounds were either not detected or were detected below corresponding RSCOs. Only the following five PAH compounds were detected in excess of corresponding RSCOs in one or more samples from the RI samples: benzo(a)anthracene, benzo(a)pyrene, benzo(b)flouranthene, benzo(k)flouranthene, dibenzo(a,h)anthracene. All PAH compounds were detected below 1 mg/kg with the exception of a single sample at the Area OP-1 which revealed concentrations of PAH compounds ranging from not detected to 1.5 mg/kg and Area MW-10D (referenced as GW-10D) which revealed concentrations of PAH compounds ranging from 0.05 mg/kg to 2.8 mg/kg. NYSDEC has not established guidance threshold values for individual PAH or total PAH compounds. However, PAH compounds are a sub-class of the SVOC class of contaminants for which the NYSDEC has established a threshold value of 50 mg/kg for a single SVOC and a contaminant class threshold of 500 mg/kg for total SVOCs. None of the samples from the RI sampling exhibited an individual concentration or total PAH concentration in excess of the guidance thresholds. A brief summary of the PAH compounds detected at each AOC is presented below.

Area B-2

No PAH compounds were detected at concentrations greater than the NYSDEC RSCOs.

Area B-3

Only one PAH compound (Benzo(a)pyrene) was detected at an elevated concentration in the samples collected from endpoint soil borings installed to delineate the B-3 location. Sample B3E3-S3 (5-5.5 feet) exhibited a concentration of 0.11 mg/kg in excess of the RSCO.

Area GW-14

Two PAH compounds were detected at concentrations greater than the NYSDEC RSCOs. Specifically, benzo(a)anthracene was detected in excess of the RSCO in three samples and benzo(a)pyrene was detected in seven samples. Benzo(a)anthracene was detected in excess of the RSCO ranging from 0.26 mg/kg in sample GW14-10 (3-3.5 feet) to 0.27 mg/kg in samples GW14-8 (3-3.5 feet) and GW14-12 (3-3.5 feet). Benzo(a)pyrene was detected in excess of the RSCO ranging from 0.062 mg/kg in sample GW-14-3 (3-3.5 feet) to 0.24 mg/kg in sample GW14-12 (3-3.5 feet).





Area OP-1

Benzo(a)anthracene, benzo(a)pyrene, and dibenzo(a,h)anthracene were detected at slightly elevated concentrations in one sample (OP1S1-S3) collected from an endpoint boring at this AOC. Total PAH compound concentrations ranged from 0.27 mg/kg in sample OP1E1-S2(2-4 feet) to 7.675 mg/kg in sample OP11-S3 (4-6 feet).

Area FS-1B

Four PAH compounds were detected in excess of the corresponding RSCOs in several samples obtained from the FS-1B Area. Specifically, benzo(a)anthracene was detected in three samples ranging from 0.27 mg/kg in sample FS1-1 (4.5-5 feet) to 2.2 mg/kg in sample FS1-4 (4.5-5 feet). Benzo(a)pyrene was detected in six samples in excess of the RSCO ranging from 0.18 mg/kg in sample FS1-8 (4.5-5 feet) to 1.7 mg/kg in sample FS1-4 (4.5-5 feet). Benzo(b)flouranthene was detected in one sample in excess of its RSCO (sample FS1-4(4.5-5 feet)). Chrysene was detected in three samples in excess of its RSCO ranging from 0.41 mg/kg in sample FS1-1 (4.5-5 feet) to 2.3 mg/kg in sample FS1-4 (4.5-5 feet).

Area MW-10D

Slightly elevated concentrations of benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(k)flouranthene, dibenzo(a,h)anthracene were detected in the samples collected from endpoint soil borings installed to delineate the GW-10D location. The highest individual concentration of a PAH compound was 4.0 mg/kg. The cumulative concentrations for each sample ranged from 0.446 mg/kg to 23.33 mg/kg.

Area GW-1

Slightly elevated concentrations of benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(k)flouranthene, 1,2 benzphenanthracene, dibenzo(a,h)anthracene were detected in the samples collected from endpoint soil borings installed to delineate southern extent of K-2, situated on Site 3. The highest individual concentration of a PAH compound was 2.30 mg/kg. The cumulative concentrations for each sample ranged from 9.017 mg/kg to 13.18 mg/kg.

9.3 Sediment and Surface Water Analytical Data

Bridge Creek is located along the western boundary of Sites 1 and 2A (Block 1400, Lot 1) and flows to the north discharging to the Arthur Kill at the northwest corner of Future Site 4 (Block 1309, Lot 10). Bridge Creek is considered a tidal, saline stream due to the influence of the Arthur KillThe NYSDEC,



Division of Water has classified surface water into fresh and saline divisions. Bridge Creek is classified as the following: SD – due to man-made/natural conditions, the stream cannot meet primary or secondary criteria. The stream can support fish survival and limited fishing. Sampling was performed adjacent to Site 1 during the SI of the HHMT-Port Ivory Facility. Information associated with the SI of Bridge Creek is provided in the Site 1 Report.

As part of the RI, rour sediment and four surface water samples were collected from Bridge Creek and submitted to Hampton-Clarke, Inc./Veritech Laboratories (NY certification number 11408) of Fairfield. New Jersey for TCL VOC, SVOC, TAL Metals, pesticides, PCBs, TPHC, O/G and pH analysis. The analytical parameters were selected to provide information on the current environmental quality of Bridge Creek. Analytical results for sediment are presented in Tables 10A-10E and surface water samples are presented in Tables 11A-11E. The sample locations and a summary of analytical results are presented on Figure 19. For discussion purposes, the results have been compared, as appropriate, to current NYSDEC Sediment Screening Criteria and Recommended Surface Water Cleanup Standards (RSWCS); in the absence of a RSWCS, results were compared to Recommended Surface Water Guidance Criteria (RSWGC). The NYSDEC Sediment Screening establishes two levels of protection for sediments; detections below the first level area considered "not contaminated"; detections above the first level but below the second are considered contaminated but tolerable by most benthic organisms; and, detections above the second level are considered to have a pronounced disturbance of the habitat. Please note, the reference of these standards in this report does not represent any agreement or concurrence that same are appropriate for usage at this site. A discussion of the analytical results from the sediment/precipitate and surface water component of the investigation is provided below.

9.3.1 Sediment

Analytical results from sediment sampling are discussed in the following subsections by contaminants class. Analytical results for sediment sampling are provided in Tables 10A-10E with exceedences of applicable standards presented on Figure 19.

9.3.1.1 Volatile Organic Compounds

VOC compounds were either not detected or detected below the Human Health Bioaccumulation criteria in all sediment samples. Please refer to Table 10A for VOC results.



ف .

	r L <u>À F</u> J	J				
RING P	ROGRA	M MA	NAGE	?		
·						
te R	evisior				Appro	ved
ERIN	IG D	EPA	RTM	ENT		
		250.00556274887483487483	ace of a track the constant of			
LOCI BLO T-PC	K 14 CK 14 ORT	00, 1330 IVO	LO7 8, L0 9RY	Г 1 / ОТ ⁻² FAC	AND I) CILI	
	SITI	Ξ 2/ ΔΙ		SI 11	٩T	
	NT /	AN[NVI) SL EST	JRF.	ACE TIO	, N
IEUI					·	
IEUI		an sati a sa k	onditio	ne in	contr	
ving su tions, re res be us	bject ideas served ed wi	to co , des to thout	signs Port its w	and Autho ritten	meth brity cons	ods and ent.
ing su tions, re res be us	bject ideas served ed wi Drc	to co , des to thout	signs Port its w	and Autho ritten Che	meth cons cked	ods and ent.
/IEUI /ing su tions, re res be us	ibject ideas served ied wi	to co , des to thout	bigns Port its w	and Autho ritten	meth cons cked	ods and ent.

Table 10A Sediment Analytical Results Volatile Organic Compounds Site 2A/2B HHMT - Port Ivory Facility

Location	Human Health	SW-1	SW-2	SW-3	SW-4
Sample Date	Bioaccumulation	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	ug/gOC	ug/gOC	ug/gOC	ug/gOC	ug/gOC
1,1,1,2-Tetrachloroethane	0.3	0.0016ND	0.0020ND	0.0024ND	0.0016ND
1,1,1-Trichloroethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
1,1,2,2-Tetrachloroethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
1,1,2-Trichloroethane	0.6	0.0081ND	0.0098ND	0.012ND	0.0079ND
1,1-Dichloroethane	0.7	0.0081ND	0.0098ND	0.012ND	0.0079ND
1,1-Dichloroethene	0.02	0.0081ND	0.0098ND	0.012ND	0.0079ND
1,2-Dichloroethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
1,2-Dichloropropane	NS_	0.0081ND	0.0098ND	0.012ND	0.0079ND
2-Butanone	NS	0.040ND	0.049ND	0.060ND	0.040ND
2-Chloroethylvinylether	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
2-Hexanone	NS	0.032ND	0.039ND	0.048ND	0.032ND
4-Methyl-2-Pentanone	NS	0.032ND	0.039ND	0.048ND	0.032ND
Acetone	NS	0.032ND	0.039ND	0.048ND	0.032ND
Acrolein	NS	0.024ND	0.029ND	0.036ND	0.024ND
Acrylonitrile	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Benzene	0.6	0.0016ND	0.020ND	0.0024ND	0.0016ND
Bromodichrloromethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Bromoform	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Bromomethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Carbon disulfide	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Carbon tetrachloride	0.6	0.0081ND	0.0098ND	0.012ND	0.0079ND
Chlorobenzene	BALCT**	0.0081ND	0.0098ND	0.012ND	0.0079ND
Chloroethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Chloroform	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Chloromethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
CIS-1,2-Dichloroethene	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
CIS-1,3-Dischloropropene	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Dibromochloromethane	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Ethylbenzene	BALCT**	0.0016ND	0.0020ND	0.0024ND	0.0016ND
M&P-Xylenes	BALCT**	0.0032ND	0.0039ND	0.0048ND	0.0032ND
Methylene chloride	NS	0.0069JB	0.0079ЈВ	0.011JB	0.0073JB
O-Xylene	BALCT**	0.0016ND	0.0020ND	0.0024ND	0.0016ND
Styrene	NS	0.0016ND	0.0020ND	0.0024ND	0.0016ND
Tetrachloroethene	0.8	0.0081ND	0.0098ND	0.012ND	0.0079ND
Toluene	BALCT**	0.0016ND	0.0020ND	0.0024ND	0.0016ND
Trans-1,2-Dichloroethene	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Trans-1,3-Dichloropropene	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Trichloroethene	NS	0.0081ND	0.0098ND	0.012ND	0.0079ND
Vinyl chloride	0.06	0.0081ND	0.0098ND	0.012ND	0.0079ND
NS No Standard					

ND

Not Detected

BALCT Benthic Aquatic Life Crhronic Toxicity Criteria

There is no criteria for Human Health Bioaccumulation * *

C

Table 10B Sediment Analytical Results Semi-Volatile Organic Coumpounds Site 2A/2B HHMT - Port Ivory Facility

Location	Human Health	SW-1	SW-2	SW-3	SW-4
Sample Date	Bioaccumulation	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	ug/gOC	ug/gOC	ug/gOC	ug/gOC	ug/gOC
1,2,4-Trichlorobenzene	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
1,2-Dichlorobenzene	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
1,2-Diphenylhydrazine	0.58	0.81ND	0.98ND	1.2ND	0.79ND
1,3-Dichlorobenzene	NS	0.81ND	0.98ND	1.2ND	0.79ND
1,4-Dichlorobenzene	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
2,4,5-Trichlorophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2,4,6-Trichlorophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2,4-Dichlorophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2,4-Dimethylphenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2,4-Dinitrophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2,4-Dinitrotoluene	NS	0.81ND	0.98ND	1.2ND	0.79ND
2,6-Dinitrotoluene	NS	0.81ND	0.98ND	1.2ND	0.79ND
2-Chloronaphthalene	NS	0.81ND	0.98ND	1.2ND	0.79ND
2-Chlorophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2-Methylnaphthalene	NS	0.082J	0.98ND	1.2ND	0.79ND
2-Methylphenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
2-Nitroaniline	NS	0.81ND	0.98ND	1.2ND	0.79ND
2-Nitrophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
3 &4-Methylphenol	NS	0.25J	0.98ND	1.2ND	0.79ND
3,3'-Dichlorobenzidine	NS	0.81ND	0.98ND	1.2ND	0.79ND
3-Nitroaniline	NS	0.81ND	0.98ND	1.2ND	0.79ND
4,6-Dinitro-2-methylphenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
4-Bromophenyl phenylether	NS	0.81ND	0.98ND	1.2ND	0.79ND
4-Chloro-3-methylphenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
4-Chloroaniline	NS	0.81ND	0.98ND	1.2ND	0.79ND
4-Chlorophenly-phenylether	NS	0.81ND	0.98ND	1.2ND	0.79ND
4-Nitroaniline	NS	0.81ND	0.98ND	1.2ND	0.79ND
4-Nitrophenol	NS	0.81ND	0.98ND	1.2ND	0.79ND
Acenaphthene	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
Acenaphthylene	NS	0.81ND	0.98ND	1.2ND	0.79ND
Anthracene	BALCT**	0.10J	0.98ND	1.2ND	0.79ND
Benzidine	0.003	1.6ND	2.0ND	1.2ND	1.6ND
Benzo {a} anthracene	BALCT**	0.48J	0.61J	0.26J	0.19J
Benzo {a} pyrene	1.3	0.36J	0.54J	0.24J	0.14J
Benzo {b} flouranthene	NS	0.63J	0.75J	0.59J	0.24J
Benzo {g,h,I} perylene	NS	0.17J	0.21J	1.2ND	0.79ND
Benzo {k} flouranthene	NS	0.23J	0.22J	0.20J	0.091J
Bis(2-Chloroethoxy)methane	NS	0.81ND	0.98ND	1.2ND	0.79ND
Bis(2-Chloroethyl)ether	0.03	0.81ND	0.98ND	1.2ND	0.79ND
Bis(2-Chloroisopropyl)ether	NS	0.81ND	0.98ND	1.2ND	0.79ND
Bis(2-Ethylhexyl)phthalate	BALCT**	0.83	0.37J	0.76J	0.49J
Butylbenzylphthalate	NS	0.33J	0.98ND	1.2ND	0.79ND
Carbozole	NS	0.81ND	0.98ND	1.2ND	0.79ND
Chrysene	<u>NS</u> _	0.54J	0.59J	0.48J	0.22J
Dibenzo[a,h]anthracene	NS	0.81ND	0.98ND	1.2ND	0.79ND
Dibenzofuran	NS	0.81ND	0.98ND	1.2ND	0.79ND
Diethylphthalate	NS	0.81ND	0.98ND	1.2ND	0.79ND

Table 10B Sediment Analytical Results Semi-Volatile Organic Coumpounds Site 2A/2B HHMT - Port Ivory Facility

Location	Human Health	SW-1	SW-2	SW-3	SW-4
Sample Date	Bioaccumulation	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	ug/gOC	ug/gOC	ug/gOC	ug/gOC	ug/gOC
Dimethyl phthalate	NS	0.81ND	0.98ND	1.2ND	0.79ND
Di-n-butylphthalate	NS	0.20J	0.19J	0.22J	0.12J
Di-n-octylphthalate	NS	0.81ND	0.98ND	1.2ND	0.79ND
Flouranthene	BALCT**	0.95	1.5	0.50J	0.49J
Flourene	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
Hexachlorobenzene	0.15	0.81ND	0.98ND	1.2ND	0.79ND
Hexachlorobutadiene	0.3	0.81ND	0.98ND	1.2ND	0.79ND
Hexachlorocyclopentadiene	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
Hexachloroethane	NS	0.81ND	0.98ND	1.2ND	0.79ND
Indeno[1,2,3-cd]pyrene	NS	0.15J	0.21J	1.2ND	0.79ND
Isophorone	NS	0.81ND	0.98ND	1.2ND	0.79ND
Naphthalene	BALCT**	0.10J	0.98ND	1.2ND	0.79ND
Nitrobenzene	NS	0.81ND	0.98ND	1.2ND	0.79ND
N-Nitrosodimethylamine	NS	0.81ND	0.98ND	1.2ND	0.79ND
N-Nitroso-di-n-propylamine	NS	0.81ND	0.98ND	1.2ND	0.79ND
N-Nitrosodiphenylamine	NS	0.81ND	0.98ND	1.2ND	0.79ND
Pentachlorophenol	BALCT**	0.81ND	0.98ND	1.2ND	0.79ND
Phenanthrene	BALCT**	0.34J	0.25J	0.21J	0.37J
Phenol	BALCT**	0.12J	0.98ND	1.2ND	0.092J
Pyrene	BALCT**	0.84	1.4	0.49J	0.42J
NS No Standard					

631

ND

Not Detected

BALCT Benthic Aquatic Life Crhronic Toxicity Criteria * *

There is no criteria for Human Health Bioaccumulation

Table 10C Sediment Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Human Health	SW-1	SW-2	SW-3	SW-4
Sample Date	Bioaccumulation	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	ug/gOC	ug/gOC	ug/gOC	ug/gOC	ug/gOC
Aldrin	0.1	0.0054ND	0.0065ND	0.0082	0.0053ND
Alpha-BHC	NS	0.0054ND	0.0065ND	0.0079ND	0.0053ND
Beta-BHC	NS	0.0054ND	0.0065ND	0.069	0.032
Chlordane	0.001	0.039	0.013ND	0.0079ND	0.011ND
Delta-BHC	NS	0.0054ND	0.0065ND	0.0079ND	0.0053ND
Dieldrin	0.1	0.0054ND	0.0065ND	0.0079ND	0.0064
Endosulfan I	BALCT**	0.0054ND	0.0065ND	0.12	0.0053ND
endosulfan II	BALCT**	0.0054ND	0.0065ND	0.0079ND	0.0053ND
Endosulfate	NS	0.0054ND	0.0065ND	0.0079ND	0.0053ND
endrin	0.8	0.0054ND	0.0065ND	0.0079ND	0.0053ND
endrin aldehyde	NS	0.074	0.0065ND	0.036	0.011
endrin ketone	NS	0.0054ND	0.048	0.0079ND	0.0053ND
Gamma-BHC	NS	0.0054ND	0.0065ND	0.0079ND	0.0053ND
Heptachlor	NS	0.0054ND	0.0065ND	0.0079ND	0.0053ND
Heptachlor Epoxide	0.0008	0.0054ND	0.0065ND	0.0079ND	0.0053ND
Methoxychlor	BALCT**	0.0054ND	0.0065ND	0.0079ND	0.0053ND
P,P'-DDD	0.001	0.016	0.0065ND	0.11	0.066
P,P-DDE	0.001	0.049 💷 👘	0.066	0.056	0.029
P,P'-DDT	0.001	0.014	0.014	0.35	0.0053ND
Toxaphene	0.02	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1016	.0008 (total PCB)	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1221	.0008 (total PCB)	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1232	.0008 (total PCB)	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1242	.0008 (total PCB)	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1248	.0008 (total PCB)	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1254	.0008 (total PCB)	0.027ND	0.033ND	0.04ND	0.026ND
Aroclor-1260	.0008 (total PCB)	0.23	0.23	0.13	0.065
Total PCBs	.0008 (total PCB)	0.23	0.23	0.13	0.065

NS No Standard

ND Not Detected

BALCT Benthic Aquatic Life Crhronic Toxicity Criteria

** There is no criteria for Human Health Bioaccumulation

Table 10D Sediment Analytical Results Metals Site 2A/2B HHMT - Port Ivory Facility

Location	Sediment Criteria	Sediment Criteria	SW-1	SW-2	SW-3	SW-4
Sample Date	Lowest Effect Level	Severe Effect Level	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	ug/g	ug/g	ug/g	ug/g	ug/g	ug/g
Antimony	2.0	25.0	3.2ND	6.9	4.8ND	3.2ND
Arsenic	6.0	33.0	20	44	62	6.2
Barium	NS	NS	65	77	190	60
Beryllium	NS	NS	0.97ND	1.2ND	1.4ND	0.95ND
Cadmium	0.6	9.0	0.97ND	1.9	1.4ND	0.95ND
Chromium	26.0	110.0	40	110	81	28
Copper	NS	NS	100	340	130	44
Lead	31.0	110.0	130	300	240	54
Mercury	0.2	1.3	0.37	0.78	0.85	0.2
Nickel	16.0	50.0	46	160	39	26
Selenium	NS	NS	2.9ND	4.5	4.3ND	2.9ND
Silver	1.0	2.2	4.0ND	4.9ND	6.0ND	4.0ND
Thallium	NS	NS	1.9ND	2.4ND	2.9ND	1.9ND
Zinc	120.0	270.0	470	1400	870	390 - 54

NS

¥

No Standard

ND

Above LEL

Not Detected

Above SEL

Table 10E Sediment Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Location	Human Health	SW-1	SW-2	SW-3	SW-4
Sample Date	Bioaccumulation	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	ug/gOC	ug/gOC	ug/gOC	ug/gOC	ug/gOC
PETROLEUM HYDROCARBONS	NS	61	110	330	440
OIL & GREASE	NS	350	520	760	760
CYANIDE	NS	0.4ND	0.8	0.59ND	0.4ND
pH	NS	8.1	7	7.4	7.9
TOTAL PHENOLICS	BALCT**	2ND	2.5ND	3ND	2ND

NS No Standard

ND Not Detected

BALCT Benthic Aquatic Life Crhronic Toxicity Criteria

** There is no criteria for Human Health Bioaccumulation





9.3.1.2 Semi–Volatile Organic Compounds

SVOC compounds were either not detected or detected at concentrations below the Human Health Bioaccumulation criteria in the sediment samples. Please refer to Table 10B for SVOC results.

9.3.1.3 Pesticides and Polychlorinated Biphenyls

Four pesticide compounds were detected at concentrations in excess of the Human Health Bioaccumulation criteria for sediment samples collected at Site 2A/2B. Chlordane was detected in one sample; SW-1 at a concentration of 0.039 ug/gOC. P,P'-DDD was detected in three samples ranging from 0.016 ug/gOC in sample SW-1 to 0.11 ug/gOC in sample SW-2. P,P-DDE' was detected in all four sediment samples ranging in concentrations from 0.029 ug/gOC in sample SW-4 to 0.066 ug/gOC in sample SW-2. P,P-DDT' was detected in three samples ranging in concentrations from 0.014 ug/gOC in samples SW-1 and SW-2 to 0.35 ug/gOC in sample SW-3. Several additional pesticides were detected, but were at concentrations well below the Human Health Bioaccumulation criteria.

One PCB compound was detected in sediment samples collected from Site 2A/2B. Aroclor-1260 was detected in all four sediment samples in excess of the Human Health Bioaccumulation criteria. Specifically, concentrations ranged from 0.065 ug/gOC in sample SW-4 to 0.013 ug/gOC in sample SW-3. The total PCB concentration, comprised entirely of Aroclor-1260, for all four samples was also in excess of the Human Health Bioaccumulation criteria. Please refer to Table 10C and Figure 19, for pesticide and PBC results.

9.3.1.4 Metals

Several TAL metals were detected in one or more sediment samples. Eight metals were detected above either the first level (LEL) or second level (SEL) of NYSDEC screening criteria in one or more sediment samples. Four metals were detected at concentrations above the LEL but below the SEL: antimony and cadmium were detected at concentrations above the LEL in sample SW-2; mercury was detected above LEL in three samples ranging from 0.37 ug/g in SW-1 to 0.86 ug/g in sample SW-3; and chromium was detected above LEL in all four samples ranging from 28 ug/g in sample SW-4 to 110 ug/g in sample SW-2. Arsenic (two samples), lead (three samples) and nickel (one sample) exceeded the LEL in all samples and the SEL in one or more sediment samples. Arsenic was detected at concentrations ranging from 6.2 ug/g in sample SW-4 to 62 ug/g in sample SW-3. Lead was detected at concentrations ranging



from 54 ug/g in sample SW-4 to 300 mg/kg in sample SW-2. Nickel was detected at concentrations ranging from 26 ug/g in sample SW-4 to 160 ug/g in sample SW-2. Zinc was detected at concentrations in excess of the SEL in all four samples with concentrations ranging from 390 ug/g in sample SW-4 to 1400 ug/g in sample SW-2. Please refer to Table 10D and Figure 19 for metals results.

9.3.1.5 Petroleum Hydrocarbons/Oil and Grease

TPHC was detected in all four sediment samples ranging from 61 ug/gOC in sample SW-1 to 440 ug/gOC in sample SW-4. Oil & Grease was detected in all four sediment samples ranging from 350 ug/gOC in sample SW-1 to 760 ug/gOC in samples SW-3 and SW-4. Please refer to Table 10E and Figure 19 for TPHC and oil and grease results.

9.3.1.6 Cyanide and Total Phenolics

Cyanide was detected in one sample, SW-2 at a concentration of 0.8 ug/gOC. Total phenolics was not detected in any sediment samples collected from Site 2A/2B. Please refer to Table 10E for cyanide and total phenolics results.

9.3.1.7 pH

The pH recorded for sediment samples ranged from 7.0 to 8.1. The lowest pH value (7.0) was recorded at sample location, SW-2 and the highest pH value (8.1) was detected at sample location SW-1. Please refer to Table 10E and Figure 19 for pH results.

9.3.2 Surface Water

Analytical results from surface water sampling are discussed in the following subsections by contaminants class. Analytical results for sediment sampling are provided in Tables 11A-11E.

9.3.2.1 Volatile Organic Compounds

VOC compounds were either not detected or detected below the RSWCS or RSWCG for all surface water samples. Please refer to Table 11A for VOC results.

Table 11A Surface Water Analytical Results Volatile Organic Compounds Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	SW-1	SW-2	SW-3	SW-4
Sample Date	Surface Water	Surface Water	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	Cleanup Standard	Cleanup Guidanc	ug/L	ug/L	ug/L	ug/L
	ug/L	ug/L				-
1,1,1,2-Tetrachloroethane	1	NG	0.35ND	0.35ND	0.35ND	0.35ND
1,1,1-Trichloroethane	5	NG	0.28ND	0.28ND	0.28ND	0.28ND
1,1,2,2-Tetrachloroethane	NS	0.2	0.16ND	0.16ND	0.16ND	0.16ND
1,1,2-Trichloroethane	5	NG	0.58ND	0.58ND	0.58ND	0.58ND
1,1-Dichloroethane	5	NG	0.19ND	0.19ND	0.19ND	0.19ND
1,1-Dichloroethene	NS	0.7	0.33ND	0.33ND	0.33ND	0.33ND
1,2-Dichloroethane	0.6	NG	0.32ND	0.32ND	0.32ND	0.32ND
1,2-Dichloropropane	1	NG	0.46ND	0.46ND	0.46ND	0.46ND
2-Butanone	NS	50	0.53ND	0.53ND	0.53ND	0.53ND
2-Chloroethylvinylether	NS	NG	0.17ND	0.17ND	0.17ND	0.17ND
2-Hexanone	NS	50	0.25ND	0.25ND	0.25ND	0.25ND
4-Methyl-2-Pentanone	NS	NG	0.26ND	0.26ND	0.26ND	0.26ND
Acetone	NS	50	27	25	23	2.1ND
Acrolein	NS	5	0.94ND	0.94ND	0.94ND	0.94ND
Acrylonitrile	NS	0.07	0.76ND	0.76ND	0.76ND	0.76ND
Benzene	1	NG	0.20ND	0.20ND	0.20ND	0.20ND
Bromodichrloromethane	NS	50	0.28ND	0.28ND	0.28ND	0.28ND
Bromoform	NS	50	0.42ND	0.42ND	0.42ND	0.42ND
Bromomethane	5	NG	0.56ND	0.56ND	0.56ND	0.56ND
Carbon disulfide	NS	NG	0.22ND	4.2	1.4	0.22ND
Carbon tetrachloride	NS	0.4	0.21ND	0.21ND	0.21ND	0.21ND
Chlorobenzene	5	NG	0.19ND	0.19ND	0.19ND	0.19ND
Chloroethane	NS	5	0.79ND	0.79ND	0.79ND	0.79ND
Chloroform	7	NG	0.18ND	0.18ND	0.18ND	0.18ND
Chloromethane	5	NG	0.22ND	0.22ND	0.22ND	0.22ND
CIS-1,2-Dichloroethene	NS	0.7	0.35ND	0.35ND	0.35ND	0.35ND
CIS-1,3-Dischloropropene	5	NG	0.23ND	0.23ND	0.23ND	0.23ND
Dibromochloromethane	NS	50	0.44ND	0.44ND	0.44ND	0.44ND
Ethylbenzene	5	NG	0.35ND	0.35ND	0.35ND	0.35ND
M&P-Xylenes	5&5	NG	0.39ND	0.39ND	0.39ND	0.39ND
Methylene chloride	5	NG	0.25ND	0.25ND	0.25ND	0.25ND
O-Xylene	5	NG	0.23ND	0.23ND	0.23ND	0.23ND
Styrene	5	NG	0.26ND	0.26ND	0.26ND	0.26ND
Tetrachloroethene	NS	0.7	0.37ND	0.37ND	0.37ND	0.37ND
Toluene	5	NG	0.19ND	0.19ND	0.19ND	0.19ND
Trans-1,2-Dichloroethene	5	NG	0.38ND	0.38ND	0.38ND	0.38ND
Trans-1,3-Dichloropropene	NS	NG	0.22ND	0.22ND	0.22ND	0.22ND
Trichloroethene	5	NG	0.22ND	0.22ND	0.22ND	0.22ND
Vinyl chloride	NS	0.3	0.22ND	0.22ND	0.22ND	0.22ND

NS No Standard

NG No Guidance

ND Not Detected

Table 11B

Surface Water Analytical Results Semi-Volatile Organic Compounds Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	SW-1	SW-2	SW-3	SW-4
Sample Date	Surface Water	Surface Water	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	Cleanup Standard	Cleanup Guidance	ug/L	ug/L	ug/L	ug/L
	ug/L	ug/L				- i
1,2,4-Trichlorobenzene	NS	5	0.35ND	0.35ND	0.35ND	0.35ND
1,2-Dichlorobenzene	3	NG	0.24ND	0.24ND	0.24ND	0.24ND
1,2-Diphenylhydrazine	0.05	NG	0.24ND	0.24ND	0.24ND	0.24ND
1,3-Dichlorobenzene	3	NG	0.31ND	0.31ND	0.31ND	0.31ND
1,4-Dichlorobenzene	3	NG	0.21ND	0.21ND	0.21ND	0.21ND
2,4,5-Trichlorophenol	NS	NG	3.2ND	3.2ND	3.2ND	3.2ND
2,4,6-Trichlorophenol	NS	NG	1.8ND	1.8ND	1.8ND	1.8ND
2,4-Dichlorophenol	NS	5	1.6ND	1.6ND	1.6ND	1.6ND
2,4-Dimethylphenol	NS	50	1.7ND	1.7ND	1.7ND	1.7ND
2,4-Dinitrophenol	NS	10	4.4ND	4.4ND	4.4ND	4.4ND
2,4-Dinitrotoluene	NS	5	0.26ND	0.26ND	0.26ND	0.26ND
2,6-Dinitrotoluene	NS	0.07	0.24ND	0.24ND	0.24ND	0.24ND
2-Chloronaphthalene	10	NG	0.17ND	0.17ND	0.17ND	0.17ND
2-Chlorophenol	NS	NG	1.2ND	1.2ND	1.2ND	1.2ND
2-Methylnapthalene	NS	NG	1.7ND	1.7ND	1.7ND	1.7ND
2-Methylphenol	NS	NG	0.98ND	0.98ND	0.98ND	0.98ND
2-Nitroaniline	NS	5	1.8ND	1.8ND	1.8ND	1.8ND
2-Nitrophenol	NS	NG	1.6ND	1.6ND	1.6ND	1.6ND
3 &4-Methylphenol	NS	NG	0.89ND	0.89ND	0.89ND	0.89ND
3,3'-Dichlorobenzidine	NS		2.8ND	2.8ND	2.8ND	2.8ND
3-Nitroaniline	NS	5	3.4ND	3.4ND	3.4ND	3.4ND
4,6-Dinitro-2-methylphenol	NS	NG	1.9ND	1.9ND	1.9ND	1.9ND
4-Bromophenyl phenylether	NS	NG	0.20ND	0.20ND	0.20ND	0.20ND
4-Chloro-3-methylphenol	NS	NG	2.0ND	2.0ND	2.0ND	2.0ND
4-Chloroaniline	NS	5	2.5ND	2.5ND	2.5ND	2.5ND
4-Chlorophenly-phenylether	NS	NG	0.33ND	0.33ND	0.33ND	0.33ND
4-Nitroaniline	NS	NG	1.8ND	1.8ND	1.8ND	1.8ND
4-Nitrophenol	NS	NG	1.4ND	1.4ND	1.4ND	1.4ND
Acenaphthene	20	NG	0.17ND	0.17ND	0.17ND	0.17ND
Acenaphthylene	NS	NG	0.15ND	0.15ND	0.15ND	0.15ND
Anthracene	NS	50	0.17ND	0.17ND	0.17ND	0.17ND
Benzidine	NS	0.02	19ND	19ND	19ND	19ND
Benzo {a} anthracene	NS	0.002	0.34ND	0.34ND	0.34ND	0.34ND
Benzo{a}pyrene	NS	0.002	0.19ND	0.19ND	0.19ND	0.19ND
Benzo {b} flouranthene	NS	0.002	0.15ND	0.15ND	0.15ND	0.15ND
Benzo {g,h,I} perylene	NS	NG	0.23ND	0.23ND	0.23ND	0.23ND
Benzo{k}flouranthene	NS	0.002	0.19ND	0.19ND	0.19ND	0.19ND
Bis(2-Chloroethoxy)methane	NS	5	0.12ND	0.12ND	0.12ND	0.12ND
Bis(2-Chloroethyl)ether	NS	0.03	0.25ND	0.25ND	0.25ND	0.25ND
Bis(2-Chloroisopropyl)ether	NS	5	0.32ND	0.32ND	0.32ND	0.32ND
Bis(2-Ethylhexyl)phthalate	5	NG	3.1B	3.0B	2.6B	2.7B
Butylbenzylphthalate	NS	50	0.44ND	0.44ND	0.44ND	0.44ND
Carbozole	NS	NG	0.19ND	0.19ND	0.19ND	0.19ND
Chrysene	NS	0.002	0.17ND	0.17ND	0.17ND	0.17ND
Dibenzo[a,h]anthracene	NS	NG	0.21ND	0.21ND	0.21ND	0.21ND

Table 11B

Surface Water Analytical Results Semi-Volatile Organic Compounds Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	SW-1	SW-2	SW-3	SW-4
Sample Date	Surface Water	Surface Water	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	Cleanup Standard	Cleanup Guidance	ug/L	ug/L	ug/L	ug/L
	ug/L	ug/L				
Dibenzofuran	NS	NG	1.7ND	1.7ND	1.7ND	1.7ND
Diethylphthalate	NS	50	0.27ND	0.27ND	0.27ND	0.27ND
Dimethyl phthalate	NS	50	0.080ND	0.080ND	0.080ND	0.080ND
Di-n-butylphthalate	NS	50	1.2B	2.0B	0.66ND	1.3B
Di-n-octylphthalate	NS	50	0.39ND	0.39ND	0.39ND	0.39ND
Flouranthene	NS	50	0.25ND	0.25ND	0.25ND	0.25ND
Flourene	NS	50	0.19ND	0.19ND	0.19ND	0.19ND
Hexachlorobenzene	0.04	NG	0.39ND	0.39ND	0.39ND	0.39ND
Hexachlorobutadiene	0.5	NG	0.33ND	0.33ND	0.33ND	0.33ND
Hexachlorocyclopentadiene	NS	5	1.9ND	1.9ND	1.9ND	1.9ND
Hexachloroethane	5	NG	0.31ND	0.31ND	0.31ND	0.31ND
Indeno[1,2,3-cd]pyrene	NS	0.002	0.22ND	0.22ND	0.22ND	0.22ND
Isophorone	NS	50	0.46ND	0.46ND	0.46ND	0.46ND
Naphthalene	10	NG	0.17ND	0.17ND	0.17ND	0.17ND
Nitrobenzene	0.4	NG	0.15ND	0.15ND	0. <u>15</u> ND	0.15ND
N-Nitrosodimethylamine	NS	NG	1.1ND	1.1ND	1.1ND	1.1ND
N-Nitroso-di-n-propylamine	NS	NG	0.32ND	0.32ND	0.32ND	0.32ND
N-Nitrosodiphenylamine	NS	50	0.25ND	0.25ND	0.25ND	0.25ND
Pentachlorophenol	1(Total Phenols)	NG	1.0ND	1.0ND	1.0ND	1.0ND
Phenanthrene	NS	50	0.21ND	0.21ND	0.21ND	0.21ND
Phenol	1(Total Phenols)	NG	0.96ND	0.96ND	0.96ND	0.96ND
Pyrene	NS	50	0.30ND	0.30ND	0.30ND	0.30ND

NS No Standard

NG No Guidance

ND Not Detected

Table 11C Surface Water Analytical Results Pesticides and PCBs Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	SW-1	SW-2	SW-3	SW-4
Sample Date	Surface Water	Surface Water	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	Cleanup Standard	Cleanup Guidance	ug/L	ug/L	ug/L	ug/L
	ug/L	ug/L	Ĩ.			
Aldrin	NS	0.002	0.02ND	0.02ND	0.02ND	0.02ND
Alpha-BHC	NS	NG	0.02ND	0.02ND	0.02ND	0.02ND
Beta-BHC	0.04	NG	0.02ND	0.02ND	0.02ND	0.02ND
Chlordane	0.05	NG	0.2ND	0.2ND	0.2ND	0.2ND
Delta-BHC	0.04	NG	0.02ND	0.02ND	0.02ND	0.02ND
Dieldrin	0.004	NG	0.02ND	0.02ND	0.02ND	0.02ND
Endosulfan I	NS	NG	0.02ND	0.02ND	0.02ND	0.02ND
endosulfan II	NS	NG	0.02ND	0.02ND	0.02ND	0.02ND
Endosulfate	NS	NG	0. 0 2ND	0.02ND	0.02ND	0.02ND
endrin	0.2	NG	0.02ND	0.02ND	0.02ND	0.02ND
endrinaldehyde	NS	5	0.02ND	0.02ND	0.02ND	0.02ND
endrin ketone	NS	5	0.02ND	0.02ND	0.02ND	0.02ND
Gamma-BHC	0.05	NG	0.02ND	0.02ND	0.02ND	0.02ND
Heptachlor	0.04	NG	0.02ND	0.02ND	0.02ND	0.02ND
Heptachlor Epoxide	0.03	NG	0.02ND	0.02ND	0.02ND	0.02ND
Methoxychlor	35	NG	0.02ND	0.02ND	0.02ND	0.02ND
P,P'-DDD	0.3	NG	0.02ND	0.02ND	0.02ND	0.02ND
P,P-DDE	0.2	NG	0.02ND	0.02ND	0.02ND	0.02ND
P,P'-DDT	0.2	NG	0.02ND	0.02ND	0.02ND	0.02ND
Toxaphene	0.06	NG	1ND	1ND	1ND	1ND "
Aroclor-1016	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Aroclor-1221	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Aroclor-1232	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Aroclor-1242	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Aroclor-1248	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Aroclor-1254	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Aroclor-1260	0.09*	NG	0.5ND	0.5ND	0.5ND	0.5ND
Total PCBs	0.09	NG	ND	ND	ND	ND

NS No Standard

NG No Guidance

ND Not Detected

* Total PCBs

239
Table 11D Surface Water Analytical Results Metals Site 2A/2B HHMT - Port Ivory Facility

Location	Recommended	Recommended	SW-1	SW-2	SW-3	SW-4
Sample Date	Surface Water	Surface Water	4/29/2003	4/29/2003	4/29/2003	4/29/2003
Concentration	Cleanup Standard	Cleanup Guidance	ug/L	ug/L	ug/L	ug/L
	ug/L	ug/L				
Antimony	3	NG	7.5ND	7.5ND	7.5ND	7.5ND
Arsenic	50	NG	16	11	14	12
Barium	1000	NG	110	93	91	94
Beryllium	NS	3	4.0ND	4.0ND	4.0ND	4.0ND
Cadmium	5	NG	2.0ND	2.0ND	2.0ND	2.0ND
Chromium	50	NG	25ND	25ND	25ND	25ND
Copper	200	NG	25ND	25ND	25ND	25ND
Lead	50	NG	15	5.0ND	5.0ND	5
Mercury	0.7	NG	0.20ND	0.20ND	0.20ND	0.20ND
Nickel	100	NG	13	10ND	10ND	10ND
Selenium	10	NG	25ND	25ND	25ND	25ND
Silver	50	NG	10ND	10ND	10ND	10ND
Thallium	NS	0.5	5.0ND	5.0ND	5.0ND	5.0ND
Zinc	NS	2000	110	31	25ND	71

-240

NS No Standard

NG No Guidance

ND Not Detected

Table 11E Surface Water Analytical Results TPHC, Oil and Grease, pH, Cyanide and Total Phenolics Site 2A/2B HHMT - Port Ivory Facility

Recommended	Recommended	SW-1	SW-2	SW-3	SW-4	
Surface Water	Surface Water	4/29/2003	4/29/2003	4/29/2003	4/29/2003	
Cleanup Standard	Cleanup Guidance					
NS	NG	1.2ND	1.1ND	1.1ND	1.2ND	
15,000 (max)	NG	1.2ND	1.2ND	1.1ND	1.2ND	
0.2	NG	0.01ND	0.01ND	0.01ND	0.01ND	
NS	NG	7.9	7.9	8.1	7.9	
0.001	NG	0.05ND	0.05ND	0.05ND	0.05ND	
	RecommendedSurface WaterCleanup StandardNS15,000 (max)0.2NS0.001	Recommended Surface WaterRecommended Surface WaterCleanup StandardCleanup GuidanceNSNG15,000 (max)NG0.2NGNSNG0.001NG	Recommended Surface WaterRecommended Surface WaterSW-1 4/29/2003Cleanup StandardCleanup GuidanceNSNG1.2ND15,000 (max)NG1.2ND0.2NG0.01NDNSNG7.90.001NG0.05ND	Recommended Surface Water Recommended Surface Water SW-1 SW-2 Cleanup Standard Surface Water 4/29/2003 4/29/2003 NS NG 1.2ND 1.1ND 15,000 (max) NG 1.2ND 1.2ND 0.2 NG 0.01ND 0.01ND NS NG 7.9 7.9 0.001 NG 0.05ND 0.05ND	Recommended Surface Water Recommended Surface Water SW-1 SW-2 SW-3 Cleanup Standard Surface Water 4/29/2003 4/29/2003 4/29/2003 NS NG 1.2ND 1.1ND 1.1ND 15,000 (max) NG 1.2ND 1.2ND 1.1ND 0.2 NG 0.01ND 0.01ND 0.01ND NS NG 7.9 7.9 8.1 0.001 NG 0.05ND 0.05ND 0.05ND	

241

NS No Standard

NG No Guidance

ND Not Detected

9.3.2.2 Semi-Volatile Organic Compounds

Hatch Mott

MacDonald

SVOC compounds were either not detected or detected at concentrations below the RSWCS or RSWCG for all surface water samples. Please refer to Table 11B for SVOC results.

9.3.2.3 Pesticides and Polychlorinated Biphenyls

No pesticides or PCBs were detected for surface water samples. Please refer to Table 11C for pesticide and PCB results.

9.3.2.4 Metals

Several TAL metals were detected in one or more samples of surface water, but all were <u>detected at</u> concentrations below the LEL. <u>Please</u> refer to Table 11D for metals results.

9.3.2.5 Petroleum Hydrocarbons/Oil and Grease

TPH and oil/grease were not detected in the four surface water samples. Please refer to Table 11E/for TPHC and oil/grease results.

9.3.2.6 Cyanide and Total Phenolics

Cyanide and total phenolics were not detected in the four surface water samples. Please refer to Table 11E for cyanide and total phenolics.

9.3.2.7 рН

The pH recorded for surface water samples ranged from 7.9 to 8.1. The pH value was recorded at 7.9 in three locations (SW-1, SW-2 and SW-4). The pH value of 8.1 was detected at sample SW-3. Please refer to Table 11E for pH results for surface water.

9.4 RI Summary

RI activities were proposed for seven general areas at Site 2A/2B: Area A-5, Area B-3, Area B-2, Area OP-1, Area GW-14 and Area FS1-1 at Site 2A and Area MW-10D at Site 2B. In addition, RI activities were also proposed to evaluate the southern limits of the K-2 location, which is situated on Site 3. Due to the presence of Buildings 74/75, the evaluation of the K-2 Area was performed within the limits of Site



Hatch Mott MacDonald

2B, in the vicinity of GW-1; the samples were designated with a GW-1 prefix to identify location and not to reference GW-1 as an AOC. Field screening identified the limits of the petroleum impacts at the above listed AOCs through visual, olfactory and field instrumentation. Analytical results confirmed the conclusions rendered through field screening activities performed during the field investigation component of the RI. Thus, the RI implemented at Site 2A/2B has successfully delineated petroleum impacts at Area A, Area B-3, Area B-2, Area OP-1, Area GW-14 and Area FS-1B on Site 2A.

With respect to Area MW-10D located on Site 2B, the RI has successfully delineated the extent of potential petroleum impacts to the north, west and east, however, the RI activities to the south were limited by the presence of an active pipeline. Based on the results of the RI, the Port Authority has reviewed remedial alternatives to address potential petroleum impacts at Site 2A/2B. Given the redevelopment plan (i.e., the contemplated use) for Site 2A/2B, it was determined that the most appropriate remedial alternative to address the petroleum-impacted areas is hot spot excavation with offsite disposal for the majority of the above listed areas. With regard to Area MW-10D further evaluation is being performed based on the presence of the nearby pipeline. A discussion of the selected remedial alternative is presented in Section 12.

In addition, the potential UST Areas were reviewed in conjunction with building/site demolition activities and resulted in the identification of two tanks at the UST7 Area. Tank removal actions are presented in Section 11 of this report.

10.0 EXPOSURE ASSESSMENT

This Exposure Assessment (EA) addresses conditions at Site 2A/2B. As previously stated, this portion of the former industrial site is being redeveloped as the intermodal component of an intermodal/container storage facility. This EA describes the exposure setting, the nature of on-site contaminants, potential exposure points and routes and identifies potential exposure populations.

10.1 Exposure Setting

The HHMT-Port Ivory Facility is situated in an industrial section in the northwestern portion of Staten Island. Generally, the site is bordered by industrial/commercial businesses, roadways, surface water bodies (Arthur Kill and Bridge Creek) and undeveloped/vacant areas. No residential populations are situated immediately adjacent to Site 2A/2B. Site 2A/2B encompasses 22.6 acres and, at the time of Port Authority purchase, was improved by approximately twenty-nine buildings. Site 2A/2B was utilized for

243



offices, boiler buildings, furnace structures, manufacturing and development buildings, locomotive maintenance, security stations, cafeteria, a sewage treatment plant and a reservoir for fire protection. Tanks, both USTs and ASTs, as well as an underground network of piping for drainage control were formerly situated at Site 2A/2B.

The entire HHMT-Port Ivory Facility including Site 2A/2B is serviced by connections to the potable water and sanitary system of New York City. No septic systems or potable water wells are reported to be located or have been located at Site 2A/2B, or elsewhere on the HHMT-Port Ivory Facility. Storm water generated on the site is directed via a sheet flow to on-site catch basins. These catch basins discharge, through the facilities underground stormwater sewer system, to the adjacent waterways, roadways, and marshland. Bridge Creek though not directly located on the site is a tidal, saline stream, which has been classified as SD by the NYSDEC. This classification indicates that due to man-made/natural conditions the stream cannot meet primary or secondary criteria. The stream can support fish survival and limited fishing.

In addition, several utility easements and pipelines traverse the subject site. Colonial Pipeline, Exxon (now known as ExxonMobil) and Texas Pipeline (maintained by SOHIO) maintain the easements. Colonial Pipeline maintains a 10-foot pipeline easement that extends in a north/south direction along the western property boundary of Sites 1 and 2A. The easement initiates in the far southwestern corner of Site 2B and extends along the southern and southwestern corner into Site 2A. The easement traverses through the southern portion of Site 2A in a northeasterly direction and enters the southwestern corner of Site 1. The pipeline continues through Site 1 extending across Richmond Terrace and through the western portion of Future Site 4 (Block 1309, Lot 10). The easement terminates at the northern end of Future Site 4 (Block 1309, Lot 10). ExxonMobil maintains an 18-foot easement that is located generally east of the Colonial Pipeline easement. This easement parallels the Colonial Pipeline easement initiates at the western portion of Site 2A, just south of Site 1 and extends through Site 1 to Richmond Terrace where it turns in a easterly direction and extends along the southern boundary of Future Site 4 (Block 1309, Lot 10). Two Texas Pipeline (maintained by SOHIO) easements extend in a east/west direction through Site 2B. In addition, Tidewater Pipe Company LTD formerly operated pipelines that extend in an east/west direction through Site 2B and continue in a northerly direction through Site 3. These abandoned pipelines continue across Richmond Terrace and extend along the easterly property boundary of Future Site 4.



Investigative activities undertaken as part of the previously described SI and RI indicate that the HHMT-Port Ivory Facility consists of non-indigenous fill on top of organic clays and peat or sand deposits. Fill was placed upon tidal salt-marsh or sand deposits to raise the elevation of the land for development. The fill is reported to have included: sand, silt, gravel mixed with debris, cinders generated from on-site coalfired boilers, calcium carbonate and other carbonate salts generated as a by-product from soap manufacturing processes, spent diatomaceous filter earth from vegetable oil refining operations, and carbonaceous filter material from glycerin recovery operations. Generally, Site 2A/2B is characterized by the presence of urban fill material including cinders. The presence of by-product fill material at Site 2A/2B is generally limited to the northwestern portion of Site 2A along the boundary line between Site 1 and Site 2A.

10.2 Nature of On-Site Contaminants

The SI/RI activities undertaken by HMM included investigation of the soil and groundwater at Site 2A/2B as well as sediment and surface water of Bridge Creek. The SI/RI for soil at Site 2A/2B included the installation of over 50 soil borings and the collection of over 80 soil samples. Groundwater investigations included the installation of 5 monitoring wells, one temporary well, recording water levels from all newly installed wells and four existing wells, reviewing wells for the presence of free phase floating product and the collection and laboratory analysis of 10 groundwater samples.

In soil, only three VOCs, trans-1,3-dichloropropene (0.43 mg/kg), dichloromethane (0.17 mg/kg) and benzene (0.32 mg/kg) were detected in excess of corresponding RSCOs in the SI. No elevated concentrations of VOCs were detected during the RI. In the instances, when the contaminant was detected, it was marginally above the MDL for the compound. The total VOCs concentration was below the NYSDEC guidance criteria of 10 mg/kg for all samples collected from Site 2A/2B. A number of SVOCs were detected in soil samples collected from Sites 2A/2B during the SI and RI. However, the vast majority of these compounds were detected below NYSDEC TAGM RSCOs. The following SVOCs were detected at concentrations above corresponding RSCOs in one or more soil samples from Site 2A/2B: acenaphthaene, anthracene, flouranthene, flourene, pyrene, phenanthrene, naphthalene, benzo(b)fluoranthene, benzo(a)pyrene, benzo(a)anthracene, dibenz[a,h]anthracene, benzo(k)flouranthene, indeno(1,2,3-cd)pyrene, and phenol. The total SVOC concentration was below the NYSDEC guidance criteria for all samples except the surface sample collected from location PG-STAIN-03, situated on Site 2A. The sample also exhibited the presence of a few PAH compounds in excess of 50 mg/kg. No PCBs

2/16



were detected at concentrations in excess of corresponding RSCOs for surface or subsurface soil, as applicable.

Five pesticides (chlordane, endrin, dieldrin, heptachlor and heptachlor epoxide) were detected in excess of corresponding TAGM RSCOs in only a few samples collected from Site 2A/2B. Specifically, Dieldrin was detected at concentrations in excess of its corresponding RSCO in three samples; chlordane and endrin were both detected in excess of their individual RSCO in two samples; and, heptachlor and heptachlor epoxide were each detected in excess of their RSCO in only a single sample.

The NYSDEC TAGM generally regards site background as an appropriate concentration for the 24 TAL metals and has only established RSCOs for a portion of the targeted metals. Therefore, analytical results from investigative efforts were compared to the upper limit of the Eastern USA Background Range in the absence of a RCSO standard. This application is particularly appropriate in this instance given the presence of fill material and the urban nature of the site and site area. Analytical results revealed exceedances of RSCOs in one or more soil samples for 12 of the 13 of the metals with established guidance criteria; vanadium as not detected in excess of its RSCO in any of the soil samples from Site 2A/and 2B. Lead is reviewed by the NYSDEC on a case-by-case basis with 500 mg/kg established as the upper limit of the Eastern USA Background Range . Levels of lead in samples collected from Site 2A/2B ranged from not detected to 950 mg/kg.

Guidance criteria have not been established for cyanide or total phenolics. The presence of these types of contaminants is reviewed on a case-by-case basis by the NYSDEC. Cyanide was detected in several soil samples collected from Site 2A/2B. In the majority of instances, cyanide was detected at a concentration of less than 1 mg/kg. Total phenolics ranged from not detected to 5.5 mg/kg in 2 samples.

Four samples exhibited concentrations of TPHC in excess of 10,000 mg/kg. All four exhibiting concentrations above 10,000 mg/kg samples were collected from Site 2A. Eight samples exhibited a concentration of oil/grease above 10,000 mg/kg.

Samples collected ranged in pH from 3.8 to 12 with a little less then the majority, approximately 46%, of the values falling between 7.0 and 8.5. It should be noted that all of the samples exhibiting pH concentrations at or above 10 were collected from soil borings installed into the by-product fill material present at the site. SI/RI activities identified three types of fill at Site 2A/2B: urban fill (historic fill)



Ż

including soil fill, vegetative debris, construction debris (wood, glass, brick, concrete); cinder fill consisting of ash and ash-type materials with some slag; and by-product fill from production activities (calcium carbonate, spent diatomaceous earth, and spent carbonaceous filter material). As previously stated, by-product fill was generally limited to the northwestern portion of Site 2A and was not identified at Site 2B. The majority of fill material encountered of Site 2A/2B consisted of urban fill and cinder fill.

Several petroleum-impacted areas were identified within areas consisting of fill material. Further sampling efforts (i.e., RI phase) performed at these petroleum-impacted areas delineated the extent of "impacted" areas, which may be unrelated to the fill material. Analytical results from endpoint sampling during the RI did not reveal the presence of any VOCs and revealed relatively low levels of PAH compounds.

Overall, very few contaminants were detected in groundwater samples collected from Site 2A/2B. No VOCs, pesticides or PCBs were detected in groundwater samples and only a single SVOC, bis(2-thylhexyl)phthalate, was detected in excess of its corresponding SVG in groundwater samples from Site 2A/2B. Bis(2ethylhexyl) phthalate is frequently identified as a laboratory contaminant and, in fact, this compound was identified as being a laboratory contaminant in other groundwater samples collected with regard to this project.

A number of TAL metals were detected in one or more groundwater samples collected as part of the groundwater investigation of Site 2A/2B. However, only six TAL metals were detected at concentrations in excess of corresponding NYSDEC SVGs: arsenic, barium, iron, magnesium, manganese, and sodium. Barium was detected in one sample whereas arsenic, iron, manganese, magnesium and sodium were detected in several samples at elevated levels. Cyanide was detected below its corresponding SVG and total phenolics were not detected in any groundwater samples. TPHC was not detected in the groundwater samples from Site 2A/2B. In contrast, O/G was detected in six of eleven groundwater samples.

Four sediment and four surface water samples were collected at locations along Bridge Creek generally within the boundary of Site 2A. Analytical results from surface water and sediment samples revealed the presence of several metals at concentrations in excess of NYSDEC screening and guidance criteria. Four pesticide compounds; chlordane, P,P'-DDD, P,P'-DDT and P,P'-DDE were detected in sediment samples in excess of the Human Health Bioaccumulation criteria during the RI. Total PCB concentrations and concentrations of Arclor-1260 were detected in all sediment samples in excess of the Human Health

247



Bioaccumulation criteria. No contaminant gradient was identified with respect to the detected contaminants indicating that the presence of same is a product of the urban nature of the stream corridor. Cessation of manufacturing activities at the site and the redevelopment of the site by the Port Authority is expected to continue to enhance the quality of Bridge Creek.

Based on the findings of the SI, HMM performed RI activities to delineate the presence of petroleum impacts at certain site locations. The RI successfully delineated the extent of petroleum impacts in soil and, in some instances, provided additional information pertaining to UST areas. The RI did not identify the presence of any VOCs in excess of NYSDEC guidance criteria or any free product conditions at Site 2A/2B. The specifics of the RI and UST evaluation efforts were presented earlier in this Report. Based on the results of the SI and RI, the Port Authority has proposed to address petroleum-impacted soil through source area excavation and removal. The remainder of the contaminants will be addressed as part of site redevelopment through the use of engineering and institutional controls.

10.3 Potential Exposure Points and Routes

The SI/RI revealed elevated concentrations (i.e., concentrations in excess of the NYDEC AGVs, RSCOs, and SVGs) of contaminants have been detected in environmental media at the site. These contaminants however are expected given the sites historical usage and its location in a historically urbanized area. In addition to contaminants that occurred directly as an outcome of historical site activities it can be assumed that contaminants at the site have migrated to the site from various sources found in historically urban areas. Generally the contaminants and levels that were found during the SI are believed to relate to both the sites historical use, location, and the fill reported to have been emplaced at the site as part of site development activities. Generally, the contaminants detected at concentrations in excess of NYSDEC soil guidance criteria included typical historic fill contaminants such as PAH compounds, metals, low levels of PCBs/pesticides, TPHC and O/G. As previously stated, non-fill contaminants such as VOCs were detected at elevated concentrations in only 3 of over 80 soil samples collected from Site 2A/2B.

In groundwater, no VOCs and only one SVOC were detected in excess of NYSDEC groundwater standards. The TAL metals iron, arsenic, barium, magnesium, manganese, and sodium were detected in excess of NYSDEC groundwater standards. In sediment, the several metals, pesticides/PCBs exceeded NYSDEC screening criteria. Analytical results did not reveal the presence of any contaminants at concentrations in excess of NYSDEC guidance criteria in one or more surface water samples.



On most sites, the most likely route of exposure for human receptors would be through ingestion of the contaminated soil, sediment or water or inhalation of airborne dust/particulates created through soil erosion in exposed areas of the site. However, on this site, a low potential exists for human contact, and thus few exposure points exist with regard to contaminants present at the site based on the two following conditions: (1) No human populations are situated in the immediate vicinity of the site. Persons present at the site are limited to Port Authority personnel or contractors retained by the Port Authority; and, (2) The Port Authority has implemented health and safety measures to minimize contact with contaminants by all persons currently performing tasks at the site. In addition, the Port Authority requires that contractors have and implement health and safety plans based on their tasks.

As previously stated, groundwater is not utilized for potable purposes and thus human populations will not contact groundwater. Bridge Creek is situated west of Site 2A. However, the Creek's physical position between the HHMT-Port Ivory Facility and the Howland Hook Marine Terminal does not provide easy access for area persons. Further, a fence is present along Bridge Creek, which further restricts access. Thus, Bridge Creek is considered to have a low potential as a point of exposure for human populations.

Any contamination remaining after the performance of remedial actions will be addressed through redevelopment efforts including engineering and institutional controls. Thus, exposure points will be eliminated in conjunction with site redevelopment.

10.4 Receptor Populations

As previously stated, no human population reside at or in the immediate vicinity of the Site 2A/2B. Further, persons currently present at the site are limited to Port Authority personnel or contractors retained by the Port Authority. To minimize exposure, the Port Authority and its contractors have implemented health and safety measures to minimize contact with contaminants by all persons currently performing tasks at the site. Additional persons will be present on site subsequent to the completion of site redevelopment. As the site will be redeveloped for industrial purposes (intermodal/container storage facility), no resident population will occupy the site. Contamination at the site will have been addressed prior to these future worker populations being present at the site.





10.5 Exposure Assessment Summary

Information gained through the SI /RI has revealed the presence of fill material and a contaminant profile, which is consistent with urban sites located in the New York Metropolitan Region. The presence of contaminants in the soil does not appear to have significantly impacted groundwater quality at the site. Based on delineation efforts, petroleum impacts (identified through assessment and investigation activities) will be addressed through excavation of source areas. Any residual contamination will be addressed through and in conjunction with site redevelopment efforts.

Human receptors have not been identified in the immediate vicinity of the site and health and safety procedures are employed by the Port Authority and its contractors to minimize exposure to persons working at the site during ongoing redevelopment efforts. The intended future redevelopment of the site as an intermodal/container storage facility will further restrict contaminant pathways/routes through the installation of pavement and other semi-impervious material, which will function as an environmental cap throughout the entire site. This action will stabilize contaminants in the soil and fill material by impending infiltration as well as minimize or, in some cases, prevent erosion, as well as forming a barrier to human exposure to impacted soil and groundwater. Redevelopment of the site also is anticipated to continue to reduce any residual contamination in sediment/surface water at Bridge Creek, thus enhancing water quality and virtually eliminating this creek as a pathway of contaminants to human receptors.

The Port Authority will continue to implement appropriate actions to minimize exposure to human populations during remedial efforts and site redevelopment. In addition, the Port Authority will monitor the integrity of any engineering controls employed as part of the overall site remedial and redevelopment strategy. Given the above, no further action is required with regard to exposure assessment for Site 2A/2B.

11.0 UST REMOVAL

During the demolition of the concrete foundation located in the vicinity of Building S-35, in September 2002, the Port Authority identified the presence of an UST. A review of available historical records revealed that the UST encountered during the demolition activities was a former #6 fuel oil tank, which had been closed in place by P&G.

Subsequently, the Port Authority removed surface soil surrounding the UST and identified a second UST situated adjacent to the originally encountered tank. Further evaluation revealed that both tanks were



located within concrete vaults and were filled with inert material (bricks, stone and sand). Subsequently, the Port Authority removed the tanks and the surrounding concrete vaults. Indications of petroleum impacts to the surrounding soil were observed during excavation activities. As a result, the Port Authority excavated visually impacted soil immediately adjacent to the vaults. Excavated soil was stockpiled on-site pending off-site disposal at an appropriate recycling/disposal facility. The excavation measured approximately 25 feet in length, 20 feet in width, and approximately 11 feet in depth; groundwater was encountered at approximately 8 feet bgs. The USTs and all connected piping were removed and set aside for off-site recycling with the other recycled materials from the demolition activities. Subsequently, the area was backfilled with existing site soil/crushed concrete. Based on the above described tank removal actions, no further action is proposed with regard to UST7.

12.0 SUMMARY OF REMEDIAL ACTIONS

12.1 Proposed Remedial Actions

The SI of Site 2A/2B revealed a variety of contaminants at a wide-range of concentrations in samples collected from soil, sediment, surface water and groundwater. The presence of these contaminants was not unexpected given the former use and location of the site. Overall, given that the subject site is located in a highly urbanized and historically industrial area, it is reasonable to assume that diffuse anthropogenic pollution has contributed, over many decades, to the contaminants present in site soil, sediment, surface water and groundwater. Diffuse anthropogenic pollution is typically defined as pollution emanating from a variety of sources including automobile exhaust and industrial smokestacks. The primary contaminants of concern associated with these types of sources are lead and PAH compounds, but it is not unusual to encounter other types of contaminants associated with sustained urban activity. Regulatory agencies have indicated that most areas are likely to have been impacted, to some degree, by anthropogenic activity, but recognize that the greatest impacts are to those sites located in urban areas such as the subject site. However, the SI also revealed the presence of two issues (the presence of several potential petroleumimpacted areas and the potential presence of USTs), which required additional investigation or delineation prior to the redevelopment of Site 2A/2B. Subsequently, the RI successfully delineated the extent of potential petroleum issues at all of the identified locations. The RI, in conjunction with other field efforts, has resolved UST related issues at the four potential UST areas (UST1, UST3, UST4 and UST7). As described in Section 11, investigations at UST 7 revealed two tanks, which had been closed in place by P&G. These tanks were removed as part of site redevelopment efforts.



Based on the information gained through the RI and the intended future usage of Site 2A/2B, hot spot excavation was identified as the appropriate remedial action to address petroleum-impacted soil at the following areas: Area B-3/B-2, Area GW-14, Area OP-1, Area A-5 and Area FS-1B. It was proposed to remove "delineated" hot spots of potentially petroleum-impacted soil at these locations in accordance with NYSDEC remedial procedures. Proposed excavation areas are presented in Figure 20. Further, it was proposed to collect samples from resultant excavation limits to confirm the success of the remedial efforts. Samples were to be analyzed for VOCs (8270) and PAH compounds (8260) based on the findings from prior sampling efforts. A summary of the actions undertaken prior to entering the VCP Program are provided in Section 12.2.

With regard to other site contaminants including fill material, the SI/RI activities identified the presence of contaminants at Site 2A/2B, which are typical to urban sites in the New York Metropolitan region. Further, the presence of contaminants in soil does not appear to have adversely impacted groundwater quality at Site 2A/2B. Overall, industrial/commercial usage such as the Port Authority's planned usage of the site as an intermodal facility and container terminal is not inconsistent with the levels of contamination noted to be present in site soil and groundwater. In fact, it is anticipated that the Port Authority's redevelopment of the site will have a positive impact on site environmental quality. In particular, the Port Authority intends to install material such as pavement and other semi-impervious material, which will function as an environmental cap throughout the entire site. This action will tend to stabilize contaminants in soil to leach from the unsaturated zone to groundwater. In addition, the placement of such materials will safeguard the public by preventing exposure to contaminants in soil and groundwater. Additional information pertaining to development actions is provided in Section 13.

12.2 Completed Remedial Actions

5 P P

To accommodate site redevelopment efforts, hot spot excavation was performed at locations within Site 2A. No remedial efforts have been proposed or performed at Site 2B. A summary of the excavation and sampling at 2A is presented by AOC in the following Sections. Excavation and sampling were performed in accordance with NYSDEC protocols. Continuous field screening, utilizing a photoionization detector (PID) was performed through excavation and sampling efforts. The limits of the hot spot excavation areas and the locations of samples are provided on Figure 21 and a summary of sampling is presented in Table 12.

252



.

--



• -

AGEF	2		
		Appro	ved
°TM	ENT		
9, 10 (1,10), 90(1,10), 10 (1,10) 10 (1,10), 10 (1,10), 10 (1,10) 10 (1,10),			
.01 , L(AND)	
(2B (28) (28)	ΓΑ (ΑΤΙ 2ΜΔ		N
AT \C1	ION FION	S, I	
ditio gns ort ts w	ns in and Autho ritten	contr meth ority cons	ods and ent.
ý	Che	cked	by
E	21		





5

Table 12 Summary of Remedial Actions and Sampling Site 2A/2B: HHMT- Port Ivory Facility

Initial SI AOC	SI Soil Boring Location	Description of Issues	Description of Actions and Sampling	Analytical Parameters
Area A	A-5	The RI delineated potential petroleum- impacted soil surrounding the borings located in Area A.	The delineated area consisting of Area A was excavated to the groundwater table to address potential petroleum impacted soil. Although not located on Site 1, the excavations for A-2 and A-5 were combined based upon field observations. Soil samples were collected from the 0.0-0.5 foot interval above the groundwater table, approximately 3.0-3.5 feet bgs.	VOC 8260; BN 8270
			The excavation measured approximately 170 feet by 147 feet. Additional excavation, measuring approximately 68 feet by 32 feet, was performed off the northwest corner to address visual signs of potential petroleum impacts. Eight confirmation samples were collected from the A-5 excavation: A5-1, A5-2, A5-3, A5-4, A5-5, A5-6, A5-7, and A5-8. All samples were submitted for laboratory analysis. Excavated soil was stockpiled on site pending off-site disposal.	
			The majority of this excavation is located on Site 1, with only a small area at Site 2A.	
Area B	B-2, B-3, GW-14	The RI delineated potential petroleum- impacted soil surrounding the borings located in Area B including sheen noted on groundwater at GW- 14.	The delineated area consisting of Area B including B-2 and B-3 and Area GW- 14 was excavated to the groundwater table to address potential petroleum impacted soil. Soil samples were collected from the 0.0-0.5 foot interval above the groundwater table. The excavated soil was stockpiled on site pending off- site disposal. The B-2/B-3 excavations joined the GW-14 excavation with resultant sampling utilized the GW-14 designation. The Area GW-14 excavation extended approximately 305 feet in length (north to south) and 110 feet in width (east to west). The excavation was extended to a depth of approximately 3.5 feet; the excavation activities encountered groundwater at some locations.	VOC 8260; BN 8270



Table 12 Summary of Remedial Actions and Sampling Site 2A/2B: HHMT- Port Ivory Facility

Initial SI AOC	SI Soil Boring Location	Description of Issues	Description of Actions and Sampling	Analytical Parameters
			is located on Site 2A, while one-quarter is located in Site 1.	
OP-1	OP-1	A sheen was noted on the groundwater surface of OP-1	The Area OP-1 excavation extended approximately 30 feet in length and 30 feet in width and was extended 3.5 feet in depth. No readings above background were recorded on the PID. The Area OP-1 excavation was located to the south of the Area GW-14 excavation, and situated entirely within Site 2A. All visually impacted soils were removed from Area OP-1. No groundwater was encountered during excavation activities. Four soil samples were collected from the sidewalls of the excavation at 3-3.5 feet bgs. Excavated soil was stockpiled onsite pending off-site disposal.	VOC 8260; BN 8270
Former Structures – FS-1B	FS-1B	The RI delineated potential petroleum- impacted soil surrounding the FS-1B location.	The delineated area surrounding FS-1B was excavated to the groundwater table to address potential petroleum impacted soil. Soil samples were collected from the 0.0-0.5 foot interval above the groundwater table. The excavation measured 100 feet by 83 feet. Eight confirmation soil samples were collected: FS1-1, FS1-2, FS1-3, FS1-4, FS1-5, FS1-6, FS1-7, and FS1-8. Soil samples were taken at the base of the sidewalls at the 0.0-0.5 foot interval above the groundwater table (approximately 4.5-5' bgs). Excavated soil was stockpiled onsite pending off-site disposal. The majority of this excavation is located on Site 1, with only a small area at Site 2A.	VOC 8260; BN 8270

12.2.1 Area B-3/B-2

The Area B-3/B-2 excavation was extended to address visual indications of petroleum impacts resulting in the joining of the Area B-3/B-2 excavation and the Area GW-14 excavation. As such, the excavation and sampling for Area B-3/B-2 are described in detail under Area GW-14 in Section 12.2.2 below.

12.2.2 Area GW-14

The Area GW-14 excavation extended approximately 305 feet in length (north to south) and 110 feet in width (east to west). The excavation was extended to a depth of approximately 3.5 feet; the excavation activities encountered groundwater at some locations. Approximately three-quarters of the excavation is located on Site 2A, while one-quarter is located in Site 1. Visually impacted soils located from within the limits of the excavation ranged from cinder and ash fill, red clays, silts and sands. PID readings were continuously recorded and ranged from not detected to 1500 parts per million (ppm). No measurable free product was observed to be present or to form on groundwater, where present.

During the removal of soil, piping was noted extending north to south along the eastern portion of the excavation. All piping was removed from the excavation. Based on field observations and historical site maps, it appears that the piping was associated with a former storm sewer line. Additional piping was uncovered in the northern corner of the excavation. The piping was traced and noted to extend to the north. The expansion of the excavation revealed the presence of a UST measuring 4 feet wide by 8 feet long by 6 feet in diameter. Based on historical information, it appeared that the UST was utilized as part of a former oil/water separator system. The UST appeared intact and additional efforts were undertaken to inspect and removed the vessel. Inspection of the tank and the underlying soil did not reveal the presence of residual materials or visually impacted soils. Field screening did not reveal the any readings above background. Due to the presence of the UST, the excavation was expanded in an easterly direction. The extension revealed the presence of three concrete tubs. The tubs were connected with piping and appeared to be part of the oil/water separators system. The system was removed from the excavation for off-site disposal.

Twelve soil samples were collected from the sidewalls of the excavation at the soil/ground water interface (3-3.5 feet bgs). All soil samples were analyzed for PAH Compounds (8260) and VO Compounds (8270). VOCs were either not detected or were detected at concentrations below corresponding RSCOs. No

257

X

Table 13A Soil Analytical Results Area GW-14 Volatile Organic Compounds Site 2A/2B - HHMT-Port Ivory Facility

Location	Recommended	GW14-1	GW14-2	GW14-3	GW14-4	GW14-5	GW14-6
Sample Date	Soil Cleanup	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003
Sample Depth (ft)	Objective	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5
Units	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
1,3,5-trimethlybenzene	NS	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
4-isopropyltoluene	NS	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0024
Benzene	0.06	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
Ethylbenzene	5.5	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
Isopropylbenzene	NS	0.0014U	0.0013U	0.0048	0.0012U	0.0012U	0.0013U
M&p-Xylenes	1.2*	0.0028U	0.0026U	0.0024U	0.0025U	0.0024U	0.0025U
Methyl-t-butyl ether	NS	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
Naphthalene	13	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
N-butylbenzene	NS	0.0014U	0.0013U	0.0013	0.0012U	0.0012U	0.0013U
N-Propylbenzene	NS	0.0014U	0.0013U	0.0047	0.0012U	0.0012U	0.0013U
O-Xylene	1.2*	0.0014U	0.0013U	0.0012U	0.0012U	0.0012U	0.0013U
Sec-butylbenzene	NS	0.0014U	0.0013U	0.0046	0.0012U	0.0012U	0.0013U
T-butylbenzene	NS	0.0014U	0.0013U	0.0049	0.0012U	0.0012U	0.0013U
Toluene	1.5	0.011	0.0084	0.011	0.0084	0.0076	0.0099
Total VOCs	10	0.011	0.0084	0.0313	0.0084	0.0076	0.0123

U Undetectable Levels

ND Not Detected

NS No Standard

* Total Xylene Recommended Cleanup Standard

258

Table 13A Soil Analytical Results Area GW-14 Volatile Organic Compounds Site 2A/2B - HHMT-Port Ivory Facility

Location	Recommended	GW14-7	GW14-8	GW14-9	GW14-10	GW14-11	GW14-12
Sample Date	Soil Cleanup	4/9/2003	4/9/2003	4/24/2003	4/24/2003	4/24/2003	4/24/2003
Sample Depth (ft)	Objective	3-3.5	3-3.5	3-3.5	.3-3.5	3-3.5	3-3.5
Units	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
1,3,5-trimethlybenzene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
4-isopropyltoluene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0088	0.0086
Benzene	0.06	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
Ethylbenzene	5.5	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
Isopropylbenzene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
M&p-Xylenes	1.2*	0.0029U	0.0025U	0.0028U	0.0022U	0.0022U	0.0022U
Methyl-t-butyl ether	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
Naphthalene	13	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
N-butylbenzene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
N-Propylbenzene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
O-Xylene	1.2*	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
Sec-butylbenzene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
T-butylbenzene	NS	0.0014U	0.0012U	0.0014U	0.0011U	0.0011U	0.0011U
Toluene	1.5	0.015	0.0054	0.0014U	0.0011U	0.0011U	0.0011U
Total VOCs	10	0.015	0.0054	ND	ND	0.0088	0.0086

U Undetectable Levels

ND Not Detected

NS No Standard

* Total Xylene Recommended Cleanup Standard

Table 13BSoil Analytical ResultsArea GW-14PAH CompoundsSite 2A/2B - HHMT-Port Ivory, Facility

Location	Recommended	GW14-1	GW14-2	GW14-3	GW14-4	GW14-5	GW14-6	GW14-7	GW14-8	GW14-9	GW14-10	GW14-11	GW14-12
Sample Date	Soil Cleanup	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003 -	4/24/2003	4/24/2003	4/24/2003	4/24/2003
Sample Depth (ft)	Objective	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5
Units	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Acenaphthene	41	0.69U	0.64U	0.61U	0.62U	0.60U	0.63U	0.096J	0.14J	0.69U	0.14J	0.088J	0.079J
Anthracene	50	0.69U	0.64U	0.61U	0.11J	0.60U	0.63U	0.71U	0.62U	0.69U	0.15J	0.062J	0.12J
Benzo(a)anthracene	0.224 or MDL	0.69U	0.64U	0.086J	0.16J	0.60U	0.082J	0.084J	0.27J	0.69U	0.26J	0.13J	0.27J
Benzo(a)pyrene	0.061 or MDL	0.69U	0.64U	0.062J	0.11J	0.60U	0.63U	0.71U	0.23J	0.69U	0.17J	0.10J	0.24J
Benzo(b)fluoranthene	1.1	0.69U	0.64U	0.14J	0.16J	0.60U	0.084J	0.18J	0.53J	0.69U	0.36J	0.18J	0.35J
Benzo(g,h,i)perylene	50	0.69U	0.64U	0.61U	0.62U	0.60U	0.63U	0.71U	0.093J	0.69U	0.56U	0.54J	0.081J
Benzo(k)fluoranthene	1.1	0.69U	0.64U	0.61U	0.070J	0.60U	0.63U	0.71U	0.14J	0.69U	0.12J	0.54J	0.12J
Chrysene	0.4	0.69U	0.64U	0.11J	0.18J	0.60U	0.075J	0.12J	0.37J	0.69U	0.31J	0.14J	0.29J
Dibenzo(a,h)Anthracene	0.014 or MDL	0.69U	0.64U	0.61U	0.62U	0.60U	0.63U	0.71U	0.62U.	0.69U	0.56U	0.54U	0.56U
Fluoranthene	50	0.69U	0.64U	0.20J	0.35J	0.60U	0.21J	2.4	8.7	0.15J	0.91	0.37J	0.86
Fluorene	50	0.69U	0.64U	0.61U	0.080J	0.60U	0.63U	0.71U	0.62U	0.69U	0.12J	0.071J	0.098J
Indeno(1,2,3-cd)pyrene	3.2	0.69U	0.64U	0.61U	0.6 2 U	0.60U	0. 63 U	0.71U	0.099J	0.69U	0.56U	0.54U	0.079J
Napthalene	13	0.074J	0.64U	0.61U	0.62U	0.60U	0.63U	0.16J	0.15J	0.095J	0.20J	0.083J	0.089J
Phenanthrene	50	0.69U	0.64U	0.15J	0.19J	0.60U	0.19J	1.5	0.62U	0.16J	0.79	0.24J	0.76
Pyrene	50	0.69U	0.64U	0.18J	0.30J	0.60U	0.17J	0.22J	0.84	0.10J	0.61	0.33J	0.55J
Total PAH Compounds	500	0.074	ND	0.928	1.71	ND	0.811	4.76	11.562	0.505	4.14	2.874	3.986

U Undetectable Levels

ND Not Detected

MDL Method Detection Limit

-

Table 14ASoil Analytical ResultsArea OP-1Volatile Organic CompoundsSite 2A/2B - HHMT - Port Ivory Facility

Location	Recommended	OP1-1	OP1-2	OP1-3	OP1-4
Sample Date	Soil Cleanup	4/24/2003	4/24/2003	4/24/2003	4/24/2003
Sample Depth (ft)	Objective	3-3.5	3-3.5	3-3.5	3-3.5
Units	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0013U	0.0011U	0.0012U	0.0011U
1,3,5-trimethlybenzene	NS	0.0013U	0.0011U	0.0012U	0.0011U
4-isopropyltoluene	NS	0.0013U	0.0011U	0.0012U	0.0011U
Benzene	0.06	0.0013U	0.0011U	0.0012U	0.0011U
Ethylbenzene	5.5	0.0013U	0.0011U	0.0012U	0.0011U
Isopropylbenzene	NS	0.0013U	0.0011U	0.0012U	0.0011U
M&p-Xylenes	1.2*	0.0025U	0.0023U	0.0023U	0.0022U
Methyl-t-butyl ether	NS	0.0013U	0.0011U	0.0012U	0.0011U
Naphthalene	13	0.0013U	0.0011U	0.0012U	0.0011U
N-butylbenzene	NS	0.0013U	0.0011U	0.0012U	0.0011U
N-Propylbenzene	NS	0.0013U	0.0011U	0.0012U	0.0011U
O-Xylene	1.2*	0.0013U	0.0011U	0.0012U	0.0011U
Sec-butylbenzene	NS	0.0013U	0.0011U	0.0012U	0.0011U
T-butylbenzene	NS	0.0013U	0.0011U	0.0012U	0.0011U
Toluene	1.5	0.0013U	0.0011U	0.0012U	0.0011U
Total VOCs	10	ND	ND	ND	ND

U Undetectable Levels

NS No Standard

ND Not Detected

* Total Xylene Recommended Cleanup Standard

Table 14BSoil Analytical ResultsArea OP-1PAH CompoundsSite 2A/2B - HHMT- Port Ivory Facility

Location	Recommended	OP1-1	OP1-2	OP1-3	OP1-4
Sample Date	Soil Cleanup	4/24/2003	4/24/2003	4/24/2003	4/24/2003
Sample Depth (ft)	Objective	3-3.5	3-3.5	3-3.5	3-3.5
Units	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Acenaphthene	41	0.63U	0.57U	0.58U	0.55U
Anthracene	50	0.63U	0.57U	0.58U	0.55U
Benzo(a)anthracene	0.224 or MDL	0.072J	0.57U	0.58U	0.55U
Benzo(a)pyrene	0.061 or MDL	0.63U	0.57U	0.58U	0.55U
Benzo(b)fluoranthene	1.1	0.63U	0.57U	0.58U	0.071J
Benzo(g,h,i)perylene	50	0.63U	0.57U	0.58U	0.55U
Benzo(k)fluoranthene	1.1	0.63U	0.57U	0.58U	0.55U
Chrysene	0.4	0.071J	0.57U	0.58U	0.060J
Dibenzo(a,h)Anthracene	0.014 or MDL	0.63U	0.57U	0.58U	0.55U
Fluoranthene	50	0.12J	0.57U	0.058J	0.18J
Fluorene	50	0.63U	0.57U	0.58U	0.55U
Indeno(1,2,3-cd)pyrene	3.2	0.63U	0.57U	0.58U	0.55U
Napthalene	13	0.064J	0.57U	0.58U	0.55U
Phenanthrene	50	0.088J	0.57U	0.58U	0.11J
Pyrene	50	0.14J	0.57U	0.58U	0.10J
Total PAH Compounds	500	0.555	ND	0.058	0.521

U Undetectable Levels

ND Not Detected

~578

MDL Method Detection Limit



samples exceeded the RSCO of 10 mg/kg for total VOCs. Only two PAH Compounds, LOW PAH benzo(a)anthracene and benzo(a)pyrene, were detected above corresponding RSCOs. Benzo(a)anthracene was detected at concentrations in excess of its RSCO in three samples ranging from 0.26 mg/kg in sample GW14-10 to 0.27 mg/kg in samples GW14-8 and GW14-12. Benzo(a)pyrene was detected at concentrations in excess of its RSCO in six samples ranging from 0.062 mg/kg in sample GW14-3 to 0.24 mg/kg in sample GW14-12. None of the samples were noted to exceed 50 mg/kg guidance for individual PAH Compounds or the 500 mg/kg guidance criteria for total PAH compounds. Please refer to Tables 13A and 13B for a summary of all analytical results.

12.2.3 Area OP-1

The Area OP-1 excavation extended approximately 30 feet in length and 30 feet in width and was extended 3.5 feet in depth. No readings above background were recorded on the PID. The Area OP-1 excavation was located to the south of the Area GW-14 excavation, and situated entirely within Site 2A. All visually impacted soils were removed from Area OP-1. No groundwater was encountered during excavation activities. Four soil samples were collected from the sidewalls of the excavation at 3-3.5 feet bgs. All samples were analyzed for PAH Compounds (8260) and VOCs (8270).

No VOCs were detected in the four samples collected from the excavation. All PAH compounds were detected at concentrations below corresponding RSCOs and no sample exceeded the 500 mg/kg guidance threshold for total PAH compounds. Please refer to Tables 14A and 14B for a summary of analytical results.

12.2.4 Area A-5

The Area A-5 excavation measured approximately 170 feet in length (east to west) and 150 feet in width (north to south) and extended 3.5 feet in depth. No readings above background were recorded on the PID. The northeastern corner of the Area A-5 excavation overlaps the southwestern excavation of the Area GW-14 (Area B-3/B-2) excavation. Approximately one-quarter of the Area A-5 excavation is situated within the limits of Site 2A, the majority of the excavation is located on Site 1. Access to the underlying soils was possible after the removal of railroad tracks and concrete slabs. Visually impacted soils located from within the limits of the excavation consisted of a mix of cinder, ash, lime sludge/by-product fill, fine black sand and tan sand. The depth of the excavation was limited by the presence of groundwater (3.5 feet



bgs) as well as the presence of lime sludge/by-product fill and numerous tree trunks (4 feet to 15 feet bgs).

Eight soil samples were collected from the interface of the sidewalls/ground water table. All samples were analyzed for PAH Compounds (8260) and VOCs (8270). No VOCs were detected at concentrations above corresponding RSCOs and no sample exceeded the total VOC guidance threshold of 10 mg/kg. PAH compounds were either not detected or detected at concentrations below the RSCO in all but one sample. Benzo(a)pyrene was detected at 0.14 mg/kg in Sample A5-5. None of the samples collected exceeded the guidance threshold of 500 mg/kg for total PAH compounds. Please refer to Tables 15A and 15B for a summary of the analytical results.

12.2.5 FS-1B Area

The Area FS-1B measured approximately 100 feet in length (east to west) and 83 feet in width (north to south) and extended approximately 5.0 feet in depth, just above the groundwater table. No readings above background were recorded on the PID. The Area FS-1B excavation is located to the southwest of the Area GW-14 excavation. Approximately one-quarter of the excavation is located on Site 2A, the majority of the excavation is located on Site 1. Visually impacted soils located from within the limits of the excavation consisted of a mix of cinder, ash, lime sludge/by-product fill material, sand, tree timbers and concrete sections.

During the soil removal effort, a concrete structure was encountered at the southeast corner. No visual indications of contaminants were noted and no readings above background were recorded on the PID. The concrete structure was removed from the excavation for off-site disposal along with other concrete demolition debris.

Eight soil samples were collected from the excavation; two from each sidewall and submitted for PAH Compound (8260) and VOC (8270) analyses. VOCs were either not detected or were detected at concentrations <u>below</u> the corresponding RSCO. None of the samples exceeded the guidance threshold of 10 mg/kg for total VOCs. Four PAH compounds were detected at concentrations in <u>excess of</u> corresponding RSCOs in several samples collected from Area FS-1B excavation. Benzo(a)anthracene was detected in excess of its RSCO in three samples with concentrations ranging from 0.27 mg/kg in sample FS1-1 to 2.2 mg/kg in sample FS1-4. Benzo(a)pyrene was detected in excess of its RSCO in six samples ranging in concentrations from 0.18 mg/kg in sample FS1-8 to 1.7 mg/kg in sample FS1-4.

Table 15A Soil Analytical Results Area A-5 Volatile Organic Compounds

Site 2A/2B - HHMT-Port Ivory Facility

Location		A5-1	A5-2	A5-3	A5-4	A5-5	A5-6	A5-7	A5-8
Sample Date	Recommended	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003
Sample Depth (ft)	Soil Cleanup	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5
Units	Objective mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
1,3,5-trimethlybenzene	NS	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
4-isopropyltoluene	NS	0.0039	0.0046	0.0035	0.012	0.0084	0.0013U	0.0014U	0.0012U
Benzene	0.06	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
Ethylbenzene	5.5	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
Isopropylbenzene	NS	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
M&P-Xylenes	1.2*	0.0031U	0.0027U	0.0027U	0.0029U	0.0025U	0.0026U	0.0027U	0.0025U
Methyl-t-butyl ether	NS	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
Naphthalene	13	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
N-butylbenzene	NS	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
N-Propylbenzene	NS	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
O-Xylene	1.2*	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
Sec-butylbenzene	NS	0.0015U	0.0014U	0.0013U	0.0014U	0.0012U	0.0013U	0.0014U	0.0012U
Toluene	1.5	0.033	0.014	0.012	0.0065	0.021	0.031	0.018	0.02
Total VOCs	10	0.0369	0.0186	0.0155	0.0185	0.0294	0.031	0.018	0.02

U Undetectable Levels

NS No Standard

* Total Xylene Recommended Cleanup Standard

Table 15BSoil Analytical ResultsArea A-5PAH CompoundsSite 2A/2B - HHMT- Port Ivory Facility

Location		A5-1	A5-2	A5-3	A5-4	A5-5	A5-6	A5-7	A5-8
Sample Date	Recommended	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003	4/9/2003
Sample Depth (ft)	Soil Cleanup	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5	3-3.5
Units	Objective mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Acenaphthene	41	0.77U	0.68U	0.67U	0 .71U	0.40U	0.66U	0.68U	0.62U
Anthracene	50	0.77U	0.68Ü	0.67U	0.71U	0.043J	0.66U	0.68U	0.62U
Benzo(a)anthracene	0.224 or MDL	0.77U	0.68U	0.67U	0.076J	0.15J	0.66U	0.68U	0.62U
Benzo(a)pyrene	0.061 or MDL	0.77U	0.68U	0.67U	0.71U	0.14J	0.66U	0.68U	0.62U
Benzo(b)fluoranthene	1.1	0.77U	0.68U	0.67U	0.71U	0.22J	0.66U	0.68U	0.62U
Benzo(g,h,i)perylene	50	0.77U	0.68U	0.67U	0.71U	0.40U	0.66U	0.68U	0.62U
Benzo(k)fluoranthene	1.1	0.77U	0.68U	0.67U	0.71U	0.083J	0.66U	0.68U	0.62U
Chrysene	0.4	0.77U	0.68U	0.67U	0.087J	0.18J	0.075J	0.68U	0.62U
Dibenzo(a,h)Anthracene	0.014 or MDL	0.77U	0.68U	0.67U	0.71U	0.40U	0.66U	0.68U	0.62U
Fluoranthene	50	0.77U	0.68U	0.67U	0.13J	0.26J	0.19J	0.14J	0.13J
Fluorene	50	0.77U	0.68U	0.67U	0.71U	0.40U	0.66U	0.68U	0.62U
Indeno(1,2,3-cd)pyrene	3.2	0.77U	0.68U	0.67U	0.71U	0.40U	0.66U	0.68U	0.62U
Napthalene	13	0.83	0.68U	0.67U	0.091J	0.070J	0.66U	0.68U	0.62U
Phenanthrene	50	0.77U	0.68U	0.67U	0.12J	0.20J	0.14J	0.68U	0.62U
Pyrene	50	0.77U	0.68U	0.67U	0.12J	0.30J	0.14J	0.10J	0.093J
Total PAH Compounds	500	0.83	ND	ND	0.624	1.646	0.545	0.24	0.223

U Undetectable Levels

ND Not Detected

MDL Method Detection Limit

266

Benzo(b)flouranthene was detected in excess of its RSCO in one sample, FS1-4 at 2.0 mg/kg. Chrysene was detected in excess of its RSCO in three samples ranging in concentrations from 0.41 mg/kg in sample FS1-1 to 2.3 mg/kg in sample FS1-4. None of the samples exhibited concentrations in excess of the guidance threshold of 500 mg/kg for total PAH Compounds. Please refer to Tables 16A and 16B for a summary of the analytical results.

13.0 SUMMARY OF PROPOSED SITE DEVELOPMENT ACTIONS

Hatch Mott

MacDonald

The Port Authority is currently redeveloping Site 2A/2B for use as an intermodal facility, which will function as part of the larger container terminal/intermodal facility including the entire HHMT-Port Ivory Facility. The findings from the assessment/investigation actions have revealed that the Port Authority's planned usage of the site as an intermodal facility and container terminal is not inconsistent with the levels of contamination noted to be present in site soil and groundwater and that contamination can be addressed through site redevelopment.

The Port Authority had developed Predminary Site Plans for the proposed redevelopment of Sites 1 and 2A/2B. Please refer to Appendix D for information related to site development including a Preliminary Site Plan dated January 2003 and a Preliminary Site Plan with Phasing, also dated January 2003. As shown on these plans, the majority of Site 2A (as well as Site 1) will be dedicated to a railway system which will continue through the southern end of Site 2A and onto Site 2B. Although the elements proposed for Site 2B are not depicted in detail on the Site Plan Figures included in Appendix D, the continuation of the rail system is illustrated on a schematic drawing designated as SK16 and dated October 13, 2003; drawing SK16 is also provided in Appendix D. At this time, it is anticipated that two buildings (Buildings 40 and 41) will remain at Site 2A and will be utilized for security and offices associated with the operation of the facility. To address structural issues presented by the presence of fill material, the Port Authority's development plan includes a process of surcharging portions of Site 1 and Site 2A/2B, with geotechnically suitable clean fill, to achieve a stable base for future construction. Figure SK1, Sequencing of Surcharge Areas along with an associated schedule, is provided in Appendix D.

As part of the geotechnical site preparation work, the Port Authority performed a surcharge pilot study at an area of Site 1 in 2002/2203. The study included the systematic placement of soil/fill over an area

267

Х

Table 16ASoil Analytical ResultsArea FS-1BVolatile Organic CompoundsSite 2A/2B - HHMT- Port Ivory Facility

Location		FS1-1	FS1-2	FS1-3	FS1-4	FS1-5	FS1-6	FS1-7	FS1-8
Sample Date	Recommended	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002
Sample Depth (ft)	Soil Cleanup	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0
Units	Objective mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
1,2,4-trimethylbenzene	3.4	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
1,3,5-trimethlybenzene	NS	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
4-isopropyltoluene	NS	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0022	0.0020U	0.0016U
Benzene	0.06	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
Ethylbenzene	5.5	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
Isopropylbenzene	NS	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
M&p-Xylenes	1.2*	0.0012U	0.0028U	0.0032U	0.0039U	0.0035U	0.0025U	0.0040U	0.0032U
Methyl-t-butyl ether	NS	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
Naphthalene	13	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
N-butylbenzene	NS	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
N-Propylbenzene	NS	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
O-Xylene	1.2*	0.0012U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
Sec-butylbenzene	NS	0.0012U	0.0014U	0.001 6 U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
T-Butlybenzene	NS	0.001 2 U	0.0014U	0.0016U	0.0020U	0.0018U	0.0013U	0.0020U	0.0016U
Toluene	1.5	0.0031	0.0037	0.001 6 U	0.0020U	0.0028	0.0025	0.0023	0.0016U
Total VOCs	10	0.0031	0.0037	ND	ND	0.0028	0.0047	0.0023	ND

U Undetectable Levels

ND Not Detected

~**

NS No Standard

* Total Xylene Recommended Cleanup Standard

- 268

Table 16BSoil Analytical ResultsArea FS-1BPAH CompoundsSite 2A/2B - HHMT-Port Ivory Facility

Location	Recommended	FS1-1	FS1-2	FS1-3	FS1-4	FS1-5	FS1-6	FS1-7	FS1-8
Sample Date	Soil Cleanup	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002	12/3/2002
Sample Depth (ft)	Objective	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0	4.5-5.0
Units	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg	mg/kg
Acenaphthene	41	0.045J	0.47U	0.52U	2.0U	0.58U	0.42U	0.27J	0.54U
Anthracene	50	0.12J	0.47U	0.52U	2.0U	0.58U	0.42U	0.67	0.063J
Benzo(a)anthracene	0.224 or MDL	0.27J	0.47U	0.1J	2.2	0.19J	0.071J	1.4	0.22J
Benzo(a)pyrene	0.061 or MDL	0.36J	0.47U	0.21J	1.7J	0.24J	0.059J	- 1	0.18J
Benzo(b)fluoranthene	1.1	0.6	0.47U	0.36J	2	0.29J	0.10J	0.96	0.31J
Benzo(g,h,i)perylene	50	0.13J	0.47U	0.52U	0.69J	0.58U	0.42U	0.2J	0.54U
Benzo(k)fluoranthene	1.1	0.27J	0.47U	0.087J	0.52J	0.069J	0.42U	0.43J	0.54U
Chrysene	0.4	0.41J	0.14J	0.15J	2.3	0.35J	0.092J	1.5	0.32J
Dibenzo(a,h)Anthracene	0.014 or MDL	0.42U	0.47U	0.52U	2.0U	0.58U	0.42U	0.67U	0.54U
Fluoranthene	50	0.66	0.47U	0.17J	0.82J	0.24J	0.11J	1.9	0.28J
Fluorene	50	0.055J	0.47U	0.52U	2.0U	0.58U	0.42U	0.29J	0.54U
Indeno(1,2,3-cd)pyrene	3.2	0.13J	0.47U	0.52U	0.55J	0.58U	0.42U	0.18J	0.54U
Napthalene	13	0.14J	0.16J	0.082J	0.52J	0.58U	0.045J	2.2	0.19J
Phenanthrene	50	0.54	0.19J	0.16J	0.66J	0.20J	0.089J	3.5	0.35J
Pyrene	50	0.81	0.47U	0.17J	2	0.35J	0.11J	3.3	0.41J
Total PAH Compounds	500	4.54	0.49	1.489	13.96	1.929	1.516	17.8	2.323

U Undetectable Levels

MDL Method Detection Limit

269



study, the Port Authority reviewed potential environmental impacts to groundwater and Bridge Creek. The environmental review for groundwater included the installation and sampling of nested monitoring well pairs (one shallow and one deep well) at four locations around the pilot study area. The wells were constructed approximately 15 feet from the edge of the surcharge material pile. Groundwater samples were collected from the eight newly installed wells as well as from two additional existing well locations, PG-CS-7 and PG-RS-1. Three rounds of groundwater sampling were performed on the two existing wells with all rounds including phenols and BTEX (benzene, toluene, ethylbenzene and xylenes) and one round including phenols, BTEX, TAL Metals and pH. One round of groundwater sampling was performed for the eight newly installed wells with analysis for TAL Metals and pH. A summary of analytical results is provided in Tables 17A (BTEX), 17B (phenols), and 17C (TAL Metals and pH). The pilot study area and well locations are presented on Figure 22 In addition, exceedences of applicable groundwater SVGs are also presented, by location, on Figure 22. A summary of analytical results is presented in Tables 17A-17C.

With respect to Bridge Creek, the environmental review included the inspection of the eastern bank of Bridge Creek for the presence of seeps, precipitate, bank failure or other evidence of mass movement of subsurface material or liquids. It should be noted that the surface water and sediment sampling undertaken as part of the RI of Site 2A/2B was performed during the surcharge pilot test period. As such, the information generated from that evaluation has been utilized as part of the environmental review for the pilot study. Information related to sampling of Bridge Creek is presented in Sections 8.3 and 9.3 of this Report. In addition, two rounds of surface water sampling were performed as part of the surcharge pilot study. Five samples were collected from representative locations during both rounds and samples were analyzed for TAL Metals and pH./The surface water locations and associated analytical results are presented on Figure 22 and Table 17C. Given the purpose of the study, the surface water samples were compared to applicable groundwater SVGs.

Overall, the environmental evaluation performed as part of the pilot study did not reveal any adverse impacts as a result of the compaction process. Inspection of the eastern bank did not reveal the presence of seeps, additional/increased occurrence of precipitate, bank failure or other evidence of mass movement of subsurface material or liquids. Analytical results from the surface water and the groundwater sampling did not reveal any increased contaminant concentrations over the period of study. It should be noted that m&p xylenes (reported as a combined concentration) was/were detected in the groundwater sample from







e 17A Surcharge Pilot Study -Groundwater Results BTEX Site 2A/2B: HHMT - Port Ivory Facility

Location			CS-7	RS-1	FB-1	TB-1	CS-7	RS-1	TB-1	FB-1	TB-1	CS-7	RS-1
Date	Recommended Groundwater	Groundwater	10/15/2002	10/15/2002	10/15/2002	10/15/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/7/2002	12/2/2002	12/2/2002
LAB ID#	Cleanup Standard	Cleanup Guidance	AB70453	AB70455	AB70457	AB70459	AB72292	AB72294	AB72304	AB72305	AB72397	AB74079	AB74081
Concentration	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L
Benzene	1	NG	0.49U	0.49U	0.49U	0.49U	0.49U	0.49U	0.49U	0.49U	0.49U	0.49U	0.49U
Ethylbenzene	· 5	NG	0.63U	0.63U	0.63U	0.63U	0.63U	0.63U	0.63U	0.63U	0.63U	2.4	0.63U
m&p-xylenes	5&5	NG	1.1U	1.1U	1.1U	1.1U	1.10	1.10	1.1U	1.1U	1.10	7	1.10
O-xylenes	5	NG	0.59U	0.59U	0.59U	0.59U	0.59U	0.59U	0.59U	0.59U	0.59U	0.59U	0.59U
Toluene	5	NG	0.79U	0.79U	0.79U	0.79U	0.79U	0.79U	0.79U	0.79U	0.79U	0.79U	0.79U

U Undetectable Levels

NG No Guidance

Surcharge Pilot Study - Groundwater Results Acid Extractables Site 2A/2B: HHMT Port Ivory Facility

Location	Recommended	Decommonded	CS-7	RS-1	FB-1	CS-7
Date	Groundwater	Groundwater	10/15/2002	10/15/2002	10/15/2002	11/6/2002
Lab ID	Cleanup	Cleanup	AB70453	AB70455	AB70457	AB72292
Concentration	Standard UG/L	Guidance UG/L	UG/L	UG/L	UG/L	UG/L
2,4,5-trichlorophenol	NS	NG	0.6U	0.6U	0.6U	0.6U
2,4,6-trichlorophenol	NS	NG	0.65U	0.65U	0.65U	0.65U
2,4-dichlorophenol	5	NG	0.57U	0.57U	0.57U	0.57U
2,4-dimethylphenol	NS	50	0.49U	0.49U	0.49U	0.49U
2,4-dinitrophenol	NS	10	3.1U	3.1U	3.1U	3.1U
2-chlorophenol	NS	NG	0.49U	0.49U	0.49U	0.49U
2-methylphenol	NS	NG	0.61U	0.61U	0.61U	0.61U
2-nitrophenol	NS	NG	0.64U	0.64U	0.64U	0.64U
3&4-methylphenol	NS	NG	5.4U	1.1J	5.4U	5.4U
4,6-dinitro-2-methylphenol	NS	NG	0.36U	0.36U	0.36U	0.36U
4-chloro-3-methylphenol	NS	NG	0.38U	0.38U	0.38U	0.38U
4-nitrophenol	NS	NG	0.27U	0.27U	0.27U	0.27U
pentachlorophenol	1 (total phenols)	NG	0.57U	0.57U	0.57U	0.57U
phenol	1 (total phenols)	NG	0.14U	0.14U	0.14U	0.14U
Total phenols	1	NG	ND	1.1	ND	ND

U Undetectable Levels

NS No Standard

w#.

NG No Guidance

ND Not Detected

1922

Table 17B Surcharge Pilot Study - Groundwater Results Acid Extractables Site 2A/2B: HHMT Port Ivory Facility

Location	Pagammandad	Decommonded	RS-1	FB-1	CS-7	RS-1
Date	Groundwater	Groundwater	11/6/2002	11/6/2002	12/2/2002	12/2/2002
Lab ID	Cleanup	Cleanup	АВ72294	AB72305	AB74079	AB74081
Concentration	Standard UG/L	Guidance UG/L	UG/L	UG/L	UG/L	UG/L
2,4,5-trichlorophenol	NS	NG	0.6U	0.6U	0.6U	0.6U
2,4,6-trichlorophenol	NS	NG	0.65U	0.65U	0.65U	0.65U
2,4-dichlorophenol	5	NG	0.57U	0.57U	0.57U	0.57U
2,4-dimethylphenol	NS	50	0.49U	0.49U	0.49U	0.49U
2,4-dinitrophenol	NS	10	3.1U	3.1U	3.1U	3.1U
2-chlorophenol	NS	NG	0.49U	0.49U	0.49U	0.49U
2-methylphenol	NS	NG	0.61U	0.61U	0.61U	0.61U
2-nitrophenol	NS	NG	0.64U	0.64U	0.64U	0.64U
3&4-methylphenol	NS	NG	5.4U	5.4U	5.4U	3.8J
4,6-dinitro-2-methylphenol	NS	NG	0.36U	0.36U	0.36U	0.36U
4-chloro-3-methylphenol	NS	NG	0.38U	0.38U	0.38U	0.38U
4-nitrophenol	NS	NG	0.27U	0.27U	0.27U	0.27U
pentachlorophenol	1 (total phenols)	NG	0.57U	0.57U	0.57U	0.57U
phenol	1 (total phenols)	NG	0.14U	0.14U	0.14U	3
Total phenols	1	NG	ND	ND	ND	6.8

U Undetectable Levels

NS No Standard

-999

NG No Guidance

ND Not Detected

Surcharge Pilot Study - Groundwater and Surface Water Results Metals and pH

Site 2A/2B:	HHMT - Port lvory	v Facility

					r	r	· · · · · · · · · · · · · · · · · · ·	T T		
Location	Recommended	Recommended	ST-SW1	ST-SW2	ST-SW3	ST-SW4	ST-SW5	CS-7	CS-7	RS-1
Date	Groundwater	Groundwater	10/21/2002	10/21/2002	10/21/2002	10/21/2002	10/21/2002	10/15/2002	10/15/2002	10/15/2002
Lab ID	Cleanup	Cleanup	AB70895	AB70896	AB70460	AB70897	AB70898	AB70453	AB70454	AB70455
Concentration	Standard	Guidance	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L
Filtered or Unfiltered	UG/L	UG/L						unfiltered	filtered	unfiltered
Aluminum	NS	NG	570	400	140	190	330	100U	100U	170
Antimony	3	NG	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U
Arsenic	25	NG	4.0U	5.4	4.0U	10	4.0U	4.0U	4.0U	4.1
Barium	1000	NG	27	25U	45	25U	25U	25U	25U	59
Beryllium	NS	3	4.0U	4.0U	4.0U	4.0U	4.0U	4.0U	4.0U	4.0U
Cadmium	5	NG	2.5	2.5	2.0U	2.5	2.5	2.0U	2.0U	2.0U
Calcium	NS	NG	150000	160000	93000	170000	160000	110000	120000	130000
Chromium	50	NG	25U	25 U	25U	25U	25U	25U	25U	36
Cobalt	NS	NG	10U	10U	10U	10U	10U	10U	10U	10U
Copper	200	NG	25U	25 U	25U	25U	25U	25U	25U	25U
Iron	300***	NG	610***	280***	790***	150U	150U	210	150U	20000***
Lead	25	NG	5.0U	5.0U	5.0U	5.0U	5.0U	5.0U	5.0U	5.0U
Magnesium	NS	35000	600000	630000	130000	610000	630000	47000	52000	27000
Manganese	300***	NG	82***	69***	260***	67	61	25U	25U	180***
Nickel	100	NG	25U	25U	25U	25U	25U	25U	25U	46
Potassium	NS	NG	250000	260000	58000	250000	260000	13000	15000	5800
Selenium	10	NG	25U	25U	25U	25U	25U	25U	25U	25U
Silver	50	NG	4800000	10U	10U	10U	10U	10U	10U	10U
Sodium	20000	NG	50000U	5100000	1300000	490000	5000000	79000	89000	29000
Thallium	NS	0.5	5.0U	5.0U	5.0U	5.0U	5.0U	5.0U	5.0U	5.0U
Vanadium	NS	NG	25U	25U	25U	25U	25U	25U	25U	25U
Zinc	NS	2000	47	34	25U	32	31	64	67	440
Mercury	0.7	NG	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U
pH	NS	NG	7.5	7.6	7.7	7.6	7.6	7	7	7.2

ND No Data

U Undetectable Levels

NS No Standard

NG No Guidance

*** Total for Iron and Manganese is > 500

Note-1: pH listed is the pH recorded in the field

Note-2: ST-SW1 through ST-SW5 represents samples

collected from surface water


Surcharge Pilot Study - Groundwater and Surface Water Results

Metals and pH

Site 2A/2B: HHMT - Port Ivory Facility

	·····	T		· · · · · · · · · · · · · · · · · · ·	· · · · · ·	Y	<u> </u>	· · · · · · · · · · · · · · · · ·	T		
Location	Recommended	Recommended	RS-1	FB-1	FB-1	ST-SW1	ST-SW2	ST-SW3	ST-SW4	ST-SW5	
Date	Groundwater	Groundwater	10/15/2002	10/15/2002	10/15/2002	11/7/2002	11/7/2002	11/7/2002	11/7/2002	11/7/2002	
Lab ID	Cleanup	Cleanup	AB70456	AB70457	AB70458	AB72569	AB72570	AB72571	AB72572	AB72573	
Concentration	Standard	Guidance	UG/L	UG/L	UG/L	ug/l	ug/l	ug/l	ug/l	ug/l	
Filtered or Unfiltered	UG/L	UG/L	filtered	unfiltered	filtered	-		_	-		
Aluminum	NS	NG	100U	100U	100U	430	420	340	550	290	
Antimony	3	NG	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	
Arsenic	25	NG	4.0U	4.0U	4.0U	4.0U	4.0U	4.1	5.4	4.0U	
Barium	1000	NG	52	25U	25U	28	25U	37	25U	25U	
Beryllium	NS	3	4.0U	4.0U	4.0U	4.0U	4.0U	4.0U	4.0U	4.0U	
Cadmium	5	NG	2.0U	2.0U	2.0U	2.6	2.8	2.6	2.8	2.8	
Calcium	NS	NG	130000	1000U	1000U	160000	170000	170000	180000	180000	
Chromium	50	NG	25U	25U	25U	25U	25U	25U	25U	25U	
Cobalt	NS	NG	10U	10U	10U	10U	10U	10U	10U	10U	
Copper	200	NG	25U	25U	25U	25U	25U	25U	25U	25U	
Iron	300***	NG	380***	150U	150U	400***	290	360***	460***	150U	
Lead	25	NG	5.0U	5.0U	5.0U	5U	5U	5U	5U	5U	
Magnesium	NS	35000	26000	1000U	1000U	620000	680000	610000	710000	730000	
Manganese	300***	NG	170***	25U	25U	72***	53	100***	48***	36	
Nickel	100	NG	25U	25U	25U	25U	<u>25U</u>	25U	25U	25U	
Potassium	NS	NG	5500	2500U	2500U	300000	340000	300000	360000	380000	
Selenium	10	NG	25U	<u>2</u> 5U	25U	25U	25U	25U	25U	25U	
Silver	50	NG	10U	10U	10U	10U	10U	10U	10U	10U	
Sodium	20000	NG	28000	25000U	25000U	5100000	5500000	5000000	5500000	5900000	
Thallium	NS	0.5	5.0U	5.0U	5.0U	5U	5U	5U	5U	5U	
Vanadium	NS	NG	25U	25U	25U	25U	<u>2</u> 5U	25U	25U	25U	
Zinc	NS	2000	25U	25U	25U	26	25U	28	25U	25U	
Mercury	0.7	NG	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	
pH	NS	NG	7.2	4.2	4.2	7.7	7.6	7.7	7.8	7.8	

ND No Data

U Undetectable Levels

190

NS No Standard

NG No Guidance

*** Total for Iron and Manganese is > 500

Note-1: pH listed is the pH recorded in the field

Note-2: ST-SW1 through ST-SW5 represents samples

collected from surface water



Surcharge Pilot Study - Groundwater and Surface Water Results

Metals and pH Site 2A/2B: HHMT - Port Ivory Facility

	· · · · · · · · · · · · · · · · · · ·	r	r	T	I	l	[1	
Location	Recommended	Recommended	CS-7	CS-7	RS-1	RS-1	ST-4S	ST-4S	ST-4D	ST-4D	
Date	Groundwater	Groundwater	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	
Lab ID	Cleanup	Cleanup	AB72292	AB72293	AB72294	AB722945	AB72296	AB72297	AB72298	AB72299	
Concentration	Standard	Guidance	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	
Filtered or Unfiltered	UG/L	UG/L	unfiltered	filtered	unfiltered	filtered	unfiltered	filtered	unfiltered	filtered	
Aluminum	NS	NG	100U	100U	100U	100U	970	2300	140	100U	
Antimony	3	NG	7.5U	7.5U	7.5U	7.5U	7.5U	75	7.5U	7.5U	
Arsenic	25	NG	7.5	5.9	4.1	4.6	15	69	8.1	8.3	
Barium	1000	NG	25U	25U	78	76	80	130	780	710	
Beryllium	NS	3	4U	4U	4U	4U	4U	40	4U	4U	
Cadmium	5	NG	2U	2U	2U	2U	2U	3.2	2U	2U	
Calcium	NS	NG	83000	81000	130000	120000	200000	90000	290000	230000	
Chromium	50	NG	25U	25U	25U	25U	25U	46	25U	25U	
Cobalt	NS	NG	10U	10U	10U	10U	10U	220	10U	10U	
Copper	200	NG	25U	25U	25U	25U	25U	160	25U	25U	
Iron	300***	NG	150U	150U	1400***	950***	3300***	550***	22000***	19000***	
Lead	25	NG	5U	5U	5U	5U	5U	5U	5U	5U	
Magnesium	NS	35000	34000	34000	18000	19000	2000	52000	97000	89000	
Manganese	300***	NG	25U	25U	170**	170***	28***	150***	5200***	4800***	
Nickel	100	NG	25U	25U	25U	25U	25U	49	25U	25U	
Potassium	NS	NG	ND	ND	ND	ND	ND	ND	ND	ND	
Selenium	10	NG	25U	25 U	25U	25U	25U	25U	25U	25U	
Silver	50	NG	10U	10U	10U	10U	10U	10U	10U	10U	
Sodium	20000	NG	130000	130000	32000	33000	400000	470000	2100000	2000000	
Thallium	NS	0.5	5U	5U	5U	5U	5U	42	5U	5U	
Vanadium	NS	NG	25U	25U	25U	25U	25U	25U	28	26	
Zinc	NS	2000	49	25U	130	34	26	920	25U	25U	
Mercury	0.7	NG	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	
pH	NS	NG	8	8	8	8	11	11	7	7	

ND No Data

U Undetectable Levels

**#

NS No Standard

NG No Guidance

*** Total for Iron and Manganese is > 500

Note-1: pH listed is the pH recorded in the field

Note-2: ST-SW1 through ST-SW5 represents samples

collected from surface water

Te 17C

Surcharge Pilot Study - Groundwater and Surface Water Results

Metals and pH Site 2A/2B: HHMT - Port Ivory Facility

Location	Recommended	Recommended	ST-1S	ST-1S	ST-1D	ST-1D	FB-1	FB-1	FB-1	FB-1	
Date	Groundwater	Groundwater	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/6/2002	11/7/2002	11/7/2002	
Lab ID	Cleanup	Cleanup	AB72300	AB72301	AB72302	AB72303	AB72305	AB72306	AB72395	AB72396	
Concentration	Standard	Guidance	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	
Filtered or Unfiltered	UG/L	UG/L	unfiltered	filtered	unfiltered	filtered	unfiltered	filtered	unfiltered	filtered	
Aluminum	NS	NG	3200	350	910	140	100U	100U	100U	100U	
Antimony	3	NG	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	
Arsenic	25	NG	90	53	8	6.2	4U	4U	4U	4U	
Barium	1000	NG	190	150	84	74	25 U	25U	25U	25U	
Beryllium	NS	3	4U	4U	4U	4U	4U	4U	4U	4U	
Cadmium	5	NG	3.2	2.5	2U	2U	2 U	2U	2U	2U	
Calcium	NS	NG	690000	350000	74000	74000	1000U	1000U	1000U	1000U	
Chromium	50	NG	25U	25U	25U	25U	25 U	25U	25U	25U	
Cobalt	NS	NG	10U	10U	10U	10U	10U	10U	10U	10U	
Copper	200	NG	25U	25U	25U	25U	25U	25U	25U	25U	
Iron	300***	NG	2200***	150U	5200***	3500***	150U	150U	150U	150U	
Lead	25	NG	5U	5U	5U	5U	5U	5U	5U	5Ú	
Magnesium	NS	35000	12000	1000U	58000	59000	1000U	1000U	1000U	1000U	
Manganese	300***	NG	54***	25U	120***	110***	25 U	25U	25U	25U	
Nickel	100	NG	92	110	25U	25U	25U	25U	25U	25U	
Potassium	NS	NG	ND	ND	ND	ND	ND	ND	ND	ND	
Selenium	10	NG	25U	25U	25U	25U	25U	25U	25U	25U	
Silver	50	NG	10U	10U	10U	10U	10U	10U	10U	10U	
Sodium	20000	NG	1500000	1600000	660000	720000	2500U	2500U	2500U	2500 U	
Thallium	NS	0.5	5U	5U	5U	5U	5U	5 U	5U	5U	
Vanadium	NS	NG	32	25U	2 5U	26	25U	25U	25U	25U	
Zinc	NS	2000	44	25U	25U	25U	25U	25U	25U	25U	
Mercury	0.7	NG	0.2U	0.21	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	
pH	NS	NG	13	13	7.3	7.3	6.5	6.5	6.8	6.8	

ND No Data

٢.

U Undetectable Levels

- 14

NS No Standard

NG No Guidance

*** Total for Iron and Manganese is > 500

Note-1: pH listed is the pH recorded in the field

Note-2: ST-SW1 through ST-SW5 represents samples

collected from surface water

278

Surcharge Pilot Study - Groundwater and Surface Water Results Metals and pH Site 2A/2B: HHMT - Port Ivory Facility

		D 11	am 20	07 00	07 00	07.00	07 20	07 30	07.20	077.20		
Location	Recommended	Recommended	ST-2S	IST-2S	ST-2D	ST-2D	ST-3D	ST-3D	51-35	51-35		
Date	Groundwater	Groundwater	11/7/2002	11/7/2002	11/7/2002	11/7/2002	11/7/2002	11/7/2002	11/7/2002	11/7/2002		
Lab ID	Cleanup	Cleanup	AB72398	AB72398	AB72400	AB72401	AB72402	AB72403	AB72404	AB72405		
Concentration	Standard	Guidance	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L	UG/L		
Filtered or Unfiltered	UG/L	UG/L	unfiltered	filtered	unfiltered	filtered	unfiltered	filtered	unfiltered	filtered		
Aluminum	NS	NG	2400	800	1400	670	680	100U	4400)0 420		
Antimony	3	NG	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U	7.5U		
Arsenic	25	NG	28	23	8.2	6.2	8.2	4U	61	9.7		
Barium	1000	NG	160	180	120	110	91	. 83	510	430		
Beryllium	NS	3	4U	4U	4U	4U	4U	4U	4U	4U		
Cadmium	5	NG	2U	2U	2U	2Ū	2U	2U	2.7	2U		
Calcium	NS	NG	420000	420000	120000	110000	220000	220000	880000	430000		
Chromium	50	NG	25U	25U	25U	25U	25U	25U	25U	25U		
Cobalt	NS	NG	10U	10U	10U	10U	10U	10U	10U	10U		
Copper	200	NG	25U	25U	25U	25U	25U	25U	25U	25U		
Iron	300***	NG	1100***	150U	6600***	4900***	8500***	8200***	2100***	150U		
Lead	25	NG	6.7	5U	5U	5U	5U	5U	5U	5U		
Magnesium	NS	35000	2400	1000U	83000	82000	130000	140000	13000	1000U		
Manganese	300***	NG	25U	25U	430***	400***	2300***	2500***	58***	25U		
Nickel	100	NG	39U	37	25U	25U	25U	25U	39	25U		
Potassium	NS	NG	52000	57000	23000	32000	72000	74000	250000	2500		
Selenium	10	NG	25U	25U	25U	25U	25U	25U	25U	25U		
Silver	50	NG	10U	10U	10U	10U	10U	10U	10U	10U		
Sodium	20000	NG	780000	850000	740000	540000	2200000	2300000	2100000	25000		
Thallium	NS	0.5	5U	5U	5U	5U	5U	5U	7.1	5U		
Vanadium	NS	NG	27	25U	25U	25U	55	52	27	25U		
Zinc	NS	2000	56	25U	25U	25U	25U	25U	130	25U		
Mercury	0.7	NG	0.2U	0.2U	0.2U	0.2U	0.2U	0.2U	1.7	0.2U		
pH	NS	NG	. 13	13	7.8	7.8	7.2	7.2	13	13		

ND No Data

U Undetectable Levels

NS No Standard

NG No Guidance

*** Total for Iron and Manganese is > 500

Note-1: pH listed is the pH recorded in the field

Note-2: ST-SW1 through ST-SW5 represents samples

and the second second

collected from surface water

2779

the third round of sampling of PG-CS-7 but not in the first or second rounds. This is not regarded as increase in contaminant concentration during the pilot study since xylenes were detected at a higher concentration in the SI sampling round performed in November 2000. As shown on the surcharging phasing study and schedule, surcharging has been completed at the two areas labeled Phase 1A has been completed and has been initiated at the area labeled Phase 1B North. Although the pilot study did not reveal the presence of adverse impacts to groundwater or Bridge Creek, additional monitoring efforts are proposed to confirm the findings of the pilot study. The proposed actions will mimic those performed during the pilot study but will utilize five existing monitoring wells situated in both Site 1 and Site 2A at locations adjacent to Bridge Creek; the proposed locations include EW-1 (Site 2A), MW-5 (Site 1) MW-6/MW-6D (Site 2A) and CS-7 (Site 2A). The proposed monitoring program will also include sampling of two sets of wells (MW-15/MW-15D located on Site 2A and MW-1/MW-1D located at Site 1) to provide additional groundwater information. The samples will be analyzed for TCL VOCs, phenols, TAL Metals and pH. Based on current information, four rounds of sampling will be performed over the next 12 months with collection occurring once per quarter.

Hatch Mott

MacDonald

The monitoring program will include a review of conditions at Bridge Creek and the sampling of both surface water and sediment/precipitate. The proposed sediment/surface water sampling will be performed in conjunction with the proposed groundwater sampling events and will include samples from five locations. Sediment and surface water samples also will be analyzed for TCL VOCs, phenols, TAL Metals and pH. In addition, monthly inspection will be performed to document conditions along the eastern bank of Bridge Creek. The inspections will be performed during low tide and will note the presence of seeps, precipitate, bank failure or other evidence of mass movement of subsurface material or liquids. All sampling will be performed in accordance with NYSDEC protocol and laboratory analysis will be performed by a NYSDEC certified laboratory. Again, information from the pilot study has not identified any adverse impacts to groundwater or Bridge Creek, however, the Port Authority intends to confirm these findings through the above-described program.





14.0 CONCLUSIONS

This report presents a summary of assessment, investigation, delineation and remedial actions which have been undertaken at Site 2A/2B from 2000 through 2003. By and large, assessment and investigation efforts identified relatively few environmental issues with respect to Site 2A/2B. Generally, the issues involved the presence of fill material, previously closed USTs and the presence of a few petroleum-impacted areas. As described herein, the environmental conditions at this Site 2A/2B as well as Sites 1 and 3 of the HHMT-Port Ivory Facility have been evaluated with respect to the HHMT-Port Ivory Facility's proposed usage. Further, the Port Authority has undertaken actions to address residual petroleum related contamination through source removal. The actions undertaken at these areas also have included the removal of two previously closed USTs (closed in place by P&G) and an abandoned oil/water separator system from Site 2A. Based on assessment and investigation activities, no remedial actions were warranted with respect to Site 2B. With regard to the presence of fill material, the SI/RI activities identified the presence of contaminants at Site 2A/2B, which are typical to urban sites in the New York Metropolitan region. The presence of the fill material and residual levels of fill-related contaminants in soil does not appear to have adversely impacted groundwater quality at Site 2A/2B or Bridge Creek, situated adjacent to the western property boundary of Site 2A/2B and Site 1.

Overall, industrial/commercial usage such as the Port Authority's planned usage of the site as an intermodal facility and container terminal is not inconsistent with the residual levels of contamination noted to be present in site soil and groundwater. The Port Authority has addressed several petroleumimpacted areas through source removal and will address fill material and residual contamination (associated with the fill material and prior industrial usage of the site by P&G) through site redevelopment including the use of engineering and institutional controls, which will minimize potential impacts to human health and the environment. Specifically, the Port Authority intends to install material such as pavement and other semi-impervious material, which will function as an environmental cap at Site 2A/2B and the entire HHMT-Port Ivory Facility. This action will tend to stabilize contaminants in soil and fill material by impeding infiltration, thereby reducing the potential for contaminants in soil to leach from the unsaturated zone to groundwater. Further, the placement of such materials will safeguard the public by preventing exposure to contaminants in soil and groundwater.

P:\232952wmd\Operable Unit Reports\Operable Unit 2\Post VCP Revisions\Report\Final Draft Report Site 2A 2B 92004.doc

GEOPHYSICAL SURVEY PROCTOR & GAMBLE PORT IVORY FACILITY STATEN ISLAND, NEW YORK

Prepared for:

Contraction of the

Killam Associates 27 Bleeker Street PO Box 1008 Millburn, New Jersey 07041-1008

Prepared by:

Hager-Richter Geoscience, Inc. 8 Industrial Way - D10 Salem, New Hampshire 03079

File 00D59 December, 2000

@2000 Hager-Richter Geoscience, Inc.

1

0. EXECUTIVE SUMMARY

Hager-Richter Geoscience, Inc. conducted a geophysical survey at the Proctor & Gamble Port Ivory Facility located on Staten Island, New York for Killam Associates (Killam)in October and November, 2000. The scope of the project and areas of interest were specified by Killam. The geophysical survey is part of a environmental investigation of the site being conducted by Killam on behalf of the Port Authority of New York and New Jersey.

The site is a large inactive industrial facility located in the northwestern portion of Staten Island. The Site consists of several buildings, gravel and paved parking areas, rail spurs, foundations and slabs of demolished buildings, and open areas. Hager-Richter was contracted by Killam to locate utilities in the vicinity of as many as 210 proposed boring locations and to locate possible USTs that may be present at nine locations identified at the Site by Killam. The locations of utilities detected as part of the boring program were marked on site as specified by Killam, and are not discussed further.

According to information provided by Killam, as many as 19 USTs might be present in nine areas of the site, designated by Killam UST Area 1 through UST Area 9. Four of the nine areas may contain multiple USTs, and five areas may contain a single UST.

The objective of the geophysical survey was to detect possible USTs in each of the nine areas of interest specified by Killam, and if any were detected, to determine the locations of each.

The geophysical survey consisted of time domain electromagnetic induction metal detector (EM61) surveys followed by focused GPR surveys in each of the areas of interest. The EM61 data were acquired at approximately 8-inch intervals along profile lines spaced 5 feet apart across the accessible portions of the areas of interest. In order to aid in the identification of the objects, a focused GPR survey was conducted at the locations of anomalies detected with the EM.

The results of the geophysical survey conducted at the Proctor & Gamble Port Ivory Facility can be summarized as follows:

• Several areas of buried metal were detected in the nine areas of interest at the site on the basis of the EM61 data. None of the identified areas of buried metal could be definitively identified as a UST due to the limited GPR signal penetration and/or surface features such as concrete slabs, metal piping, and rail spurs. Whether the buried metal is a UST is present cannot be determined on the basis of the geophysical data alone.

HAGER-RICHTER GEOSCIENCE, INC.

(

ŝ

N.

Several other EM61 anomalies are interpreted as possible utilities.

HAGER-RICHTER GEOSCIENCE, INC.

Sec.

ti den

TABLE OF CONTENTS

0.	Execut	tive Summ	nary			 		•••		·			 	•••				•	• •			•	i
1.	Introdu	uction			<i>.</i>	 					•		 					•			• •		1
2.	Equipr	nent and I	Proce	edui	res	 							 										2
	2.1	General				 							 										2
	2.2	EM61				 							 										2
	2.3	GPR				 							 										2
	2.4	Site Spec	cific		•	 • •	•••				• •		 	• •				•				•	4
3.	Results	s and Disc	ussic	on		 							 										5
	3.1	General				 							 										5
	3.2	UST Are	ea 1			 							 										5
	3.3	UST Are	ea 2			 							 										6
	3.4	UST Are	ea 3			 							 										6
	3.5	UST Are	ea 4			 							 										7
	3.6	UST Are	ea 5			 							 										7
	3.7	UST Are	ea 6			 							 										8
	3.8	UST Are	ea 7			 							 										8
	3.9	UST Are	ea 8			 							 										9
	3.10	UST Are	ea 9			 						•••	 • •	• •		•••	• •	•	• •				9
4.	Conclu	isions				 			• •				 		•					•		. 1	.0
5.	Limita	tions				 							 									. 1	1

FIGURES

1.	Site Location
2.	EM61 - UST Area 1
3.	GPR survey & Integrated Interpretation - UST Area 1
4. ·	EM61 - UST Area 2
5.	GPR survey & Integrated Interpretation - UST Area 2
6.	EM61 - UST Area 3
7.	GPR survey & Integrated Interpretation - UST Area 3

HAGER-RICHTER GEOSCIENCE, INC.

Geophysical Survey Proctor & Gamble Port Ivory Facility Staten Island, New York File 00D59 December, 2000

Ľ

g

S. C. S. S.

- 8. EM61 UST Area 4
- 9. GPR survey & Integrated Interpretation UST Area 4
- 10. EM61 UST Area 5
- 11. GPR survey & Integrated Interpretation UST Area 5
- 12. EM61 UST Area 6
- 13 GPR survey & Integrated Interpretation UST Area 6
- 14. EM61 UST Area 7
- 15. GPR survey & Integrated Interpretation UST Area 7
- 16. EM61 UST Area 8
- 17. GPR survey & Integrated Interpretation UST Area 8
- 18. EM61 UST Area 9
- 19. GPR survey & Integrated Interpretation UST Area 9

PLATE

1. Site Plan

APPENDIX

1. EM61 Surveys

2. GPR Surveys

HAGER-RICHTER GEOSCIENCE, INC.

1. INTRODUCTION

Hager-Richter Geoscience, Inc. conducted a geophysical survey at the Proctor & Gamble Port Ivory Facility located on Staten Island, New York for Killam Associates (Killam) October 25 -November 15, 2000. The scope of the project and areas of interest were specified by Killam. The geophysical survey is part of a environmental investigation of the site being conducted by Killam on behalf of the Port Authority of New York and New Jersey.

The site is a large inactive industrial facility located in the northwestern portion of Staten Island. The general location of the Site is shown in Figure 1, and Plate 1 is a site plan. The Site consists of several buildings, gravel and paved parking areas, rail spurs, foundations and slabs of demolished buildings, and open areas. Hager-Richter was contracted by Killam to locate utilities in the vicinity of as many as 210 proposed boring locations and to locate possible USTs that may be present at nine locations identified at the Site by Killam. The locations of utilities detected as part of the boring program were marked on site as specified by Killam, and are not discussed further.

According to information provided by Killam, as many as 19 USTs might be present in nine areas of the site, designated by Killam as UST Area 1 through UST Area 9. Four of the nine areas may contain multiple USTs, and five areas may contain a single UST. The locations of the nine areas specified by Killam are shown as hatched areas on Plate 1.

The objective of the geophysical survey was to detect possible USTs in each of the nine areas of interest specified by Killam, and if any were detected, to determine the locations of each.

The geophysical survey consisted of time domain electromagnetic induction metal detector (EM61) surveys followed by focused GPR surveys in each of the areas of interest. The EM61 survey detects and outlines areas containing buried metal. However, the EM method cannot provide information on the type of objects causing the EM anomaly. In order to aid in the identification of the objects, a focused GPR survey was conducted at the locations of anomalies detected with the EM61.

James Coffman, Jeffrey Reid, P.G., and Jeffrey Sullivan of Hager-Richter conducted the field operations on October 30, November 8, 9, 14, and 15, 2000. The project was coordinated with Ms. Jennifer Kohlsaat of Killam. Mr. Daniel Davis and Mr. Charles Springer, both of Killam, specified the areas of interest for the survey and were present for portions of the field work.

- 1 -

HAGER-RICHTER GEOSCIENCE, INC.

2. EQUIPMENT AND PROCEDURES

2.1 General

The equipment, limitations, and general procedures of EM61 high sensitivity metal detector and GPR surveys are described below. Details specific to this project are given in the Site Specific section below.

2.2 EM61

Equipment. The EM survey was conducted using a Geonics Model EM61 time domain electromagnetic induction metal detector, the industry standard for this type of geophysical survey. The EM61 produces a pulsed primary magnetic field in the earth that induces eddy currents in the ground and in nearby metal objects. The receiver is timed to measure the secondary magnetic field produced by eddy currents after those in the ground have dissipated, i.e., only the current in the metal objects. The data are recorded on a digital data logger. The EM61 is relatively insensitive to nearby cultural interferences such as buildings.

Limitations of the Method. The data from an EM61 survey are affected by surface metal debris in the survey area, and its depth sensitivity is limited to about 15 feet. The instrument is relatively cumbersome, and works best where the 1-meter square transmit and receive coils can be hand pulled in a small trailer.

Detection and identification should be clearly differentiated. Detection is the recognition of the presence of a metal object, and the electromagnetic method is excellent for such purposes. Identification, on the other hand, is determination of the nature of the causative body (i.e., what is the body -- a cache of drums, UST, automobile, white goods, etc.?). Although the EM61 data cannot be used to *identify* all buried metal objects, they provide excellent guides to the identification of some objects. For example, buried metal utilities produce anomalies with lengths many times their widths.

2.3 ^{\\} GPR

Equipment. The GPR survey was conducted using a Geophysical Survey Systems SIR-2 digital GPR system equipped with a survey wheel to trigger recording of data at equal horizontal distances. The GPR system was used with a 500 MHz antenna and a 60 nsec time window. The GPR traverses were spaced approximately 5 feet apart, and were conducted at the locations of EM61 anomalies.

Limitations of the Method. There are limitations of the GPR technique as used to detect and/or locate targets such as those of the subject Site: (1) surface conditions, (2) electrical conductivity of the ground, (3) contrast of the electrical conductivities of the targets and the ground, and (4) spacing between lines. Of these limitations, only the fourth, line spacing, is controlled by the operator.

The condition of the ground surface can affect the quality of the GPR data and the depth of penetration of the GPR signal. Sites covered with high grass, bushes, landscape structures, debris, obstacles, soil mounds, etc. limit the survey access and the coupling of the GPR antenna with the ground. In many cases, the GPR signal will not penetrate below concrete pavement, especially inside of buildings, and a target may not be detectable.

The electrical conductivity of the ground determines the attenuation of the GPR signals, and thereby limits the maximum depth of exploration. The GPR signal does not penetrate clay-rich soils, and targets buried in clay can be missed.

A definite contrast in the electrical conductivities of the ground and the target is required to obtain a reflection of the GPR signal. If the contrast is too small, possibly due to construction details or extremely corroded conditions of metal targets, then the reflection may be too weak to recognize, and the target can be missed.

The spacing between lines is under control of the GPR operator, and the design of the survey is based on the dimensions of the smallest feature of interest. Targets with dimensions smaller than the spacing between GPR survey lines can be missed.

2.4 Site Specific

As noted in the Introduction, Killam specified nine areas of interest for the geophysical survey. A local survey grid was established in each of the UST survey areas and tied to fixed landmarks.

EM61 data were acquired at approximately 8-inch intervals along lines spaced 5 feet apart in the accessible portions of each area. The EM61 was operated with the 1-meter square transmit/receive coils mounted on a hand-drawn trailer with a survey wheel that measures distance and triggers data collection at equal intervals. The EM61 data were recorded digitally and processed in the field using software provided by the manufacturer. A color contour plot of the data was generated using commercially available software (Geosoft).

A focused GPR survey was conducted at the locations of anomalies detected by the EM61 survey to attempt to identify the causative body(ies). GPR traverses were located along the same

Sec. Sec.

lines as the EM61 survey and spacing was variable based on the size of the EM anomalies and surface conditions. The GPR antenna was pulled by hand for all traverses.

GPR data were acquired with a 300 MHz antenna and a 60 nsec time window. GPR signal penetration varied significantly at the Site. Based on handbook values of time-to-depth conversions for the GPR signal in average soils, the GPR signal penetration is estimated to have varied from about 1 foot to about 5 feet.

HAGER-RICHTER GEOSCIENCE, INC.

3. RESULTS AND DISCUSSION

3.1 General

The geophysical survey consisted of a time domain electromagnetic induction metal detector (EM61) survey followed by a focused GPR survey where the EM61 survey indicated possible buried metal. Plate 1 is a Site Plan provided by Killam showing the locations of the survey areas.

Interpretation of EM61 data is based on the *relative* response (in millivolts) of the top and bottom instrument coils to local conditions. The differential response, the difference between the top and bottom coils, is typically used as the best indication of the location of buried metal objects, and is shown in the figures for this report. The instrument is not calibrated to provide an absolute measure of a particular property, such as the conductivity of the soil or of buried metal objects. Subsurface metal objects produce sharply defined positive anomalies when the EM61 is positioned directly over them. Such anomalies are colored red and pink on the color plots presented herein. Acquiring data at short intervals along closely spaced lines, as was done at the present site, provides high spatial resolution of the location and footprint of the targets. Thus, buried metal is recognized in contour plots of EM61 data by positive anomalies (red or pink zones) roughly corresponding to the dimensions of the buried metal.

Many surface metal objects and objects containing metal are present in the UST survey areas such as manhole covers, railroad tracks, fences, and reinforced concrete. The locations of such objects are shown on the figures for each of the areas. Because these objects contain metal, they can produce significant EM anomalies. The presence or absence of buried metal in these areas cannot be determined due to the anomalies caused by such surface objects.

In general, GPR signal penetration at the site was limited, with reflections received for less than about 30 nsec. The limited signal penetration is likely due to conductive soils, and in many places, concrete at the surface. Based on handbook time-to-depth conversions for the GPR signal in average soils, the GPR signal penetration is estimated to have been no more than about 2 to 3 feet for most of the areas of interest.

3.2 UST Area 1

UST Area 1 is located on the north side of Building 20, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 2 is a color contour plot of the EM61 data for UST Area 1, and Figure 3 shows the locations of the GPR traverses and the

interpretation of both the EM61 and GPR data. Five areas of possible buried metal were detected within the survey area, and their locations are shown on Figure 3. GPR traverses were conducted in the central portion of the area. GPR signal penetration was limited to less than 2 feet. Therefore, no additional information regarding the causative bodies was determined for this area with the GPR traverse.

Based on the shapes and sizes of the EM anomalies for UST Area 1, we infer that a utility and several other buried objects are present. Whether the objects are USTs cannot be determined on the basis of the geophysical data alone. If any of the buried metal objects is a UST, its capacity is likely 1000 gallons or less.

3.3 UST Area 2

UST Area 2 is located south of a wood shavings stockpile area, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 4 is a color contour plot of the EM61 data for UST Area 2, and Figure 5 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data. One area of possible buried metal was detected within the survey area. The area is located about 35 feet south of a concrete pad. GPR traverses were conducted over the location of the EM anomaly. GPR signal penetration is estimated to have been about 4 to 5 feet for this area, but GPR reflections typical of a UST were not detected in the area included in the GPR survey.

Based on the presence of the EM anomaly in UST Area 2, we infer that a buried metal object is present. Whether the object is a UST cannot be determined on the basis of the geophysical data alone. Because no GPR reflections typical of a UST were observed in the records for the effective depth of penetration of the GPR signal (about 5 feet), and the EM anomaly is small in amplitude, we conclude that if a UST is present, it would likely be located at a depth greater than 5 feet.

3.4 UST Area 3

UST Area 3 is located north of the northeast corner of Building 13, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 6 is a color contour plot of the EM61 data for UST Area 3, and Figure 7 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data. Two areas of possible buried metal were detected within the survey area as well as a possible utility. One buried metal object is located about 25 feet east of a trailer, the other is located about 60 feet east of the trailer, and the locations of both are shown on Figure 7.

GPR signal penetration is estimated to have been about 2 to 3 feet for this area. GPR reflections typical of a UST were not detected in the area included in the GPR survey. GPR reflections typical of a flat structure, such as a concrete pad, are present at the location of the southern end of the EM anomaly closer to the trailer.

Based on the presence of the EM anomalies in UST Area 2, we infer that two buried metal objects are present. The GPR data indicate that at least part of one of the EM anomalies may be related to a flat concrete-like structure. Whether the concrete object is a UST cannot be determined on the basis of the geophysical data alone.

3.5 UST Area 4

UST Area 4 is located west of Buildings 34 and 38 and north of a former floor slab for a demolished building, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 8 is a color contour plot of the EM61 data for UST Area 4, and Figure 9 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data.

The western portion of the survey area is covered by a concrete pad. Three significant EM anomalies are present in this portion of the survey area and one large EM anomaly is present along the southeast edge of the survey area. The areas of the EM anomalies are shown as areas of buried metal on Figure 9. The large EM anomalies may be caused by structures located under the concrete slab. The GPR signal penetration over the concrete slab is limited to less than about 1 foot and GPR reflections typical of USTs were not detected. Whether USTs are located under the slab cannot be determined on the basis of the geophysical data alone. The remaining portion of UST Area 4 is generally free of buried metal.

3.6 UST Area 5

UST Area 5 is located along a rail spur southwest of Building 17, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 10 is a color contour plot of the EM61 data for UST Area 5, and Figure 11 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data. Two rail spurs and a reinforced concrete surface drainage swale are present in the area. High amplitude EM anomalies are present near the concrete drainage swale and low amplitude negative EM anomalies are observed for the rail spurs.

- 7 -

GPR traverses were conducted in the northwest corner of the survey area, but the GPR signal penetration was limited to less than about 1 foot and no GPR reflection typical for a UST were detected.

3.7 UST Area 6

UST Area 6 is located along a rail spur west of Building 17, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 12 is a color contour plot of the EM61 data for UST Area 6, and Figure 13 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data. A rail spur and iron rimmed surface drain are present along the east side of the survey area.

Five EM anomalies not related to the surface features were identified, and their locations are shown on Figure 13. The two large circular anomalies located in the northeast portion of the survey area are likely caused by buried concrete. A small portion of a slab was visible on site and its presence was confirmed with the GPR. The remaining three anomalies are low amplitude and small in extent and are likely too small to be caused by USTs.

3.8 UST Area 7

UST Area 7 is located south of Building S-#35, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 14 is a color contour plot of the EM61 data for UST Area 7, and Figure 15 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data. Surface objects such as a rail spur, a concrete loading dock, a steel plate, transformers, and a tower are present in the survey area. The EM data were adversely affected at such locations.

Four EM anomalies not related to the surface features were identified, and their locations are shown on Figure 15. A large EM anomaly is present in the central portion of the survey area. The GPR data for the area of the large anomaly indicate the presence of a shallow buried reinforced concrete slab or structure at a depth of about 1 foot in the southern part of the anomaly. GPR records for the traverses conducted in the vicinity of the remaining anomalies contain no reflections characteristic of USTs. Such areas are shown as areas of buried metal. Whether the buried metal objects are USTs cannot be determined on the basis of the geophysical data alone.



3.9 UST Area 8

ľ

18

UST Area 8 is located at the northeast corner of Building 55, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 16 is a color contour plot of the EM61 data for UST Area 8, and Figure 17 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data. Surface objects such as a concrete pad and vertical pipes cut at grade are present in the survey area and such objects are shown on Figure 17.

Three anomalies attributed to buried metal objects were identified by the EM survey and their locations are shown on Figure 17. EM anomalies attributed to subsurface utilities were also identified. GPR signal penetration in the areas of the EM anomalies was limited to a depth of about 1 foot and no GPR reflections typical of a UST were detected. Therefore, no information regarding the causative bodies could be determined. Whether the buried metal objects are USTs cannot be determined on the basis of the geophysical data alone.

3.10 UST Area 9

UST Area 9 is located between Buildings 52 and 53, and its location is shown on Plate 1. EM61 data were acquired along survey lines spaced 5 feet apart, and GPR data were acquired at most locations where the EM data indicated the presence of buried metal. Figure 18 is a color contour plot of the EM61 data for UST Area 9, and Figure 19 shows the locations of the GPR traverses and the interpretation of both the EM61 and GPR data.

Several surface metal objects, such as valve box covers, transformers, and overhead pipes are present in the survey area. Four 4-inch pipes, cut at the surface, are present in the southeast corner of the survey area. Significant EM anomalies are present at the locations of the surface features and may mask the presence of buried metal objects, if any, at such locations.

Three anomalies not associated with surface metal were identified by the EM survey. These anomalies have been attributed to buried metal objects. GPR signal penetration in the areas of the EM anomalies was limited to a depth of about 1 foot and no GPR reflections typical of a UST were detected. Therefore, no information regarding the causative bodies could be determined. Whether the buried metal objects are USTs cannot be determined on the basis of the geophysical data alone.

-9-



4. CONCLUSIONS

Based on the geophysical survey conducted at the Proctor & Gamble Port Ivory Facility located on Staten Island, New York, we conclude:

Several areas of buried metal were detected in the nine areas of interest at the site on the basis of the EM61 data. None of the identified areas of buried metal could be definitively identified as a UST due to the limited GPR signal penetration and/or surface features such as concrete slabs, metal piping, and rail spurs. Whether the buried metal is a UST is present cannot be determined on the basis of the geophysical data alone.

Several other EM61 anomalies are interpreted as possible utilities.



2

5. LIMITATIONS

This letter report was prepared for the exclusive use of Killam Associates and the Port Authority of New York and New Jersey (Client). No other party shall be entitled to rely on this Report or any information, documents, records, data, interpretations, advice or opinions given to Client by Hager-Richter Geoscience, Inc. (H-R) in the performance of its work. The Report relates solely to the specific project for which H-R has been retained and shall not be used or relied upon by Client or any third party for any variation or extension of this project, any other project or any other purpose without the express written permission of H-R. Any unpermitted use by Client or any third party shall be at Client's or such third party's own risk and without any liability to H-R.

H-R has used reasonable care, skill, competence and judgment in the performance of its services for this project consistent with professional standards for those providing similar services at the same time, in the same locale, and under like circumstances. Unless otherwise stated, the work performed by H-R should be understood to be exploratory and interpretational in character and any results, findings or recommendations contained in this Report or resulting from the work proposed may include decisions which are judgmental in nature and not necessarily based solely on pure science or engineering. It should be noted that our conclusions might be modified if subsurface conditions were better delineated with additional subsurface exploration including, but not limited to, test pits, soil borings with collection of soil and water samples, and laboratory testing.

The detection of subsurface utilities and/or other subsurface objects was not an objective of this portion of the geophysical survey, and the survey was not designed to detect such. However, some utilities and/or other subsurface objects were detected and their locations are provided as a courtesy. Other utilities and/or other subsurface objects may be present and the Client or any third party shall not rely on this report for information on such.

Except as expressly provided in this limitations section, H-R makes no other representation or warranty of any kind whatsoever, oral or written, expressed or implied; and all implied warranties of merchantability and fitness for a particular purpose, are hereby disclaimed.



......









<u>LEGEND</u>



V

+

AREA OF POSSIBLE BURIED METAL

GPR TRAVERSE

EM ANOMALY ATTRIBUTED TO EFFECTS OF SURFACE OBJECTS. THE PRESENCE OR ABSENCE OF BURIED METAL WITHIN THIS AREA CANNOT BE DETERMINED ON THE BASIS OF THE GEOPHYSICAL DATA ALONE.



-H-A Z

APPROX.





Site sketch generated from field notes.

MONITORING WELL

WATER VALVE



















LEGEND

AREA OF POSSIBLE BURIED METAL

GPR TRAVERSE



EM ANOMALY ATTRIBUTED TO EFFECTS OF SURFACE OBJECTS. THE PRESENCE OR ABSENCE OF BURIED METAL WITHIN THIS AREA CANNOT BE DETERMINED ON THE BASIS OF THE GEOPHYSICAL DATA ALONE.

POSSIBLE UTILITY

MANHOLE



RAILROAD TRACK



NOTE:

Site sketch generated from field notes.















HAGER-RICHTER GEOSCIENCE, INC. Salem, New Hampshire












.







LEGEND

•	EM DATA STATION
P	PIPE
S	OVERHEAD SUPPORT



NOTES:

1. Site sketch generated from field notes.

2. Contour Interval = 20 mV.

Figure 18 EM61 Survey - UST Area 9 Procter & Gamble Port Ivory Facility Staten Island, New York File 00D59 December, 2000 HAGER-RICHTER GEOSCIENCE, INC. Salem, New Hampshire









APPENDIX EM61 Metal Detector Surveys

Equipment. The Geonics EM61Metal Detector is a time-domain electromagnetic induction type instrument designed solely for detecting buried metal objects. The manufacturer's specifications are attached. An air-cored 1-meter square transmitter coil generates a pulsed primary magnetic field in the earth, thereby inducing eddy currents in nearby metal objects. The decay of the eddy current produces a secondary magnetic field that is sensed by two receiver coils, one coincident with the transmitter and one positioned 40 cm above the main coil. By measuring the secondary magnetic field after the current in the ground has dissipated but before the current in metal objects has dissipated, the instrument responds only to the secondary magnetic field produced by metal objects. Two channels of secondary response are measured in mV and are recorded on a digital data logger. The system is generally operated by pulling the coils as a trailer with an odometer mounted on the axle to trigger the data logger automatically at 20-cm intervals.

Data Analysis and Interpretation. EM61 survey data are most commonly plotted as color contour plots of Channel 2, the lower of the two receiver coils, and the difference between Channel 1 and Channel 2. The differential plot suppresses the effects of surface metal objects.

A buried metal object produces a single, sharply defined, positive peak response when the EM61 is located directly over the object. Thus, the interpretation of the plotted data is relatively straightforward in terms of the presence and location of buried metal objects. The depth of metal objects can be estimated by the width or "footprint" of the peak response.

According to the manufacturer's literature, the EM61 can detect a single 55-gallon drum buried at a depth of 10 feet. The instrument provides excellent lateral location accuracy and discrimination of multiple targets due to the data density (20 cm) possible along each traverse. The EM61 is not as affected by interference from surface metal and electrical objects as other geophysical methods and has the advantage of detecting both ferrous and non-ferrous metal objects.

Limitations of the Method. The EM61 detects metal objects that are present below the 1meter square coils of the instrument, but it is not very sensitive to the presence of small metal objects located to the sides of the coils. It is possible, then, that metal objects could be missed in an EM61 survey if the survey data are collected at intervals greater than 1 meter.

Detection and identification should be clearly differentiated. Detection in this context is the recognition of the presence of a metal object, and the EM61 is excellent for such purposes. Identification, on the other hand, is determination of the nature of the causative body (i.e., what is the body -- a cache of drums, UST, automobile, white goods, etc.?), and the EM61 cannot identify the buried metal object.

APPENDIX GROUND PENETRATING RADAR SURVEYS

Field Work. A Geophysical Survey Systems, Inc. Model SIR-2 ground penetrating radar system was used for this survey. The SIR-2 is a fully digital system and includes a color monitor, grey-scale thermal printer, and 10-Gbyte digital tape backup system. The transmit/receive antenna is housed in a box that is moved across the surface. The antenna transmits electromagnetic signals into the subsurface and then detects, amplifies, and displays reflections of the signals in real-time on the color monitor. The result is a radar record of the subsurface.

The maximum depth of penetration of the GPR signal and the resolution of the reflections are controlled in part by the frequency of the antenna used and in part by the electrical properties of the subsurface. Hager-Richter owns antennas with the following center frequencies: 120 MHz, 300 MHz, 500 MHz, and 1000 MHz. The total time during which radar signals are recorded can be varied from a few to 1,000 nanoseconds (nsec). However, there is a trade-off between total time, corresponding to depth range, and resolution. As the total time of recording is increased, the resolution of the GPR records decreases. For a given site, the total time window is set to detect features located somewhat below the maximum expected target depths.

Interpretation. The horizontal axis of a GPR record represents distance across the surface and the vertical axis represents round-trip travel time of the radar signal. The round-trip travel time can be converted to approximate depth by correlating with reflections from targets of known depth or by using handbook values of velocities for materials in the subsurface. For those sites where the subsurface is electrically heterogeneous, the travel times of the radar signal may be different in the various materials, and the vertical scale for the radar records is not necessarily uniform with depth.

The reflections in a GPR record are produced by spatial changes in the physical properties (e.g., type of material, subsurface fluids, porosity, etc.) and related changes in the electrical properties (dielectric constant) of the subsurface materials in the path of the signals. The greater the difference in electrical properties between two materials in the subsurface, the stronger the reflection observed in the GPR record.

The size, shape, and amplitude of the GPR reflections are the characteristics that are considered in the interpretation of the data from any site. Because the electrical properties of metal USTs, utilities, and conduits different significantly from those of the soils in which they are buried, such objects produce GPR reflections with high amplitude and distinctive shapes that permit identification with a high degree of reliability. Most other objects, although readily detectable, require "ground truth" for identification. Only excavations provide positive identification for most objects identified in GPR surveys.

For GPR profiles oriented perpendicular to the long axis of a tank, the signature is similar to a hyperbola, the shape of which is a function of the diameter and depth of burial of the tank. For GPR profiles oriented parallel to the long axis of a tank, the signature is a set of parallel, high amplitude reflections that terminate sharply at the ends of the tank. GPR, then, is useful for determining the exact location and dimensions of USTs.

Limitations of the Method. The maximum depth to which GPR signals can penetrate depends on the electrical properties of the subsurface materials. The higher the electrical conductivity of the subsurface materials, the lower the radar signal penetration. Clay minerals and/or brackish water in the subsurface, for example, attenuate the GPR signal, so reflections are not received from materials at greater depths.

There are limitations of the GPR technique as used to detect and/or locate particular targets: (1) surface conditions, (2) electrical conductivity of the ground, (3) contrast of the electrical conductivities of the targets and the ground, and (4) spacing between lines. Of these limitations, only the fourth, line spacing, is controlled by the operator.

The condition of the ground surface can affect the quality of the GPR data and the depth of penetration of the GPR signal. Sites covered with high grass, bushes, landscape structures, debris, obstacles, soil mounds, etc. limit the survey access and the coupling of the GPR antenna with the ground. In many cases, the GPR signal will not penetrate below concrete pavement, and a target may not be detectable.

The electrical conductivity of the ground determines the attenuation of the GPR signals, and thereby limits the maximum depth of exploration. The GPR signal does not penetrate clayrich soils, and targets buried in clay can be missed.

A contrast in the electrical conductivities of the ground and the target is required to obtain a reflection of the GPR signal. If the contrast is too small, possibly due to extremely corroded conditions of a metal target, then the reflection may be too weak to recognize, and the target can be missed.

The spacing between lines is under control of the GPR operator, and the design of the survey is based on the dimensions of the smallest target of interest. Targets with dimensions smaller than the spacing between GPR survey lines can be missed.

Accurate determination of the depth to any interface requires calibration of the site specific GPR signal velocity. Where targets of a known depth are not available at a site, the time-to-depth conversion of the GPR signal can be estimated from handbook values, but such depth estimations might contain significant error.

Interpretation of GPR data is subjective. As noted above, "ground truth" through correlation with borings and excavations is required for positive identification of most objects detected on the basis of GPR data.

PA 547

6-90

THE PORT AUTHORITY OF NY & RU

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

	_									SHEET 7 OF 3
PROJECT					NAME OF CONTRACTOR				BORING NO.	SURFACE ELEV.
PROC	TOR an	1 GAMBLE			CRAIG RR				RRI	
LOCATION				1.					CONTRACT NO.	DATE
AS m	ARKED	out in TI	he fie	19 2	7	KILLIA	m Asso	C.	426-99-006	11-3-00
SPOON	. /	CASING SE	ZE HOLE	TYPE	4			GRO	UND WATER LEVEL	
3:0	.D. 278	"1.D. H.S. Aug	ers	1		Date	Time	Depth	Re	emarks
HAMMER (S	afety)	HAMMER'				,				
140 #	FALL	30. 4	FALL	-		11/3	Am	3.2		
DRILLER		\sim			ļ		1			
· ·	r.	Pennell								
INSPECTOR	M.	Dudeh		_						
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- '	SAMP.2			3SA	MPLE DES	SCRIPTION AND REMARK	(S E do
4.5	0	ifard	8.11	1	F.	"U · Jack	brown	5000.50	me crushed Rock, Pit	He cravel frace Silt 12'
augers		Augen		2	F.	H- brow	Jr c-1 4	SPND.	trove Grovel, tro	e Silt
					1	-,, 1			Good Luck	1 Louis CiAL
				3		n- da	<u>(K Drow</u>	<u>~~</u> B	SAND, HOU DIE	Use Trace Six1
					6	- <u>-</u>		<		
	►5 ◄		<u> </u>		F	<u></u>				
		<u> </u>	¥	9						
		7-9	21"		T.	File		· <	Anak	
			01-	5	<u> </u>			2		
		11-13								
		4.10	20"		F	Eu- b	(~) ~ (15	AND Las Gran	Ofrace Silt
		11.4		6	F			5-51	It's a for the second the	
╞╌╂───┣	► 10 <	11-15	·		-					
		5-11	∂Y"	2	F	ill -		4	SAME	
		13.10		1 +						
┠┈┨───┼					┢─					
		10-12	24"	V	Ĺź	<u>-ill-</u>			SAME	
		لاريور		0						
┝╌╉╼──┼		10.17			╆					
┝╶╃────┪	►15 ◄	1-9	21"	0	Li	<u>-11-5</u>	1- 100 - 41	ey ef	GAND, trace Orau.	ex, trace D'XT
		11-15		7			U	N U	•	
		5			\uparrow				Dulla .	
┝╌╂───┤		3-15		\Box	μŕ	· LI- Ced	Lown (ND, KITTLE Graven	+ Train Crusted Hock
		20-19		\mathcal{O}		Lit	the Sil	240		
		8-11	20"		c	 			1.140 E: 11.	Lacourt
┝╌╂───┤				(1	F	162. 100	Dom's w	<u>A DA</u>	ND, LITTLE SULT	I GO CAMUPA
V I	> 20	14-16								
[~	27-17	21"	A	15	- 11 -		Sam	26	
├─── ┼			Væ	121						
╞────┤		10-20		E		rd-been	24 5	JLT_	and a second	22.0'
					11	1 Samol	· Laind :	n 1 m	ENTER & 1 UD	batton at mire \$
						- are pr	An -	<u>. a </u>		
					<u>µ2</u>	ith the	Kollowing	except	Tions: 5#4 wes	rot
					50	urd. S	#12B	765 501	on I one Dt. J	AP \$ 1 4 OF JAR.
	- 65 4	0.1.1	L					باللو د .	- <u>provide priva</u>	

1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

THE PORT AUTHORITY OF NY& NJ

Engineering Department Construction Division **Materials Engineering Section**

BORING REPORT

-									SHEET) OF 4
PROJECT					NAME OF CONT	RACTOR	-	BORING NO.	SURFACE ELEV.
Howlan	of Nock	for Thory	16S,7	0	Craix	Urilha	×	<u> </u>	
LOCATION	70	,/			Ū	-		CONTRACT NO.	DATE
F10 (NOS/ of	orig Loc	-	· <u> </u>			l	416-11-00D	1119100
SPOON 72	23/		ZE HOLE	TYPE		T 1	GRO	UND WATER LEVEL	
7 "C	<u>D.D. 018</u>	"I.D. // W		/	Date	Time	Depth		
140 #	SGTID FALL 30		FALL	-	11/4/00	2 pm	2.9	while Mond	Bugging
DRILLER	5	Burns				-			- 2
INSPECTOR	<u></u>	Have				··· <u></u>			
CARING		- CON	DE 1	CAMP 2		30 41			
BLOWS/FT.	DEPTH	BLOWS/6"	COV'D	NO.		LII	NE LOCAT	TES CHANGE OF PROFIL	
Kand		Hand Augy	Full						
Augy				<u> </u>	F,11-C	indor			
	<u> </u>			2	Score	 2			
XU		TannelSom	24"			<			
(gung	2	J- Mar Mach	01	3	Samo				
	— —	2-2	<u> </u>		June	<u> </u>			
	<u>⊢</u> . –	2-1	12"	4	Grain		m. ch	I t Brown H	Port 6.2 8.0
	<u> </u>	WOR				2010 31	<u></u>	y J Proud 1	
Drili		1-8	9"	5	Black	Braun	Pat		6.2 10.0
Bhad		3-8		1			·		
with		10-10	10"	6	FGra	Sand:	TY Sili	Г	12,0
Much		5-7		-					
		c-D	144	/	ER.	5.1	Tus n		
	└ <u>─</u> ──	1 1 2	-' <i>''</i>		I Prow.	v sang	<u> </u>		
	- 15	6-8	<u>-</u>	Ŷ	L				
		8-5	9"	0	Same			<u> </u>	16.0
		<u> </u>							
ļ		_			L			Bullomot	-Boving /
			L			JJE S	# 6-	8 Taken WI	The 2" Speare
	201				L				
]			C	Samalo	. # 1.	-5 checkall	The PIDMotor
						CHILU CUIDE France T.T.			
∦	<u> </u>	<u> </u>			IT IT Saved ter ENLING Lastry -				
		 			<u>├</u>	sec jaci	<u>h San</u> g	plos on 194	
							•	r 	
	~~~								
L		0. 1			at Osmala T			<u></u>	

Length recovered; 0" — Loss of Sample, T — Trap used
 U = undisturbed; A = auger; OER = open end rod; V = vane
 Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

<u>4 547</u> 3-90			T	NE PO	RT AUTHO Engineering C Construction aterials Engine BORING F	RITY () Department Division Pering Sect REPOR	FRIVE Ion C		_4/SU (pages)
ROJECT		······			NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
1-120	3CT 3R	E GAMB	LE		CRAI	ය		CONTRACT NO	DATE
AS	MARKE	DOUT IN	THÉ	FIEL	0 31 K,	Ilian 1	95502.	426-94-506	11-3-00
POON	72/	CASING S	ZE HOLE	TYPE			GR	DUND WATER LEVEL	
S C	SAFETS	HAMMER	ers		Date	Timo	Depth	Re	marks
140 .	FALL ?	SU .	# FALL	-	11/3/00	2:45 p	4.5	InS#3	
RILLER	<b>?</b>	Pennell							
SPECTOR		Duder							
CASING .OWS/FT.	DEPTH	SPOON BLOWS/6"	RE- ' COV'D	SAMP. ² NO.		3SA LI	MPLE DE NE LOCA	SCRIPTION AND REMARK TES CHANGE OF PROFIL	(S E <i>).</i> (
		HAND	Full		L		SHED		
		AUGER		1	Fill- dark -	LTUD - C-	5 SAN	D. Some Gravel, tr	ove Silt 2.
		1			Fill. Low	- c-f	SAND.	trace Gravel In	silt
				Ζ		0	/_		
			<u> </u>		<u> </u>		 c A		
-	- 5 -			3	<u> <u> </u></u>			<u>^c</u>	
ans		12-13	24"	. 1	Fill dark	= trin	<u>-</u>	SAND, from Grave	el treis Silt
-j		20-27		4			- — -		
		5-5	ay"	_	Firl- bo	on c-f	<u>SAN</u>	> trace Gravel	frace Silt_
	- 0 -	7-19		>					
		6-18		1	fill-Bro	ion M-F	Sal :	t Siet to Grave	l
		21-20	18"	6					
		10 - 12	1-1	_	Sam	re —			
$\sqrt{2}$		15-15	20"	7					
¥.		()=[S	0						
	► 15 <	>-7	1	8	san	<u>u</u>			&
		10-12	<i>1†</i>						16.01
		· · · · · · · · · · · · · · · · · · ·							~
					· 		Bo	Tomor Bonning	
								D I	· ·
								· · · · · · · · · · · · · · · · · · ·	
	- 20 -								
					fote: Z	2-16 2 ample	jaro_	¿ Ivoa jan	ros swedfor
									- <u> </u>
	_		1						

			Ma	Engineering Department Construction Division	
				BORING REPORT	
PROJECT P'É	G			NAME OF CONTRACTOR BORING NO.	SURFACE ELEV.
LOCATION As laid out h	Killan	25.02		CONTRACT NO. 426-99-006	DATE / 4/00
SPOON 3 .0.D. 21/8	U CASING SI "I.D. H.S.Q	ZE HOLE	TYPE	GROUND WATER LEVEL Date Time Depth	Remarks
HAMMER 140 # FALL 37	HAMMER	# FALL		11/4/00 11:57 m 40' In Sta	t
J.C.	uj				
NSPECTOR T.R.	<u> </u>			-	·
	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	3SAMPLE DESCRIPTION AND REMA LINE LOCATES CHANGE OF PROF	rks ile o.
	Hand	full	) -	Till-Rom-EC Day Confitte Cub	Ra and Barry
		Nec.		Same	3.0
		-/	L	Fill - Re Br clay Silt little G.	ti Cindero
			3	Michil- Ke M Clay Silt, some on filone,	40 md, 4pm
sero -	6-9		U	Gray FSand to Plant fibers	
- <b>`</b>	9-8 7-4	<i>  </i> ''		Bom-FS Dt Sit to Gual	
	4-5	12''	5		
/0	2-4	1.11	6	Son	
	<u> </u>	10"	4	Lanu	
	11-11	24"	+		
	3-5	10"	8	Same	
				····· ····· ····· ····· ····· ···· ·····	
<b></b>				Note: Sample #1-5 were	aved for
<b></b>				environtenting to per C Springer (K	illo Osson)
				all other somples were screened in)	MUL Then_
■					

THE PORT AUTHORITY OF NY & NJ

• •

2 - U = undisturbed; A = auger; OER = open end rod; V = vane, 3 - Log depth of change in color of wash water, loss of water, artestan water, sand heave in casing, etc.

2

<u>PA 547</u>

, **`** 

•

PA 547			П	ie poi	<b>RT AUTHORITY</b> OF	NY&N	J		
6-90					Engineering Department				
-				Ма	Construction Division	on			
						, ,			
								SHEET ) OF	2
PROJECT					NAME OF CONTRACTOR	B	DRING NO.	SURFACE ELEV.	2
Howlow	Nove Nove	Port Thory	Ptc	م آرا	Craig Prilling		RR-5	1	
LOCATION						C	DNTRACT NO.	DATE	
Floo	South of	- Olde de		Block	1400 LoT/		176-79-006	11/100	
SPOON	34	CASING SI		1	Date Time	GROU	ND WATER LEVEL	Remarks	<u></u>
HAMMER	5.0.012 Cafat	HAMMER				Doput			·
140 .	FALL 30		FALL		11/100 77	20	Lehilo Havo	1 Accering	
DRILLER		S D			75			ð	
INCREATOR		> BUINS					·····	<u></u>	
INSPECTOR	L	Mare							
CASING		SPOON	RE- 1	SAMP.2	3SAI	MPLE DESC	RIPTION AND REMAI	RKS	
BLOWS/FT.	DEPTH	BLOWS/6"	COA.D	NO.	LIN	E LOCATES	S CHANGE OF PROF		<u>v.0</u>
Mard		Hand Auger	Full		Fill-Cinders			·	0
Aug			1	/	Fill- M-F Brow	1 Saud	TV Europen Th	SulT	
	<u> </u>							<u> </u>	
	┝	+		2				<u> </u>	~
├ <u></u>	╄	<u>↓                                     </u>		<b> </b>	Same		······································	78	0_
	5			2					
-V					X-P Brown Sal	vol. Tr	SIT		
lley		2-2							
GTan	<u>}</u> − −	3-7	1.11	19	ER. S. /	Troit			
) JJ m	<u>}</u>		<u>, ' '</u>		1- UPOUN JANA	11 311			
Hyp	┣─ ─	6-0	<b> </b>	5					·
		10-13	224		Same	<u></u>			·
		10-13	[	1					
		20-24	24"	6	Samo				
		101-13	- <u> </u>						. <u></u>
			0.11	7	└── <i>─</i> ───────────────────				
	┟── ──	16-11	04'	<u> </u>					
		5-7	L						
		8-12	74"	ک ا				1	br.
				<b> </b>		کنار بدین است بر پیدید است ا		1	) 1
	╂╌╸ ──	ć	<u> </u>	1			R.T. J.D		′
ļ	┝─ ─	<u> </u>	ļ	4	└── ── ·── ·── ·		HO DOMOT DO	ING	. <b></b> -
ļ	┟── ──	<u> </u>		1					
	20-			ļ	DII Samo	<u>les ch</u>	lockal with	PIO NOTOL	
	r a				5# 1-2	Sala	for TosTing.	5# 122	
	<u>├</u>	<u> </u>		1	R. Tota	P		ale Asourtel	,
}	┢─ ─	<u> </u>				1_ rom	awing samp	10000 CON (0)	
		<b> </b>	ļ	ł	└── ── ── ── ── <i>──</i>				
	┞	ļ		]					
	1 ac								
L	► 40 <	•	•	L					~~~~~

• •

•

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

# THE PORT AUTHORITY OF NYG RU

Engineering Department Construction Division Materials Engineering Section **BORING REPORT** 

										SHEET / OF 3
PROJECT	1 14 1	×	~		NA	ME OF CONT	ACTOR		BORING NO.	SURFACE ELEV.
Houland	1 Nouk,	Port Ivory	Ptes	5,50	L	Craig	UnIlla	y	<u>KK-6</u>	
LOCATION			2			, <del>- 0 -</del>		· T	CONTRACT NO.	DATE
South	Side	Bldg #70	<u> </u>	lock 1	100	1 LaTI			426-99-006	11/6/00
SPOON	~3,		E HOLE	TYPE				GROL	JND WATER LEVEL	
D. C	).D. 0 %	"I.D. HUAR	1 1	2		Date	Time	Depth	<u> </u>	Hemarks
140 #	<del>آمارکر</del> FALL 30	7   TAMMER 	FALL			11/1/0	35	6.0	s [≠] y	
DRILLER	5	5 Burns					-			
INSPECTOR		Dowe								
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	'  '		3SAI	MPLE DES	CRIPTION AND REM	IARKS DFILE 0.0
Vand	$\sim$	Mardon	1-11		$\frac{1}{c}$		100			
riano	┝── ──	our might	1011		H		- n		· ·	<u> </u>
Hupr		<u>├</u>		ļ	ļŀ	<u>-11- M-</u>	+ Brown	Sand L	ITTO Cudors	1 <u>r Silf</u>
				<b>_</b>			_	·	/	
					E		> Ru	5.11	The las The	it tran
╞──╂──┤	<b>┝</b>	<u>├ </u>			+'			- que l	- man IN	· · · · · · · · · · · · · · · · · · ·
├	5-	ŀł	└─╂──┤	3	-					·
			¥		1	Same	2			6.0
<b>M</b> u		1-0							_	
SIM		(1 - 1)	201	1 4		MEL	2	J TU C	 זר	
N.	┞ <b>───</b>				╉──	_ [	www Jan	<u>" </u>	<u>ч —                                   </u>	
picons	<u> </u>	1-4		5	$\vdash$					
		3-10	211		L	<u> </u>	e			
		10-8		,						
		10-20	2-21	6		NER	~ ~ ~	17		
- -		6-111	<u> </u>	<b> </b>	┫	_ <u>[ ×1 – [ – ]</u>	10 <u>WN 7</u> d	vor 1r 1x	- ury say	<u></u>
┝╌┟╌╌╎		5-17		17	<u> </u>		<u> </u>			
		24-40	241	/		M-F!	Your S	and, T	FSIT	
	. )	9-10		C-	1					
	► 15 <	16-201	19.	8	┢─					16.00
┝		10 00		<b> </b>	┝					2
				ł	┣_					
				ſ	L				BoDomof	Baring
					[					0
				ł	[	<u>k</u>	711 Sam	Alar a	hat II. Th	PID Mata
└──── <b>─</b> ┦	▶ २८◀	▶		ļ	$\vdash$	<u></u>	#1-2	s d	Cut 1 T	
		┝ <u>-</u>			┣			Javas	tor ENUITO 125	Ing ONLY 102 X
				ļ		!	30 T-sta	<u>I k</u>	mapping Sam	Mas Piscardod
								1	0 - 7	×
					$\vdash$					· ······ ······· ·······
			{		┝					
-	- 25				L			<u> </u>		
-	NOTE	S: 1 - Length reg	overed: 0	)" — Loss	s of	Samole, T	Tran used			

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

-

## THE PORT AUTHORITY OF MYG NJ

....

Engineering Department Construction Division Materials Engineering Section **BORING REPORT** 

					SHEET	DF_}_
PROJECT	1				NAME OF CONTRACTOR BORING NO. SURFACE ELEV.	
Noula	and No.	ok, Ports	wory Po	GSIJe	Cracy Vrilling FR-7	
LOCATION	1 -	7	) =	<u> </u>	CONTRACT NO. DATE	
Hs Lan	d ovsh	y Fillom	Hosa	c Eas	5 Side Bldg 20 426-99-006 11/6/a	2
SPOON	~ 31		G SIZE   HO		GROUND WATER LEVEL	
	D.D. 078	("I.D. 0"		<u> レ</u>	Date Time Depth Remarks	
140 #	FALL 30	·/7 ,	# FALL		11/6/00 1 pm 6.0 Stry	
DRILLER	•	S BUKIS				
INSPECTOR		P Now	0			
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6	RE- COV	D SAMP.	² ³ SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE	00
Hand		Hand Au	cor Ful	/		
A	<u> </u>				FIL-M-F Bran Co. 11 T. C.F. Tre	
TUGY	┟─ ─				ITO WA SONA LI 16 211, 1100000	
		- <b>  </b>		- 7		
					Fill MF Brown Sand, 4TTIO Gravel TISill, Tr Clay ch	Ings II
						71.
	▶ 5 ◄	┫───┤,─		- 3		
V.			V		M-F Drown Javor, TrSilly IrGraw/	•
and the second s	L _	11-10				
STAM		8-8	15	, 7	Same	8.0
Auas		6-4		1		
g-		9-13	22	<u>, 5</u>	F Braunch Gen Sand Some Silf Trans	0.0
	<b>•</b> 10 •	2-11			· Uround - y and arrive still fir brauge	
	┝	17-19	2	76		
	┝─ ─	11-17			- [ Prown Sandy It Silt	•
	L	5-17				
		25-29	24	1 /	Same	
*	·	11-17				
	► 15 -			$\neg \land$		1/
<u>```</u>	┝────	18-39	1 24		Same	16.0
						1_
		]			Reitten of Reining	
		+			BUS I LI PIDIO	<u>-</u>
	► 20 -	┫			JI samples creckal with Meter_	
	L				ST 1-3 Saud for Ta Drive, S'It To	to Ty
					Ramaladay Samala Discorda	
	— —	1				
	 	<u> </u>			<u></u>	
	25		\			
	NOT	ES: 1 - Lena	h recovere	d: 0" — Los	ss of Sample, T — Tran used	

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

THE PORT AUTHORITY OF MY & MJ 6-90 **Engineering Department Construction Division Materials Engineering Section BORING REPORT** SHEET OF 3 ۱ NAME OF CONTRACTOR BORING NO. SURFACE ELEV. PROJECT PD -1 P ľ Port Craig drilling LVOry CONTRACT NO. DATE LOCATION Block 1400 Lot 426-99-006 Bldg <u>+</u> 30 31 11-20-00 /11-21-00 HOLE TYPE SPOON CASING SIZE **GROUND WATER LEVEL** 3 Augers Date Time Depth Remarks "O.D. "I.D. HAMMER Autom, HAMMER 5.5' # FALL 11-20-00 12:20pM SAMPLE # 3 140 30 # FALL DRILLER raig raic INSPECTOR arks SAMPLE DESCRIPTION AND CONTRACT OF PROFIL CASING SPOON SAMP. RE-DEPTH BLOWS/FT BLOWS/6" COV'D NO. . CONCrete Full. 0.7 Hand ange Handang Gravel , tr Sill SAND little Fil brown Ť SAME 5 3 SAME 4 M ŋ AUGERS SAME 10" 4 3 Z 5 <u>2</u>Å Fill Gravel Sill SAN brown C D little te -5 10 ų 9 14 SAND 6 bravel 12. ٩ 9 24" Red 13-13 Siltu brown 15 for testing Note: 2 samples saved All other samples screened with PID meter 2 dis cardeal of Boing Bottom 20. NOTES: 1 -- Length recovered; 0" -- Loss of Sample, T -- Trap used 2 - U = undisturbed; A = auger; OER = open end rod; V = vane

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547

PA 547 6-90

# THE PORT AUTHORITY OF MY & NJ

Engineering Department Construction Division Materials Engineering Section BORING REPORT

		SHEET OF 3
ROJECT DIO	NAME OF CONTRACTOR BORING NO.	SURFACE ELEV.
ort Ivory 1 & G	Craig drilling PD 3	DATE
+9514 ABIDO 77 A Block	$\frac{1}{2}$	6 11-28-00/1.99
POON CASING SIZE HOLE TYPE	GROUND WATER LEVEL	
3 .0.D. 2 3/8-1.D. Augers 1	Date Time Depth	Remarks
IAMMER Safety HAMMER		
AO # FALL SO # # FALL	11-28-00 2.21 1.0 Sample #	: [
A.Kides		
NSPECTOR 7 7		
J. Larks		
LOWS/FT. DEPTH BLOWS/6" COV'D NO.	LINE LOCATES CHANGE OF PRO	DFILE 0.4
utter Head Cutter Head Fuli Rec	CONCRETE	0.
andouger	THE CANA COLOUR	
<u>_</u>	prill grey c-f salver or brever, Lobb	les, tr. 5111
	Itill grey & brown c - I' SAND som	e browel little Sill
		,
	Reddich brown Silly CLAV come (	SAND L. G.
		· · · · · · · · · · · · · · · · · · ·
14-20 12 4		
21-27 8" 5	SAME	
19 16 94" 0	SAME	
14-20 24 1	SAME	14.
	Note: 2 samples saved for testing	
	All samples checked with PID me	er /
	line land	Bottom of Bori
	eniscardea	
—— <b>)</b>		
		······
		· · · · · · · · · · · · · · · · · · ·
	<u> </u>	
	]	

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

.1

٤.

PA 547			TH	ie po	Drt Authority of NY& Ru
6-90					Engineering Department
_		، م		м	Construction Division Jaterials Engineering Section
				141	
					SHEET J OF 3
PROJECT					NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
Part	lvaru	PEG			Craia drilling PD 4
LOCATION	J				CONTRACT NO. DATE
20'W	<u>of BH</u>	PD 4 A 5 25	N of BI	dq 36	(NW corner) Block 1400 lot 1 426-99-006 12-2-00
SPOON	1	CASING SI	ZEHOLE	TVPE	GROUND WATER LEVEL
	D.D. 2	78 "I.D. Augel	rs I		Date Time Depth Remarks
	FALLO	3.0.7	# FAL1		19 9 00 112 PM 6.0' Ton sounds # 4
DRILLER	1 I		TALL		
	A K	lides			
INSPECTOR		7 1			
	<u> </u>	Larks			
CASING BLOWS/FT.	DEPT	SPOON BLOWS/6"	RE- 1	SAMP.	² SAMPLE DESCRIPTION AND REMARKS
Handauger	<b>•</b> •	Handauger	Full Rec		
	<b>}</b>		┟╌┾╌┤		Fill grey c- 2 SAND & Gravel some Cobbles tr Sill 10"
		_		1	Red - brown c- I SAND with silty CLAY some bravel. 20
	}				
	<u> </u>			9	P $H$ $L$ $C$ $H$ $C$ $H$ $C$ $A$ $V$ $C$ $L$ $C$ $A$ $V$ $A$ $C$ $A$
	<u> </u>				- Redaish - brown DITH UNT some C
	5	◀		i	
4				3	LSAME
		1 _ 9			
WICERS	<u> </u>		0_H	1.	SIMC
		3-5	10	7	
		3-2		_ *	*
	L 10	2-2	24"	_5	SAME 120
	F	7.9			
		11-2		1_	PILL CINCY SIIT
	<b> </b>	4-5		<u> </u>	negaish - brown Cinici
	<u> </u>	3-9		• .	
		9-16	2.0	7	SAME 14.0
				1-	Note a comple so year the testing All cleaked w PID wety ?!
	- 15				now: sumple survey for testing all carefully
	<b> </b>			•	Remaining Samples discarded
ļ	L_				
					Batton & Boring
			┟┥		
			┟────┤		
	L				
				l	
	<b>—</b>		<b>├</b>		
	<b>L_</b>				
				# 1 m	ee of Complex T Tree wood

•••

S: 1 — Length recovered; 0" — Loss of Sample, T — Trap used
 2 — U = undisturbed; A = auger; OER = open end rod; V = vane
 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

ï

------

•

٦

r	A	3	•7
	6	90	)

# THE PORT AUTHORITY OF MY & NJ

Engineering Department Construction Division Materials Engineering Section

## **BORING REPORT**

-					SHEET 1 OF 2
PROJECT	010		NAME OF CONTRACTOR	BORING NO.	SURFACE ELEV.
Port Ivory	1 4 G		Uraig drilling	PD 4A	
LOCATION		,		CONTRACT NO.	DATE
2 W of BH	PD 4 BBlock 1	400	Lot 1	426-99-006	12-2-00
SPOON Hand auger	CASING SIZE HOLE TY	PE	GR	OUND WATER LEVEL	
*O.D. J	"I.D. Handauger )		Date Time Deptr	n Re	marks
HAMMER	, HAMMER J				
# FALL	* # FALL				
DRILLER A.Kid	es				
INSPECTOR 7 7	-kc				
CASING J. CO	SPOON DE 1 e				(5
BLOWS/FT. DEPTH	BLOWS/6" COV'D	NO.		TES CHANGE OF PROFIL	۵.۵ E
Cutter thead	cuter Head Full Rec			AS DNAIT Concrete	0.7
Handenger	HANDAUGER				
<u> </u>	┟───┼───┼──┝━	<u> </u>	till greyish - black C-f.	_SAND tr bravel,	Ŀ Sill
		0			
		h	DAME		3,5
	· · · · · · · · · · · · · · · · · · ·				
			Note: Obstruction at	3.51 A pipeline in	/+ BH. 5_
			Move + 2'East		
	<u>+</u>		· · · · · · · · · · · · · · · · · · ·		
				U	occom of Doning
	<u> </u>		·		
	<u> </u>				
			]		
	<u> </u>		┝ <b></b>		
				· · · · · · · · · · · · · · · · · · ·	
			·		
			[		
┝───┣  ◀			· · · · · · · · · · · · · · · · · · ·		
┝───┼── ──	┝		└── ── ~─ ·── ·── · ·		
<b>├───</b>	┟╍╍╍╍╍╍╌┥		└── ── ── ── ·── ·		
┝───▶ ◀			· · · · · · · · · · · · · · · · · · ·		
					Х. Х
┝───┼─ ──			L	, 	
	<u>├</u> ────┤				- ;
			<u> </u>		
NOT	<b>10</b>				

1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

BORING REPORT      BORING NO,      BORING	OF 2 IV.
Construction Division Materials Engineering Section BORING REPORT FROMEET FROMEET FROMEET FROMEET FROMEET FROMEET FOR EVENT FOR EVENT	οF 2 
BORING REPORT BORING REPORT SHEET I BORING NO. BORING NO. BORING NO. BORING NO. BORING NO. BORING NO. BORING NO. BORING NO. BIOWSON CONTRACT NO. BIOWSON CASING SIZE HOLE TYPE SPOON CO. 2 %8 'I.D. AUQUYS I HAMMER AUGOT HAMMER AUGOT INSPECTOR A. Kides INSPECTOR COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN BLOWSON COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN COVEN	OF 2 IV.
BUCHING HEPOHI BORING NO BORING NO BORING NO BORING NO BORING NO BORING NO BUCKTION J COSING SIZE LOCATION J COSING SIZE HOLE TYPE SROON SROON SROON SROON CO.D. 2. 3/8 *1.0. AUG25 HAMMER ALL Date Time Depth BLOWSRE CONTRACTOR CONTRACTOR DATE J COSING SIZE HOLE TYPE GROUND WATER LEVEL BANDER ALL DATE CONTRACTOR CONTRACTOR DATE CONTRACTOR DATE CONTRACTOR DATE DATE DATE CONTRACTOR DATE DATE DATE SUMPACE ELL SROON SPOON SPOON SPOON SPOON SPOON SPOON SPOON CONTRACTOR DEPTH BLOWSRE COVO NO. LINE LOCATES CHANGE OF PROFILE CASING BLOWSRE CONTRACTOR DATE CASING SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON SPOON S	OF 2 EV.
PROJECT PROJECT PROJECT PORT Ivory P & C LOCATION + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 + 2.0' N of Bidg 3.6 (cents, Block 400 lot 1 - 2.0 0.0 (cents, Block 400 lot 1 - 2.0 0.0 (cents, Block 400 lot 1 - 2.0 0 (cents,	<u> </u>
Provention     Pert Ivory     Pert G     Inner to contraction     Burnation     Southact no.       PD 4 B     Incashe size     Block     Aco     Incashe size     Block     Aco     Date       + 20' N of Blog 36 (center)     Block     Hole type     GROUND WATER LEVEL     Date       Spoon     2 % 10.     Augers     Incashe size     Hole type       3     Costing alors     Incashe size     Incashe size     Incashe size       HAMMER     Augers     Incashe size     Incashe size     Incashe size       Hammer     A Kides     Incashe size     Incashe size     Incashe size       Hammer     A Kides     Incashe size     Incashe size     Incashe size       BLOWS/FT     Depth     Blows/set     Covrb     No.       Casing BLOWS/FT     Depth     Blows/set     Covrb     Incashe size       Blows/set     Covrb     No.     Inte Locates change of PROFILe       Casing     Blows/set     Covrb     No.     Inte Locates change of PROFILe       Casing     Blows/set     Covrb     No.     Inte Locates change of PROFILe       Casing     Blows/set     Covrb     No.     Inte Locates change of PROFILe       Casing     Inte Locates change of PROFILe     Intill greyish-brown c f SAND & Grav	20 
LICRTING LIVERY 1 Y U LIFELIG DITILING CONTRACT NO. 1 20' N of Bildg 36 (cente) Block 400 lot 1 426-99-006 12-2-1 SPOON 2 Casina sze hole type GROUND WATER LEVEL 3 "O.D. 2 3/8 "LD. Augers 1 Date Time Depth Remarks HAMMER Aukorn HAMMER AUKOR PEALL Date Time Depth Remarks 1 AD FALL 30 "FALL ODITION AND REMARKS DATE LEVEL BOOMS FOR THE SAMP.2 SAMPLE DESCRIPTION AND REMARKS UNRE LOCATES CHANGE OF PROFILE CALLS Head Full Handonger Hand Full And Source SCHANGE OF PROFILE SCHANGE OF PROFILE SCHANGE OF PROFILE SCHANGE OF PROFILE CALLS Head Handonger Handonger Source SAND & Gravel +tr. Silt SAMPLE DESCRIPTION AND REMARKS SCHANGE OF PROFILE SCHANGE OF PR	20 
Image: Second State Sta	20 
10     10     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000     1000 <td< td=""><td></td></td<>	
3     O.D.     2     3     No.     Date     Time     Depth     Remarks       140     #FALL     30     #FALL     -     -     -     -     -       000     #FALL     30     #FALL     -     -     -     -     -       010     #FALL     30     #FALL     -     -     -     -     -       010     #FALL     30     #FALL     -     -     -     -     -       010     #FALL     30     #FALL     -     -     -     -     -       010     #FALL     30     #FALL     -     -     -     -     -       010     #FALL     -     #FALL     -     -     -     -     -       010     #FALL     #FALL     -     -     -     -     -     -       010     #FALL     #FALL     -     -     -     -     -     -     -       140     #FALL     #FALL     #FALL     -     -     -     -     -     -       160     #FALL     #FALL     #FALL     -     -     -     -     -     -       160     #FALL     #FA	<u> </u>
HAMMER Autorn IAn #FALL 30 #FALL DRILLER A. Kides INSPECTOR J. Zarks CASING BLOWSKT. DEPTH BLOWSK' COV'D NO. Cuttor Head Handauger Handauger Handauger Handauger Handauger Handauger 1 Till greyish-brown c f SAND & Gravel with Sifty ( 2 Reddish - brown C - J SAND & Gravel with Sifty ( 3 Reddish - brown SITTY CIAY some Gravel & c. CONCRETE SIAB 10 10 Refusal Bottom of	
140     # FALL     30     # FALL       DRILLER     A. Kides       INSPECTOR     Zarks       BLOWS/FT.     DEPTH       Station     Train       Stat	<u> </u>
DRILLER A. Kides INSPECTOR INSPECTOR Casing BLOWS/FT. DEPTH BLOWS/6" Cutter Head Handauger Handauger I Till greyish-brown c f SAND & Gravel +tr. Silt Cutter Head Handauger I Till greyish-brown c f SAND & Gravel +tr. Silt 2 Reddish - brown c f SAND & Gravel with Silty 1 2 Reddish - brown SITTY CIAY some Gravel & c. CONCRETE SIAB I O I O I O I O I O I O I O I O	<u> </u>
A. Kides INSPECTOR J. Zarks CASING BLOWS/FT. DEPTH BLOWS/6" COV'D SAMP? Cuites Head Handowger Handowger 1 Fill greyish-brown c f SAND & Groyel +r. Silt 2 Reddish - brown c f SAND & Groyel +r. Silt 5 Reddish - brown c f SAND & Groyel with Silty 1 5 Reddish - brown SITTY CIAY some Groyel & c. CONCRETE SIAB 10 Refusal Bottom of	<u> </u>
INSPECTOR Zarks CASING BLOWS/6" BLOWS/6" COVD NO. LINE LOCATES CHANGE OF PROFILE COVD Hendowger LINE LOCATES CHANGE OF PROFILE Cutter. Head Handowger LINE LOCATES CHANGE OF PROFILE LINE LOCATES CHANGE OF PROFI	<u> </u>
CASING BLOWS/FT. DEPTH BLOWS/FT. DEPTH Cutter Head Handowger 1 Till greyish-brown c f SAND & Gravel , tr. Silt Cutter Head Handowger 5 2 2 Reddish - brown C - f SAND & Gravel with Silty 1 2 Reddish - brown SITTY CIAY some Gravel & c CONCRETE SIAB 1 0 Refusal Bottom of	0.0 0.1 
BLOWS/FT. BLOWS/FT. BLOWS/6" BLOWS/6" COU'D NO. COU'D NO. COU'D NO. COU'D NO. LINE LOCATES CHANGE OF PROFILE LINE LOCATES CHANGE OF PROFILE COUNCACETS CHANGE OF PROFILE COUNCACE OF PROFIL	
Cetter Head Cetter Head Handowger Handowger 1 Fill greyish-brown c & SAND & Gravel , tr. Silt 2 Reddish - brown C - J SAND & Gravel with Silty 1 2 Reddish - brown SITTY CIAY some Gravel & c CONCRETE SIAB 10 Refusal Bottom of	0.0 0.7
Handouger Handouger Handouger 1 Fill greyish-brown c & SAND & Gravel , tr. Silt 2 Reddish-brown c & SAND & Gravel with Silty 1 5 7 7 8 Reddish-brown SITTY CLAY some Gravel & c. CONCRETE SLAB 10 Refusal Bottom of	- <u></u>
1     Fill greyish-brown c. f SAND & Gravel , tr. Silt       2     Reddish-brown c. f SAND & Gravel with Silty (       5     3       8     8       10     10	·`
2 Reddish-brown c-f SAND & Gravel with Sifty ( 5 3 Reddish-brown SITTY CLAY some Gravel & c. CONCRETE SLAB 10 Refusal Bottom of	
2 Reddish-brown C-f SAND & Grovel with Silfy 1 5 3 Reddish-brown SITTY CLAY some Grovel & c CONCRETE SLAB 10 Refusal Bottom of Refusal Bottom of	
Keddish-brown c-f SAND & liravel with Silty 1       5     3       8     3       8     8       10     10	MAL T
5 7 8 8 10 10 10 10 10 10 10 10 10 10	JIAY
Boltom     Boltom	
Image: Neddish - brown Silly UIAY some Gravel & c       Image: Concrete Slab       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Concrete Slab       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c       Image: Neddish - brown Silly UIAY some Gravel & c </td <td>D CAND</td>	D CAND
CONCRETE SLAB	1 STING
10 Refusal Bottom of	<u> </u>
10 Refusal Bottom of	2
10 Refusal Bottom of	$\neg -$
10 Kefusal Dottom of	<u>/</u>
	Doring .
	Ŭ
$ \begin{tabular}{cccccccccccccccccccccccccccccccccccc$	
┝╼╍╌╼╍┼╍╴╶┥╼╍╴╼╸┥╴╴╸╸╸╸╸╸╸╸╸╸╸╸╸╸╸╸	
┝───┼╸ ┽────┼──┤  ┝─╴── ── ── ── ── ── ── ── ── ──	
┝	
┝╾───┼╸ ╺┼────┼╼──┤ ╵ ┝╼╴╼╸╺━╸╺━╸╼╸╺━╸╼╸╼╸╼╸	

.

١

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

## THE PORT AUTHORITY OF NY & NJ

5

Engineering Department Construction Division Materials Engineering Section BORING REPORT

							SHEET   OF Z
ROJECT		~	1 0				NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
Port Iv	oru	P	专日				Craig drilling PD-5
OCATION	J						CONTRACT NO. DATE
\$ 10 W of	Blda .	43 A	2±50	'S .	INV	N corner	urbld. 40 lock 1400 Lot 1 426-99-006 12-02-00
SPOON T	<u> </u>		CASING S	ZE	HOLE	TYPE	GROUND WATER LEVEL
<u> </u>	2 ^{_3} /8	ζ "I.D.	Auge	rs		1	Date Time Depth Remarks
AMMER			HAMMER				
140 #FA	LL 30	<b>)</b> '		<u># FAI</u>	.L	#	12-2-00 9:32 7.0' SANDE# 4
DRILLER A							
<u> </u>	nides	)					╡ ┟╼╴┈╾┟╼╴┈╼╴┟╸┈╸╼┈╼
SPECTOR	7	1					
	<u>, La</u>	rKS		т			
	DEPTH	BL	OWS/6"		יב- עיע	SAMP.	LINE LOCATES CHANGE OF PROFILE
utter head	0.4	Cutte	y Head	Fu	1		CONCRETE
Handauger		Har	<u>idauger</u>	ļ			Ett around the proved Cobbles to Sill
				ļ		1 7	Plathan Silvy Clay Etter C C SAUD & Grand
				1			The survey have survey and the survey
				+			
						21	* Red-brown C- [ SAND & Gravel with SILLY CLAY
			1				
►	5 ◄						
<u>+</u>			¥			3	J SAME
		I۲.	- 5				
EM				1	, 11		
Pagers		110 -	- 10	!	4	4	I readish - brown Dilly LIN/
		8_	- 13				
			19		" م	5	Redded Leave CINCA
<b>&gt;</b>	10 4		- 1.	†	ю		- readish - brown Silly Silly
		ļ		· [			Note: 2 samples saved for testing
				ł			All checked w. P.T. D meter.
		<u> </u>					<u>Kemaining</u> <u>Samples</u> <u>aliscardion</u> <u>R</u> U. C. D
· ·		ļ		<b>_</b>			
l	. –						
	154			1-			
				┨			
	- —						
		<b> </b>		4			
►				†			
		1					
		t		1		T.	
<u> </u>				╂			
	_	1		1			
		[		1			
				1			
		- A. 4				W F.e.	as of Comple T. Tran used

PA 547 6-90 PA 547 6-90

)

# THE PORT AUTHORITY OF MY & NU

Engineering Department Construction Division Materials Engineering Section

## **BORING REPORT**

			-								SHEET OF 3
PROJECT		<u>.</u>	~ 1 /	^		N	AME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
1 Po	srt I	.VOI	ru Pell			1 (	Crain dr	illina		F2-2	
LOCATION			3-1-	¥		-				CONTRACT NO.	DATE
+ 75 "	12 W	BID	a 31 Block	1400	lot	1	-			426-99-006	- 20 = 00
SPOON		<b>Cursa</b>	J CASING SI	ZE HOLE	TYPE		Į		GRC	OUND WATER LEVEL	
3 0	.D. 2	3/8	".D. Auger	s I	1		Date	Time	Depth	Rei	marks
HAMMER	Saleti	U	HAMMER								
140 #	FALE	J 30		FALL	-		11-20-00	10:45	7.0	5#4	
DRILLER	C						+				
1 9	. Lro	aig					· · ·				
INSPECTOR	, – – –	7	1								
	<u> </u>	_ <u>_</u> _	arks	· · ·							
CASING	J		SPOON	RE- 1	SAMP.2	2		3SAI	APLE DE	SCRIPTION AND REMARK	S
BLOWS/FT.		ਸ਼⋠	BLOWS/6"	COV.D	NO.	_		LIN	E LOCA	TES CHANGE OF PROFILE	
Gutter Head	0	1	Cutter Head	rul		+-	Fill brown	C. T SAN	Lonc	nuel tr.Sill	<u>Q.P</u>
- and and -		-†	1				T-11 1		0 C A A	NICIIO	: 1-1
		-+	<del> </del>	<b>  </b>		┢	_till_br	ewn C-	<u>t jan</u>	ly tr bravel, tr. S	
					*	Ί			+ . 		
		-			2		TA 2	1E -		· ·	
- + - +		-+			~	╋					
	2				5		SAN	AF -		· · · · · · · · · · · · · · · · · · ·	
		-+		Y		┢		<u> </u>			
EM			4 - 4								·
Augers			67	1811	4	5	RIA L.			C CAND L (	Sugar L Sill
		·-+		10		<u>,                                    </u>		w <u>n</u> ey	rey i	C- J-DEILY JEIL	
			11-11'		× ~ ×	٩					
			19.90	24"	5		Fill or		0 9A	ND tr. Gravel	$1 \leq  T $
<b> </b>	► 10	•				ᡟ	<u> </u>	g =	<u>}⊻'-</u>	<u>, , , , , , , , , , , , , , , , , , , </u>	- 14
Jł		_	11 - 12		1						
			20-47	14"	6		SAME				
			a			T					
<u>├──</u> ┤		-+	<u>B-20</u>			┣			·		
			26-30	24"	7_		<u>_5AME</u>				
		. T.			A '		SAME				14.5
┝╾┼╾╼┩	▶ 15		1-0	1	k O	F	ō				
		_4	0-13	24	OB		<u>Seddish</u> -	- brann	Silly	CIAY with tr. Br	avellea
									J		<b>A</b>
<u>├</u> ────┤		-+				┢	···· ····· ····				<u></u>
┝		-+-	~, 			$\vdash$	Note: 29	iam ples_	saved	for resting	///
		- 1					Se 1	on hold.	. All ~	they samples	Bottom of Be
		1						لىرەنىغ «ئىتىتىدى». 1	<u>م کر</u>		F
	► ೩۵	∍◀				$\vdash$	<u>Sc</u>	recored u	<u>. [] _</u>	meter y discardes	
.3											
			*			Γ	••••				
}		-+				$\vdash$					
						L					
		T				Γ					
		-+				⊢					
											·
-		~		465							

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

*

<u>547</u> 90			TH	I <b>E POI</b> Ma	<b>Engineering Departmen</b> Construction Division terials Engineering Sec BORING REPOR	に nt ction て に に に し に の の の の の の の の の の の の の の の	RU	SHEET OF 3
ROJECT	A. 0	110			NAME OF CONTRACTOR		BORING NO. #7	SURFACE ELEV.
NOTION	y P71	20			inaig		CONTRACT NO.	DATE ) /
Leid out	tinsile	Red 20 in	Pit (	South	End) (Black-1401	•)	426-99-006	1/1/100
POON 3 "0	~ z1/8		ze hole	TYPE	Date Time	GR0	DUND WATER LEVEL	lemarks
AMMER	Safer	Y HAMMER			Julula 10 00			1.0
<u>140 #</u>	FALL 30	·	FALL		11/11/00 120	0 6.0	while Nav	d Higorday
MELEN	0.00	uch						、
ISPECTOR	$\overline{D}$	lau.a						
		SPOON	RE- 1	SAMP.2	35	AMPLE DE	SCRIPTION AND REMAR	KS
.OWS/FT.		BLOWS/6"	COV'D	NO.		LINE LOCA	TES CHANGE OF PROFI	
						f Flo	or in side Bldg	— — <u> </u>
		<u>.                                    </u>			Hole in IN	spection	<u>_ RT FlouF</u>	<u> </u>
								, ·
	$\sim$							c
					CONCRETE			/.
nd		Hand Aug	Fill	1	Fill-M-For	NSC. Br	own Sand, LITLe	Sit 2
Augor				2	M-F Black E	Brown Sa	nd LITTIE STILL TH	Upor 31
Ů								· · · · · · · · · · · · · · · · · · ·
				3	F Brown Sa	nd Tr	SIT, Tr Var	
								· · · ·
1/			1	1	Samo			
fucors		13-24			×			
1		29-28	244	5	Samo	·		
		16-15		,				
	 ,	15-16	194	Ð	Samo			
		· · ~	<b>-</b>		Source	<u> </u>		7
		-					B.T. A.R.	
					<u> </u>		VO UOMOJ UOY	
	► 15 <		· ·				11. ~	DID N/ T
		· · · · · · · · · · · · · · · · · · ·				2 6	1 ct 1-1	· · ····
——					$-\frac{5}{2}$	<u>7 _2</u>	100 10 10	<u>C 10 105/2</u>
					Komak	VINX 89	mpyg uscald	od

, bellikinin oʻ madilikini N

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.



NOTES: 1 -- Length recovered; 0" -- Loss of Sample, T -- Trap used

2-U = undisturbed; A = auger; OER = open end rod; V = vane

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547

、.... i

								÷	
<u>PA 547</u> 6-90			TH	ie po		<b>RITY</b> (1)	3 NM & I	CHN CHN	
					Constructio	n Division			
				Ma	aterials Engine	eering Sect	tion		
					BORING	REPOR	Т		CHEET OF 1
PPO IECT				· <u> </u>	NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
lor T	Europy 0	PdG S.To			Crais	· Dr.ll	AC .	Stame 3 A	
LOCATION							·y	CONTRACT NO.	DATE / /
<u>→3, ,</u>	shot	STaw 3	East	Bay,	Bldg DU	··		426-99-000	1/11/00
SPOON	-	CASING SI		TYPE	Data	Time	GRO	OUND WATER LEVEL	
HAMMER		HAMMER			Date				
#	FALL		# FALL	-	11/11		1 Pr	/	
DRILLER	2	Da i					/		
INSPECTOR		U Such							<u> </u>
		Nowe							
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.		*S/	AMPLE DE	SCRIPTION AND REMAR	RKS LE UU
Mand		Hand Buger	Fuli			for	5 Jrg Ta	0-4' See	<u>Log</u>
Augr						for	<u>- SJS</u>	11N2	
Ĵ				<u> </u>		Na	Samo	los Takon	
									<u> </u>
		<u> </u>	<u>├──¥</u> ──					P.T. of	- 10
	- 5 -		<u> </u>		┝			_ pow	<u>potray</u>
	<u> </u>	<u> </u>	ļ					rics non	)-e/Mote
			<b> </b>			<u></u>			
			<b></b>						
			1						
			<u> </u>		<u>├</u>				
		+							
	— —	+	<u> </u>		<u>├</u>				
	— —	ļ			ļ		··· ······		
	- 15 -								
	—— <u>—</u>								
		+	<u> </u>		<u>├</u>				
		+	<b> </b>		<u>├</u>				
	- 70 -	<b>_</b>	<b> </b>						
	<u></u>								· ·
		<b> </b>	<u> </u>		<u> </u>				
	25		<u> </u>	L	l	<u></u>		<u></u>	

÷

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

# THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section

ų

1

-									SHEET / OF 3
PROJECT	~~~	2100	_		NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
Porl	Dory	POG SI	Te		Craig	Prilling	·	STAIN 3B	
LOCATION					1 ~	0		CONTRACT NO.	DATE
Eqsi	Bay DI	dy 20 7	-1/ >	outh	et STGIN.	3A		476-71-00	1//// 00
SPOON	234	CASING S		: түре 1		-	GRO	UND WATER LEVEL	
	$\frac{D.D.}{Cat}$	T.O. HANNER	<u>n</u>	<u>L</u>	Date	lime	Deptn	n	emarks
140 .			# 6411		Infutoc	95	55'	Lubil Hand	Associat
DRILLER	17466	<u>_</u>							<u>13 - 91 - 9</u>
		1 Osuch							
INSPECTOR	ľ	) Koup							
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- COV'D	SAMP. NO.	2	3°SA LI	MPLE DES	CRIPTION AND REMAR	кs _E <u>С,</u> с/
Hand	0	Hand Auga	FII		Misc Fi	11, Gro	115 , S	and, Graud ETC	5
Aucor				]/-	EU ED				<u> </u>
1	<u> </u>	[· ]	++-			OWN SAN			
		<b>├</b> ─── <i>}</i> ───	┼╌╌┠╌	12	<u> </u>	´			
					<u>  EII-E</u>	Brown	Sand_		
					+ Same				
))	- '5' -			13					
	·		+	<u> </u>	F flack U	ray Savor			
- Car		6-7		44					
SJann		3-3	16'		Forou	IN Sand	<u>4116</u>	<u>sı/</u>	
Augus		2-4		-					
		6-7	18"	Э	Sam	Q			
	10	812	1						
		8-10	2.21'	6					
		(-1	00		<u></u>	<u> </u>			
		S12	24	7.	<u> </u>				
		12 12	101		Iam	<u>e</u>			
		12-12	120	X	Sam	nc			15:0
					L				
			1					Battomat Bi	ring
		- <u> </u>	1	]	<u>├</u> ─ ─ ─				<u> </u>
			<u> </u>	}	<u>├</u>				
			<u> </u>		L			<u> </u>	
	201		1	.					
				1	<u> </u>	FII Sam	mala c	hataluith	PIN Motor
						H- 10		1 ctt 12	TOT.T.
	<b>→</b>		<b> </b>		┝─>	<u>- 1- 1- 1-</u>	<u> </u>	alo 1 - 1 - 1 2	- parala 20
			<u> </u>	Į		omalur	s Sam	pla Viscandor	
			]				0 (	, 1	
	► 25 <	S: 1 - Longth -		L	a of Samala T	Tran wood			
	NUL	2 - U = und	isturbed; /	v Lus ∖ = auge	r; OER = open e	nd rod; V ≕	vane	(	
		3 — Log dept	h of chang	e in color	r of wash water, I	oss of water,	, artesian w	rater, sand heave in casir	ng, etc.

.

,

PA 547 6-90

## THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section BORING REPORT

								SHEET OF
ROJECT	_	0, 0,	~	_	NAME OF CONTRACTOR		BORING NO.	SURFACE ELEV.
PorTI	-Vor-1	PGG SI	Ĩ.		Crais Drilling		Q1-1	
OCATION		·					CONTRACT NO.	DATE
Southero	lof (0]	, NexTron I	ruch s	cale	Bluck 1338	611	426-77-006	1130100
POON	~ 3/			TYPE		GR	OUND WATER LEVEL	
<u> </u>	<u>, 718</u>	"I.D. 19081			Date Time	Depth		Remarks
AMMER	Satel.	7 HAMMER			11/2 1014	1.d	In Crusterd	Stan
	ALL JU		FALL				10 010100	)100g
	S	Buris						
NSPECTOR	n	<u>//</u>						
<u> </u>		oue		r				
CASING LOWS/FT.	DEPTH	SPOON BLOWS/6"	RE-' COV'D	SAMP. ² NO.	s اا	AMPLE DE LINE LOCA	SCRIPTION AND REMA TES CHANGE OF PROF	RKS 1LE <u>0.0</u>
	0			-	Concrete			1.0
Kand		Hand Aus	FJI		Crushed Storie			20
Pico	<u>-</u>	1	1	,		<u> </u>		
J.				1	FII- M-PC	ires Sa	nd La Min Siti. T	r Cucler Tr Ground
Yellon .		Tamad			<u> </u>		al show a	Sendlo
STem	• 'S ` ◄	Span	184	2	FBran	Savol. 1	MOSTI, TURIA	
		810						
		12-15	5"	3	P Brown	Sand	I The SIT. T. C.	
		4-3						
		3 -6	192	4	P-Brown (	Toral T	SITI TUKAT.	Tr Gray.
		6-17				in with a		
		27-27	200	5	P Brown	and, I	rSIII, Tr Grou	
		6-12						
		19-21	15"	6	P Brown S	not De	PGray Sound Dr	SIT TY Gray 1 14
-v- -		5-12						und fille the state of the second
	15	12-14	187	7	Far. C			
						wey I	<u>/_}///</u>	1
							Battan + Bar	
							www.	~~~
					All Ca	mal	charte 1. m	PID Mata
	<i>70</i> <		<u> </u>		$- \frac{1}{\sqrt{1-3}}$		S. I Can Ear	Tation
					$\frac{3}{2}$ $+^2$	ey.	Jugar Tor Ell	1 1
					Komainin	y Jan	flos Us card	≤ <u> </u>
[	_					-	•	
							· · · · · · · · · · · · · · · · · · ·	
- L	20							

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

Same:

4

PA 547 6-90

٠

<u>A 547</u> 6-90			TH	IE PU	KI AU INUMITY 他F NY 街水田 Engineering Department Construction Division aterials Engineering Section
					BORING REPORT
PROJECT	Č.				NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
Fort Ive	sry 🏽	46			Uraig drilling PI
LOCATION			betw. Rai	( sed of t	CONTRACT NO. DATE
between	bldg 49	A& bldg !	52 BI	ock 14	00 lot 1 426-99-006 11-22-00
SPOON	- 3/-	CASING S		TYPE	GROUND WATER LEVEL
<u>5 °0.</u>	<u>b. 7. 78</u>	"I.D. Auger	<u>s                                      </u>		Date lime Deptn Hemarks
					1, 22 00 1. 69 PM 1 . S++ 6
	ALL SU		# FALL		11-22-00 1:43 6.0 JH 7
NSPECTOR	D. Co	ok			
	77	ark			
CASING LOWS/FT.		SPOON BLOWS/6"	RE- COV'D	SAMP. ² NO.	SAMPLE DESCRIPTION AND REMARKS     LINE LOCATES CHANGE OF PROFILE     O
			Full		BALLAST & Crushed Rock
				1	Ell arou a I SAND & Gravel 1. CilT
-++		<u>├──                                   </u>	<del>       </del>	<u>_</u>	<u></u>
			+ + + + + + + + + + + + + + + + + + +	*	
				2	Fill reddish - brown c. I SAND to Gravel to SIT
	- 5 -	·		7	
		<b></b>		<u> </u>	<u></u>
		4-6			
Augens		E /.	9 011	1.	B D SAND I C I L SIT
	<b>-</b>				- vrown <u>c</u> -f value, fr vale, "
		1-2		*	
	- 10 -	4-8	24"	5	Brown c - [ SAND tr Gravel tr. CLAY & SIII _
·		6.9			
<del></del>			0.11	L	
		15-22	<u>  14" -</u>	<u>v</u>	
			-	•	<u> </u>
					Note & cound for halfing 7
					filore: Normples sowed for resting
	15 -		ļ		All other samples checked
^* · [`					w PiD meter & discarded Bottom of Bo
	- ·		· · · ·		
<del></del> +					
		1			
	• •	<b></b>	<u> </u>		
		·····	<u> </u>		┝ · · · · ·
			ļ	2	
		1	{.		• · · ·
	<u> </u>	<u> </u>	†		
- 1		1			

1 - Length recovered; 0" - Loss of Sample, 1 - Irap used
 2 - U = undisturbed; A = auger; OER = open end rod; V = vane
 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

## THE PORT AUTHORITY OF MY& MJ

**Engineering Department Construction Division Materials Engineering Section BORING REPORT** 

							SHEET 1 OF 3
PROJECT		· <u> </u>	NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
Port Ivon, PdG	Site		Crais Dr.	Iling	_	P-2	
LOCATION						CONTRACT NO.	DATE
Boturen Bldg 32A	242B		Black 1	yee Los	$\Gamma/$	426-99-006	11/30/00
SPOON CA	SING SIZE HOLI	TYPE			GRO	UND WATER LEVEL	
3 "O.D. 218 "I.D.	Huprs	1	Date	Time	Depth		Remarks
HAMMER Safer HA	MMER				7~		10.
140 # FALL 30 "	# FALL				10	Unily Mon	nd Bugering
DRILLER S Burns					ł		
INSPECTOR							
	ON RF-	SAMP	2	3SA			RKS
BLOWS/FT. DEPTH BLOW	S/6* COV'D	NO.		Ŭ	NE LOCAT	ES CHANGE OF PROFI	LE 0,0
			AspleN	~~~~			0,3
		4	ļ,			d	
Navel Mand H	typer Full	L	Crushia	Slone		· 1 22	
Arren	Ŭ	Γ.	1			loom	
	├──┼─┼─	11	NI. E				
			1×1/SCF	11- 20	and Si	Linden Or	106 - CIC
							,
		12			R	LODS /THO	ist. C. I
	├───┼──┼──	+		Uray 1	010000	O ED SAND, ITS	1111110Vaver
		] 3	Samo				Sec.
		+		<u> </u>			01)_
STOM 6-8		11	L			~	
Arcon S-1	4 144	Y	Red Ru	our Su	IT. Cla	V TYFSand	TriGuard
10 5-					$d \neq d$	71-1	
		5			- <u> </u>		
	-2 20"		Sam	e			150
		4	┝─ ── ──				<i>#</i>
·		4				Mallamot Boy	ing
			1				
		1	<u>├──</u> ── ──				
<b>├</b> ───┤── ─┤───					-,,		
		J	LP1	1 Samo	los ch	ected with	PID Metor
		]	<#	125	Sai	al for Fam	to Deline
<u>}</u>		4		102	<u> uu</u>	Tor Level	
		1	Kar Kar	MAINING	<u>, Jan</u>	fles Uscarde	24
			1	6	2	V	1
		1	<u>├</u> ── ── ──				
		4	<u> </u>				
	ł						
		1	F				
h		4	<u>├</u>				
		1					
▶ 26 ◀	<u>l</u>	1			·		

NULES:

1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

.

PA 547			TH	ie po	RT AUTHORITY OF NYG NJ	
6-90		. ·			Engineering Department	
-				м	Construction Division aterials Engineering Section	and the second se
					BORING REPORT	ि - ¹ र्भि 
						SHEET OF 3
PROJECT					NAME OF CONTRACTOR BORING NO.	SURFACE ELEV.
Port .	Lvory	<u> </u>			Craig drilling P3	DATE
Baluna	111. /0	er bloka zz A	. 9 . 1	N C	Rida 1. Black 1600 Late 696 99-006	11-29-00
SPOON	bidg 4x	OASING SI	ZE HOLE	TYPE	GROUND WATER LEVEL	
3 .0	D.D. 23/	8 .1.D. Auge	rs		Date Time Depth R	emarks
HAMMER A	utom.	HAMMER		_	ALEN IS THE AVE	
DRILLER	FALL S		FALL		11-22-00 $3:50$ $4.5$ $100$ $5# 5$	·
	D.	Cooks				
INSPECTOR		-7				
	<u> </u>	Larks				<u></u>
BLOWS/FT.	DEPTH	BLOWS/6"	COV'D	NO.	LINE LOCATES CHANGE OF PROFIL	E 0.0
No and augo		Handause	Full Rec		Fill and C & CAND AND Ground In SUIT	0.2
no.no.avyr	F	I Manabulger		1 1	FILL C- SAND AND COMPLETE SINC	
		++			Keddish = brown c-1 SAND and Gravel som	2 Clayey 3111
				2		
		}		4	SAME	4,0
	5.			~		
•		1 🖌		5	Red-brown Silty CLAY with Grovel k	pulders little c-(SAND)
Wow	<b></b>	97.91			,	
mers		77 30	26+			CLAVEN SUT
		35-38	~~~		Red brown C-1 AND and brower, bounders	Some VINYET JUI
		<u>  B _ 12</u>	()	-		
	- 10 -	15-23	24	5	<u></u>	
·		42-24		,		
¥.		20 - 18	24	6	SAME	12.01
				-	NI is coupled and D takes	
	<u></u>	· · · · · · · · · · · · · · · · · · ·		<b>-</b> .	Note 2 Samples saved for resing	
,	► 15 -	◀			- All other samples checked w_PIL	meint
	<u> </u>				è discarded	_ Bottom of Baring
	> 20 -	<b>4</b>				
·						
					· · · · · · · · · · · · · · · · · · ·	
	▶. ◄	◀	LI		I	

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

8:28 11.00

÷.

.

PA 547 6-90

## THE PORT AUTHORITY OF NYS NJ

**Engineering Department Construction Division** 

**Materials Engineering Section** 

#### **BORING REPORT**

			۸		SHEET OF
PROJECT		210.00			NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
Por Ju	01-1	PECSIR			Charge Utilling BBMG-70
LOCATION	י ר ה ה מ	LNV. 7 Wast	F. A. RI	1.22	RI LUKA LATI YOL 90-00K UL/12/C-
SPOON				TYPE	0100E 1900 LCJ 1 176-17-000 11773700
2 .O.D.	1%	-1.D. Hu	P'N	Aonitor	Date Time Depth Remarks
HAMMER	Sator	/ HAMMER			
140 # FAI	<u>ш Эо</u>		FALL		
	t	Finch			
INSPECTOR	Q	Moue			
CASING BLOWS/FT.		SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	² ³ SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE O.
CUTTer	0		}		
heid					for strate a-16 See los, for
		1/2 / D		1	PANW-7 No South EL (1-14'
ngha mgi		I MANA DUCT		{	10 Jan 0 - 10
┝━┼━┼╸					
	5 <				
	6				
(Sing)	0				
				]	
}	·			1	
<b>├</b> ─┤── <b>▶</b>	75 -				
					///////////////////////////////////////
4,11		11-16			
Phad		93-21	241		Red Brown S, IN, Clay, Tr FSque 18.
with	_	11-16			
Rectors	20	16-24	12"	2	F PK Bround Scrol TV SIT
		12-15			
		16-18	11"	3	FDL Bury Sand to SUT TLG
<u>├</u> ┼┼-		10 - 14			
┝━┼─┼		16-21	17//	4	PDF Que Su / TVSUT
+ + +	····-	9-11		<u> </u>	1 VL Orown Jand, IV JILI
	25 <	-1-19		5	Nulo #++Ter 5 5 074 goluge cal from 16-5 24'
└──┼──┼─		16-4	16"	ļ	Same H.
		23-29		1	
		19-19	174	6	F Brown Sand Little SIT
		8-8		7	
	2	13-23	94	/ /	ISame.
	$\sim$			I	

\$

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

<u>PA 547</u> 6-90			T	<b>IE POI</b> Ma	TI AU I MUNIII Y UP INY 造水日 Engineering Department Construction Division laterials Engineering Section BORING REPORT						
PROJECT	<u>~</u>				NAME OF CONTRACTOR BORING NO.	SURFACE ELEV.					
LOCATION	Luory	POG-Sila			Craig Urilling Contract NO.	DATE / ,					
201 m	estof	PD-MU-7	lups)	of BIJ	533 Plack 1400 Lot 1 426-99-00	6 11/14/ac					
SPOON	D. 13/8	"I.D. Ravent	-  'B"N	(UNITOr	GROUND WATER LEVE	L. Remarks					
HAMMER	Safe	HAMMER									
DRILLER	FALL 20	<u> </u>	FALL			<u></u>					
INCRECTOR		J FINCH									
INSPECTON		D Nowe									
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- '	SAMP. ²	³ SAMPLE DESCRIPTION AND RE LINE LOCATES CHANGE OF P	MARKS ROFILE					
Drill	30	12-22									
Bhead	<u> </u>	20-32	164	8	Former Sand LITTIO SIT						
with		14-16		9							
RAUNT	<del></del> <del></del>	28-28	121	11	Same						
1		25-23			Noto: After St la Hu casure a lug	col from 24 to 34'					
	> 35 ~	13-12	10"	10	PBrown Sand TrSill	36.0					
		5-9		· · ·							
		12-14	15"	ų	F Roof Brown Sand, Some clayer						
		20-11		17							
		11-11	221	12	FRed Braun Sand Some Sill						
	90 -	14-15		10	<i>y</i>						
		14-15	20"	り	Same	42.0					
		7-6									
		5-5	31"	14	F.R. Brown Sand, Same clara	SIT 440					
	- Kr-	7-8		15							
	_ ~	11-9	241	13	FRed Brown Savel, LITTLe claye	x Sin 46.0					
		10-10		10		, 					
		10-12	19"	10	FRed Brown Sand, Little Sitt _						
		9-8		17							
	► 150 <	11-11	24"	1/	F Rad Brown Sand Some SIT						
		11-12		10							
		13-18	24"	10							
		7-8		19 A	Same						
	<del>~ -</del>	11-10	22'	B	FRed Brown Sand, LITTLE CLAYER SIVIT, T						
		10-19	20"	20	Same						

į

- Menter

¥

.

,

NOTES: 1 – Length recovered; 0" – Loss of Sample, T – Trap used 2 – U = undisturbed; A = auger; OER = open end rod; V = vane 3 – Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA	547
6-	90

- A Strategy and the same second data of the second s second s •

## THE PORT AUTHORITY OF MY & NJ

. 1

400

Engineering Department Construction Division Materials Engineering Section

## **BORING REPORT**

						SHEET 3 OF 5
PROJECT					NAME OF CONTRACTOR BORING NO.	SURFACE ELEV.
Port Ivory P&G Site					Craig Drilling PH-MW-10	ļ
LOCATION			<b>¬</b> • •		CONTRACT NO.	DATE
+20'W	osic+P	BMW 7, m	slot B	dg 33	Block 1400 Lot 1 426-79-006	1116(00
SPOON	17/		ZE HOLE	TYPE	GROUND WATER LEVEL	
<u>, , , , , , , , , , , , , , , , , , , </u>	0.D. 178	"I.D. Keliert	HN	(IN) IN	Date Time Depth Re	marks
HAMMER	تغابة 26 FALL		FALL	Π		·
DRILLER		J Fluch				
INSPECTOR	<u></u>	DHorwe				
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	³ SAMPLE DESCRIPTION AND REMARK LINE LOCATES CHANGE OF PROFIL	KS E
Phil	- 55 -	30-32	22"	20	FRed Brown Sand, Little Clayer Silt	D'Grauf
Mad		15-18	[.			
Lum		21-24	111	181		
0 -	┝	112 10	•••	<u> </u>		
Newy	<u> </u>	11-19	<b> </b>	22		
	1.01	27-30	7″		F-Rad Brown Sand, Some clayer, SITT, Tr Ra	Istalo, Tr Grant
		7-10		0.2		
		20-61	12"	63	FRed Brown Sand. Some clayar S. ITV. 4De	Croud TI Raishale
		23-20				
		1.115-27	1211	124	T=P-1 Rtm S 11. Ht C 11 R China	ST. T. D Ich
├──-{		1411 2-	<u> </u>		I NO OTOWN DANO, LING WYOUPI, LITTE CIAYES	2111/ I Ket male
	65.	74-55		20		
	└ <u>·</u> -	34-48	14"	(0)	FRAN Brown Sand, Some Gravel, Tr Clayer,	tr Rod Shale bb.
		70-29			· · · · · · · · · · · · · · · · · · ·	
		87-58	フペ	66	Recomposed Read Shale. Suma Createl LITTLE	FSand
		27-68				
<b> +</b>	⊢ 	1000 14. 24	14"	127	Scim a	
<b>├</b> ── <b>├</b> ──		24-94	<u> </u>		n_	chan s. IT
		96.58	12"	28	EPHP SIS Prove 10 101	L L THE C Th
		54-100 4/1"	10 ¹¹	20	1-North Drawn - and some the mais a sta	it To Par 1 770
<b>├</b> ── <b>↓</b> ∕──		12/ 59/24		07	vycomposar 100 37512, LIUG CRYLY)	<u>11 1 - 132</u>
<u> </u>	┝	X	<b> </b>	1		
	75	A.			<u></u>	Ing Kepoart
ļ	┣─────	<u> </u>			·	
ş						
	┝── ─					<i>A</i>
	├ <u>-</u> -		**:-		Bollimof Bo	ring
	60.	┫	ŀ	L		

: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc. # 300 // Hann more USed

# THE PORT AUTHORITY OF NYS RU

		·		Engine	ering Department		
			Ma	aterials	Engineering Section		
			E	RILL	ING REPORT		
				NAME OF	CONTRACTOR	BORING NO.	SHEET OF
PorTILory PJG Site					14 Utilling	PAME-70	
+ (	DA NUC	7 44-5	0.01.0	~	RI ( liferer 1 to)	CONTRACT NO.	
LL RIG.	FD JAU-	1 W95Ja	COE BARF	<u>う</u> iel	CORE DRILLS	SIZE CONDITION OF I	HAMOND BIT
Mobile	B-58		-	SINGLE TUBE	TUBE	line 1.D.	ood
OM CASING	DEPTH	START CORING	DRILLE	R		INSPECTOR	/
34.0' 13.2'			J Finch V Norse				
AE 1	DEPTH		WASH ³	4	ROCK-DESCRIPTION A	ND REMARKS	73:
	► 73.2.◄	DEMAILON		<u> </u>			
Mus/PT		Sleady	Losinguri	¥			
		<u> </u>			KUNE Ke	d Shile, Segn	my Fractured
			obset we up				· · · · · · · · · · · · · · · · · · ·
			Cosnalata				
			Blackingup	<u>├</u> ── ─			75
¥	>78.2						
				<u> </u>			
, 		L				Battion of	Boring
		ļ					
				}			
				<u>├</u> ─ -		=	· · · · · · · · · · · · · · · · ·
	-832-	ļ		┼			
		<u>-</u>		┣		· /	· · · · · ·
				ļ			
					-		
				<u>}</u>			· · · · ·
	► 882-			<u> </u>			
		·		┟─── ──		<u></u>	
			 	<u> </u>	<b> t</b>		
·							
				├	· · · · · · · · · · · · · · · · · · ·	- <u></u>	
	- 43.2	,   		L		·· <u></u>	
RUN NO.	FROM	то	LENGTH D	RILLED	LENGTH RECOVERED	% RECOVERED	NO. PIECES
	732	78.2	E.C	7	145	90 %	17 Que to
<del></del>	1,0						11 10050010
				=			
							,
					<u>↓</u>		
1	1						
			·····				
ł					ll	<u></u>	
					1		
· · · · · · · · · · · · · · · · · · ·	RUN NO.	<u>LUOIY</u> <u>VUU JIII</u> <u>ust of PH-MU</u> <u>Noble B-S8</u> OM CASING DEPTH <u>4.0'</u> <u>782</u> <u>782</u> <u>782</u> <u>782</u> <u>882</u> <u>882</u> <u>882</u> <u>882</u> <u>882</u> <u>882</u> <u>732</u> <u>732</u> <u>732</u> <u>732</u> <u>732</u> <u>732</u> <u>732</u>	LDOTY       VOC JINE         usst of PB-MU-7 Wossta         IL RIG.       Machile B-S8         OM CASING       DEPTH START CORING         Y(c)'       73,2'         Mus/PT       DEPTH         BEHAVIOR       STeady         Mus/PT       STeady         Nus/PT       93.2         RUN NO.       FROM       TO         1       73,2       78.2         93.2       RUN NO.       FROM         1       73,2       78.2	LUOINY       VOID SITE         ust of PH-MUL-7       Wastof Aldred Graph         Machile       B-58         OM CASING       DEPTH START CORING         Mashing       DEPTH         DEPTH       DRILL 2         WASH 3       WASH 3         Mus/PT       Steady         Losinglu/R         Nus/PT       Steady         Blacknown       Usinglu/R         Nus/PT       Steady         Steady       Losinglu/R         Nus/PT       Steady         Steady       Losinglu/R         Blacknown       R         Steady       Losinglu/R         Steady       Steady         Steady       Losinglu/R         Steady       Steady </td <td>Interpretation       Interpretation       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation</td> <td>DOTY       0.00 - 3112       1 Crimy       0111102         urs1 of PP-Muc-7       vors1 of PILs       Block 1400 Lot 7         Mobile       8-58       000000000000000000000000000000000000</td> <td>Dery     Vor Jie     Charge of the mark to compare the comparence of the mark to compare the comparence of the mark to compare the comparence of the comparen</td>	Interpretation       Interpretation       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation         Instruction       Instruction       Interpretation       Interpretation	DOTY       0.00 - 3112       1 Crimy       0111102         urs1 of PP-Muc-7       vors1 of PILs       Block 1400 Lot 7         Mobile       8-58       000000000000000000000000000000000000	Dery     Vor Jie     Charge of the mark to compare the comparence of the mark to compare the comparence of the mark to compare the comparence of the comparen

NOTES: 1 — Record the time to start and end of each foot of drilling 2 — Log drill behavior (i.e., steady, chatter, grinding, etc.)

.

.

 3 — Log wash water return (i.e., color, loss, blocking, etc.)
 4 — Log type, color and condition of rock (i.e., broken, soft, seamy, hard, etc.), log character of wash return solids . .....

N AND L

# PORT AUTHORITY OF NY & NJ Engineering Department - Materials Division


6-90	•			Ma	Engineering Department Construction Division faterials Engineering Section BORING REPORT	
					SHEET ) OF L	$\overline{I}$
PROJECT	•	Dol			NAME OF CONTRACTOR BORING NO. SURFACE ELEV.	
lort 1	vory				Uraig drilling	{
3'CW	al aris	ainal loca	tion	В	Block 1400 Lot 1 426-99-006 11-11-00	
SPOON	-	CASING SU	ZE HOLE	TYPE	GROUND WATER LEVEL	
3 10	.D. 2 3/8	3 -1.D. Augers	μ' M	ONITON	Date Time Depth Remarks	
HAMMER	Safe FALL 30	Ty HAMMER	FALL	-	11/13/0c 815 7.0 5#3	
DRILLER	Ţ	Finh				
INSPECTOR	ກ	Vana				
CASING		SPOON	RE- 1	SAMP.2	² ³ SAMPLE DESCRIPTION AND REMARKS	
BLOWS/FT. Cutter -		BLOWS/6" Cutter Head	COV'D	NO.	LINE LOCATES CHANGE OF PROFILE	0.0
Head	`					{
						5
Kand		Hand Auge	FUY	é	·	
HC87					Fill-M-P Brown Sond, Tr SIIT	
	5 -					
	·		$\Psi_{-}$	<i>"</i> n	hill - M-F Black Sand, 185115	-
Sign		54-82	194	38	B F/1-E DEGER Sand Treat	5
Arcars		97-12		- P	A some	2
1		52-60	244	°Y B	B Reddish Brown Cluyar SiTT & Save, The Fillack Scard 10	20
		4-8		-		
		14-20	18''	>	Rod Brown S. IT, Clay, LITTL FSand, To Grand, To FBlackSana	
!.		15-17		6		
$-\Psi$		24-28	भ		Bed Brown SITY Clay, To FSand TV Grown	{
	► 15 <	11-15	0.11	7		<u></u>
		85-98	14"		Jame 16.0	<u> </u>
				-	Bo Derrot Boring	
					All Sample Checked with PID Motor	
					- S" 1-4 Sund for Bostlas + 1-4 Nor Be Forskon -	$\neg$
	21					
	NOT	ES: 1 — Length re	covered; C	)" — Los:	ss of Sample, T Trap used	

PÀ 547

2.00

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

# PORT AUTHORITY OF NY & NJ Engineering Department - Materials Division

----

Vell Installation Repor	rt			Sheet 2 of 4
PorT Every, Pd	G-SITE			476-99-00
TELL NO. PANULT WELL T WELL T WELL T WELL T WELL T	- Rloct 1400 A ^{NPE} ManiTor	INSPECTOR Decor	DRILLER OF F	inch 11/13/ac
Vell Development Rep	Ort (NOTE: WATER L	EVEL READINGS FROM TOP OF PV	· · ·	
IIII3/00 WATERLEY	EL BEFORE 6.2	WATER LEVEL AFTER	7,5	TAKEN 20 MINUTES AFTER
<u>9</u> " ( <u>3''</u> " dia. PVC	dia. Manhole cover pipe w/ locking cap	)		,
$L1 = 0.3^{9}$	L1		Top c & cen	of surface nent grout
$L2 = \frac{317'}{1000}$	L2			of bentonite seal
L3 = (0.0)			3.d Top o	f well gravel filter
	L3	<b>Spenings</b>		
	Can		$\frac{14.0'}{16.0}$ Botton	n of well n of boring
		Boring diameter		
Hole Back Fille	14-16 WIT	h well Graud		
			* *	、

				м	Construction Division	ion		
					BORING REPOR	Г		
		- <u></u>			1			SHEET / OF 5
PROJECT	1 Hack	Post Turn .	Pte	SITA	Croce Dullia		PA-MW 150	SUMPAGE ELEV.
LOCATION	la noct	July Dory		517,0			CONTRACT NO.	DATE
AsLan	d outb	y Killes	Hssac		· · · · · · · · · · · · · · · · · · ·		426-99-006	11/3/00
SPOON	3/4			TYPE	Data Time	GRO	UND WATER LEVEL	amarke
HAMMER	Safat-1	HAMMER		<u>-</u>		Depin		
140 #	FALL JU	-	# FALL	*		- 1	·	
DRILLER		5 Burus						
INSPECTOR	·····	0 Howe				1		
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. NO.	² 3SA	MPLE DES	SCRIPTION AND REMAR	KS LE <i>O</i> .O
HW								
Casing					No S	amples	Taken	
)			1	1	for S	Trata	0-16 Soclar	~ — — — — — &
		1		1		for	PA-MW-K	<b>6</b>
				1				
	5 -			-			·····	
			╂────	{				
	<u> </u>		<u> </u>	ł				
			<b> </b>	}				
		·	ļ		· · · · · · · · · · · · · · · · · · ·			
	10							
		1				_		
				1			· <u> </u>	
		<u>+</u>	1	1				
		<u> </u>		1	├			
<u></u>				}	<u>├</u>			
	► IS ◄							16.0
Dill		11 - 12						
Ahad		15-13	14"	/	F Brown Sc	and Tr	Sili, Tr Gravel	
with	<u> </u>	6-8			1		<u> </u>	·····
Rows	20	11-13	141	2	Same			
)		11-12				· · · · · ·		
	 	13-17	18/1	3	- P. Rinner C	11.7	The SJT. Tr Grace	/
	<u> </u>	10-12.		├	<u></u>	mar LI	10001 11 11 01001	
		10-10	151	4	<u> </u>			
		9-10	13	5	<u></u>			
	> 25	1-10	10	<u> </u>	1 same			

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

·

PA 547 6-90

- A constraint with researches damaged but high ` ·

# THE PORT AUTHORITY OF NY & NJ

Engineering Department

0-50					Engineering Department Construction Division		
				M	aterials Engineering Section		
incases.					BORING REPORT		SHEET DOF
PROJECT					NAME OF CONTRACTOR	BORING NO.	SURFACE ELEV.
Howland	Houk, 1	POITILONY	PDG-5	SiJe	Craig Dulling	PAMIN ISD	_
	1 T	h. F.11.	И.			CONTRACT NO.	DATE
SPOON	<u>(a ou</u>	CASING SI	ZE HOLE	TYPE	GF	OUND WATER LEVEL	111/100
2.0	D.D. 1/2	"I.D. Revort	-		Date Time Dept	nF	Remarks
HAMMER	Safet	HAMMER		_			
DRILLER	FALL 70		FALL			·	
	<u> </u>	Burus					
INSPECTOR		O Howe					
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	³ SAMPLE D	ESCRIPTION AND REMAR	iks Le
Dill	[	12-12	16"	5	F Brown Sand, Lil	The STIF, Tr Growe	<u> </u>
Pheed		13-18		1			
with		24-22	)5'1	p p	Same		
Revort		11-16		7			
		12-12	8"	/	Same		30.0
		12-11		ø			
		13-14	11"	8	Brown SilT; LITIL 1	- Son of	
		9-11	ļ	9			
		13-12	15"	/	Brown S, IT & FS	and	34.01
	30	10-17		10		•	
		13-16	18"	10	F Brown Sand Sam	a S/17	36.0'
		8-8		1)	7		
		7-6	18"	111	F Brown Soud Som	SITY Clay	
		2-2	†				
		3-4	15"	12	Samo		· · ·
	- 40 <b>-</b>	3-2					
		3-4	16"	13	Samo,		
		4-2	† <u>`</u>				
		1-1	19"	14	Same	······································	
		3-2					
		1-4	24"	15	Same	· · · · · · · · · · · · · · · · · · ·	460
		5-4		11			
		3-4	22'1	16	F Brown Sand, LITIS	, S, IT	48.0
		3-4		17	M-F Roddish Brown	Soud LITTLE Sill	Fr Red Shaley
		5-6	23"	( /	1/4 Lews of FGIRY Some	1, 15" Layor of Gray	clayory SIT
-	NOTE	S: 1 - Length re	covered; (	0" Los	s of Sample, T — Trap used N	The after S#17 4	Into Consolin

1.1

THE PORT AUTHORITY OF NY & NJ

OTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

• ,

Engineering Department Construction Division Materials Engineering Section **BORING REPORT** 

PROJECT				·	NAME OF CONTI	RACTOR	T	BORING NO.	SURFACE ELEV.
Nowland	1 Hook.	POIT IVOIN	Ptas	Te	Crain	Dr. Ilman	,	PA MW-150	
LOCATION				<u></u>		- <u></u>		CONTRACT NO.	DATE
BS Lan	LOUT	by tillow	Assac					486-99-006	11/3/00
SPOON	00	CASING SI	ZE HOLE	TYPE			GRO	UND WATER LEVEL	
2 .0	.D.)%	"I.D. ReverT	-  'p' M	ONITOY	Date	Time	Depth	R	lemarks
HAMMER	C f	T HAMMER							
140 #	FALL S		# FALL	-					
DRILLER		S Burns							
INSPECTOR	· · · · · · · · ·	Dellaure	. <u></u>						
	DEPTH	SPOON BLOWS/6"	RE- 1	SAMP.2		³ SA	MPLE DES	CRIPTION AND REMAR	LE SO.C
Prill	- 20	6-9		, D	F Brown	Sand.	LITILe	SIT	- Slid
Bhead		21-22	2011	18 B	M-FR	& Brown	Sand, Se	amo clayer Sitt L.	The Rel Shale Gravel 52
with		8-3		la	Noto !	HFTer "	5# 18 4	Holo Cosal from 1	1を51-
Rount		8-10	12"	17	Red Brow	<u>v clorpy</u>	SILT	LITTLOM-FSOND, L	- TTe Gravel 540
	- 55	9-13		20					Redsholo
		77-22	181		Raddish Br	our Clay	15 Jung Silly	LIDLE M-FSand, 7	2 Decomposar 56:0
		10-13	101	31					
		6-11	(1)		Red Bro	ww 271	<u> </u>	FSond	580
		19-11	111	22	P 10				
	► 60	10-26			1) and Brown	- ( layoy >	JIJ Som	o 19-7-Song Ir Gro	will, I'r Rad Shale 040
		22-2/	1117	23	M-F-Ral	Brown S.	I So	SITE LITTLe Greek	d. Little Ratshele 620
		7-10	11"	24	Decompo	sol Rea	1 5/9/0	, Li Die M-+ Rad	Brown Sand 13-
V		100/4° 41/5 1/2			p				
		¥.					Ret	Usal, Seo Prille	ng Roport
									· · · · · · · · · · · · · · · · · · ·
<u> </u>						<u> </u>			
									68,5
		· · · · · · · · · · · · · · · · · · ·						Rattin A R	
<b> </b>	-70	◀						Udijamaj U	
<u>├</u> ───── <u></u>									
								· · · · · · · · · · · · · · · · · · ·	
		<b></b>	L						

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

PA 2255				INE PUN			JIND CHAI	
6-90					Enginee	ering Department		
				Ma	Const toriale	ruction Division		
				ma				·
DOO IECT					NAME OF		BORING NO	SHEET 4 OF
Houl	and Hook	E POITI	vor, Pd	GSITE	C,	ais Prilling	PANW 150	
LOCATION	·		F11	here			CONTRACT NO.	DATE
TYPE OF D	RILL RIG.	<i>"</i>	MILLON	COE BARRI	EL	CORE DRILL SIZ	E CONDITION OF DIAM	OND BIT
	Mobi	le 19-5	58		single Tube	(TUBE) 2"WITE/IN	vi I.D. Good	1
DEPTH BOT	TOM CASING	DEPTH	START CORING	DRILLEF		S B.	INSPECTOR	
T								
Start	Ene-		BEHAVIOR	WATER		LINE LOCATED	END OF RUN	63.5
1	Mus/FT		STondy	Full Ray				
10		L	/			Run #1 Red	Shale, Seamy	+ FracTurod
11						<u> </u>		
15			Chattering	Blockston				:
13		100	Ч	Losing wo Ta				68.5
		- 6823						1
		<u> </u>					RT AR	
							<u>De cjone of</u> 170	<u> </u>
	+							
	+							
	+	73,5			<u> </u>			
		ļ	··				· · · ·	
	+	ļ	}		<u> </u>			
		L ~				· · · · · · · · · · · · · · · · · · ·		
		75						
		- /8,3 -						
							<u></u>	
	-							
			<u> </u>					
		┝						
L		835		I I				
	RUN NO.	FROM	то	LENGTH DR	ILLED	LENGTH RECOVERED	% RECOVERED	NO. PIECES
100000	11	63.5'	68.5	ちの	/	3.9'	78%	7 Piecos + Frace
NUIES	<u> </u>							
			<u>}</u>					
RUNS								
			╂─────┤	<u></u>		<u>├</u>		
						<u> </u>		

NOTES: 1 — Record the time to start and end of each foot of drilling 2 — Log drill behavior (i.e., steady, chatter, grinding, etc.)

.

3 — Log wash water return (i.e., color, loss, blocking, etc.)
 4 — Log type, color and condition of rock (i.e., broken, soft, seamy, hard, etc.), log character of wash return solids

# PORT AUTHORITY OF NY & NJ Engineering Department - Materials Division

Ŷ



.

,								i -		
PA 547 6-90				ie po	RT AUTHO	RITY OF	'nysi	Ъ		
0-90					Engineering I Construction	Department				
				Ma	aterials Engine	ering Secti	ion F			
					BUNING				SHEET )	OFY
PROJECT	I VI I-	Patt	ptc s	<u>л</u>	NAME OF CONT	RACTOR		BORING NO.	SURFACE ELE	EV.
LOCATION	TIOUR	1045 Luon	FVC SI	10	Craig	Winnes		CONTRACT NO.	DATE	
BS Lain	lowthy	Killam ASS	C	TYPE				426-99-006	11/3/0	2
3	D.D. 03/8	"I.D. Augus			Date	Time	Depth	R	emarks	
	Safar	HAMMER	# EAL		Ulston	85	5.9	upila Hand	Acrive	
DRILLER	FALL JU		# <b>FALL</b>		11100		<u> </u>		1) 471 -22	
INSPECTOR		DUYNS			· · · · · · · · · · · · · · · · · · ·		<u> </u>			
		V Nowe					<u> </u>			
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- ' COV'D	SAMP. ⁴ NO.		3SA Lii	MPLE DES	CRIPTION AND REMAR	KS .E	940
Hand		Hand Accor	Full.	,						
Augr	L		1		Misc Fi	11-Cino	lors, D.	tos Saud Grau	1 Etc	2.a
		ļ		7_						
		<u> </u>			<u> </u>	M-F 6	Brown !	Sand, Trs. 15,	TrGravel	, 4.c
	- 2 -		↓	3						
	ļ				Fill -	M-F Br	LUI Samo	1 LIDIO DE Brown	Send, Jr	SIT-6C
de	<u> </u>	1-0		Ϋ́					·	
Slam			18"		<u>  F.//- F</u>	- Bicu	<u>v Sand</u>	(LiDlo Siti, Tr	Logt	80
HIGH		(-0)	1.14	2						
	- 10 -				<u> БЛ- </u>	Prour	Saral L	16 SIT, 12 SII	y (lay	0.0
		5-11	101	6						<u> </u>
	<u></u> −−	5 1/	181		<u> -</u>	Drown S	Dan 9, 11	S, IT, Tr Gravel		· <u>·</u> ·····
		18 20	24	7		d Samp	16 <u>_Satu</u>	<u>ــــــــــــــــــــــــــــــــــــ</u>	·	
·		4-101	041		same		; ;;			<u> </u>
	15 -	1591	2011	8			<u>~</u>			16.0
					June	<del></del>		<u></u>	· <u> </u>	1
								BerThemat 7	Parise ~	+-
					<u>├</u>					
		· <u>+</u>			- A	11 Sam	abs cl	antentier The P	ID Motor	
					5'	+ 1-6	78	Sacoch fer P	sting/	
*		1			R		501/	Ascardod	·····	
*										
	2									
					a di Osmala T					

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

.

and the second second

# PORT AUTHORITY OF NY & NJ Engineering Department - Materials Division



.

(

Engineering Department Construction Division Materials Engineering Section

#### **BORING REPORT**

											SHEET OF 3
ROJEÇT			0	NAME OF CONTRACTOR						BORING NO.	SURFACE ELEV.
HOWW	MID HO	2K .	Yora	I VORY P.	+6 51TE		LRAIG	VILILL'HE		UST7-1	
CATION		1		2-1						CONTRACT NO.	DATE
NORN	OF BUIL	DING	537	- 15	WEST	OF	UST7-	1A		466-49-006	11/20/00
	73/2			ZE HOLE	TYPE				GRO	OUND WATER LEVEL	<u></u>
<u>)"</u> •0.	. <b>D.</b> 718	"I.D.	HUGER		l	İ	Date	Time	Depth		Remarks
<b>Ammer</b> †0 <u>#1</u>	FALL SAFET	-	HAMMER	FALL	-		11/28/00	15:30	6to7	OBSALVED DU	LING SPOONING
RILLER	J	(R	AG								
SPECTOR	DAN	$\overline{)}$	)A/15							and the second states	
ASING	DEPTH	SI	POON DWS/6"	RE- ' COV'D	SAMP. ² NO.		L	3SALI	MPLE DE NE LOCA	SCRIPTION AND REMA TES CHANGE OF PRO	RKS FILE
NO AUGGA		HAN	2 Aurice	For		ŀ	ASPHIALT EAU - DAN	AN STONE K BROWN	HND (	BLATCK FINE TO Y	NECIVIN SAND +
1				1			GOEN	<u>المعام</u>		········ ·····························	
	The second s				Λ	1	GAME				/
· [			1		1 1	-	<u></u>				3.5
++	- 75	1 200	+;		<u> </u>	┢	hu- Kel	ADSH/ KROW	H CLAY	EX SIM LITLE	WIT TEL AND
1			$+\sim$		1 2		FINE	SAND		· · · · · · · · · · · · · · · · · · ·	5.2'_
	_ ^ _		↓	_ ⊻_			FILL - BI	AUK-STAINEP	LAVER	1/16"-THICK), GRA	DES TO BROWN 90
		2	3	21		Γ	SAME		. <b>X</b>		FINE SAND, SOME SILT
	•	11	1		4	$\vdash$				/	TRACE CLAN
		1	<u> </u>		<u> </u>	╞				<u>````````````````````````````````</u>	
		4	<u> </u>	5"	$\langle$		R	B_CIA	IEV SILT	- SOME SMALL	TO LARDE GNAVEL
	IN -	5	8				AND	FNE <	AND ,	STRONG PETROLEUM	1 ODOR 10'
		ΪΛ	10	18"		-	DENKE	REDNICH	RODUN	SILF TRAIL O	LAY
		10	21		6	┢		Laupon	Nom14	+ Inne C	<u></u>
<b>V</b>		19	_34		·	┢					12
				L			1_				
					1		L	Barrow	OF PAN	anh.	
					1	$\vdash$			<u> / 1</u>		
	• •					┝					
						$\vdash$					
					ł		— — ни	<	ES_S	UPENED WITH	IV_METER
					1	L	<u> </u>	<u>maes #5</u>	AND 6	_SAVED_FOR ET	WIRONMENTAL TESTIN
				1			- An	L PEMAIN	NK CAR	MAS DURA	
					1	$\vdash$		- NEUDIN	110 21	inc marry	Y
	• •		,	ļ	ł					······ ······ ······	
					]						
					1						
					ł	$\vdash$					
									·		
• ·					}						
					1						·······
				1	1	I.					

3 -- Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

<u>\ 547</u> ⊱90

Engineering Department Construction Division Materials Engineering Section

**BORING REPORT** 

				_						SHEET OF 3
DJECT		~	41 1	~		NAME OF C	ONTRACTOR		BORING NO.	SURFACE ELEV.
YORT	VORV	P	265	ITE	<u>.</u>	(RAIG	DRILLING		UST 7-1 H	
ATION		7		7 21.					CONTRACT NO.	DATE
<u>No</u>	RTH OF	DU	LDING	5 27		<u> </u>	· · · · · · · · · · · · · · · · · · ·		1426-41-006	11/20/00
ON	1		CASING SI					GRO	OUND WATER LEVEL	
. <u> </u>	<u>),D.</u>	"I.D.	HANNER		<u> </u>	Date	Time	Depth		emarks
₩	EALL	-		FALL						
		<u> </u>	1	TALL				+		
I	CRA	16								
FCTOR	\	<u> </u>						1		
LA.	NI	1411	5							
SING WS/FT.	DEPTH	S BL	SPOON OWS/6"	RE- 1 COV'D	SAMP, ² NO.		³ S/	MPLE DE	SCRIPTION AND REMAR TES CHANGE OF PROFIL	KS _E
D		1 hon	DANGE	Fur	1*	Fill	DARK -	REDW	N F-M SANT	. Some
ĒR		<u>† ""</u>	1		11					
	<u> </u>	<u> </u>			<b> </b>		<u>&gt; (olla</u>	VEL -		
	· ·	<u> </u>		ļ	2	Sm	<u> </u>			
		<u> </u>		t			1			
					1		L Barren		NI / Contraction Dat	-2/
	► 5 ◄	<b> </b>			-	<u> </u>		E DURI	No floncheir Im	
	$\vdash$ $-$	ļ			ł					
					[					
	<u> </u>	1	· · · · · · · · · · · · · · · · · · ·		1					
		<b> </b> -		<u> </u>	1					
	 	<u> </u>			ł	·			·	
		<u>^</u>				1				
					]				- <u> </u>	
	<b></b>				1			- ,		
			· · · · · · · · · · · · · · · · · · ·		4	<u> </u>				
	*				]					
					]					
				 	·		- AU CAME		SCREENED WITH	PID WETA?
	- 15 -						- SMAPLE 1	_ SAIET	2 FIR ENVIRONM	NTAL FESTING
			·			• • • • • • • • • • • • • • • • • • • •	- REMAIN	IN G	CAMPLES DICT	ARDED
		[			1.		<u> </u>	<u> </u>		<u> </u>
	<u> </u>				ł	·				
					1	·				
Į	30		_	]						
	► W <	<u> </u>			1					
				<b> </b>	4	·		·		
		L			[					
										,
	<u></u>				1					+
	- /-				{				<u></u>	/ 

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

547 i-90

5-90				Ma	Eng C ater <b>BC</b>	gineering l onstruction ials Engine DRING 1	Departmen n Division eering Sec	tion <b>T</b>		
								•		SHEET OF 3
PROJECT	Junary	Pt6 SI	 		NA	ME OF CONT	RACTOR		BORING NO. UST 7-1 R	SURFACE ELEV.
LOCATION	iont of	BULDING	537	->8	'w	EST OF	0577-1	A	CONTRACT NO. 47.6-99-006	DATE 71/28/00
SPOON		CASING S	ZE HOLE	TYPE				GR	OUND WATER LEVEL	
*(	D.D	"I.D.		/		Date	Time	Depth		Remarks
HAMMER		HAMMER								
#	FALL	•	# FALL					<u> </u>		
DRILLER	$\tau$	RAIL								
INSPECTOR	<u> </u>	/14/16				· · ·				
$\widehat{\mathbf{D}}$	DAVIS			i						
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6*	RE- 1 COV'D	SAMP. ² NO.	'  		3S.	AMPLE DE	SCRIPTION AND REMAN	RKS ILE
HAND	L H	Aver.			F	4. STANE	AND WO	Toc		
Dalling	<u> </u>	1	1	1 (	<u> </u>	<u></u>	<u></u>			
Tivber		<u>├──</u>			+	En BLA	K STAINE	D FINE	TO MEDIUM SAN	D-DETROIEUM CIDA
		<u>-</u> +/		12						
			-{		1					
<u> </u>	► 5 ◄			1				·		
					<u> </u>					
		l					-Au	Sami	NES SCREE	NED WITH
							- 4 1/10	mE TCK	······································	•_
				}				C 2		
	► 10 ◄				<u> </u>		<u> mp</u>	UE L	SAVED FOR	ENVIRON MIEWIAL
<u> </u>							_ 1BI	V6		<u></u>
		· ·		ł			- Au	RAMA	NINL SAMPLE	3 DICARDED
<u> </u>			1							
·	┝─ ─-									
	15							·		
	┝ ─				-		<u> </u>			
<u> </u>	┝── ──							·		
<u></u>										
	-									
<u>.                                    </u>	▶ ◄		+		$\vdash$					
	┝		<b>_</b>							
	└ <u>─</u> ` ──		1		$\vdash$					
								<u> </u>		
	1	1			1				-	

1

•

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

<u>a 547</u> 5-90

•

547			П	<del>i</del> e po	RT AUTHO	<b>RITY</b> OF NY	78 NJ	•
90					Engineering I	Department		
					Constructio	n Division		
				Ma	aterials Engine	ering Section		
					BORING I	REPORT		
								SHEET OF 2
ROJECT	. 1				NAME OF CONT	RACTOR	BORING NO.	SURFACE ELEV.
Hauha	JD Hor	141.5	175		(2011)	Dallin	115T7-1 C	
OCATION	NY TWO	+ 110 2			1_C19110	MERCING		DATE .
lanat	17. 2011	NUV C 21	-> 0	1 50	IT OF U	K+7-1.A	426-99-006	11/28/00
POON	IF DULL	VING J ST		TYPE			CROWNE WATER LEVEL	1 11/2 - 1
				1	Data		GROUND WATER LEVEL	Pomarka
	0.0.	-I.D.		L	Date			Incinai Ka
AMMEN .		- Incomment		_			Î Î	
	FALL		FALL					
niccen	$ \prec  $	DA11						
ISDECTOR		19116	<u>.</u>			<u> </u>		
SPECIUP	Day	Dudic						
	VAN	VHID	T				l	
	DEDTU	SPOON		SAMP.		³ SAMPLE	E DESCRIPTION AND REMA	
		DLUHS/0		1.0.			UNATES CHANGE OF PROF	
Awy A	<u> </u>	HAND AVGER	TVU	<b>」</b>	JTANE A	ND DARK BROW	N JANJU	<u></u>
PUPPO .		1			F14 - 11	HT BROWN FI	NE TO MEDIUMSAND	· ·
_	<u> </u>	<u>∤</u>	<u> </u>	<u> </u>		1.11	NE TO MEDINAL CAILO	- DESTRUCTION ODAR
				12	1 714 - 24	tur statiney f	INE TO TREDITIN SHAD	
		V		<u> </u>				
	<u> </u>		<u> </u>	1.			to to come	
					- 001	tum of B	OICING (CONCRETE	(, (, , , , , , , , , , , , , , , , , ,
	F 2 -			]	· · · · · ·			
	<u> </u>			$\left\{ \right.$	<u>├</u>			
	Ţ			]				
	├			{			· ·	
	[			]				
	▶ ◄		<u> </u>	4	<u>}</u>			
	I			]	{			
				]	- A.			
	┝────		}	-	<u>۱</u>	- SIMPLES	XREENEY WITH	PID MEJEIC
					- Al	L SAMPLE	S DISCARDED	
								· · · · ·
	<u>├-</u>	<b> </b>	<u> </u> ····	{	├			
				]				
				1				
	├			1	<u>├</u>			
				]			<u>:</u>	
	┝── ──	l		{	├			
		L		]	1 '			
				]				
	▶ ◄			4	├			
				1				
<u>}</u>	┝── ──			ł				
				1				
				1				
	L			Į	L			
	Į		l.	J	]			
	▶ <			L	L			

- Internation of an an and the state of the

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

•• • • • • •

lind.

and a national

LOW PORCOVERS, 100

Engineering Department Construction Division Materials Engineering Section BORING REPORT

									SHEET OF 3
ROJECT	<del></del>				NAME OF CONT	RACTOR	В	ORING NO.	SURFACE ELEV.
Port	lvory	PEG			Craig a	Irilling	[ ]	UST 7-2	
		011 -118=	15'E 0	railrow	d. J:∘	J		CONTRACTINO.	DATE
<u>± 15 N</u>	lorth of	Didg 361	Dlock	140	o Lot I			<u>426-JJ-006</u>	11-21-00
7	- 9 3/o		ZE HOLE	TYPE	Data	Time	GROU	ND WATER LEVEL	amarka
	D. 270	HAMMER	<u>SI</u>	·	Date	I Ime	Depth		
160			# FALL	-	11.91.00	1. Sand		SAMOL HL	
RILLER	0	<u> </u>			11-21-00				
-	. L'ra	iq							
SPECTOR	ר ר <b>ר</b>	J						· · ·	
r	<u> </u>	arks	· · · · ·	]	<u> </u>	L			
	DEPTH	SPOON BLOWS/6"	RE- 1	SAMP.2		· 3SA	MPLE DESC	RIPTION AND REMAR	KS
utter Head	► 0 <	Culter Head	Full	<u> </u>					<u> </u>
andaugo		Handauger	<u>   </u>					ABC	
				1	Fill groush-	black c-1	SAND 5	ome Gravel to SIT	Γ
T							C C TT		
			<del>     </del>	9	tull reddish-1	prown Clay	<u> ч</u> эг., і	Line oravel, some in	
		┟────┼────	$\left\{ \begin{array}{c} \\ \end{array} \right\}$		SAME	· · · · · · · · · · · · · · · · · · ·			<u> </u>
	- F -								
	5			ス	F: 11 J. 1: - 1		Group	and c-l CAN	UD some CLAVEY O
allow			¥		I'll readis	V- ntomi	VIWIEI	, and a sm	Lo some ompere
		6-6		1					
		10-10	24'	4	Realdish -	brown	Silly C	<u>IAY and Gra</u>	ve
				*			· 1	•	
			9/11	E	CANE				
		10-10	~4	2	SAME		- <u></u>	- <u> - Loil /</u> -	
		5-7							
		11-11	2.4"	6	SAME			(ail)	•
		5 -	_~~						····
		5-1							
		8-8	24"	7	SAME			<u></u>	<u></u>
		10-15		1					
	15	9	91.11	እ	CAME				
		10-10	<u>~H."</u>						1
									k
					Note: 250	imples s	aved	for testino	7
					All -			T and will	
						mer sam	165 <u>501</u>		
	- 20 -				<u></u>	eter 2	discarde	al P	occom of Borin
Ţ									
			ļ						

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

<u>A 547</u> 6-90 n es**e**ncer

	·		Ma	Construction Di aterials Engineerin	vision ng Section		
			. 1	BURING RE	PURI		SUFET OF 7
	- <u> </u>			NAME OF CONTRAC			
PIT	PIC			MAME OF CONTRAC			SUNFACE ELEV.
ATION	ry 140			Craig ari		CONTRACT NO.	DATE
Roll of	RIJ RO R	515	- P = = =	ilmand	U	426-99-00	6 11-20-00
ON N U	CASING SI		TVPE	II POGO			
λ το 7	3/8 10 Augo	rs	<b>i</b>	Date	Time Depti		Remarks
MER	HAMMER		- <b>I</b>			1 1/	
O # FALL	30 -	# FALL		11-20-00 h	:30 5.0	Sample#	5
	raig						
ECTOR -	Zarks .						
SING	SPOON	RE- 1	SAMP.2		SAMPLE D	SCRIPTION AND REMA	RKS
DEPTI	H BLOWS/6"	COV'D	NO.		LINE LOCA	LES CHANGE OF PROF	الت ۵.1
andre	mandaliger	rwi L	]				<u></u>
				Fill black	c-fSA	ND some brave	Lt. Silt
			*	L	· · · · · · · · · · · · · · · · · · ·	<u> </u>	
			2	SAME			
<del>  -</del>		+ + +			<u></u>		
5	◀	┼╌╂──					
	¥		<u> </u>	SAME		·	
	スース	-	°,				
acro		. 0 11			$\overline{1}$ $\overline{1}$ $\overline{0}$	CAID III O	
<u>e</u>	1 3 - 2	1 10	1 11	INGO THE P		THANK (ITTO TERAID	
		10			mach c=1	Unite prave	1-11- Mood uno
	3-2				mack C=		
	3-2	24"	5	SAME	( NO V		
10	3-2 8-5	24"	5	SAME	( NO V	<u>vood</u>	
10	3-2 8-5 5-8	24"	5	SAME Fill black c=	f SAND & C	vood) Gravel, tr Sill.	
10	3-2 8-5 5-8 10-11	24"	5	SAME Fill black c- Reddish bro	f SAND & C	<u>uood</u> ) <u>irauel</u> , <u>tr Sill</u> .	
10	3-2     8-5     5-8     10 - 11 $4-8 $	24"	5 6	SAME Fill black c- Reddish bro	f SAND & C	<u>vood</u> ) <u>bravel</u> , <u>tr SilT.</u> ( lay ,	
10	3 - 2 8 - 5 5 - 8 10 - 11 4 - 8	24"	5	SAME Fill black c= Reddish bro SAME	f SAND & C	<u>vood</u> )	Linders
10	3-2     8-5     5-8     10-11     4-8     10-17     10-17     1	24" 24" 20"	5 6 7	SAME Fill black c- Reddish bro SAME Black c-S	f SAND & C wn Silty C SAND , sen	uood) ravel, tr Sill. lay	Linders
	3-2     8-5     5-8     10 - 11     4-8     10 - 17     4-2     4-2     4-2	24" 24"	5 6 7 * 4	SAME Fill black c- Reddish bro <u>SAME</u> Black c- Same	f SAND & C SAND & C SAND & C	vood) Gravel, tr Sill. ay,	Linders
10	3-2     8-5     5-8     10-11     4-8     10-17     4-2     3-3	24" 24" 20"	5 6 7 * 4 88	SAME Fill black c= Reddish bro SAME Black c= Same Brown Pee	f SAND & C SAND & C SAND & C SAND , sen	Lood) Gravel, tr Sill. Lay Le Gravel, tr Silt.	
10	$     \begin{array}{r}       3 - 2 \\       3 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\     \end{array} $	24" 24" 20"	5 6 7 * 4 88	SAME Fill black c= Realdish bro SAME Black c= Sane Brown Pee	f SAND & C f SAND & C wn Silty C SAND , sen t w. Silti	Lood) Gravel, tr Sill. Lay, Le Gravel, tr Silt. Clay & fiber Destrolies	Linders
10	3-2     8-5     5-8     10-11     4-8     10-17     4-2     3-3	24" 24" 20" 20	5 6 7 * 4 88	SAME Fill black c- Reddish bro SAME Black c- Sane Brown Pee Ibote: 3 Sam	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>wn Silty</u> <u>C</u> <u>SAND</u> , <u>sen</u> <u>t w. Silti</u> <u>ples</u> <u>save</u> a	Lood) Gravel, tr Sill. Lay	Linders
10	3 - 2     3 - 2     8 - 5     5 - 8     10 - 11     4 - 8     10 - 17     4 - 2     3 - 3	24" 24" 20"	5 6 7 * A 88	SAME Fill black c= Realdish bro SAME Black c= Same Brown Pee Note: 3 sam All other	<u>f</u> SAND <u>e</u> <u>f</u> SAND <u>e</u> <u>sand</u> <u>silty</u> <u>C</u> <u>sand</u> <u>sand</u> <u>samples</u> <u>s</u>	Vood) Gravel, tr Sill. Lay, Le Gravel, tr Sill Clay & fiber foc testing reened with PID n	Cinders mich
10	3-2     8-5     5-8     10-11 $4-8     10-17     4-2     3-3     $	24" 24" 20"	5 6 7 * 4 88	SAME Fill black c- Reddish bro <u>SAME</u> Black c- Sane Brown Pee Note: 3 sam All other & d	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>sand</u> <u>sand</u> <u>samples</u> <u>sa</u> <u>iscardeol</u>	Lood) Iravel, tr Sill. Lay, Le Gravel, tr Sill Clay & fiber foc testing ceened with PID 10	Linders
10	$     \begin{array}{r}       3 - 2 \\       3 - 2 \\       8 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\     \end{array} $	24" 24" 20"	5 * 6 * 6 * 6 * 6 * 6	SAME Fill black c- Reddish bro SAME Black c- Sane Brown Pee Note: 3 Sam All other	<u>(No</u> <u>f SAND</u> <u>e</u> <u>SAND</u> <u>e</u> <u>SAND</u> <u>sen</u> <u>samples</u> <u>samples</u> <u>samples</u> <u>saue</u> <u>iscardeol</u> .	Clay & fiber foc testing	Cinders
10	3 - 2     3 - 2     8 - 5     5 - 8     10 - 11     4 - 8     10 - 17     4 - 2     3 - 3	24" 24" 20"	5 6 7 * 4 88	SAME Fill black c= Realdish bro SAME Black c= Same Brown Pee Note: 3 sam All other & other	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>sand</u> <u>san</u> <u>sand</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> <u>san</u> (san <u>san</u> (san <u>san</u> (san <u>san</u> (san san san san san san san san san san	Dravel, tr Sill. Dravel, tr Sill. Lay	Cinders
	$     \begin{array}{r}       3 - 2 \\       8 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\     \end{array} $	24" 24" 20"	5 16 7 * 4 88	SAME Fill black c= Realdish bro SAME Black c= Same Brown Pee Note: 3 sam All other	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>sand</u> <u>san</u> <u>sand</u> <u>san</u> <u>samples</u> <u>s</u> <u>iscardeol</u>	Vood) Pravel, tr Sille Lay, Le Gravel, tr Sill Clay & fiber foc testing reened with PID n	Cinders Cinders Cinders brick Cinders brick S relin Rottom of Boring
10	$     \begin{array}{r}       3 - 2 \\       3 - 2 \\       8 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3     \end{array} $	24" 24" 20"	5 6 7 * 4 88	SAME Fill black c= Reddish bro <u>SAME</u> <u>Black c=</u> Black c= Brown Pee Note: 3 sam <u>All other</u> <u>Brown Pee</u>	<u>f SAND</u> <u>v</u> <u>f SAND</u> <u>v</u> <u>sand</u> <u>silty</u> <u>C</u> <u>sand</u> <u>sand</u> <u>samples</u> <u>se</u> <u>liscordeol</u>	vood) iravel, tr Sill. ay, e Gravel, tr Sill. (lay & fiber for testing reened with PID n	Cinders II.
10	$     \begin{array}{r}       3 - 2 \\       3 - 2 \\       8 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       3 - 3 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       5 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\       \hline       4 - 2 \\       5 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       5 - 8 \\       10 - 17 \\       4 - 2 \\       5 - 8 \\       4 - 2 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 \\       5 - 8 $	24" 24" 20"	5 <i>6</i> <i>7</i> <i>*</i> <i>8</i> <i>8</i>	SAME Fill black c- Reddish bro SAME Black c- Sane Brown Pee Note: 3 Sam All other	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>wn Silty</u> <u>C</u> <u>SAND</u> <u>sen</u> <u>samples</u> <u>se</u> <u>iscardeol</u>	Clay & fiber foc testing	Cinders II.S Cinders II.S Ci
	3 - 2     3 - 2     8 - 5     5 - 8     10 - 11     4 - 8     10 - 17     4 - 2     3 - 3 $     3 - 3     $	24" 24" 20"	5 6 7 * 4 8 8	SAME Fill black c= Reddish bro SAME Black c= Sane Brown Pee Note: 3 sam All other	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>wn Silty</u> <u>C</u> <u>SAND</u> <u>sen</u> <u>st w. Silti</u> <u>st w. Silti</u> <u>samples</u> <u>se</u> <u>iscardeol</u>	Clay & fiber foc testing	Cinders II.
	$     \begin{array}{r}       3 - 2 \\       3 - 2 \\       8 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\     \end{array} $	24" 24" 20"	5 16 7 * 4 88	SAME Fill black c= Realdish bro SAME Black c= Same Brown Pee Note: 3 sam All other	<u>f</u> SAND <u>b</u> <u>f</u> SAND <u>b</u> <u>san Silty</u> <u>SAND</u> <u>son</u> <u>san Silty</u> <u>san Silti</u> <u>san Silti</u> <u>san Silti</u> <u>san Silti</u> <u>san Silti</u> <u>san Silti</u> <u>san Silti</u>	Vood) Gravel, tr Sill. Lay, Le Gravel, tr Silt Clay & fiber foc testing reened with PID n	Cinders wick
	$     \begin{array}{r}       3 - 2 \\       3 - 5 \\       5 - 8 \\       10 - 11 \\       4 - 8 \\       10 - 17 \\       4 - 2 \\       3 - 3 \\       \\       \\       4 - 2 \\       3 - 3 \\       \\       \\       \\       \\       $	24" 24" 20"	5 * 6 * 6 * 4 * 4 * 8	SAME Fill black c= Reddish bro <u>SAME</u> Black c= Sane Brown Pee Note: 3 sam All other	<u>f SAND</u> <u>e</u> <u>f SAND</u> <u>e</u> <u>sand</u> <u>silty</u> <u>C</u> <u>sand</u> <u>sand</u> <u>samples</u> <u>s</u> <u>iscardeol</u>	Clay & fiber foc testing	Linders III

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

.

<u>A 547</u> 6-90

`



3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

E	<u>^</u>	<u>547</u>
	6	90

ĺ٦

•--

;

à

Engineering Department Construction Division **Materials Engineering Section** 

# **BORING REPORT**

			- • • •			-			SHEET	OF
PROJECT	010	······································	NAME	OF CONT	RACTOR		BORING NO.		SURFACE EL	EV.
Port Ivory	$F \neq G$		Cro	h oic	rilling		FS-1			
LOCATION U				J.			CONTRACT NO	. ,	DATE	
±15'S of	Bldg 12 (From	t side } B	Slock	140	o lat 1		426-99-	006	11-17-	00
SPOON	VASING SIZE   H	OLE TYPE				GRO	UND WATER	LEVEL		
*O.D.	"I.D.			Date	Time	Depth		Re	marks	
HAMMER	HAMMER									
# FALL	# # FALL	-								
<u> </u>	urns								·	
INSPECTOR T 7	sele									
		1 SAMD 2	┍━┖─		354				S	·
BLOWS/FT. DEPTH	BLOWS/6" COV	D NO.			Ľ	NE LOCAT	ES CHANGE	OF PROFILE		
ANDALIGER	HANDAUGER Full	Rec A	Fill	black 1	, dark bro	WD C-	SAND S	ome Grav	el tr. SiTT co	al cinder
			511	11.1				C CAND	Grand	
-+	┼──┼─┤			readish	.prown_a	and car		1-24120	_ w_vravel	•
		9.	SAI	<u>1E</u>						<u> </u>
V	¥ \$	~~~~								5
					<u> </u>	- <u> </u>	CIAR	1.6	- <u>-</u>	
▶ 5 ◄					Lonc	r <u>ete</u>	JAD -	obsir	u crien	7
									<b>0</b> .	7-
						- <u> </u>			Bottom	of Bo
										1
								· ····		
								- <u> </u>		
	┝ <i>┈┈╸</i> ╸╸╸╸╸╸									
ſ										
	<b>├</b> ··· <b>-</b>	-				·				<u> </u>
	·									
	<b></b>									
'										
								·		
								· <u> </u>		
·								·		
		-7 I								
								·		
			<u> </u>			. <u></u>		. <u> </u>		
		الــــــ								

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.



2-U = undisturbed; A = auger; OER = open end rod; V = vane

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547			T	IE PO	RT AUTHORITY OF NY & NJ
6-90					Engineering Department Construction Division
				М	laterials Engineering Section
					BORING REPORT
PROJECT					NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
Po	rt Ivo	ry PelG			Craig drilling FSIB
LOCATION	111 (	HUTCIL	1. 011	a	ALL D D L L L L CONTRACT NO. DATE
40 SPOON	Wol			TYPE	$\frac{1019913}{3} = 1000000000000000000000000000000000000$
3 .0	.D. 23	18-1.D. Auger	8 1		Date Time Depth Remarks
HAMMER C	FALL 7	HAMMER	FALL		11-17-00 11:20 7.0' Sample #4
DRILLER	SP	Durns	<u> </u>		
INSPECTOR	<b>ゴ</b> フ	arks			
CASING	J. 4	SPOON	RE- 1	SAMP.	² ³ SAMPLE DESCRIPTION AND REMARKS
BLOWS/FT.		BLOWS/6"	COV'D	NO.	LINE LOCATES CHANGE OF PROFILE
nandauge		nandauger	rul nec	4	Fall
					Fill greyish - yellow e. CSAND AND Silty CIAY, cinders, wood, cool
				2	Fill previse block fraud come of SAND to Sill gindere
	► 5 ·			3	Fill grouish black Gravel some c - I SAND + SITT cinders 6.0'
4		4 - 7		* , A	Mise Fill Willowish - white dietomaccous, grey e - [SAND, Gravel, cinders, wood
AUGERS		4-3	18"	4 2	5 Fill greyish - black Gravel some c- ( SAND + sill cinder
		1-4			
	- 10 -	1-0	24"	5	Fill areyish - White Dietomaceous - with gray c. ( SAND & Gravel.
	10	Wight OC		1	
		HAMPLE R	0	6	(Sample fall into the hole )
	<u> </u>	W o H	18 "	* ۸	Fill grey Dietomaceous - with grey C. I SAND & Growel 13.
		18" Pounce	100/	7	<u> </u>
					+
	- 15-		- <del></del>		Refusal - Bottom of Baring
					Note: 3 SAMPles saved for testing
	4				All samples screened with PID meter
					the the complex licensed
	<del>.</del> -				The owner sumples discrided
	- 20 -	◀			
					<u>├</u>
×					
	• -				
	NOT	ES: 1 - Length re	covered: 0	" - Los	ss of Sample, T Trap used

1

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

AN AN

ś 200

			,		· · · · · ·
<u>A 547</u> 6-90			T	ie poi	RT AUTHORITY OF NY & NJ
•				· .	Engineering Department Construction Division
				Ma	Iterials Engineering Section
				a de la companya de	SHEET I OF 3
PROJECT	-	PIC	- 151		NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
LOCATION	vory		<u>al (</u>		CONTRACT NO. DATE
±50	NE of	Blog 12	Plock	1400 TYPE	$\frac{1}{426 - 99 - 006}   11 - 17 - 00$
3 .(	D.D. 2 3/8	-I.D. Auge	rs I		Date         Time         Depth         Remarks
LAMMER 9	Safety Z	HAMMER			N-17 or PM + 0 Complette
DRILLER		<u> </u>	FALL		1. 1. 1. Someren 1
NSPECTOR	J. DL	arns			
	TZ	ARKS			
CASING	DEPTH	SPOON BLOWS/6*	RE- 1 COV'D	SAMP. ² NO.	³ SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE
· ·		HANDAUCE	Full Re	د ا	crushed Rock DGABS
					BALLAST - Crushed Rock 20
	F -				Fill red-brown C-P SAND tr. Gravel tr. SI/T
				2	Same
	5				
		*	¥	3	SAME
5W EM		3 - 3		1	
Augus		3 - 4	12"	4	Fill grey c-[ SAND tr Gravel tr SIT
	·	8-9		*	
	16	9-10	24"	5	_ SAME (with wood splinters)
		5-3		1	'
		8 - 25	20"	6	SAME (with wood)
	┡	40-28			
	└─	38-25	24"	_/_	till grey c-f SAND & Gravel tr Sill wood
	15	20-12		R	
	∟	18-27	20"	0	Ell much block of CANN & Count I Silt of the
	┝─ ──	24-38		í a Î	
*	┡	68-88	<u>24"</u>	xJB	Greyish-green c- (SAND
		· · · · ·			L
	20 -				Note: 3 samples saved for Testing
	└── ⁻ ──				All other samples screened with Bottom of Bari
·					VID meter 2 discarded
					· · · · · · · · · · · · · · · · · · ·

3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-00			TH	IE PO Ma	RT AUTHORITY OF RYGRU Engineering Department Construction Division aterials Engineering Section BORING REPORT
	ort Ivo	ry P&	G		NAME OF CONTRACTOR     BORING NO.     SURFACE ELEV.       Craig     drilling     FS-2-3       CONTRACT NO.     DATE
SPOON	As laid	<u>casing si</u>	the fi ze hole	TYPE	GROUND WATER LEVEL
3 "0	.D. 2 3/8	-1.D. Auger	s I		Date Time Depth Remarks
HAMMER	FALL 3		# FALL		11-15-00 11:45 8.5' Sample # 5
INSPECTOR	<u>).0si</u> 77	ich	<u> </u>		
CASING	J-40	SPOON	RE- 1	SAMP.2	2 3SAMPLE DESCRIPTION AND REMARKS
BLOWS/FT.		BLOWS/6"	COV'D	NO.	LINE LOCATES CHANGE OF PROFILE
nunualiger		Handauger		1	Hise Fill prevish-black C- SAND& Gravel to Sill cinders wood coal etc
				<b>B</b>	
				2	Fill brown C-PSAND tr. Gravel tr. Silty CLAY tr. Cool
				3	SAME
<b>N</b>		4 - 4		· , ×	6.5
AUGERS		5 - 5	14"	4	Ell reddish - brown CLAYEY SILT : Little C. C SAND to Grave
	·	4-4			fur - where a contract of the set of the set
		17_14	9.11	5	SAME
	- 10 -	4-5			
		6.5	20"	6	Fill brown C-C SAND AND Rod-brown Claver Sill little
		2 - 9			SAME 12.8
		3-4	24 "	7	Brown PEAT Some area CIAY 14.0
					1
	- 15 -				Note: Samples 2:4 27 were saved for testing 2
					all samples were screened with Pin meters
					and the other samples were discarded.
			·		
					Bottom of Boring
		S ¹ 1 — Length re	covered: 0	# _ 1.0e	e of Sample T Tran used

1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

د برور د رویهمهر ورونه

参照る

ŝ. ŝ

1.

Ξ.

Engineering Department Construction Division **Materials Engineering Section** 

 $\hat{k}^{i}$ 

#### **BORING REPORT**

		,								SHEET OF Z
PROJECT	T		<u>~ ^</u>		N/	AME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
L Te	ort L	Voru Y	& G_		L	Craia	drilling		<u> 15-4</u>	
LOCATION	1.	,	0	• •		- J		)	CONTRACT NO.	DATE
A	à laid	out in t	he I	ield		T			426-33-006	11-15-00
SPOON	. 3/.	CASING SI	ZE HOLE	TYPE		·	·	GRO	UND WATER LEVEL	
5	D.D. L 10	"I.D. Auger	5			Date	Time	Depth	Re	marks
HAMMER	7	HAMMER					, pM	σ.	C. 1 16 - 1	r
	FALL		FALL			11-15-00	1.40	0.0	Jample 7 D (	top
UNILLER	D 0:	such								
INSPECTOR						}	<u>├</u>			
		Zarks								
CASING	J	SPOON	RE- '	SAMP.2			3SAN	IPLE DES	CRIPTION AND REMARK	(S
BLOWS/FT.	DEPTH	BLOWS/6"	COA.D	NO.			LIN	E LOCAT	ES CHANGE OF PROFIL	E 0.0
HANDAUG	n U	HAND ANGER	Full Rec	*	-	<u></u>		GABC		ers
┠╍╍╉╼╾┥			┼╌╂╌╌		-		0.0	·		F
	L					Ill grey	<u>c-f San</u>	10 and	L'Growel, tr. SII	
					1		·			
	<u> </u>			9	2			ACA		
			┠──┠───	~~~~~	+	III _greyis	b. black	c-ton	NU some brovel sr. 2	ill, cin all's, coal, word
	5			_			· <u> </u>	· ·		
<b>J</b>		1 [		3		SAME				
Here .	· ·				1					
M I		4-5	l	/.	$\vdash$					
AUGER		5-5	スイ"	4		SAM	E			
					T.				· · ·	
		4-4		1	$\vdash$		<b>_</b>	·		
	10	9-3	20	5	L	<u> </u>	i (w	2ea		
		8 - 12.						,	•	
· · ·		a /	a N	6	Γ	SAME	,	· <u> </u>	· · · · · · · · · · · · · · · · · · ·	
<b>├</b> ── <b>├</b> ───	┝	0-6	6		┢╌	_JAPIC	- <u>(wor</u>	×q_} _		
		9-9		_						
		4-4	9 21	17		SAME				16.01
			<b>⊢</b> ₽		T					
	► 15·	2-2		*						
		2-2	20"	8		Brown	reat	tr. on	w silty CLAY	Ka'
								0	J J	
					$\vdash$					
L	L					lote: _	amples 1	;2.L_	8 saved for t	esting
						All sor	noles si	TRENA	al with Pill me	ten /
					F	 	n [	1 1	ан - наран - н Гр	11 12
	► 70 ·	┫────	·		┝	_ ana	discar	ded_	<b>p</b>	tom_of_poring
	L		<u> </u>		L				<u></u>	
		•					-			
Pi	┝	+			$\vdash$		_ <u></u>			
	└──								·	
		1					÷.,		·	
	⊢				F					
L		┫		L		- <u>.</u>			· · · · · · · · · · · · · · · · · · ·	
	NOT	ES: 1 - Length re	covered: (	)" Los	s of	Sample, T	- Trap used		;	1

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

**Engineering Department Construction Division Materials Engineering Section** 

#### **BORING REPORT**

									SHEET OF 3
PROJECT		2		``	NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
1 4	011 T	vouy P+6 S	ite		Crase	Duillin	8	FS-7	
LOCATION					0		0	CONTRACT NO.	DATE
	vest o	F RL2 55						416-99.001	12/1 100
SPOON	~3	CASING SI	ZE HOLE	TYPE			GRC	UND WATER LEVEL	·
3.0	. <u></u>	8-1.D. Auc 01		1	Date	Time	Depth		Remarks
HAMMER	S&F	HAMMER			12/1/00	0825	5'	HA	
DRILLER	FALL 7		TALL						<u> </u>
Det	- CVE	1ig		<u>.</u>		ļ			
INSPECTOR	Charl.	e Sovin	e V						
CASING	<u> </u>	SPOON	RE- 1	SAMP.	2	3SA	MPLE DES	SCRIPTION AND REMAI	aks
BLOWS/FT.	DEPTH	BLOWS/6"	COV'D	NO.		LI	NE LOCAT	TES CHANGE OF PROF	
Venterne		Hend energy	Full		Concione				.55'
TRACEV		MANC OUJEV			Grant Fill	- ufford N	attox -		<u> </u>
				*ر	Fine TO ME	d Save	] TVARA	SIT/Some Gungl	DAKBrown
				4			•		40
				~	·		<u> </u>		
}	► S ·	┫		3					
V		<u> </u>	$\downarrow$		Fines	T SNO	Voce	SILT LT	BIOWN
yev.		8-14	21	4					
		Ц-10		1	Sane				
		12-14	21	54	·				
		16-18		5	Sine				
	10	4-6	2/	1					
		10-7	1	6	Sine				120
		1				BOTIUM	oF	Boring	
					└── ──	<u>v</u>			
		1			F				
	ר כ <b>ו</b> -	¶			-4151	ples c	۔۔ ۔۔۔۔ ۔۔ لہمیروں	WPTD	
					- Sanph	\$ 2.5	Ś	loved For En	V. Jossing
					- 01	Remarin	IS SAI	nlos Discur	Led
		1			F		Ч —— Ч		<u> </u>
		]			<u>├</u> ────				
	> 20 -	┫			h				
		+			┝	<u> </u>			
						~			
									-
					<u>├</u> ─ ─ ─				
					<u>├</u>		<u> </u>		

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

ļ

Engineering Department Construction Division **Materials Engineering Section** 

#### **BORING REPORT**

								SHEET / OF 3
PROJECT	····· <u>·</u> ······························				NAME OF CONTRACTOR	B	ORING NO.	SURFACE ELEV.
Handa	of Marik	Pert Timer	Pta	Site	Crais Drilling		Ħ11-1	
LOCATION					Book 1400 L	UTI C	ONTRACT NO.	DATE / /
1 75	South	wast of 1	Truck	Scale	Near South Ga	re' l'	126-99-006	11/7/00
SPOON	, つ	CASINGSI	ZE HOLE	TYPE	7	GROU	ND WATER LEVEL	
3.	D.D. 18	"I.D. Augus	1	Ľ	Date Time	Depth	R	emarks
HAMMER 140 #	FALL 30	HAMMER	# FALL	#	11/1/2 975	3.0	while Hand	Accoring
DRILLER	<u></u>	5 Burns						<u>0</u> 0
INSPECTOR		PHove						
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	3S	AMPLE DESC	RIPTION AND REMARI S CHANGE OF PROFIL	ks E Öc
Vand	<b>P</b> 0 <b>1</b>	May Apres	5.11	1	Crushod STUR	······		0,5
n	<u> </u>	I I I I I I I I I I I I I I I I I I I		1				
Huger	┣				FILL- M-1- BIOW	in Sonol	INSIT, In Crush	a sine
						·		
				12	MER	< ./ 1		
├	<u> </u>	<u> </u>	+ + -	<u>├</u>	proux	_ sanot _ J		
/								
				13	Samo			
		U_3	<b>*</b>	<u> </u>			- <u> </u>	 הר
06	₋	1)		44	Same			//0
STAron		3-3	194		M-F Grey	Sand IV	<u>SIIT</u>	
Auron		3-3			Note S#	4_"	Taken with ?	7' Spering
-	<u></u>	6-12-	104	15		1 6. 0		// /~
<b>├</b> ── <b>├</b> ──	► 10 ◄	10	10	$\vdash$	Prownia	h very =	DONOT LIVE SIT	/
	L	1-9		1				
		13-11	18'1	0	F Brown	Sand,	Tr SILT	
		6-11						
		11/14	164	17				
<b>├</b> ──¥─	┟		1.0					
ļ	- 21	4-6	ļ	8	Some		. <u></u>	15.0
		7-9	44	0	F Brown Sand,	Some SI	7	1513
								7
	╂			{				/
	<u> </u>		ļ				Balamot Ba	IN INT
			1					•
	<u>├</u> ─ _ ─			-	Alle	- aha	n 1 1 1 1 1 1 1 1 1	
	> 20 <			1	Sample	es <u>chect</u>	a wingrie	
	┣━. ━		<u> </u>	4		2 29000	for ENUIRC	<u>isjilizi</u>
	L				Remaining	4 Sam	165 Aiscarde	d
1			]			0 7		- <b>-</b>
			<u> </u>		└── -── -── -── <del>-</del> -			
	L		<b> </b>	1				
	74 -					_		

à.

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

۰.

Engineering Department Construction Division **Materials Engineering Section** 

#### **BORING REPORT**

									SHEET / OF 3
PROJECT					NAME OF CON	TRACTOR	1	BORING NO.	SURFACE ELEV.
Pr	octor.	2 GAME	slé		CRAI	67		Fill-2	
LOCATION		<del></del>			·			CONTRACT NO.	DATE
AS	MARKE	D out in	THE	FIELD	> RYKI	LLIAM A	ssoc.	426.99.006	11.3.00
SPOON	1.1.2.2.2.	CASING SI		TYPE	<del>````</del> _		GROI	IND WATER   EVEL	
2.	n 73/2	NID US AG	os.		Date	Time	Depth		Bemarks
HANNER /	alater	HAMMER							
140 #	FALL	30.	# FALL		11.3	Am	<u> 4.1'</u>	Found in	s#3
DRILLER	<i>.</i> ?	Pennell				1			
INSPECTOR	M.	Dudeh							
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.		3SA	MPLE DES	CRIPTION AND REMAI	rks ILE ,
1	$\triangleright $		1	ļ		ASPho	H Vaue	ment	Oiz
F.)	$\vdash$ $\sim$ $-$	ILAND	C.AD	<u> </u>	<u> </u>	CKUS	THE R		04
Augers	·	AUGHER	Fure		Fill- day!	= trail c	& SAND	some Grouply from	o Silt frace Caused Back
		/	└──┟──	<b></b>	<b>↓</b>				
		{			Fill - brai	Dr c. l	SAND .	ray Gravel 1	race Silt
			† <b>†</b>	12		<u> </u>	لے تشت ک	له الجنبية "منتشد" مستك" الم	<u></u>
	5 <	I	<b>↓↓</b>	10	<u> </u>				
	-								
		11.1		<u> </u>					
	<del>_</del>		20	へ	Fill-		_ > Ar	<u>ne</u>	
-1		67							
		5-6	20"	u	Fill-		> <u>A</u> ,	re	
	SID.	5-12		7					
		8.4	2.11	[					
		8-11	104	5	<u> </u>		Av	<u>nt</u>	
		14-17				0			
		2.2	21/1		5.11				
└─ <b>─</b> ┤		10-10	<u> ~7''</u> _	10	- <u></u>	_\		Ime	
. ↓		13-15							
		9-11	ייער		5.1				
	►  S  ◄		121-	12			<u>}P</u>	m <u>e</u>	
		18.22		/					16.0'
					[				$\wedge$
			<u> </u>	{	├				
									Dottom
									1his J
			<u> </u>	1			—,		og oring
				ł	<u> </u>	an plas	Wer ?	sourd	
	do			ļ	-	and of	Teal e	1 4 22 410	
			<u> </u>	{ ·	for the	ore pri-	J 1 2 3 5	- 1-7 <u>or var.</u> -	
								·	
			{						
			<b>├</b> ────						
				Î I					
	2	1							
	► 85 ◀		L	L	L				
	NOTE	S: 1 - Length re	covered; (	)" — Loss	s of Sample, T -	- Trap used			. · ·

÷ ·· · ø 1. 200 A.

A.S.

100

2

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

- ----

<u>547</u> 90				ie pui	KIAUIHUKIIY UP NY 图 NU
					Construction Division
				Ma	aterials Engineering Section
		:			BORING REPORT
OJECT	010				NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
414-	146				Crais Fill #3
	0 -+	1 11:01:	G	• • 1	+ 2' ORATE + 11 1 to 1/2(-69-02)
<u>Istai</u> 100N	d out	CASING SI	ZE HOLE	TYPE	GROUND WATER LEVEL
3 .0	D.D. 2 /8	"LD. HS-Que	en 1	-	Date Time Depth Remarks
MMER		HAMMER			11/100 10:050 75' 1 5#4
170 #	FALL 7		# FALL		1110100 10 - A 7.5 m ~ 1
	J. Craf	<u></u>			
SPECTOR	-0	3			
ASING	1.190	SPOON	BE_ 1	SAMD 2	2 3SÀNDI E DESCRIPTION AND REMARKS
OWS/FT.	DEPTH	BLOWS/6"	COV.D	NO.	LINE LOCATES CHANGE OF PROFILE
		Hand	Jull		Mise hill- Sand, Endro Brink Core
		Quan	Rec.		
		10	1		Sam ulto met.
		1		2	
	— —	1-1			
	- 5 -			3	
					La Grand
eno I		7-13			Sance
	<b>_</b>	N-24	11	A STATE	
	 	13-31		-	Dam!
_	10 -	54-45	22"	<u> </u>	· · · · · · · · · · · · · · · · · · ·
1		1 7-11		1	Same
1	· · ·	18-31	20"	6	
- i.		11-17		~	Same
$\mathbf{V}$		22-28	14"	1	
		4-10	12"	8	Sang 150
	- 15 -				A A A A A A A A A A A A A A A A A A A
······	<u> </u>				Patta IBari
·	·				
					······································
	<u> </u>		ļ		
					1 plo: Jamples - 4 were same for testing. I
	L				other samples were screened w/ PIDG the discardes
	L				Sample #4 was saved & placed on Hold.
	▶ ◄	t	I		<u></u>

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc. X - Tarks have been removed, foundation only effects.

**Engineering Department Construction Division Materials Engineering Section** PODING DEDODT

				1		OF 5
HH	- Pig				NAME OF CONTRACTOR BORING NO. 4 4 SURFACE ELE	<b>v</b> .
OCATION	id out	by Kallon Casing si		TYPE	103'So. of Ridg 12) HOUVELONTRACT NO. DATE 11610 GROUND WATER LEVEL	0
<u> </u>	D.D. 278	"I.D. augers	(44) 7		-4 Date Time Depth Remarks	
140*	FALL 37		# FALL		11/ 100 1:40 pm 5.9' Jm S#3	
RILLER	P. Per	nell				
ISPECTOR	Τ.	from				
CASING .OWS/FT.		USPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	² ³ SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE (Grove Work	und)
	- °	Hand	Jul	,	Till-Cinders Send, little Br, little Class	
		unjer	Kuc	<u> (</u>		
		<u>├}</u>	$\left  - \right\rangle$	2	Same	
		<u> _</u>			Samo	5.0'
	-5 -		$\bigvee$	3	Hell-BAM-F Sand little silt, to Grovel	
yrs		2-3		4	Same	
- <b>i</b>		3-6	<u>  19</u> .	<i>(</i>	Kn FSe D pome Sitt, Dome flent fibero (org. odur)	<u> </u>
		1-1	- 011	$\langle \langle \rangle$	Brill FSand, to Seet, to Gravel	
	/0 -	2-2	<i> 1</i> "_			
		7-7	1211	6	- deme	
		10-15			Sone	
$\mathbf{V}$		11-15	24"	1		
		11-14	12"	8	Same	15.0'
						₽-
	<b>—</b> . <b>—</b>					<b>├</b> ─ -
					Bourn groung-	
		·····				
			1. 			
					THE ATTENT	<del></del> _
	<u> </u>	······································			110U: Damples # 1- 5 were sever for the	An ALA
					E the disconded Sanali Swa sand i dand	mhol.
	NOTE	S: 1 — Length re	covered; (	0" Loss	as of Sample, T Trap used	
	b	2 U = undi , 3 Log depth	sturded; A n of chang	auger au	er; UEH = open end rod; V = vane r of wash water, loss of water, artesian water, sand heave in casing, etc.	

Ł

PA 547 6-90

• •

<u>°A 547</u> 6-90

١,

THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section

# **BORING REPORT**

								_				SHEET	OF	3
PROJECT	2	<u> </u>			N	ME OF CONT	RACTOR		BOR	ING NO.		SURFACE I	ELEV.	
Port Ivory	<u> </u>	6				raig d	Irilling		Lti	ill 5				
LOCATION		1.0.			Γ.	J	J		CON	TRACT NO.	1	DATE	•	
+ 180 L of	Bldg	2 - Bloc	k 140	o Lo		· · · · · · · · · · · · · · · · · · ·			4	16-33-0	00	<u> </u>	8-0	0
SPOON /	3/2	CASING SI	ZE   HOLE 	TYPE				GRO		WATER LEVEL			·	
3 "O.D. Z	78 "I.D.	Auger	<u>s</u>	L		Date	Time	Depth			Ren	narks		
HAMMER Salety	7	<b>NAMARC</b>					IN PM		1	Ballid	.(	Saul	41	
140 # FALL			FALL			11-10-00	16:30	1.5	-+-	Poulon	<u>_</u> )	Jamp	2#7	·
June June	es P	inch							- [	•				
							<u> </u>	ļ						
1	Lark	<u>،</u> ۲			ĺ									
	5	SPOON	RE- 1	SAMP.	1		3SA	MPLE DE	SCRI	PTION AND REM	ARK	S		]
BLOWS/FT. DEPT	TH BL	.OWS/6"	COV'D	NO.	4_		<u> </u>	NE LOCA	TES	CHANGE OF PRO	FILE			0.0
Handauger	HAN	idau gek ľ	Full Kec	ł	-		Crus	neol Roy	ck	LOGABC				_0.5
				1	T			<u> </u>	A 711	VIH &	1 0	it and		
<b>├</b> ── <b>┼</b> ── <b>├</b> ─			┟╌┼╌╸		+1	يلا _عدمين	5N-brown	r c-t л	AIV	2, un re braver	, tr	21 Cinde	<u> </u>	
				,	Ĺ					<u></u>	. <u> </u>	-,		
				2	P	T reddict	brown	29-0	MA	Dtr. Gravel	L S	it woo	al	
			1-1		╵	<u>n ream</u>	<u></u>						<u> </u>	
5			┠╌┠┯──	<b>_</b> _	-					<u> </u>				
		¥	V	5		<u>- SA</u>	ME_							
H	5	- 5 -		,										· .
SHEM	-17	 /	0 11		5			NN C	· ·	$r \rightarrow r$	11	 /// \\	Λ.	(+2"
AUGERY	- 6	- 6	120		Ц	Drow	<u>v c-</u>	SAND	,tr.	bravel, tr Ji	<u>II w</u>	ith bl. o	ganic ti	bers
	4.	- 3				<u> </u>	!			<u> </u>				
	2	5	20	5		SAMF	-							}
		<u></u>			┢		<b></b>							
┝╼╍╃╼╍┽╾	3	- 4		/	⊢				<u> </u>		·		<u> </u>	<u> </u>
		7 - 11	10	0		<u>SAME</u>	<u> </u>							·
	11	12												
		= =	11	7	5				0 0	AND I C	·		:1	_
		5-21	24		╀	ill <u>grey</u>	V browt	r <del>_ c .</del>	┢╴	SAND Tr UI	ave	왼 <del>, 1</del> 다 그	ЦЦ	
15	1	- 10	<u> </u>	0					<u> </u>			·		
		12	2.4"	Ŏ	(	THEN C	1A29	1D +	r.G	rated to	SI	IT		16.0
			~ ·				1-2-1-			, <u>, , , , , , , , , , , , , , , , , , </u>	<u> </u>	•••••		
<b>├</b> ───┤──		<u>-</u>	<u> </u>		$\vdash$						·			F
					Ц	lote: 25	amples_	Save	d	for testing				2
		4-	1			۸ <b>۱</b> ۱	ather E	om la	٦	Jaken J				1
<u>├</u>		·	<u> </u>		F	 	יד, גשוי צ ו	emples	_ <u>}</u>	I /			-	
20	> ◀	•	<b> </b>		$\vdash$	Y ID	meter 3	kdis	CA	colled			_/	
										Do	tto n	of Bo	iring	
												1	5	
<u>₽</u>		~	t		$\vdash$				<del></del>					
	_							- <del></del> -			·			
						•						· · · -		
					Γ									
L			L		1	0	<b>T</b>	<u></u>						]
. N	101E5: 1 - 2-	— Length re — U = undi	covered; ( isturbed: A	J ⁻ LOS \ = auge	s ol r; C	ER = open (	– ।rapused endrod:V ≕	vane						
	PD 3-	- Log depti	n of chang	e in colo	r of	wash water,	loss of water,	artesian	water	r, sand heave in c	asing	), etc.		
	V - [												*	
	T O.	<i>q</i>												

A 547				<b>IE PO</b>	rt authority of NM& NJ	
6-90					Engineering Department	
				Ma	aterials Engineering Section	
					BORING REPORT	
				•	SHEET	) OF 3
	T	plc er			NAME OF CONTRACTOR BORING NO. SURFACE	ELEV.
LOCATION	LUORY		<u> </u>		O CONTRACT NO. DATE	
Wisto	Awood	chipper,	Block	1400	LOT 1 476-99-00G 121	4/00
SPOON २	23			TYPE	GROUND WATER LEVEL	
HAMMER	Sata	HAMMER	<u>_</u>	e	1 005 11 ctt	
140 #	FALL 30	· /	FALL	"	124 9 10.0 5-6	·.
DRILLER		S Bure				
INSPECTOR		nu				<u> </u>
	·····	VNowe	<u> </u>			<u>.</u>
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- ' COV'D	SAMP. ² NO.	³ SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE	· 00
	$\mathbf{P} \mathbf{O}$				Con cristie	0,7
Houd		May AA	5.II		Nice Ell S d Gard C la luc / Site Elc	<u> </u>
N.		riano ijuer		17	Inix III, Sana, Orawi, Chang, Wood, Siri Jere _	
Hugh		<u> </u>		2	<u> </u>	
		+	├- <i>┟</i>		Jame	7,0
¹				3		
	L				Misc F. 11, Grobis Grew/ ETC	
Ilcu	·	16-11		υ		
STem		16-18	19"	1	Same	
Aurs		17-9		Ľ.		
		4-3	<i>J</i> 0'	5	Same	100
		3-2		/		
		1-1	24"	6	Eller D Targer Early	
-1-	┝ <u></u> ── ~──	1404				
	└	10011	2141	7		
		111011	54		<u>anne</u>	
	- 15 -	- HOM-1	0.01	8	<u> </u>	
	└ <u></u> ·	6-2	23"	ļ	<u>Scime</u>	
	<u>_</u> _	2-2		9		
		2-2	2411	(	<u>Same</u>	
	 	9-18		1.		
	2	25-45	15"	10	Fill- White Protomacove Earth LiDio Cudar Tr G	avel
	<0	7-3				
	<u></u>	2-7-	184	11	Ell-lehtaton O France Faith	
¥	└── <u></u> ──	3.7	10	N.	Can Canada Cara Cara Cara Cara Cara Cara Cara C	
	<u></u>	1_1	191	20	P. D.	20
·		''	10	5	Urown Ken	- 10
	► 25 <				Bottom of Boring	<u> </u>

449

2-U = undisturbed; A = auger; OER = open end rod; V = vane 3-Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc. P11 Samples Checked WTM P10 MeTer J# 1,3,26 Saud for Environ Tontal Romannia Samples Descarded

\$

÷

THE P	ORTA	JTHOR		NNGNU
-------	------	-------	--	-------

Engineering Department Construction Division Materials Engineering Section / **BORING REPORT** 

									SHEET OF 3					
PROJECT					NAME OF CONT	RACTOR	В	ORING NO.	SURFACE ELEV.					
Port	Lor,	VHC SITE			Craig	Vailling		<u>Fill-8</u>						
LOCATION	/	T orla	A		RI I		C	ONTRACT NO.	DATE					
NU C	GYNUY OT	Ste, 15 South	TE HOLE	MW-1	Ulack	1400 CC	,) /	486-97-006	1010100					
3-00				I IFE	Date	Time	GHUU	ND WATEH LEVEL	7 WATCH LEVEL Remarks					
HAMMER	S-fo	HAMMER	<u></u>			2	·							
140 #	FALL 30		FALL		12h	川方	3.0	while Hand	Acseria					
DRILLER	<	5 Burns												
INSPECTOR		0 Howe												
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- ' COV'D	SAMP. ² NO.		³ SA LI	MPLE DESC	RIPTION AND REMAP S CHANGE OF PROFI	rks Le oc					
Hand		Hand Auger	Full	)										
Buger				ļ	Misc F	<u></u>	andy Si	IT, Bret, Grac	of Motel, ER					
<b>├ )</b>				2										
						· · · · · · · · · · · · · · · · · · ·			<u> </u>					
	5			3	- FII-	Grey R	Tam ac	en Entla latt						
bio		1-1				<u> </u>		2005 6 4714 6101						
Jom		1-1	19"	9	<u> </u>	Gray P	atomi	cools Eatth						
Augus		woy		-		· ·			·					
		wal	23''	5	Fill - Le	hits & Gr	A, AsTo	macous Ear	<u> </u>					
<b> </b>		Woll		6										
		HOW	254		<u>S</u>	ame								
		leal												
		Worl	22'1		<u> </u>	ame								
	5	WOH		$\boldsymbol{\varsigma}$				~ ~						
		hoy	394	<u> </u>	<u> </u>	2me								
		Woll		9 #	<u></u>	ame			<u> </u>					
<b> </b>		won-1	<i>2</i> 4*	<u> </u>	Black	Poat	D	π. Λn.	180 7					
	<u> </u>			<b> </b>	┝── ──		<u>^</u>	DUMAY Mar	vy					
	- 20 -	◀												
		+		1		BIIS	Samale	cherkertun	The PIO Motor					
						57-1	24 Sa	ved for En	AK TosJuj					
						Rema	ININS S	amples asca	nded					
							0-2							
			mond l	17 1.00	of Sample T	Tran used								

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-00

Engineering Department Construction Division **Materials Engineering Section** 

### **BORING REPORT**

										SHEET	OFS	
PROJECT	「丁丁」	vov/	P+4	SITE	2	NAME OF CONT	RACTOR	_	BORING NO. FV11-10	SURFACE	ELEV.	
LOCATION			· · · · · · · · · · · · · · · · · · ·			0	<u> </u>		CONTRACT NO.	DATE	· · ·	
5	outh	OFF	3LD 7	2					426.99.006	12	100	
SPOON	~ 7	de la	CASING SI	ZE HOLE	TYPE			GRC	OUND WATER LEVEL		• 	
3.0	<u>.в. 27</u>	Y "I.D.	Augol			Date	Time	Depth	R	emarks		
HAMMER 140 #	FALL	30.	HAMMER	# FALL		12/1/00	1025	5.8'				
	eff (	Chais										
INSPECTOR	how	e Sr	NIMOL	,								
CASING BLOWS/FT. DEPTH BLOWS/6" COV'D NO.						³ SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE						
HA.		11	4.	Full		Fire See 5	0 7 July 0	-www.			912	
			<u></u>		1	FINE SAN	Same	 SI.T	Rod Bra			
		-			*		<u> </u>					
<b>├├│</b>					2	Fino S	INC CO	M SIL	T w/Grand -	Radi	2	
<u>├</u> ──- <u>}</u>					~							
			1		ک							
		5	5-8	1,3'	11 0	ار السرع		- Jundar			<u>bi</u>	
a gr		S	-23	-	4	-> arc 4	SLig	hTod	eV	e	C -	
	'	4	-4	21	C ·	2			· · · · · · · · · · · · · · · · · · ·		<u> </u>	
	10	4	- 4		ר   	- Fez		- <u></u> -			190	
					-		Eı	NOF	Boung 10'			
	_ <b></b>					Ļ						
	<u> </u>		•			<u> </u>					<u>_</u>	
			. <u> </u>			┝	<u>_ #1   56</u>	mp 2	Schenod u			
	► 1S						_ Sayph		5 + Y 5 md		~ [3] M	
	<b></b> .		<u>-</u>			L	- 011 R	nom	15 Saplos [	)isciv	Lord _	
				7								
			<u></u>									
	·					┝── ──						
	20	4	<u> </u>			└── ──						
						L						
			<u> </u>			┝						
	•	-				<u> </u>						
-	NC	TES 1 -	_ longth re	covered. (	# I oee	of Samola T	- Trop used					

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

÷

Engineering Department Construction Division **Materials Engineering Section** 

# **BORING REPORT**

										SHEET OF 3
PROJECT	. 1	$\rho = \rho$		/	NA	ME OF CONTR	ACTOR		BORING NO.	SURFACE ELEV.
YOR	1	IORY PTG	SITE			KAIG	KIUM	6	151	
LOCATION	ANZ IL	100 Rulin	211.	1615 1	<u>.</u>	2.11.0.	1 Ezi	15 lock	LIZ 96-004	17-14100
SPOON	jun 1.	CASING		TYPE		DUILIN			OUND WATER LEVEL	
3 .	.o. 73	8 "1.D. ANGE	RI			Date	Time	Depth	N NATER LEVEL	Remarks
HAMMER	5	AFETT HAMME	R			12/4/00	1230	ID to 12		<u> </u>
DRILLER	FALL	30 -1	# FALL				107-	10 10 11		
	JIN	n FINCH								
INSPECTOR	Da	N DAVIS	, 7							· .
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- 1 COV'D	SAMP. ² NO.	2		35A	MPLE DE	ESCRIPTION AND REM	ARKS DFILE 00
						CONCRETE				
									· · · · · · · · · · · · · · · · · · ·	2.0
Hono Ave	4	Harly AUGER	- Fur	7*		CINPERS I	ento sma	n Gra	Na	
				1						UT
				7 A	$\square$	SAME				4.5
	- 5			JB.	R	aprist Brow	N CLAVE	I SIUT, L	ITTLE GRAVE	
	<u> </u>	34	16"	1.*	- 2					· · ·
	·	56		4		20100				<u></u>
		3.5	15"	CA	4	SAME				9.0
		78		) B		BLACK FIN	E SMP, T	RACE SH	я 	
	10	34	14"	1		SAME				11.0
		46		0.	<u> </u>	Brown f	INE TO r	NEDIUM	SAND, TRACE SIL	я 120
		78	8"	~	Ľ	BROWN	PEAT_			
		9 11		<i>†</i>						14.0
					Ţ	1				
	- 15			1		TBM	tim 0	e 1	BORING	
				1				L ·		······································
	· ·			1		- Au	Some	165	SCREENED WITH	PID METER
				1		- Sem	ones 2	2.4	AND SB SI	WED FOR
	- 20			1		KAI	Vikasime	NTAL	TISTING	
				1		- Au	RCMA	nh.	CAMPIES DIS/	ARDED
ļ				1				INHNO	Sharren Min	
	<u> </u>	_		4	<u> </u>				<u></u> <u></u>	·
			_		<u> </u>					
	►	<b></b>					<b>T</b> ana			

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, retc.

PA 547 6-90

				Ma	Construct	tion Division	lion		
				1			T		
							1		SHEET , OF 2
PROJECT					NAME OF C	ONTRACTOR		BORING NO	SURFACE ELEV.
Part	Thank	RHG SIT	0		Char	a Drillian		B·J	
LOCATION	2017	100 517	<u> </u>			8 2111111		CONTRACT NO.	DATE ) ,
Ground	Area	HIST of H	14.3	2				426-98-006	11/10/00, 11/16/00
SPOON	DIZE	CASING SU	ZE HOLE	TYPE			GRO	UND WATER LEVEL	
3 .0	.D. 2 3/8	3 m.D. Auger	s 2	1	Date	Time	Depth	Re	marks
HAMMER I	afity	HAMMER							
140 #	FALL 3	6 '	FALL		11-16.	no 9:08 #	<u>'  9.0'</u>	Sample #5	
DRILLER	5	ρ.			1				а. — •
	<u> </u>	Duris			<u> </u>		5.2	Upen hole	
INSPECTOR	r (	Maria	7 7						
CACING	N			arks	<u>l</u>				/6
BLOWS/FT.	DEPTH	BLOWS/6"	COV'D	NO.			INE LOCAT	ES CHANGE OF PROFIL	
Entles Heal		Cuttor Head	Full Rec		FILL BEORN	n Cat SAND te	DGABC .	SUL SUL	0.1
		+	<u> </u>						
				1	Conc	rete Sle	sb w.	ribars	20'
NANDAUCER	•	HANDAUGER		*					
		<u> </u>		9			<u> </u>	1 0 . 6.4.0	
		<u> </u>	╎╌┠─╸	$ \mathcal{N} $	_[i]_gr	eyish-bloc	K_C=¥_	SANY & Grovel, +	r. Silt, Linders, Coal, bric
e for Area				-	•	0			
	5			5	SΔM	ר ב			
		+				· · · · ·			
M M		5-5	L	/ *		75.20	pm		·
AHUEK3		6-6	24	4	SAN	1E. W.	/oil		
		9 19							
		0-1/	11			<u> </u>	$10^{-1}$		
	10 -	13-13	1'8		_5AM	<u>E_w</u> /	wood		
	10	100/311	3" (	1				<u></u>	4
	<u> </u>			hb					
		<u> </u>				<u></u>			//
				i i					
				!				Refusal.	Botton of Boring
				1	AV.L				
	► 15 <	•		4	IVØCA	- 7 SHM	<u> 162 –</u>	saved for ter	11, ng
	<u> </u>					_All_San	oples_	checked with 1	10 meter
						R die	Cardea		
		<u>}</u>				<u></u>		••••••••••••••••••••••••••••••••••••••	
								· ···· ····· ···· ···· ·····	
	<b>^</b>								ć j
	- 20 -			1					
		ļ	ļ						/
									Λ
					- <u> </u>	· · ·			
		<b></b>	<u> </u>		· 				
- · [	2								
		S: 1 - Length re	covered ·	)" 1 000	of Sample	T - Tran used			

**Engineering Department** 

1 1

- 10

۱

1 — Length recovered;  $0^{*}$  — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90



 $P \rightarrow A$ 

**Engineering Department Construction Division** 

**Materials Engineering Section** 

#### **BORING REPORT**

PROJECT Part Ivory P&G LOCATION 10'S of BH.B-2 SPOON 3 '0.D. 2 ³ /8 '1.D. Augers I HAMMER Safety HAMMER Safety HAMER Safety HAMER Safety HAMMER Safety HAMER Safety HAMER Safety HAMMER Safety HAMER SAFETY
Port Lyory       P & G       Craig drilling       B - 2 A         10'S of BH.B-2       CASING SIZE       HOLE TYPE       CONTRACT NO.       DATE         3POON       CASING SIZE       HOLE TYPE       GROUND WATER LEVEL       DATE         3 'O.D. 2 3/8 'LD.       Augers       I       Date       Time       Depth       Remarks         HAMMER Safety       HAMMER       III-16-00       III-16-00       III-16-00       III-16-00         JO # FALL       30 '       #FALL       III-16-00       III-16-00       IIII-16-00       IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
LOCATION 10'S of BH.B-2 SPOON CASING SIZE HOLE TYPE 3 '.O.D. 2 3/8 'L.D. AUGERS J HAMMER Sofety 140 "FALL 30 ' "FALL ' Date Time Depth Remarks 140 "FALL 30 ' "FALL ' Date Time Depth Remarks 11-16-00 10:30 5,1' Sample #3 DRILLER C.MC. Angny INSPECTOR CASING BLOWS/FT. DEPTH BLOWS/6" COV'D NO. Catba Nead Full NO. Catba Nead Full NO. Catba Nead Full NO. Catba Nead Full NO. Catba Nead Full NO. Catba Nead Full NO. Catba Nead Full Catba Nead Full Catba Nead Full NO. Catba No. Catba Nead Full NO. C
10' S of BH.B-2       426-99-006       11-16-00         SPOON         CASING SIZE HOLE TYPE         GROUND WATER LEVEL         3 "O.D. 2 3/8 "I.D. Augers I         HALL TYPE         HAMMER         INSPECTOR         J. Zarkg       "SAMP.2         CASING         BLOWS/6" COV'D NO.         Torkg         CASING BLOWS/6" COV'D NO.         SAMPLE DESCRIPTION AND REMARKS         BLOWS/6" COV'D NO.         LINE LOCATES CHANGE OF PROFILE         Cuttor Head         MANDE RUGER         HANDE RUGER         Auger Solo         SEE Boring report
ID IT - CASING SIZE       HOLE TYPE       GROUND WATER LEVEL         3 *0.0. 2 3/8 *1.0.       Augers       1       Date       Time       Depth       Remarks         HAMMER         HAMMER         JAG *1.0.       Augers       1         HAMMER         GROUND WATER LEVEL         Date         Time         Depth         BLOWS/FT.         DEPTH         BLOWS/6"         COVID         NO.         LINE LOCATES CHANGE OF PROFILE         CASING         BLOWS/6"         COVID         NO.         For
3       O.D.       2 3/8 "I.D.       Augers       I         HAMMER Safety       HAMMER       HAMMER       Depth       Remarks         140       #FALL       30       #FALL       III-16-00       10:30       5,1'       Sample # 3         DRILLER       G. Mc. Aneny       III-16-00       10:30       5,1'       Sample # 3         INSPECTOR       J. Zarks       III-16-00       10:30       5,1'       Sample # 3         BLOWS/6"       SPOON       RE-'       SAMP.'       III-16-00       III-16-00       III-16-00         BLOWS/6"       GOV'D       NO.       III-16-00       III-16-00       III-16-00       IIII-16-00       IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
HAMMER Safety HAMMER 140 # FALL 30 # FALL "HAMMER 140 # FALL 30 # FALL "HAMMER 11-16-00 10:30 5.1 Sample # 3 III-16-00 10:30 5.1 Sample # 3 IIII-16-00 10:30 5.1 Sample # 3 IIIII-16-00 10:30 5.1 Sample # 3 IIIII-16-00 10:30 5.1 Sample # 3 IIIII-16-00 10:30 5.1 Sample # 3 IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
Idommen     Salety     Instruction       140 # FALL     30 " # FALL     "III-16-00 10:30 5,1' Sample # 3       DRILLER     G. Mc. Aneny     III-16-00 10:30 5,1' Sample # 3       DRILLER     G. Mc. Aneny     III-16-00 10:30 5,1' Sample # 3       INSPECTOR     J. Zarkg     SPOON       BLOWS/FT.     DEPTH     BLOWS/6" COV'D       BLOWS/FT.     DEPTH     BLOWS/6" COV'D       Cutto. Head     Full     III       Mandauger     Haub Auger     I       Mandauger     Haub Auger     I       2     See Boring report     B.H.2
140       # FALL       30       # FALL       111-16-00       10:52       5.1       Sample # 5         DRILLER       G. Mc. Aneny       III-16-00       10:52       5.1       Sample # 5         INSPECTOR       J. Zarkg       III-16-00       10:52       5.1       Sample # 5         CASING       J. Zarkg       SPOON       RE-1       SAMP.2       3SAMPLE DESCRIPTION AND REMARKS         BLOWS/FT.       DEPTH       BLOWS/6"       COV'D       NO.
DHILLEH G. Mc. Aneny INSPECTOR J. Zarks CASING BLOWS/6" COV'D NO. Cution Head HAND RUGER HAND RUGER HAND RUGER HAND RUGER Cution Head HAND RUGER HAND RUGER HA
U. IIC. Aneny INSPECTOR J. Zarks CASING BLOWS/FT. DEPTH BLOWS/6" COV'D NO. Cuttor Head Full HAND AUGER HAND AUGER HAND AUGER 2
INSPECTOR       J. Zarks         CASING BLOWS/FT.       SPOON BLOWS/6"       RE-1 COV'D       SAMPLE DESCRIPTION AND REMARKS LINE LOCATES CHANGE OF PROFILE         Cutto: Head       Full
CASING BLOWS/FT. DEPTH BLOWS/6" COV'D NO. Cutto: Head Full HANDAUGER For Soil Strate 0.0' - 8:0' 2 SER Boring repart B.H2
CASING BLOWS/FT. DEPTH BLOWS/6" COV'D NO
BLOWS/FT.     DEPTH     BLOWS/6"     COV'D     NO.       Cuttor Head     Full     I       HANDAUGER     I       HANDAUGER     For     Soil       See     Boring     repart       B.H2
Cuttor Head Full Handlau GER
Handlauger Haub Auger I Handlauger For Soil Strata o.o' = 8.o' 2 See Boring report BH-2
Handlaucer HAND AUGER For Soil Strata o.o' = 80' 2 See Boring report BH-2
HANDANGER
2
See_Boring_report_BH-2
Auger -
4-5 * 66.5000
C N EN EN CANN C LIGITO LO
10 7-8 10 - 7-8 10 - 1115c till black c-1 SAND some bravel tr SIII, unders Loa
100/34
When a country 45 parent O. C. al Bill
Noce i sample no savea Ketusal = porton
for testing. All other samples Dor
<u>Screened for Il Preadings Kaliscarded</u>
┝━━━━╋╸╴╺╋╍━╍╼╋╍━╍┪╴╴┝╍╴━╸━╸━╸━╸━╸╾╸╍╸╍╸╍╸╍╸╍╸╍╸
┝╾╍╍╌╋╼╶╶┊╍╋╼╍╍╌┉╋╍╧╬╣┊┊┊╞╾╸╤╧╴╤╼╴╾╸╾╸╾╸╼╸╺╍╴╍╸╍╸╺╍╴╍╸╸╸
┝────┼── ──┼─────┼────┤
NOTES: 1 - Length recovered: 0" - Loss of Sample. T - Trap used

2 - U = undisturbed; A = auger; OER = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc. . .

.

6-90	Ei Mate	ngineering Department Construction Division erials Engineering Section ORING REPORT		
	. —			SHEET OF
PROJECT PL	) N	AME OF CONTRACTOR	BORING NO. RZ	SURFACE ELEV.
	J	craig arilling	CONTRACT NO.	DATE
40'5 of Bldg 16 \$ 200 Wol	Bldo 33A Bloc	ck 1400 lot 1	426-99-006	12-4-00
SPOON CASING SIZ	E HOLE TYPE		GROUND WATER LEVEL	
3 "O.D. 2 78 "I.D. Auge	rs	Date Time	Depth R/	emarks
140 # FALL 30 "	FALL 7	12-4-00 9:30	3.5   Sample # 9	
DRILLER				
INSPECTOR	·			
CASING J. ZOrks	RE- 1 SAMP.2	3SAMPL	E DESCRIPTION AND REMAR	KS
BLOWS/FT. DEPTH BLOWS/6"	COV'D NO.	LINE L	OCATES CHANGE OF PROFIL	-E 0.0
handauger handauger	full Kec	CONC		<u></u>
		Fill dark aren	c- PSAND tr. Grav	Let tr Sill cinders con
	×		1 7	<b>)</b>
		SAME		
<b>└───</b> ► 5 <b>◀</b> ──┼				
	<u> </u>	SAME		
2-3	/*			
AUGERS 3-3	20" 4 F	Ell_c-C SAND +	r Grave tr. Sill	cinders coal
12-18				
92 - 24	9419 5	SAME	······································	
	24 6			÷
12-15	20 7 9	<u>SAME</u>		
15-3-5			* · · · · · · · · · · · · · · · · · · ·	
13 6-7	20" 8	SAME		
		DAME		
		Brown Peat	ο η L.	10.0 A
<u>├</u>	N	lote: 2 Samples saved	for testing	
20		_ All samples chec	ked w. PhD meter	/
			discarded	
		J		Boltom of Boring
			· · · · · · · · · · · · · · · · · · ·	+ <b></b> ]
	—			
			·	:

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc. 

.

C

•

۰.

#### THE PORT AUTHORITY OF NYS NU

1

۰.

PA 547

Engineering Department Construction Division **Materials Engineering Section** 

#### **BORING REPORT**

							<u> </u>				SHEET OF S
PROJECT	<u>ר</u>	1		0 /			NAME OF CONTI	RACTOR	E	BORING NO. TRI	SURFACE ELEV.
	DRT	VOA		146	SITE		CRAIG	PRIMA	6	64	*
LOCATION									C	CONTRACT NO.	DATE
WEST	<u> </u>	TUR	ONG	¥ 53	3HILK	1408	Black	- 1400		426-99-006	1014/00
SPOON	้า	3/		CASING SE	ZE HOLE	TYPE			GROU	IND WATER LEVEL	
<u> </u>	D.D. 🖌	18	"I.D.	HUGER			Date	Time	Depth	l	Remarks
HAMMER 140	FALL	infery 30	, -	HAMMER	FALL		12/4/00	2:00	41.5	PURING HAND	ALCERING
DRILLER	5m	Fin	VU								
INSPECTOR	<u></u>	11	in	<del></del>							
	Va	$\mathcal{N}$	N	ANIS							
	DE	этн	SI	POON	RE- 1	SAMP.2		3SA	MPLE DESC	RIPTION AND REMAI	RKS ILE
A				A. 0	0010		Arabelt and	10015			
How Albert	L		HANNY	PRICER		1 A		JONE			
T I			İ	1	ļ	T B	Pat CINTIGRE	anD LAM	VEL		115
	1-						- Datan E	un conc	· · ·		
<u>├</u> ─- <u></u>	+-					2*		750-	-460 001		
							<u> </u>				
	- 6					ZA				0.000	3.0
1 de la		· _		V		- 6	RINGRAY FINE	SAND JA	ALE SILT	- VETTUTUT DUOK.	10-100 6.0
tel			5	6	15"	1 At	Fill REDOKH BR	own) clate	ey shar		73
			6	5	· ·	18	hy Rould		mes		
	<u> </u>		2	2	11	Y	INT PROWN P	THE ARTY	10000 3	<u> </u>	
<b>├</b> ── <b>┤</b> ─	_			2	U	$\zeta$	Sme				
		0 🚽	5	8							
			4	4	10"	1.	BROWN PE	1			10.9
			7	11		6	BROWN F	INE SAN	D. TRA	TESIL	12.8
							1				
								Butim	UF	DORING	
	<u> </u>	$/\neg$					<u>├──</u> ── ──	* THE IT	_*		
	► [!	5 ◀				,	┝── ──				
	<del> </del>	-						. CAnnl	nes cr	RAPHED WITH	PID METER
	<del> </del> -							C _ ALLY	# 7.2	3 And the Ka	KD FOR
	<u> </u>							his alies	<u>- <u>-</u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>	Kerny	
	<u>+</u>						⊢ <i>⊌</i>	MKONM		ICONNO	
ļ	► 2	.0 🚽					<u> </u>	1-1451	MAINING	SAMRES T	DISCHARDEY
	L						·				
	<u> </u>	-				1					
	┣						└── ── <del>──</del>				
	<b> </b>	·									
	L						ł				
·····		NOTE	0. 4					-			

1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

'A 547 6-90
THE PO	RT AL	THOR	<b>ITY</b> OF	nnent
--------	-------	------	---------------	-------

Engineering Department Construction Division Materials Engineering Section

#### **BORING REPORT**

								SHEET OF S
PROJECT	NT.	$\overline{\mathcal{D}}$	D+1		NAME OF CONTRACTOR	e BC	DRING NO.	SURFACE ELEV.
	201	_vary_			Crait Ound	8-1-		DATE
LUCATION	N	RW I-A	Bla	xK 1L	601		426-99-006	12/2 AND 12/4/00
SPOON	- 2/	CASING S	ZE HOLE	ТҮРЕ		GROU	ND WATER LEVEL	
3.0	<u>р.р. 278</u>	"I.D. QUEON		1	Date Time	Depth	R	emarks
HAMMER	FALL 3	PT7 HAMMER	# FALL		12/4/00 7:45 6	8,0	DURING 3PU	t= Spooning
	A7							
INSPECTOR	houl	ie Spring	201					
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- '	SAMP.	³ SAMPL Line		RIPTION AND REMAR	KS _E
11A		NA	Fal	,				
			<u> `•</u> {−		Grand ul Sand S			
	·							
			┼╌┼─╴	2	Gund at charlons -	Gate	SET MOTOX B	Woth
<b>├</b> ─- <b>├</b> ─-	· ·		$\left  - \right  $		- diotomshors - Eau	-11-	white	
	5			3	CINEOUS + GUNEL	1360		
C.L		78	15"		SAME	- <u></u>		
	·	12-9	<u> </u>	4				
	· ·	97	131	C		·····		
		33	-13	5	<u></u>			
	• 10	II.	20"		Lines with 111 10		DIATION HOUR B	ADTL TOD DE
<u>}</u>		53		6	Con 1		MATOM MORO US EN	
		31	211		JIWN			
		11	2	7				
		<u> </u>	1010			······································		
	15	- 20	14"	8	BROWN TEAT		·	
		LL_	*					(LeC)
			ļ		L			
					Burrom 0	<u>F</u> I	BORING	
			<u> </u>					
	20		ļ		- ALL SAMPLES	_Sre	ENED WITH P	17 METER
			L		- SAMPLE #2	SAVED	FOR ENTIRON	NMENTAL TESTING
					- ALL REMAININ	<u>NG S</u>	AMPLES TASCH	HRDED
	- •							
L		<b>4</b>	I	l	L	. <u> </u>		

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

ł,

**Engineering Department** Construction Division Materials Engineering Section

#### **BORING REPORT**

					SHEET   OF 3
PROJECT	A TYAN	n 9+ (	i sit	e	NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
10					Craig Dulling A-2
LOCATION	SW of	Bailding	16	Copr	me Gofeet) O CONTRACT NO. DATE U12912000
SPOON	21	CASING SI	ZE HOLE	TYPE	GROUND WATER LEVEL
3 .	<u>אר א. ס.כ.</u>	"I.D.		]	Date Time Depth Remarks
HAMMER	لمی FALL 3	(-hy   HAMMER 10 "   140 #	FALL	-	11/29/50 0815 4.0'
DRILLER	David	cooks			
INSPECTOR	m. Po	tel			
	OEDTH	SPOON BLOWS/67	RE- 1	SAMP.2	³ SAMPLE DESCRIPTION AND REMARKS
Hama		Hand	Full	140.	
A.4-	<u> </u>	Aye	<u> </u>	1	Fill Black and Brownish Yellow CML SAND
					trace Silt trace mf Gravel, Ruck, cinder, etc 2.0'
	L			2	white Dictomacaceous earth material
					Shites Gray Same
	- 2 -			3	
W	<u>├-</u>		N.		
Jer Jer	┝───	ωο	24	4	White Diatamaca cus earth matinal
·		Н			
		10	24"		SAME
		10		5	
			20'		
		1.0		6	
╞	┝	<b>},}</b>			
	L	1,1	16		SAME
	,	ર ૨			
		9,15	18		SAME ISO
		20,15		8	Black cinf GRAVEL, base conf SAND Roak, cincles
		5.6	10"	~	SAME
		10		4	
		1, 1	18"		SAME
		1.1		10	Amuan Organic STIT is decembred where CPLAT
¥	► do ◄				Bothum of Hole at 20.0
	<u> </u>				
	┝─ ──				All soil samples checked for PID meters
	L				Sample at 1 2 2 save of for Environmentel testing
	L				Remaining samples are alicandeal.
					·
			· · · · · · · · · · · · · · · · · · ·		

-

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PA 547 6-90

.

									``			
547				TH	<b>IE PO</b>	11	<b>AUTHO</b>	RITY OF	ny&i	Ϋ́Ρ		
90						Er	gineerina D	Department	. –			
						-0	Construction	Division				
					i Ma	ate	rials Engine	ering Secti	on	419 A		
					•	B	ORING F	<b><b><i>EPORT</i></b></b>	-			
•						_				· · · · · · · · · · · · · · · · · · ·	SHEET   O	F 3
ROJECT						I N	AME OF CONT	RACTOR	ŕ	BORING NO.	SURFACE ELEV.	
P.J.T.		ΡĮ	G			1	Crain	illing	1	A - 3		
CATION	VOT Y	<u>-'}</u> .				-L	ciald a	riting		CONTRACT NO.	DATE	
Å	c hid	~. i ł	in 4	ha lia	10 -	•	•	. •	1	10/ 09 00/0	11-16-0	00
N NOON	2 14101	Jul	CASING SI		TYPE	·	<u>,                                     </u>			11ND WATER LEVEL	<u></u>	
7 10	vn 93/	9 m	A	.	1		Date	Time	Depth	R	marks	
MMER Q	aletu	0_1.0.	HAMMER	<u> </u>	<u> </u>							
140 -	FALL	ا م		E FALL	-		11. 16	1:05 PM	20	SAND # 2		
	0			- FALL			11-10-00	~ 0 3	<u> </u>	- Jenne H		
	G.M.	Anen	4.1									
SPECTOR			<del>,</del>									
	77	ARK	Ś									
ASING			POON	RE- 1	SAMP 2	5	J	3541		CRIPTION AND REMARK	(S	<i>v</i>
OWS/FT.	DEPTH	BLO	DWS/6"	COV'D	NO.			ĽI	NE LOCAT	ES CHANGE OF PROFIL	E ^{ite} t	0
ndaneu		Cutte	r Head	Full Re	· ·	Т		(	concret	e		C
+	┝			<u>├</u>		F						
	L .						il Arei	<u>c - (</u> SA	HND an	d Gravel tr. SI	Ш	
					d ²	Γ	1.1	15 1				
+	┝	+		k	9*	F	<u>_</u>	<u>.</u>				
							ill greui	sh - area	en Sil	t. tr. m- C SAI	V D	4
						Γ	3-3-	0		, ,		
	▶ 5 -	◀		┝━╂╌╌		1-	70		<u> </u>			
_ <b>y</b> (	Ļ		¥	V	5	L	ill_gre	tish b	lack_	DIT & CLAY		
W.		1		1 -		Γ	0-1			· · ·		
JERS	<u>⊢</u> –	<i>0</i>	- 0		/ / *	1	ना न					
		<u> </u>	_ 10	24"	4	$\downarrow$	rill blo	ack -	SILL	X UAY		8.
		1 2	2				N. and		• <b>V</b>	* 3# 	1	
-1	<b>⊢</b> −	1-7		0/	ト	17			atta		<u></u>	
_ <b>_</b>	10 -	4 9	-15	14		Ļ	ill_ <u>Grey</u>	1 C-1-2	ZANU	, greenish - White	J'etomoceou	<u>~ -</u>
	-	4	- 5		/ ×		0 1	1		r e∎ K		
		1 1		. 11	6	1		F grync	TINT-			
		16	-6	24"	ļ	+1	III While	e c-fi	ZHNY,	bravel, Viatous	er eacis	· 19
	•	2	- 2				x	1	•	,		_
	<u> </u>	5	/	9 _ 11	7	5			$\Pi$		-1 $-1$ $-1$ $-1$	
		1-2	-4	10"	-/	₽		ACK 3	비뷰 기	ruy , some _	chunks	
	15	- 1-	- 2			1				/ ·		
	- 13 -		2	o /11	K	[7	FIT -					
						╞	<u>-1117 —</u>	PLACK	<u> </u>			<del></del>
		2-	2		0	L						
	· · · ·		2	01.4		$\left[ \right]$	$\overline{\Box}$	ME				
╉──┼		+		1.24	<u> </u>	┢╸						
		-	- 2			L						<u>    19:</u>
~ <b> </b>	<b>X</b>	1	2	24"	O		B	Ω	FAT			
_ <u></u>	► 20 -				-	ŧ	OWI					
		1				L						_ 2
.T.			:				lote: 3 9	samples	saved	for Testing		7
		╂────				F	Ait	other sa	moles -	screened with -		-/-
						L	?iD	meter	2 disco	rdeo	_ Bottom _c	L'Ba
		1				Γ						1
	<u> </u>			I								<u> </u>
						1						
				······		•						

{ }

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

Aller

PA 547 6-90

- 4

#### THE PORT AUTHORITY OF NYG NJ

Engineering Department Construction Division Materials Engineering Section

#### **BORING REPORT**

							·	SHEET OF 3
PROJECT				NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
Port Tuari	Péli			Craig c	Irillina		A - 4	1 . 1
LOCATION				<u> </u>			CONTRACT NO.	DATE
Ac Toic	1 aut in H	an (lin				1	696-99-006	11-16-00
SPOON	CASING SI	TE HOLE	TYPE				UND WATER LEVEL	
7				Data	Time	Death		marke
D -0.0. <u>178</u>	TID. AULOLIS	2 1				Осран	······································	
namen Sheey	, promition		-		/AM	20	CAMPLE #	9
140 # FALL 30	)	FALL		11-10-00	11.24	<u> </u>		<u>~</u>
C. M. A.	•••					l		
	eny							
T7.								
I. Lari	<u> <u> </u></u>			<u> </u>				
	SPOON		SAMP.		"SA	MPLE DES	CRIPTION AND REMARI	KS And
Cuthelled O	BLOWSIO	CUVD	<b>NO.</b>				LO ORANGE OF THOME	- 0.0
CUMILL THERE	Cutter Head-	PUI NC					RETE	
			1	N. T.O		D CAL		
HANDAUGER	HANDDIGEP	┝╌┤╌╌╴┑	<b>-</b>	Fire till de	eyish-plack	لالتهت	Ny & Gravel, ir Silt, Vool	Linders , Drick , Wood ele
				ľ. ľ	-			
			2.	CAME				
┝━━╇╍━╋╍╸╺━╸	<b>├───</b> ┤───────	┟╴╎╶╼╼	ļ	SULE				
			1	ł.				
			コ	CAME				
	· · · · · · · · · · · · · · · · · · ·	<b>└──</b>		LOAME_				
W S	4-4	••	*					
STEM		0.0		CAMP	· · · · · · · · · · · · · · · · · · ·			
Augors -	3-3	18.	4		<u> </u>			
, ingeno	4.3		]				. ·	
			5	GAME				
	3-3	14"	<u> </u>	DALIC				
	4-2			1			-	·
		0.1	16	CIME	• :			
┝━━╋━━┥╼╴ ╺━	5-4	20"	10	1- SALIE	·	- <u> </u>		
	4-6		1				· · ·	
		0 P	<b>-</b> .	CANE				
┝━━╉━━┼╸╴╺━	6-4	20		1 June				
	2 - 2			SAME			·	
	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	1.01	R		2-14	==		/
└ ─ ┸──┤────	2-2	18	\vdash	Prown_	L'EA	<u> </u>		16,
			}					. 1
├─── ├ ── ──			1	N 4.		· · · ·		
<u>├</u>				Noce: Samp	<u>us # 4 _ </u>	⊻_ #7	Savedfor_test	ing/_
			•	L All Å	her comp		eened with PID ~	eter /
·	1.		1			=		
┝───▶. ◀			{	<u> </u>	<u>iscarde</u>	a		/
		(1			Bot	tom of Boring
			1					
∮ →	·	· · ·						
	1			└── ── ──				
				<u></u>	· · · · · · · · · · · · · · · · · · ·			
	C. 1. Landb		L	e of Compton T				

1 — Length recovered; 0" — Loss of Sample, 1 — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

<u>547</u>			Ţ	ie po	RT AUTHORITY OF NY & NU
90				1997. V	Engineering Department
				L.	Construction Division
					SHEET 1 OF 2
OJECT	_ <u></u>				NAME OF CONTRACTOR BORING NO. SURFACE ELEV.
F	ort Tu	oru P	& G		Crain drilling A-5
CATION		J.			CONTRACT NO. DATE
<u> </u>	ls laid	out/in t	he fi	eld.	426-99-006 11-14-00
OON	- 3/.	CASING SI	ZE HOLE	TYPE	GROUND WATER LEVEL
<u>5 "(</u>	D.D. 278	"I.D. Auger	r <u>\$</u>	L	Date Time Depth Remarks
		C .	4 2411	· .	1115 at 7 1/6 8.0' Samula #5
	TALL 3	<u> </u>	FALL		11-13-00 146 010 Sumple H 0
	J. Usuc	h			
PECTOR	7 -	7 }			
·	<u>J. 4</u>	-arks	<u> </u>		
ASING	DEPTH	SPOON BLOWS/6"	RE- 1	SAMP.	² 3SAMPLE DESCRIPTION AND REMARKS
dauger	►"°°"<	Handauger	Full		Fill DGABC
+			┦╌┡╌─		Fill dark brown c - [SAN], Gravel, tr Sill, cinders
1	L _				<u>SAME</u>
				×	
1	+	<u> </u>		9	THE LILL CONNECTION
╉────	┼── ──	<u>}</u> −−−−	┟╌┠╼──	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	fill greyish - black c-f SHIV, browel, tr. SIII, cinders, coal-
ļ	5				
				3	SAME
W	t	<u> </u>			
A		3-0	- 11	,	
		12-17	12"	-4	Ilisc Till areyish black c- SAND, triavel, tr. DIT, cinders, wood,
	L _	7-8			
		7-9	20	5	SAME
		10			
	<u> </u>	12-6	11	1	
	└── [`] ──	6-7	24	N .	<u>SAME</u>
	L	10-11			
1		11 - 14	12."	7	SAME
-	<u> </u>				SAME 14
<u> </u>	▶ 15 ◄	4-5		0	
1	L _	3-4	24	0	Brown PEAt little Silty CIAY
		,			
					IN he as he flow and the harding
	┝ ─				liete: Sample # 20 Saved for resting
	┝		 		All samples were screened with PLD meters
	20-				and discarded Doccom of Dor
					х х
			·		
	<u> </u>	<u> </u>		·	
	1		i 1	1	
	L				
	· · ·				
	· · · · · · · · · · · · · · · · · · ·				

· · · ·

.

.

•

PA 547 6-90

THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

									SHEET / OF 3
PROJECT					NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
Port	Ivory	VJG SITE			Crais	Villing		H-6	· · · · · · · · · · · · · · · · · · ·
LOCATION		, , , , , , , , , , , , , , , , , , ,	<u>.</u>			0		CONTRACT NO.	DATE
Eag 7	of Kld	<u>c 1+</u>	Block	1400	LOT/			<u> 426 - 99 - 006</u>	11/10/00 - 11/11
SPOON			ZE HOLE	TYPE			GRO	UND WATER LEVEL	
3.0	.D. 2.3/8	3 "I.D.		1	Date	Time	Depth		Remarks
HAMMER		HAMMER) (lun)en	115	10	1.11.11	(N. a)
140 #	FALL 30		# FALL		11110100	ΓP	610	While Mi	and Hugering
DHILLEH	S	Russe			ulul.		1. 5'	Samle #	6
INSPECTOR					~ 11/00	1:05	(v. j	Sample #	7
	Ľ	Marco 1	T 7	Le					
CASING		SPOON	RE- 1	SAMP.2	1	3SA	MPLE DES	CRIPTION AND REM	ARKS
BLOWS/FT.	DEPTH	BLOWS/6"	COV.D	NO.		LI	NE LOCAT	ES CHANGE OF PRO	OFILE
Hand	0	Mard Augr	Ful	ļ	Crushe	1 STON	a . LT	The Mischie	O.Lorg
n		1	1 1				= -+= -		
TUGA		<u>├</u>	┼──┼──	4	<u> </u>				
		·			Mix Fi	<u>11 (ud</u>	AS Cra	no Sand Brick	<u></u>
					1		•	, ,	
				2				<u> </u>	
	> 7 <		┼──┦──		<u> </u>	<u>e</u>			
		<u> </u>	<u> </u>	3	Sam	<u>re</u>			·
		4-3							
TEN		4-5-							· · ·
Augers		3-0	20	1	↓⊇/	IME_			
- u	_• .	4-5	·	ł					
		6 1	901	5	SA	MF			
		<u> </u>	~~		~ 1				
	~	6-8		1	└				
		6-11	24"	0	<u>SAM</u>	E			
		10 18		ļ					
			. 0"	17	M: T			1. 0.014	
		12-3	10	ļ	MISC F	II Lind	ers, OI	ack c - f JAI	Ul bravel etc 140
		W.O						· · · · · · · · · · · · · · · · · · ·	
		<u>ч і</u>	2.4"	8	Brown	1 PEAT	۲ «	ne aren sit	tu CIAY 160
		! `!	+ <u>/ / / / / / / / / / / / / / / / / / /</u>	<u>├───</u> ──				- Jeg - pi	A
				4	└				
·			<u> </u>	ļ					
								-	Bottom of Borin
				1	<u>├</u> ───				
	20 🗸		 	ł [.]	└ <u></u>		·		
				[AIIS	Sample.	s chooked us	Th OID Motic
]		571	$12 \sqrt{c}$	I for ToT	And
				1	┝── ──		<u> </u>	WOO TOP 1-11	*
			 	1	<u> </u>				
						sample =	# 2	saved (On h	old) for testing
				1			······································		
	> 25	····	<u> </u>	<u> </u>	L				

愛いた

ŝ

NOTES: 1 — Length recovered; 0" — Loss of Sample, T — Trap used 2 — U = undisturbed; A = auger; OER = open end rod; V = vane 3 — Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

MATERIALS ENGINEERING DIVISION WELL MONITORING DATA SHEET

PROJECT	: HH - P	RT IVOR	V PEG	SITE	JOB NO:	501-7	27-90	96	
WELL DE	SIGNATION	: TW2	· _ ·		DATE:	12-2-00		2	
CHECK	BOXFOR	LOWFLOW	1: RATE	(HL):	CASING D	IAMETER:	2	Inch	
WEATH	IER CON	DITIONS:	SUNN	28%	51	CL-DPDIST	ANKE:	2.0	
STATIC WAT		<u>zPSCC</u>	ORD'	:10	N(T.:	L	AT.:		
-	·		DISTANC	E FROM T	OP OF PIPE	TO:			
	· ·	TIME	WATER	(FEET.)	PRODUCT	(FEET)			
PRE-PUR	GE:	11:151		77				• •	
POST PU	KGE:	12:201/5	6.	55					
	DEPTH O	FWELL		F	11 03			EFFT 1	
	DEPTH TO	D WATER			6.17				
	DEPTH O	F WATER C	OLUMN		4.28			FEET	
	FACTO	R #		X	0.612				
WELL PURG	VOLUM	LE TO REI	REMOVED		2.65		<u>.</u>		
TIME	DH	TEMP	CONDUC	TIVITY	SALINITY	TURBI		DISS. 02	
	(SU)	(C)	(umohs	tan) uS	(0/00)			(mg/l)	
1:21AM	7.18	16.20	29	50	1.5	TE Z			
11:30A	7.28	16.2.	310	57	1.2	1,006			
1:40	7.10	17.1		38	1.7	73	5 *		
		· · ·							
				· · ·	·				
	_								
					-{	· · · · · · · · · · · · · · · · · · ·	4		
		1			1				
		1						1	
			 						
									
			 						
L	L	<u></u>	<u> </u>						
SAMPLE	<u>D BY:</u>								
COMMEN	ITS:	Ċ	5 heen	Visib	e in	PUR	ed	Volumes	
				· · · · · · · · · · · · · · · · · · ·	······				
-		*	METALS	SAM	PLED 4	,iTH	FILTE	۹	
			Well s	ampled	at 12:0	орм			
				(

* FACTOR = 0.618 FOR LINCH DIAMETER WELL CASENG ¥, ٨

THE PORT AUTHORITY OF N.Y. & N.J. MATERIALS ENGINEERING DIVISION

WELL MONITORING DATA SHEET

PROJECT:	PROJECT: HH - PORT IVORY P&G SITE JOB NO: SO1-233-295										
WELL DES	GNATION	MV	PZ-1		DATE:	11-24-00					
CHECKE	SOXFOR	LOWFLOW	A KATE	(HL):	CASING D	IAMETER: 2	Inch				
WEATH	RCON	DHIONS:	SUM1)	<u>34°F</u>	51	CA-UPDISTANCE:	2.4				
STATIC WATE		PSCC	ORD'	<u>s</u> :Lor	S(T:	LAT .:					
-			DISTANC	E FROM TO	DP OF PIPE	TO:					
		TIME	WATER	(FEET)	PRODUCT	(FEET)					
PRE-PURGE: 210PM 4.63											
POST PUR											
	DEDTH OF	- \ A / - 1 A	<u> </u>	·		14					
	DEPTH UP	WELL		}		15					
	DEPTH OF	WATER (}	<u> </u>	5	FEET				
	EA(1)	D .#		X	5) (I	<u> </u>					
WELL PURGE	I I W	D T	7544-1600	<u> </u>	58	-7,					
	VOLUP	12 10 ISCI	KITTO VED	1		_/ !					
TIME	рH	TEMP	CONDUC	TIVITY	SALINITY	TURBIDITY	DISS. O2				
1.105/20	(SU)	(C)	(umohs		(0/00)		(mg/l)				
h:15Pph	1.52	13.60	<u>1 d4</u>	<u>5</u> 9	0.6	<u> </u>					
2 35Pm	7.02	15 60	120	<u>/</u>	01	110 #					
<u>A.5201</u>	1.00-	12.0	. / 00	<u> </u>	<u> </u>	<u>//U_T</u>					
	 			·		.6					
	 										
	{				<u> </u>	······································					
		<u> </u>									
		1			11						
	l										
	ļ	·	L			····					
ļ		<u> </u>									
				•							
L		1	<u> </u>				l				
SAMPLED	<u>BY:</u>	/	6. /£.	<u>M.</u>							
COMMENT	<u>rs:</u>	W	e/1 .	Sampled	at	2:45PM					
		<u> </u>	METALS	FiL	<i>lerul</i>						
	``					·····					
			······								
* FACTO	pR = (7 618	FOR	TNUM	DIAMET	FRUCHEA	ST.V-				

* (

MATERIALS ENGINEERING DIVISION

WELL MONITORING DATA SHEET

	PROJECT:	HH - Po	RTIVOR	r P\$G	SITE	JOB NO:	501-2	33-295		
	WELL DES	IGNATION:	PAMW	~15	7	DATE:	11/2	0/00		
	L TRACIN	SOX FOR	Lowflow	H IKAIE	(11,):	ICASING D	AMELEK:	2		
	STATIC WATE	RLEVEL	DCC		= = 1 =	101	CAULINIA	NKE.	5.20	
	-		TFS CO	OKDS	-LO	VIT:	L/	47.:		
				DISTANCI	E FROM T	OP OF PIPE	TO:			, , , , , , , , , , , , , , , , , , ,
		r .	TIME	WATER	(FEET)	PRODUCT	(FEET)			
	PRE-PUKG		10.07	9	-39					•
	1001101			•••••••• <u>•</u> ••				_		
	:	DEPTH OF	WELL	· · · · · · · · · · · · · · · · · · ·	[16 *	00		FEET	· ·
		DEPTH TO	WATER			9-1	T-9		FEET	
		DEPTH OF	WATER C	OLUMN		6.4	41		FEET	
		FACTO	<u>R * .</u>			0.61	18			
	WELL PURGE	VOLUM	ETOREF	REMOVED		.3.9	6			
	TIME	DH	TEMP	CONDUC	TIVITY	ISALINITY	TURBIDI	TY	DISS 02	
		(SU)	(C)	(umohs	/cm)	(0/00)			(mg/l)	·
	10:44	6-48	16.0	37	9	0.2	05			
-	11:00	6-48	15.3	36	4	0.2	05			
	11:10	6146	16+4		2	02	650		-	
	19/16	6.56	16-0		[]	0.2	220			provillien P.
	12-13	- 6 - 3 /	_/ C /	2 (· · · · ·				· · · · · · · · · · · · · · · · · · ·	US S. M. Saule
						1				•
										,
						-{		·*		
	}			7	• <u></u>		······································			
								·····		
				<u>`</u>						
		ļ								
- 1-	L	L	L						L	
\$	SAMPLED	BY:	A	7 <i>4 É</i>	E. M.,		<u>مەرىخى مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ مەرىپ م</u>	ting and a state of the state o	• • • • • • • • • • • • • • • • • • •	· .
	COMMENT	rs·	Well	1-	14	1 3	1 - V	Printing and	1. 16.1	in part
	S CHINLIN	<u></u>	Tent P	por de	7 4	· coury	al fresh	- J.T.	in 4 Dulle	Crossellos,
			- colle	of can	men S	molini	an new	ed 719	2.16->	Setting & ajed

chil

* FACTOR = 0.618 FOR LINCH DIAMETER WELL CASENG

Used 0,45 micron

)

MATERIALS ENGINEERING DIVISION

THE MELL MONTONING DATA SHEET

PROJECT: HH - PORT I VORY PEG SITE	JOB NO: SO/-233-295
WELL DESIGNATION: FAMW-15D	DATE: 11/20/00
CHECK BOX FOR LOWFLOW: RATE(HL):	CASING DIAMETER: 2" Inch
WEATHER CONDITIONS:	STKA-UP DISTANCE:
STATIC WATER LEVEL GPS COORD'S : LC	NG: LAT.:

		DISTANCE FROM	FOP OF PIPE TO:
	TIME	WATER (FEET)	PRODUCT (FEET)
PRE-PURGE:	10:08	9,99	
POST PURGE:			

DEPTH OF WELL		52.49		FEET
DEPTH TO WATER		9199		FEET
DEPTH OF WATER COLUMN		42-50		FEET
FACTOR #	X	0.618	·	
WELL PURGE VOLUMETOREREMOVED		.26-27		

	TIME	pН	TEMP	CONDUCTIVITY	SALINITY	TURBIDITY	DISS. O2
		(SU)	(C) ·	(umohs/cm)	(0/00)		(mg/l)
	12-23	6:83	15-7	2132	1.1	0.5-	
	13:05	14-76.9	0 14-7	2162	1-1	Or S-	\sim
	13:48	6 87	15.4	2257	1.2	0-5-	
	14:15	6-99	15-6	E264	1-2	0.5.	
	14:20	6-99	15-7	2284	1-2	<u> </u>	
Ì	14:35	7.00	155	22 40	1.2	0+ 5 -	\sim
					<u> </u>		
						· 5	
						· · · · · · · · · · · · · · · · · · ·	
		<u> </u>					
1							
			1		1		
		1	1				
	·	1	1		1		

SAMPLED BY:

AZ dEM

OMMENTS:

Very nuddy Filtered Metals

* FACTOR = 0.618 FOR LINCH DIAMETER WELL CASENG

MATERIALS ENGINEERING DIVISION WELL MONITORING DATA SHEET

PROJECT:	HH - Po	RTIVOR	Y PEG	SITE	JOB NO:	501-233-2	295		
WELL DESI	WELL DESIGNATION: PAMW-15D (Regent DATE: 11/30/00								
CHECKH	CAPUCA BOX FOR LOWFLOW, IKATE(AL); CASING DIAMETER: 2 Inch								
STATIC WATE		ITIONS:			151	CA-UP DISTANCE:			
<u></u>	16	PSCC	DOKD.2	LON	J(T:	LAT.:			
	·		DISTANCE	FROM TO	OP OF PIPE	TO:			
		TIME	WATER	(FEET)	PRODUCT	(FEET)			
PRE-PURG	E:	8:16	1	3.23			•		
POSTPUR	PUST PURGE: 9:56 13-55								
1					691		ECET 1		
DEPTH TO WATER 13-22 FFFT									
	DEPTH OF	WATER C	OLUMN		39	26	FEET		
	FACTO	R# .		X	0.61	8	· · · ·		
WELL PURGE	Valum	KTORE	CAM 1/4		.94.	96			
		101001			<u>2</u> 7				
IME	PH (SUN	IEMP				TURBIDITY	DISS. OZ		
Quit	7-5-3	14.9		74	(0/00)	67. C.	(119/1)		
4:18	7,43	16.4	23	37	1.2	c.R.			
9:33	7.40	18.0	23	32	1.2	O.R.			
			ļ						
		·		· · · · ·					
·			}	·····		······································			
	·		[
									
\									
			<u> </u>				·································		
	{					 			
		<u> </u>				O.R-Over	range		
·	1	<u> </u>	1						
Company and the second s	Lawrence		1	مالا المستجلية المراجعة من المراجعة المراجعة عنها.	1		diadi		

SAMPLED BY:

AZYEM

COMMENTS:

Very anddy - ; low	recevery, W	ell your dry
between volumes.)	
Filtered metal.		

* FACTOR = 0.618 FOR LINCH DIAMETER VELL CASENG

MATERIALS ENGINEERING DIVISION

DPO JECT.	Hil-P.	OF T	017	(LIOP NO.		1 0.00	
WELL DESIGNATION: PAMW-10D DATE: 1/20								
CHECKE	LON LOR	1050	- IRAT	(41):	CASING DI	AMETER	80/00	toch
LIFATU	FR CONT	THOME .			ा दगा	IPDICTA	VC.	
STATIC WATE	STATIC WATER LEVEL COCCORD'S CLOWNS							
-			URU	<u> </u>	<u>V(7</u>		1	
		·	DISTANC	E FROM T	OP OF PIPE	TO:		
		TIME	WATER	(FEET.)	PRODUCT	(FEET)		
PRE-PURG	<u>E:</u>	14:36		7-18				
POSTPOR	GE:	13:30		-68]		
1	DEDTU OF			r	37	- 62		
	DEPTH TO	WATER				-18	ר 	
	DEPTH OF	WATER C	OLUMN		91.	1412	F	EET
	LACIO	Ω .#		X	<u>~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~</u>	6.18		
WELL PURGE	1/ 1 or		2544-16	<u> </u>		.79		
	VOLUM	12 TO ISE	KITOVED	l	. 16	> ८		
TIME	pН	TEMP	CONDUC	TIVITY	SALINITY	TURBIDIT		ISS. 02
	(SU)	(C)	(umohs	/cm)	(0/00)			(mg/l)
14:46	6-16	14-8	27	42	1.4	32		
14-16	6-14	15.6	26	<u>86</u>	1-4	40		
15201	6-10	8		2-1	1-1-4			
				<u> </u>	-{			
				·····				
					4+			
			·					
							.9	
	 							
ļ	l	 	 					
	·}	<u> </u>	 					
			 					
	1	<u>.</u>	L		<u></u> ł			والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع وال
SAMPLED	BY:	_ A 2	4 F.,	4		ې د د د د د د د د د د د د د د د د د د د		na ang kanalang kana
COMMENT	<u>[S:</u>	60	od re	rore	~ej			

* FACTOR = 0.618 FOR LENCH DIAMETER WELL CASENG

MATERIALS ENGINEERING DIVISION

WELL MONITORING DATA SHEET

PROJECT:	HH - Pe	RTIVOR	Y PEG	SITE	JOB NO:	501-233-2	95		
WELL DES	IGNATION:	MANU	J 10 4-T	<u></u>	DATE:	11/271	03		
CHECKE	CHECK BOX FOR LOWFLOWS, INATE(AL). TOASING DAMETER:								
WEATH STATE WATE	STATE WATER LEVEL								
STATE HALL	GPS COOKD'S CONG: LAT.								
-		· · · · ·	DISTANCE	FROM T	OP OF PIPE	TO:			
1		TIME	WATER	(FEET)	PRODUCT	(FEET)			
PRE-PURG	E:	0:10	6.63	\$			· .		
POST PUR	GE:	10:25	7.0	2		<u>_</u>			
	DEPTH OF	WELL		12.			- EEET I		
DEPTH TO WATER 7. 7.3 FEFT									
	DEPTH OF	WATER C	OLUMN		H7	· · · · · · · · · · · · · · · · · · ·	FEFT		
4	FACTO	Ω .#	フル	X 0	.1.18				
WELL PURGE	FACIO		2544	<u> </u>	- CQ				
	Valor	FJOREI	KIMOVED	L	3.77	·····	J		
TIME	pH	TEMP	CONDUC	TIVITY	SALINITY	TURBIDITY	DISS. O2		
	(SU)	(C)	(umo hs	ICM (mS	(0/00)		(mg/l)		
:25 *	7.06	+4.9				700	1.06		
10:50	7.40	15.6	1.00		0.5	270			
10:55	7.41	15.6	0.96		0.5	120	·		
и:00	7.42	15.8	0.98	<u> </u>	0.5	210			
					i	· · · · · · · · · · · · · · · · · · ·			
					l				
				· · · · · · · · · · · · · · · · · · ·					
	 								
	ł				· · · · · · · · · · · · · · · · · · ·				
		{							
						·····			
			<u> </u>						
			t						
			1						
			1		· · · · ·				
	1	1	1						
B				و بادواند آنوزوی برای امیر بیرمید اور .					
SAMPLED	<u>BY:</u>								
		<i>C</i> ;	0 1	1 0 1	, , ,		المتبادة ومناصلة في بين والتي وفي التاريخ الما		
COMMEN	<u>TS:</u>	Dample	for met	als fil	tered, hi	gh turbitity.			
		* Well Bui	meed cumpin	ucusl. 1 to	1) next in	Hing			
			1			· · · · · · · · · · · · · · · · · · ·			

* FACTOR = 0.618 FOR LINCH DIAMETER WELL CASENG

24

MATERIALS ENGINEERING DIVISION

PROJECT: HH-PORT IVORY PEG SITE WELL DESIGNATION: PAMW-7D JOB NO: 501 - 233-295 DATE: 30/00 11 CASING DIAMETER: CHECK BOX FOR LOWFLOW: IRATE(HL): 911 Inch STKK-OPDISTANCE: EATHER CONDITIONS: GPS COORD'S : LONG :: LAT .:

		DISTANCE FROM TOP OF PIPE TO:				
	TIME	WATER (FEET)	PRODUCT (FEET)			
PRE-PURGE:	10:08	11.01	` `			
POST PURGE:	11=25	16.37	-			

				_			روی بر بین ویشمال میدخد به ایس ۲۰۰۰ مید انگریز بر ۲۰۰۰ میشند.	and the second second second second second second second second second second second second second second secon
		DEPTH OF	WELL		I	46-4	0	FEET
		DEPTH TO	WATER			[].c]		
•	DEPTH OF WATER COLUMN				35-34			
•	FACTOR #				X.	0.618	?	
	WELL PURGE	VOLUM	ETOBER	REMOVED		21.8	7	
	TIME	pН	TEMP	CONDUC	TIVITY	SALINITY	TURBIDITY	DISS. O2
		(SU)	(C)	(umohs	/cm)	(0/00)		(mg/l)
	10.22	7.78	15.2	35	54	1.4	110	-
	10-35	7.32	16.8	44	86	2.4	950	- 1
	11'00	7.32	17.0	30	94	1.6	O.R.	-
	11:16	7-31	16-9	31	23	1-6	OrRe	- 1
								· · ·
								1
						·		
•								
							ş	·
i			_					
			1	1				
		1	<u> </u>	1				11
		1				1		
		1	1		•			
-	· ·		1	1				

SAMPLED BY:

AZ & E.M.

COMMENTS:

V	erg	5	10 4	re re	cover	2-1£	runs	almost.	diz
L	dr	14de	tes a	rater					
	Filter	مولم	5	etal.	2				

1

* FACTOR = 0.618 FOR LINCH DIAMETER WELL CASENG

MATERIALS ENGINEERING DIVISION

WELL DESIGNATION: $4\omega - Gw - 7$ DATE: $11/24/0b$ CHECK BOX FOR LOWFLOW:IRATE(HL):CASING DIAMETER: $2^{\prime\prime\prime}$ InchWELL DESIGNATION:IRATE(HL):CASING DIAMETER: $2^{\prime\prime\prime}$ InchWEATHAR CONDITIONS:ISTANCE FROM TOP OF PIPE TO:IATI:IATI:DISTANCE FROM TOP OF PIPE TO:IATI:IATI:PRE-PURGE:11/14 $g.et1$ -POST PURGE:12:36 9.26 -DEPTH OF WELL14.98FEETDEPTH TO WATER $g.et2$ -DEPTH OF WELL14.98FEETDEPTH OF WATER COLUMN 6.16 FEETFACTOR *X0.618WELL PURGE(SU)(C)(SU)(C)(umohs/cm)(0/00)(mg/l)(mg/l)11:34 6.87 IGM/4.710:39 6.60 7.22 0.3 11:39 6.61 7.02 0.3 11:39 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50 6.61 16.7 7.04 11:50	WELL DES	JON STONE						~ (Z		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	WELL DESIGNATION: MW-GW-7 DATE: 11/24/00									
IME ATHER CONDITIONS:ISTRICURTER LEVELSTATIC WATER LEVELGPS COORD'S : LON(τ :LAT::DISTANCE FROM TOP OF PIPE TO:PRE-PURGE:III/I 4G'R2DEPTH OF WELLIVI-UP DISTANCE :DEPTH OF WELLIVI-UP PODUCT (FEET)PRE-PURGE:IIII 19DEPTH OF WELLIVI 98DEPTH OF WATER COLUMN6.16MELL PURGEIVI 0000IME PH TEMP CONDUCTIVITYSALINITYTURBIDITYDISS. 02III: 366.87IVI 0000III: 366.87IVI 0000III: 366.87IVI 0000III: 366.87IVI 0000III: 366.87IVI 0000III: 366.87IVI 0000IVI: 36IVI 0	CHECK	BOXFOR	LOWFLOW	1. RATE	(HL):	CASING D	IAMETER:	2″ Inch		
STATE WATER LEVEL GPS COORD'S : LONG: LAT: DISTANCE FROM TOP OF PIPE TO: TIME WATER (FEET) PRODUCT (FEET) PRE-PURGE: 11/14 8.42 - POST PURGE: 12:36 9.26 - DEPTH OF WELL 14.98 FEET DEPTH OF WATER COLUMN 6.16 FEET DEPTH OF WATER COLUMN 6.16 FEET FACTOR * X 0.618 - WELL PURGE VOLUME TO BEREMOVED 3.81 TIME PH TEMP CONDUCTIVITY SALINITY TURBIDITY DISS.02 (mg/l) 11:36 6.87 167 707 0.4 60 - 11:39 6.60 16.7 704 0.3 5.1 - 11:45 6.6/ 16.7 704 0.3 5.1 - 11:45 6.6/ 16.7 704 0.3 5.1 - 11:45 0.66/ 16.7 704 0.3 5.1 -	WEATH	ER COPI	DITIONS:				CL-UP DISTANCE:			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	STATIC WAT	STAIL WATER LEVEL GPS COORD'S : LONG: LAT.:								
PRE-PURGE:TIMEWATER (FEET)PRODUCT (FEET)POST PURGE: $1!:14$ $g \cdot g \cdot 1$	-		·	DISTANC	E FROM TO	OP OF PIPE	TO: 1			
PRE-PURGE: $1 :14$ $g \cdot g_1$ POST PURGE: $12:36$ $q \cdot 26$ DEPTH OF WELL $14 \cdot qg$ FEETDEPTH TO WATER $g \cdot g_2$ FEETDEPTH OF WATER COLUMN $6 \cdot 16$ FEETVELL PURGE $FACTOR *$ X $0'61g$ WELL PURGEVOLUME TO BEREMOVED $3'81$ TIMEpHTEMPCONDUCTIVITY(SU)(C)(umohs/cm)(0/00) $(1:36)$ $6 \cdot g7$ $16 \cdot g$ $11:39$ $6 \cdot 61$ 702 $0 \cdot 3$ $11:39$ $6 \cdot 61$ $16 \cdot 7$ 704 $0 \cdot 3$ $11:39$ $6 \cdot 61$ $16 \cdot 7$ 704 $0 \cdot 3$ $11:39$ $6 \cdot 61$ $16 \cdot 7$ 704 $0 \cdot 3$ $11:39$ $6 \cdot 61$ $16 \cdot 7$ 704 $0 \cdot 3$ $11:45$ $6 \cdot 61$ $16 \cdot 7$ 704 $0 \cdot 3$ $11:570$ $16 \cdot 61$ $16 \cdot 7$ 704 $0 \cdot 3$ $11:570$ $16 \cdot 7$ 704 $0 \cdot 3$ $5 \cdot 1$ $11:570$ $16 \cdot 7$ 704 $0 \cdot 3$ $5 \cdot 1$			TIME	WATER	(FEET.)	PRODUCT	(FEET)			
POST PURGE: $12:36$ 9.26 $-$ DEPTH OF WELL 14.98 FEETDEPTH TO WATER 8.82 FEETDEPTH OF WATER COLUMN 6.16 FEETFACTOR *X 0.618 WELL PURGEVOLUME TO BEREMOVED 3.81 TIMEpHTEMPCONDUCTIVITY(SU)(C)(umohs/cm)(0/00) $11:36$ 6.87 16.87702 0.4 $11:39$ 6.60 16.67702 0.3 $11:39$ 6.60 16.77704 0.3 $11:45$ 6.61 16.77704 0.3 $11:50$ 6.60 16.77704 0.3 $11:50$ 6.60 16.77704 0.3 $11:50$ 6.60 16.77704 0.3 $11:50$ 16.77704 0.3 5.1 $11:50$ 16.77704 0.3 5.1 $11:50$ 16.77704 0.3 5.1 $11:50$ 16.77704 0.3 5.1	PRE-PURG	ie:	11:14	8.	\$2	~		•		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	POST PUR	GE:	12:36	9.	26	-		·		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $										
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		DEPTHON	WELL			14.9	8	FEET		
Interference Interference Interference Interference WELL PURGE FACTOR # X 0.618 VOLUME TO BEREMOVED 3.81 TIME pH TEMP CONDUCTIVITY SALINITY TURBIDITY DISS. 02 (SU) (C) (umohs/cm) (0/00) (mg/l) 11:36 6.87 102714.7 707 0.4 60 11:39 6.60 16.6 702 0.3 14 - 11:45 6.61 16.7 704 0.3 5.1 - 11:50 16.7 704 0.3 5.1 -	DEPTH OF WATER COLUMN 6.16 FEET									
WELL PURGE PACION + Solution VOLUME TO BEREMOVED 3.81 TIME pH TEMP CONDUCTIVITY SALINITY TURBIDITY DISS. 02 (SU) (C) (umohs/cm) (0/00) (mg/l) (mg/l) 11:36 6.87 16.7 707 0.4 60 $-$ 11:39 6.60 16.6 702 0.3 14 $-$ 11:45 6.61 16.7 704 0.3 5.1 $-$ 11:50 6.6 16.7 704 0.3 5.1 $-$		TA(TO	D .*	<u>O LONAT</u>	X	016	18			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	WELL PURGE	FACIO		2544-164	<u> </u>	2.0	21			
TIME pH TEMP CONDUCTIVITY SALINITY TURBIDITY DISS. 02 (SU) (C) (umohs/cm) (0/00) (mg/l) (mg/l) 11:36 $6 \cdot 87$ 1637/14.7 707 0.4 60 $-$ 11:39 6.60 16.6 702 0.3 14 $-$ 11:45 6.61 16.7 704 0.3 5.1 $-$ 11:50 6.60 16.7 704 0.3 5.1 $-$	<u>.</u>	Valor	ETOISE	KITOVED	L	• 7*8	-1			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	TIME	pН	TEMP	CONDUC	TIVITY	SALINITY	TURBIDITY	DISS. O2		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(SU)	(C)	(umohs	/cm)	(0/00)		(mg/l)		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	11:36	6.8/	199719	1./ 10		0.4	60			
11:50 6.60 16.7 DOF 0.3 C.9 -	11.37	6.60	16.7	70	<u> </u>	0.3	<u> </u>			
	11:50	6.60	16.7	701	7	0.3	<u> </u>			
			J							
		<u> </u>				1.				
		l		·		 				
						<u> </u>				
							· · · · · · · · · · · · · · · · · · ·			
		1								
							·			
				 						
				 						
					<u> </u>					
						لم میں محمد میں میں ان				
SAMPLED BY: $A Z \neq D P$	SAMPLED	BY:	A	24DF	2					
			and the second second second second second second second second second second second second second second secon		يسين الكرية المسترجات المانية موردة	۵×۳۳, بنور او الابر او می او ا		الم معن من الم من الم الم الم الم الم الم الم الم الم الم		
COMMENTS:	COMMEN	TS:								

* FACTOR = 0.618 FOR LENCH DIAMETER VELL CASENG

MATERIALS ENGINEERING DIVISION

PROJECT: HH - PORT IVORY PEG SITE	JOB NO: 501-233-295
WELL DESIGNATION: GW - 10	DATE: 11-24-00
CHECK BOXFOR LOWFLOW: IRATE(HL):	CASING DIAMETER: 2 Inct
WEATHER CONDITIONS: SUNAY . 34°F	STKI-UPDISTANCE: 0.0
STATIC WATER LEVEL GPS COORD'S : LO	NG:: LAT.:
-	

· · · · ·		DISTANCE FROM T	OP OF PIPE TO:
si	TIME	WATER (FEET)	PRODUCT (FEET)
PRE-PURGE:	3'26PM	6.98	
POST PURGE:	4:19PM	7.81	

•	DCOTL OF	~ \ \ \ \ T	۲ ـ مر		12 112		
	DEPTHO	VYELL			13.42		reel
	DEPTH TC	WATER			6.98	·	FEET
	DEPTH OF	WATER C	OLUMN		·	FEET	
	FACTO	R # .		X	0.618	?	
WELL PURGE	Valum	etober	REMOVED		3.98		
TIME	рН	TEMP	CONDUC	XTIVITY	SALINITY	TURBIDITY	DISS. O2
	(SU)	(C)	(umohs	/cm) (v 5)	(0/00)		(mg/ī)
3.30Pm	8.13	15.20	2621)	1.4	50	
3:42 m	8.07	15.10	252	2	1.3	30	-
3:50pm	7.69	15.2°	2718	>	.1.4	a5	
3:59PM	7.72	15.30 .	2712		1.4	23	
							·
						· · · · · · · · · · · · · · · · · · ·	11
				· ·	·		
				······			
	1					· {	
	1						11
	1						11
	1						11
							11
	1	1					
	1			···· <u>·</u> ·······························			
 		<u> </u>					
				······			
l		·	 				

SAMPLED BY:

R.6 / Em.

SOMMENTS:

well sampled @ 4:17PM

* FACTOR = 0.618 FOR LINCH DIAMETER VELL CASENG









a,

P.02

PA 547 6-90

THE PORT AUTHORITY OF NY & NJ

EngIneering Department Construction Division Materials Engineering Section

BORING REPORT

									SHEET OF 3
PROJECT	~				NAME OF CONT	RACTOR	E	ORING NO.	SURFACE ELEV.
HHMT-	Port	FLIDTY S.	urchan	, T.osT	Crocri	Or, Iling		PG-ST-15	
LOCATION		2				•	C	ONTRACT NO.	DATE
Aslanda	JIN	Field as por	Otoget	h h	avTh Side			426-99-006	10/23/02
SPOON	13/	CASING S	IZE HOLE	TYPE			GROU	ND WATER LEVEL	
"O.D.	118	"I.D. Hryprz	H M	Norlar	Date	Time	Depth	/ /	Remarks
140 # FAI	#050 11.30	#	# FALL	-	10/21/02	12 25	3.7	Lehilo Kan	ALGULL.
DRILLER	1	2 Carlos							
INSPECTOR	<u>م</u>	NV.				······	·····		
	l	love		1				1	······································
CASING BLOWS/FT.	EPTH_	SPOON BLOWS/6"	RE- COV'D	SAMP." NO.	·	"SA! LIN	NPLE DESC	RIPTION AND REMAR S CHANGE OF PROFI	LE O,¢
Mand	0	MardArca	R11			10	-		
Auger					F.11 - K	CA-R	cyclad	Concrete Par	
1				~ A	Same	 رو	· · · · ·		3.0
-+-+-				Lh	E.II - (e] / e-	-	······································	
			<u>├</u>				Phila	· · · · · · · · · · · · · · · · · · ·	
	5 -			3			·		
					Same				6.0
Nollae		1-coot		•	·	·		·	
STORY		WOH	19'1	9	Fill wh	To 2 Gra	1 4mes	Paro Storns	
Aura		k.							
1		O H	221	5	<u> </u>		··		
	10 -	₩. ₩			Zame				
				6					
		Н	24"	. *	Same				· ·
		les .							
		V	244	7	<u> </u>		· ······		
		م ا				· ·····	<u> </u>		
	15			8					
	_	Н	241		Same	. <u></u>	<u> </u>		
		2		a	,,				
	•	H	20"	7	Same			· · · · · · · · · · · · · · · · · · ·	
VT	-	4		A	Fill - Cor	ease lim	tano S		/9.//
	2	H	16"	10 B	Black	Post		×	200
					VICE		······		
				f					#
			{	ŀ				_ Kallanet Ba	±145
				Ļ	All S			TI PINNT.	
				F	N. S.	unter Sai	1 DI	Preco 1	
		S: 1 - Length me	overed: 0	·····	I Comple F	407 -90	we Th	wiscurdar	



.

THE PORT AUTHORITY OF N.Y & N.J.

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION PID READINGS

FID READ	NGS BY	Ditan			PID Model	Nr. R.D.	
	SAMPLE	IN-SITU Split Spoon	HEAD-	BREATHING		REMARKS	
TIME	No.	Reading	Reading	Reading	ļ		
Ax	1.		0.0				
) .	2A		U.C	•			
J	2B		0.0				•
PN	3.		. 00			· · •	
·	4		00		,		· · · · · · · · · · · · · · · · · · ·
	5		0.0			•	·
11	6		0.0		•		
	7		0.0				* <u></u>
	8.		00				
	9		0.0		······································		
	10A		0.0				<u> </u>
	IUR		0.3		Part	· · · · ·	·
						······	
					······································	· · · · · · · · · · · · · · · · · · ·	
		<u>-</u>				·····	
.							••••••
				 			
						· · · · · · · · · · · · · · · · · · ·	
						المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحم المحمد المحمد	

^

CHEFT

~ -

PA 547 6-90

دم

THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

						····			
PROJECT		^			NAME OF CONT	RACTOR	B	IORING NO.	SURFACE ELEV.
HHMT	- POIT IL	ioni Surchers	e Tost	-	Crais D	rilling		PG-ST-10	·
LOCATION	·				0	0 -	C	ONTRACT NO.	DATE
BSLaid	OUT in f	old 45 May Dray	mar/	North	1 Sito			426-99-006	10/24/02
SPOON	.3,	CASING SI	ZE HOLE	TYPE			GROU	ND WATER LEVEL	
9 "	0.D. 178	"1.D. Hw	B°,	MUNIJON	Date	Time	Depth	Re	marks :
HAMMER	AMMER HAMMER					1 34	-		
140	FALL 30		FALL	-	(2)24	102	5,0	Lehilo Head Aico	kid
DRILLER	0.4							0	•
	p_{Q}	ore							
INSPECTOR	NH	,							
		0400	·	,İ		<u> </u>		<u>]</u>	
CASING		SPOON	RE-	SAMP. ²		³ SA	MPLE DESC	RIPTION AND REMARK	S · · ·
BLOWS/FT.		BLOWS/6"	COVID	NU.	}		NE LOCATE	S CHANGE OF PHOFILE	00
Hand		Hand Durge	For	,					
N		1 0	1		Ell-F	CAR	and I	Circuit R	
TUER	h				'- <u>-</u>		-yeled	CULLING BEE	
	·			~			·	· · · · · · · · · · · · · · · · · · ·	
					<u> </u>	- <u> </u>			
	╆╍╸╷ ╺━		{		Jame				
				2					
				2	Ell-C	rach. 15-	5. S.1T	S. IFT.	
	- 6 -					103100131	che sing	- Jang LI (
Ju					L				
Casing					Brr	STr.To 1	· -za's	- La for PG-ST-1	15
	└── <u>─</u> ─					<u>,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,</u>			
	L				· · ·				
	1 -			1				,	
	- 00 -	1 0						• •••••••• •••••••	· ······ ······· ······· ·······
				4				· ·	
		1 -	8'	/	Black of	Brown	Post		
		4.04-1					. <u> </u>		
		10011-1		5					
		2-2	234		Sa	me			
		140Ha							
}	► 7×. <		1.1	6					
		2-2	117		San	ne.			260
		3-2]						
				7				· · ·	
		7-5	197		- torey	Jand Tr	-SiT_		
			ĺ		,	1			
				ţ					
	► 30. ◄	<u>-</u>							
		3-5		~				·	
		6-7	10"	ð	Same				
			- <u>`</u> -						
		· · · · ·		ļ					
	 -			F		* ********	<u> </u>		·
	► 35-								
	NOTE	S: 1 - Length rec	overed; 0	" — Loss	of Sample, T	Trap used			

2 - 0 = undisturbed; A = auger; OEH = open end rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artestan water, sand heave in casing, etc. _

-

г	-	٦.	F	۲	۱

PA 547 6-90

ر.

THE PORT AUTHORITY OF MYG MJ

Engineering Department Construction Division Materials Engineering Section **BORING REPORT**

										<u> </u>		SHEET 2 OF
PROJECT						NAI	ME OF CONT	RACTOR		E	ORING NO.	SURFACE ELEV.
HMMI	- POITI	Juary	1 Surc	hango J.	<u> </u>	16	rais Vr.	live			10-21-10	
LOCATION	~	, , ^T	0	*	Б		, -			C	CONTRACT NO.	DATE
BSLAID.	CUTIN \$10	d q	spar 1/1	auits	North	Sid	6		_,		<u>426-99-006</u>	10/24/02
SPOON	31		CASING S	IZE HOLE	TYPE			· · · · · · · · · · · · · · · · · · ·		GROU	ND WATER LEVEL	
0 - 6	.0. 118	°1.D.	ne	<u> </u>	MULLION		Date	Time	·	Depth		Remarks
HAMMER	Agic		HAMMER									
190 #	FALL 30	"	<u> </u>	# FALL		· -		<u> </u>			+	
DHILLER		0 0	Colle							· · ·	· · · · · · · · · · · · · · · · · · ·	
NSPECTOR		O H	oue	· · · · · · · · · · · · · · · · · · ·								
CASING		S BL	POON OWS/6"	RE- 1 COV'D	SAMP.'				SAMF	LE DESC	RIPTION AND REMAP S CHANGE OF PROFI	iks Le 3:
HU	- 32 -	14	-4	ĺ	0		••••					
Casing		5	5-5	10"			M-FI	Brew	Serc	l, tisi	17	3
							·			· .		91
			-	1					•••••••••••••••••••••••••••••••••••••••		Batton & Ba	
				+			• • • • • • • • • • • • • • • • • • •				10 10 10 10 10	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
	-40 -			· · · · · · · · · · · · · · · · · · ·		<u> </u>						
							Ð	1 Sam	nles	check	ed with PID M	Tou
							11	. <. '		C	AID	
						┣		<u>0 20m</u>	pre-	<u> </u>	DI MISCArde	24
						 	·				-	
, }												
												
	► Y5 ◀			<u>├</u>		<u> </u>						
											·	· .
1												
	~					—	·				•	<u></u>
				 							·	
_												
	- 50 -							•				
								· · · · · · ·				
}												
							<u></u>		·····			
l	\sim								_			
	-2014			{	ł							
					-							
	. 1				ſ							
	+				. F							
								-			-	
L	1 a			ł								
								Tar				

3 - Log depth of change in calor of wash water, loss of water, artesian water, sand heave in casing, etc.



SEP-02-2004 15:57

PROJECT: HMMT- Port Ivory Surcharge Test

THE PORT AUTHORITY OF N.Y & N.J.

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION PID READINGS

	Sheet	4 of 4
DATE:	10/24/02	
Model:	Mini RALD	
	REMARKS	

BORING No	PG- 57	. 20	0		DATE: 10/24/02
FIELD READ	DINGS BY:	D How			PID Model: Mini RALD
τιμε	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Space Reading	BREATHING Zonc Reading	REMARKS
AM	1.		0,0	<u> </u>	
	2		00	-	
	3		00		
	4		0.0		
L	5		00		
Pm	6		0.0		•
1	. 7		0.0		
	8		0.0		
	9		0.0		
·					
•					
		•			

-

P.09

Ρ	A	547	
	6-	90	

· +.

THE PORT AUTHORITY OF NYS NJ

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

· .									SHEET	OF 3
PROJECT					NAME OF CONT	RACTOR	B	ORING NO.	SURFACE	ELEV.
HHMT	-POITI	Vory Surcha	10 Tosi	+	Craig 1	Irilly.		PGST25		
LOCATION			/				C	ONTRACT NO.	DATE /	
As Laid	cutin fi	old go por d	rquing	wos	r Sida	<i></i>		426-99-006	(0)2	sloz
SPOON	. 7/	CASING SI	ZE HOLE	TYPE			GROU	ND WATER LEVEL		
2 .	D.D. 178	"1.D. Hugers	7.1	Xcator	Date	Time	Depth		Remarks	
HAMMER	Auto	HAMMER				45			·	
140 #	FALL 30	•	# FALL	•	(0/25/02	812	3,0	while do.	ud Augorla	
ORILLER	D	Carlo							Ŭ	
INSPECTOR	\mathcal{D}	Howe								
CASING BLOWS/FT.	DEPTH	SPOON BLOWS/6"	RE- ' COV'D	SAMP,3 NO.		3SA	MPLE DESC	RIPTION AND REM.	ARKS FILE	0,
H. 1	0	Harlp	Eu	, B	Ell-Cin	due las	1 Coul	let la a	+ 577	
oland	<u>├</u> —	UIANA DUAN	1 1		1 111 112	un ural	, crush	Jong COUCY	RIG EIC	
Augo	L			B	_Fill - U	thite of C.	rey Lime	STORE STUTTY		<u>.</u>
۰,							/	,		
			<u> </u>	2		•				
		┝━━━━┝━━━━━	├)ame		•	• •••••		
							<u>_</u>			
	►· > ◄	<u>,</u>		3	San			· · ·		
	├ ,	¥			- una					
Nollow		w q		u		·		·		
STam		1	241	7	Sama					• • •
n		· · · · · ·				· <u></u> ·	· ·			
mors	,	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		$\overline{\mathbf{\nabla}}$		·		·		
		H	24"	5	Same					
		2-2-								
			2.1	6		. <u> </u>				
		1-1	82'		<u> </u>	_ Dr. ka	oud			
	· .	-0		-						
		1-1	727			· ·····				
			~~		same	. <u> </u>				
		4-2		8.1			· ······			
		1-7	6"	0	5-	T	2,			
							w000			
				9						
V		H	121		Same					
		2-1								
	- 20.	1-2	75	10	Blacko	Brown	Post			20.0
								Batton	R	1
			{	ŀ	······			D CM CJ	LOTINE _	
				Ļ						
			l		BIISO	mails c	hockalu	UTH BID MO	7.02	
				ľ	N C	, , , , , , , , , , , , , , , , , , , ,		0 11		
- <u></u>	<u> </u>			ŀ		inger 2	allow 171	UISCADDOD		
1	2-		1	ſ		v		· .		

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.



*

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION PID READINGS

	·			U HEADINGS	Sheet 3 of		
ROJECT:	HAMT - F	CIT IVONY	Surchaise	Tost			
ORING No	. PG ST	25			DATE: 10/25/22		
IELD REAL	DINGS BY:	p How		PID Model: Min, PAG			
TIME	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Space Reading	BREATHIN Zone Reading	IG		
AM	.A		00				
1	1B		00				
	2		0.0				
	3		00				
•	4		0.0				
	5		0'0	4.			
	6		0.0				
	7		0.0				
	४		0.0				
	5		0.0	•			
\vee	0		0.7		Post		
				· · ·			
·	. \				·		
					·		
	·						
		<u>·</u>			·		
		1					
	·						
		· · ·					

.

•

. . .

PA	547
6	·90

THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section BORING REPORT

									SHEET UP	- 3
PROJECT	<u> </u>	· · · ·		~	NAME OF CONT	RACTOR		BORING NO.	SURFACE ELEV.	
HAMT	- Parl	Ivory Su	rcharge	Test	1 Craige V	rilling		86-5T-2D		
LOCATION	(· ·		0			CONTRACT NO.	DATE	
Hs Laid	low in t	iold as Por 1	rowing	leves	S.da			186-71-0CG	10/25/02	
SPOON	13/	CASING S	IZE HOLE	TYPE		·	GROL	JND WATER LEVEL		
d -0	n.d. 1/8	"I.D. 19	5 12	ONIJOY	Date	Time	Depth	Re	emarks	
HAMMER	H-JJ	B HAMMEH			1.br	, on	30	. h/. V. /	ρ.	
1-10 #	FALL 20	<u> </u>	# FALL		10100	<u> </u>	,	Dunile Blond	Hugoting	
DRICCER	D	Carlo								
INSPECTOR										
	0.	Houe								
CASING		SPOON	RE- '	SAMP.2	f	3SAI	MPLE DESC	RIPTION AND REMARK	(S	
BLOWS/FT.	DEPTH	BLOWS/6"	COA.D	NO.		LI)	NE LOCATE	S CHANGE OF PROFIL	E	0.0
Main		Hand Press	P.II	A, F	F. 11- Gara	lana Corquet	Cuchel	Star Sad ETC		t.o
N			1			The second	1 rus red	JULIO, MANO, CIC		
Huger		<u> </u>	<i> </i>	U	F.II Whi	To forsy	Lingester	- Sturmy		
								,		
		1		2	<					
<u> </u>			┼─-┼				·			
				5	Same					
N	— 6 —	¥	<u></u>							
nu	·	<u> </u>								
CaSING					For	STr-Ja	5-20' 5	Toolog for PGST	20	
10-1										
		ļ			······································		·	······································		
	► <u>2</u> ~. ◄						· •		· · · · · · · · · · · · · · · · · · ·	
l [00	Wall		U						
		1.1	14'1	7	<u> </u>	<u> </u>	·			
		+			Djock of	Prouse Pr	2e 1		, 	
		1-1-2		<u>_ #</u>	Same	2				23.0
		2-7_	19"	3 3	MF A	Recht Sr.	1 The	с. /Г		
						<u></u>	ap ins			2
>	- 25 -				·				>	<u>~.</u> ,
		2-2		1						
		3-5	7'1	Ø		c. 1 7			······································	
					- i- oray	- 2010 - 1	r>11_			
		L		L	· · ·				-	
			· 7	、						
				ł	nuc				· ·····	
	30 -			Ļ	<u></u>	mples ch	school in	th VIO MOTR		
					Ne S	angle Sc	und , D	+11 Discardod		
				Γ		-v				2
							· · · · · · · · · · · · · · · · · · ·		ر	12.0
		2-3		7						
		5-4	91		FBra	ww Som	I, Jr S.	1J	3	14.0
							1	DTT P A	. 0	7
	- 35	l				·		pollom of Bor	·**	
	NULE	2 - U - undia	covered; 0	- Loss	of Sample, T -	Trap used				

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.



- .

THE PORT AUTHORITY OF N.Y & N.J.

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION

PID READINGS

						Sheet	3 of ;
PROJECT:	HH MT- 1	Pari Ivory	Surcharge	Tor			
BORING No.	- PG 55	5 20			DATE:	10/25/02	
HELD READ	INGS BY:	Охоцо			PID Model:	Min, PBE	
TIME	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Space Reading	BREATHING Zone Reading	5	REMARKS	
PM	1.4		0.0				
1	1B		0.0				
	2		6.0				
	3		0.0				
	.4		0.2		Post		
	54		0.3		· /		
	SB		0.0		,		
	6		() ro		<u></u>		
	7		0.0				··
				•			
						<u>، میں ایک بی پی میں ^{پر} بالی کی محمد اور پر میں ان محمد ہے ہ</u> یں ا	
						*** :: ····****************************	•
							<u></u>
						•	•
					x	-	
						·	
					• •		
							<u>ة</u>

.

_

PΑ	547
_	
6-	90

THE PORTAUTHORITY OF MYG MJ

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

					•	· · · · · · · · · · · · · · · · · · ·	SHEET / OF 3
ROJECT				NAME OF CONTRACTOR	B	ORING NO.	SURFACE ELEV.
HUMT-POTI	Sure	haven the	st	Crace Dr. 11100		PG-57 - 35	
OCATION		large			C	ONTRACT NO.	DATE
\mathcal{D}_{-}	211an	2	S.	th S (-		421-99-00C	10/22/02
TS Laid OUT IN J	CASING SI	ZE HOLE	TYPE	J Juse	GPOU	ND WATED LEVEL	30000
2 11/2	n	mr.		Data Time	GROUI	WATER LEVEL	
J "0.0. 1 %	"I.D. HUCALS	0/	Mar Tor	Date sime	Deptn		iemarks
IAMMER DUTO	HAMMER			11/21 615	27	1111	10
140 # FALL 30		FALL		10/22/2 8-	215	Letilo hard	Hisorius
	\sim 1						<i>a</i> ,
V	Coore						
NSPECTOR	10						· · ·
\mathcal{D}	dove					· · · ·	
CASING	SPOON	RE- '	SAMP,2	³ SA	MPLE DESCH	RIPTION AND REMAR	KS
LOWSIFT. DEPTH	BLOWS/6"	COV.D	NO.	· L	NE LOCATES	CHANGE OF PROFI	
VIII	11. in	E,					
nana	Mand Huger	141	\ .				
ALCON	l í		¥	FII-Charle Ist	SIT.C	iracal Caliblas	ETC 2-
		 -	~~~~~				
					<u> </u>		
			2	Ell. L. L.	14-1	-T C/	
				_1 L' Urey The	In la Li	mesione Jun	¥
			~	/			,
-+			3	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~			
		V		Jame		·	
Hallows	k.		,,			-	
	_ 		4			· • • •	
STEM	- 1	24		Same		·	
	1.	·					
Jugers _	<u> </u>		5				
	U U	24		Same			
		<u> </u>					
	n		1				
	И	777	b				
		oc.		<u></u>			
	LCH		_				
-+			7				
	1-1	209		Jama			
						· · · · · · · · · · · · · · · · · · ·	
> 15 ◄	-Vo		81				······
	Н	154	5	Same			
)		7		· · ·		
	4 mm		0 B	Jame			
\vee	~ H	181	7 -	Rigel D			/75
	- ,		¥	DIACK POST			18.0
							a
			t				
			Ļ			Ballomat	-Bering -
							<i>u</i>
			F	 ·/·			
			1	<u>H11_Samo</u>	to check	Luith PID M	et s
		- 7		NC		1 n. n.	1,
┝────┽── ╺──┼			ŀ	- in -angle	5_2quel	HUN USCOLO	red
					,		
			F				<u></u>
	1		- e e -				

3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

P.16



÷

THE PORT AUTHORITY OF N.Y & N.J.

.

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION PID READINGS

			· · · ·			Sheet	3 of
PROJECT:	JUMI-6	BIT TUOIN	Surcharge	Tost			
ORING No	. PG57	35			DATE:	10/22/02	
IELD READ	DINGS BY:	- O'Hour	~~~~~~		PID Model:	MILI RDE	
TIME	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Space Reading	BREATHING Zone Reading		REMARKS	
AN	1.		6.0				
1	2		00	1 ·			
	3		0.0				
	Ý		.00				·····
<u>`-</u> /	5		0.0				· · · · ·
1	6		G.c	·	·		
+}	7		00		· · · · · · · · · · · · · · · · · · ·		
+	8		00				
+	94		00		· · · · · · · · · · · · · · · · · · ·		<u></u>
	- GR		02		Part -		<u></u>
-V							`
						****	•
						<u></u>	
					~**	مىي يەرە رە مەرە بىرىمىيە - بىلى يەر	

				_		· · · · · · · · · · · · · · · · · · ·	<u> </u>
						······	
							·····
		· · · · · · · · · · · · · · · · · · ·					
				· · ·			ì
							<u>.</u>
					<u></u>		
• 4

.

PA 547 6-90

THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

PROJECT <u>HHMT</u> - LOCATION <u>DSLACOC</u> SPOON <u>JOD</u> HAMMER <u>JYC</u> # FAI DRILLER INSPECTOR CASING LOWS/FT. <u>Mand</u>	Porty 13/2 13/2 DOEPTH	- Lory CASING SI CASING SI "I.D. HUE HAMMER - Coule SPOON BLOWS/6" Hand n.	Prowing ZE MOLE B'	Scutter Manta	NAME OF CONTRACTOR BORING NO. SURFACE ELEV. Craig Drilling PG-ST-30 DATE TG S.Lo CONTRACT NO. DATE GROUND WATER LEVEL Date Icil 22/c2 Date Time Depth Remarks Ici2 2/c2
HANDER HAMMER JYU # FAL DRILLER NSPECTOR CASING HOWS/FT. D Hand D	DEPTH	Locite Locite Locite Locite HAMMER - Locite Houe SPOON BLOWS/6" Hard n.	Prowing IZE HOLE B'	TYPE Mausta	Craig Villing PG-ST-30 CONTRACT NO. DATE 10/22/02 GROUND WATER LEVEL Date Time Depth Remarks 10/2702 11 ^{an} 275 While Hand Augering
LOCATION D-SLandon SPOON J-O.D. HAMMER JYC # FAIL DRILLER INSPECTOR CASING HOWS/FT. D Hand D	DEPTH	Coll 95 Dor CASING SI "I.D. He HAMMER - Colle Houe SPOON BLOWS/6" Hourd n.	Prowing IZE HOLE B'	TYPE Manita	CONTRACT NO. 12/22/02 CONTRACT NO. 12/22/02 Date Date CONTRACT NO. 12/22/02 CONTRACT NO. CONTRACT NO. 12/22/02 CONTRACT NO. CONTRACT NO. CO
D-SLAWOC SPOON J-O.D. HAMMER 140 # FAL DRILLER INSPECTOR CASING ILOWS/FT. D Hand D	DEPTH	Casing si Casing si -1.D. He HAMMER - Coult Howe SPOON BLOWS/6" Hourd n.	RE-	TYPE Mausta	Th S. So Y26-99-006 10/22/02 GROUND WATER LEVEL Date Time Depth Remarks Id2tor Id2tor Id2tor Id2tor Id2tor Id2tor
SPOON 	13/5 DUDC 11 30 0 0 0 0 0 0 0 0 0 0 0 0 0	Casing si "I.D. He HAMMER " Coola Moue SPOON BLOWS/6" Hard n.	RE- 1	type Mausta	GROUND WATER LEVEL Date Time Depth Remarks Idztor 11 ^{an} 275 while Hand Aussetike
J O.D. HAMMER 140 # FAL DRILLER NSPECTOR CASING LOWS/FT. Mand		"I.D. He HAMMER " Coole Move SPOON BLOWS/6" Herel n.	# FALL	*	Date Time Depth Remarks 10/27/02 11 an 275 while Hand Ausertine
HAMMER 140 # FAL DRILLER NSPECTOR CASING ELOWS/FT. Hend D		HAMMER - Coula Doue SPOON BLOWS/6" Herel n.	# FALL		Idata 11° 255 while Hard Augerine
140 # FAI DRILLER NSPECTOR CASING BLOWS/FT. D Hend		- Xoue SPOON BLOWS/6" Herel n.	RE- 1		Idada 11 255 while Hard Augerina
CASING LOWS/FT. D		Louie Houe SPOON BLOWS/6" Herd n.	RE- 1		
CASING LOWS/FT. D Mand		Louiz Houz SPOON BLOWS/6" Hourd n.	RE- 1		
CASING ILOWS/FT. D Mand		House SPOON BLOWS/6" Housel n.	RE- 1		
CASING ELOWS/FT. D Mend		SPOON BLOWS/6" Hard n	RE- 1		
LOWS/FT. E Herd		SPOON BLOWS/6" Hard D.	RE- 1	-	
Hend		BLOWS/6"	1 COVPD	SAMP.2	SAMPLE DESCRIPTION AND REMARKS
Mand	_	Hourd no	COVD	NO.	LINE LOCATES CHANGE OF PROFILE 0.0
n			5.1	•	
$\boldsymbol{\Omega}$		and break	1 1		
Flear				۱ 	till- (rushed stone, SIT, Growl, Sand wood, ETC 2.0
0					· · · · · · · · · · · · · · · · · · ·
			┟╌╌┨───┥	>	
				<u>د</u>	Fill-Gray + White Limos Dove Skirws
	-				
	6 4		├├	2	
	,			>	Same
diar					P TO (1/2'S / C P/SZ 20
	,	······································			For Strila 0-18 Jer Log dur FU-51-35
osws.	10				
	18-	4/			
		G		U I	
		\mathcal{H}	12"	'	Black & Brown Past
	20 -				
		- year-			
		H	13	5	Sama
) - /			
		l = l		6	
1 1		2-3	23 1	Q	Some Trof F Alach Soul intin of moon 2110
		1 1		~f	
	25-4	(-)		7	
		2-2	71	,	F Brown Gray Sound IN SILT
	+				
<u>_</u>				ļ	
					HI Samples charles in The PID Matter
1	. –†				
					NO Samples squad, 441 Uiscatchad
Ι Ι.					
	30 🖣	2			
	-+	3-3		\sim	
		8-8	7"	0	F Gren Sand Tr Silt
1	+	- <i>v</i>			
	ļ	2-3 1			
1,1	-+	1 7	1211	91	
	354	6-1	10.	!	Jame 35.0

PORT AUTHORITY OF NY & NJ

Engineering Department - Materials Division Well Installation Report Sheet 2 of ? PROJECT CONTRACT NO. HAMT- Port Twony Surcharge Tast-426-95-006 CONTRACTOR LOCATION AS Laid OUT IN Field as per Prouse, South Side IL NO. WELL TYPE Crois Prilling DRILLER WELL NO. DATE B' Munto 1 House O Coole PG-ST-10/23/0-Well Development Report (NOTE: WATER LEVEL READINGS FROM TOP OF PVC) DATE 13.5 13.5 10/23/02 15 WATER LEVEL BEFORE WATER LEVEL AFTER TAKEN MINUTES AFTER ____ dia. PVC pipe w/steel locking cap 11 = 3.0'L1 L2 = 25.0" Top of surface L3 = L0.0& cement grout <u>م</u> $\frac{22.0'}{2}$ Top of bentonite seal L2 $\frac{\partial 4.\sigma'}{\partial t}$ Top of well gravel filter ٠. spenings 020 L3 35.0' Bottom of well <u>35.01</u> Bottom of boring Cap 14 Boring diameter APPRS:

THE PORT AUTHORITY OF N.Y & N.J.

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION PID READINGS

						Sheet	3 of 3
PROJECT:	HUMT-	Port Ivon	1 Surc	have Tost	· · · · · · · · · · · · · · · · · · ·		
BORING NO	. PG-5	<u>J.30</u>	····		DATE:	10/22/02	
FIELD READ	DINGS BY:	DHOUR			PID Model:	MIN RAE	
TIME	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Spece Reading	BREATHING Zone Reading		REMARKS	
PM	1.		0.0				
	2		0.0	•			
	3		0.0				
	4		0.0				
-1	5		0.0				
PM	6		ÛU -				
)	7		G.C		•	· · · · · · · · · · · · · · · · · · ·	
	8		0.0				
J	9		00		· · ·		
					· · · · · · · · · · · · · · · · · · ·	-	
						· · ·	
							•
· · · · ·						······································	
	ľ					`	
					· · · · · · · · · · · · · · · · · · ·		
	4						
					· · · · · · · · · · · · · · · · · · ·		

]

N

PA	547	
6	-90	

THE PORT AUTHORITY OF NY & NJ

Engineering Department **Construction Division** Materials Engineering Section

BORING REPORT

										SHEET OF 3
PROJECT					NAME	OF CONT	RACTOR		BORING NO.	SURFACE ELEV.
HAMT-0	6TI IVO	14 Surcher	4 TAT	-		raig	Urilling		46-55-45	
LOCATION	_ (<u>^</u>	Έ.		(CONTRACT NO.	DATE
HrLaid	outin +	-lold sspar	Itowicy	, <u> </u>	<u>, < 15</u>	do			786-77-006	10118107.
SPOON	3/1	CASING S	ימי א			Data	Time	GRO	UND WATER LEVEL	Jamoslur
J O	.D. 178	HAMMER		Marije W		Date	Line .	Depui		10mmras
1401 4	HUID FALL 31		# FALL		11	0/18	11 2	3.0	1. hile Hand A	martin A
DRILLER	-									21.48
	<u>()</u>	Coolea								
INSPECTOR	n	Maria								
CASING	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	SPOON	RE- 1	SAMP.2	<u></u>	I	³ SA	MPLE DES	CRIPTION AND REMAR	IKS
BLOWS/FT.	DEPTH	BLOWS/6"	COV'D	NO.			LII	NE LOCAT	ES CHANGE OF PROFIL	LE GC
Hord		Mand Auger	RII	<u></u>	Fil	-PCF	+ Recy	clad co	increto Agy	1.0
Deer	_			B	Fil	- Crus	had sten	, SIIT, S	and Uncisto . ET	<u></u>
		1 1		n n				7 / ==		
		++	├┼	2 5			·	,		<i></i>
		<u> </u>	├}_	5	Fil	1 lehi	Cot Gray	L/me st	ene Sturry	••••••••••••••••••••••••••••••••••••••
				_				<u> </u>	-	
				3		Samo				
1 1/	····· •	1-1						•		
a allow				4		~				
STOM		Won	101		·	Same	 _			
Augo		14-		F				_		
1	10	J UN	Jer	0		Samo				
		W C. H								
			110	6		<u> </u>	·			 ,
		1-1-1	11			Same		<u></u>		
		Wa								
		H	91			Samo				
	<u>~~</u>	h				`` ``` `	,			
	-15-		177	8				· ·····		,
		HH	<u>,</u>			Sama	· · ·			
		4		a						
		H	151	1		Sama-				
		Way		A		5				
			2.00	101		Di .				17.0
	- 20 -		80.	B		Black	Peot	·		200
	<u> </u>			l					Better of	Baring
										·
						- pii	Samolo	schal	La Lunth Pin h	
	<u> </u>			ł	<u>-</u>	<u> </u>	Sand	C. I	Ala A	/
				ŀ		110	-umples	sallar	HUI VIS Calde	1
	- 25-						·····			
	- ŇOTE	S: 1 - Length red	covered; 0	" Loss	of Sam	ple, T	Trap used			•

2 - U = undisturbed; A = auger; OER = open and rod; V = vane 3 - Log depth of change in color of wash water, loss of water, artesian water, sand heave in casing, etc.

PORT AUTHORITY OF NY & NJ Engineering Department - Materials Division



THE PORT AUTHORITY OF N.Y & N.J.

ENGINEERING DEPARTMENT MATERIALS ENGINEERING DIVISION PID READINGS

)		•				Sh	eet 3 o	17
PROJECT	HHMT-	Port Ivony	Surcha	the Tors				
BORING N	10. PG-5	T-45			DATE:	Wiston		
RELD REA	DINGS BY:	OHoure			PID Model:	Mill RAFE		
TIME	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Space Reading	BREATHING Zone Reading	3	REMARK	S	
AM	IA.		0.0					
	18		0.0					
	2A		00				· · ·	
	28		0.0				· · ·	
	3		0.0					
	4		0.0					
Ţ	5		0.0			······································		·
PM	6		0.0		· · · · · · · · · · · · · · · · · · ·			
1	7		0.0					
	8		0.0			,		
	9		0.0		·	· · · · · · · · · · · · · · · · · · ·		
	lot		0.0			-	· ·	
	108		0.0			•		
	ľ					•		
		· · ·				-		
								.
				<u> </u>			، خــــــ	

-

PA 547
6-90

THE PORT AUTHORITY OF NY & NJ

Engineering Department Construction Division Materials Engineering Section

BORING REPORT

		·····							SHEET / OF 3
PROJECT	0	۲	,	T		NAME OF CONTRACTOR		BORING NO.	SURFACE ELEV.
<u>AHMT</u>	-Port Li	101.y 30	it char	Q.10	57-	Craig Stilling	}	PG- 57 - 41	DATE
	1. + 6		<i>D</i> .		—	t s.l.		421.98-C	Id/20100
SPOON	CUI INT	CASING	SIZE	HOLE T	(PE		80	DIND WATER LEVEL	11 01102
2	Ple	ID HU		Bina	Dor	Date Time	Depth	OUND WATER LEVEL	Remarks
HAMMER	Acto	НАММЕ							
140 #	FALL 30	-	# FAL	L	•	10/2/02 9	3.0	while Hand A	top sidd
DRILLER	~								Jack Strategy and
	V	Cocke							
INSPECTOR	Ĺ	House							
CASING	DEDTH	SPOON	R	E- 1 S	AMP,2	3(SAMPLE DE	SCRIPTION AND REMA	RKS
May 1		Hand D		U I	<u>но.</u> Д	FII- RCA 6	Carles	Cauge T. Mag	
8 iurof		orana prog			15		ecycled.	CONCESS TES	
Huger					3	Fill Crushed 5	Jone_		
Ň					>8	Same			3.0
					- p	Ell What the		Terle	
		<u> </u>				<u></u>	ray Lime	Store Sluth	· · ·
	• 5 •	d/			3				
					/	Samo			
111.	- 6 -	1					· · · · · · · · · · ·		
na		1	ⁱ						
-951mg									
						Ne Sam	plue 6	-20 Soolor f	7 Up
		1		.		·PC ST	110 6	T. A. 1. 2.	
	≥.20.~						73_70	5/14/2 6-80	
		A. A.	_	·	Ч				
		11-2	2:	3"	1	Brown & Black	Part		
		1-2							
					5				
		2-3	80			Same Tr F OK	- Gray San	div Tipot spoo	e24.0
		hour	1		/		,		-
	- 25° -	Usert - 7	19	4	6	FNLCC	1 7.5		
		2001 2			·	- VE UNAL DON	4 <u>1</u> 2		
			_	<u> </u>		AIL Sompto_	checter_	4 The PID NoT-2	
			1			No Samples	Saund	All Discordant	
11			T						
		<u> </u>							
_ <u>_</u> _	- 30 -								30
	50	2-3			7				
		4-3	1-	2/1	/	PRX	1 +		
		ļ/				urour sand	لالالالا		
	- 33								
		3-3							
1/		8-8	12	1	8	12 Ru 1/	5.17	The Suit	
Y	- 354				ليسب	i stown o ury	-und, I	- J/15	



EH INT INJ

-

CH NT NJ

-

THE PORT AUTHORITY OF N.Y & N.J.

ENGINEERING DEPARTMENT

MATERIALS ENGINEERING DIVISION

PID READINGS

ELD REAL	DINGS BY:	Pola			PID Model:	Mr. PAE	
τιμε	SAMPLE No.	IN-SITU Split Spoon Reading	HEAD- Space Reading	BREATHING Zone Reading		REMARKS	
Pin	1.2		0.0				
	18		0.0				
	2A		0.0			·	
	28		· 0.0 ·				•
•	3		U.a				:
	4		0.2		Pat		
	S.		0.3		Pat		
	6		0.0				
	7		0.0		· · ·		
V	8		0.0				
·	·	·			``````````````````````````````````````		
					·		
					ء مرتبع <u>محمد من المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحم</u>	· · · · · · · · · · · · · · · · · · ·	•
	j						
							i